Algebra & Number Theory

Volume 9 2015 _{No. 5}

Algebra & Number Theory

msp.org/ant

EDITORS

MANAGING EDITOR Bjorn Poonen Massachusetts Institute of Technology Cambridge, USA EDITORIAL BOARD CHAIR David Eisenbud University of California Berkeley, USA

BOARD OF EDITORS

Georgia Benkart	University of Wisconsin, Madison, USA	Susan Montgomery	University of Southern California, USA
Dave Benson	University of Aberdeen, Scotland	Shigefumi Mori	RIMS, Kyoto University, Japan
Richard E. Borcherds	University of California, Berkeley, USA	Raman Parimala	Emory University, USA
John H. Coates	University of Cambridge, UK	Jonathan Pila	University of Oxford, UK
J-L. Colliot-Thélène	CNRS, Université Paris-Sud, France	Anand Pillay	University of Notre Dame, USA
Brian D. Conrad	Stanford University, USA	Victor Reiner	University of Minnesota, USA
Hélène Esnault	Freie Universität Berlin, Germany	Peter Sarnak	Princeton University, USA
Hubert Flenner	Ruhr-Universität, Germany	Joseph H. Silverman	Brown University, USA
Sergey Fomin	University of Michigan, USA	Michael Singer	North Carolina State University, USA
Edward Frenkel	University of California, Berkeley, USA	Vasudevan Srinivas	Tata Inst. of Fund. Research, India
Andrew Granville	Université de Montréal, Canada	J. Toby Stafford	University of Michigan, USA
Joseph Gubeladze	San Francisco State University, USA	Ravi Vakil	Stanford University, USA
Roger Heath-Brown	Oxford University, UK	Michel van den Bergh	Hasselt University, Belgium
Craig Huneke	University of Virginia, USA	Marie-France Vignéras	Université Paris VII, France
Kiran S. Kedlaya	Univ. of California, San Diego, USA	Kei-Ichi Watanabe	Nihon University, Japan
János Kollár	Princeton University, USA	Efim Zelmanov	University of California, San Diego, USA
Yuri Manin	Northwestern University, USA	Shou-Wu Zhang	Princeton University, USA
Philippe Michel	École Polytechnique Fédérale de Lausan	ne	

PRODUCTION

production@msp.org

Silvio Levy, Scientific Editor

See inside back cover or msp.org/ant for submission instructions.

The subscription price for 2015 is US \$255/year for the electronic version, and \$440/year (+\$55, if shipping outside the US) for print and electronic. Subscriptions, requests for back issues and changes of subscribers address should be sent to MSP.

Algebra & Number Theory (ISSN 1944-7833 electronic, 1937-0652 printed) at Mathematical Sciences Publishers, 798 Evans Hall #3840, c/o University of California, Berkeley, CA 94720-3840 is published continuously online. Periodical rate postage paid at Berkeley, CA 94704, and additional mailing offices.

ANT peer review and production are managed by EditFLOW[®] from MSP.

PUBLISHED BY mathematical sciences publishers nonprofit scientific publishing

http://msp.org/ © 2015 Mathematical Sciences Publishers



p-adic Hodge-theoretic properties of étale cohomology with mod p coefficients, and the cohomology of Shimura varieties

Matthew Emerton and Toby Gee

We prove vanishing results for the cohomology of unitary Shimura varieties with integral coefficients at arbitrary level, and deduce applications to the weight part of Serre's conjecture. In order to do this, we show that the mod p cohomology of a smooth projective variety with semistable reduction over K, a finite extension of \mathbb{Q}_p , embeds into the reduction modulo p of a semistable Galois representation with Hodge–Tate weights in the expected range (at least after semisimplifying, in the case of the cohomological degree greater than 1).

1.	<i>p</i> -adic Hodge theoretic properties of mod <i>p</i> cohomology	1039
2.	Breuil modules with descent data	1056
3.	The cohomology of Shimura varieties	1061
4.	Group theory lemmas	1077
Appendix: Cohomology of pairs		1083
Acknowledgements		1085
References		1085

The aim of this paper is to establish vanishing results for the cohomology of certain unitary similitude groups. For example, we prove the following result:

Theorem A. Let X be a projective U(2, 1)-Shimura variety of some sufficiently small level, and let \mathcal{F} be a canonical local system of $\overline{\mathbb{F}}_p$ -vector spaces on X. Let \mathfrak{m} be a maximal ideal of the Hecke algebra acting on the cohomology $H^{\bullet}(X, \mathcal{F})$, and suppose that there is a Galois representation $\rho_{\mathfrak{m}} : G_F \to \operatorname{GL}_3(\overline{\mathbb{F}}_p)$ associated to \mathfrak{m} . If we suppose further that we have $\operatorname{SL}_3(k) \subset \rho_{\mathfrak{m}}(G_F) \subset \overline{\mathbb{F}}_p^{\times} \operatorname{SL}_3(k)$ for some finite extension k/\mathbb{F}_p , and that $\rho_{\mathfrak{m}}|_{G_{\mathbb{Q}_p}}$ is 1-regular and irreducible, then the localisations $H^i_{\text{ét}}(X_{\overline{\mathbb{Q}}}, \mathcal{F})_{\mathfrak{m}}$ vanish in degrees $i \neq 2$.

The first author was supported in part by NSF grant DMS-1003339, and the second author was supported in part by NSF grant DMS-0841491 and EPSRC Mathematics Platform grant EP/I019111/1. *MSC2010:* 11F33.

Keywords: p-adic Hodge theory, Shimura varieties.

(See Corollary 3.5.1 and Lemma 4.1.9 below, and see Sections 2 and 3 for the precise definitions that we are using; for simplicity we work with U(2, 1)-Shimura varieties¹ over a quadratic imaginary field F. Note that "sufficiently small level" means that the compact open subgroup defining the level is sufficiently small. We say that a Galois representation is associated to a maximal ideal of a Hecke algebra if there is the usual relation between Hecke polynomials and characteristic polynomials of Frobenius elements at unramified places; see Section 3.4 for a precise definition.)

In fact, we prove a version of this result for U(n - 1, 1)-Shimura varieties under weaker assumptions on ρ_m ; however, in general we can only prove vanishing in degrees outside of the range [n/2, (3n - 4)/2].

We also prove the following result, which makes no explicit reference to a maximal ideal in the Hecke algebra:

Theorem B. Let X and \mathcal{F} be as in the statement of Theorem A. If ρ is a 3dimensional irreducible sub- G_F -representation of the étale cohomology group $H^1_{\acute{e}t}(X_{\overline{\mathbb{Q}}}, \mathcal{F})$, then either every irreducible subquotient of $\rho|_{G_{\mathbb{Q}_p}}$ is 1-dimensional, or else $\rho|_{G_{\mathbb{Q}_p}}$ is not 1-regular, or else $\rho(G_F)$ is not generated by its subset of regular elements.

Note that in neither theorem do we make any assumption on the level of the Shimura variety at p.

A Galois representation ρ_m as in the statement of Theorem A is known to exist if m corresponds to a system of Hecke eigenvalues arising from the reduction mod p of the Hecke eigenvalues attached to some automorphic Hecke eigenform. Furthermore, recent work of Scholze [2013] (which appeared after the first version of this paper was written) implies that such a representation exists for *any* maximal ideal m.

It seems reasonable to believe that any irreducible sub- G_F -representation of any of the étale cohomology groups $H^i_{\acute{e}t}(X_{\overline{\mathbb{Q}}}, \mathcal{F})$ for any of the Shimura varieties under consideration should in fact be a constituent of ρ_m for some maximal ideal m of the Hecke algebra. However, this doesn't seem to be known, and relating the "abstract" G_F -representations ρ_m to the "physical" G_F -representations appearing on étale cohomology is one of the problems we have to deal with in proving our results.

Application to Serre-type conjectures. We are able to combine our results with those of [Emerton et al. 2013] so as to establish cases of the weight part of the Serre-type conjecture of [Herzig 2009] for U(2, 1). More precisely, we have the following result (where the assertion that $\bar{\rho}$ is modular means that the corresponding system of Hecke eigenvalues occurs in the mod p cohomology of some U(2, 1)-Shimura variety; see Theorem 3.5.6 and Lemma 4.1.9.)

¹These Shimura varieties might more properly be called GU(2, 1)-Shimura varieties; see Section 3 for their definition.

Theorem C. Suppose $\rho: G_F \to \operatorname{GL}_3(\overline{\mathbb{F}}_p)$ satisfies $\operatorname{SL}_3(k) \subset \rho(G_F) \subset \overline{\mathbb{F}}_p^{\times} \operatorname{SL}_3(k)$ for some finite extension k/\mathbb{F}_p , that $\rho|_{G_{\mathbb{Q}_p}}$ is irreducible and 1-regular, and that ρ is modular of some strongly generic weight. Then the set of generic weights for which ρ is modular is exactly the set predicted by the recipe of [Herzig 2009].

Relationship with a mod p analogue of Arthur's conjecture. Arthur [1989, §9] made a quite precise conjecture regarding the systems of Hecke eigenvalues that appear in the L^2 -automorphic spectrum of any reductive group over a number field, which has consequences for the nature of the Hecke eigenvalues appearing in the cohomology of Shimura varieties. For our purposes it suffices to describe a qualitative version of these consequences: namely, Arthur's conjecture implies that if λ is a system of Hecke eigenvalues appearing in the degree-*i* cohomology, where *i* is less than the middle dimension, then λ is attached (in the sense of, e.g., [Buzzard and Gee 2014; Johansson 2013]) to a reducible Galois representation (i.e., one which factors through a parabolic subgroup of the *L*-group).

The fragmentary evidence available suggests that a similar statement will be true for the mod p cohomology of Shimura varieties. Our Theorems A and B give further evidence in this direction.

p-adic Hodge theory. In order to prove these theorems, we establish some new results about the *p*-adic Hodge-theoretic properties of the étale cohomology of varieties over a number field or *p*-adic field with coefficients in a field of characteristic *p*. In the first section we establish results about the mod *p* étale cohomology of varieties over number fields or *p*-adic fields which, although weaker in their conclusions, are substantially broader in the scope of their application than previously known mod *p* comparison theorems. For example, we prove the following result (see Theorem 1.4.1 below):

Theorem D. Let K be a finite extension of \mathbb{Q}_p , and write G_K for the absolute Galois group of K. If X is a smooth projective variety over K which has semistable reduction, and if ρ is an irreducible subquotient of the G_K -representation $H^i_{\text{ét}}(X_{\overline{K}}, \overline{\mathbb{F}}_p)$, then ρ also embeds as a subquotient of a G_K -representation over $\overline{\mathbb{F}}_p$ which is the reduction modulo the maximal ideal of a G_K -invariant $\overline{\mathbb{Z}}_p$ -lattice in a G_K -representation over $\overline{\mathbb{Q}}_p$ which is semistable with Hodge–Tate weights contained in the interval [-i, 0].

Both the hypotheses and the conclusions of our theorems are rather precisely tailored to maximise (as far as we are able) their utility in applications to the analysis of Galois representations occurring in the cohomology of Shimura varieties, which we give in the third section.

The remaining two sections of the paper are devoted respectively to using integral *p*-adic Hodge theory (Breuil modules with descent data) to establish a result related

to the reductions of tamely potentially semistable *p*-adic representations of $G_{\mathbb{Q}_p}$ (Section 2) and to proving some technical results about group representations (Section 4). The result of Section 2 is an essential ingredient in the arguments of Section 3, while the results of Section 4 provide sufficient conditions for the various representation-theoretic hypotheses appearing in the results of Section 3 to be satisfied.

Remark on related papers. Very general vanishing theorems for the mod p cohomology of Shimura varieties have been proved by Lan and Suh [2013]; however, their results apply only in situations of good reduction and for coefficients corresponding to small Serre weights, which makes them unsuitable for the kinds of applications we have in mind, e.g., to the weight part of Serre-type conjectures. In the ordinary case there is the work of Mokrane and Tilouine [2002, §9] in the Siegel case and Dimitrov [2005, §6.4] in the case of Hilbert modular varieties. Finally, in a recent preprint, Shin [2013] proved a general vanishing result for cohomology outside of middle degree for the part of the mod p cohomology which is supercuspidal at some prime $l \neq p$, by completely different methods from those of this paper. It seems plausible that, via the mod p local Langlands correspondence for GL_n(\mathbb{Q}_l), Shin's hypothesis could be interpreted as a condition on the restriction to a decomposition group at l of the relevant mod p Galois representations, whereas our conditions involve the restriction to a decomposition group at p, so our results appear to be complementary.

Conventions. For any field K we let G_K denote a choice of absolute Galois group of K.

If K is a finite field, then by a Frobenius element in G_K we will always mean a geometric Frobenius element. We extend this convention in an evident way to Frobenius elements at primes in Galois groups of number fields, and to Frobenius elements in Galois groups of local fields.

If K is a local field, then we denote by \mathbb{O}_K the ring of integers of K, by I_K the inertia subgroup of G_K , by W_K the Weil group of K (the subgroup of G_K consisting of elements whose reduction modulo I_K is an integral power of Frobenius), and by WD_K the Weil–Deligne group of K.

If K is a number field and v is a finite place of K, then we will write K_v for the completion of K at v and \mathbb{O}_{K_v} for its ring of integers. We will write $\mathbb{O}_{K,(v)}$ for the localisation of \mathbb{O}_K at the prime ideal v.

We will write $\overline{\mathbb{Z}}_p$ for the ring of integers in $\overline{\mathbb{Q}}_p$ (a fixed algebraic closure of \mathbb{Q}_p), and $\mathfrak{m}\overline{\mathbb{Z}}_p$ for the maximal ideal of $\overline{\mathbb{Z}}_p$.

We let ω denote the mod p cyclotomic character. We will denote a Teichmüller lift with a tilde, so that for example $\tilde{\omega}$ is the Teichmüller lift of ω .

We use the traditional normalisation of Hodge–Tate weights, with respect to which the cyclotomic character has Hodge–Tate weight 1.

By a *closed geometric point* \bar{x} of a scheme X, we mean a morphism of schemes \bar{x} : Spec $\Omega \to X$ for a separably closed field Ω , whose image is a closed point x of X, and such that the induced embedding $\kappa(x) \hookrightarrow \Omega$ (where $\kappa(x)$ denotes the residue field of x) identifies Ω with a separable closure of $\kappa(x)$. If \bar{x} is a closed geometric point of a Noetherian scheme X, then we let $\mathbb{O}_{X,\bar{x}}$ denote the local ring of X at \bar{x} , i.e., the stalk, in the étale topology on X, of the structure sheaf of X at \bar{x} ; we let $(\mathbb{O}_{X,\bar{x}})^{\wedge}$ denote the completion of $\mathbb{O}_{X,\bar{x}}$, and we write $(X_{\bar{x}})^{\wedge} := \text{Spf}((\mathbb{O}_{X,\bar{x}})^{\wedge})$, and refer to $(X_{\bar{x}})^{\wedge}$ as the formal completion of X along the closed geometric point \bar{x} .

The symbol G will always denote a group; in Section 3 it will be a certain algebraic group, and in Section 4 it will be a finite group.

1. *p*-adic Hodge theoretic properties of mod *p* cohomology

1.1. *Introduction.* We now describe in more detail our results on the integral *p*-adic Hodge theory of the étale cohomology of projective varieties, which are perhaps the most novel part of this paper.

It is well-known that integral *p*-adic Hodge theory is less robust than the corresponding theory with rational coefficients; for example, the comparison theorems for integral and mod *p* étale cohomology due to Fontaine and Messing [1987] and Faltings [1989] involve restrictions both on the degrees of cohomology and the dimensions of the varieties considered, and they also require that the field *K* be absolutely unramified and that the variety under consideration be of good reduction. More recently, Caruso [2008] has proved an integral comparison theorem in the case of semistable reduction for possibly ramified fields *K*, but there are still restrictions: his result requires that ei , where*e*is the absolute ramification index of*K*, and*i*is the degree of cohomology under consideration.

These restrictions are unfortunate, since mod p and integral p-adic Hodge theory are among the most powerful local tools available for the analysis of Galois representations occurring in the mod p étale cohomology of varieties. The premise that underlies the present work is that frequently in such applications one does not need a precise comparison theorem relating the mod p étale cohomology to an analogous structure involving mod p de Rham or crystalline cohomology. Rather, one often uses the comparison theorem merely to draw much less specific conclusions, such as that the Galois representations occurring in certain mod p étale cohomology spaces are in the essential image of the Fontaine–Laffaille functor, applied to Fontaine–Laffaille modules whose Fontaine–Laffaille numbers lie in some prescribed range. Our aim is to establish results of the latter type in more general contexts than they have previously been proved.

The precise direction of our work is informed to a significant extent by the fairly recent development of a rich *internal* integral *p*-adic Hodge theory, by Breuil

[2000], Kisin [2006], Liu [2008] and others. What we mean here by the word "internal" is that these developments have been directed not so much at applications to comparison theorems, but rather at the purely Galois-theoretic problem of giving a *p*-adic Hodge-theoretic description of Galois-invariant lattices in crystalline or semistable Galois representations, and of the mod *p* Galois representations that appear in the reductions of such lattices. These tools, especially the theory of *Breuil modules* [2000], which provides the desired description of the mod *p* representations arising as reductions of such lattices, have proved very useful in arithmetic applications. Because of the availability of these tools, it has become both possible and worthwhile to move beyond the Fontaine–Laffaille context in integral *p*-adic Hodge theory. While Caruso's work mentioned above is a significant step in this direction, an important aspect of the present work will be the consideration of situations in which the bound ei < p-1, required for the validity of the comparison theorem of [Caruso 2008], does not hold.

Our goal, then, is to establish in various situations that a Galois representation appearing in the mod p étale cohomology of a variety can be embedded in the reduction of a Galois-invariant lattice contained in a crystalline or semistable Galois representation, with Hodge–Tate weights lying in some specified range (namely, the range that one would expect given the degree of the cohomology space under consideration). Since, in arithmetic situations, one frequently has to make a ramified base change in order to obtain good or semistable reduction, and since the resulting descent data on the associated Breuil module typically then play an important role in whatever analysis has to be undertaken, we also prove results in certain cases of potentially semistable reduction which allow us to gain some control over these descent data.

The idea underlying our approach is very simple. Suppose that X is a variety over a p-adic field K. If i is some degree of cohomology, then we have a short exact sequence

$$0 \longrightarrow H^{i}_{\mathrm{\acute{e}t}}(X_{\overline{K}}, \mathbb{Z}_{p})/pH^{i}_{\mathrm{\acute{e}t}}(X_{\overline{K}}, \mathbb{Z}_{p}) \longrightarrow H^{i}_{\mathrm{\acute{e}t}}(X_{\overline{K}}, \mathbb{F}_{p}) \longrightarrow H^{i+1}_{\mathrm{\acute{e}t}}(X_{\overline{K}}, \mathbb{Z}_{p})[p] \longrightarrow 0,$$

as well as an isomorphism

$$\mathbb{Q}_p \otimes_{\mathbb{Z}_p} H^1_{\text{\'et}}(X_{\overline{K}}, \mathbb{Z}_p) \xrightarrow{\sim} H^1_{\text{\'et}}(X_{\overline{K}}, \mathbb{Q}_p).$$

Thus, if both $H^i_{\text{ét}}(X_{\overline{K}}, \mathbb{Z}_p)$ and $H^{i+1}_{\text{ét}}(X_{\overline{K}}, \mathbb{Z}_p)$ are torsion-free, then we see that $H^i_{\text{ét}}(X_{\overline{K}}, \mathbb{F}_p)$ is the reduction mod p of $H^i_{\text{ét}}(X_{\overline{K}}, \mathbb{Z}_p)$, which *is* a Galois-invariant lattice in $H^i_{\text{ét}}(X_{\overline{K}}, \mathbb{Q}_p)$. Furthermore, the usual comparison theorems of *rational* p-adic Hodge theory [Faltings 1989; Tsuji 1999] can be applied to conclude that this latter representation is, e.g., crystalline (if X is proper with good reduction) or semistable (if X is proper with semistable reduction).

The obstruction to implementing this idea is that we have no reason to believe in general that $H^i_{\text{ét}}(X_{\overline{K}}, \mathbb{Z}_p)$ and $H^{i+1}_{\text{ét}}(X_{\overline{K}}, \mathbb{Z}_p)$ will be torsion-free. To get around this difficulty, we engage in various dévissages using the weak Lefschetz theorem. To explain these, first consider the case when X is a projective curve and i = 1. In this case all the cohomology with \mathbb{Z}_p -coefficients is certainly torsion-free, and so $H^1_{\text{ét}}(X_{\overline{K}}, \mathbb{F}_p)$ is the reduction of a Galois-invariant lattice in $H^1_{\text{ét}}(X_{\overline{K}}, \mathbb{Q}_p)$. Now a simple induction using the weak Lefschetz theorem shows that for *any* smooth projective variety X over K there is an embedding

$$H^1_{\mathrm{\acute{e}t}}(X_{\overline{K}}, \mathbb{F}_p) \hookrightarrow H^1_{\mathrm{\acute{e}t}}(C_{\overline{K}}, \mathbb{F}_p),$$

where C is a smooth projective curve. Furthermore, if X has good (respectively semistable) reduction, we can ensure that the same is true of C. This gives the desired result in the case of H^1 (ignoring for a moment the problem of obtaining a refinement dealing with descent data in the potentially semistable case).

For higher degrees of cohomology, a more elaborate dévissage is required. The key point, again established via the weak Lefschetz theorem, is that if X is smooth and projective of dimension d, and if Y and Z are sufficiently generic hyperplane sections of X, then the cohomology of the pair $((X \setminus Y)_{\overline{K}}, (Z \setminus Y)_{\overline{K}})$, with either \mathbb{Z}_p or \mathbb{F}_p coefficients, vanishes in degrees other than d (see Section A.3 of the appendix; note that $X \setminus Y$ is affine), so that $H^d_{\acute{e}t}((X \setminus Y)_{\overline{K}}, (Z \setminus Y)_{\overline{K}}, \mathbb{Z}_p)$ is torsion-free and is thus a Galois-invariant lattice in $H^d_{\acute{e}t}((X \setminus Y)_{\overline{K}}, (Z \setminus Y)_{\overline{K}}, \mathbb{Q}_p)$, which is potentially semistable by [Yamashita 2011], and whose reduction is equal to $H^d_{\acute{e}t}((X \setminus Y)_{\overline{K}}, (Z \setminus Y)_{\overline{K}}, \mathbb{F}_p)$. Such relative cohomology spaces are the essential ingredient of the *basic lemma* of [Beilinson 1987], and we learned the idea of using them as building blocks for the cohomology of varieties from Nori [2002], who has used the basic lemma as the foundation of his approach to the construction of motives. Indeed, our present approach to integral *p*-adic Hodge theory was inspired by Beilinson's and Nori's work.

1.2. Bertini-type theorems. We begin by giving a straightforward generalisation of some of the results of [Jannsen and Saito 2012], which build on the results of [Poonen 2004] to prove Bertini-type theorems for varieties with semistable reduction over a discrete valuation ring. It will be convenient to allow K to denote either a number field, or a field of characteristic zero that is complete with respect to a discrete valuation with perfect residue field k_K of characteristic p. We abbreviate these two situations as "the global case" and "the local case" respectively, and in the former case we will let v denote a place of K dividing p.

We recall the following definition:

Definition 1.2.1. Suppose first that we are in the local case. We then say that a projective \mathbb{O}_K -scheme \mathscr{X} is *semistable* if it is regular and flat over $\operatorname{Spec} \mathbb{O}_K$, and if the special fibre \mathscr{X}_s is reduced and is a normal crossings divisor; equivalently, a finite-type \mathbb{O}_K -scheme \mathscr{X} is semistable if, at each closed geometric point \overline{x} of \mathscr{X}_s , there is an isomorphism of complete local rings

 $(\mathbb{O}_{\overline{x},\mathscr{X}})^{\wedge} \cong ((\mathbb{O}_{K}^{\mathrm{sh}})^{\wedge})[[x_{1},\ldots,x_{n}]]/(x_{1}\cdots x_{m}-\varpi_{K}),$

where $(\mathbb{O}_{K}^{sh})^{\wedge}$ is the completion of the strict Henselisation of \mathbb{O}_{K} (equivalently, the completion of the ring of integers in the maximal unramified algebraic extension of K), the element $\overline{\varpi}_{K}$ is a uniformiser of $(\mathbb{O}_{K}^{sh})^{\wedge}$, and $1 \leq m \leq n$. We say that \mathscr{X} is strictly semistable if it is semistable and if the special fibre \mathscr{X}_{s} is a strict normal crossings divisor.

Again in the local case, we say that a smooth projective *K*-scheme has *good reduction* if it admits a smooth projective model over \mathbb{O}_K , and that it has (*strictly*) *semistable reduction* if it admits an extension to a projective \mathbb{O}_K -scheme which is (strictly) semistable in the sense of the preceding definition.

In the global case, we say that a smooth projective K-scheme has good reduction at v if it admits a smooth projective model over $\mathbb{O}_{K,(v)}$, and that it has (strictly) semistable reduction at v if it admits a (strictly) semistable projective model over $\mathbb{O}_{K,(v)}$, i.e., a projective model over $\mathbb{O}_{K,(v)}$ whose base change over \mathbb{O}_{K_v} is (strictly) semistable in the sense of the preceding definition.

Remark 1.2.2. Note that our definition of a semistable \mathbb{O}_K -scheme (putting ourselves in the local case) includes the requirement that the scheme be regular. This is the definition that is frequently adopted in the theory of semistable reduction, and it is well-suited to our intended applications. Recall that, with this definition, semistability is *not* preserved under the base change to \mathbb{O}_L , if *L* is a finite extension of *K*, unless L/K is unramified or the original scheme is in fact smooth over \mathbb{O}_K ; see also Remark 1.5.2 below.

Proposition 1.2.3. Let X be a smooth projective variety over K with strictly semistable (respectively good) reduction (at v, in the global case). Then there are smooth hypersurface sections Y and Z of X (with respect to an appropriately chosen embedding of X into some projective space) such that Y and Z intersect transversely, and all of Y, Z, and $Y \cap Z$ have strictly semistable (respectively good) reduction (at v, in the global case).

Proof. We first handle the local case. Choose an extension \mathscr{X} of X to an \mathbb{O}_K -scheme that is projective and smooth (in the good reduction case) or strictly semistable (in the strictly semistable reduction case), and fix an embedding of \mathscr{X} into some projective space over \mathbb{O}_K . By Corollaries 0 and 1 of [Jannsen and Saito 2012]

(or, perhaps more precisely, by their proofs) we can find a hypersurface section \mathfrak{V} of \mathscr{X} such that \mathfrak{V} is again either smooth or strictly semistable over \mathbb{O}_K . We take Y to be the generic fibre of \mathfrak{V} . By Remark 0(ii), together with Lemma 1 and the remark immediately before Corollary 1, of [Jannsen and Saito 2012], we see that in order to find Z it is enough to check that, given a finite collection X_1, \ldots, X_n of smooth projective schemes in $\mathbb{P}^N_{/k_K}$, there is a common hypersurface section meeting each of them transversely. This is an immediate consequence of Theorem 1.3 of [Poonen 2004], taking the set U_P there to be the subset of the completion $\widehat{\mathbb{O}}_P$ consisting of the f such that f = 0 is transverse to each X_i at P. (Since this set contains all the f which do not vanish at P, and in particular contains all the f congruent, modulo the maximal ideal, to a particular choice of f, it has positive Haar measure.)

We now pass to the global case. Let \mathscr{X} be a smooth (in the good reduction case) or strictly semistable (in the strictly semistable reduction case) projective model of X over $\mathbb{O}_{K,(v)}$. Let P_d^* denote the projective space (over $\mathbb{O}_{K,(v)}$) of degree-d hypersurfaces in the ambient projective space containing \mathscr{X} . Applying the argument in the local case to the base change $\mathscr{X}_{/\mathbb{O}_{K_v}}$, we see that, for some $d \geq 1$, there is a K_v -valued point of P_d^* corresponding to a hypersurface section of $\mathscr{X}_{/\mathbb{O}_{K_v}}$ having either smooth or semistable intersection (depending on the case we are in) with $\mathscr{X}_{/\mathbb{O}_{K_v}}$. Furthermore, this point lies in an affinoid open subset of P_{d/K_v}^* (the preimage of an open set in the special fibre of P_d^*), all of whose points correspond to hyperplane sections of $\mathscr{X}_{/\mathbb{O}_{K_n}}$ with either smooth or strictly semistable intersection. (See Remark 0(i) and the proofs of Theorems 0 and 1 of [Jannsen and Saito 2012].) The set of K_v -points of this affinoid open set is a nonempty open subset of $P_d^*(K_v)$. Since K is dense in K_v , we see that this intersection also contains a K-point of P_d^* , which gives the required hypersurface section Y. We find the hypersurface section Z by applying the same argument.

1.3. Cohomology in degree 1. Our arguments in degree 1 are rather simpler than in general degree, so we warm up with this case. (In fact, our result in this case is slightly stronger than our result in general degree, as we do not need to semisimplify the representation, so this result is not completely subsumed by our later results in general degree.) Fix a prime p. Let K denote a field of characteristic zero, complete with respect to a discrete valuation, with ring of integers \mathbb{O}_K and residue field k, assumed to be perfect of characteristic p. Let \overline{K} denote an algebraic closure of K, and set $G_K := \text{Gal}(\overline{K}/K)$.

Theorem 1.3.1. If X is a smooth projective variety over K which has good (resp. strictly semistable) reduction, then $H^1_{\acute{e}t}(X_{\overline{K}}, \mathbb{F}_p)$ embeds G_K -equivariantly into the reduction modulo p of a G_K -invariant lattice in a crystalline (resp. semistable) p-adic representation of G_K whose Hodge–Tate weights are contained in [-1,0].

Proof. We proceed by induction on the dimension d of X. If $d \leq 1$ then $H^1_{\text{ét}}(X_{\overline{K}}, \mathbb{F}_p)$ is isomorphic to the reduction modulo p of $H^1_{\text{\acute{e}t}}(X_{\overline{K}}, \mathbb{Z}_p)$, and the latter space (being torsion-free, by virtue of our assumption on d) is in turn a lattice in $H^1_{\text{\acute{e}t}}(X_{\overline{K}}, \mathbb{Q}_p)$, which is crystalline or semistable, respectively, with Hodge–Tate weights lying in [-1, 0], by the main result of [Tsuji 1999].

Suppose now that d > 1. It follows from Corollary 0 (resp. Corollary 1) of [Jannsen and Saito 2012] that if X has good reduction (resp. strictly semistable reduction) then we may choose a smooth hypersurface section Y of X defined over K which has good (resp. strictly semistable) reduction. Our induction hypothesis applies to show that $H^1_{\text{ét}}(Y_{\overline{K}}, \mathbb{F}_p)$ embeds as a subobject of a G_K -representation over \mathbb{F}_p which is the reduction modulo p of a G_K -invariant lattice in a crystalline (resp. semistable) p-adic representation of G_K whose Hodge–Tate weights are contained in [-1, 0]. On the other hand, the weak Lefschetz theorem with \mathbb{F}_p -coefficients [SGA 4₃ 1973, Exposé XIV, Corollaire 3.3] implies that the natural (restriction) map $H^1_{\text{ét}}(X_{\overline{K}}, \mathbb{F}_p) \to H^1_{\text{ét}}(Y_{\overline{K}}, \mathbb{F}_p)$ is an embedding (because $1 \le d - 1$ by assumption). This completes the proof.

1.4. Cohomology of arbitrary degree. As always we fix a prime p. The necessary dévissages in this subsection will be more elaborate than in the previous one, and so, to maximise the utility of our results for later applications, it will be convenient to again allow K to denote either a number field or a field of characteristic zero that is complete with respect to a discrete valuation with perfect residue field of characteristic p. In applications it will also be useful to have flexibility in the choice of coefficients in the various cohomology spaces that we consider, and to this end we fix an algebraic extension E of \mathbb{Q}_p , with ring of integers \mathbb{O}_E and residue field k_E . (In applications, E will typically either be a finite extension of \mathbb{Q}_p , or else will be $\overline{\mathbb{Q}_p}$.)

We now recall some consequences of the weak Lefschetz theorem. Among other notions, we will use the étale cohomology of a pair consisting of a variety and a closed subvariety; a precise definition of this cohomology, and a verification of its basic properties (such as those recalled in the next paragraph), is included in the Appendix.

Let X be a smooth projective variety of dimension d over K, and suppose that Y and Z are two smooth hypersurface sections of X, chosen so that $Y \cap Z$ is also smooth. Let A denote either E, \mathbb{O}_E , or k_E . In either the first or last case, the spaces $H^i_{\text{ét}}((X \setminus Y)_{\overline{K}}, (Z \setminus Y)_{\overline{K}}, A)$ and $H^{2d-i}_{\text{ét}}((X \setminus Z)_{\overline{K}}, (Y \setminus Z)_{\overline{K}}, A)(d)$ are naturally dual to one another for each integer i. The weak Lefschetz theorem implies that the former space vanishes when i > d and the latter space vanishes when 2d - i > d, i.e., when i < d. Thus, in fact, both spaces vanish unless i = d. It then follows that both spaces vanish unless i = d in the case when A is taken to be \mathbb{O}_E as well, and hence that when i = d both spaces are torsion-free.

Let \overline{K} denote an algebraic closure of K, and set $G_K := \text{Gal}(\overline{K}/K)$. Now let $\rho: G_K \to \text{GL}_n(k_E)$ be irreducible and continuous. In the global case, we fix a place v of K lying over p, and a decomposition group $D_v \subset G_K$ for v.

Theorem 1.4.1. If X is a smooth projective variety over K which has strictly semistable (resp. good) reduction (at v, if we are in the global case), and if ρ embeds as a subquotient of $H^i_{\acute{e}t}(X_{\overline{K}}, k_E)$, then ρ also embeds as a subquotient of a G_K -representation over k_E which is the reduction modulo the uniformiser of a G_K -invariant \mathfrak{O}_E -lattice in a G_K -representation which is semistable (resp. crystalline) (at v, in the global case) with Hodge–Tate weights contained in the interval [-i, 0].

Proof. We proceed by induction on the dimension of X. Suppose initially that we are in the strictly semistable reduction case. By Proposition 1.2.3, we can and do choose smooth hypersurface sections Y and Z, having smooth intersection, and such that Y, Z, and $Y \cap Z$ all have strictly semistable reduction.

We then consider the long exact sequences

$$\cdots \longrightarrow H^{i}_{Y_{\overline{K}}, \acute{e}t}(X_{\overline{K}}, A) \longrightarrow H^{i}_{\acute{e}t}(X_{\overline{K}}, A) \longrightarrow H^{i}_{\acute{e}t}((X \setminus Y)_{\overline{K}}, A) \longrightarrow \cdots,$$

$$\cdots \longrightarrow H^{i}_{(Y \cap Z)_{\overline{K}}, \acute{e}t}(Z_{\overline{K}}, A) \longrightarrow H^{i}_{\acute{e}t}(Z_{\overline{K}}, A) \longrightarrow H^{i}_{\acute{e}t}((Z \setminus Y)_{\overline{K}}, A) \longrightarrow \cdots,$$

$$\cdots \longrightarrow H^{i}_{\acute{e}t}((X \setminus Y)_{\overline{K}}, (Z \setminus Y)_{\overline{K}}, A) \longrightarrow H^{i}_{\acute{e}t}((X \setminus Y)_{\overline{K}}, A) \longrightarrow H^{i}_{\acute{e}t}((Z \setminus Y)_{\overline{K}}, A) \longrightarrow \cdots,$$

with A taken to be either E or k_E (see [Milne 1980, Chapter III, Proposition 1.25] for the first two, which are local cohomology long exact sequences, and the Appendix for the third, which is the long exact sequence of the pair $(X \setminus Y, Z \setminus Y)$). We also recall (see [SGA 4₃ 1973, Exposé XIV, §3]) that there are canonical isomorphisms

$$\begin{aligned} H^{i-2}_{\acute{e}t}(Y_{\overline{K}},A)(-1) &\xrightarrow{\sim} H^{i}_{Y_{\overline{K}},\acute{e}t}(X_{\overline{K}},A), \\ H^{i-2}_{\acute{e}t}((Y \cap Z)_{\overline{K}},A)(-1) &\xrightarrow{\sim} H^{i}_{(Y \cap Z)_{\overline{K}},\acute{e}t}(Z_{\overline{K}},A). \end{aligned}$$

When A = E, all the cohomology spaces that appear are potentially semistable [Yamashita 2011]. Since $H^i_{\acute{e}t}(X_{\overline{K}}, E)$, $H^i_{\acute{e}t}(Y_{\overline{K}}, E)$, and $H^i_{\acute{e}t}(Z_{\overline{K}}, E)$ are semistable with Hodge–Tate weights lying in [-i, 0], we see that $H^i_{\acute{e}t}((X \setminus Y)_{\overline{K}}, E)$, $H^i_{\acute{e}t}((Z \setminus Y)_{\overline{K}}, E)$, and $H^i_{\acute{e}t}((X \setminus Y)_{\overline{K}}, E)$ are semistable, with Hodge–Tate weights lying in [-i, 0].

Now taking $A = k_E$, we see that, since ρ is irreducible, it appears as a subquotient of $H^{i-2}_{\text{ét}}(Y_{\overline{K}}, k_E)(-1)$, of $H^{i}_{\text{\acute{e}t}}(Z_{\overline{K}}, k_E)$, of $H^{i-1}(Y \cap Z)(-1)$, or of $H^{i}_{\text{\acute{e}t}}((X \setminus Y)_{\overline{K}}, (Z \setminus Y)_{\overline{K}}, k_E)$. In the first three cases, the theorem follows by induction on the dimension. In the final case, the conclusion follows from the vanishing theorem noted above; namely, $H^{i}_{\text{\acute{e}t}}((X \setminus Y)_{\overline{K}}, (Z \setminus Y)_{\overline{K}}, E)$ is the desired semistable representation, with invariant lattice $H^{i}_{\text{\acute{e}t}}((X \setminus Y)_{\overline{K}}, (Z \setminus Y)_{\overline{K}}, \mathbb{O}_E)$, whose reduction $H^{i}_{\text{\acute{e}t}}((X \setminus Y)_{\overline{K}}, (Z \setminus Y)_{\overline{K}}, k_E)$ contains ρ . Matthew Emerton and Toby Gee

Finally, suppose we are in the good reduction case. Again, by Proposition 1.2.3, we can and do choose smooth hypersurface sections Y and Z, having smooth intersection, such that Y, Z, and $Y \cap Z$ all have good reduction. Applying the same argument as in the previous paragraph, we see by induction on the dimension of X that it is enough to check that $H^i_{\text{ét}}((X \setminus Y)_{\overline{K}}, (Z \setminus Y)_{\overline{K}}, E)$ is crystalline, but this follows immediately from Theorem 1.2 of [Yamashita 2011]. (Note that if in the notation of that work we take $D^1 = Z$ and $D^2 = Y$, then by (A.1.2) below we see that $H^i_{\text{ét}}((X \setminus Y)_{\overline{K}}, (Z \setminus Y)_{\overline{K}}, E)$ appears on the left side of the Hyodo–Kato isomorphism in the statement of Theorem 1.2 of [Yamashita 2011], and since we have already shown that $H^i_{\text{ét}}((X \setminus Y)_{\overline{K}}, (Z \setminus Y)_{\overline{K}}, E)$ is semistable, it is enough to show that the monodromy operator N vanishes on the right side of the Hyodo–Kato isomorphism. This follows easily from the definition of this operator as a boundary map, as all objects concerned arise from base change from objects with trivial log structures.)

1.5. *Equivariant versions.* In practice, we will need equivariant analogues of the preceding results. As in the preceding section, we let K denote either a number field ("the global case") or a field of characteristic zero that is complete with respect to a discrete valuation with perfect residue field of characteristic p ("the local case"). We let \overline{K} denote an algebraic closure of K, and set $G_K := \text{Gal}(\overline{K}/K)$. In the global case, we fix a place v of K lying over p, and a decomposition group $D_v \subset G_K$ for v.

We now put ourselves in the following (somewhat elaborate) situation, which we call a *tamely ramified semistable context*, or a *tame semistable context* for short.

We suppose that X_0 and X_1 are smooth projective varieties over K, that G is a finite group which acts on X_1 , and that $\pi : X_1 \to X_0$ is a finite étale morphism which intertwines the given G-action on X_1 with the trivial G-action on X_0 , making X_1 an étale G-torsor over X_0 .

We suppose further that X_0 admits a semistable projective model \mathscr{X}_0 over \mathbb{O}_K (in the local case) or over $\mathbb{O}_{K,(v)}$ (in the global case). We also suppose that there is a finite Galois extension L of K, and (in the global case) a prime w of L lying over v, such that $(X_1)_{/L}$ admits a semistable projective model \mathscr{X}_1 over \mathbb{O}_L (in the local case) or over $\mathbb{O}_{L,(w)}$ (in the global case) to which the G-action extends, such that π extends to a morphism $\mathscr{X}_1 \to (\mathscr{X}_0)_{/\mathbb{O}_L}$ which intertwines the G-action on its source with the trivial G-action on its target, and such that the action of the (opposite group of) the inertia group $I(L/K)^{\text{op}}$ (or $I(L_w/K_v)^{\text{op}}$ in the global case) on $(X_1)_{/L}$ extends to an action on \mathscr{X}_1 .²

²Note that the tameness condition that we are going to require below ensures that L/K is in fact tamely ramified, and hence that I(L/K) is abelian. Thus passing to the opposite group is not actually necessary here when passing from the action on rings to the action on their Specs, but we will keep the superscript op in the notation for the sake of conceptual clarity.

Finally (and most importantly), we assume that the composite morphism

$$\mathscr{X}_1 \to (\mathscr{X}_0)_{/\mathbb{O}_L} \to \mathscr{X}_0$$
 (1.5.1)

(the first being the extension of π , and the second being the natural map) is tamely ramified along the special fibre $(\mathscr{X}_0)_s$, in the sense of [Grothendieck and Murre 1971, Definition 2.2.2].

In fact, in our applications we will consider the case that \mathscr{X}_0 is furthermore strictly semistable, in which case we will say that we are in a *tame strictly semistable context*.

Remark 1.5.2. The notion of a tame semistable context is somewhat rigid, as we will see in the following lemma, and would perhaps not be of much interest if it did not occur naturally in the Shimura variety context (as we will see Section 3.1). As one example of this rigidity, note that if G = 1, i.e., if X_0 and X_1 coincide, then the only way to achieve a tame semistable context is if L/K is unramified, or if \mathscr{X}_0 is smooth over \mathbb{O}_K . (Indeed, since a tamely ramified morphism is finite, and since the base change $\mathscr{X}_{0/\mathbb{O}_L}$ over the semistable \mathbb{O}_K -scheme \mathscr{X}_0 is normal, we see that if X_0 and X_1 coincide then the morphism $\mathscr{X}_1 \to \mathscr{X}_{0/\mathbb{O}_L}$ is necessarily an isomorphism. This implies that the semistable \mathbb{O}_K -scheme \mathscr{X}_0 has a semistable base change over \mathbb{O}_L , which, as we noted in Remark 1.2.2, is possible only if L/K is unramified or \mathscr{X}_0 is smooth over \mathbb{O}_K . Another point of view on this case is as follows: if \mathscr{X}_0 is semistable but not smooth, then in order to construct a semistable model \mathscr{X}_1 of $\mathscr{X}_{0/\mathbb{O}_L}$, we must perform some nontrivial blow-ups, and the resulting morphism $\mathscr{X}_1 \to \mathscr{X}_0$ is not finite, and in particular not tamely ramified.)

The following lemma gives a more concrete interpretation of the stipulation that (1.5.1) be tamely ramified along $(\mathscr{X}_0)_s$.

Lemma 1.5.3. In the above setting, the morphism (1.5.1) is tamely ramified along $(\mathscr{X}_0)_s$ if and only if the following conditions hold:

- (1) L (resp. L_w in the global case) is tamely ramified over K (resp. K_v in the global case), of ramification degree e, say.
- (2) For each closed geometric point \$\overline{x}_1\$ of the special fibre \$(\mathcal{X}_1)_s\$, with image \$\overline{x}_0\$ in \$(\mathcal{X}_0)_s\$, and for some choice of isomorphism

$$(\mathbb{O}_{\overline{x}_0,\mathscr{X}_0})^{\wedge} \cong (\mathbb{O}_K^{\mathrm{sh}})^{\wedge} \llbracket x_1, \dots, x_n \rrbracket / (x_1 \cdots x_m - \varpi_K), \tag{1.5.4}$$

where ϖ_K is a uniformiser of $(\mathbb{O}_K^{sh})^{\wedge}$ and $1 \leq m \leq n$, there is a corresponding isomorphism

$$(\mathbb{O}_{\overline{x}_1,\mathscr{X}_1})^{\wedge} \cong (\mathbb{O}_L^{\mathrm{sh}})^{\wedge} \llbracket y_1, \ldots, y_n \rrbracket / (y_1 \cdots y_m - \varpi_L),$$

where $\overline{\varpi}_L$ is a uniformiser of $(\mathbb{O}_L^{sh})^{\wedge}$, such that the induced morphism

$$((\mathscr{X}_1)_{\bar{x}_1})^{\wedge} \to ((\mathscr{X}_0)_{\bar{x}_0})^{\wedge}$$

is defined by the formula $x_j = y_j^e$ for $1 \le j \le m$ and $x_j = y_j$ for $m < j \le n$. Furthermore, if these equivalent conditions hold, then condition (2) holds for every

choice of isomorphism (1.5.4).

Proof. We first note that if we are in the global case, then, relabelling K_v as K and L_w as L, we may reduce ourselves to proving the lemma in the local case. Thus we assume that we in the local case from now on.

If conditions (1) and (2) (for some choice of isomorphism (1.5.4)) hold, then the morphism (1.5.1) is certainly tamely ramified along $(\mathscr{X}_0)_s$. (This amounts to the claim that we can verify tame ramification by passing to formal completions of closed geometric points, which is indeed the case, as follows from [Grothendieck and Murre 1971, Corollary 4.1.5].)

Suppose conversely that (1.5.1) is tamely ramified along $(\mathscr{X}_0)_s$. Since this morphism factors through the natural morphism $(\mathscr{X}_0)_{/\mathbb{O}_L} \to \mathscr{X}_0$, it follows from [Grothendieck and Murre 1971, Lemma 2.2.5] that this latter morphism is tamely ramified, and hence (e.g., by Proposition 2.2.9 of that reference, although our particular situation is much simpler than the general case of faithfully flat descent for tamely ramified covers considered in that proposition) that Spec $\mathbb{O}_L \to \text{Spec }\mathbb{O}_K$ is tamely ramified, i.e., that *L* is tamely ramified over *K*, of some ramification degree *e*. Thus condition (1) holds.

Now choose a closed geometric point \overline{x}_1 of $(\mathscr{X}_1)_s$ lying over the closed geometric point \overline{x}_0 of $(\mathscr{X}_0)_s$, and fix an isomorphism of the form (1.5.4). Since $\mathscr{X}_1 \to \mathscr{X}_0$ is tamely ramified along the divisor $\overline{w}_K = 0$ of \mathscr{X}_0 , Abhyankar's lemma [SGA 1 1971, Exposé XIII, Corollaire 5.6] (see also [Grothendieck and Murre 1971, Theorem 2.3.2] for a concise statement) implies that we may find regular elements $\{a_j\}_{j=1,...,k}$ of $(\mathbb{O}_K^{\text{sh}})^{\wedge}[[x_1,\ldots,x_n]]/(x_1\cdots x_m-\overline{w}_K)$ such that $a_1\cdots a_k$ generates the ideal (\overline{w}_K) of $(\mathbb{O}_K^{\text{sh}})^{\wedge}[[x_1,\ldots,x_n]]/(x_1\cdots x_m-\overline{w}_K)$, exponents e_1,\ldots,e_k all coprime to p, and a subgroup $H \subset \mu_{e_1} \times \cdots \times \mu_{e_k}$, such that the $(\mathbb{O}_K^{\text{sh}})^{\wedge}[[x_1,\ldots,x_m-\overline{w}_K)$ -algebra $(\mathbb{O}_{\overline{x}_1})^{\wedge}$ is isomorphic to

$$((\mathbb{O}_{K}^{\mathrm{sh}})^{\wedge}[[x_{1},\ldots,x_{n}]][T_{1},\ldots,T_{k}]/(x_{1}\cdots x_{m}-\varpi_{K},T_{1}^{e_{1}}-a_{1},\ldots,T_{k}^{e_{k}}-a_{k}))^{H}.$$

(Here $\mu_{e_1} \times \cdots \times \mu_{e_k}$, and hence *H*, acts on

$$(\mathbb{O}_{K}^{\mathrm{sh}})^{\wedge}[[x_{1},\ldots,x_{n}]][T_{1},\ldots,T_{k}]/(x_{1}\cdots x_{m}-\varpi_{K},T_{1}^{e_{1}}-a_{1},\ldots,T_{k}^{e_{k}}-a_{k})$$

in the obvious manner: namely, an element $(\zeta_1, \ldots, \zeta_k)$ acts on T_j via multiplication by ζ_j .)

Since each e_j is prime to p and $(\mathbb{O}_K^{\text{sh}})^{\wedge}[[x_1, \ldots, x_n]]/(x_1 \cdots x_m - \varpi_K)$ is strictly Henselian, any unit in this ring has an e_j -th root, and thus we are free to multiply any of the a_j by a unit. Consequently, we may assume that in fact $a_1 \cdots a_k = \varpi_K = x_1 \cdots x_m$, and hence (again taking advantage of our freedom to modify the a_j by

units) that $\{1, ..., m\}$ is partitioned into k sets $J_1, ..., J_k$, such that $a_j = \prod_{i \in J_j} x_i$. Now if we extract the e_j -th roots of each x_i for $i \in J_j$, the resulting extension of $(\mathbb{O}_K^{sh})^{\wedge}[[x_1, ..., x_n]]/(x_1 \cdots x_m - \varpi_K)$ contains

$$(\mathbb{O}_{K}^{\mathrm{sh}})^{\wedge}[[x_{1},\ldots,x_{n}]][T_{1},\ldots,T_{k}]/(x_{1}\cdots x_{m}-\varpi_{K},T_{1}^{e_{1}}-a_{1},\ldots,T_{k}^{e_{k}}-a_{k});$$

thus it is no loss of generality to assume that k = m and that $a_j = x_j$, and so we conclude that $(\mathbb{O}_{\overline{x}_1,\mathscr{X}_1})^{\wedge}$ is isomorphic, as an $(\mathbb{O}_K^{sh})^{\wedge}[[x_1,\ldots,x_n]]/(x_1\cdots x_m-\varpi_K)$ -algebra, to

$$((\mathbb{O}_{K}^{\mathrm{sh}})^{\wedge}[[x_{1},\ldots,x_{n}]][T_{1},\ldots,T_{m}]/(x_{1}\cdots x_{m}-\varpi_{K},T_{1}^{e_{1}}-x_{1},\ldots,T_{m}^{e_{m}}-x_{m}))^{H},$$

for some subgroup $H \subset \mu_{e_1} \times \cdots \times \mu_{e_m}$.

Let I_j denote the subgroup $1 \times \cdots \times \mu_{e_j} \times \cdots \times 1$ of $\mu_{e_1} \times \cdots \times \mu_{e_m}$; this is the inertia group of the divisor (x_j) with respect to the cover

$$\operatorname{Spec}((\mathbb{O}_{K}^{\operatorname{sh}})^{\wedge})[[x_{1},\ldots,x_{n}]][T_{1},\ldots,T_{m}]/(x_{1}\cdots x_{m}-\varpi_{K},T_{1}^{e_{1}}-x_{1},\ldots,T_{m}^{e_{m}}-x_{m})$$

$$\rightarrow \operatorname{Spec}((\mathbb{O}_{K}^{\operatorname{sh}})^{\wedge})[[x_{1},\ldots,x_{n}]]/(x_{1}\cdots x_{m}-\varpi_{K}).$$

If we write $H_j = H \cap I_j$, then $H' := H_1 \times \cdots \times H_m$ is a subgroup of H, and the cover

$$Spec((\mathbb{O}_{K}^{sh})^{\wedge}[[x_{1},...,x_{n}]][T_{1},...,T_{m}]/(x_{1}...x_{m}-\varpi_{K},T_{1}^{e_{1}}-x_{1},...,T_{m}^{e_{m}}-x_{m}))^{H'} \rightarrow Spec((\mathbb{O}_{K}^{sh})^{\wedge}[[x_{1},...,x_{n}]][T_{1},...,T_{m}]/(x_{1}...x_{m}-\varpi_{K},T_{1}^{e_{1}}-x_{1},...,T_{m}^{e_{m}}-x_{m}))^{H}$$

is unramified in codimension 1. Since \mathscr{X}_1 is regular, being semistable over \mathbb{O}_L , so is the target of this map (since we recall that this target is isomorphic to $((\mathscr{X}_1)_{\overline{x}_1})^{\wedge})$). The purity of the branch locus then implies that this cover is étale, and hence is an isomorphism (since its target is strictly Henselian). Consequently H = H'.

If we write

$$H_j = 1 \times \cdots \times \mu_{e'_j} \times \cdots \times 1 \subset 1 \times \cdots \times \mu_{e_j} \times \cdots \times 1 = I_j,$$

and set $d_j = e_j / e'_j$ and $S_j = T_j^{e'_j}$, then we conclude that $(\mathbb{O}_{\overline{x}_1, \mathscr{X}_1})^{\wedge}$ $\cong ((\mathbb{O}_K^{\text{sh}})^{\wedge} [[x_1, \dots, x_n]] [T_1, \dots, T_m] / (x_1 \cdots x_m - \varpi_K, T_1^{e_1} - x_1, \dots, T_m^{e_m} - x_m))^{H'}$ $\cong (\mathbb{O}_K^{\text{sh}})^{\wedge} [[x_1, \dots, x_n]] [S_1, \dots, S_m] / (x_1 \cdots x_m - \varpi_K, S_1^{d_1} - x_1, \dots, S_m^{d_m} - x_m).$

Now \mathscr{X}_1 is an \mathbb{O}_L -scheme with reduced special fibre (again because it is semistable over \mathbb{O}_L). Since $(\mathbb{O}_{\overline{x}_1,\mathscr{X}_1})^{\wedge}$ is strictly Henselian, it contains $(\mathbb{O}_L^{sh})^{\wedge}$, and we may choose a uniformiser ϖ_L of this ring such that $\varpi_L^e = \varpi_K$. Looking at the above description of $(\mathbb{O}_{\overline{x}_1,\mathscr{X}_1})^{\wedge}$, and taking into account that its reduction modulo ϖ_L

1050

must be reduced, we see that this special fibre must be the zero locus of the element $S_1 \cdots S_m$, hence that $S_1 \cdots S_m = u \varpi_L$ for some unit u, and thus that $(S_1 \cdots S_m)^e = u^e \varpi_K$. We conclude that $d_1 = \cdots = d_m = e$ and that $u^e = 1$, and hence, replacing ϖ_L by $u \varpi_L$, we find that $S_1 \cdots S_m = \varpi_L$. This shows that (2) holds (for our given choice of isomorphism (1.5.4)).

Remark 1.5.5. Note that we could avoid the appeal to the general theory of tame ramification (in particular, to Abhyankar's lemma) by just directly stipulating in our context that conditions (1) and (2) of Lemma 1.5.3 hold; indeed, in the proof of Theorem 1.5.15 below, we will work directly with these conditions, and in our applications to Shimura varieties we will also see directly that these conditions hold. Nevertheless, we have included Lemma 1.5.3 as an assurance to ourselves (and perhaps to the reader) that these conditions are somewhat natural.

Lemma 1.5.6. In a tame strictly semistable context as above, the \mathbb{O}_L -scheme \mathscr{X}_1 is also strictly semistable.

Proof. Since \mathscr{X}_1 is semistable by assumption, it is enough to show that the components of the special fibre $(\mathscr{X}_1)_s$ are regular. Suppose that D is a nonregular component of $(\mathscr{X}_1)_s$, and let \overline{x}_1 be a closed geometric point of D whose local ring on D is not regular. If we let \overline{x}_0 denote the image of \overline{x}_1 in $(\mathscr{X}_0)_s$, then Lemma 1.5.3(2) shows that we may find isomorphisms $(\mathbb{O}_{\overline{x}_1,(\mathscr{X}_1)_s})^{\wedge} \cong \overline{\mathbb{O}_K/\varpi_K}[[y_1,\ldots,y_n]]/(y_1\cdots y_m)$ and $(\mathbb{O}_{\overline{x}_0,(\mathscr{X}_0)_s})^{\wedge} \cong \overline{\mathbb{O}_K/\varpi_K}[[x_1,\ldots,x_n]]/(x_1\cdots x_m)$, with $1 \le m \le n$, such that the morphism $(\mathbb{O}_{\overline{x}_1,(\mathscr{X}_1)_s})^{\wedge} \to (\mathbb{O}_{\overline{x}_0,(\mathscr{X}_0)_s})^{\wedge}$ is given by $x_j = y_j^e$ for $1 \le j \le m$ and $x_j = y_j$ for $m < j \le n$. Since by assumption \overline{x}_1 is not a regular point of D, we find that necessarily $m \ge 2$, and that (possibly after permuting indices) there is an isomorphism $(\mathbb{O}_{\overline{x}_1,D})^{\wedge} \cong \overline{\mathbb{O}_K/\varpi_K}[[y_1,\ldots,y_n]]/(y_1\cdots y_{m'})$, where $2 \le m' \le m$. If we let D' denote the image of D in $(\mathscr{X}_0)_s$, we conclude that there is an isomorphism $(\mathbb{O}_{\overline{x}_0,D'})^{\wedge} \cong \overline{\mathbb{O}_K/\varpi_K}[[x_1,\ldots,x_n]]/(x_1\cdots x_{m'})$, and thus that D' is not regular. Hence \mathscr{X}_0 is not strictly semistable, a contradiction.

We now suppose that we are in a tame semistable context, as described above, and suppose for the moment that we are in the local case. Then we have an action of $I(L/K)^{\text{op}} \times G$ on the special fibre $(\mathscr{X}_1)_s$. Let D be an irreducible component of the special fibre of $(\mathscr{X}_0)_s$, and let \tilde{D} denote its preimage in $(\mathscr{X}_1)_s$, so that \tilde{D} is an $I(L/K)^{\text{op}} \times G$ -invariant union of irreducible components of $(\mathscr{X}_1)_s$.

Lemma 1.5.7. If G is abelian, then there is a homomorphism $\psi : I(L/K) \to G$ such that the action of $I(L/K)^{\text{op}}$ on \tilde{D} is given by composing the action of G with ψ ; i.e., if $i \in I(L/K)$, then the action of i on \tilde{D} coincides with the action of $\psi(i)$.

We first prove a general lemma:

Lemma 1.5.8. Let *S* be a connected Noetherian scheme, let *G* and *I* be finite groups, and let $f : T \to S$ be a finite étale morphism with the property that $I^{op} \times G$ acts on *T* over *S* in such a way that *T* becomes a *G*-torsor over *S*. If *G* is abelian, then there exists a morphism $\psi : I \to G$ such that the action of *I* on *T* is given by composing the action of *G* on *T* with the morphism ψ .

Proof. For clarity, we will not impose the assumption that G is abelian until required.

If we fix a geometric point \bar{s} of S, then the theory of the étale fundamental group [SGA 1 1971, Exposé V, Théorème 4.1] shows that passing to the fibre over \bar{s} gives an equivalence of categories between the category of finite étale covers of S and the category of (discrete) finite sets with a continuous action of $\pi_1(S, \bar{s})$. In this way, T is classified by an object P of this latter category equipped with an action of $I^{\text{op}} \times G$, with respect to which the G-action makes P a principal homogeneous G-set.

If we fix a base point $p \in P$, then we may identify P with G, thought of as a principal homogeneous G-set via left multiplication. As the automorphisms of G as a principal homogeneous G-set are naturally identified with G^{op} acting by right multiplication, we obtain a homomorphism $\psi_p : I^{op} \to G^{op}$, or equivalently a homomorphism $\psi_p : I \to G$, describing the action of I^{op} on P. If we replace p by $g \cdot p$ (for some $g \in G$), then one finds that $\psi_{gp} = g\psi_p g^{-1}$. Thus, if we now assume furthermore that G is abelian, then $\psi_p = \psi_{gp}$, and so it is reasonable in this case to write simply ψ for this homomorphism, which is well-defined independently of the choice of base point for P. Furthermore, when G is abelian, left and right multiplication by an element $g \in G$ coincide, and so the action of I^{op} on P is given by the formula $i \cdot p = \psi(i) \cdot p$ for all $p \in P$. Since the automorphisms of T over S induced by i and $\psi(i)$ coincide on P, they in fact coincide on all of T.

Proof of Lemma 1.5.7. The morphism $\mathscr{X}_1 \to (\mathscr{X}_0)_{/\mathbb{O}_L}$ is étale on generic fibres, and the explicit local formulas for this morphism provided by Lemma 1.5.3 show that it is in fact étale over an open subset \mathscr{U}_0 of $(\mathscr{X}_0)_{/\mathbb{O}_L}$ whose intersection with the special fibre $((\mathscr{X}_0)_{/\mathbb{O}_L})_s$ is Zariski dense. Replacing \mathscr{U}_0 with the intersection of all of its $I(L/K)^{\text{op}}$ -translates, we may furthermore assume that \mathscr{U}_0 is invariant under the action of $I(L/K)^{\text{op}}$ on $(\mathscr{X}_0)_{/\mathbb{O}_L}$.

If we let \mathcal{U}_1 denote the preimage of \mathcal{U}_0 in \mathcal{X}_1 , then \mathcal{U}_1 is invariant under the $I(L/K) \times G$ -action on \mathcal{X}_1 , and the morphism $\mathcal{U}_1 \to \mathcal{U}_0$ is a finite étale cover, for which the corresponding map $U_1 \to U_0$ on generic fibres realises U_1 as a *G*-torsor over U_0 . It follows that the *G*-action on \mathcal{U}_1 realises \mathcal{U}_1 as an étale *G*-torsor over \mathcal{U}_0 , and hence, passing to special fibres, that $(\mathcal{U}_1)_s$ is an étale *G*-torsor over $(\mathcal{U}_0)_s$.

Now the induced $I(L/K)^{\text{op}}$ -action on $(\mathfrak{U}_0)_s$ is trivial, and so $I(L/K)^{\text{op}}$ acts on $(\mathfrak{U}_1)_s$ as a group of automorphisms of the *G*-torsor $(\mathfrak{U}_1)_s$ over $(\mathfrak{U}_0)_s$. If $D' := D \cap (\mathfrak{U}_0)_s$, then D' is an irreducible component of $(\mathfrak{U}_0)_s$, and $\tilde{D}' := \tilde{D} \cap (\mathfrak{U}_1)_s$

is the restriction of $(\mathfrak{A}_1)_s$ to D'. Thus $\widetilde{D}' \to \widetilde{D}$ is again an étale *G*-torsor, with an action of $I(L/K)^{\text{op}}$ via automorphisms. Lemma 1.5.8 then shows that there is a homomorphism $\psi : I(L/K)^{\text{op}} \to G^{\text{op}}$, or equivalently a homomorphism $\psi : I(L/K) \to G$, such that the action of $I(L/K)^{\text{op}}$ on the points of \widetilde{D}' is given by composing the action of *G* with the homomorphism ψ . Since \widetilde{D} is equal to the Zariski closure of \widetilde{D}' in $(\mathscr{X}_1)_s$, the claim of the lemma follows. \Box

Lemma 1.5.9. Suppose that we are in a tame strictly semistable context. If $g \in I(L/K) \times G$ and D is a component of $(\mathcal{X}_1)_s$, then D and gD either coincide or are disjoint.

Proof. The images of *D* and *gD* in $(\mathscr{X}_0)_s$ coincide, and it follows from Lemma 1.5.3 that two distinct components of $(\mathscr{X}_1)_s$ that have nonempty intersection must have distinct images in $(\mathscr{X}_0)_s$.

If we now suppose that we are in the global case, then the discussion applies with L/K everywhere replaced by L_w/K_v , and in particular for each component D we may define a character $\psi : I(L_w/K_v) \to G$ describing the action of $I(L_w/K_v)$ on D.

Our next result describes how our tame semistable context behaves upon passage to a semistable hypersurface section of \mathscr{X}_0 . In its statement we assume for simplicity that we are in the local case.

Proposition 1.5.10. Suppose that we are in the tamely ramified semistable context described above, and let \mathfrak{V}_0 be a regular hypersurface section of \mathfrak{X}_0 such that the union of \mathfrak{Y}_0 and $(\mathfrak{X}_0)_s$ forms a divisor with normal crossings on \mathfrak{X}_0 . Let Y_0 denote the generic fibre of \mathfrak{Y}_0 , let Y_1 denote the preimage of Y_0 under the morphism $\pi : X_1 \to X_0$, and let \mathfrak{Y}_1 be the preimage of \mathfrak{Y}_0 under (1.5.1) (so that $(Y_1)_{/L}$ is the generic fibre of the \mathbb{O}_L -scheme \mathfrak{Y}_1). Then:

- (1) The complement of Y_1 in X_1 is affine.
- (2) The generic fibre Y₀ of 𝔅₀ is smooth over K, the morphism Y₁ → Y₀ is an étale G-torsor (so in particular Y₁ is also smooth over K), 𝔅₀ is a semistable model of Y₀ over 𝔅_K, 𝔅₁ is a semistable model for (Y₁)/L over 𝔅_L, and the morphism 𝔅₁ → 𝔅₀ is tamely ramified; consequently 𝔅₁ → 𝔅₀ is again a tamely ramified semistable context.
- (3) Suppose that G is abelian. If D' is an irreducible component of $(\mathfrak{V}_1)_s$, contained in an irreducible component D of $(\mathfrak{X}_1)_s$, then the homomorphism $\psi: I(L/K) \to G$, which describes the action of I(L/K) on D, also describes the action of I(L/K) on D'.

Proof. Since \mathfrak{V}_0 is a hypersurface section of \mathscr{X}_0 , its generic fibre Y_0 is a hypersurface section of X_0 . Thus its complement is affine. Since π is a finite morphism by assumption, the complement of Y_1 in X_1 is again affine. Since Y_0 is a regular projective *K*-scheme (being the generic fibre of \mathfrak{V}_0 , which is regular by assumption), it is in fact smooth over *K*. By definition, Y_1 is the preimage of Y_0 under the morphism $X_1 \to X_0$, which is an étale *G*-torsor by assumption. Thus $Y_1 \to Y_0$ is indeed an étale *G*-torsor (and so Y_1 is also smooth over *K*).

Let \overline{x}_0 denote a closed geometric point of the special fibre $(\mathfrak{Y}_0)_s$. Since $(((\mathfrak{X}_0)_{\overline{x}_0})^{\wedge})_s \cup ((\mathfrak{Y}_0)_{\overline{x}_0})^{\wedge}$ forms a divisor with normal crossings, since each component of $(((\mathfrak{X}_0)_{\overline{x}_0})^{\wedge})_s$ is regular, and since \mathfrak{Y}_0 is regular by assumption, it follows from [Grothendieck and Murre 1971, Lemma 1.8.4] that $(((\mathfrak{X}_0)_{\overline{x}_0})^{\wedge})_s \cup ((\mathfrak{Y}_0)_{\overline{x}_0})^{\wedge}$ is in fact a divisor with strictly normal crossings in $((\mathfrak{X}_0)_{\overline{x}_0})^{\wedge}$, and hence the local equation ℓ of $((\mathfrak{Y}_0)_{\overline{x}_0})^{\wedge}$, together with the elements x_1, \ldots, x_m that cut out the irreducible components of $(((\mathfrak{X}_0)_{\overline{x}_0})^{\wedge})_s$, form part of a regular system of parameters for $(\mathbb{O}_{\overline{x}_0,\mathfrak{X}_0})^{\wedge}$. Thus we may choose a model of the form (1.5.4) for $((\mathfrak{X}_0)_{\overline{x}_0})^{\wedge}$ is the zero locus of the element x_n .

We now choose a closed geometric point \overline{x}_1 of $(\mathscr{X}_1)_s$ lying over \overline{x}_0 , as well as a model for the tamely ramified morphism $((\mathscr{X}_1)_{\overline{x}_1})^{\wedge} \to ((\mathscr{X}_0)_{\overline{x}_0})^{\wedge}$ as in part (2) of Lemma 1.5.3. Thus this morphism has the form

$$\operatorname{Spec}((\mathbb{O}_L^{\operatorname{sh}})^{\wedge})[[y_1,\ldots,y_n]]/(y_1\cdots y_m-\varpi_L) \to \operatorname{Spec}((\mathbb{O}_K^{\operatorname{sh}})^{\wedge})[[x_1,\ldots,x_n]]/(x_1\cdots x_m-\varpi_K),$$

with $x_j = y_j^e$ for $1 \le j \le m$ and $x_j = y_j$ for $m < j \le n$. In particular, we see that $x_n = y_n$, and thus we see that the induced morphism

$$((\mathfrak{Y}_1)_{\overline{x}_1})^{\wedge} \to ((\mathfrak{Y}_0)_{\overline{x}_0})^{\wedge} \tag{1.5.11}$$

can be written as

$$\operatorname{Spec}((\mathbb{O}_{L}^{\operatorname{sh}})^{\wedge})[[y_{1},\ldots,y_{n-1}]]/(y_{1}\ldots y_{m}-\varpi_{L})$$

$$\rightarrow \operatorname{Spec}((\mathbb{O}_{L}^{\operatorname{sh}})^{\wedge})[[x_{1},\ldots,x_{n-1}]]/(x_{1}\ldots x_{m}-\varpi_{K}). \quad (1.5.12)$$

Thus we see that \mathfrak{Y}_0 and \mathfrak{Y}_1 are indeed semistable models of their generic fibres (over \mathbb{O}_K and \mathbb{O}_L respectively), and that the morphism $\mathfrak{Y}_1 \to \mathfrak{Y}_0$ is tamely ramified. This completes the verification of (2). The claim of (3) follows from the fact that the action of $I(L/K)^{\text{op}} \times G$ on $(\mathfrak{Y}_1)_s$ is the restriction of the corresponding action on $(\mathfrak{X}_1)_s$, together with the fact that any component of $(\mathfrak{Y}_1)_s$ is contained in a component of $(\mathfrak{X}_1)_s$.

We now suppose that *E* is an algebraic extension of \mathbb{Q}_p containing K_0 . Recall that if $\rho : G_K \to \operatorname{GL}_n(E)$ is a potentially semistable representation, then we may

attach a Weil–Deligne representation WD(ρ) to ρ by first passing to the potentially semistable Dieudonné module $D_{pst}(\rho)$ of ρ , which is a module over $E \otimes_{\mathbb{Q}_p} K_0$, then fixing an embedding $K_0 \hookrightarrow E$, and hence a projection pr : $E \otimes_{\mathbb{Q}_p} K_0 \to E$, and, finally, forming WD(ρ) := $E \otimes_{E \otimes_{\mathbb{Q}_p} K_0, pr} D_{pst}$. Although WD(ρ) depends on the choice of the embedding $K_0 \hookrightarrow E$, up to isomorphism it is independent of this choice, as the Frobenius ϕ on $D_{pst}(\rho)$ provides isomorphisms between the different choices. (See for example [Conrad et al. 1999, Appendix B] and [Taylor 2004, p. 78–79] for discussions of this construction and its properties.)

In the tame strictly semistable case, the following result will allow us to describe the inertial part of the Weil–Deligne representation associated to the *p*-adic étale cohomology of X_1 , or of a pair (X_1, Y_1) that arises in the context of the preceding proposition. Before stating the result we introduce some additional notation, and an additional assumption.

Assume that *G* is abelian, and let *J* denote the set of $I(L/K) \times G$ -orbits on the set of irreducible components of $(\mathscr{X}_1)_s$, and let D_j (for $j \in J$) denote the union of the components lying in the orbit labelled by *j*. Let $\psi_j : I(L/K) \to G$ be the homomorphism provided by Lemma 1.5.7, describing the action of I(L/K) on the points of D_j .

Proposition 1.5.13. Suppose that we are in a tame strictly semistable context as above. Either let W denote the Weil–Deligne representation associated to the potentially semistable G_K -representation $H^i_{\text{ét}}((X_1)_{/\overline{K}}, E)$, or else suppose that we are in the context of Proposition 1.5.10, and let W denote the Weil–Deligne representation associated to the potentially semistable G_K -representation $H^i_{\text{ét}}((X_1)_{/\overline{K}}, E)$, in either $H^i_{\text{et}}((X_1)_{/\overline{K}}, (Y_1)_{/\overline{K}})$, E (here i is some given degree of cohomology); in either case, W is a representation of the product $WD_K \times G$. Assume furthermore that G is abelian.

Then, if, as in the above discussion, \tilde{J} denotes the set of $I(L/K) \times G$ -orbits of irreducible components of $(\mathscr{X}_1)_s$, we may decompose W as a direct sum $W = \bigoplus_{\tilde{j} \in \tilde{J}} W_{\tilde{j}}$, such that on $W_{\tilde{j}}$ the action of the inertia group in W_K is obtained by composing the G-action on $W_{\tilde{j}}$ with the homomorphism $I_K \to I(L/K) \xrightarrow{\psi_{\tilde{j}}} G$.

Proof. Since the action of the inertia subgroup of W_K on W factors through a finite group, and representations of a finite group over a field of characteristic zero are semisimple, the claimed property of W is stable under the formation of subobjects, quotients, and extensions (in the category of $W_K \times G$ -representations). A consideration of the long exact sequence of cohomology associated to the pair (X_1, Y_1) (see the Appendix) then reduces the claim for the cohomology of the pair to the claim for the cohomology of X_1 and Y_1 individually. Since Proposition 1.5.10 shows that the strictly semistable model \mathfrak{P}_1 of $(Y_1)_{/L}$ behaves in an identical manner

to the strictly semistable model of \mathscr{X}_1 of $(X_1)_{/L}$, it in fact suffices to consider the case of X_1 .

Thus we now restrict our attention to the W_K -representation W underlying the potentially semistable Dieudonné module associated to $H^i_{\text{ét}}((X_1)_{/\overline{K}}, E)$. By [Tsuji 1999], this Dieudonné module is naturally identified with the log-crystalline cohomology $H^i((\mathscr{X}_1)_s^{\times}/W(k)^{\times}) \otimes_{W(k)} E$ of the special fibre $(\mathscr{X}_1)_s$ with its natural log-structure. Lemma 1.5.9 shows that if an intersection of distinct components of the special fibre $(\mathscr{X}_1)_s$ is nonempty, then the various components appearing must lie in mutually distinct orbits of $I(L/K) \times G$ acting on the set of components. Recalling that J denotes the indexing set for the collection $\{D_j\}_{j \in J}$ of $I(L/K) \times G$ -orbits of components of $(\mathscr{X}_1)_s$, this log-crystalline cohomology may be computed by the following spectral sequence of [Mokrane 1993]:

$$E_1^{-m,i+m} = \bigoplus_{\substack{l \ge \max\{0,-m\}\\ \{j_1,\dots,j_{2l+m+1}\} \subset J}} H_{\text{cris}}^{i-2l-m}(D_{j_1} \cap \dots \cap D_{j_{2l+m+1}}/W(k)) \otimes_{W(k)} E(-l-m)$$
$$\Longrightarrow H^i((\mathscr{X}_1)_s^{\times}/W(k)^{\times}) \otimes_{W(k)} E.$$

The constructions of [Tsuji 1999; Mokrane 1993] are both functorial, so that everything here is compatible with the $I(L/K) \times G$ -actions. Each of the summands in the E_1 -term is naturally an $I(L/K) \times G$ -representation, and furthermore the action of I(L/K) is given by the composite of the action of G with one of the characters $\psi_{\tilde{j}}$. Thus the E_1 -terms of this spectral sequence satisfy the claimed property of W. Thus W also satisfies this property, since it is obtained as a successive extension of subquotients of these E_1 -terms.

We are now ready to prove our equivariant versions of Theorems 1.3.1 and 1.4.1. For the first result, we place ourselves in the local case (since the global case immediately reduces to the local case by passing from K to K_v):

Theorem 1.5.14. Suppose that we are in the tame strictly semistable context described above. Then the $G_K \times G$ -representation $H^1_{\text{ét}}((X_1)_{\overline{K}}, \mathbb{F}_p)$ embeds $G_K \times G$ -equivariantly into the reduction modulo the uniformiser of a $G_K \times G$ -invariant \mathbb{O}_E -lattice in a representation V of $G_K \times G$ over E, having the following properties:

- (1) The restriction of V to G_L is semistable, with Hodge–Tate weights contained in the interval [-1, 0].
- (2) The Weil–Deligne representation associated to V, which is naturally a representation of $WD_K \times G$, when restricted to a representation of $I_K \times G$ can be written as a direct sum $\bigoplus_{\tilde{j} \in \tilde{J}} W_{\tilde{j}}$ of $I_K \times G$ -representations, where \tilde{j} runs over the same index set that labels the set of $I(L/K) \times G$ -orbits of irreducible

components of $(\mathscr{X}_1)_s$, such that on $W_{\tilde{j}}$ the action of the inertia group in W_K is obtained by composing the G-action on $W_{\tilde{j}}$ with the homomorphism $I_K \to I(L/K) \xrightarrow{\Psi_{\tilde{\tau}}} G.$

Proof. We follow the proof of Theorem 1.3.1, proceeding by descending induction on the dimension of X_0 and X_1 , and passing to appropriately chosen hypersurface sections \mathfrak{V}_0 of \mathfrak{X}_0 and their corresponding preimages Y_1 in X_1 and \mathfrak{V}_1 in $(\mathfrak{X}_1)_{/L}$. Taking into account Proposition 1.5.10, we thus reduce to the case when X_0 and X_1 are curves, so that $H^1_{\text{ét}}((X_1)_{/\overline{K}}, \mathbb{F}_p)$ is the reduction mod p of $H^1_{\text{ét}}((X_1)_{/\overline{K}}, \mathbb{Z}_p)$, which is in turn a lattice in $H^1_{\text{\acute{et}}}((X_1)_{/\overline{K}}, \mathbb{Q}_p)$. This latter representation is potentially semistable with Hodge–Tate weights in [-1, 0], by [Tsuji 1999]; the claim regarding Weil–Deligne representations follows from Proposition 1.5.13.

For our second result, we allow ourselves to be in either the local or global context.

Theorem 1.5.15. Suppose that we are in the tame strictly semistable context described above, and let $\rho: G_K \times G \to \operatorname{GL}_n(k_E)$ be an irreducible and continuous representation that embeds as a subquotient of $H^i_{\operatorname{\acute{e}t}}((X_1)_{\overline{K}}, k_E)$. Then ρ also embeds as a subquotient of a $G_K \times G$ -representation over k_E which is the reduction modulo the uniformiser of a $G_K \times G$ -invariant \mathbb{O}_E -lattice in a representation V of $G_K \times G$ over E, having the following properties:

- (1) The representation V becomes semistable when restricted to G_L (resp. the decomposition group $D_w \subset G_L$ in the global case), with Hodge–Tate weights contained in the interval [-i, 0].
- (2) The Weil–Deligne representation associated to V, which is naturally a representation of $WD_K \times G$ (resp. $WD_{K_v} \times G$ in the global case), when restricted to a representation of $I_K \times G$ (resp. $I_{K_v} \times G$ in the global case) can be written as a direct sum $\bigoplus_{\tilde{j}} W_{\tilde{j}}$ of $I_K \times G$ -representations (resp. of $I_{K_v} \times G$ representations), where \tilde{j} runs over the same index set that labels the set of $I(L/K) \times G$ -orbits (resp. of $I_{K_v} \times G$ -orbits) of irreducible components of $(\mathfrak{X}_1)_s$, such that on $W_{\tilde{j}}$, the action of the inertia group is obtained by composing the G-action on $W_{\tilde{j}}$ with the homomorphism $I_K \to I(L/K) \xrightarrow{\psi_{\tilde{j}}} G$ (resp. the homomorphism $I_{K_v} \to I(L_w/K_v) \xrightarrow{\psi_{\tilde{j}}} G$).

Proof. We can be proved in exactly the same way as Theorem 1.4.1, taking into account Propositions 1.5.10 and 1.5.13. \Box

2. Breuil modules with descent data

In this section we establish a result (Theorem 2.2.4) which imposes some constraints on the reductions of certain tamely potentially semistable *p*-adic representations of $G_{\mathbb{Q}_p}$.

2.1. *Preliminaries.* We begin by recalling some results from Section 3 of [Emerton et al. 2013]. To this end, let p be an odd prime, let $\overline{\mathbb{Q}}_p$ be a fixed algebraic closure of \mathbb{Q}_p , and let E and K be finite extensions of \mathbb{Q}_p inside $\overline{\mathbb{Q}}_p$. Assume that E contains the images of all embeddings $K \hookrightarrow \overline{\mathbb{Q}}_p$. Let K_0 be the maximal absolutely unramified subfield of K, so that $K_0 = W(k)[1/p]$, where k is the residue field of K. Let K/K' be a Galois extension, with K' a field lying between \mathbb{Q}_p and K. Assume further that K/K' is tamely ramified with ramification index e, and fix a uniformiser $\pi \in K$ with $\pi^e \in K'$. Let $E(u) \in W(k)[u]$ be the minimal polynomial of π over K_0 .

Let k_E be the residue field of E, and let $0 \le r \le p-2$ be an integer. Recall that the category k_E -BrMod^{*r*}_{dd} of *Breuil modules of weight r with descent data* from Kto K' and coefficients k_E consists of quintuples $(\mathcal{M}, \mathcal{M}_r, \varphi_r, \hat{g}, N)$, where:

- \mathcal{M} is a finitely generated $(k \otimes_{\mathbb{F}_p} k_E)[u]/u^{ep}$ -module, free over $k[u]/u^{ep}$.
- \mathcal{M}_r is a $(k \otimes_{\mathbb{F}_p} k_E)[u]/u^{ep}$ -submodule of \mathcal{M} containing $u^{er}\mathcal{M}$.
- $\varphi_r : \mathcal{M}_r \to \mathcal{M}$ is k_E -linear and φ -semilinear (where $\varphi : k[u]/u^{ep} \to k[u]/u^{ep}$ is the *p*-th power map) with image generating \mathcal{M} as a $(k \otimes_{\mathbb{F}_p} k_E)[u]/u^{ep}$ -module.
- $N : \mathcal{M} \to \mathcal{M}$ is $k \otimes_{\mathbb{F}_p} k_E$ -linear and satisfies N(ux) = uN(x) ux for all $x \in \mathcal{M}, u^e N(\mathcal{M}_r) \subset \mathcal{M}_r$, and $\varphi_r(u^e N(x)) = cN(\varphi_r(x))$ for all $x \in \mathcal{M}_r$. Here, $c = \overline{F}(u)^p \in (k[u]/u^{ep})^{\times}$, where $E(u) = u^e + pF(u)$ in W(k)[u].
- \$\heta\$: \$\mathcal{M}\$ → \$\mathcal{M}\$ are additive bijections for each \$g \in Gal(K/K')\$, preserving \$\mathcal{M}\$_r\$, commuting with the \$\varphi\$_r\$- and \$N\$-actions, and satisfying \$\hat{g}\$_1 \circ \$\hat{g}\$_2 = (\$g_1 \circ \$g_2\$)^\$ for all \$g_1, \$g_2 \in Gal(K/K')\$. Furthermore, if \$a \in k \otimes \$\varksim_p\$ \$k_E\$ and \$m \in \mathcal{M}\$ then \$\hat{g}\$(au^i m) = \$g(a)((\$g(\pi)/\pi)^i \otimes 1)u^i\$ \$\hat{g}\$(m)\$.

There is a covariant functor $T_{st}^{*,r}$ from k_E -BrMod^{*r*}_{dd} to the category of k_E -representations of $G_{K'}$.

Lemma 2.1.1. Suppose that $\mathcal{M} \in k_E$ -BrMod^r_{dd} and T' is a $G_{K'}$ -subrepresentation of $T_{st}^{*,r}(\mathcal{M})$ (so that in particular T' has the structure of a k_E -vector space). Then there is a unique subobject \mathcal{M}' of \mathcal{M} such that, if $f : \mathcal{M}' \to \mathcal{M}$ is the inclusion map, then $T_{st}^{*,r}(f)$ is identified with the inclusion $T' \hookrightarrow T_{st}^{*,r}(\mathcal{M})$. (Here \mathcal{M}' is a subobject of \mathcal{M} in the naive sense that it is a sub- $(k \otimes_{\mathbb{F}_p} k_E)[u]/u^{ep}$ -module of \mathcal{M} , which inherits the structure of an object of k_E -BrMod^r_{dd} from \mathcal{M} in the obvious way.)

Proof. This is Corollary 3.2.9 of [Emerton et al. 2013].

We now specialise to the particular situation of interest to us in this paper; namely, we let K_0 be the unique unramified extension of \mathbb{Q}_p of degree d, we take $K = K_0((-p)^{1/(p^d-1)})$, and we set $K' = K_0$, so that $e = p^d - 1$. Fix $\pi = (-p)^{1/(p^d-1)}$. We write $\tilde{\omega}_d : \operatorname{Gal}(K/K_0) \to K_0^{\times}$ for the character $g \mapsto g(\pi)/\pi$, and we let ω_d be the reduction of $\tilde{\omega}_d$ modulo π . (By inflation we can also think of $\tilde{\omega}_d$ and ω_d as characters of $I_{K_0} = I_{\mathbb{Q}_p}$. Note that ω_d is a tame fundamental character of niveau d and that $\tilde{\omega}_d$ is the Teichmüller lift of ω_d .) Note that when d = 1, we have $\omega_1 = \omega$, the mod p cyclotomic character.

Let φ be the arithmetic Frobenius on k, and let $\sigma_0: k \hookrightarrow k_E$ be a fixed embedding. Inductively define $\sigma_1, \ldots, \sigma_{d-1}$ by $\sigma_{i+1} = \sigma_i \circ \varphi^{-1}$; we will often consider the numbering to be cyclic, so that $\sigma_d = \sigma_0$. There are idempotents $e_i \in k \otimes_{\mathbb{F}_p} k_E$ such that if M is any $k \otimes_{\mathbb{F}_p} k_E$ -module, then $M = \bigoplus_i e_i M$, and $e_i M$ is the subset of M consisting of elements m for which $(x \otimes 1)m = (1 \otimes \sigma_i(x))m$ for all $x \in k$. Note that $(\varphi \otimes 1)(e_i) = e_{i+1}$ for all i.

If $\rho: G_{K_0} \to \operatorname{GL}_n(E)$ is a potentially semistable representation which becomes semistable over K, then the associated inertial type (that is, the restriction to I_{K_0} of the Weil–Deligne representation associated to ρ) is a representation of I_{K_0} which becomes trivial when restricted to I_K , so we can and do think of it as a representation of $\operatorname{Gal}(K/K_0) \cong I_{K_0}/I_K$.

Proposition 2.1.2. Maintaining our current assumptions on K, suppose that ρ : $G_{K_0} \to \operatorname{GL}_n(E)$ is a continuous representation whose restriction to G_K is semistable with Hodge–Tate weights contained in [0, r], where $r \leq p - 2$, and let the inertial type of ρ be $\chi_1 \oplus \cdots \oplus \chi_n$, where each χ_i is a character of I_{K_0}/I_K . If $\bar{\rho}$ denotes the reduction modulo \mathfrak{m}_E of a G_{K_0} -stable \mathbb{O}_E -lattice in ρ , then there is an element \mathcal{M} of k_E -BrMod^r_{dd}, admitting a $(k \otimes_{\mathbb{F}_p} k_E)[u]/u^{ep}$ -basis v_1, \ldots, v_n such that $\hat{g}(v_i) = (1 \otimes \overline{\chi}_i(g))v_i$ for all $g \in \operatorname{Gal}(K/K_0)$, and for which $T^{*,r}_{st}(\mathcal{M}) \cong \bar{\rho}$.

Proof. This is Proposition 3.3.1 of [Emerton et al. 2013]. (Note that the conventions on the sign of the Hodge–Tate weights in that work are the opposite of the conventions in this paper.) \Box

Lemma 2.1.3. Maintain our current assumptions on K, so that in particular we have $e = p^d - 1$. Then every rank-1 object of k_E -BrMod^r_{dd} may be written in the form

- $\mathcal{M} = ((k \otimes_{\mathbb{F}_p} k_E)[u]/u^{ep}) \cdot m,$
- $e_i \mathcal{M}_r = u^{r_i} e_i \mathcal{M},$

•
$$\varphi_r\left(\sum_{i=0}^{d-1} u^{r_i} e_i m\right) = \lambda m \text{ for some } \lambda \in (k \otimes_{\mathbb{F}_p} k_E)^{\times},$$

- $\hat{g}(m) = \left(\sum_{i=0}^{d-1} (\omega_d(g)^{k_i} \otimes 1)e_i\right) m$ for all $g \in \text{Gal}(K/K_0)$, and
- N(m) = 0.

Here the integers $0 \le r_i \le (p^d - 1)r$ and k_i satisfy $k_i \equiv p(k_{i-1} + r_{i-1}) \mod (p^d - 1)$ for all *i*. Conversely, any module \mathcal{M} of this form is a rank-1 object of k_E -BrMod^r_{dd}. Furthermore,

$$\mathbf{T}_{\mathrm{st}}^{*,r}(\mathcal{M})|_{I_{K_0}} \cong \sigma_0 \circ \omega_d^{\kappa_0},$$

where $\kappa_0 \equiv k_0 + p(r_0 p^{d-1} + r_1 p^{d-2} + \dots + r_{d-1})/(p^d - 1) \mod (p^d - 1)$.

Proof. This is Lemma 3.3.2 of [Emerton et al. 2013].

Remark 2.1.4. In the sequel, we will only be interested in the case that for each *i* we have $k_i = (1 + p + \dots + p^{d-1})x_i$ for some $0 \le x_i . In this case, the condition that <math>pr_i \equiv k_{i+1} - pk_i \mod (p^d - 1)$ implies that $r_i \equiv (1 + p + \dots + p^{d-1})(x_{i+1} - x_i) \mod (p^d - 1)$, so the condition that $0 \le r_i \le (p^d - 1)r$ means that we can write $r_i = (1 + p + \dots + p^{d-1})(x_{i+1} - x_i) + (p^d - 1)y_i$, with $0 \le y_i \le r$. An elementary calculation shows that we then have

$$\kappa_0 \equiv x_0 + y_0 + p^{d-1}(x_1 + y_1) + \dots + p(x_{d-1} + y_{d-1}) \mod (p^d - 1).$$

2.2. *Regularity.* Let \mathbb{Q}_{p^n} denote the unique unramified extension of \mathbb{Q}_p of degree *n*, with residue field \mathbb{F}_{p^n} . Regarding \mathbb{F}_{p^n} as a subfield of $\overline{\mathbb{F}}_p$, we may then regard ω_n as a character $I_{\mathbb{Q}_p} \to \overline{\mathbb{F}}_p^{\times}$.

Definition 2.2.1. Let $\bar{\rho}: G_{\mathbb{Q}_p} \to \operatorname{GL}_n(\overline{\mathbb{F}}_p)$ be an irreducible representation, so that $\bar{\rho} \cong \operatorname{Ind}_{G_{\mathbb{Q}_pn}}^{G_{\mathbb{Q}_p}} \chi$ for some character $\chi: G_{\mathbb{Q}_p} \to \overline{\mathbb{F}}_p^{\times}$. If we write

$$\chi|_{I_{\mathbb{Q}_p}} = \omega_n^{(a_0 + pa_1 + \dots + p^{n-1}a_{n-1})},$$

where each $a_i \in [0, p-1]$ and not all the a_i are p-1, then the multiset of *exponents* of $\overline{\rho}$ is defined to be the multiset of residues of the a_i in $\mathbb{Z}/p\mathbb{Z}$.

Definition 2.2.2. Let $\bar{\rho}: G_{\mathbb{Q}_p} \to \operatorname{GL}_n(\overline{\mathbb{F}_p})$ be a representation. Then the multiset of *exponents* of $\bar{\rho}$ is the union of the multisets of exponents of each of the Jordan–Hölder factors of $\bar{\rho}$.

Definition 2.2.3. Let $\bar{\rho} : G_{\mathbb{Q}_p} \to \operatorname{GL}_n(\overline{\mathbb{F}}_p)$ be a representation, and let r be a nonnegative integer. Then we say that $\bar{\rho}$ is r-regular if the exponents a_1, \ldots, a_n of $\bar{\rho}$ are such that the residues $a_i + k \in \mathbb{Z}/p\mathbb{Z}$, $1 \le i \le n, 0 \le k \le r+1$, are pairwise distinct.

The following theorem is the main result we will need from explicit *p*-adic Hodge theory:

Theorem 2.2.4. Let r be a nonnegative integer, and let $s : G_{\mathbb{Q}_p} \to \operatorname{GL}_m(\overline{\mathbb{Q}_p})$ be a potentially semistable representation with Hodge–Tate weights contained in the interval [0, r] and inertial type $\chi_1 \oplus \cdots \oplus \chi_m$. Suppose that there are (not necessarily distinct) integers $0 \le a_1, \ldots, a_n < p-1$ such that each χ_i is equal to some $\widetilde{\omega}^{a_j}$.

Suppose that $\bar{\rho}: G_{\mathbb{Q}_p} \to \mathrm{GL}_n(\overline{\mathbb{F}}_p)$ is a subquotient of \bar{s} , the reduction mod $\mathfrak{m}_{\mathbb{Q}_p}$ of some $G_{\mathbb{Q}_p}$ -stable $\overline{\mathbb{Z}}_p$ -lattice in s. Suppose also that

- det $\overline{\rho}|_{I_{\Omega,n}} = \omega^{a_1 + \dots + a_n + n(n-1)/2}$,
- $r \le (n-1)/2$, and
- p > n(n-1)/2 + 1.

If r = (n-1)/2 then assume further that some irreducible subquotient of $\overline{\rho}$ has dimension greater than 1. Then $\overline{\rho}$ is not r-regular.

Proof. Modifying the choice of $\overline{\mathbb{Z}}_p$ -lattice if necessary, it suffices to treat the case that $\overline{\rho}$ and \overline{s} are semisimple. Let $\overline{\rho}' \cong \operatorname{Ind}_{G_{\mathbb{Q}_pd}}^{G_{\mathbb{Q}_p}} \chi$ be an irreducible subrepresentation of $\overline{\rho}$. Take $K' = K_0 = \mathbb{Q}_{p^d}$ in the above notation, so that $e = p^d - 1$. Taking *E* to be sufficiently large so that *s* is defined over *E* and $\overline{\rho}$ is defined over k_E , and applying Proposition 2.1.2 to $s|_{G_{\mathbb{Q}_pd}}$, we see that there is an element \mathcal{M} of k_E -BrMod^{*r*}_{dd} with

$$\mathrm{T}^{*,r}_{\mathrm{st}}(\mathcal{M}) \cong \bar{s}|_{G_{\mathbb{Q}_{n^d}}},$$

such that \mathcal{M} has a $(k \otimes_{\mathbb{F}_p} k_E)[u]/u^{ep}$ -basis v_1, \ldots, v_m with $\hat{g}(v_i) = (1 \otimes \overline{\chi}_i(g))v_i$ for all $g \in \text{Gal}(K/K_0)$. Since $\overline{s}|_{G_{\mathbb{Q}_{p^d}}}$ contains a subrepresentation isomorphic to χ , we see from Lemma 2.1.1 that there is a rank-1 subobject \mathcal{N} of \mathcal{M} for which $T_{\text{st}}^{*,r}(\mathcal{N}) \cong \chi$.

Since $\mathcal{N}/u\mathcal{N}$ embeds into $\mathcal{M}/u\mathcal{M}$ (as \mathcal{N} is a free $k[u]/u^{ep}$ -submodule of the free $k[u]/u^{ep}$ -module \mathcal{M}), we see from Lemma 2.1.3 and our assumption on the characters $\overline{\chi}_i$ that we may write \mathcal{N} in the form

- $\mathcal{N} = ((k \otimes_{\mathbb{F}_p} k_E)[u]/u^{(p^d-1)p}) \cdot w,$
- $e_i \mathcal{N}_r = u^{r_i} e_i \mathcal{N},$
- $\varphi_r\left(\sum_{i=0}^{d-1} u^{r_i} e_i w\right) = \lambda w$ for some $\lambda \in (k \otimes_{\mathbb{F}_p} k_E)^{\times}$,
- $\hat{g}(w) = \left(\sum_{i=0}^{d-1} (\omega_d(g)^{k_i} \otimes 1) e_i\right) w$ for all $g \in \text{Gal}(K/K_0)$, and

•
$$N(w) = 0.$$

Here the integers $0 \le r_i \le (p^d - 1)r$ and k_i satisfy $k_i \equiv p(k_{i-1} + r_{i-1}) \pmod{p^d - 1}$ for all *i*, and each k_i is equal to some $(1 + p + \dots + p^{d-1})a_j$. (The conditions on the r_i come from Lemma 2.1.3, and the fact that each k_i is equal to some $(1 + p + \dots + p^{d-1})a_j$ comes from the fact that \mathcal{N} is a submodule of \mathcal{M} which has a basis v_1, \dots, v_n such that $\hat{g}(v_i) = (1 \otimes \overline{\chi}_i(g))v_i$ and the assumption that each $\overline{\chi}_i$ is equal to some ω^{a_j} .) Writing $k_i = (1 + p + \dots + p^{d-1})x_i, 0 \le x_i , we see as$ $in Remark 2.1.4 that we can write <math>r_i = (1 + p + \dots + p^{d-1})(x_{i+1} - x_i) + (p^d - 1)y_i$, with $0 \le y_i \le r$. By Lemma 2.1.3 and Remark 2.1.4, we have

$$\chi|_{I_{\mathbb{Q}_p}} = \sigma_0 \circ \omega_d^{x_0 + y_0 + p^{d-1}(x_1 + y_1) + \dots + p(x_{d-1} + y_{d-1})}$$

Since $r \le p-2$, we have $0 \le y_i \le p-2$, and we conclude (after allowing for "carrying") that each exponent of $\overline{\rho}$ is of the form $a_i + k$ with $0 \le k \le r+1$.

Suppose that $\bar{\rho}$ is *r*-regular. It must then be the case that the x_i above are all distinct. Applying this analysis to each irreducible subrepresentation of $\bar{\rho}$, we conclude that the a_i are all distinct. Since we have $\det(\bar{\rho}')|_{I_{\mathbb{Q}_p}} = \chi^{1+p+\dots+p^{d-1}}|_{I_{\mathbb{Q}_p}} = \omega^{(x_0+y_0)+\dots+(x_{d-1}+y_{d-1})}$, we conclude that $\det \bar{\rho}|_{I_{\mathbb{Q}_p}} = \omega^{a_1+\dots+a_n+y}$ for some

 $0 \le y \le nr \le n(n-1)/2$. The assumption on det $\overline{\rho}$ then implies that $y \equiv n(n-1)/2 \pmod{p-1}$. If in fact r < (n-1)/2, then we have $0 \le y < n(n-1)/2$, which contradicts the assumption that p > n(n-1)/2 + 1.

It remains to treat the case that r = (n-1)/2, where we may assume (by the additional hypothesis that we have assumed in this case) that the representation $\overline{\rho}'$ above has dimension d > 1. By the above analysis we must have y = n(n-1)/2, so that each $y_i = r$. Since we have $r_i \le (p^d - 1)r$, we must have $x_{i+1} - x_i \le 0$ for each *i*, so that in fact $x_0 = x_1 = \cdots = x_{d-1}$, a contradiction (as we already showed that the x_i are distinct).

3. The cohomology of Shimura varieties

3.1. *The semistable reduction of certain* U(n-1, 1)*-Shimura varieties.* Fix $n \ge 2$, and fix an odd prime p.³ We now recall the definitions of the U(n-1, 1)-Shimura varieties with which we will work, and some associated integral models. For simplicity we work over \mathbb{Q} (or rather an imaginary quadratic extension of \mathbb{Q}) rather than over a general totally real field.

For the most part we will follow Section 3 of [Haines and Rapoport 2012] (which uses a similar approach to [Harris and Taylor 2002]), with the occasional reference to [Harris and Taylor 2001]. Fix an imaginary quadratic field F in which the prime p splits, say $(p) = p\bar{p}$ for some choice of p, let $x \mapsto \bar{x}$ be the nontrivial automorphism of F, and regard F as a subfield of \mathbb{C} via a fixed embedding $F \hookrightarrow \mathbb{C}$.

Let *D* be a division algebra over *F* of dimension n^2 , and let * be an involution of *D* of the second kind (that is, $*|_F$ is nontrivial). Assume that *D* splits at \mathfrak{p} (and hence at $\overline{\mathfrak{p}}$), and fix isomorphisms $D_{\mathfrak{p}} \cong M_n(\mathbb{Q}_p)$ and $D_{\overline{\mathfrak{p}}} \cong M_n(\mathbb{Q}_p)$ with the property that, under the induced isomorphism

$$D \otimes \mathbb{Q}_p \cong M_n(\mathbb{Q}_p) \times M_n(\mathbb{Q}_p)^{\mathrm{op}},$$

the involution * corresponds to $(X, Y) \mapsto (Y^t, X^t)$.

Let $G_{/\mathbb{Q}}$ be the algebraic group whose *R*-points are

$$G(R) = \{ x \in (D \otimes_{\mathbb{Q}} R)^{\times} \mid x \cdot x^* \in R^{\times} \}$$

for any \mathbb{Q} -algebra R. Thus our fixed isomorphism $D_{\mathfrak{p}} \cong M_n(\mathbb{Q}_p)$ induces an isomorphism $G \times_{\mathbb{Q}} \mathbb{Q}_p \cong \operatorname{GL}_n \times \mathbb{G}_m$.

Now let $h_0 : \mathbb{C} \to D_{\mathbb{R}}$ be an \mathbb{R} -algebra homomorphism with the properties that $h_0(z)^* = h_0(\overline{z})$ and the involution $x \mapsto h_0(i)^{-1}x^*h_0(i)$ is positive (that is, $\operatorname{tr}_{B/\mathbb{Q}}(xh_0(i)^{-1}x^*h_0(i)) > 0$ for all nonzero x). Let $B = D^{\operatorname{op}}$ and let V = D, which we consider as a free left *B*-module of rank 1 by multiplication on the right.

³The reason for assuming that the prime p is odd is that below we will want to apply the discussion and results of Section 2, in which this assumption was made.

Then $\operatorname{End}_{B}(V) = D$, and one can find an element $\xi \in D^{\times}$ with the properties that $\xi^{*} = -\xi$ and such that the involution ι of *B* defined by $x^{\iota} = \xi x^{*} \xi^{-1}$ is positive (see Section I.7 of [Harris and Taylor 2001] or Section 5.2 of [Haines 2005] for the existence of such a ξ).

We have an alternating pairing $\psi(\cdot, \cdot) : D \times D \to \mathbb{Q}$ defined by $\psi(x, y) = \operatorname{tr}_{D/\mathbb{Q}}(x\xi y^*)$, and one sees easily that $\psi(bx, y) = \psi(x, b^t y)$ and that $\psi(\cdot, h_0(i) \cdot)$ is either positive- or negative-definite. After possibly replacing ξ by $-\xi$, we can and do assume that it is positive-definite.

It is easy to see that one has

$$G(\mathbb{R}) \cong \mathrm{GU}(r,s)$$

for some r, s with r + s = n. We impose the additional assumption that in fact $\{r, s\} = \{n - 1, 1\}$. Note that by Lemma I.7.1 of [Harris and Taylor 2001] one can find division algebras D for which this holds. We say that a compact open subgroup $K \subset G(\mathbb{A}^{\infty})$ (resp. $K^p \subset G(\mathbb{A}^{p,\infty})$) is *sufficiently small* if for some prime q (resp. some prime $q \neq p$) the projection of K (resp. K^p) to $G(\mathbb{Q}_q)$ contains no element of finite order other than 1. If K is sufficiently small, we will consider the Shimura variety $\mathrm{Sh}(G, h_0|_{\mathbb{C}^{\times}}^{-1}, K)$. It has a canonical model over F, which we denote by X(K) (note that if n > 2 the reflex field is F, while if n = 2 the reflex field is \mathbb{Q} , and we let X(K) denote the base change of the canonical model from \mathbb{Q} to F).

We say that a compact open subgroup K of $G(\mathbb{A})$ is of *level dividing* N, for some integer $N \ge 1$, if for all primes $l \nmid N$ we can write $K = K_l K^l$, where K_l is a hyperspecial maximal compact subgroup of $G(\mathbb{Q}_l)$ and K^l is a compact open subgroup of $G(\mathbb{A}^{\infty,l})$. (Note then that in fact $K = K_N \times \prod_{l \nmid N} K_l$, for some compact open subgroup K_N of $\prod_{l \mid N} G(\mathbb{Q}_l)$.) If K is of level dividing N, then we similarly refer to X(K) as a U(n-1, 1)-Shimura variety of level dividing N.

We will now define integral models of these Shimura varieties for two specific kinds of level structure. We begin by introducing notation related to the level structures in question.

We write I_0 for the Iwahori subgroup of $\operatorname{GL}_n(\mathbb{Q}_p) \times \mathbb{Q}_p^{\times}$; namely, I_0 is the subgroup of $\operatorname{GL}_n(\mathbb{Z}_p) \times \mathbb{Z}_p^{\times}$ consisting of elements whose first factor lies in the usual Iwahori subgroup of matrices which are upper-triangular mod p. We write I_1 to denote the pro-p-Iwahori subgroup of $\operatorname{GL}_n(\mathbb{Q}_p) \times \mathbb{Q}_p^{\times}$; namely, I_1 is the (unique) pro-p Sylow subgroup of I_0 , and consists of those elements of $\operatorname{GL}_n(\mathbb{Z}_p) \times \mathbb{Z}_p^{\times}$ whose first factor is upper-triangular unipotent mod p, and whose second factor is congruent to 1 mod p. We write I_1^* to denote the subgroup of I_0 consisting of those matrices in $\operatorname{GL}_n(\mathbb{Z}_p) \times \mathbb{Z}_p^{\times}$ whose first factor is upper-triangular unipotent mod p.

There is a natural isomorphism $\mathbb{Z}_p^{\times} = \mathbb{F}_p^{\times} \times (1 + p\mathbb{Z}_p)$, and this induces a natural isomorphism $I_1^* = \mathbb{F}_p^{\times} \times I_1$. If we let *T* denote the diagonal torus in GL_n , then there is also a natural isomorphism $I_0 = T(\mathbb{F}_p) \times I_1^*$.

We will define integral models for $X(I_0K^p)$ and $X(I_1^*K^p)$ over the local rings $\mathbb{O}_{F,(\mathfrak{p})}$ and $\mathbb{O}_{F(\xi_{p-1}),(v)}$ respectively, where ξ_{p-1} denotes a primitive (p-1)-st root of unity, and v is some fixed place in $F(\xi_{p-1})$ above \mathfrak{p} ; here K^p is a sufficiently small compact open subgroup of $G(\mathbb{A}^{\infty,p})$, and we consider I_0 and I_1^* as subgroups of $G(\mathbb{Q}_p) \cong \operatorname{GL}_n(\mathbb{Q}_p) \times \mathbb{Q}_p^{\times}$. We will typically not include K^p in the notation, and we will write $\mathscr{X}_0(p), \mathscr{X}_1(p)$ for our integral models of $X_0(p) := X(I_0K^p)$ and $X_1(p) := X(I_1^*K^p)$ respectively.

Remark 3.1.1. Note that in [Haines and Rapoport 2012], [Harris and Taylor 2002], and [Harris and Taylor 2001], the authors work over \mathbb{Z}_p , but we follow [Kottwitz 1992] in working over $\mathbb{O}_{F,(p)}$, so as to satisfy the hypothesis required to be in the global case of Section 1. The appearance of ζ_{p-1} in the ring of definition of $\mathscr{X}_1(p)$ is a consequence of our use of Oort–Tate theory in the definition of the integral model in this case.

In order to define these integral models, we first recall a certain category of abelian schemes (up to isogeny) with polarisations and endomorphisms. If \mathcal{F} is a set-valued functor on the category of connected, locally noetherian $\mathbb{O}_{F,(p)}$ -schemes, we will also consider it to be a functor on the category of all locally noetherian $\mathbb{O}_{F,(p)}$ -schemes by setting

$$\mathscr{F}(\coprod S_i) := \prod \mathscr{F}(S_i).$$

Let \mathbb{O}_B be the unique maximal $\mathbb{Z}_{(p)}$ -order in B which under our fixed identification $B \otimes_{\mathbb{Q}} \mathbb{Q}_p = M_n(\mathbb{Q}_p) \times M_n(\mathbb{Q}_p)^{\text{op}}$ is identified with $M_n(\mathbb{Z}_{(p)} \times M_n(\mathbb{Z}_{(p)})^{\text{op}}$. Let S be a connected, locally noetherian $\mathbb{O}_{F,(p)}$ -scheme, and let AV_S be the category whose objects are pairs (A, i), where A is an abelian scheme over S of dimension n^2 and $i : \mathbb{O}_B \to \text{End}_S(A) \otimes \mathbb{Z}_{(p)}$ is a homomorphism. We define homomorphisms in AV_S by

$$\operatorname{Hom}((A_1, i_1), (A_2, i_2)) = \operatorname{Hom}_{\mathbb{O}_B}((A_1, i_1), (A_2, i_2)) \otimes \mathbb{Z}_{(p)}$$

(that is, the elements of $\operatorname{Hom}_S(A_1, A_2) \otimes \mathbb{Z}_{(p)}$ which commute with the action of \mathbb{O}_B). The dual of an object (A, i) of AV_S is (\hat{A}, \hat{i}) , where \hat{A} is the dual abelian scheme of A and $\hat{i}(b) = (i(b^i))^{\wedge}$. A polarisation of (A, i) is a homomorphism $\lambda : (A, i) \to (\hat{A}, \hat{i})$ in AV_S with the property that, for some $n \ge 1$, $n\lambda$ is induced by an ample line bundle on A. A principal polarisation is a polarisation which is also an isomorphism in AV_S . A Q-class of polarisations is an equivalence class of homomorphisms $(A, i) \to (\hat{A}, \hat{i})$ which contains a polarisation, under the equivalence relation of differing by a Q[×]-scalar.

Fix K^p a sufficiently small open compact subgroup of $G(\mathbb{A}^{\infty,p})$. Let \mathcal{A}_0 be the set-valued functor on the category of locally noetherian schemes over $\mathbb{O}_{F,(\mathfrak{p})}$

which sends a connected, locally noetherian scheme S over $\mathbb{O}_{F,(\mathfrak{p})}$ to the set of isomorphism classes of the following data:

• A commutative diagram of morphisms in the category AV_S of the form

where each α_i is an isogeny of degree p^{2n} and their composite is just multiplication by p. In addition, λ_0 is a \mathbb{Q} -class of polarisations containing a principal polarisation. Furthermore, we require that each A_i satisfies a compatibility between the two actions of \mathbb{O}_F on the Lie algebra of A_i (one action coming from the structure morphism $\mathbb{O}_{F,(\mathfrak{p})} \to \mathbb{O}_S$, and the other from the \mathbb{O}_B -action; see [Harris and Taylor 2001, §III.4] for a discussion of this condition).

• A geometric point s of S, and a $\pi_1(S, s)$ -invariant K^p -orbit of isomorphisms

$$\overline{\eta}: V \otimes_{\mathbb{Q}} \mathbb{A}^{p,\infty} \xrightarrow{\sim} H_1((A_0)_s, \mathbb{A}^{p,\infty})$$

which are \mathbb{O}_B -linear and up to a constant in $(\mathbb{A}^{p,\infty})^{\times}$ take the ψ -pairing on the left side to the λ_0 -Weil pairing on the right side. (This data is canonically independent of the choice of *s*; see the discussion on pp. 390–391 of [Kottwitz 1992].)

An isomorphism of this data is one induced by isomorphisms in AV_S which preserve the λ_i up to an overall \mathbb{Q}^{\times} -scalar.

The functor \mathcal{A}_0 is represented by a projective scheme $\mathscr{X}_0(p)$ over $\mathbb{O}_{F,(p)}$, which is an integral model for $X(I_0K^p)$. (See the proof of Lemma 3.2 of [Taylor and Yoshida 2007], which shows that $\mathscr{X}_0(p)$ is projective over the usual integral model at hyperspecial level. At hyperspecial level, quasiprojectivity is proved on p. 391 of [Kottwitz 1992], and projectivity can be checked via the valuative criterion for properness as on p. 392 of the same work. More properly, $\mathscr{X}_0(p)$ is an integral model for a disjoint union of a number of copies of $X(I_0K^p)$, due to the possible failure of the Hasse principle; see for example Section 7 and the discussion on p. 400 of [Kottwitz 1992]. Since the cohomology of a disjoint union of spaces is the direct sum of the cohomologies of the individual spaces, this does not affect our arguments, and we will not dwell on this point in the following.) The proof of Proposition 3.4(3) of [Taylor and Yoshida 2007] (which goes over unchanged in our setting) shows that the special fibre of $\mathscr{X}_0(p)$ is a strict normal crossings divisor.

Our next goal is to describe an integral model $\mathscr{X}_1(p)$, over $\mathbb{O}_{F(\zeta_{p-1}),(v)}$, for $X(I_1^*K^p)$ (or rather, as in the previous paragraph, an integral model of a disjoint union of a number of copies of $X(I_1^*K^p)$). Recalling that *T* denotes the diagonal

torus in GL_n , we let $\iota_i : \mathbb{G}_m \to T$ denote the embedding of tori identifying \mathbb{G}_m with the subgroup of *T* consisting of elements which are 1 away from the *i*-th diagonal entry. We use the same notation ι_i to denote the map $\mathbb{F}_p^{\times} \to T(\mathbb{F}_p)$ induced by the map of tori.

The quotient I_0/I_1^* is naturally identified with $T(\mathbb{F}_p)$, and so $T(\mathbb{F}_p)$ acts on $X_1(p)$, with quotient isomorphic to $X_0(p)$.

Given an *S*-valued point of \mathcal{A}_0 , let $A_i(p^{\infty})$ be the *p*-divisible group associated to A_i for i = 0, ..., n-1. Each $A_i(p^{\infty})$ has an action of $\mathbb{O}_B = M_n(\mathbb{Z}_{(p)}) \times M_n(\mathbb{Z}_p)^{\text{op}}$. Let $X_i = e_{11}A_i(p^{\infty})$, where e_{11} is the usual idempotent in $M_n(\mathbb{Z}_{(p)})$ (and is zero on the second factor). Then each X_i is a *p*-divisible group of height *n* and dimension 1, and we obtain a chain of isogenies of degree *p*

$$\mathscr{C}: X_0 \xrightarrow{\alpha_0} X_1 \xrightarrow{\alpha_1} \cdots \xrightarrow{\alpha_{n-2}} X_{n-1} \xrightarrow{\alpha_{n-1}} X_n := X_0,$$

whose composite is equal to multiplication by *p*.

We let OT denote the Artin stack over $\mathbb{O}_{F(\zeta_{p-1}),(v)}$ given by

$$OT := \left[\operatorname{Spec} \mathbb{O}_{F(\zeta_{p-1}), (v)}[X, Y] / (XY - w_p) / \mathbb{G}_m \right],$$

where \mathbb{G}_m acts via $\lambda \cdot (X, Y) = (\lambda^{p-1}X, \lambda^{1-p}Y)$ and w_p is some explicit element of $\mathbb{O}_{F(\xi_{p-1}),(v)}$ of valuation 1. Oort–Tate theory shows that OT classifies finite flat group schemes of order p over $\mathbb{O}_{F(\xi_{p-1}),(v)}$ -schemes. The universal group scheme over OT is the stack

$$\mathcal{G} := \left[\operatorname{Spec} \mathbb{O}_{F(\zeta_{p-1}), (v)}[X, Y, Z] / (XY - w_p, Z^p - XZ) / \mathbb{G}_m \right],$$

where \mathbb{G}_m acts on X and Y as above, and on Z via $\lambda \cdot Z = \lambda Z$. The morphism $\mathcal{G} \to OT$ is the evident one, the zero section of \mathcal{G} is cut out by the equation Z = 0, and we let \mathcal{G}^{\times} denote the closed subscheme of \mathcal{G} cut out by the equation $Z^{p-1} - X = 0$; this is the so-called *scheme of generators* of \mathcal{G} . (See Theorem 6.5.1 of [Genestier and Tilouine 2005] for these facts, which are a restatement of Theorem 2 of [Tate and Oort 1970] in the language of stacks.)

We define $\mathscr{X}_1(p)$ via the Cartesian diagram

$$\begin{aligned}
\mathscr{X}_{1}(p) & \longrightarrow \mathscr{G}^{\times} \times_{\mathbb{O}_{F(\zeta_{p-1}),(v)}} \cdots \times_{\mathbb{O}_{F(\zeta_{p-1}),(v)}} \mathscr{G}^{\times} \\
\downarrow & \downarrow \\
\mathscr{X}_{0}(p)_{\mathbb{O}_{F(\zeta_{p-1}),(v)}} & \longrightarrow \operatorname{OT} \times_{\mathbb{O}_{F(\zeta_{p-1}),(v)}} \cdots \times_{\mathbb{O}_{F(\zeta_{p-1}),(v)}} \operatorname{OT}
\end{aligned}$$
(3.1.2)

where the bottom horizontal arrow is given by

$$\mathscr{C} \mapsto (\ker(\alpha_0), \ldots, \ker(\alpha_{n-1})).$$

Note that the right-hand vertical arrow is finite and relatively representable by construction, and so the left-hand vertical arrow is a finite morphism of schemes. The action of $T(\mathbb{F}_p)$ on $X_1(p)$ extends to an action on $\mathscr{X}_1(p)$, namely, the action pulled back from the action of $T(\mathbb{F}_p)$ on $\mathscr{G}^{\times} \times_{\mathbb{O}_{F(\zeta_{p-1}),(v)}} \cdots \times_{\mathbb{O}_{F(\zeta_{p-1}),(v)}} \mathscr{G}^{\times}$ over $OT \times_{\mathbb{O}_{F(\zeta_{p-1}),(v)}} \cdots \times_{\mathbb{O}_{F(\zeta_{p-1}),(v)}} OT$.

Let $\pi := (-p)^{1/(p-1)}$, and let w be the unique finite place of $L := F(\zeta_{p-1}, \pi)$ lying over our fixed place v of $F(\zeta_{p-1})$. We let \mathbb{O} denote the localisation of \mathbb{O}_L at w, and we let $\mathscr{X}_1(p)_{\mathbb{O}}$ denote the normalisation of the base change $\mathscr{X}_1(p)_{/\mathbb{O}}$ of $\mathscr{X}_1(p)$ over \mathbb{O} . We write $I := \operatorname{Gal}(L/F(\zeta_{p-1}))$; this is also the inertia group at w in $\operatorname{Gal}(L/F)$. The group I acts naturally on $\mathscr{X}_1(p)_{\mathbb{O}}$. The mod p cyclotomic character ω induces an isomorphism (which we continue to denote by ω)

$$\omega: I \xrightarrow{\sim} \mathbb{F}_p^{\times}.$$

For each i = 0, ..., n-1, we let $\alpha_i : I \to T$ denote the composite $\iota_i \circ \omega^{-1}$.

Lemma 3.1.3. The scheme $\mathscr{X}_1(p)_{\mathbb{O}}$ is a semistable projective model for $X_1(p)_{/L}$ over \mathbb{O} , the natural morphism

$$\mathscr{X}_1(p)_{\mathbb{C}} \to \mathscr{X}_0(p) \tag{3.1.4}$$

is tamely ramified, and the action of $I \times T$ on $X_1(p)_{/L}$ extends to an action on $\mathscr{X}_1(p)_{\mathbb{O}}$. Furthermore, on each irreducible component of its special fibre, the inertia group I acts through the composite of the T-action with one of the characters α_i .

Proof. We will apply a form of Deligne's homogeneity principle, as described in the proof of [Taylor and Yoshida 2007, Proposition 3.4], to the morphism (3.1.4). The scheme here denoted $\mathscr{X}_0(p)$ is there denoted X_U (and the integral model there is considered over \mathbb{Z}_p rather than $\mathbb{O}_{F,(\mathfrak{p})}$, but this is immaterial for our present purposes), while the scheme there denoted X_{U_0} is an integral model of the Shimura variety (in the notation of the present paper) $X(K_p K^p)$, where $K_p = \operatorname{GL}_n(\mathbb{Z}_p) \times \mathbb{Z}_p^{\times}$. We let $\mathscr{X}_0(p)_s^{(h)}$ (for $0 \le h \le n-1$) denote the locally closed subset of the special fibre $\mathscr{X}_0(p)_s$, obtained by pulling back the locally closed subset $\overline{X}_{U_0}^{(h)}$ defined in Section 3 of [Taylor and Yoshida 2007] under the natural projection $\mathscr{X}_0(p)_s = \overline{X}_U \to \overline{X}_{U_0}$.

We will show that the morphism (3.1.4) is tamely ramified in the formal neighbourhood of any closed geometric point of the special fibre $\mathscr{X}_0(p)_s$, and hence (by Lemma 1.5.3) that it is tamely ramified. We first note that the morphism $\mathscr{X}_0(p)_{/\mathbb{O}_{F(\zeta_{p-1}),(v)}} \to \mathscr{X}_0(p)$ induces an isomorphism on special fibres, and so we are free to replace $\mathscr{X}_0(p)$ by $\mathscr{X}_0(p)_{/\mathbb{O}_{F(\zeta_{p-1}),(v)}}$ in our considerations. We next note that the completion of the bottom arrow of (3.1.2) at a closed geometric point \bar{x}_0 of $\mathscr{X}_0(p)_s$ depends up to isomorphism only on the value of *h* for which $\bar{x}_0 \in \mathscr{X}_0(p)_s^{(h)}$ (since the *p*-divisible group attached to the point \bar{x}_0 depends only on

the value of h), and hence that the restriction of (3.1.4) to a formal neighbourhood of \bar{x}_0 depends only on the value of h. Then, by [Taylor and Yoshida 2007, Lemma 3.1], the closure of $\mathscr{X}_0(p)_s^{(h)}$ contains $\mathscr{X}_0(p)_s^{(0)}$. Since being tamely ramified is an open condition, we conclude from these two conditions that, in order to prove the lemma, it suffices to show that the restriction of (3.1.4) to a formal neighbourhood of \bar{x}_0 is tamely ramified at closed geometric points \bar{x}_0 of $\mathscr{X}_0(p)_s^{(0)}$. (As already indicated, this argument is a variation on Deligne's homogeneity principle.)

Thus, consider a closed geometric supersingular point \overline{x}_0 of $\mathscr{X}_0(p)_s^{(0)}$, so that \overline{x}_0 admits a formal neighbourhood of the form

Spec
$$W(\mathbb{F}_p)[[T_1,\ldots,T_n]]/(T_1\cdots T_n-w_p).$$

The proof of [Taylor and Yoshida 2007, Proposition 3.4] shows that the T_i may be taken to be the matrix of α_{i-1} on tangent spaces, so that the map $\mathscr{X}_0(p) \rightarrow$ OT $\times_{\mathbb{O}_{F(\zeta_{p-1}),(v)}} \cdots \times_{\mathbb{O}_{F(\zeta_{p-1}),(v)}}$ OT may be defined in the formal neighbourhood of \overline{x}_0 by the map

$$(T_1,\ldots,T_n)\mapsto ((T_1,U_1),\ldots,(T_n,U_n)),$$

where $U_i = T_1 \cdots \hat{T}_i \cdots T_n$ (and, as is usual in these situations, a hat on a variable denotes that that variable is omitted in the expression). Thus a formal neighbourhood of a closed geometric point lying over \bar{x}_0 in $(\mathscr{X}_1(p)_{/\mathbb{C}})_s$ is isomorphic to

Spec
$$W(\overline{\mathbb{F}}_p)[\pi][[V_1,\ldots,V_n]]/((V_1\cdots V_n)^{p-1}-w_p).$$

If we write $u := V_1 \cdots V_n / \pi$, then we see that $u^{p-1} = -w_p / p$, and hence that u lies in the normalisation of this formal neighbourhood. Furthermore, on each component of this normalisation, u is equal to one of the (p-1)-st roots of $-w_p / p$ lying in $W(\overline{\mathbb{F}}_p)$. Thus the normalisation of this formal neighbourhood is a union of components, each isomorphic to

Spec
$$\mathbb{O}[[V_1,\ldots,V_n]]/(V_1\cdots V_n-u\pi),$$

with the morphism (3.1.4) being given by $T_i = V_i^{p-1}$. Thus this morphism is indeed tamely ramified in the formal neighbourhood of \bar{x}_0 .

It is clear that the $I \times T$ -action on $X_1(p)_L$ extends to an action on $\mathscr{X}_1(p)_{/\mathbb{O}}$, and hence to its normalisation $\mathscr{X}_1(p)_{\mathbb{O}}$. As for the final statement, note that I acts on π via ω , and hence on u via ω^{-1} , while I fixes each V_i . Also T acts on V_i through multiplication by the *i*-th diagonal entry, and so acts on u via multiplication by the determinant. Combining these facts, we see that I acts on the components of the special fibre on which $V_i = 0$ via $\iota_i \circ \omega^{-1}$. **Remark 3.1.5.** This lemma (and the fact that $\mathscr{X}_0(p)$ is strictly semistable) shows that the map $\mathscr{X}_1(p)_0 \to \mathscr{X}_0(p)$ provides a tame strictly semistable context, in the sense of Section 1.5. In particular, we can apply Theorem 1.5.15 in this setting (of course taking the group *G* to be the abelian group *T*), and we see that each character ψ_i as in the statement of that theorem is equal to one of the characters α_i .

3.2. *Canonical local systems.* If *K* is a sufficiently small compact open subgroup of $G(\mathbb{A}^{\infty})$, and *V* is a continuous representation of *K* on a finite-dimensional $\overline{\mathbb{F}}_p$ -vector space (this vector space being equipped with its discrete topology), then we may associate to *V* an étale local system \mathcal{F}_V of $\overline{\mathbb{F}}_p$ -vector spaces on X(K) as follows: Choose an open normal subgroup $K' \subset K$ lying in the kernel of *V*, and regard *V* as a representation of the quotient K/K'. Since X(K') is naturally an étale K/K'-torsor over X(K), we may form the étale local system of $\overline{\mathbb{F}}_p$ -vector spaces over X(K) associated to the K/K'-representation *V*. This local system is independent of the choice of K', up to canonical isomorphism, and we define it to be \mathcal{F}_V .

Definition 3.2.1. We refer to the étale local systems \mathcal{F}_V that arise by the preceding construction as the *canonical local systems* on X(K). If we may choose K' in the kernel of V to be of level dividing N (so that in particular X(K) is of level dividing N), then we say that \mathcal{F}_V can be *trivialised at level* N.

3.3. The Eichler–Shimura relation. Let X be a U(n-1, 1)-Shimura variety of level dividing N. Let w be a place of F such that $l := w|_{\mathbb{Q}}$ splits in F and does not divide N. There is a natural action via correspondences on X of Hecke operators $T_w^{(i)}$, $0 \le i \le n$, where $T_w^{(i)}$ is the double coset operator corresponding to

$$\begin{pmatrix} l1_i & 0\\ 0 & 1_{n-i} \end{pmatrix} \times 1 \in \operatorname{GL}_n(\mathbb{Q}_l) \times \mathbb{Z}_l^{\times},$$

where we use the assumption that $l \nmid N$ and identify a hyperspecial maximal compact subgroup of $G(\mathbb{Q}_l)$ with $\operatorname{GL}_n(\mathbb{Z}_l) \times \mathbb{Z}_l^{\times}$ via an isomorphism $D_w \cong M_n(\mathbb{Q}_l)$. These correspondences then act on the cohomology $H_{\acute{e}t}^j(X_{\overline{\mathbb{Q}}}, \overline{\mathbb{F}}_p)$.

More generally if \mathcal{F}_V is a canonical local system on X that can be trivialised at level N, then we obtain an action of the double coset operators $T_w^{(i)}$ on \mathcal{F}_V , and hence on the cohomology $H^j_{\acute{e}t}(X_{\overline{\mathbb{O}}}, \mathcal{F}_V)$.

The following theorem regarding this action is then an immediate consequence of the main result of [Wedhorn 2000] (which proves the Eichler–Shimura relation for PEL Shimura varieties at places of good reduction at which the group is split).

Theorem 3.3.1. Let X be a U(n-1, 1)-Shimura variety of level dividing N and \mathcal{F}_V a canonical local system on X. Let w be a place of F such that $w|_{\mathbb{Q}}$ splits in F and does not divide Np. Then $\sum_{i=0}^{n} (-1)^i (\operatorname{Norm} w)^{i(i-1)/2} T_w^{(i)} \operatorname{Frob}_w^{n-i}$ acts as 0 on each $H_{\text{ét}}^j(X_{\overline{\mathbb{Q}}}, \mathcal{F}_V)$.
3.4. Vanishing and torsion-freeness of cohomology for certain U(n - 1, 1)-Shimura varieties. Let X denote a U(n - 1, 1)-Shimura variety as above, and let \mathcal{F}_V denote a canonical local system on X. Choose N so that X has level dividing N, so that \mathcal{F}_V can be trivialised at level N, and so that p divides N. Assume that the projection of the corresponding level K to $G(\mathbb{A}^{p,\infty})$ is sufficiently small.

Let $\mathbb{T} = \overline{\mathbb{Z}}_p[T_w^{(i)}]$ be the polynomial ring in the variables $T_w^{(i)}$, $1 \le i \le n$, where w runs over the places of F such that $w|_{\mathbb{Q}}$ splits in F and does not divide N. Let \mathfrak{m} be a maximal ideal in \mathbb{T} with residue field $\overline{\mathbb{F}}_p$, and suppose that there exists a continuous irreducible representation $\rho_{\mathfrak{m}} : G_F \to \operatorname{GL}_n(\overline{\mathbb{F}}_p)$ which is unramified at all finite places not dividing N, and which satisfies $\operatorname{char}(\rho_{\mathfrak{m}}(\operatorname{Frob}_w)) \equiv \sum_{i=0}^{n} (-1)^i (\operatorname{Norm} w)^{i(i-1)/2} T_w^{(i)} X^{n-i} \mod \mathfrak{m}$ for all $w \nmid N$ such that $w|_{\mathbb{Q}}$ splits in F. Continue to fix a choice of a place \mathfrak{p} of F dividing p, and write $G_{\mathbb{Q}_p}$ for G_{F_p} from now on. Recall that the choice of \mathfrak{p} also gives us an isomorphism $G(\mathbb{Q}_p) \cong \operatorname{GL}_n(\mathbb{Q}_p) \times \mathbb{Q}_p^{\times}$ as in Section 3.1.

We consider the following further hypothesis on $\rho_{\mathfrak{m}}$ (this is Hypothesis 4.1.1):

Hypothesis 3.4.1. If $\theta : G_F \to \operatorname{GL}_m(\overline{\mathbb{F}}_p)$ is any continuous, irreducible representation with the property that the characteristic polynomial of $\rho_{\mathfrak{m}}(g)$ annihilates $\theta(g)$ for every $g \in G_F$, then θ is equivalent to $\rho_{\mathfrak{m}}$.

We will now prove our first main result, a vanishing theorem for the cohomology of X with \mathcal{F}_V -coefficients:

Theorem 3.4.2. Suppose that $\rho_{\mathfrak{m}}$ satisfies Hypothesis 3.4.1, that $\rho_{\mathfrak{m}}|_{G_{\mathbb{Q}_p}}$ is rregular for some $r \leq (n-1)/2$, that p > n(n-1)/2+1 and, if r = (n-1)/2, suppose in addition that $\rho_{\mathfrak{m}}|_{G_{\mathbb{Q}_p}}$ contains an irreducible subquotient of dimension greater than 1. Then the localisations $H^i_{\text{ét}}(X_{\overline{\mathbb{Q}}}, \mathcal{F}_V)_{\mathfrak{m}}$ vanish for $i \leq r$ and $i \geq 2(n-1)-r$.

Remark 3.4.3. In Section 4.1 we will show that Hypothesis 3.4.1 is satisfied if either $\rho_{\mathfrak{m}}$ is induced from a character of G_K for some degree-*n* cyclic Galois extension K/\mathbb{Q} , or if $p \ge n$ and $\operatorname{SL}_n(k) \subseteq \rho_{\mathfrak{m}}(G_F) \subseteq \overline{\mathbb{F}}_p^{\times} \operatorname{GL}_n(k)$ for some subfield $k \subset \overline{\mathbb{F}}_p$.

Remark 3.4.4. While we work here with étale local systems and étale cohomology, by virtue of Artin's comparison theorem [SGA 4₃ 1973, Exposé XI, Théorème 4.3] our vanishing results are equivalent to vanishing results for the cohomology of the complex U(n - 1, 1)-Shimura varieties with coefficients in the corresponding canonical local systems for the complex topology.

Proof of Theorem 3.4.2. First, note that it suffices to prove vanishing in degree $i \le r$ for all V, as vanishing in degree $i \ge 2(n-1) - r$ then follows by Poincaré duality. (Note that the dual of the canonical local system \mathcal{F}_V attached to a representation V is the canonical local system attached to the contragredient representation V^{\vee} .)

We prove the theorem for $i \leq r$ by induction on i, the case when i < 0 being trivial. We begin by reducing to the case when \mathcal{F}_V is trivial. To this end, write X = X(K), let K' be an open normal subgroup of K of level dividing N such that V is representation of K/K', and choose an embedding of K/K'-representations $V \hookrightarrow \overline{\mathbb{F}}_p[K/K']^n$ for some n; denote the cokernel by W. Let π denote the projection $X(K') \to X(K)$. Passing to the associated canonical local systems, we obtain a short exact sequence $0 \to \mathcal{F}_V \to \pi_* \overline{\mathbb{F}}_p^n \to \mathcal{F}_W \to 0$, which gives rise to an exact sequence of cohomology

$$H^{i-1}_{\mathrm{\acute{e}t}}(X_{\overline{\mathbb{Q}}}, \mathcal{F}_W) \longrightarrow H^{i}_{\mathrm{\acute{e}t}}(X_{\overline{\mathbb{Q}}}, \mathcal{F}_V) \longrightarrow H^{i}_{\mathrm{\acute{e}t}}(X_{\overline{\mathbb{Q}}}, \pi_*\overline{\mathbb{F}}_p^n) = H^{i}_{\mathrm{\acute{e}t}}(X(K')_{\overline{\mathbb{Q}}}, \overline{\mathbb{F}}_p)^n.$$

Localising at m and applying our inductive hypothesis (with \mathcal{F}_W in place of \mathcal{F}_V), we reduce to the case of constant coefficients (with X(K') in place of X). We therefore turn to establishing the claimed vanishing in this case.

Suppose now that K' is any open normal subgroup of K of level dividing N. Combining the Hochschild–Serre spectral sequence

$$E_2^{m,n} = H^m_{\text{\'et}}(K/K', H^n(X(K')_{\overline{\mathbb{Q}}}, \overline{\mathbb{F}}_p)_{\mathfrak{m}}) \Longrightarrow H^{m+n}_{\text{\'et}}(X(K)_{\overline{\mathbb{Q}}}, \overline{\mathbb{F}}_p)_{\mathfrak{m}}$$

with our inductive hypothesis, we find that vanishing of $H^i_{\text{ét}}(X(K'), \overline{\mathbb{F}}_p)_{\mathfrak{m}}$ implies the vanishing of $H^i_{\text{ét}}(X(K)_{\overline{\mathbb{Q}}}, \overline{\mathbb{F}}_p)_{\mathfrak{m}}$. Thus, without loss of generality, we may and do assume that $K = K_p K^p$, where K_p is an open normal subgroup of I_1 and K^p is a sufficiently small compact open subgroup of $G(\mathbb{A}^{\infty,p})$.

We again consider a Hochschild–Serre spectral sequence, this time the one relating the cohomology of X(K) and $X(I_1K^p)$, which takes the form

$$E_2^{m,n} = H^m(I_1/K_p, H^n_{\text{\'et}}(X(K)_{\overline{\mathbb{Q}}}, \overline{\mathbb{F}}_p)_{\mathfrak{m}}) \Longrightarrow H^{m+n}_{\text{\'et}}(X(I_1K^p)_{\overline{\mathbb{Q}}}, \overline{\mathbb{F}}_p)_{\mathfrak{m}}.$$

Once more taking into account our inductive hypothesis, we obtain an isomorphism $H^i_{\text{ét}}(X(I_1K^p)_{\overline{\mathbb{Q}}}, \overline{\mathbb{F}}_p)_{\mathfrak{m}} \xrightarrow{\sim} H^i_{\text{ét}}(X(K)_{\overline{\mathbb{Q}}}, \overline{\mathbb{F}}_p)_{\mathfrak{m}}^{I_1/K_p}$. Since I_1/K_p is a *p*-group, while $H^i_{\text{ét}}(X(K)_{\overline{\mathbb{Q}}}, \overline{\mathbb{F}}_p)_{\mathfrak{m}}$ is a vector space over a field of characteristic *p*, we see that the latter space vanishes if and only if its space of I_1/K_p -invariants does. Thus we are reduced to establishing the theorem in the case when $K = I_1K^p$.

Now recall that $I_1^* = \mathbb{F}_p^{\times} \times I_1$, and that the projection onto the first factor arises from the similitude projection $\operatorname{GU}(n-1,1) \to \mathbb{G}_m$. From this it follows that $X(I_1K^p)$ is isomorphic to the product $X(I_1^*K^p) \times_F$ Spec *A*, where A := $F[x]/\Phi_p(x)$ (where $\Phi_p(x)$ denotes the *p*-th cyclotomic polynomial; the action of $\mathbb{F}_p^{\times} = I_1^*/I_1$ on $X(I_1K^p)$ is induced by the action of \mathbb{F}_p^{\times} on *A* given by $x \mapsto x^a$ for $a \in \mathbb{F}_p^{\times}$). Consequently there is an isomorphism of Galois representations

$$H^{i}(X(I_{1}K^{p})_{\overline{\mathbb{Q}}},\overline{\mathbb{F}}_{p})_{\mathfrak{m}} \xrightarrow{\simeq} \bigoplus_{j=0}^{p-2} H^{i}(X(I_{1}^{*}K^{p})_{\overline{\mathbb{Q}}},\overline{\mathbb{F}}_{p})_{\mathfrak{m}_{j}} \otimes \omega^{j},$$

where \mathfrak{m}_j is the maximal ideal in \mathbb{T} which corresponds to the twisted Galois representation $\rho_{\mathfrak{m}_j} := \rho_{\mathfrak{m}} \otimes \omega^{-j}$. Since each of the Galois representations $\rho_{\mathfrak{m}_j}$ satisfies the hypotheses of the theorem (as these hypotheses are invariant under twisting), we are reduced to proving the theorem in the case of $X(I_1^*K^p)$.

Consider an irreducible $G_F \times I_0/I_1^*$ -subrepresentation of $H_{\text{ét}}^i(X(I_1^*K^p)_{\overline{\mathbb{Q}}}, \overline{\mathbb{F}}_p)_{\mathfrak{m}}$, which we may write in the form $\theta \otimes \beta$, where θ is an absolutely irreducible G_F -representation and β is a character of the abelian group $T(\mathbb{F}_p) = I_0/I_1^*$. Theorem 3.3.1 and Hypothesis 3.4.1 taken together imply that θ is equivalent to $\rho_{\mathfrak{m}}$. We consider β as an *n*-tuple of characters β_1, \ldots, β_n of \mathbb{F}_p^{\times} . Write $\tilde{\beta}_j$ for the Teichmüller lift of β_j and χ_j for the character of $I_{\mathbb{Q}_p}$ given by β_j . By Lemma 3.1.3, Remark 3.1.5 and Theorem 1.5.15, we see that θ can be embedded as the reduction mod p of a potentially semistable representation with Hodge–Tate weights in the range [-i, 0], whose inertial type is a direct sum of characters belonging to the collection $\{\tilde{\beta}_j\}$. (To see this, note that as θ is in the β -part of the cohomology, it necessarily occurs in the reduction of the $\tilde{\beta}$ -part of the $G_F \times T$ -representation provided by Theorem 1.5.15.) We claim that det $\theta|_{I_{\mathbb{Q}_p}} = \det \rho_{\mathfrak{m}}|_{I_{\mathbb{Q}_p}} = \overline{\chi}_1 \cdots \overline{\chi}_n \omega^{-n(n-1)/2}$. Admitting this for the moment, we may apply Theorem 2.2.4 to the representation $\rho_{\mathfrak{m}}|_{G_{\mathbb{Q}_p}}^{\vee}$, is not *r*-regular. Equivalently, we see that $\rho_{\mathfrak{m}}|_{G_{\mathbb{Q}_p}}$ is not *r*-regular. Equivalently, we see that

It remains to establish the equality det $\theta|_{I_{\mathbb{Q}_p}} = \det \rho_{\mathfrak{m}}|_{I_{\mathbb{Q}_p}} = \overline{\chi}_1 \cdots \overline{\chi}_n \omega^{-n(n-1)/2}$. To see this, note that Theorem 3.3.1 implies that, for each place $w \nmid N$ of F such that $w|_{\mathbb{Q}}$ splits in F, we have det $\rho_{\mathfrak{m}}(\operatorname{Frob}_w) = (\operatorname{Norm} w)^{n(n-1)/2} T_w^{(n)}$. Let $\psi_{\mathfrak{m}}$ be the character of $\mathbb{A}_{F}^{\times}/F^{\times}$ corresponding to the character $\omega^{n(n-1)/2}$ det $\rho_{\mathfrak{m}}$ by global class field theory; then we need to prove that $\psi_{\mathfrak{m}}|_{\mathbb{Z}_n^{\times}} = \beta_1 \cdots \beta_n$. The centre of $G(\mathbb{A}_{\mathbb{Q}})$ is \mathbb{A}_{F}^{\times} , so by the definition of the Shimura variety X there is a natural action of $\mathbb{A}_F^{\times}/F^{\times}$ on $H^i_{\text{ét}}(X(I_1^*K^p)_{\overline{\mathbb{Q}}}, \overline{\mathbb{F}}_p)_{\mathfrak{m}}$. By the definition of the Hecke operators, we see that if ϖ_w is a uniformiser at a place $w \nmid N$ of F such that $w|_{\mathbb{Q}}$ splits in F, then $\overline{\omega}_w$ acts as $T_w^{(n)}$. By the Chebotarev density theorem, we deduce that the action of $\mathbb{A}_F^{\times}/F^{\times}$ on the underlying vector space of θ (which by definition is a subspace of $H^i_{\text{ét}}(X(I_1^*K^p)_{\overline{\mathbb{Q}}}, \overline{\mathbb{F}}_p)_{\mathfrak{m}})$ is via the character $\psi_{\mathfrak{m}}$. In order to compute $\psi_{\mathfrak{m}}|_{\mathbb{Z}_{p}^{\times}}$, it is thus sufficient to compute the action of \mathbb{Z}_{p}^{\times} on $H_{\text{ét}}^{i}(X(I_{1}^{*}K^{p})_{\overline{\mathbb{Q}}}, \overline{\mathbb{F}}_{p})_{\mathfrak{m}}$, and in particular sufficient to compute the action of the Iwahori subgroup I_0 . Now, since θ is assumed to be in the β -part of the cohomology, I_0 acts via the character β of I_0/I_1^* , so that $\psi_{\mathfrak{m}}|_{\mathbb{Z}_p^{\times}} = \beta_1 \cdots \beta_n$, as required.

Corollary 3.4.5. Suppose that $\rho_{\mathfrak{m}}$ satisfies Hypothesis 3.4.1, that $\rho_{\mathfrak{m}|G_{\mathbb{Q}_p}}$ is rregular for some $r \leq \min\{(n-1)/2, p-2\}$, and, if r = (n-1)/2, suppose in addition that $\rho_{\mathfrak{m}|G_{\mathbb{Q}_p}}$ contains an irreducible subquotient of dimension greater than 1. Then the localisation $H^i_{\acute{e}t}(X_{\overline{\mathbb{Q}}}, \overline{\mathbb{Z}}_p)_{\mathfrak{m}}$ vanishes for $i \leq r$, while $H^{r+1}_{\acute{e}t}(X_{\overline{\mathbb{Q}}}, \overline{\mathbb{Z}}_p)_{\mathfrak{m}}$ is torsion-free. *Proof.* This follows at once from Theorem 3.4.2 and the short exact sequence

$$0 \longrightarrow H^{i}_{\text{\acute{e}t}}(X_{\overline{\mathbb{Q}}}, \overline{\mathbb{Z}}_{p})_{\mathfrak{m}}/\mathfrak{m}_{\overline{\mathbb{Z}}_{p}}H^{i}_{\text{\acute{e}t}}(X_{\overline{\mathbb{Q}}}, \overline{\mathbb{Z}}_{p})_{\mathfrak{m}} \longrightarrow H^{i}_{\text{\acute{e}t}}(X_{\overline{\mathbb{Q}}}, \overline{\mathbb{F}}_{p})_{\mathfrak{m}} \longrightarrow H^{i+1}_{\text{\acute{e}t}}(X_{\overline{\mathbb{Q}}}, \overline{\mathbb{Z}}_{p})_{\mathfrak{m}}[\mathfrak{m}_{\overline{\mathbb{Z}}_{p}}] \longrightarrow 0.$$

Remark 3.4.6. As already remarked in the introduction, we expect some kind of mod p analogue of Arthur's conjectures to hold, and so in particular we expect that stronger results than Theorem 3.4.2 and Corollary 3.4.5 should hold. In particular, if m is any maximal ideal in the Hecke algebra attached to an irreducible continuous representation $\rho_{\mathfrak{m}}: G_F \to \operatorname{GL}_n(\overline{\mathbb{F}}_p)$, then we expect that the localisations $H^i_{\text{ét}}(X_{\overline{\mathbb{Q}}}, \overline{\mathscr{F}}_V)_{\mathfrak{m}}$ and $H^i_{\text{ét}}(X_{\overline{\mathbb{Q}}}, \overline{\mathbb{Z}}_p)_{\mathfrak{m}}$ should vanish in degrees i < n - 1.

On the other hand, it need not be the case that (for example) $H_{\text{ét}}^i(X_{\overline{\mathbb{Q}}}, \overline{\mathbb{F}}_p)$ vanishes in all degrees in which $H_{\text{\acute{e}t}}^i(X_{\overline{\mathbb{Q}}}, \overline{\mathbb{Q}}_p)$ vanishes. For example, in the case n = 3, for the unitary Shimura varieties that we consider here, namely those that are associated to division algebras, it is known that $H_{\text{\acute{e}t}}^1(X_{\overline{\mathbb{Q}}}, \overline{\mathbb{Q}}_p) = 0$ [Rogawski 1990, Theorem 15.3.1]. (Under additional restrictions on the division algebra allowed, an analogous result is known for all values of n [Clozel 1993, Theorem 3.4].) On the other hand, one can construct examples for which $H_{\text{\acute{e}t}}^1(X_{\overline{\mathbb{Q}}}, \overline{\mathbb{F}}_p) \neq 0$, and hence for which $H_{\text{\acute{e}t}}^2(X_{\overline{\mathbb{Q}}}, \overline{\mathbb{Z}}_p)$ is not torsion-free, via congruence cohomology. (See e.g., the proof of [Suh 2008, Theorem 3.4].) The existence of such classes does not contradict our theorems or expectations, since congruence cohomology is necessarily Eisenstein (i.e., gives rise to Eisenstein systems of Hecke eigenvalues, in the sense that the associated Galois representation ρ_{m} is completely reducible).

3.5. On the mod p cohomology of certain U(2, 1)-Shimura varieties. Let X := X(K) denote a U(2, 1)-Shimura variety, with K of level dividing N for some natural number N divisible by p, and such that the projection of K to $\mathbb{G}(\mathbb{A}^{p,\infty})$ is sufficiently small. Let \mathcal{F}_V be a canonical local system on X, which may be trivialised at level N. The results of Section 3.4 are particularly powerful in this case, as we now demonstrate.

Corollary 3.5.1. Suppose that $\rho_{\mathfrak{m}}$ satisfies Hypothesis 3.4.1, that $\rho_{\mathfrak{m}}|_{G_{\mathbb{Q}_p}}$ is 1-regular, and that $\rho_{\mathfrak{m}}|_{G_{\mathbb{Q}_p}}$ contains an irreducible subquotient of dimension greater than 1. Then the localisations $H^i_{\acute{e}t}(X_{\overline{\mathbb{Q}}}, \mathscr{F}_V)_{\mathfrak{m}}$ vanish for $i \neq 2$.

Proof. This follows immediately from Theorem 3.4.2, noting that the hypothesis that $\rho_{\mathfrak{m}}|_{G_{\mathbb{Q}_p}}$ is 1-regular implies that p > 4 (indeed, that $p \ge 11$).

We now prove a result which does not require the existence of a Galois representation ρ_m . We begin with a lemma:

Lemma 3.5.2. If \mathfrak{m} is a maximal ideal of \mathbb{T} with residue field $\overline{\mathbb{F}}_p$, such that $H^0_{\mathrm{\acute{e}t}}(X_{\overline{\mathbb{Q}}}, \mathcal{F}_V)_{\mathfrak{m}} \neq 0$, then there is an abelian representation $\overline{\rho}_0 : G_F \to \mathrm{GL}_3(\overline{\mathbb{F}}_p)$ such that $\mathrm{char}(\overline{\rho}_0(\mathrm{Frob}_w)) = \sum_{i=0}^3 (-1)^i (\mathrm{Norm} \, w)^{i(i-1)/2} T^{(i)}_w X^{n-i} \pmod{\mathfrak{m}}$ for all split places w of E for which $w \nmid Np$.

Proof. This is standard and follows for example from [Deligne 1979, Section 2.1].

Theorem 3.5.3. If ρ is a 3-dimensional irreducible sub- G_F -representation of the étale cohomology group $H^1_{\acute{e}t}(X_{\overline{\mathbb{Q}}}, \mathscr{F}_V)$, then every irreducible subquotient of $\rho|_{G_{\mathbb{Q}_p}}$ is 1-dimensional, or else $\rho|_{G_{\mathbb{Q}_p}}$ is not 1-regular, or else $\rho(G_F)$ is not generated by its subset of regular elements.

Remark 3.5.4. Recall that a square matrix is said to be regular if its minimal and characteristic polynomials coincide. In Section 4.2 we will show that $\rho(G_F)$ is generated by its subset of regular elements if either ρ_m is induced from a character of G_K for some cubic Galois extension K/\mathbb{Q} , or if $\rho(G_F)$ contains a regular unipotent element.

Remark 3.5.5. In the proof of the theorem we use some of the results of Section 4.

Proof of Theorem 3.5.3. The argument follows similar lines to the proof of Theorem 3.4.2, although it is slightly more involved, since we are not giving ourselves the existence of the Galois representation ρ_m .⁴ The key point will be that, in the Hochschild–Serre spectral sequences that appear, the only other cohomology to contribute besides H^1 will be H^0 , and, for maximal ideals of \mathbb{T} in the support of H^0 , we do have associated Galois representations, by Lemma 3.5.2.

We first show that if $H^0_{\text{ét}}(X_{\overline{\mathbb{Q}}}, \mathcal{F}_W)_{\mathfrak{m}} \neq 0$ for some canonical local system \mathcal{F}_W on X and some maximal ideal \mathfrak{m} of \mathbb{T} , then $\operatorname{Hom}_{G_F}(\rho, H^1_{\text{\acute{et}}}(X_{\overline{\mathbb{Q}}}, \mathcal{F}_V)[\mathfrak{m}]) = 0$. To see this, note that if $H^0_{\text{\acute{et}}}(X_{\overline{\mathbb{Q}}}, \mathcal{F}_W)_{\mathfrak{m}} \neq 0$ and $\operatorname{Hom}_{G_F}(\rho, H^1_{\text{\acute{et}}}(X_{\overline{\mathbb{Q}}}, \mathcal{F}_V)[\mathfrak{m}]) \neq 0$, then Lemma 3.5.2 and Theorem 3.3.1 together imply that there exists an abelian representation $\overline{\rho}_0 : G_F \to \operatorname{GL}_3(\overline{\mathbb{F}}_p)$ such that, for all $g \in G_F$, the characteristic polynomial of $\overline{\rho}_0(g)$ annihilates $\rho(g)$. By Lemma 4.1.3 this implies that ρ is abelian, which is impossible as ρ is irreducible.

Now $H^1_{\text{\acute{e}t}}(X_{\overline{\mathbb{Q}}}, \mathcal{F}_V)$ is the direct sum of its localisations at the various maximal ideals m of \mathbb{T} , and hence, since $\text{Hom}_{G_F}(\rho, H^1_{\text{\acute{e}t}}(X, \mathcal{F}_V)) \neq 0$ by hypothesis, we see that $\text{Hom}_{G_F}(\rho, H^1_{\text{\acute{e}t}}(X, \mathcal{F}_V)_{\mathfrak{m}}) \neq 0$ for some maximal ideal m of \mathbb{T} . Since this is a finite-length $\mathbb{T}_{\mathfrak{m}}$ -module, we see that its $\mathbb{T}_{\mathfrak{m}}$ -socle $\text{Hom}_{G_F}(\rho, H^1_{\text{\acute{e}t}}(X, \mathcal{F}_V)[\mathfrak{m}])$ must also be nonzero, and hence, by the preceding paragraph, we conclude that $H^0_{\text{\acute{e}t}}(X, \mathcal{F}_W)_{\mathfrak{m}} = 0$ for any canonical local system \mathcal{F}_W on X.

⁴In fact, as noted in the introduction, recent work of Scholze [2013] implies that ρ_m exists for any maximal ideal m in the Hecke algebra. We have left our argument as originally written.

As in the proof of Theorem 3.4.2, choose a short exact sequence of canonical local systems $0 \to \mathcal{F}_V \to \pi_* \overline{\mathbb{F}}_p^n \to \mathcal{F}_W \to 0$, for some $\pi : X' \to X$. Passing to the long exact sequence

$$H^{0}_{\mathrm{\acute{e}t}}(X, \mathcal{F}_{W})_{\mathfrak{m}} \longrightarrow H^{1}_{\mathrm{\acute{e}t}}(X, \mathcal{F}_{V})_{\mathfrak{m}} \longrightarrow H^{1}_{\mathrm{\acute{e}t}}(X, \pi_{*}\overline{\mathbb{F}}_{p})^{n} = H^{1}_{\mathrm{\acute{e}t}}(X', \overline{\mathbb{F}}_{p})^{n},$$

and using the result of the preceding paragraph, namely that $H^0_{\text{ét}}(X, \mathcal{F}_W)_{\mathfrak{m}} = 0$, we conclude that ρ embeds into $H^1_{\text{\acute{e}t}}(X', \overline{\mathbb{F}}_p)$. Thus, replacing X by X', we reduce to the case when \mathcal{F}_V is constant, which we assume from now on.

We now suppose that ρ is a subrepresentation of $H^1_{\text{ét}}(X_{\overline{\mathbb{Q}}}, \overline{\mathbb{F}}_p)_{\mathfrak{m}}$. We will prove that ρ is then necessarily a subrepresentation of $H^1_{\text{\acute{et}}}(X(K^p I_1), \overline{\mathbb{F}}_p)_{\mathfrak{m}}$ for some sufficiently small open subgroup K^p of $G(\mathbb{A}^{\infty, p})$. The result will then follow from Lemmas 3.1.3 and 4.2.2 and Theorems 1.5.15, 3.3.1, and 2.2.4.

As in the proof of Theorem 3.4.2, we write X = X(K), and choose a normal open subgroup $K' := K^p K_p$ of K, with $K_p \subset I_1$. The Hochschild–Serre spectral sequence associated to the cover $X(K') \to X(K)$ gives rise to an exact sequence

$$0 \longrightarrow H^{1}(K/K', H^{0}(X(K'), \overline{\mathbb{F}}_{p})_{\mathfrak{m}}) \longrightarrow H^{1}_{\acute{e}t}(X(K), \overline{\mathbb{F}}_{p})_{\mathfrak{m}} \longrightarrow H^{1}_{\acute{e}t}(X(K'), \overline{\mathbb{F}}_{p})_{\mathfrak{m}}^{K/K'} \longrightarrow H^{2}(K/K', H^{0}(X(K'), \overline{\mathbb{F}}_{p})_{\mathfrak{m}}).$$

The same argument as above, using Lemma 3.5.2 (applied now with X(K') in place of X), Theorem 3.3.1, and Lemma 4.1.3 below, shows that

$$H^{1}(K/K', H^{0}(X(K'), \overline{\mathbb{F}}_{p})_{\mathfrak{m}}) = H^{2}(K/K', H^{0}(X(K'), \overline{\mathbb{F}}_{p})_{\mathfrak{m}}) = 0.$$

Thus in fact we have an isomorphism $H^1_{\text{\'et}}(X(K), \overline{\mathbb{F}}_p) \xrightarrow{\sim} H^1_{\text{\'et}}(X(K'), \overline{\mathbb{F}}_p)_{\mathfrak{m}}^{K/K'}$, and hence an isomorphism

$$\operatorname{Hom}_{G_F}(\rho, H^1_{\operatorname{\acute{e}t}}(X(K), \overline{\mathbb{F}}_p)) \xrightarrow{\sim} \operatorname{Hom}_{G_F}(\rho, H^1_{\operatorname{\acute{e}t}}(X(K'), \overline{\mathbb{F}}_p)_{\mathfrak{m}}^{K/K'}) \\ = \operatorname{Hom}_{G_F}(\rho, H^1_{\operatorname{\acute{e}t}}(X(K'), \overline{\mathbb{F}}_p)_{\mathfrak{m}})^{K/K'}.$$

In particular, if ρ appears as a subrepresentation of $H^1_{\text{\'et}}(X(K), \overline{\mathbb{F}}_p)_{\mathfrak{m}}$, then it appears as a subrepresentation of $H^1_{\text{\'et}}(X(K'), \overline{\mathbb{F}}_p)_{\mathfrak{m}}$.

Now, considering the Hochschild–Serre spectral sequence for the cover $X(K') \rightarrow X(K^p I_1)$, and using the fact that if $\operatorname{Hom}_{G_F}(\rho, H^1_{\operatorname{\acute{e}t}}(X(K'), \overline{\mathbb{F}}_p)_{\mathfrak{m}}) \neq 0$, then also $\operatorname{Hom}_{G_F}(\rho, H^1_{\operatorname{\acute{e}t}}(X(K'), \overline{\mathbb{F}}_p)_{\mathfrak{m}})^{I_1/K_p} \neq 0$ (since I_1/K_p is a *p*-group), we conclude that if ρ appears as a subrepresentation of $H^1_{\operatorname{\acute{e}t}}(X(K'), \overline{\mathbb{F}}_p)_{\mathfrak{m}}$, then it appears as a subrepresentation of $H^1_{\operatorname{\acute{e}t}}(X(K'), \overline{\mathbb{F}}_p)_{\mathfrak{m}}$.

Arguing exactly as in the proof of Theorem 3.4.2, we then deduce that some twist of ρ appears in $H^1_{\text{ét}}(X(K^p I_1^*), \overline{\mathbb{F}}_p)$, and so, replacing ρ by this twist, it suffices to prove that if ρ is an irreducible 3-dimensional representation ρ of $H^1_{\text{ét}}(X(K^p I_1^*), \overline{\mathbb{F}}_p)$ that is generated by its regular elements, then either every irreducible subquotient of $\rho_{|G_{\mathbb{Q}_p}}$ is 1-dimensional, or else $\rho_{|G_{\mathbb{Q}_p}}$ is not 1-regular.

This follows from Lemma 3.1.3 and Theorems 1.5.15, 3.3.1, and 2.2.4 exactly as in the proof of Theorem 3.4.2, replacing the appeal to Hypothesis 3.4.1 with one to Lemma 4.2.2 below. \Box

Our other main theorem concerns the weight part of the Serre-type conjecture of [Herzig 2009] for U(2, 1). It is proved by combining our techniques with those of [Emerton et al. 2013], where a similar theorem is proved for U(3) (which is simpler, because one has vanishing of cohomology outside of degree 0). We begin by recalling some terminology from that work. We will call an irreducible $\overline{\mathbb{F}}_p$ -representation of $GL_3(\mathbb{F}_p)$ a *Serre weight*. Fix an irreducible representation $\overline{\rho}: G_F \to GL_3(\overline{\mathbb{F}}_p)$. Let X := X(K) be a U(2, 1)-Shimura variety such that Kis of level dividing N and has sufficiently small projection to $\mathbb{G}(\mathbb{A}^{p,\infty})$, where now we assume that (N, p) = 1. Assume furthermore that $\overline{\rho}$ is unramified at all places not dividing Np, and define a maximal ideal \mathfrak{m} of \mathbb{T} with residue field $\overline{\mathbb{F}}_p$ by demanding that, for each place $w \nmid Np$ of F such that $w|_{\mathbb{Q}}$ splits in F, the characteristic polynomial of $\overline{\rho}(\operatorname{Frob}_w)$ is equal to the reduction modulo \mathfrak{m} of $\sum_{i=0}^3 (-1)^i (\operatorname{Norm} w)^{i(i-1)/2} T_w^{(i)} X^{n-i}$.

Let V be a Serre weight; since (N, p)=1, we may write $K = K_p K^p$, where $K_p \subset G(\mathbb{Q}_p) \cong \operatorname{GL}_3(\mathbb{Q}_p) \times \mathbb{Q}_p^{\times}$ is conjugate to $\operatorname{GL}_3(\mathbb{Z}_p) \times \mathbb{Z}_p^{\times}$, and we may regard V as a representation of K_p via the projection $\operatorname{GL}_3(\mathbb{Z}_p) \twoheadrightarrow \operatorname{GL}_3(\mathbb{F}_p)$. As usual, write \mathcal{F}_V for the canonical local system associated to V. We say that $\overline{\rho}$ is *modular* of weight V if for some N, X as above and for some $0 \le i \le 4$ we have

$$H^{l}_{\mathrm{\acute{e}t}}(X_{\overline{\mathbb{O}}}, \mathcal{F}_{V})_{\mathfrak{m}} \neq 0.$$

Assume now that $\bar{\rho}|_{G_{\mathbb{Q}p}}$ is irreducible. Definition 6.2.2 of [Emerton et al. 2013] defines what it means for a Serre weight to be (strongly) generic, and Section 5.1 there (using the recipe of [Herzig 2009]) defines a set $W^{?}(\bar{\rho})$ of Serre weights in which it is predicted that $\bar{\rho}$ is modular. Let $W_{\text{gen}}(\bar{\rho})$ be the set of generic weights for which $\bar{\rho}$ is modular.

Theorem 3.5.6. Suppose that $\bar{\rho}$ satisfies Hypothesis 4.1.1 and $\bar{\rho}|_{G_{\mathbb{Q}_p}}$ is irreducible and 1-regular. Suppose that $\bar{\rho}$ is modular of some strongly generic weight. Then $W_{\text{gen}}(\bar{\rho}) = W^?(\bar{\rho})$. In fact, for each $V \in W_{\text{gen}}(\bar{\rho})$, we have

$$H^i_{\mathrm{\acute{e}t}}(X_{\overline{\mathbb{Q}}}, \mathcal{F}_V)_{\mathfrak{m}} \neq 0$$

if and only if i = 2 *and* $V \in W^{?}(\bar{\rho})$ *.*

Proof. By the definition of \mathfrak{m} , the representation $\overline{\rho}$ satisfies the defining properties of the representation $\rho_{\mathfrak{m}}$ considered in Section 3.4. Applying Corollary 3.5.1, we see that for any Serre weight V we have $H^i_{\text{ét}}(X_{\overline{\mathbb{Q}}}, \mathcal{F}_V)_{\mathfrak{m}} = 0$ if $i \neq 2$. We will now deduce the result from Theorem 6.2.3 of [Emerton et al. 2013] (taking \overline{r} there to be our $\overline{\rho}$). By Theorem 4.3.3 of that reference, we see that it suffices to show that we can define S and \tilde{S} as in Section 4 there, so that Axioms $\tilde{A}1-\tilde{A}3$ of Section 4.3 of that work are satisfied. Following [Emerton et al. 2013], we define S and \tilde{S} using completed cohomology in the sense of [Emerton 2006] (in [Emerton et al. 2013] the use of completed cohomology was somewhat disguised, but the constructions with algebraic modular forms there are equivalent to the use of completed cohomology of U(3) in degree 0). In fact, given our vanishing results the verification of the axioms of [Emerton et al. 2013] is very similar to that carried out in that work for U(3), and we content ourselves with sketching the arguments.

From now on we regard the prime-to-*p* level structure K^p of *X* as fixed, and we will vary K_p in our arguments. We will write $K_p(0)$ for $GL_3(\mathbb{Z}_p) \times \mathbb{Z}_p^{\times} \subset G(\mathbb{Q}_p)$. We fix a sufficiently large extension E/\mathbb{Q}_p with ring of integers \mathbb{O}_E , residue field k_E , and uniformiser ϖ_E , and we define

$$S := \varinjlim_{K_p} H_{\text{\'et}}^2 (X(K^p K_p)_{\overline{\mathbb{Q}}}, \overline{\mathbb{F}}_p)_{\mathfrak{m}},$$
$$\widetilde{S} := \left(\left(\varinjlim_s \varinjlim_{K_p} H_{\text{\'et}}^2 (X(K^p K_p)_{\overline{\mathbb{Q}}}, \mathbb{O}_E / \varpi_E^s)_{\mathfrak{m}} \right) \otimes_{\mathbb{O}_E} \overline{\mathbb{Z}}_p \right)^{\text{l.alg}}$$

(that is, the locally algebraic vectors in the localisation at m of the completed cohomology of degree 2). Using the Hochschild–Serre spectral sequence and the vanishing of $H_{\text{ét}}^i(X_{\overline{\mathbb{Q}}}, \mathcal{F}_V)_{\mathfrak{m}} = 0$ if $i \neq 2$, it is straightforward to verify Axioms $\tilde{A}1-\tilde{A}3$ of Section 4.3 of [Emerton et al. 2013], as we now explain.

First, we need to check that our definition of "modular" is consistent with that of Definition 4.2.2 of [Emerton et al. 2013]. This amounts to showing that for any Serre weight V

$$(S \otimes \overline{\mathbb{F}}_p V)^{K_p(0)} = H^2_{\text{\'et}}(X(K_p(0)K^p)_{\overline{\mathbb{Q}}}, \mathcal{F}_V)_{\mathfrak{m}}$$

To see this, note that since any sufficiently small K_p acts trivially on V, we have

$$S \otimes \overline{\mathbb{F}}_p V = \varinjlim_{K_p} H^2_{\text{\'et}}(X(K^p K_p)_{\overline{\mathbb{Q}}}, \overline{\mathbb{F}}_p)_{\mathfrak{m}} \otimes V \xrightarrow{\simeq} \varinjlim_{K_p} H^2_{\text{\'et}}(X(K^p K_p)_{\overline{\mathbb{Q}}}, \mathcal{F}_V)_{\mathfrak{m}},$$

so it is enough to check that for all compact open subgroups $K_p \subset K_p(0)$ we have

$$H^{2}_{\mathrm{\acute{e}t}}(X(K^{p}K_{p})_{\overline{\mathbb{Q}}}, \mathcal{F}_{V})^{K_{p}(0)}_{\mathfrak{m}} = H^{2}_{\mathrm{\acute{e}t}}(X(K^{p}K_{p}(0))_{\overline{\mathbb{Q}}}, \mathcal{F}_{V})_{\mathfrak{m}},$$

which is an easy consequence of the Hochschild–Serre spectral sequence and our vanishing result. We also need an embedding $S \hookrightarrow \widetilde{S} \otimes \overline{\mathbb{Z}}_p \overline{\mathbb{F}}_p$ which is compatible with the actions of $GL_3(\mathbb{Q}_p)$ and the Hecke algebra. In fact, it is easy to see that we have $\widetilde{S} \otimes \overline{\mathbb{Z}}_p \overline{\mathbb{F}}_p = S$. For example, there is a natural isomorphism

$$H^{3}_{\mathrm{\acute{e}t}}(X_{\overline{\mathbb{Q}}}, \mathbb{O}_{E})_{\mathfrak{m}}/\varpi_{E}H^{3}_{\mathrm{\acute{e}t}}(X_{\overline{\mathbb{Q}}}, \mathbb{O}_{E})_{\mathfrak{m}} = H^{3}_{\mathrm{\acute{e}t}}(X_{\overline{\mathbb{Q}}}, k_{E})_{\mathfrak{m}},$$

and hence the vanishing of $H^3_{\text{ét}}(X_{\overline{\mathbb{Q}}}, k_E)_{\mathfrak{m}}$ implies that of the finitely generated \mathbb{O}_E -module $H^3_{\text{ét}}(X_{\overline{\mathbb{Q}}}, \mathbb{O}_E)_{\mathfrak{m}}$. One then sees that for all *s* we have a natural isomorphism

$$H^{2}_{\mathrm{\acute{e}t}}(X_{\overline{\mathbb{Q}}}, \mathbb{O}_{E})_{\mathfrak{m}} / \varpi^{s}_{E} H^{2}_{\mathrm{\acute{e}t}}(X_{\overline{\mathbb{Q}}}, \mathbb{O}_{E})_{\mathfrak{m}} = H^{2}_{\mathrm{\acute{e}t}}(X_{\overline{\mathbb{Q}}}, \mathbb{O}_{E} / \varpi^{s}_{E})_{\mathfrak{m}},$$

from which the claim follows easily.

We now examine Axiom A1. We must show that if \widetilde{V} is a finite free $\overline{\mathbb{Z}}_p$ -module with a locally algebraic action of $K_p(0)$ (acting through $\operatorname{GL}_3(\mathbb{Z}_p)$), then $(\widetilde{S} \otimes \overline{\mathbb{Z}}_p \widetilde{V})^{K_p(0)}$ is a finite free $\overline{\mathbb{Z}}_p$ -module, and for $A = \overline{\mathbb{Q}}_p$, $\overline{\mathbb{F}}_p$ we have

$$(\widetilde{S} \otimes \overline{\mathbb{Z}}_p \widetilde{V})^{K_p(0)} \otimes \overline{\mathbb{Z}}_p A = (\widetilde{S} \otimes \overline{\mathbb{Z}}_p \widetilde{V} \otimes \overline{\mathbb{Z}}_p A)^{K_p(0)}.$$

This is straightforward, the key point being that if $\mathcal{F}_{\widetilde{V}}$ denotes the lisse étale sheaf attached to \widetilde{V} , then a straightforward argument with Hochschild–Serre as above gives

$$(\widetilde{S} \otimes \overline{\mathbb{Z}}_p \widetilde{V})^{K_p(0)} = H^2_{\text{\'et}}(X(K^p K_p(0))_{\overline{\mathbb{Q}}}, \mathcal{F}_{\widetilde{V}})_{\mathfrak{m}},$$

which is certainly a finite free $\overline{\mathbb{Z}}_p$ -module (it is torsion-free by the proof of Corollary 3.4.5).

The verification of Axioms $\tilde{A}2$ and $\tilde{A}3$ is now exactly the same as in Proposition 7.4.4 of [Emerton et al. 2013], as the Galois representations occurring in the localised cohomology module $H^2_{\text{ét}}(X(K^pK_p(0))_{\overline{\mathbb{Q}}}, \mathcal{F}_{\widetilde{V}})_{\mathfrak{m}}$ are associated to automorphic forms exactly as in that work.⁵ (In fact, at least for Axiom $\tilde{A}2$ this is a rather roundabout way of proceeding, as the Galois representations in question are constructed in [Harris and Taylor 2001] by using $H^2_{\text{ét}}(X(K^pK_p(0))_{\overline{\mathbb{Q}}}, \mathcal{F}_{\widetilde{V}})$, and one can read off the required properties directly from the comparison theorems of *p*-adic Hodge theory. For Axiom $\tilde{A}3$ we are not aware of any comparison theorems in sufficient generality, so it is necessary at present to take a lengthier route through the theory of automorphic forms.)

4. Group theory lemmas

The theorems of Section 3 contain certain hypotheses on the Galois representations involved. Our goal in this section is to establish some group-theoretic lemmas which give sufficient criteria for these hypotheses to be satisfied. Throughout the section G is a finite group and k is an algebraically closed field of characteristic p. For any square matrix A with entries in k, we write char(A) to denote the characteristic polynomial of A.

⁵In the interests of full disclosure, we are not aware of a reference in the literature giving the precise base change result from U(2, 1) to GL₃ that we need, but it seems to be well-known to the experts, and will follow from the much more general work in progress of Mok and Kaletha, Minguez, Shin and White.

4.1. Characterising representations by their characteristic polynomials. Let ρ : $G \rightarrow GL_n(k)$ be an irreducible representation. In this subsection we establish some criteria for ρ to satisfy the following hypothesis:

Hypothesis 4.1.1. If $\theta : G \to \operatorname{GL}_m(k)$ is irreducible, and if $\operatorname{char}(\rho(g))$ annihilates $\theta(g)$ for every $g \in G$, then θ is equivalent to ρ .

Remark 4.1.2. Any irreducible ρ of dimension 2 satisfies Hypothesis 4.1.1, as was proved by Mazur [1977, proof of Proposition 14.2]. However, it is not satisfied in general if the dimension *n* of ρ is greater than 2 (for instance, this already fails if ρ is the irreducible 3-dimensional representation of A_4 ; see Section 5 of [Boston et al. 1991] as well as Remark 4.1.7 below).

Lemma 4.1.3. Let $\rho : G \to \operatorname{GL}_n(k)$ and $\theta : G \to \operatorname{GL}_m(k)$ be two representations. If θ is irreducible, and if $\operatorname{char}(\rho(g))$ annihilates $\theta(g)$ for every $g \in G$, then the kernel of θ contains the kernel of ρ .

Proof. If $\rho(g)$ is trivial, then the assumption implies that every eigenvalue of $\theta(g)$ is equal to 1, and hence that $\theta(g)$ is unipotent, and so of order a power of p. Thus the image of ker (ρ) under θ is a normal subgroup H of $\theta(G)$ of p-power order, and we see that the space of invariants $(k^m)^H$ is a nontrivial subspace of k^m . Since H is normal in $\theta(G)$, we see that $\theta(G)$ leaves $(k^m)^H$ invariant, and hence, since θ is assumed to be irreducible, we see that in fact $(k^m)^H = k^m$. Thus H is trivial, which is to say that ker $(\rho) \subset \ker(\theta)$, as claimed.

Lemma 4.1.4. If $\rho : G \to \operatorname{GL}_n(k)$ is a direct sum of 1-dimensional characters of G, and if $\theta(g) : G \to \operatorname{GL}_m(k)$ is an irreducible representation of G such that $\operatorname{char}(\rho(g))$ annihilates $\theta(g)$ for every $g \in G$, then m = 1, so that θ is a character, and every element of G lies in the kernel of at least one of the summands of $\rho \otimes \theta^{-1}$.

Proof. Since ρ is a direct sum of characters, it factors through G^{ab} . Lemma 4.1.3 then shows that θ also factors through G^{ab} . Since θ is also assumed to be irreducible, we find that θ must be a character. Twisting by ρ by θ^{-1} , we may in fact assume that θ is trivial, and, writing $\rho = \chi_1 \oplus \cdots \oplus \chi_n$, we find that for each $g \in G$, the value $\chi_i(g)$ is equal to 1 for at least one value of *i* (since char($\rho(g)$) = $(X - \chi_1(g)) \cdots (X - \chi_n(g))$ annihilates $\theta(g) = 1$). Thus *G* is equal to the union of its subgroups ker(χ_i). \Box

Remark 4.1.5. In the context of the preceding proposition, we can't conclude in general that θ coincides with one the summands of ρ . For example, if *G* denotes the Klein four-group, if *p* is odd, and if ρ denotes the 3-dimensional representation obtained by taking the direct sum of the three nontrivial characters of *G*, then, taking θ to be the trivial representation, the hypotheses of the proposition are satisfied, but θ is certainly not one of the summands of ρ .

Lemma 4.1.6. If $\rho: G \to \operatorname{GL}_n(k)$ is irreducible, and is isomorphic to an induction $\operatorname{Ind}_H^G \psi$, where H is a cyclic normal subgroup of G of index n and $\psi: H \to k^{\times}$ is a character, then ρ satisfies Hypothesis 4.1.1.

Proof. The restriction $\rho|_H$ is isomorphic to the direct sum $\bigoplus_{g \in G/H} \psi^g$. If we let θ' be a Jordan–Hölder constituent of the restriction $\theta|_H$, then Lemma 4.1.4 (applied to the representations $\rho|_H$ and θ' of H) implies that θ' is a character of H and (because H is cyclic) that $\theta' = \psi^g$ for some $g \in G/H$. The H-equivariant inclusion $\psi^g = \theta' \rightarrow \theta|_H$ then induces a nonzero G-equivariant map $\rho = \operatorname{Ind}_H^G \psi = \operatorname{Ind}_H^G \psi^g \rightarrow \theta$, which must be an isomorphism, since both its source and target are irreducible by assumption. This proves the lemma.

Remark 4.1.7. If we take $G = A_4$ and H to be the normal subgroup of G of order four (so that H is a Klein four-group), then the induction of any nontrivial character of H gives an irreducible representation $\rho : G \to SO_3(k)$. For every $g \in G$, the characteristic polynomial of $\rho(g)$ thus has 1 as an eigenvalue, and so, if θ denotes the trivial character of G, the element $\theta(g)$ is annihilated by char($\rho(g)$) for every $g \in G$. Thus the analogue of Lemma 4.1.6 does not hold in general if H is not cyclic.

We thank Florian Herzig for providing the proof of the following lemma:

Lemma 4.1.8. Suppose that G is a finite subgroup of $GL_n(k)$ which contains $SL_n(k')$ for some subfield k' of k and is contained in $k^{\times} GL_n(k')$.

- (1) Any irreducible representation of G over k remains irreducible upon restriction to $SL_n(k')$.
- (2) Given any two irreducible representations of G which become isomorphic upon restriction to $SL_n(k')$, one can be obtained from the other via twisting by a character of G that is trivial on $SL_n(k')$.

Proof. Let *G* act via θ on the *k*-vector space *V*, and let (θ, W) be an irreducible subrepresentation of $\theta|_{SL_n(k')}$. Then *W* is obtained by restriction from a representation of the algebraic group SL_n / k' (see Section 1 of [Jantzen 1987]), so the action of $SL_n(k')$ on *W* may be extended to an action of $GL_n(k)$ and thus of *G*. By Frobenius reciprocity we obtain a surjective map $(Ind_{SL_n(k')}^G 1) \otimes W \to V$ of *G*-representations. Since $G/SL_n(k')$ is a finite abelian group of prime-to-*p* order, we see that $(Ind_{SL_n(k')}^G 1)$ is a direct sum of 1-dimensional representations, so that *V* is a twist of *W* by some character which is trivial on $SL_n(k')$. Thus the restriction of θ to $SL_n(k')$ is just *W*, which is irreducible, proving (1).

This same argument also serves to establish (2).

Lemma 4.1.9. Assume that $p \ge n$. If $\rho : G \to \operatorname{GL}_n(k)$ is irreducible, and if $\operatorname{SL}_n(k') \subseteq \rho(G) \subseteq k^{\times} \operatorname{GL}_n(k')$ for some subfield k' of k, then ρ satisfies Hypothesis 4.1.1.

Proof. The case n = 1 follows from Lemma 4.1.4, and, as remarked above, the case n = 2 is proved in the course of the proof of Proposition 14.2 of [Mazur 1977], so we may assume that $n \ge 3$. By Lemma 4.1.3, we may assume that ρ is faithful, so that we can identify G with $\rho(G)$. In particular, $SL_n(k')$ is a subgroup of G, and, by Lemma 4.1.8, the restriction of θ to $SL_n(k')$ remains irreducible.

Since *G* is finite, our assumption that $SL_n(k') \subset G$ implies that k' is finite; suppose that k' has cardinality *q*. We recall some basic facts about the representation theory of $SL_n(k')$; see for example Section 1 of [Jantzen 1987]. The irreducible *k*-representations of $SL_n(k')$ are obtained by restriction from the algebraic group SL_n/k' , and are precisely those representations whose highest weights are *q*restricted. (With the usual choice of maximal torus *T* of SL_n , if we identify the weight lattice with \mathbb{Z}^n modulo the diagonally embedded copy of \mathbb{Z} , a weight (a_1, \ldots, a_n) is *q*-restricted if $0 \le a_i - a_{i+1} \le q - 1$ for all $1 \le i \le n - 1$.) Suppose that θ has highest weight (a_1, \ldots, a_n) . Let $g \in SL_n(k')$ be a semisimple element with eigenvalues $\alpha_1, \ldots, \alpha_n$. Then, since *g* is conjugate to an element of T(k), by considering the formal character of the corresponding representation of SL_n/k' , we see that among the eigenvalues of $\theta(g)$ are each of the quantities

$$\prod_{i=1}^n \alpha_i^{x_i},$$

where the x_i are a permutation of a_1, \ldots, a_n .

Our assumption on θ and ρ implies that, for each such permutation, $\prod_{i=1}^{n} \alpha_i^{x_i}$ must be one of $\alpha_1, \ldots, \alpha_n$. In particular, if we let α be a primitive $(q^n - 1)/(q - 1)$ -st root of unity, we may consider a semisimple element g with eigenvalues

$$\alpha, \alpha^q, \ldots, \alpha^{q^{n-1}}$$

Then, for any x_1, \ldots, x_n as above, there must be an integer $0 \le \beta \le n-1$ such that

$$q^{n-1}x_1 + q^{n-2}x_2 + \dots + x_n \equiv q^\beta \pmod{(q^n - 1)/(q - 1)}.$$

Fix some $1 \le i \le n-1$, and consider two permutations x_1, \ldots, x_n and x'_1, \ldots, x'_n as above which satisfy $x_i = x'_i$ for $1 \le i \le n-2$, $x_{n-1} = a_i$, $x_n = a_{i+1}$, $x'_{n-1} = a_{i+1}$ and $x'_n = a_i$. Taking the difference of the two expressions

$$q^{n-1}x_1 + q^{n-2}x_2 + \dots + x_n$$
 and $q^{n-1}x'_1 + q^{n-2}x'_2 + \dots + x'_n$,

we conclude that there are integers $0 \le \beta$, $\gamma \le n-1$ such that

$$(q-1)(a_i - a_{i+1}) \equiv q^\beta - q^\gamma \pmod{(q^n - 1)/(q - 1)}.$$

Since $n \ge 3$, we have $0 \le (q-1)(a_i - a_{i+1}) \le (q-1)^2 < (q^n - 1)/(q-1)$, and we conclude that either $\beta \ge \gamma$ and $(q-1)(a_i - a_{i+1}) = q^\beta - q^\gamma$, or $\beta < \gamma$ and $(q-1)(a_i - a_{i+1}) = (q^n - 1)/(q-1) + q^\beta - q^\gamma$. In the second case, we have

$$(q-1)^{2} \ge (q-1)(a_{i} - a_{i+1})$$

= $(q^{n} - 1)/(q-1) + q^{\beta} - q^{\gamma}$
 $\ge (q^{n} - 1)/(q-1) + 1 - q^{n-1}$
= $q^{n-2} + \dots + q + 2$.

This is a contradiction if $n \ge 4$. If n = 3, $(q^3 - 1)/(q - 1) \equiv 3 \pmod{q - 1}$, so that $(q - 1) \mid 3$, which is a contradiction as p > 2.

Thus it must be the case that $\beta \ge \gamma$ and $(q-1)(a_i - a_{i+1}) = q^\beta - q^\gamma$, so that $a_i - a_{i+1}$ is congruent mod p to 0 or 1, and thus $a_i - a_{i+1} = 0$ or 1. In particular, for each i we have $0 \le a_i - a_n \le n - 1 \le p - 1$.

We now repeat the above analysis. Fix some $1 \le i \le n-1$, and consider two permutations x_1, \ldots, x_n and x'_1, \ldots, x'_n as above which satisfy $x_i = x'_i$ for $1 \le i \le n-2$, $x_{n-1} = a_i$, $x_n = a_n$, $x'_{n-1} = a_n$ and $x'_n = a_i$. Taking the difference of the two expressions $q^{n-1}x_1 + q^{n-2}x_2 + \cdots + x_n$ and $q^{n-1}x'_1 + q^{n-2}x'_2 + \cdots + x'_n$, and using that $0 \le a_i - a_n \le p - 1 \le q - 1$, we conclude as before that each $a_i - a_n = 0$ or 1. Thus there must be an integer $1 \le r \le n$ with $a_1 = \cdots = a_r = a_n + 1$, $a_{r+1} = \cdots = a_n$.

Returning to the original congruences

$$q^{n-1}x_1 + q^{n-2}x_2 + \dots + x_n \equiv q^{\beta} \pmod{(q^n - 1)/(q - 1)},$$

we see that the left side is congruent to a sum of precisely *r* distinct values q^j , $1 \le j \le n-1$. Thus r = 1, and $\theta|_{SL_n(k')}$ is the standard representation of $SL_n(k')$, i.e., $\theta|_{SL_n(k')} \cong \rho|_{SL_n(k')}$. By part (2) of Lemma 4.1.8, we see that there is a character $\chi : G \to k^{\times}$ with $\chi|_{SL_n(k')} = 1$ such that $\theta \cong \rho \otimes \chi$.

To complete the proof, we must show that χ is trivial. Take $g \in G$; we will show that $\chi(g) = 1$. If $g' \in G$ has det $g' = \det g$, then $g(g')^{-1} \in \operatorname{SL}_n(k')$, so $\chi(g') = \chi(g)$. First, note that by assumption we can write $g = \lambda h$, with $\lambda \in k^{\times}$, $h \in \operatorname{GL}_n(k')$. Choose $h' \in \operatorname{GL}_n(k')$ to have eigenvalues $\{1, \ldots, 1, \det(h)\}$. Then $h(h')^{-1} \in \operatorname{SL}_n(k') \subset G$, so $g' := \lambda h'$ is an element of G. Then the hypothesis on θ and ρ shows that the eigenvalues of $\chi(g')g' = \chi(g)g'$ are contained in the set of eigenvalues of g', so that if $\chi(g) \neq 1$ we must have $\chi(g) = \det(h) = -1$. Assume for the sake of contradiction that this is the case. If n is odd, then we now choose h' to have eigenvalues $\{-1, \ldots, -1\}$, and we immediately obtain a contradiction from the same argument. If n is even then since $p \ge n$ we have $p \ge 5$ (recall that we are assuming $n \ge 3$), and we may choose $a \in (k')^{\times}$, $a \neq \pm 1$. Then choosing h'to have eigenvalues $\{1, \ldots, 1, a, -1/a\}$ gives a contradiction. **4.2.** *Representations whose image is generated by regular elements.* Recall that a square matrix with entries in *k* is said to be *regular* if its minimal polynomial and characteristic polynomial coincide.

Lemma 4.2.1. If $\rho: G \to \operatorname{GL}_n(k)$ and $\theta: G \to \operatorname{GL}_n(k)$ are representations such that the image $\theta(G)$ is generated by its subset of regular elements, and for every $g \in G$ the characteristic polynomial of $\rho(g)$ annihilates $\theta(g)$, then det $\rho = \det \theta$.

Proof. Let $g \in G$ be an element such that $\theta(g)$ is regular. Then the characteristic polynomials of $\rho(g)$ and $\theta(g)$ must be equal (since the minimal and characteristic polynomials of $\theta(g)$ coincide and the characteristic polynomial of $\rho(g)$ annihilates $\theta(g)$), so det $\rho(g) = \det \theta(g)$. Since $\theta(G)$ is generated by its subset of regular elements, the result follows.

In fact, we actually need a slight generalisation of this result, where we simply have a collection of characteristic polynomials, rather than a representation ρ . Suppose that for each $g \in G$ we have a monic polynomial

$$\rho_g(X) = X^n - a_1(g)X^{n-1} + \dots + (-1)^n a_n(g) \in k[X]$$

of degree *n* with the property that for all $g, h \in G$, we have $a_n(gh) = a_n(g)a_n(h)$.

Lemma 4.2.2. Suppose that θ : $G \to \operatorname{GL}_n(k)$ is a representation with the property that $\theta(G)$ is generated by its subset of regular elements, and that for each $g \in G$ we have $\rho_g(\theta(g)) = 0$. Then for each $g \in G$ we have $\det \theta(g) = a_n(g)$.

Proof. This may be proved in exactly the same way as Lemma 4.2.1. \Box

Let $\rho: G \to GL_3(k)$ be irreducible. Our goal is to give criteria for ρ to satisfy the following hypothesis, in order to apply the previous lemmas:

Hypothesis 4.2.3. The image $\rho(G)$ is generated by its subset of regular elements.

Lemma 4.2.4. If $\rho : G \to GL_3(k)$ is irreducible, and if either

- (1) ρ is isomorphic to an induction $\operatorname{Ind}_{H}^{G} \psi$, where *H* is a normal subgroup of index 3 in *G* and $\psi : H \to k^{\times}$ is a character, or
- (2) $\rho(G)$ contains a regular unipotent element,

then ρ satisfies Hypothesis 4.2.3.

Proof. Suppose first that ρ is isomorphic to an induction $\operatorname{Ind}_{H}^{G} \psi$. Since H is a proper subgroup of G, the set of elements G - H generates G. If $g \in G - H$ then the characteristic polynomial of $\rho(g)$ is of the form $X^{3} - \alpha$. If $p \neq 3$ then this has distinct roots, so $\rho(g)$ is regular, and if p = 3 then it is easy to check that $\rho(g)$ is the product of a scalar matrix and a unipotent matrix, and is regular.

Suppose now that $\rho(G)$ contains a regular unipotent element. For ease of notation, we will refer to $\rho(G)$ as G from now on. Let H be the subgroup of G generated

by the regular elements, and assume for the sake of contradiction that H is a proper subgroup of G. We claim that H contains every scalar matrix in G; this is true because the product of a scalar matrix and a regular matrix is again a regular matrix. Consider an element $g \in G - H$; since it is not regular, and not scalar, it acts as a scalar on some unique plane in k^3 . We write ℓ_g for the corresponding line in $\mathbb{P}^2(k)$.

Let *h* be the given regular unipotent element in *H*. Then *h* stabilises a unique line in k^3 , so a unique point $P \in \mathbb{P}^2(k)$. As $g \in G - H$, we also have $gh \in G - H$. Then $\ell_g \cap \ell_{gh}$ is nonempty, so there is a point $Q \in \mathbb{P}^2(k)$ which is fixed by *g* and *gh*. It is thus also fixed by *h*, so in fact Q = P. Since *g* was an arbitrary element of G - H, we see that every element of G - H fixes *P*, and since *G* is generated by G - H, this implies that every element of *G* fixes *P*. This contradicts the assumption that ρ is irreducible.

Appendix: Cohomology of pairs

A.1. Étale cohomology of a pair. Let X be a scheme, finite-type and separated over a field, let Z be a closed subscheme, and write $j : U \hookrightarrow X$ for the open immersion of the complement $U := X \setminus Z$ into X. As in Section 1, we let E be an algebraic extension of \mathbb{Q}_p , where p is invertible on X, let k_E denote the residue field of E, and we let A be one of E or k_E . We define the étale cohomology of the pair (X, Z) with coefficients in A to be the étale cohomology of the sheaf $j_!A$ on X, i.e., we write

$$H^{\bullet}_{\text{\'et}}(X, Z, A) := H^{\bullet}_{\text{\'et}}(X, j_! A).$$

If $i : Z \hookrightarrow X$ is the closed immersion of Z, then the short exact sequence

$$0 \longrightarrow j_! A \longrightarrow A \longrightarrow i_* A \longrightarrow 0$$

gives rise to a long exact sequence

$$\cdots \longrightarrow H^m_{\mathrm{\acute{e}t}}(Z,A) \longrightarrow H^{m+1}_{\mathrm{\acute{e}t}}(X,Z,A) \longrightarrow H^{m+1}_{\mathrm{\acute{e}t}}(X,A) \longrightarrow H^{m+1}_{\mathrm{\acute{e}t}}(Z,A) \longrightarrow \cdots,$$

which is the long exact cohomology sequence of the pair (X, Z).

We are particularly interested in the case of a pair $(X \setminus Y, Z \setminus Y)$, where X is a smooth projective variety over a separably closed field, and Y and Z are smooth divisors on X which meet transversely. In this case we have a Cartesian diagram of open immersions

$$\begin{array}{ccc} X \setminus (Y \cup Z) \xrightarrow{j} & X \setminus Y \\ & & \downarrow^{k'} & \downarrow^{k} \\ & X \setminus Z \xrightarrow{j'} & X \end{array} \tag{A.1.1}$$

According to the above definition, the cohomology of the pair $(X \setminus Y, Z \setminus Y)$ is computed as the cohomology of the sheaf $j_!A$ on $X \setminus Y$, which is canonically isomorphic to the cohomology of the complex $Rk_*j_!A$ on X. An important point is that there is a canonical isomorphism

$$j'_{1}Rk'_{*}A \xrightarrow{\sim} Rk_{*}j_{!}A. \tag{A.1.2}$$

(See the discussion of §III (b) on p. 44 of [Faltings 1989].)

A.2. *Verdier duality.* Verdier duality [SGA 4₃ 1973, Exposé XVIII; Verdier 1967] states that if $f: X \to S$ is a morphism of finite-type and separated schemes over a separably closed field k, then, for any constructible étale *A*-sheaves \mathcal{F} on *X* and \mathcal{G} on *S*, there is a canonical isomorphism (in the derived category) of complexes of étale sheaves on *S*

$$RHom(Rf_!\mathcal{F},\mathcal{G}) \cong Rf_*RHom(\mathcal{F}, f^!\mathcal{G}).$$

We recall some standard special cases of this isomorphism, in the context of the diagram (A.1.1).

Taking f to be k' (and recalling that k' is an open immersion), we obtain an isomorphism

$$RHom(k'_1A, A) \cong Rk'_*RHom(A, A) = Rk'_*A,$$

and hence, by double duality, an isomorphism

$$RHom(Rk'_*A, A) \cong k'_1A. \tag{A.2.1}$$

Next, taking f to be j', and taking into account (A.1.2) and (A.2.1), we obtain isomorphisms

$$RHom(Rk_*j_!A, A) \cong RHom(j'_!Rk'_*A, A) \cong Rj'_*RHom(Rk'_*A, A) \cong Rj'_*k'_!A.$$

Finally, taking f to the natural map $X \rightarrow \operatorname{Spec} k$, and recalling that in this case we have

$$f^! A = A[2d](d)$$

(where d is the dimension of X; see [SGA 4₃ 1973, Exposé XVIII, Théorème 3.2.5]) and that $Rf_* = Rf_!$ (since f is proper, the variety X being projective by assumption), we find that

$$RHom(Rf_*Rk_*j_!A, A) \cong Rf_*RHom(Rk_*j_!A, A[2d](d))$$
$$\cong Rf_*Rj'_*k'_!A[2d](d).$$

Passing to cohomology, we find that $H^m_{\text{ét}}(X \setminus Y, Z \setminus Y, A)$ is in natural duality with $H^{2d-m}(X \setminus Z, Y \setminus Z, A)(d)$.

A.3. Vanishing outside of, and torsion-freeness in, the middle degree. We continue to assume that X is a smooth projective variety of dimension d over the separably closed field k, and that Y and Z are smooth divisors on X which meet transversely, but, in addition, we now assume that the complements $X \setminus Y$ and $X \setminus Z$ are affine (and hence also that $Z \setminus Y$ and $Y \setminus Z$ are affine). This latter assumption implies that $H^m_{\text{ét}}(X \setminus Y, A)$ vanishes if m > d and that $H^m_{\text{ét}}(Z \setminus Y, A)$ vanishes if $m \ge d$ [SGA 4₃ 1973, Exposé XIV, Corollaire 3.3]. By the long exact cohomology sequence of the pair $(X \setminus Y, Y \setminus Z)$, we see that $H^m_{\text{ét}}(X \setminus Y, Z \setminus Y, A)$ vanishes if m > d. Similarly, we see that $H^{2d-m}(X \setminus Z, Y \setminus Z, A)(d)$ vanishes if m < d. Hence, by the duality between $H^m_{\text{ét}}(X \setminus Y, Z \setminus Y, A)$ and $H^{2d-m}(X \setminus Z, Y \setminus Z, A)(d)$, we find that both vanish unless m = d.

Let \mathbb{O}_E denote the ring of integers in *E*. Suppose momentarily that E/\mathbb{Q}_p is finite, and let ϖ be a uniformiser of \mathbb{O}_E . From a consideration of the cohomology long exact sequence arising from the short exact sequence of sheaves

$$0 \longrightarrow \mathbb{O}_E / \varpi^n \xrightarrow{\varpi} \mathbb{O}_E / \varpi^{n+1} \longrightarrow k_E \longrightarrow 0,$$

and arguing inductively on *n*, we find that $H^m_{\text{ét}}(X \setminus Y, Z \setminus Y, \mathbb{O}_E/\varpi^n)$ vanishes in degrees other than m = d for all *n*. Passing to the projective limit over *n*, we see that the same is true of $H^m_{\text{ét}}(X \setminus Y, Z \setminus Y, \mathbb{O}_E)$. Finally, a consideration of the cohomology long exact sequence arising from the short exact sequence

$$0 \longrightarrow \mathbb{O}_E \xrightarrow{\varpi} \mathbb{O}_E \longrightarrow k_E \longrightarrow 0$$

shows that $H^d_{\text{ét}}(X \setminus Y, Z \setminus Y, \mathbb{O}_E)$ is torsion-free.

By passage to the direct limit over subfields of E which are finite over \mathbb{Q}_p , we see that these properties continue to hold for arbitrary algebraic extensions E/\mathbb{Q}_p .

Acknowledgements

We would like to thank Brian Conrad, Florian Herzig, Mark Kisin, Kai-Wen Lan, Madhav Nori, Bjorn Poonen, Michael Rapoport, Sug Woo Shin, Junecue Suh, and Teruyoshi Yoshida for helpful correspondence and conversations on the subject of this article. We also thank an anonymous referee for their critical reading of the paper, which led to the filling of some gaps in the arguments of Section 1.5.

References

[[]Arthur 1989] J. Arthur, "Unipotent automorphic representations: conjectures", pp. 13–71 in *Orbites unipotentes et représentations, II*, Astérisque **171–172**, 1989. MR 91f:22030 Zbl 0728.22014

[[]Beilinson 1987] A. A. Beilinson, "On the derived category of perverse sheaves", pp. 27–41 in *K-theory, arithmetic and geometry* (Moscow 1984–1986), edited by Y. I. Manin, Lecture Notes in Math. **1289**, Springer, Berlin, 1987. MR 89b:14027 Zbl 0652.14008

- [Boston et al. 1991] N. Boston, H. W. Lenstra, Jr., and K. A. Ribet, "Quotients of group rings arising from two-dimensional representations", *C. R. Acad. Sci. Paris Sér. I Math.* **312**:4 (1991), 323–328. MR 92c:11057 Zbl 0718.16018
- [Breuil 2000] C. Breuil, "Groupes *p*-divisibles, groupes finis et modules filtrés", *Ann. of Math.* (2) **152**:2 (2000), 489–549. MR 2001k:14087 Zbl 1042.14018
- [Buzzard and Gee 2014] K. Buzzard and T. Gee, "The conjectural connections between automorphic representations and Galois representations", pp. 135–187 in *Automorphic forms and Galois representations*, *I*, edited by F. Diamond et al., Lond. Math. Soc. Lecture Notes **414**, Cambridge Univ. Press, 2014.
- [Caruso 2008] X. Caruso, "Conjecture de l'inertie modérée de Serre", *Invent. Math.* **171**:3 (2008), 629–699. MR 2008j:14034 Zbl 1245.14019
- [Clozel 1993] L. Clozel, "On the cohomology of Kottwitz's arithmetic varieties", *Duke Math. J.* **72**:3 (1993), 757–795. MR 95b:11055 Zbl 0974.11019
- [Conrad et al. 1999] B. Conrad, F. Diamond, and R. Taylor, "Modularity of certain potentially Barsotti–Tate Galois representations", *J. Amer. Math. Soc.* **12**:2 (1999), 521–567. MR 99i:11037 Zbl 0923.11085
- [Deligne 1979] P. Deligne, "Variétés de Shimura: interprétation modulaire, et techniques de construction de modèles canoniques", pp. 247–289 in *Automorphic forms, representations and L-functions, Part 2* (Corvallis, OR, 1977), edited by A. Borel and W. Casselman, Proc. Sympos. Pure Math. 33, Amer. Math. Soc., Providence, R.I., 1979. MR 81i:10032 Zbl 0437.14012
- [Dimitrov 2005] M. Dimitrov, "Galois representations modulo *p* and cohomology of Hilbert modular varieties", *Ann. Sci. École Norm. Sup.* (4) **38**:4 (2005), 505–551. MR 2006k:11100 Zbl 1160.11325
- [Emerton 2006] M. Emerton, "On the interpolation of systems of eigenvalues attached to automorphic Hecke eigenforms", *Invent. Math.* **164**:1 (2006), 1–84. MR 2007k:22018 Zbl 1090.22008
- [Emerton et al. 2013] M. Emerton, T. Gee, and F. Herzig, "Weight cycling and Serre-type conjectures for unitary groups", *Duke Math. J.* 162:9 (2013), 1649–1722. MR 3079258 Zbl 1283.11083
- [Faltings 1989] G. Faltings, "Crystalline cohomology and *p*-adic Galois-representations", pp. 25–80 in *Algebraic analysis, geometry, and number theory* (Baltimore, MD, 1988), edited by J.-I. Igusa, Johns Hopkins Univ. Press, Baltimore, MD, 1989. MR 98k:14025 Zbl 0805.14008
- [Fontaine and Messing 1987] J.-M. Fontaine and W. Messing, "p-adic periods and p-adic étale cohomology", pp. 179–207 in *Current trends in arithmetical algebraic geometry* (Arcata, CA, 1985), edited by K. A. Ribet, Contemp. Math. 67, Amer. Math. Soc., Providence, RI, 1987. MR 89g:14009 Zbl 0632.14016
- [Genestier and Tilouine 2005] A. Genestier and J. Tilouine, "Systèmes de Taylor–Wiles pour GSp₄", pp. 177–290 in *Formes automorphes, II: Le cas du groupe* GSp(4), Astérisque **302**, 2005. MR 2007k:11086 Zbl 1142.11036
- [Grothendieck and Murre 1971] A. Grothendieck and J. P. Murre, *The tame fundamental group of a formal neighbourhood of a divisor with normal crossings on a scheme*, Lecture Notes in Mathematics **208**, Springer, Berlin-New York, 1971. MR 47 #5000 Zbl 0216.33001
- [Haines 2005] T. J. Haines, "Introduction to Shimura varieties with bad reduction of parahoric type", pp. 583–642 in *Harmonic analysis, the trace formula, and Shimura varieties*, edited by J. Arthur et al., Clay Math. Proc. **4**, Amer. Math. Soc., Providence, RI, 2005. MR 2006m:11085 Zbl 1148.11028
- [Haines and Rapoport 2012] T. J. Haines and M. Rapoport, "Shimura varieties with $\Gamma_1(p)$ -level via Hecke algebra isomorphisms: the Drinfeld case", *Ann. Sci. Éc. Norm. Supér.* (4) **45**:5 (2012), 719–785. MR 3053008 Zbl 06155585

- [Harris and Taylor 2001] M. Harris and R. Taylor, *The geometry and cohomology of some simple Shimura varieties*, Annals of Mathematics Studies **151**, Princeton University Press, 2001. MR 2002m:11050 Zbl 1036.11027
- [Harris and Taylor 2002] M. Harris and R. Taylor, "Regular models of certain Shimura varieties", *Asian J. Math.* **6**:1 (2002), 61–94. MR 2003d:11085 Zbl 1008.11022
- [Herzig 2009] F. Herzig, "The weight in a Serre-type conjecture for tame *n*-dimensional Galois representations", *Duke Math. J.* **149**:1 (2009), 37–116. MR 2010f:11083 Zbl 1232.11065
- [Jannsen and Saito 2012] U. Jannsen and S. Saito, "Bertini theorems and Lefschetz pencils over discrete valuation rings, with applications to higher class field theory", *J. Algebraic Geom.* **21**:4 (2012), 683–705. MR 2957692 Zbl 1267.14010
- [Jantzen 1987] J. C. Jantzen, "Representations of Chevalley groups in their own characteristic", pp. 127–146 in *The Arcata Conference on Representations of Finite Groups* (Arcata, CA, 1986), edited by P. Fong, Proc. Sympos. Pure Math. 47, Amer. Math. Soc., Providence, RI, 1987. MR 89g:20076 Zbl 0652.20042
- [Johansson 2013] C. Johansson, "A remark on a conjecture of Buzzard–Gee and the cohomology of Shimura varieties", *Math. Res. Lett.* **20**:2 (2013), 279–288. MR 3151647 Zbl 1284.11093
- [Kisin 2006] M. Kisin, "Crystalline representations and *F*-crystals", pp. 459–496 in *Algebraic geometry and number theory*, edited by V. Ginzburg, Progr. Math. **253**, Birkhäuser, Boston, 2006. MR 2007j:11163 Zbl 1184.11052
- [Kottwitz 1992] R. E. Kottwitz, "Points on some Shimura varieties over finite fields", *J. Amer. Math. Soc.* **5**:2 (1992), 373–444. MR 93a:11053 Zbl 0796.14014
- [Lan and Suh 2013] K.-W. Lan and J. Suh, "Vanishing theorems for torsion automorphic sheaves on general PEL-type Shimura varieties", Adv. Math. 242 (2013), 228–286. MR 3055995 Zbl 1276.11103
- [Liu 2008] T. Liu, "On lattices in semi-stable representations: a proof of a conjecture of Breuil", *Compos. Math.* 144:1 (2008), 61–88. MR 2009c:14087 Zbl 1133.14020
- [Mazur 1977] B. Mazur, "Modular curves and the Eisenstein ideal", *Inst. Hautes Études Sci. Publ. Math.* 47 (1977), 33–186. MR 80c:14015 Zbl 0394.14008
- [Milne 1980] J. S. Milne, *Étale cohomology*, Princeton Mathematical Series **33**, Princeton University Press, 1980. MR 81j:14002 Zbl 0433.14012
- [Mokrane 1993] A. Mokrane, "La suite spectrale des poids en cohomologie de Hyodo–Kato", *Duke Math. J.* **72**:2 (1993), 301–337. MR 95a:14022 Zbl 0834.14010
- [Mokrane and Tilouine 2002] A. Mokrane and J. Tilouine, "Cohomology of Siegel varieties with *p*-adic integral coefficients and applications", pp. 1–95 in *Cohomology of Siegel varieties*, Astérisque **280**, 2002. MR 2004c:11081 Zbl 1078.11037
- [Nori 2002] M. V. Nori, "Constructible sheaves", pp. 471–491 in Algebra, arithmetic and geometry, *I–II* (Mumbai, 2000), edited by R. Parimala, Tata Inst. Fund. Res. Stud. Math. 16, Tata Inst. Fund. Res., Bombay, 2002. MR 2003m:14027 Zbl 1059.14025
- [Poonen 2004] B. Poonen, "Bertini theorems over finite fields", Ann. of Math. (2) 160:3 (2004), 1099–1127. MR 2006a:14035 Zbl 1084.14026
- [Rogawski 1990] J. D. Rogawski, Automorphic representations of unitary groups in three variables, Annals of Mathematics Studies 123, Princeton University Press, 1990. MR 91k:22037 Zbl 0724.11031
- [Scholze 2013] P. Scholze, "On torsion in the cohomology of locally symmetric varieties", preprint, 2013. arXiv 1306.2070

- [SGA 1 1971] A. Grothendieck, Revêtements étales et groupe fondamental (Séminaire de Géométrie Algébrique du Bois Marie 1960–1961), Lecture Notes in Math. 224, Springer, Berlin, 1971. MR 50 #7129 Zbl 0234.14002
- [SGA 43 1973] M. Artin, A. Grothendieck, and J. L. Verdier (editors), Théorie des topos et cohomologie étale des schémas, Tome 3: Exposés IX-XIX (Séminaire de Géométrie Algébrique du Bois Marie 1963–1964), Lecture Notes in Math. 305, Springer, Berlin, 1973. MR 50 #7132 Zbl 0245.00002
- [Shin 2013] S. W. Shin, "Supercuspidal part of the mod l cohomology of GU(n-1,1)-Shimura varieties", J. Reine Angew. Math. (online publication July 2013).
- [Suh 2008] J. Suh, "Plurigenera of general type surfaces in mixed characteristic", Compos. Math. 144:5 (2008), 1214-1226. MR 2009g:14059 Zbl 1153.14028
- [Tate and Oort 1970] J. Tate and F. Oort, "Group schemes of prime order", Ann. Sci. École Norm. Sup. (4) 3 (1970), 1-21. MR 42 #278 Zbl 0195.50801
- [Taylor 2004] R. Taylor, "Galois representations", Ann. Fac. Sci. Toulouse Math. (6) 13:1 (2004), 73-119. MR 2005a:11071 Zbl 1074.11030
- [Taylor and Yoshida 2007] R. Taylor and T. Yoshida, "Compatibility of local and global Langlands correspondences", J. Amer. Math. Soc. 20:2 (2007), 467-493. MR 2007k:11193 Zbl 1210.11118
- [Tsuji 1999] T. Tsuji, "p-adic étale cohomology and crystalline cohomology in the semi-stable reduction case", Invent. Math. 137:2 (1999), 233-411. MR 2000m:14024 Zbl 0945.14008
- [Verdier 1967] J.-L. Verdier, "A duality theorem in the etale cohomology of schemes", pp. 184–198 in Proceedings of a Conference on Local Fields (Driebergen, 1966), edited by T. A. Springer, Springer, Berlin, 1967. MR 37 #6292 Zbl 0184.24402
- [Wedhorn 2000] T. Wedhorn, "Congruence relations on some Shimura varieties", J. Reine Angew. Math. 524 (2000), 43-71. MR 2001j:11045 Zbl 1101.14033
- [Yamashita 2011] G. Yamashita, "p-adic Hodge theory for open varieties", C. R. Math. Acad. Sci. Paris 349:21-22 (2011), 1127-1130. MR 2855488 Zbl 1252.14016

Communicated by Brian Conrad Received 2013-10-24 Revised 2015-03-13 Accepted 2015-04-10 emerton@math.uchicago.edu Mathematics Department, University of Chicago, Chicago, IL 60637, United States toby.gee@imperial.ac.uk Imperial College London, London SW7 2AZ, United Kingdom





Homotopy exact sequences and orbifolds

Kentaro Mitsui

We generalize the homotopy exact sequences of étale fundamental groups for proper separable fibrations to the case where fibrations are not necessarily proper and separable. To treat the case where fibrations admit nonreduced geometric fibers, we introduce orbifolds within the framework of schemes and study their fundamental groups. As an application, we give a criterion for simpleconnectedness of elliptic surfaces over an algebraically closed field by classifying simply connected orbifold curves.

1.	Introducti	on	1089
2.	Notation and conventions		1094
3.	Orbifolds		1095
4.	Homotopy exact sequences		1100
5.	Orbifold trivializations of orbifold curves		1110
6.	Fundamental groups of elliptic fibrations		1118
Appendix A. Triangle groups and projective special linear groups		1130	
Appendix B. Comparison between orbifolds and stacks		1133	
Acknowledgments			1134
References			1134

1. Introduction

In algebraic geometry, the determination of fundamental groups of algebraic varieties is a classical problem. However, the problem is difficult, especially in the positivecharacteristic case, where few results are known except for the one-dimensional case. In this paper, we develop a method to compute étale fundamental groups of fibered regular schemes, and apply the method to study elliptic surfaces, which provides insight into the computation of étale fundamental groups of fibered varieties.

We denote the étale fundamental group of a pointed connected locally Noetherian scheme (X, \bar{x}) by $\pi_1(X, \bar{x})$. In the introduction, we omit the geometric point \bar{x} for simplicity. Let $f: X \to S$ be a proper separable morphism between connected

MSC2010: primary 14F35; secondary 14D06, 14J27, 14H30.

Keywords: orbifolds, elliptic surfaces, coverings of curves, étale fundamental groups, homotopy exact sequences, simply connected.

locally Noetherian schemes with $\mathbb{O}_S = f_*\mathbb{O}_X$. The fibration $f: X \to S$ may be characterized by the following conditions:

- (1) S is a connected locally Noetherian scheme.
- (2) $f: X \to S$ is a faithfully flat proper morphism.
- (3) The homomorphism $\mathbb{O}_S \to f_*\mathbb{O}_X$ associated to f is an isomorphism.
- (4) Any geometric fiber of f is reduced.

Choose a geometric fiber $i: X_0 \to X$ of f. In this case, Grothendieck showed in [SGA 1 1971, X.1] that the morphisms i and f induce a homotopy exact sequence

$$\pi_1(X_0) \xrightarrow{i_*} \pi_1(X) \xrightarrow{f_*} \pi_1(S) \longrightarrow 1.$$

However, Condition (4) is too strong to compute fundamental groups of fibered varieties, e.g., elliptic surfaces, which may admit nonreduced geometric fibers.

Introducing orbifolds and their fundamental groups, we generalize the above homotopy exact sequence to the case where fibrations admit nonreduced geometric fibers. Instead of considering all general connected locally Noetherian schemes, we restrict ourselves to regular integral schemes. We consider a fibration $f: X \rightarrow S$ satisfying the following conditions (see Definition 4.17 for slightly weaker conditions):

- (1') X and S are regular integral schemes.
- (2') $f: X \to S$ is a faithfully flat morphism of finite type.
- (3') \mathbb{O}_S is integrally closed in $f_*\mathbb{O}_X$.
- (4') The geometric generic fiber of f is reduced.

For example, all elliptic fibrations over curves satisfy these conditions (Section 6A). In order to give a similar exact sequence with the same homomorphism i_* in the general case (Section 4C), we have to replace $\pi_1(S)$ by its extension. To this end, we introduce orbifolds within the framework of schemes and study their fundamental groups (Section 3).

An orbifold (S, B) is defined as a locally Noetherian normal scheme S with data of ramifications B (Definition 3.6). Any orbifold curve over an algebraically closed field may be regarded as a DM stack (Theorem B.1). However, in the higher dimensional case, our orbifolds are different from DM stacks and more suitable than DM stacks for our studies (Remark B.2). We denote the fundamental group of a pointed orbifold (S, B, \bar{s}) by $\pi_1(S, B, \bar{s})$ (Definition 3.22). The local invariants associated to the nonreduced geometric fibers of f (Section 4B) endow S with an orbifold structure. Our main result is the following (Section 4C):

Theorem 1.1. Let (X, S, f) be a triple satisfying Condition (C^*) (Definition 4.17). Take the orbifold (S, B) associated to f (Definition 4.23). Choose a connected reduced geometric fiber $i : X_0 \to X$ of f over a regular point (e.g., the geometric generic fiber of f). Take a geometric point \bar{x}_0 on X_0 . Put $\bar{x} := i(\bar{x}_0)$ and $\bar{s} := f(\bar{x})$. The morphisms i and f induce canonical homomorphisms $i_* : \pi_1(X_0, \bar{x}_0) \to \pi_1(X, \bar{x})$ and $f_*^{orb} : \pi_1(X, \bar{x}) \to \pi_1(S, B, \bar{s})$, respectively (Definition 4.25). Then the sequence

$$\pi_1(X_0, \bar{x}_0) \xrightarrow{i_*} \pi_1(X, \bar{x}) \xrightarrow{f_*^{\text{orb}}} \pi_1(S, B, \bar{s}) \longrightarrow 1$$

is exact.

Next, we apply the above theorem to the case where $f: X \to S$ is the structure morphism of an elliptic surface over an algebraically closed field (Section 6). We determine the data of ramifications *B* of the orbifold (*S*, *B*) induced by *f* (Section 6B), and determine which orbifold is induced by an elliptic surface (Section 6C). As a result, we obtain a criterion for simple-connectedness of elliptic surfaces (Section 6D):

Theorem 1.2. Let k be an algebraically closed field of characteristic $p \ge 0$. Let C be a connected proper smooth k-curve. Let (X, C, f) be a relatively minimal elliptic fibration (Definition 6.1). For each closed point s on C, we set

$$m_s := \begin{cases} m & \text{if } f^{-1}(s) \text{ is of type } {}_m \mathbf{I}_n \ (n \ge 0) \ (\text{the Kodaira symbol}), \\ 1 & \text{otherwise.} \end{cases}$$

By n_s we denote the maximum integer satisfying $p \nmid n_s$ and $n_s \mid m_s$ (if p = 0, then $n_s = m_s$). Then X is simply connected if and only if all of the following conditions are satisfied:

- (1) $\chi(\mathbb{O}_X) > 0.$
- (2) $C \cong \mathbb{P}^1_k$.
- (3) $\#\{s \in C(k) \mid n_s > 1\} \le 2.$
- (4) $gcd(n_s, n_t) = 1$ for $s \neq t$.
- (5) If p > 0 and $f^{-1}(s)$ is of type ${}_m I_n$ (n > 0), then $p \nmid m$.
- (6) If p > 0 and $(f^{-1}(s))_{red}$ is isomorphic to an ordinary elliptic curve, then the $\mathbb{O}_{C,s}$ -module $(R^1 f_* \mathbb{O}_X)_s$ is torsion-free.

Furthermore, each of Conditions (1)–(6) is necessary.

We make a remark on Conditions (5) and (6) in the above theorem, which appear only in the positive-characteristic case. Under certain technical assumptions, Katsura and Ueno [1985, §§6–7] observed that an elliptic surface admits a nontrivial étale covering if one of Conditions (5) and (6) is not satisfied. Localizing elliptic

surfaces with respect to the base curve and using Galois cohomology groups, we study this phenomenon in a systematic way (Section 6B).

We may give plenty of examples of elliptic fibrations with multiple fibers by means of the algebraic analog [Lang 1986; Cossec and Dolgachev 1989, V, §4] (Section 6C) and the rigid analytic analog [Mitsui 2013] of Kodaira's logarithmic transformation [1964, §4] for complex analytic elliptic fibrations. As for topological fundamental groups in the complex analytic case, Moishezon [1977, Theorem 11, II, §2, p. 191] gave a similar criterion by means of deformations of elliptic fibrations in the category of differentiable manifolds. Although the determination of topological fundamental groups is an old problem, no references can be found for étale fundamental groups. Our proof is purely algebraic and applies in any characteristic.

Let us briefly review the studies on the fundamental group of an elliptic surface $f: X \to C$ in the complex analytic case. Take a smooth fiber $i: X_0 \to X$ of f. The morphism *i* induces a homomorphism $i_*: \pi_1(X_0) \to \pi_1(X)$. In order to study $\pi_1(X)$, Iitaka [1971, §4] determined Coker i_* . He reduced the problem to the case where X does not admit any multiple fiber by using Kodaira's logarithmic transformation and van Kampen's theorem. In a similar way, Xiao [1991] studied the case of more general compact complex analytic fibered surfaces. In another point of view, the group Coker i_* is relatively easy to deal with because it may be interpreted as the fundamental group of the orbifold curve induced by the elliptic fibration f [Ue 1986, §1; Friedman and Morgan 1994, 1.3.6]. If $\chi(\mathbb{O}_X) = 0$, then the map f between the underlying topological spaces may be regarded as a higher dimensional analog of a Seifert fibration [Seifert 1933; Thornton 1967]. Thurston [1980, §13.4] studied circle bundles over two-dimensional orbifolds in the context of the geometry of three-manifolds, which clarified the structure of Seifert fibrations: a Seifert fibration may be regarded as a circle bundle over a two-dimensional orbifold. After these studies, Ue [1986, §1] showed that $\pi_1(X)$ is isomorphic to the fundamental group of the orbifold curve induced by f whenever $\chi(\mathbb{O}_X) > 0$. Using the orbifold curve, Friedman and Morgan [1994, 2.2.1 and 2.7.2] discussed the general case in a systematic way. In the present paper, we develop this idea of using orbifolds within the framework of schemes, and give homotopy exact sequences as explained above.

As for Im i_* , no difference appears between the characteristic-zero case and the positive-characteristic case (Theorem 6.23). However, some differences appear between the algebraic case of characteristic zero and the complex analytic case (Remark 6.24). As for Coker i_* , no difference appears between the algebraic case of characteristic zero and the complex analytic case. As mentioned above, the group Coker i_* is isomorphic to the fundamental group of the orbifold curve induced by f. Thus, it follows from the fact that any compact complex analytic curve is algebraic.

However, several differences appear between the characteristic-zero case and the positive-characteristic case, as explained below.

In the algebraic case of characteristic zero and in the complex analytic case, the local structure of the orbifold curve is determined by the multiplicities of the multiple fibers. On the other hand, in the positive-characteristic case, the local structure is more complicated. This is because the completion of the local ring of the base curve at any point admits lots of finite coverings even if we fix the degree of the covering. For example, the completion admits infinitely many Artin-Schreier coverings. In particular, the resolution of multiple fibers in the positive characteristic case is much more difficult [Katsura and Ueno 1985, §§6–7; 1986, §2; Liu et al. 2004, §8.6] than that in the algebraic case of characteristic zero and in the complex analytic case [Kodaira 1963, §6]. In order to determine the local structure of the orbifold, we develop the above resolution of multiple fibers and study the minimal regular models of torsors of elliptic curves (Section 6B). In conclusion, multiple fibers of additive type [Katsura and Ueno 1986] do not affect the local structure of the orbifold, and the local uniformizations of the orbifold are given by certain finite cyclic coverings (Proposition 6.11).

In order to show Theorem 1.2, we classify simply connected orbifold curves that are locally uniformized by finite cyclic coverings (Theorem 1.3(1)). More precisely, we prove a generalized Fenchel conjecture (Theorem 1.3(2)). The original conjecture states that any finitely generated Fuchsian group admits a torsion-free subgroup of finite index, which was proved by a purely group-theoretic approach in [Fox 1952] and [Chau 1983]. The conjecture is equivalent to the following: any compact complex analytic orbifold curve minus finitely many points may be trivialized by a finite branched covering except for some trivial cases. In other words, except for some trivial cases, there exists a finite branched covering of a given compact complex analytic curve minus finitely many points with prescribed ramifications. Using the geometry of orbifold curves, we generalize this result in any characteristic (Section 5):

Theorem 1.3. Let (C, B) be a connected cyclic orbifold k-curve (Definition 5.1). Take the tame part (C, B^t) and the wild part (C, B^w) of (C, B) (Definition 3.8). Put $M := \# \operatorname{Supp} B$ and $N := \# \operatorname{Supp} B^t$ (Definition 3.6). For each $s \in \operatorname{Supp} B^t$, we put $n_s := [B_s^t : K_s]$. Then:

- (1) The orbifold (C, B) is simply connected (Definition 3.19) if and only if one of the following conditions is satisfied:

 - (a) $C \cong \mathbb{A}_{k}^{1}, M = 0, and p = 0.$ (b) $C \cong \mathbb{P}_{k}^{1}, B^{t} = B, M \leq 2, and gcd(n_{s}, n_{t}) = 1 for s \neq t.$
- (2) There exists an orbifold trivialization of (C, B) (Definition 3.10) if and only if neither of the following conditions are satisfied:

Kentaro Mitsui

(a)
$$C \cong \mathbb{P}_k^1$$
, $M = N = 1$, and $e(C, B^w) > 0$ (Definition 5.14).
(b) $C \cong \mathbb{P}_k^1$, $B^t = B$, $M = 2$, and $n_s \neq n_t$ where $\text{Supp } B = \{s, t\}$.

Let us explain our proof. The problem may be reduced to showing the existence or nonexistence of the following four types of coverings:

- (1) a covering of the projective line with at most two tame branch points;
- (2) a covering of the projective line with three tame branch points;
- (3) a covering of a curve with one tame branch point;
- (4) a covering of a curve with one wild branch point.

Case (1) is easy. In the other cases, difficulties arise when we construct a covering of a curve with prescribed ramifications. In Case (2), we use the techniques of degeneration of a covering of a curve over a mixed characteristic ring [Raynaud 1994, §6]. In Cases (3) and (4), we produce rational functions on étale coverings with prescribed zeros and poles in order to apply Kummer theory and Artin–Schreier–Witt theory.

Finally, the classification of simply connected orbifolds and the above studies on the homotopy exact sequences give the desired criterion for simple-connectedness of elliptic surfaces.

2. Notation and conventions

We denote the cardinality of a set A by #A and the degree of a finite field extension L/K by [L : K]. We denote the field of fractions of an integral domain R by Frac R and the strict Henselization of a local ring R by R^{sh} . For a ring R, an *R*-curve is a faithfully flat separated *R*-scheme of finite type and of pure relative dimension one. We denote the geometric genus of a proper curve C over a field by g(C). The multiplicity of a nonzero Weil divisor D on a locally Noetherian normal scheme X is the maximum positive integer m such that there exists a Weil divisor D' on X satisfying D = mD'. A scheme Y over a scheme X is called an étale covering space of X if the structure morphism $Y \to X$ is finite, étale, and surjective. A scheme X is said to be simply connected if X is connected and does not admit any nontrivial connected étale covering space of X.

Let X be a connected locally Noetherian scheme. Take a geometric point \bar{x} : Spec $\Omega \to X$ on X, where Ω is a separably closed field. The pair (X, \bar{x}) is called a *pointed connected locally Noetherian scheme*. We denote the étale fundamental group of (X, \bar{x}) by $\pi_1(X, \bar{x})$. We sometimes omit \bar{x} and denote $\pi_1(X, \bar{x})$ by $\pi_1(X)$ for simplicity.

3. Orbifolds

Definition 3.1. A morphism f between schemes is said to be *separable* if f is flat and the fiber of f over any point is geometrically reduced [SGA 1 1971, X.1.1]. A morphism $f: X \to Y$ between schemes is said to be *generically separable* if f maps any point x of codimension zero to a point y of codimension zero and induces a separable morphism Spec $\mathbb{O}_{X,x} \to \text{Spec } \mathbb{O}_{Y,y}$. A morphism $f: X \to Y$ between schemes is called a *quasiseparable-covering (qsc) morphism* if f is a locally quasifinite generically separable morphism. We say that a morphism fbetween locally Noetherian schemes *preserves codimensions* if f maps any point to a point of the same codimension.

Remark 3.2. A separable-covering morphism between integral schemes is conventionally defined as a finite generically separable morphism. The notion of a qsc morphism is a generalization of this notion.

We frequently use the following:

- Lemma 3.3. (1) Any locally Noetherian normal scheme is the disjoint union of locally Noetherian integral schemes [Matsumura 1989, Exercise 9.11].
- (2) The normalization of any locally Noetherian normal integral scheme X in any finite separable field extension of the function field of X is finite over X [Matsumura 1989, §33, Lemma 1].
- (3) Any separated qsc morphism X → Y between connected locally Noetherian normal schemes decomposes into an open immersion X → Z and a finite qsc morphism Z → Y where Z is a connected locally Noetherian normal scheme [EGA IV₃ 1966, 8.12.11].

Lemma 3.4 [Matsumura 1989, 9.4 and 15.1]. Let $\phi : A \to B$ be a homomorphism between Noetherian rings, and Q a prime ideal of B. Put $P := \phi^{-1}(Q)$. Then:

- (1) ht $Q \leq$ ht $P + \dim B_Q / PB_Q$.
- (2) If ϕ is flat, then the equality in (1) holds.
- (3) Any qsc morphism between locally Noetherian normal schemes preserves codimensions.

Lemma 3.5 (Zariski–Nagata purity [SGA 1 1971, X.3.1]). Let $f : X \to Y$ be a *qsc morphism between locally Noetherian schemes. Assume that X is normal and Y is regular. Then any irreducible component of the non-étale locus of f is of codimension one.*

Definition 3.6. Let *S* be a locally Noetherian normal scheme. By P(S) we denote the set of all points on *S* of codimension one. For each $s \in P(S)$ put $K_s := \operatorname{Frac} \mathbb{O}_{S,s}^{\operatorname{sh}}$. Take a separable closure $\overline{K_s}$ of K_s . Let *B* be a map that associates $s \in P(S)$ with

a finite Galois extension B_s/K_s in $\overline{K_s}$ and satisfies the following condition: let Supp $B := \{s \in P(S) \mid B_s \neq K_s\}$; then Supp *B* is locally finite. The pair (S, B)is called an *orbifold*. For a locally Noetherian normal scheme *S*, we denote the orbifold obtained by equipping *S* with the trivial orbifold structure by (S). Let *P* be a property of schemes (e.g., connected, quasicompact, regular). We say that an orbifold (S, B) is *P* if *S* has the property *P*. Let *P* be a property of finite field extensions (e.g., trivial, tame, wild, cyclic). We say that an orbifold (S, B) is *P* if B_s/K_s has the property *P* for any $s \in P(S)$.

Remark 3.7. In the above definition, we restrict ourselves to the case where the data of ramifications are given only in codimension one for our purpose of application to homotopy exact sequences (see Remark 4.26 for a generalization).

Definition 3.8. Let (S, B) be an orbifold. By B^t we denote the map that associates $s \in P(S)$ with the maximal tame field extension B_s^t/K_s in B_s . Then (S, B^t) is a cyclic orbifold. The orbifold (S, B^t) is called the *tame part* of (S, B). Assume that (S, B) is cyclic. By B^w we denote the map that associates $s \in P(S)$ with the minimum field extension B_s^w/K_s in B_s such that the equality $B_s^t B_s^w = B_s$ holds. Then (S, B^w) is a cyclic orbifold. The orbifold (S, B^t) is called the *wild part* of (S, B).

Lemma 3.9. Let $u: S' \to S$ be a qsc morphism between locally Noetherian normal schemes. Take $s' \in P(S')$. Put s := u(s'). Then $s \in P(S)$. Put $K_s := \text{Frac } \mathbb{O}^{\text{sh}}_{S,s}$ and $K'_{s'} := \text{Frac } \mathbb{O}^{\text{sh}}_{S',s'}$. Then u induces a finite field extension $K'_{s'}/K_s$.

Proof. We may assume that S and S' are affine. The first statement follows from Lemma 3.4(3). The last statement follows from Lemma 3.3(3). \Box

Definition 3.10. Let (S, B) and (S', B') be two orbifolds. We use the notation introduced in Lemma 3.9. Composing the field extensions $K'_{s'}/K_s$ and $B'_{s'}/K'_{s'}$, we obtain a field extension $B'_{s'}/K_s$. Assume that there exists a K_s -algebra homomorphism $\phi_{s'}: B_s \to B'_{s'}$ for all $s' \in P(S')$. Then u is called an *orbifold morphism* $(S', B') \to (S, B)$. If $\phi_{s'}$ is an isomorphism for all $s' \in P(S')$, then u is called an *orbifold étale morphism* $(S', B') \to (S, B)$. If u is finite, orbifold étale, and surjective, then (S', B') is called an *orbifold étale covering space* of (S, B). If (S', B') is a trivial orbifold and an orbifold étale covering space of (S, B), then uis called an *orbifold trivialization* of (S, B).

Remark 3.11. The homomorphisms $\phi_{s'}$ are *not* part of the data of an orbifold morphism. The image of $\phi_{s'}$ does not depend on the choice of $\phi_{s'}$ since B_s/K_s is Galois. The composite of any two orbifold (étale) morphisms is an orbifold (étale) morphism.

Definition 3.12. Let (S, B) be an orbifold. We use the notation introduced in Lemma 3.9. Assume that there exists a K_s -algebra homomorphism $\psi_{s'}: K'_{s'} \to B_s$

for all $s' \in P(S')$ (e.g., *u* is étale). Since $B_s/K'_{s'}$ is Galois, we may define an orbifold (S', B') by putting $B'_{s'} := B_s$ for all $s' \in P(S')$. Then *u* is an orbifold étale morphism $(S', B') \to (S, B)$. We say that *u* induces an orbifold étale morphism $(S', B') \to (S, B)$.

Lemma 3.13. Let $f : X \to S$ and $g : Y \to S$ be two qsc morphisms between locally Noetherian normal schemes. Take the normalization Z of $X \times_S Y$ and the canonical projections $f' : Z \to Y$ and $g' : Z \to X$. Then f' and g' are qsc morphisms between locally Noetherian normal schemes.

Proof. We may assume that X, Y, and S are affine. By (1) and (3) of Lemma 3.3, we have only to show the case where f and g are finite qsc morphisms between integral schemes. In this case, the lemma follows from Lemma 3.3(2).

Lemma 3.14. Take a separable closure \overline{L} of a field L. Let M and N be two finite field extensions of L in \overline{L} . Put $P := M \cap N$ and Q := MN. By \widetilde{Q} we denote the Galois closure of Q/L. Then the L-algebra $M \otimes_L N$ is L-isomorphic to a finite product of Q-subfields of \widetilde{Q} . If M/L is Galois then $M \otimes_L N \cong Q^{[P:L]}$ over L, where the right-hand side is the product of [P:L] copies of Q. If M/L and N/Lare Galois, then $\widetilde{Q} = Q$.

Proof. This follows from the *L*-algebra isomorphism $M \otimes_L N \cong P \otimes_L Q$. \Box

By \mathscr{C} (resp. $\mathscr{C}_{\acute{e}t}$) we denote the category consisting of locally Noetherian normal schemes and qsc morphisms (resp. étale morphisms). By \mathscr{C}_{orb} (resp. $\mathscr{C}_{orb,\acute{e}t}$) we denote the category consisting of orbifolds and orbifold morphisms (resp. orbifold étale morphisms). We define a faithful functor $F_{orb} : \mathscr{C} \to \mathscr{C}_{orb}$ by $S \mapsto (S)$. In the same way, we define a faithful functor $F_{orb,\acute{e}t} : \mathscr{C}_{\acute{e}t} \to \mathscr{C}_{orb,\acute{e}t}$. By $G : \mathscr{C}_{\acute{e}t} \to \mathscr{C}$ and $G_{orb} : \mathscr{C}_{orb,\acute{e}t} \to \mathscr{C}_{orb}$ we denote the canonical faithful functors. Then $F_{orb} \circ G$ is naturally isomorphic to $G_{orb} \circ F_{orb,\acute{e}t}$.

Proposition 3.15. The categories \mathcal{C} , $\mathcal{C}_{\text{ét}}$, \mathcal{C}_{orb} , and $\mathcal{C}_{\text{orb},\text{ét}}$ admit any finite fiber product. In any case, the (underlying) scheme of any finite fiber product is isomorphic to the normalization of the fiber product of the (underlying) schemes, and any base change of any (orbifold) étale morphism is an (orbifold) étale morphism. The functors F_{orb} , $F_{\text{orb},\text{ét}}$, G, and G_{orb} preserve any finite fiber product.

Proof. Let us show the first statement. Lemma 3.13 shows the cases of \mathscr{C} and $\mathscr{C}_{\text{ét}}$. Let (S_1, B_1) and (S_2, B_2) be two orbifolds over an orbifold (S_0, B_0) . We define an orbifold (S_3, B_3) in the following way. Take the normalization S_3 of $S_1 \times_{S_0} S_2$ and the canonical projection $p_i : S_3 \to S_i$ for i = 0, 1, and 2. Take $s \in P(S_3)$. Put $K_i := \operatorname{Frac} \mathbb{O}_{S_i, p_i(s)}^{\text{sh}}$ and $K_3 := \operatorname{Frac} \mathbb{O}_{S_3, s}^{\text{sh}}$. The morphism p_i induces a field extension K_3/K_i . Put $L_i := B_{i, p_i(s)}$. Take a separable closure $\overline{K_3}$ of K_3 . By L'_i/K_3 we denote the unique Galois extension in $\overline{K_3}$ that is K_3 -isomorphic to the lifting of the Galois extension L_i/K_i via K_3/K_i . We define $B_{3,s}/K_3$ as the Galois extension $L'_1L'_2/K_3$. We apply the same procedure to all $s \in P(S_3)$. Then we obtain an orbifold (S_3, B_3) and an orbifold morphism $p_i : (S_3, B_3) \to (S_i, B_i)$ for i = 0, 1, and 2. By construction, the orbifold (S_3, B_3) is the fiber product of (S_1, B_1) and (S_2, B_2) over (S_0, B_0) in \mathscr{C}_{orb} . Thus, the category \mathscr{C}_{orb} admits any finite fiber product. Assume that (S_1, B_1) is orbifold étale over (S_0, B_0) . By definition, we may regard L_0, L_1 , and L_2 as field extensions of K_0 satisfying $L_1 = L_0 \subset L_2$. Since $K_3 \subset K_1 \otimes_{K_0} K_2 \subset L_1 \otimes_{K_0} L_2 \cong L_2^{[L_1:K_0]}$ (Lemma 3.14), we may regard K_3 as a K_2 -subfield of L_2 . Then $L_2 \cong B_{3,s}$ over K_2 . Thus, the morphism $p_2 : (S_3, B_3) \to (S_2, B_2)$ is orbifold étale. In particular, the category $\mathscr{C}_{orb,\text{ét}}$ admits any finite fiber product. The other statements follow from the construction. \Box

Definition 3.16. Let (S, B_1) and (S, B_2) be two orbifolds. For i = 1 and 2, the identity on S is an orbifold morphism $(S, B_i) \rightarrow (S)$. We define the *composite orbifold* (S, B_1B_2) of (S, B_1) and (S, B_2) as the fiber product of (S, B_1) and (S, B_2) over (S) in \mathcal{C}_{orb} .

Remark 3.17. By the proof of Proposition 3.15, the field $(B_1B_2)_s$ is equal to the composite field of $B_{1,s}$ and $B_{2,s}$ for any $s \in P(S)$, where we regard the extensions $B_{1,s}$ and $B_{2,s}$ of K_s as subfields of a fixed separable closure $\overline{K_s}$.

Proposition 3.18. For i = 1 and 2, let $u_i : (S_i) \to (S, B_i)$ be an orbifold trivialization. Take the normalization S_3 of $S_1 \times_S S_2$ and the canonical projection $p_i : S_3 \to S_i$ for i = 1 and 2, which is an orbifold morphism $(S_3) \to (S_i)$. Then the orbifold morphism $(S_3) \to (S, B_1B_2)$ induced by $u_1 \circ p_1$ and $u_2 \circ p_2$ is an orbifold trivialization.

Proof. The proposition follows from Lemma 3.14 and Remark 3.17.

Definition 3.19. An orbifold (S, B) is said to be *simply connected* if (S, B) is connected and does not admit any nontrivial connected orbifold étale covering space. Let (S, B) be a connected orbifold. Take a geometric point \overline{s} : Spec $\Omega \to S$ on S, where Ω is a separably closed field. Assume that the image of \overline{s} is a regular point on S and not contained in the closure of Supp B (e.g., the image is the generic point of S). The triple (S, B, \overline{s}) is called a *pointed connected orbifold*. Let $\mathcal{G} = (S, B, \overline{s} : \operatorname{Spec} \Omega \to S)$ and $\mathcal{G}' = (S', B', \overline{s}' : \operatorname{Spec} \Omega' \to S')$ be two pointed connected orbifolds. A *pointed orbifold* (*étale*) morphism $\mathcal{G}' \to \mathcal{G}$ is a pair of an orbifold (*étale*) morphism $u : (S', B') \to (S, B)$ and a morphism $\Phi : \operatorname{Spec} \Omega' \to \operatorname{Spec} \Omega$ between schemes such that the diagram

Spec
$$\Omega' \xrightarrow{\overline{s}'} S'$$

 $\downarrow \Phi \qquad \qquad \downarrow u$
Spec $\Omega \xrightarrow{\overline{s}} S$

is commutative.

Remark 3.20. By Zariski–Nagata purity (Lemma 3.5), any orbifold étale morphism to (S, B) is étale over the image of \overline{s} .

Let (S, B, \bar{s}) be a pointed connected orbifold. By $\mathscr{C}_{(S,B)}$ we denote the category of finite orbifold étale (S, B)-orbifolds and orbifold étale (S, B)-morphisms. We define a functor $F_{\bar{s}}$ from $\mathscr{C}_{(S,B)}$ to the category of finite sets by sending an object $(S', B') \rightarrow (S, B)$ to the underlying set of $S' \times_S \bar{s}$. We refer to [SGA 1 1971, V.5] for the definition of a Galois category and a fiber functor (a fundamental functor).

Theorem 3.21. The category $\mathscr{C}_{(S,B)}$ is a Galois category with the fiber functor $F_{\overline{s}}$.

Proof. We show the theorem in the same way as in the case of the category of finite étale schemes over a pointed connected locally Noetherian scheme (see [SGA 1 1971, V.7]). We have only to verify that the pair ($\mathscr{C}_{(S,B)}$, $F_{\bar{s}}$) satisfies Axioms (G1)–(G6) in [loc. cit., V.4]:

- (G1) The orbifold (S, B) is the final object in $\mathscr{C}_{(S,B)}$. The category $\mathscr{C}_{(S,B)}$ admits any finite fiber product (Proposition 3.15).
- (G2) The empty set equipped with the trivial orbifold structure is the initial object in $\mathscr{C}_{(S,B)}$. Take an object $u : (S', B') \to (S, B)$ in $\mathscr{C}_{(S,B)}$. Assume that a finite group *G* acts on an orbifold (S', B') over (S, B) in $\mathscr{C}_{(S,B)}$. By S''we denote the spectrum of the \mathbb{O}_S -algebra of *G*-invariant sections of $\mathbb{O}_{S'}$. Then S'' is the quotient of S' by *G* in the category of *S*-schemes [loc. cit., V.1.8]. By $v : S' \to S''$ we denote the morphism induced by the canonical inclusion homomorphism $\mathbb{O}_{S''} \to \mathbb{O}_{S'}$. The morphism *u* factors as $u = w \circ v$. Since *u* is finite, the morphism *v* is finite. Since *S* is locally Noetherian and *u* is finite, the morphism *w* is finite. Take a point *s'* on *S'* of codimension one. By $I(s') \subset G$ we denote the inertia group of *s'*. We use the notation $B'_{s'}/K'_{s'}$ introduced in Definition 3.10. Put s'' := v(s'), $B''_{s''} := B'_{s'}$, and $K'''_{s''} := \operatorname{Frac} \mathbb{O}_{S'',s''}^{\operatorname{sh}}$. Then the image under the homomorphism $K''_{s''} \to K'_{s''}$ induced by *v* is equal to $(K'_{s'})^{I(s')}$. Furthermore, the extension $B''_{s''}/K''_{s''}$ is finite and Galois. By construction, the pair (S'', B'') is an orbifold that is a quotient of (S', B') by *G* in $\mathscr{C}_{(S,B)}$.
- (G3) Lemma 3.3(1) implies that any morphism $u: (S', B') \to (S, B)$ in $\mathscr{C}_{(S,B)}$ factors as $w \circ v: (S', B') \to (S'', B'') \to (S, B)$ where v is a strict epimorphism, w is a monomorphism, and (S'', B'') is a direct summand of (S, B).
- (G4) By definition, the functor $F_{\bar{s}}$ is left-exact.
- (G5) By definition, the functor $F_{\bar{s}}$ preserves any finite direct sum and preserves the quotient by any action of any finite group. Take a strict epimorphism u in $\mathscr{C}_{(S,B)}$. We have to show that $F_{\bar{s}}(u)$ is surjective. By s we denote the image of \bar{s} on S. By base change, we may replace S by Spec $\mathbb{O}_{S,s}$. Then uis étale (Remark 3.20). Thus, the surjectivity follows from [loc. cit., V.3.5].

(G6) Take a morphism u in $\mathscr{C}_{(S,B)}$. Assume that $F_{\bar{s}}(u)$ is an isomorphism. We have to show that u is an isomorphism. By s we denote the image of \bar{s} on S. By base change, we may replace S by Spec $\mathbb{O}_{S,s}$. Then u is étale (Remark 3.20). Thus, it follows from [loc. cit., V.3.7].

Therefore, the pair $(\mathscr{C}_{(S,B)}, F_{\overline{s}})$ satisfies Axioms (G1)–(G6) in [loc. cit., V.4], which proves the theorem.

Definition 3.22. Let (S, B, \bar{s}) be a pointed connected orbifold. The functor $F_{\bar{s}}$ is pro-representable by a profinite group (see [loc. cit., V.5]). This group is called the *fundamental group* of (S, B, \bar{s}) and denoted by $\pi_1(S, B, \bar{s})$. We sometimes omit \bar{s} and denote $\pi_1(S, B, \bar{s})$ by $\pi_1(S, B)$ for simplicity.

Let $\mathcal{G} = (S, B, \bar{s})$ and $\mathcal{G}' = (S', B', \bar{s}')$ be two pointed connected orbifolds. Any pointed orbifold (étale) morphism $u : \mathcal{G}' \to \mathcal{G}$ induces an (injective) homomorphism $u_* : \pi_1(\mathcal{G}') \to \pi_1(\mathcal{G})$ (see [loc. cit., V.6]). Since any connected finite étale *S*-scheme induces a connected finite orbifold étale (S, B)-orbifold, we obtain a canonical surjective homomorphism $\phi_{\mathcal{G}} : \pi_1(\mathcal{G}) \to \pi_1(S, \bar{s})$. If the regular locus of *S* is open (e.g., *S* is excellent), then the singular locus of *S* is a closed subset of codimension at least two. Thus, Zariski–Nagata purity (Lemma 3.5) shows the following:

Proposition 3.23. Let $\mathcal{G} = (S, B, \bar{s})$ be a pointed connected orbifold. By S_0 we denote the regular locus of S. Assume that S_0 is an open subset of S. By $u : \mathcal{G}_0 \to \mathcal{G}$ we denote the pointed orbifold étale morphism induced by the inclusion morphism $S_0 \to S$. Then the homomorphism $u_* : \pi_1(\mathcal{G}_0) \to \pi_1(\mathcal{G})$ induced by uis an isomorphism. If \mathcal{G} is trivial and regular, then $\phi_{\mathcal{F}}$ is an isomorphism and, in particular, $\pi_1(\mathcal{G}) \cong \pi_1(S)$.

Example 3.24. The homomorphism $\phi_{(S)} : \pi_1((S)) \to \pi_1(S)$ is not injective in general, where (S) is the trivial orbifold associated to S and we omit the geometric points. Let k be an algebraically closed field of characteristic zero, n an integer greater than one, and ζ a primitive n-th root of unity. Put $S' := \mathbb{A}_k^2$. Take the coordinate functions (x, y) of S'. We define an automorphism σ on S' by $(x, y) \mapsto (\zeta x, \zeta y)$. Take the quotient $u : S' \to S$ of σ . The scheme S is normal but not regular. The morphism u ramifies only at the origin o of S'. Since $\pi_1(S' \setminus \{o\}) \cong \pi_1(S') \cong 1$ by Zariski–Nagata purity (Lemma 3.5), we obtain the isomorphisms $\pi_1((S)) \cong \mathbb{Z}/n\mathbb{Z}$ and $\pi_1(S) \cong 1$. Thus, the homomorphism $\phi_{(S)}$ is not injective.

4. Homotopy exact sequences

4A. Coverings of fibrations.

Lemma 4.1. Let $f : X \to S$ be a quasicompact morphism between locally Noetherian schemes. Assume that X is reduced. Then the following are equivalent:

- (1) S is reduced and f is dominant.
- (2) The homomorphism $\mathbb{O}_S \to f_*\mathbb{O}_X$ associated to f is injective.

Assume that (1) and (2) hold and that X is normal and integral. Then S is integral and the following statements are equivalent:

- (3) *S* is normal and the function field of *S* is algebraically closed in that of *X*.
- (4) \mathbb{O}_S is integrally closed in $f_*\mathbb{O}_X$.
- If f is generically separable, then (3) and (4) are equivalent to the following:

(5) *S* is normal and the generic fiber of f is geometrically integral.

Proof. We may assume that *S* is affine. Put $R := \Gamma(S, \mathbb{O}_S)$ and $W := \Gamma(X, \mathbb{O}_X)$. Since *W* is reduced and the kernel of the homomorphism $R \to W$ associated to *f* is the defining ideal of the closure of f(X), the first equivalence holds. Let us show the other statements. Since *X* is irreducible and *f* is dominant, the scheme *S* is irreducible, which implies that *S* is integral. By *K* and *L* we denote the function fields of *S* and *X*, respectively. By *K'* and *R'* we denote the algebraic closure of *K* in *L* and the integral closure of *R* in *K'*, respectively. Then R' = R if and only if *R* is normal and K' = K. Since *X* is normal and integral, the ring *W* is normal and integral [Liu 2002, 4.1.5], which implies that *R'* is the integral closure of *R* in *W*. Thus, the second equivalence holds. The last statement follows from [EGA IV₂ 1965, 4.6.3].

Lemma 4.2. Let $u : X \to Y$ and $v : Y \to Z$ be morphisms between integral schemes. Assume that $v \circ u$ is dominant and generically separable and that v is integral. Then u is dominant and generically separable and v is surjective and generically separable.

Proof. Since $v \circ u$ is dominant and v is closed, the morphism v is surjective. Since v is integral and dominant, the preimage of the generic point of Z under v consists of the generic point of Y [Matsumura 1989, 9.3 (ii)]. Furthermore, since $v \circ u$ is dominant and generically separable, the morphism u is dominant and the morphism v is generically separable. Note the following: for any field extensions L/K, M/L, and N/L, the ring $M \otimes_L N$ may be regarded as a subring of $M \otimes_K N$ since $M \otimes_K N \cong M \otimes_L N \otimes_L (L \otimes_K L)$ and $L \cong L \otimes_K K \subset L \otimes_K L$. Thus, the morphism u is generically separable.

Definition 4.3. Condition (D) on morphisms $f : X \to S, u' : X' \to X, u : S' \to S$, and $f' : X' \to S'$ between locally Noetherian schemes consists of the following conditions:

- (1) f is quasicompact, surjective, and generically separable.
- (2) u' is finite, surjective, and generically separable.

- (3) u is integral.
- (4) The homomorphism O_S → f_{*}O_X associated to f is injective and O_S is integrally closed in f_{*}O_X.
- (5) The homomorphism $\mathbb{O}_{S'} \to f'_* \mathbb{O}_{X'}$ associated to f' is injective and $\mathbb{O}_{S'}$ is integrally closed in $f'_* \mathbb{O}_{X'}$.
- (6) The diagram



is commutative.

Remark 4.4. Conditions (1)–(3) imply that f' is quasicompact. Conditions (1)–(6) imply that u is given by the integral closure of \mathbb{O}_S in $(f \circ u')_* \mathbb{O}_{X'}$.

Proposition 4.5. Let $f : X \to S$, $f' : X' \to S'$, $u : S' \to S$, and $u' : X' \to X$ be morphisms between locally Noetherian schemes satisfying Condition (D). Suppose that X is normal and that X' is connected and normal. Then:

- (1) X, X', S, and S' are normal and integral.
- (2) *u* is finite, surjective, and generically separable.
- (3) f' is quasicompact, surjective, and generically separable.

Proof. Since X' is connected and u' is surjective, the scheme X is connected. Lemma 3.3(1) shows that X and X' are integral. Thus, Lemma 4.1 shows that S and S' are integral and normal. Therefore, Statement (1) holds. Since $f \circ u'$ is surjective and generically separable and u is integral, Lemma 4.2 shows that f' is dominant and generically separable and u is surjective and generically separable. Thus, Lemma 3.3(2) shows that u is finite. Therefore, Statement (2) holds. Let us show that f' is surjective. We may assume that the finite covering X'/X is Galois after replacing X' and S' by finite coverings. The Galois group G of X'/X faithfully acts on the finite covering S'/S such that f' is G-equivariant. Since f'(X') is stable under any element of G and the equalities u(f'(X')) = f(u'(X')) = S hold, the morphism f' is surjective. Thus, Statement (3) holds.

Proposition 4.6. We use the same notation and assumption as in Proposition 4.5. Let $v: T \rightarrow S$ be one of the following morphisms between schemes:

- (a) a smooth morphism;
- (b) the localization at a point;
- (c) the strict Henselization (or the Henselization) of the spectrum of a local ring;
- (d) the completion of the spectrum of an excellent local ring at the closed point.

By $f_T: X_T \to T$, $f'_T: X'_T \to S'_T$, $u_T: S'_T \to T$, and $u'_T: X'_T \to X_T$ we denote the base changes of the S-morphisms f, f', u, and u' via v, respectively. Take connected components Z and Z' of T and S'_T , respectively. Then:

- (1) $Z, f_T^{-1}(Z), Z', and (f_T')^{-1}(Z')$ are locally Noetherian, normal, and integral.
- (2) f_T , f'_T , u_T , and u'_T satisfy Condition (D).
- (3) If v is surjective, then u' is étale if and only if u'_T is étale.

Proof. By U we denote any of T, X_T , S'_T , and X'_T . Then U is locally Noetherian. Since any fiber of v is geometrically regular, the scheme U is normal [Matsumura 1989, 23.9]. The schemes Z and Z' are integral (Lemma 3.3(1)). We may assume that Z = T and $f_T^{-1}(Z) = X_T$. Since v is a dominant flat morphism between integral schemes, Lemma 4.1 implies that X_T is integral. In the same way, we may show that $(f'_T)^{-1}(Z')$ is integral. Thus, Statement (1) holds. Let us show Statement (2). We have only to show that \mathbb{O}_T and $\mathbb{O}_{S'_T}$ are integrally closed in $f_T * \mathbb{O}_{X_T}$ and $f'_T * \mathbb{O}_{X'_T}$, respectively. Thus, Statement (2) follows from Lemma 4.1. Statement (3) follows from faithfully flat descent for étale morphisms.

Lemma 4.7. Let $u : X \to Y$ and $v : Y \to Z$ be morphisms between locally Noetherian schemes. Put $w := v \circ u$. Assume that Y and Z are normal. Suppose that u is dominant, v is affine, and w is finite, étale, and dominant. Then u and v are finite, étale, and surjective.

Proof. Since Z is normal and w is étale, the scheme X is normal. We may assume that X, Y, and Z are integral (Lemma 3.3(1)). Then u and v are finite surjective morphisms between locally Noetherian normal integral schemes. We have only to show that u and v are étale over any point z on Z. Since w is finite, étale, and surjective, there exists an étale morphism $t : U \to Z$ such that $z \in t(U)$ and the restriction of the base change $X_U \to U$ of w via t to any connected component of X_U is an isomorphism [Bosch et al. 1990, 2.3.8]. Thus, by faithfully flat descent for étale morphisms, we may assume that w is an isomorphism. Then u and v induce isomorphisms between the function fields of X, Y, and Z. Since X, Y, and Z are normal and integral and u and v are finite, the morphisms u and v are isomorphisms, which implies that u and v are étale.

Proposition 4.8. Let $f: X \to S$ and $u: S' \to S$ be morphisms between locally Noetherian normal integral schemes. Assume that f is quasicompact, surjective, and generically separable, u is finite, surjective, and generically separable, and \mathbb{O}_S is integrally closed in $f_*\mathbb{O}_X$. Then there exist a locally Noetherian normal integral scheme X' and morphisms $f': X' \to S'$ and $u': X' \to X$ satisfying the following:

(a) f, f', u, and u' satisfy Condition (D).

Kentaro Mitsui

(b) For any normal integral scheme Y and any dominant morphisms h: Y → S' and ξ : Y → X satisfying f ∘ ξ = u' ∘ h, there exists a unique morphism ξ' : Y → X' such that u' ∘ ξ' = ξ and f' ∘ ξ' = h.

Furthermore, the following statements hold:

- (1) If ξ in (b) is finite and étale, then u' and ξ' are finite, étale, and surjective.
- (2) If u is étale, then u' is étale.
- (3) If f is separable and of finite type, then the converse of (2) holds.

Proof. By K, K', and L we denote the function fields of S, S', and X, respectively. Put $L' := L \otimes_K K'$. Since K is algebraically closed in L (Lemma 4.1) and the extension K'/K is finite and separable, the ring L' is a field. Furthermore, the field K' is algebraically closed in L' and the extension L'/L is finite and separable. Take the normalization $u': X' \to X$ of X in L'. Since u' is finite (Lemma 3.3(2)), the scheme X' is locally Noetherian, normal, and integral. Take the unique morphism $f': X' \to S'$ such that $f \circ u' = u \circ f'$. Then $\mathbb{O}_{S'}$ is integrally closed in $f'_*\mathbb{O}_{X'}$ (Lemma 4.1). Thus, Condition (a) is satisfied. By construction, Condition (b) is satisfied. Statement (1) follows from Lemma 4.7. By $v: Z \to X$ and $w: Z \to S'$ we denote the base change of u via f and the base change of f via u, respectively. Then v is finite, and Z is an integral scheme with function field L'. Let us show Statement (2). Assume that u is étale. Then v is étale and Z is normal. Thus, the scheme Z is X-isomorphic to X', which implies that u' is étale. Let us show Statement (3). Assume that f is separable and of finite type and that u' is étale. Replacing X by the smooth locus of f, we may assume that f is smooth ([Bosch et al. 1990, 2.2.16] and Lemma 4.1). Then Z is normal. Thus, the scheme Z is X-isomorphic to X', which implies that v is étale. Therefore, Statement (3) follows from faithfully flat descent for étale morphisms.

4B. Base spaces of local étale coverings.

Definition 4.9. Let *R* be a strictly Henselian Noetherian normal local ring with field of fractions *K*. Take a separable closure \overline{K} of *K*. Put S := Spec *R*. Let $f : X \to S$ be a surjective generically separable morphism between connected Noetherian normal schemes. Assume that \mathbb{O}_S is integrally closed in $f_*\mathbb{O}_X$. We define the *maximal base field* \widetilde{K} (of étale coverings of the total space) of f in the following way. Let $\xi : Y \to X$ be a connected étale covering space. Then Y is normal. Take the normalization

$$Y \xrightarrow{h} S' \xrightarrow{u} S$$

of *S* in the composite $f \circ \xi : Y \to X \to S$. By K_{ξ} we denote the function field of *S'*. Then *u* induces a finite separable field extension K_{ξ}/K (Proposition 4.5).
We define \widetilde{K} as the composite field of all *K*-embeddings of the finite separable extensions K_{ξ}/K in \overline{K} for all connected étale covering spaces $\xi : Y \to X$ of *X*.

Remark 4.10. By definition, the field extension \tilde{K}/K is algebraic and Galois.

We use the notation introduced in Definition 4.9.

Proposition 4.11. Let $g: X' \to X$ be a proper birational morphism between regular integral schemes. Then the maximal base field of $f \circ g$ in \overline{K} is equal to \widetilde{K} .

Proof. By Zariski–Nagata purity (Lemma 3.5; see also [SGA 1 1971, X.3.3]), the base change of finite étale X-schemes via g induces an equivalence of categories between the category of finite étale X-schemes and the category of finite étale X'-schemes, which proves the proposition.

Lemma 4.12. Let L/K be a finite field extension in \overline{K} . Then $L \subset \widetilde{K}$ if and only if there exists a connected étale covering space of X that induces the extension L/K.

Proof. The "if" part follows from the definition of \tilde{K} . Since any finite fiber product of étale covering spaces of X over X is an étale covering space of X, the "only if" part follows from Proposition 4.8(1).

By k we denote the residue field of R. By $\{Z_i\}_{i \in I}$ we denote the set of all irreducible components of the special fiber X_k of f with the reduced structures. Take the integral closure k_i of k in $\Gamma(Z_i, \mathbb{O}_{Z_i})$.

Lemma 4.13. The ring k_i is a field. If f is of finite type, then the field extension k_i/k is finite and purely inseparable for any $i \in I$.

Proof. Since k_i is an integral extension of the field k, the integral domain k_i is a field. Assume that f is of finite type. Since Z_i is finite type over k, the function field K_i of Z_i is finitely generated over k. Since $k_i \subset K_i$ and k is separably closed, the last statement holds.

Suppose that *R* is a discrete valuation ring. Then *f* is flat. The closed subscheme X_k is a divisor on *X* and, for any $i \in I$, the closed subscheme Z_i is a prime divisor on *X* (Lemma 3.4(2)). We may write $X_k = \sum_{i \in I} m_i Z_i$.

Lemma 4.14. Suppose that *R* is a discrete valuation ring. Assume that *f* is of finite type. Put $n_i := [k_i : k]$, which is finite by Lemma 4.13. Let $\xi : Y \to X$ be a connected étale covering space. Take the normalization

$$Y \xrightarrow{h} S' \xrightarrow{u} S$$

of S in the composite $f \circ \xi : Y \to X \to S$. Then the degree of u divides $gcd(m_i)_{i \in I} \cdot gcd(n_i)_{i \in I}$.

Proof. By *m* and *n* we denote the ramification index of S'/S and the degree of the residue field extension of S'/S, respectively. Then $m \mid m_i$ and $n \mid n_i$ for any $i \in I$ since ξ is étale and *k* is separably closed. Since *R* is a Henselian discrete valuation ring, the degree of *u* is equal to *mn*, which concludes the proof.

Proposition 4.15. We use the notation introduced in Definition 4.9. By k we denote the residue field of R. Suppose that R is a discrete valuation ring. Assume that f is of finite type. Then:

- (1) The field extension \tilde{K}/K is finite and Galois.
- (2) If f is separable, then $\tilde{K} = K$.
- (3) If k is perfect, then $[\tilde{K} : K]$ divides the multiplicity of the special fiber of f (Section 2).
- (4) If f is proper and a finite field extension K'/K in \overline{K} satisfies $X(K') \neq \emptyset$, then $\widetilde{K} \subset K'$.

Proof. Statement (1) (resp. (2) and (3)) follows from Lemma 4.14 since \tilde{K}/K is Galois (Remark 4.10) (resp. $m_i = 1$ and $k_i = k$ for any $i \in I$, and $k_i = k$ for any $i \in I$ (Lemma 4.13)). Let us show Statement (4). By Proposition 4.8(1), we may assume that $\tilde{K} \cap K' = K$. By S' and \tilde{S} we denote the normalizations of S in K' and \tilde{K} , respectively. Take the scheme X' (resp. \tilde{X}) and the morphism $u' : X' \to X$ (resp. $\tilde{u} : \tilde{X} \to X$ and $\tilde{f} : \tilde{X} \to \tilde{S}$) given by Proposition 4.8. Then $\tilde{u} : \tilde{X} \to X$ is an étale covering space. Since the base change of \tilde{u} via u' induces $\tilde{K}K'/K'$, we have only to show that $\tilde{K} = K$ whenever $X(K) \neq \emptyset$. Assume that $X(K) \neq \emptyset$. Then f admits a section by the valuative criterion for properness. Since the pullback of any section of f via \tilde{u} induces a section of \tilde{f} and S is strictly Henselian, the degree of \tilde{u} is equal to 1, which concludes the proof.

Example 4.16. Let us give an example of a morphism $f: X \to S$ of finite type with $[\tilde{K}:K] = \infty$ when dim S > 1. Assume that k is algebraically closed, the characteristic of k is not equal to 3, and $R = k[[x, y, z]]/(x^3 + y^3 + z^3)$. By s we denote the closed point of S. Put $S_0 := S \setminus \{s\}$. Then $\pi_1((S)) \cong \pi_1(S_0)$ (Proposition 3.23). Take the blowing-up $f: X \to S$ of S at s. Then X is regular. Put $E := f^{-1}(s)$. The reduction E_{red} of E is k-isomorphic to an elliptic curve over k and the multiplicity of E is equal to 3. The morphism f is not flat at any point on E (Lemma 3.4(2)). The inclusion morphisms $S_0 \to X$, $E_{\text{red}} \to E$, and $E \to X$ induce a surjective homomorphism $\pi_1(S_0) \to \pi_1(X)$, an injective homomorphism $\pi_1(E_{\text{red}}) \to \pi_1(E)$, and an isomorphism $\pi_1(E) \cong \pi_1(X)$, respectively [SGA $4\frac{1}{2}$ 1977, IV.2.2]. Since $\pi_1(E_{\text{red}})$ is not finite, the extension \tilde{K}/K is infinite.

4C. *Homotopy exact sequences.* In this subsection, we give homotopy exact sequences for fibrations satisfying the following conditions:

Definition 4.17. Condition (C) on a triple (X, S, f) consists of the following conditions:

- (1) X and S are locally Noetherian normal integral schemes.
- (2) $f: X \to S$ is a surjective morphism of finite type.
- (3) \mathbb{O}_S is integrally closed in $f_*\mathbb{O}_X$ (Conditions (1) and (2) imply that the homomorphism $\mathbb{O}_S \to f_*\mathbb{O}_X$ associated to f is injective (Lemma 4.1)).
- (4) The geometric generic fiber of f is reduced.

Condition (C^*) on a triple (X, S, f) is Condition (C) and the following conditions:

- (5) X is regular.
- (6) f is flat in codimension one.

Remark 4.18. In the case where f is proper, Conditions (2) and (3) are equivalent to the following conditions:

(2') $f: X \to S$ is proper.

(3') The homomorphism $\mathbb{O}_S \to f_*\mathbb{O}_X$ associated to f is an isomorphism.

Remark 4.19. In our studies on homotopy exact sequences, Condition (C^{*}) is necessary. This condition is used to describe the effect of the nonreduced geometric fibers of f on étale covering spaces of X in terms of an orbifold (S, B). Conditions (1)–(3) are used to produce a finite covering space of S by taking the normalization of S in the composite of a finite covering map of X and f. In the case where f is proper, this normalization may be given by the Stein factorization of the composite. Condition (5) is used to apply Zariski–Nagata purity (Lemma 3.5) to a finite covering space of X. In particular, the condition that the finite covering map is étale may be checked in codimension one. Condition (4) enables Condition (6) to encode this condition as the data of ramifications B of an orbifold (S, B). See Theorem 4.22 and Remark 4.26 for Condition (C).

Example 4.20. Take $k, n, u : S' \to S$, and σ as in Example 3.24. Let E be an elliptic curve over k. Put $X' := E \times_k S'$. By $f' : X' \to S'$ we denote the second projection. Choose a primitive *n*-torsion point P on E. We define an action τ on X' as the product of the translation by the addition of P on E and the action of σ on S'. We take the quotient $u' : X' \to X$ of τ . Since f' is equivariant with respect to τ and σ , we obtain a morphism $f : X \to S$. The triple (X, S, f) satisfies Condition (C). The morphisms f, f', u, and u' satisfy Condition (D) (Definition 4.3). Since X is regular and S is not regular, the morphism f is not flat [Matsumura 1989, 23.7 (i)]. However, the morphism f is flat in codimension one since f is flat over the regular locus of S [Matsumura 1989, 23.1]. In particular, the triple (X, S, f) satisfies Condition (C^{*}).

Example 4.21. Take $k, u: S' \to S$, and o as in Example 3.24. Take the blowing-ups $f: X \to S$ and $f': X' \to S'$ of S and S' at u(o) and o, respectively. Then X and X' are regular. The universal property of blowing-up shows that there exists a unique morphism $u': X' \to X$ such that $f \circ u' = u \circ f'$. The morphism u' ramifies along the exceptional divisor of f'. The morphisms f, f', u, and u' satisfy Condition (D) (Definition 4.3). The triple (X, S, f) satisfies Condition (C). However, the morphism f is not flat in codimension one (Lemma 3.4(2)). In particular, the triple (X, S, f) does not satisfy Condition (C^{*}).

We first generalize Grothendieck's homotopy exact sequence to the case where fibrations are not necessarily proper:

Theorem 4.22. Let (X, S, f) be a triple satisfying Condition (C) (Definition 4.17). Assume that f is separable. Choose a connected geometric fiber $i : X_0 \to X$ of f (e.g., the geometric generic fiber of f). Take a geometric point \bar{x}_0 on X_0 . Put $\bar{x} := i(\bar{x}_0)$ and $\bar{s} := f(\bar{x})$. The morphisms i and f induce canonical homomorphisms $i_* : \pi_1(X_0, \bar{x}_0) \to \pi_1(X, \bar{x})$ and $f_* : \pi_1(X, \bar{x}) \to \pi_1(S, \bar{s})$, respectively. Then the sequence

$$\pi_1(X_0, \bar{x}_0) \xrightarrow{i_*} \pi_1(X, \bar{x}) \xrightarrow{f_*} \pi_1(S, \bar{s}) \longrightarrow 1$$

is exact.

Proof. We have only to show the exactness at $\pi_1(X)$. Since X_0 is a geometric fiber of f, the relation $\operatorname{Im} i_* \subset \operatorname{Ker} f_*$ holds. Let us show that $\operatorname{Ker} f_* \subset \operatorname{Im} i_*$. Take $\eta \in \operatorname{Ker} f_*$. We have only to show the following: for any connected Galois étale covering space $\xi : Y \to X$, the element η acts trivially on $\pi_0(\xi^{-1}(X_0))$, where we denote the base change of X_0 via ξ by $\xi^{-1}(X_0)$. Take the normalization

$$Y \xrightarrow{h} S' \xrightarrow{u} S$$

of *S* in the composite $f \circ \xi : Y \to X \to S$. Then $u : S' \to S$ is an étale covering space (Proposition 4.5(2) and Proposition 4.8(3)). The action of η on Y/X induces an action of η on S'/S. Since $\eta \in \text{Ker } f_*$, the element η acts trivially on S'/S. Thus, the element η acts trivially on $\pi_0(\xi^{-1}(X_0))$, which implies that $\eta \in \text{Im } i_*$. Therefore, the sequence is exact.

Definition 4.23. Let (X, S, f) be a triple satisfying Condition (C) (Definition 4.17). We define the *orbifold* (S, B) associated to f in the following way. By P(S) we denote the set of all points on S of codimension one. Take $s \in P(S)$. Put $K_s :=$ Frac $\mathbb{O}_{S,s}^{sh}$. By f_s we denote the base change of f via the composite Spec $\mathbb{O}_{S,s}^{sh} \rightarrow$ Spec $\mathbb{O}_{S,s} \rightarrow S$ of the canonical morphisms. Take the maximal base field \widetilde{K}_s of f_s (Definition 4.9). Proposition 4.15(1) shows that the field extension \widetilde{K}_s/K_s is finite and Galois. We define a map B on P(S) by $s \mapsto \widetilde{K}_s/K_s$ (Definition 3.6). Let us show that the pair (S, B) is an orbifold. By S_0 we denote the open subscheme of S

that is the complement of the closure of Supp *B*. Take a nonempty open subscheme S_1 of *S* over which *f* is separable. Then $S_1 \subset S_0$ (Proposition 4.15(2)), which implies that Supp *B* is locally finite. Thus, the pair (S, B) is an orbifold.

Using the above orbifold, we give an étaleness criterion for finite coverings of X:

Theorem 4.24. Let (X, S, f) be a triple satisfying Condition (C^*) (Definition 4.17). Take the orbifold (S, B) associated to f (Definition 4.23). Let $u : S' \to S$ be a finite surjective generically separable morphism between locally Noetherian normal integral schemes. Take the scheme X' and the morphism $u' : X' \to X$ given by Proposition 4.8. Then u' is étale if and only if u induces an orbifold étale morphism $(S', B') \to (S, B)$ (Definition 3.12).

Proof. First, we assume that S = Spec Q and S' = Spec Q' where Q and Q' are discrete valuation rings. The morphism u induces a finite flat extension Q'/Q of discrete valuation rings. Put J := Frac Q, J' := Frac Q', $K := \text{Frac } Q^{\text{sh}}$, and $K' := J' \otimes_J K$. The field extension J'/J induced by u is finite and separable. Take the maximal unramified extension I of J in J'. We may embed I in K over J. By (1) and (2) of Proposition 4.8, we may assume that J = I. Then K' is a field. By Proposition 4.6, we may assume that J = K and J' = K'. Then the theorem follows from Lemma 4.12.

Next, let us show the general case. The "only if" part follows from the first case and Proposition 4.6. Let us show the "if" part. Since f maps any point of codimension one to a point of codimension at most one (Lemma 3.4(2)), the first case and Proposition 4.6 show that u' is étale in codimension one. Thus, Zariski–Nagata purity (Lemma 3.5) shows that u' is étale, which proves the "if" part. \Box

Definition 4.25. Let (X, S, f) be a triple satisfying Condition (\mathbb{C}^*) (Definition 4.17). Take the orbifold (S, B) associated to f (Definition 4.23). Choose a geometric point \bar{x} on X. Put $\bar{s} := f(\bar{x})$. Assume that the image of \bar{s} on S is a regular point on S (e.g., the generic point of S). We define the *homomorphism* $f_*^{orb} : \pi_1(X, \bar{x}) \to$ $\pi_1(S, B, \bar{s})$ *induced by* f in the following way. Let $u : (S', B') \to (S, B)$ be a connected orbifold étale covering space. Take the scheme X' and the morphism $u' : X' \to X$ given by Proposition 4.8. Then $u' : X' \to X$ is a connected étale covering space (Theorem 4.24). Thus, we obtain a surjective homomorphism $f_*^{orb} : \pi_1(X, \bar{x}) \to \pi_1(S, B, \bar{s})$.

Proof of Theorem 1.1. We may show the theorem in the same way as in the proof of Theorem 4.22. We have only to use Theorem 4.24 instead of Proposition 4.8(3). \Box

Remark 4.26. We generalize the definition of an orbifold (S, B) (Definition 3.6) by the following two modifications: replace P(S) by all points on S; remove the finiteness assumption on B_s/K_s . We may define the fundamental group of (S, B) in the same way as in the case of an orbifold. The morphism f induces

a generalized orbifold (S, B). Note that if f is separable, then (S, B) is trivial (Proposition 4.8(3)). Replacing an orbifold by a generalized orbifold, we may show Theorems 4.24 and 1.1 for any triple (X, S, f) satisfying Condition (C) (Definition 4.17) without the regularity assumption on X and the flatness assumption on f in codimension one (Examples 4.16 and 4.21).

5. Orbifold trivializations of orbifold curves

In this section, we fix an algebraically closed field k of characteristic $p \ge 0$. We study orbifold trivializations of orbifold k-curves and classify simply connected cyclic orbifold k-curves.

Definition 5.1. An orbifold (C, B) (Definition 3.6) is called an *orbifold* k-curve (resp. a proper orbifold k-curve) if C is a k-curve (resp. a proper k-curve). If p > 0 and $[B_s : K_s]$ is power of p for any $s \in P(C)$, we say that an orbifold (C, B) is a p-orbifold k-curve.

Since the underlying scheme of any orbifold k-curve is a smooth k-curve, we study ramified coverings of smooth k-curves. The Riemann–Hurwitz formula shows the following:

Lemma 5.2. Let $u : C \to \mathbb{P}^1_k$ be a finite tamely ramified k-morphism of degree d between connected proper smooth k-curves. By N we denote the number of the branched points of u. Then:

- (1) $N \neq 1$.
- (2) If N = 2, then C is isomorphic to \mathbb{P}_k^1 , u ramifies at exactly two points, and both of the two ramification indices are equal to d.

Proposition 5.3. Let (\mathbb{P}^1_k, B) be a proper tame orbifold k-curve. If # Supp $B \le 1$, then (\mathbb{P}^1_k, B) is simply connected. Assume that Supp $B = \{0, \infty\}$. For s = 0 and ∞ we put $n_s := [B_s : K_s]$. Then:

- (1) The orbifold (\mathbb{P}^1_k, B) is simply connected if and only if $gcd(n_0, n_\infty) = 1$.
- (2) There exists an orbifold trivialization of (\mathbb{P}^1_k, B) if and only if $n_0 = n_{\infty}$.

In Statement (2), the restriction of the orbifold trivialization to any connected component ramifies at exactly two points.

Proof. The first statement follows from Lemma 5.2(1). Let us show the other statements. Put $d := \gcd(n_0, n_\infty)$. Take a parameter t of \mathbb{P}^1_k so that t(0) = 0 and $t(\infty) = \infty$. Then the k-morphism $\mathbb{P}^1_k \to \mathbb{P}^1_k$, $t \mapsto t^d$ induces an orbifold étale morphism $(\mathbb{P}^1_k, B') \to (\mathbb{P}^1_k, B)$, where the equalities

$$[B'_s:K_s] = \begin{cases} n_i/d & \text{if } s = 0 \text{ or } \infty, \\ 1 & \text{otherwise.} \end{cases}$$

hold. Thus, Lemma 5.2 proves the proposition.

Proposition 5.4. Let (\mathbb{P}_k^1, B) be a proper tame orbifold k-curve with Supp $B = \{0, 1, \infty\}$. For $s = 0, 1, and \infty$, we put $n_s := [B_s : K_s]$. Assume that $n_0, n_1, and n_\infty$ are pairwise coprime. Then there exists a Galois orbifold trivialization of (\mathbb{P}_k^1, B) with noncommutative simple Galois group.

Proof. First, we consider the case p = 0. By $F_2 = \langle x_0, y_0 \rangle$ we denote the free group of rank two. Take the elements x, y, and z of the triangle group $\Delta(n_0, n_1, n_\infty)$ in the definition of $\Delta(n_0, n_1, n_\infty)$ in Section A. We define a homomorphism $\phi : F_2 \rightarrow \Delta(n_0, n_1, n_\infty)$ by $x_0 \mapsto x$ and $y_0 \mapsto y$. Then the equality $\phi(y_0^{-1}x_0^{-1}) = z$ holds. Since the étale fundamental group of $\mathbb{P}^1_k \setminus \{0, 1, \infty\}$ is isomorphic to the profinite completion of F_2 (see [Lieblich and Olsson 2010] for a purely algebraic proof), Theorem A.9 shows the proposition.

Next, we consider the case p > 0. Take a complete discrete valuation ring R of characteristic zero whose residue field is isomorphic to k (e.g., the ring of Witt vectors over k). Put K := Frac R. The case p = 0 shows the following: replacing R by a finite extension of R, there exist a finite noncommutative simple group G and a K-morphism $w_K : Y_K \to \mathbb{P}^1_K$ between connected proper smooth K-curves satisfying the following condition:

(0) w_K is the quotient morphism $Y_K \to \mathbb{P}^1_K = Y_K/G$ whose branch points are equal to 0, 1, and ∞ , over which each ramification index is equal to n_0 , n_1 , and n_∞ , respectively.

In the following, we take an appropriate *R*-model *w* of w_K in the same way as in [Raynaud 1994, §§6.1–6.3]. Remark that in [Raynaud 1994, §6] the group *G* is a quasi-*p*-group and the ramification indices of w_K are equal to powers of *p*. However, these conditions are used only from Lemma 6.3.6. Replacing *R* by a finite extension of *R*, we obtain an *R*-morphism $w : Y \rightarrow P$ between *R*-schemes satisfying Conditions (1)–(6):

- (1) Y is a projective normal semistable R-curve that is an R-model of Y_K .
- (2) *P* is a projective normal semistable *R*-curve that is an *R*-model of \mathbb{P}^1_K .
- (3) w is the quotient morphism $Y \rightarrow P = Y/G$.
- (4) The restriction of w to the generic fibers is equal to w_K .
- (5) The closure of each branch point of w_K in P is contained in the smooth locus $P_{\rm sm}$ of the R-scheme P.

To state Condition (6) below, we introduce notation. Condition (2) shows the following two statements on the special fiber P_k of P: any irreducible component is isomorphic to \mathbb{P}_k^1 , and the dual graph of the irreducible components is a tree Γ_P . By 0_R , 1_R , and ∞_R we denote the closures of 0, 1, and ∞ on \mathbb{P}_k^1 in P, respectively.

For s = 0, 1, and ∞ , we denote the reduction of s_R by s_k . Condition (5) implies that s_k is contained in exactly one irreducible component C_s . We denote the vertex of Γ_P corresponding to C_s by e_s . Any two vertices e_s and e_t are connected by a unique line l_{st} on Γ_P . The intersection $l_{01} \cap l_{1\infty} \cap l_{\infty 0}$ is exactly one vertex $e_{01\infty}$. We denote the irreducible component corresponding to $e_{01\infty}$ by $C_{01\infty}$.

(6) P is a successive blowing-up of P¹_R, and the strict transform of the special fiber of P¹_R is equal to C_{01∞}.

Take a subgroup *H* of *G*. By Condition (1), we obtain an *R*-morphism $u: Y \to X$ between *R*-schemes satisfying Conditions (7) and (8) [Raynaud 1990, Corollaire of Proposition 5]:

- (7) X is a proper normal semistable R-curve with connected smooth generic fiber.
- (8) *u* is the quotient morphism $Y \to X = Y/H$.

Conditions (3) and (8) give an *R*-morphism $v : X \to P$ between *R*-schemes such that $w = v \circ u$. Furthermore, the following condition is satisfied:

(9) u, v, and w are finite and surjective.

Since v and w are finite (Condition (9)), X and Y are Cohen–Macaulay (Conditions (1) and (7) and [EGA IV₂ 1965, 5.8.6]), and $P_{\rm sm}$ is regular (Condition (2)), the following condition is satisfied [Matsumura 1989, 23.1]:

(10) v and w are flat over $P_{\rm sm}$.

By Γ_Y and Γ_X we denote the dual graphs of the irreducible components of the special fibers Y_k and X_k of Y and X. The group G acts on Γ_Y . The quotient morphisms u and v induce the quotient maps of the actions of G and H from the vertices of Γ_Y to the vertices of Γ_P and Γ_X , respectively. If an element g of G fixes an edge e of Γ_Y , then g does not exchange the two vertices on the edge e [Raynaud 1994, 6.3.5]. Thus, the quotients of the actions of G and H on Γ_Y are canonically isomorphic to Γ_P and Γ_X , respectively. Therefore, the morphisms u and v induce maps $\Gamma_Y \to \Gamma_X \to \Gamma_P$, respectively, that preserve the vertices and the edges.

Take the generic point λ of an irreducible component of P_k . Choose the generic point η of an irreducible component of Y_k over λ . Conditions (9) and (10) show that $\mathbb{O}_{Y,\eta}/\mathbb{O}_{P,\lambda}$ is a finite flat extension of discrete valuation rings. Conditions (1) and (2) imply that the ramification index of $\mathbb{O}_{Y,\eta}/\mathbb{O}_{P,\lambda}$ is equal to 1. Thus, the inertia group I_η of $\mathbb{O}_{Y,\eta}/\mathbb{O}_{P,\lambda}$ is a *p*-group. Since *G* is a noncommutative simple group and any simple *p*-group is commutative, the inequality $G \neq I_\eta$ holds. We take the above subgroup *H* of *G* so that $H = I_\eta$. Put $\theta := u(\eta)$. Then the extension of the residue fields of $\mathbb{O}_{X,\theta}/\mathbb{O}_{P,\lambda}$ is separable. Conditions (2) and (7) imply that the ramification index of $\mathbb{O}_{X,\theta}/\mathbb{O}_{P,\lambda}$ is equal to 1. Thus, the extension $\mathbb{O}_{X,\theta}/\mathbb{O}_{P,\lambda}$ is étale.

Take $g \in G$. Replacing η and H by $g\eta$ and gHg^{-1} , respectively, we obtain X^{g} and θ^{g} corresponding to X and θ , respectively, in the same way as in the above argument. Furthermore, the extension $\mathbb{O}_{X^g,\theta^g}/\mathbb{O}_{P,\lambda}$ is étale. Since v is finite (Condition (9)) and R is excellent, the normalization Y' of the fiber product of X^g for all $g \in G$ over P is finite over P. By Y'_K we denote the generic fiber of Y'. Since G is simple and $G \neq H$, the intersection $\bigcap_{g \in G} gHg^{-1}$ is the trivial group. Thus, any connected component of Y'_K is P_K -isomorphic to Y_K (Lemma 3.14). Since Y is normal (Condition (1)) and w is finite (Condition (9)), any connected component of Y' is P-isomorphic to Y. Thus, the product of θ^g for all $g \in G$ over λ gives a point η_0 on Y over λ such that the extension $\mathbb{O}_{Y,\eta_0}/\mathbb{O}_{P,\lambda}$ is étale. Since w is a Galois covering, the extension $\mathbb{O}_{Y,g\eta_0}/\mathbb{O}_{P,\lambda}$ is étale for any $g \in G$. Since λ is arbitrary, the morphism w is étale at the generic point of any irreducible component of Y_k . Put $P' := P_{sm} \setminus (0_R \cup 1_R \cup \infty_R)$. By w' we denote the restriction $w|_{w^{-1}(P')}: w^{-1}(P') \to P'$. Condition (0) implies that the restriction of w' to the generic fibers is étale. Thus, the morphism w' is étale in codimension one. Therefore, Zariski–Nagata purity (Lemma 3.5) shows that w'is étale. Thus, by Lemma 5.2 and the same method as above, we may assume that $\Gamma_P = l_{01} \cup l_{1\infty} \cup l_{\infty 0}$ after successive blowing-down of exceptional curves on Y and P (Condition (6)).

The normalizations of the preimages of C_s and $C_{01\infty}$ under w are proper smooth k-curves C'_s and $C'_{01\infty}$, respectively. The covering C'_s/C_s branches at s_k , over which each ramification index is equal to n_s ([Raynaud 1994, 6.3.2] and Conditions (0), (4), and (5)). Since w' is étale, Lemma 5.2 implies that the preimage $C'_{01\infty}$ is connected and the covering $C'_{01\infty}/C_{01\infty}$ branches at exactly three points, over which each ramification index is equal to n_0 , n_1 , and n_∞ , respectively. Therefore, the covering $C'_{01\infty}/C_{01\infty}$ induces a desired orbifold trivialization.

Lemma 5.5. Let C be a connected proper smooth k-curve of positive genus. Take a closed point s on C and an integer n. Assume that $p \nmid n$. Then there exists a connected étale covering space $u : C' \rightarrow C$, a divisor D on C', and a rational function h on C' such that $u^*[s] - nD = (h)$, where [s] and (h) are the divisors defined by s and h, respectively.

Proof. We may assume that *n* is positive. Since the genus of *C* is positive and $p \nmid n$, we may take a connected étale covering space $u : C' \to C$ of degree *n*. We denote the genus of *C'* by *g*. By *J* we denote the Jacobian variety of *C'* over *k*. Take a closed point s_0 on *C'*. Since the morphism $(C')^g \to J$ defined by $(s_i)_{i=1}^g \mapsto \sum_{i=1}^g ([s_i] - [s_0])$ is surjective and the multiplication of *J* by *n* is surjective, there exists $(s_i)_{i=1}^g \in (C'(k))^g$ such that $u^*[s] - n[s_0]$ is linearly equivalent to $n \sum_{i=1}^g ([s_i] - [s_0])$, which proves the lemma.

Proposition 5.6. Let (C, B) be a connected proper tame orbifold k-curve with # Supp B = 1. Assume that the genus of C is positive. Then there exists an orbifold trivialization of (C, B) that is the composite $u \circ v : C'' \to C' \to C$ of two finite coverings where u is étale and v is totally ramified over each branch point.

Proof. Take $s \in \text{Supp } B$. Put $n := [B_s : K_s]$. Take a connected étale covering space $u : C' \to C$ and a rational function h on C' given by Lemma 5.5. By $v : C'' \to C'$ we denote the normal model of the equation $z^n = h$. Then $u \circ v$ induces a desired orbifold trivialization.

Proposition 5.7. Let (C, B) be a connected tame orbifold k-curve. Put M :=#Supp B. For each $s \in$ Supp B, we put $n_s := [B_s, K_s]$. Then there exists an orbifold trivialization of (C, B) if and only if neither of the following conditions are satisfied: (a) $C \cong \mathbb{P}^1_k$ and M = 1; (b) $C \cong \mathbb{P}^1_k$, M = 2, and $n_s \neq n_t$ where Supp $B = \{s, t\}$.

Proof. In the proof of the "if" part, we may replace *C* by the smooth compactification of *C*. Thus, we may assume that *C* is proper over *k*. By Proposition 3.18, we have only to consider the following cases: (1) g(C) = 0 and $M \le 2$; (2) g(C) = 0, M = 3, and $gcd(n_s, n_t) = 1$ for $s \ne t$; (3) g(C) > 0 and M = 1. Cases (1), (2), and (3) follow from Propositions 5.3, 5.4, and 5.6, respectively.

In the following, we provide steps in order to prove Proposition 5.12. Assume that p > 0. Let *C* be a connected proper smooth *k*-curve. We denote the sheaf of rational functions on *C* by \mathcal{M}_C . Put $\mathcal{P}_C := \mathcal{M}_C / \mathbb{O}_C$. The exact sequence of abelian sheaves $0 \to \mathbb{O}_C \to \mathcal{M}_C \to \mathcal{P}_C \to 0$ induces a long exact sequence

$$H^0(C, \mathcal{M}_C) \xrightarrow{\phi_C} H^0(C, \mathcal{P}_C) \xrightarrow{\psi_C} H^1(C, \mathbb{O}_C).$$

By F_C we denote the absolute Frobenius endomorphism of C and its actions on the cohomology groups $H^0(C, \mathcal{P}_C)$ and $H^1(C, \mathbb{O}_C)$. We define F_C^0 as the identity map and, for each positive integer d, we inductively define F_C^d by $F_C^d := F_C^{d-1} \circ F_C$.

Lemma 5.8. For any $\xi \in H^0(C, \mathcal{P}_C)$, there exists a connected étale covering space $u: C' \to C$, a nonnegative integer d, and a rational function h on C' such that $u^* F_C^d \xi = \phi_{C'}(h)$, where u^* is the homomorphism $H^0(C, \mathcal{P}_C) \to H^0(C', \mathcal{P}_{C'})$ induced by u.

Proof. Since the *k*-vector space $H^1(C, \mathbb{O}_C)$ is finite-dimensional, we may take a nonnegative integer *d* and a polynomial $G(X) = \sum_{i=0}^n c_i X^i \in k[X]$ so that $c_0 \neq 0$ and $G(F_C)\eta = 0$, where we put $\eta := F_C^d \psi_C \xi$. Take an affine covering $\mathfrak{U} = \{U_i\}$ of *C* and a representative $\{a_{ij}\} \in C^1(\mathfrak{Q}, \mathfrak{O}_C)$ of η . We define a polynomial $G^{(p)}(X) \in k[X]$ by $G^{(p)}(X) := \sum_{i=0}^n c_i X^{p^i}$. Since $G(F_C)\eta = 0$, there exists

 $\{a_i\} \in C^0(\mathcal{U}, \mathbb{O}_C)$ such that $G^{(p)}(a_{ij}) = a_j - a_i$ for all *i* and all *j*. Thus, we may define an étale covering space $u : C' \to C$ by the equations

$$\begin{cases} G^{(p)}(z_i) = a_i & \text{on } U_i, \\ z_j - z_i = a_{ij} & \text{on } U_i \cap U_j. \end{cases}$$

By definition, the pullback of η via u splits, which proves the lemma.

We denote the cokernel of an endomorphism ϕ of a module M by M_{ϕ} .

Lemma 5.9. Let R be an excellent discrete valuation ring of positive characteristic with separably closed residue field. By \hat{R} we denote the completion of R with respect to the maximal ideal. Put $K := \operatorname{Frac} R$, $K^{\operatorname{sh}} := \operatorname{Frac} R^{\operatorname{sh}}$, and $\hat{K} := \operatorname{Frac} \hat{R}$. We denote the Frobenius endomorphisms on these fields by F. The canonical inclusion homomorphisms $K \to K^{\operatorname{sh}}$ and $K^{\operatorname{sh}} \to \hat{K}$ induce the canonical homomorphisms $\alpha : K \to K^{\operatorname{sh}}_{F-1}$ and $\beta : K^{\operatorname{sh}}_{F-1} \to \hat{K}_{F-1}$, respectively. Then:

- (1) α is surjective and $R \subset \text{Ker } \alpha$.
- (2) β is an isomorphism.

Proof. Since K^{sh} is algebraically closed in \hat{K} by the approximation property [Bosch et al. 1990, 3.6.9], Artin–Schreier theory shows that β is injective. Since \hat{R} is isomorphic to the formal power series ring over the separably closed residue field of R, the relation $\hat{R} \subset (F-1)\hat{K}$ holds, which implies that $\beta \circ \alpha$ is surjective. Thus, Statement (2) holds, which implies Statement (1).

We recall the definition of the addition of the ring of Witt vectors W(A) with coefficient ring A. Let n be a nonnegative integer. Put

$$W_n(X_0,...,X_n) := \sum_{i=0}^n p^i X_i^{p^{n-i}} \in \mathbb{Z}[X_0,...,X_n].$$

We inductively define S_n as the unique polynomial in $\mathbb{Z}[X_0, \ldots, X_n, Y_0, \ldots, Y_n]$ satisfying the equality $W_n(S_0, \ldots, S_n) = W_n(X_0, \ldots, X_n) + W_n(Y_0, \ldots, Y_n)$. For $a = (a_0, \ldots, a_n, \ldots) \in W(A)$ and $b = (b_0, \ldots, b_n, \ldots) \in W(A)$, the addition of Witt vectors is defined by

$$a + b := (S_0(a_0, b_0), \dots, S_n(a_0, \dots, a_n, b_0, \dots, b_n), \dots).$$

Lemma 5.10. We denote the ideal of $\mathbb{Z}[X_0, \ldots, X_n, Y_0, \ldots, Y_n]$ generated by $\{X_i Y_j\}_{0 \le i,j \le n}$ by *I*. Then the equality $S_n \equiv X_n + Y_n \mod I$ holds. In particular, the equality

 $(a_0, \ldots, a_{n-1}, a_n, \ldots) + (0, \ldots, 0, b_n, \ldots) = (a_0, \ldots, a_{n-1}, a_n + b_n, \ldots)$

holds in W(A).

Kentaro Mitsui

Proof. Let us show the first equality by induction on *n*. The case n = 0 is clear. Assume that the case *i* is proved for any i < n. By the induction hypothesis, the equality $p^n S_n \equiv p^n X_n + p^n Y_n \mod I$ holds, which proves the case *n*. Thus, the first equality holds for any *n*. The first equality shows the last equality.

For a positive integer *n*, we denote the ring of length-*n* Witt vectors with coefficient ring *A* by $W_n(A)$. We denote the Frobenius endomorphism on $W_n(A)$ by *F*. Take a connected étale covering space $u : C' \to C$, a closed point *s* on *C*, and $s' \in u^{-1}(s)$. Put $K_s := \operatorname{Frac} \mathbb{O}_{C,s}^{\operatorname{sh}}$ and $K'_{s'} := \operatorname{Frac} \mathbb{O}_{C',s'}^{\operatorname{sh}}$. The extensions $K'_{s'}/K_s$ for all $s' \in u^{-1}(s)$ induce a homomorphism

$$\Delta_{u,s,n}: W_n(K_s)_{F-1} \to \bigoplus_{s' \in u^{-1}(s)} W_n(K'_{s'})_{F-1}$$

Put $A_{u,s} := \mathbb{O}_{C'}(C' - u^{-1}(s))$. The canonical homomorphisms $A_{u,s} \to K'_{s'}$ for all $s' \in u^{-1}(s)$ induce a homomorphism

$$\phi_{u,s,n}: W_n(A_{u,s})_{F-1} \to \bigoplus_{s' \in u^{-1}(s)} W_n(K'_{s'})_{F-1}.$$

Put $\Delta_{C,s,n} := \lim_{u \to u} \Delta_{u,s,n}$ and $\phi_{C,s,n} := \lim_{u \to u} \phi_{u,s,n}$, where $u : C' \to C$ runs through all connected étale covering spaces of *C*. By construction, the homomorphisms $\Delta_{C,s,n}$ and $\phi_{C,s,n}$ are compatible with the reductions of the rings of Witt vectors.

Lemma 5.11. *The relation* Im $\Delta_{C,s,n} \subset \text{Im } \phi_{C,s,n}$ *holds.*

Proof. Take $\xi \in W_n(K_s)_{F-1}$. Put $\eta := \Delta_{C,s,n}(\xi)$. We have to show that $\eta \in \text{Im } \phi_{C,s,n}$. By induction on *n*, we have only to consider the following cases: (1) n = 1; (2) n > 1 and ξ is contained in the kernel of the reduction homomorphism $W_n(K_s) \to W_{n-1}(K_s)$. Note that $W_1(A) = A$ for any ring *A*. Lemma 5.10 reduces Case (2) to Case (1). Thus, we may assume that n = 1. Lemma 5.9(1) shows that the canonical homomorphism $\mathcal{M}_{C,s} \to K_s$ induces a surjective homomorphism $P : \mathcal{P}_{C,s} \to (K_s)_{F-1}$. Thus, we have only to show that $\Delta_{C,s,1}(P(\theta)) \in \text{Im } \phi_{C,s,1}$ for any $\theta \in \mathcal{P}_{C,s}$. Since the equality $X^d = 1 + (\sum_{i=0}^{d-1} X^i)(X-1)$ holds in the polynomial ring k[X] for any positive integer *d*, we may replace θ by $F^d \theta$ for any positive integer *d*. Thus, the lemma follows from Lemma 5.8. □

Proposition 5.12. Assume that p > 0. Let (C, B) be a connected proper cyclic *p*-orbifold *k*-curve with # Supp B = 1. Then there exists an orbifold trivialization of (C, B) that is the composite $u \circ v : C'' \to C' \to C$ of two finite coverings where *u* is étale and *v* is totally ramified over each branch point.

Proof. Take $s \in \text{Supp } B$. Take an integer n so that $[B_s : K_s] = p^n$. Since $H^1(K_s, W_n) = 0$, the exact sequence of $\text{Gal}(\overline{K_s}/K_s)$ -modules

$$0 \longrightarrow \mathbb{Z}/p^n \mathbb{Z} \longrightarrow W_n(\overline{K_s}) \xrightarrow{F-1} W_n(\overline{K_s}) \longrightarrow 0$$

induces an isomorphism $H^1(K_s, \mathbb{Z}/p^n\mathbb{Z}) \cong W_n(K_s)_{F-1}$. In particular, the field extension B_s/K_s is induced by an element $\xi \in W_n(K_s)_{F-1}$. Lemma 5.11 gives a connected étale covering space $u : C' \to C$ and an element $\eta \in W_n(A_{u,s})_{F-1}$ such that the equality $\Delta_{u,s,n}(\xi) = \phi_{u,s,n}(\eta)$ holds. By K' we denote the function field of C'. The image of η under the canonical homomorphism $W_n(A_{u,s})_{F-1} \to W_n(K')_{F-1}$ induces a cyclic extension K''/K' of degree p^n . Take the normalization $v : C'' \to C'$ of C' in K''. By the choice of η , the morphism $u \circ v$ induces a desired orbifold trivialization.

Proposition 5.13. *Let* (C, B) *be a connected p-orbifold k-curve. Then there exists an orbifold trivialization of* (C, B)*.*

Proof. We may replace *C* by the smooth compactification of *C*. Thus, we may assume that *C* is proper over *k*. By Proposition 3.18, we may assume that # Supp B = 1. Take $s \in$ Supp *B*. Put $m := [B_s : K_s]$. Let us show the proposition by induction on *m*. The case m = 1 is clear. Assume that m > 1. Since B_s/K_s is solvable, we may take a Galois extension B'_s/K_s of degree *p* in B_s . For each $t \in P(C) \setminus \{s\}$, we put $B'_t := K_t$. We define a map B' on P(C) by $t \mapsto B'_t/K_t$ (Definition 3.6). Then the pair (C, B') is an orbifold. Applying Proposition 5.12 to (C, B'), we may reduce the case *m* to the case m/p. Thus, the case *m* holds by the induction hypothesis.

Definition 5.14. The *Euler characteristic* e(C) of a proper smooth *k*-curve *C* is the ℓ -adic Euler characteristic of *C*, which does not depend on the choice of the prime number ℓ that is prime to *p*. Let (C, B) be a proper orbifold *k*-curve. Take $s \in \text{Supp } B$. We use the notation B_s/K_s introduced in Definition 3.6. By B_s° and K_s° we denote the valuation rings of the discrete valuation fields B_s and K_s , respectively. We define the *orbifold Euler characteristic* e(C, B) of (C, B) by

$$e(C, B) := e(C) - \sum_{s \in \text{Supp } B} \frac{1}{[B_s : K_s]} \operatorname{length}_{K_s^{\circ}}(\Omega^1_{B_s^{\circ}/K_s^{\circ}}).$$

The Riemann–Hurwitz formula shows the following:

Proposition 5.15. Let (C, B) and (C', B') be two proper orbifold k-curves. If there exists an orbifold étale morphism $(C', B') \rightarrow (C, B)$ of degree n, then the equality e(C', B') = ne(C, B) holds. In particular, if e(C, B) > 0 and C is connected, then the underlying curve of any connected orbifold étale covering space of (C, B) is isomorphic to \mathbb{P}^1_k .

Proof of Theorem 1.3. Let us show Statement (2). Assume that C is not proper over k. Take the smooth compactification \overline{C} of C. We may choose an extension \overline{B} of B to $P(\overline{C})$ so that the orbifold $(\overline{C}, \overline{B})$ satisfies neither Condition (a) nor Condition (b). Thus, we may assume that C is proper over k. First, we consider

the "if" part. By Propositions 3.18, 5.15, and 5.13, we may assume that $B^t = B$. In that case, the "if" part follows from Proposition 5.7. Next, we consider the "only if" part. Take $s \in \text{Supp } B$ and an orbifold trivialization $C_1 \to (C, B)$. By Proposition 5.7, we have only to show that Condition (a) is not satisfied. Assume that Condition (a) is satisfied. Since *C* is simply connected, Proposition 5.12 gives an orbifold trivialization $u : C_2 \to (C, B^w)$ that is totally ramified over the unique branch point *s*. Condition (a) and Proposition 5.15 imply that $C_2 \cong \mathbb{P}^1_k$. Take the normalization C_3 of $C_1 \times_C C_2$ and the canonical projection $v : C_3 \to C_2$. Lemma 3.14 shows that *v* is tamely ramified over the unique branch point $u^{-1}(s)$, which contradicts Lemma 5.2. Thus, Condition (a) is not satisfied.

Let us show Statement (1). The "if part" follows from Proposition 5.3. We consider the "only if" part. Assume that Condition (a) is not satisfied. Statement (2) implies that $C \cong \mathbb{P}^1_k$ and $M \leq 2$. Proposition 5.13 implies that $B^t = B$. Proposition 5.3 implies that $gcd(n_s, n_t) = 1$ for $s \neq t$. Thus, Condition (b) is satisfied. Therefore, the "only if" part holds.

6. Fundamental groups of elliptic fibrations

6A. *Elliptic fibrations.* We study elliptic surfaces by localizing the fibrations with respect to the base curves. To this end, we generalize the definition of elliptic surfaces. We refer to [Liu 2002, §§8–9] for fibered surfaces.

Definition 6.1. An *elliptic fibration* is a triple (X, C, f) satisfying the following conditions:

- (1) C and X are excellent regular integral schemes of dimension one and two, respectively.
- (2) $f: X \to C$ is a proper morphism.
- (3) The homomorphism $\mathbb{O}_C \to f_*\mathbb{O}_X$ associated to f is an isomorphism.
- (4) The generic fiber of f is a proper smooth curve of genus one.

Let (X, C, f) be an elliptic fibration. A prime divisor D on X is said to be a (-1)-curve if the following conditions are satisfied. Put $k := \Gamma(D, \mathbb{O}_D)$. Then D is k-isomorphic to \mathbb{P}_k^1 and deg $\mathbb{O}_X(D)|_D = -1$. If any fiber of f does not contain a (-1)-curve, then (X, C, f) is said to be *relatively minimal*. The *multiplicity* of a closed fiber F of f is the multiplicity of the divisor F on X (Section 2). The minimal regular C-model of the Jacobian of the generic fiber of f is called the *Jacobian fibration* of f.

Remark 6.2. Conditions (2) and (3) show that f is surjective. Thus, Condition (1) shows that f is flat. The multiplicity of F does not depend on the choice of the proper regular C-model of the generic fiber of f [Liu 2002, 9.2.7].

Lemma 6.3. Let (X, C, f) be an elliptic fibration and $\xi : Y \to X$ a connected étale covering space. Take the Stein factorization

$$Y \xrightarrow{h} D \xrightarrow{v} C$$

of the composite $f \circ \xi : Y \to X \to C$. Then:

(1) (Y, D, h) is an elliptic fibration.

(2) v is finite, flat, surjective, and generically separable.

Choose an integral scheme C' and a finite flat morphism $u : C' \to C$ such that v factors through u. Take the normalization X' of $X \times_C C'$, the canonical projections $u' : X' \to X$ and $f' : X' \to C'$, and the unique morphism $\xi' : Y \to X'$ satisfying $\xi = u' \circ \xi'$ and $h = f' \circ \xi'$. Then:

- (3) (X', C', f') is an elliptic fibration.
- (4) u' and ξ' are finite, étale, and surjective.

Proof. Since X is regular and ξ is étale, the scheme Y is regular. Thus, Statements (1) and (2) follow from Proposition 4.5. Statement (4) follows from Lemma 4.7. Since X is regular and u' is étale, the scheme X' is regular. Thus, Statement (3) follows from Proposition 4.5.

We frequently use the following:

Proposition 6.4 [Liu et al. 2004, 6.6]. Let *C* be the spectrum of a complete discrete valuation ring with algebraically closed residue field and field of fractions *K*. Let (X, C, f) be a relatively minimal elliptic fibration with Jacobian fibration (E, C, g). By X_K and E_K we denote the generic fibers of f and g, respectively. Then the special fiber of f is of type $_m T$ (the Kodaira symbol) if and only if the special fiber of g is of type T and the order of the torsor $[X_K] \in H^1(K, E_K)$ is equal to m.

We refer to [Liu 2002, 8.3.39, 8.3.44, 9.3.31, and 9.3.32] for desingularizations and the minimal desingularizations of fibered surfaces.

Lemma 6.5. Let (X, C, f) be an elliptic fibration with generic fiber X_K and $\xi_K : Y_K \to X_K$ a finite morphism between geometrically connected K-curves of genus one. Take the normalization $\xi : Y \to X$ of X in ξ_K . Assume that f is smooth and that the residue field at any closed point on C is algebraically closed. Then the triple $(Y, C, f \circ \xi)$ is a relatively minimal elliptic fibration.

Proof. We may assume that *C* is the spectrum of a discrete valuation ring. Take the minimal desingularization $\lambda : \hat{Y} \to Y$ of *Y*. By X_k , Y_k , and \hat{Y}_k we denote the special fibers of f, $f \circ \xi$, and $f \circ \xi \circ \lambda$, respectively. Choose an irreducible component *D* of Y_k and an irreducible component \hat{D} of \hat{Y}_k dominating *D*. Since $1 = g(X_k) \le g(D) \le g(\hat{D})$, the Néron–Kodaira classification of singular fibers implies that \hat{D} is the unique irreducible component of \hat{Y}_k whose geometric genus is equal to 1. Thus, the component D is the unique irreducible component of Y_k , and the component \hat{D} is the unique irreducible component of \hat{Y}_k that dominates D. Since λ is the minimal desingularization, the morphism λ is an isomorphism. Therefore, the triple $(Y, C, f \circ \xi)$ is a relatively minimal elliptic fibration.

Lemma 6.6. Let (X, C, f) be an elliptic fibration with Jacobian fibration (E, C, g). Assume that the reduction of any closed fiber of f is isomorphic to an elliptic curve and that the residue field at any closed point on C is algebraically closed. By X_K and E_K we denote the generic fibers of f and g, respectively. Take a positive multiple n of the order of the torsor $[X_K] \in H^1(K, E_K)$. Then there exists a C-morphism $X \to E$ whose restriction to the generic fibers induces the multiplication of their Jacobian E_K by n.

Proof. Take a finite Galois extension K'/K so that $X(K') \neq \emptyset$. Put G := Gal(K'/K) and $X_{K'} := X_K \times_K K'$. Choose a cocycle $c \in Z^1(G, E(K'))$ representing $[X_K]$. The curve X_K/K may be obtained as the quotient of a *G*-equivariant action on $X_{K'}/K'$ induced by c (see Section 6B). Moreover, an endomorphism of $Z^1(G, E(K'))$ induces a *G*-equivariant endomorphism of $X_{K'}$, whose quotient is a *K*-morphism between torsors of E_K . Since the endomorphism on $H^1(K, E_K)$ induced by the multiplication of E_K by n maps the torsor $[X_K]$ to the trivial torsor $[E_K]$, the endomorphism induces a *K*-morphism $\xi_K : X_K \to E_K$. Take the normalization $Y \to E$ of E in ξ_K . Proposition 6.4 shows that g is smooth. By Lemma 6.5, the scheme Y is the minimal regular *C*-model of X_K . Since the minimal regular *C*-model of X_K is unique up to unique *C*-isomorphism, the *C*-scheme Y is *C*-isomorphic to X, which concludes the proof. \Box

Corollary 6.7. Let C be the spectrum of a complete discrete valuation ring with algebraically closed residue field. Let (X, C, f) be a relatively minimal elliptic fibration with Jacobian fibration (E, C, g). Then the reduction of the special fiber of f is isomorphic to an ordinary elliptic curve if and only if the special fiber of g is an ordinary elliptic curve.

6B. *Étale coverings of local elliptic fibrations.* Let *R* be a complete discrete valuation ring with algebraically closed residue field *k* of characteristic $p \ge 0$ and field of fractions *K*. Put C := Spec *R*. Let (X, C, f) be an elliptic fibration with Jacobian fibration (E, C, g). By X_K and E_K we denote the generic fibers of *f* and *g*, respectively. By $[X_K]$ we denote the element of $H^1(K, E_K)$ corresponding to the torsor X_K of E_K . Take a separable closure \overline{K} of *K*. Take the maximal base field \widetilde{K} of *f* in \overline{K} (Definition 4.9). In this subsection, we determine the extension \widetilde{K}/K .

Take a finite Galois extension K'/K in \overline{K} so that $X(K') \neq \emptyset$. Put $G_{K'/K} :=$ Gal(K'/K). The group $H^1(G_{K'/K}, E(K'))$ may be regarded as a subgroup of

 $H^1(K, E_K)$ by the inflation homomorphism. Then the torsor $[X_K] \in H^1(K, E_K)$ is contained in $H^1(G_{K'/K}, E(K'))$ since $[X_K]$ splits over K'. Choose a cocycle $c \in Z^1(G_{K'/K}, E(K'))$ representing $[X_K]$. The extension K'/K induces a finite covering C'/C. By (E', C', g') we denote the Jacobian fibration of $X_K \times_K K'$. By the uniqueness of the normalization C' of C in K' and the Jacobian fibration g', we obtain a homomorphism $\rho : G_{K'/K} \to \operatorname{Aut}(C'/C) \to \operatorname{Aut}(E'/C)$, where the first arrow is induced by the automorphisms on the generic point of C' and the second arrow is induced by the base change of the automorphisms via g'. Furthermore, we obtain a map $\tau : G_{K'/K} \to E(K') \to \operatorname{Aut}(E'/C')$, where the first arrow is given by c and the second arrow is induced by the translation by addition. Since c is a cocycle, the map $\tilde{\tau} : G_{K'/K} \to \operatorname{Aut}(E'/C)$ defined by $\sigma \mapsto \tau(\sigma) \circ \rho(\sigma)$ is a homomorphism. By $\chi : E' \to Z := E'/\operatorname{Im} \tilde{\tau}$ we denote the quotient morphism of the action $\tilde{\tau}$. The quotient Z is a normal scheme over C whose generic fiber is isomorphic to X_K .

Lemma 6.8. If any element of $\text{Im } \tilde{\tau}$ fixes any closed point on E', then $\tilde{K} = K$.

Proof. We may assume that X is the minimal desingularization of Z. Since Z is normal, we may take a regular closed point z on Z. By $x \in X$ we denote the preimage of z. The extension \tilde{K}/K induces a finite covering \tilde{C}/C . Take the minimal desingularization X' of $X \times_C C'$ (resp. \tilde{X} of $X \times_C \tilde{C}$) and the canonical projection $u' : X' \to X$ (resp. $\tilde{u} : \tilde{X} \to X$). By the choice of x, the preimage $(u')^{-1}(x)$ consists of one closed point on X'. By Proposition 4.15(4) and the definitions of u' and \tilde{u} , the morphism u' factors through the finite étale surjective morphism \tilde{u} . Thus, the degree of \tilde{u} is equal to one, which implies that $\tilde{K} = K$. \Box

Assume that (X, C, f) is relatively minimal. Put $G_K := \text{Gal}(\overline{K}/K)$. By ${}_mT$ we denote the type of the special fiber of f (the Kodaira symbol). The type T is divided into the following three cases, (A), (M), and (E):

Case (A). Additive type: $T \neq I_n$ ($n \ge 0$). Since the residue field k of R is algebraically closed and the special fiber of f is simply connected, Lemma 6.3 implies that X is simply connected. In particular, the equality $\tilde{K} = K$ holds.

Case (M). Multiplicative type: $T = I_n$ (n > 0). Tate's uniformization gives an exact sequence of G_K -modules

$$0 \longrightarrow \mathbb{Z} \longrightarrow \mathbb{G}_{m,K}(\overline{K}) \xrightarrow{\pi} E_K(\overline{K}) \longrightarrow 0,$$

where π maps 1 to $q \in K$ satisfying 0 < |q| < 1. The exact sequence induces a long exact sequence

$$H^1(K, \mathbb{G}_{m,K}) \longrightarrow H^1(K, E_K) \longrightarrow H^2(K, \mathbb{Z}) \longrightarrow H^2(K, \mathbb{G}_{m,K}).$$

Kentaro Mitsui

Since $H^1(K, \mathbb{G}_{m,K}) = H^2(K, \mathbb{G}_{m,K}) = 0$ and $H^2(K, \mathbb{Z}) \cong \text{Hom}(G_K, \mathbb{Q}/\mathbb{Z})$, we obtain an isomorphism $\phi_M : H^1(K, E_K) \cong \text{Hom}(G_K, \mathbb{Q}/\mathbb{Z})$. Put $\gamma_M := \phi_M([X_K])$. The group Im γ_M is finite and cyclic. The Galois extension L/K corresponding to Ker γ_M is the minimum separable field extension that splits $[X_K]$. By D we denote the normalization of C in L. Put d := [L : K]. Then the normalization Y of $X \times_C D$ is a relatively minimal elliptic fibration over D with special fiber of type I_{dn} , and the induced morphism $Y \to X$ is étale (see the proof of [Liu et al. 2004, 8.3(b)]). In particular, the relation $\tilde{K} \supset L$ holds. Proposition 4.15(4) gives the relation $\tilde{K} \subset L$. Thus, the equality $\tilde{K} = L$ holds.

Case (*E*). *Elliptic type*: $T = I_0$. By \hat{E} we denote the formal group law associated to *E*. By \overline{R} and \overline{m} we denote the integral closure of *R* in \overline{K} and the maximal ideal of \overline{R} , respectively. Then \hat{E} gives a group structure on \overline{m} . By $\hat{E}(\overline{m})$ we denote this group. Since the canonical homomorphism $E(\overline{R}) \to E(\overline{K})$ is a G_K -module isomorphism by the valuative criterion for properness, we obtain an exact sequence of G_K -modules

$$0 \longrightarrow \widehat{E}(\overline{\mathfrak{m}}) \longrightarrow E(\overline{K}) \longrightarrow E(k) \longrightarrow 0.$$

The exact sequence induces a long exact sequence

$$0 \longrightarrow H^{1}(K, \widehat{E}(\overline{\mathfrak{m}})) \stackrel{\iota}{\longrightarrow} H^{1}(K, E_{K}) \longrightarrow H^{1}(K, E(k)) \stackrel{\Psi}{\longrightarrow} H^{2}(K, \widehat{E}(\overline{\mathfrak{m}})).$$

Since G_K acts trivially on E(k), we obtain an isomorphism $H^1(K, E(k)) \cong$ Hom $(G_K, E(k))$.

Lemma 6.9. If p = 0, then the group $H^i(K, \hat{E}(\overline{\mathfrak{m}}))$ is trivial for any positive integer *i*. Otherwise the group $H^i(K, \hat{E}(\overline{\mathfrak{m}}))$ is *p*-primary for any positive integer *i*.

Proof. Any *i*-th Galois cohomology group is torsion for any positive integer *i*. Take an integer *n* so that $p \nmid n$. Then the multiplication of $\hat{E}(\overline{\mathfrak{m}})$ by *n* is an isomorphism [Silverman 2009, IV.2.3(b)]. These facts show the lemma.

Lemma 6.10. The homomorphism ψ is the zero map.

Proof. By Lemma 6.9, we may assume that p > 0. Furthermore, we have only to show the image of any element of $H^1(K, E(k))$ of *p*-power order under ψ is equal to zero. If the special fiber of *E* is ordinary, then the statement follows from [Raynaud 1970, 9.4.1(iii)]. Otherwise, the group E(k) is *p*-torsion free. Thus, the group $H^1(K, E(k))$ is *p*-torsion free, which implies that the statement holds. \Box

From Lemma 6.10 we get a surjective homomorphism $\phi_E : H^1(K, E_K) \to$ Hom $(G_K, E(k))$. Put $\gamma_E := \phi_E([X_K])$. The group Im γ_E is finite and cyclic. The Galois extension L/K corresponding to Ker γ_E is the minimum separable field extension that splits the image of $[X_K]$ in $H^1(K, E(k))$. By D we denote the normalization of C in L. Put d := [L : K]. Then the normalization Y of $X \times_C D$ is a relatively minimal elliptic fibration over D with special fiber of type $_{m/d}I_0$, and the canonical projection $Y \to X$ is étale. In particular, the relation $\widetilde{K} \supset L$ holds. If $[X_K] \in \iota(H^1(G_K, \hat{E}(\overline{\mathfrak{m}})))$, then any element of Im $\tilde{\tau}$ fixes any closed point on E'. Thus, Lemma 6.8 gives the equality $\widetilde{K} = L$.

We summarize the above results:

Proposition 6.11. Assume that (X, C, f) is relatively minimal. By $_mT$ we denote the type of the special fiber of f. Take the maximal base field \tilde{K} of f (Definition 4.9). Then $[\tilde{K} : K]$ divides m (Proposition 4.15(3)). Moreover:

- (A) If $T \neq I_n$ $(n \ge 0)$, then X is simply connected and $\tilde{K} = K$.
- (M) If $T = I_n$ (n > 0), then \tilde{K}/K corresponds to Ker γ_M in Case (M) and $[\tilde{K}:K] = # \operatorname{Im} \gamma_M = m$.
- (E) If $T = I_0$, then \tilde{K}/K corresponds to Ker γ_E in Case (E), $[\tilde{K} : K] = \# \operatorname{Im} \gamma_E$, and one of the following statements holds: (1) p = 0 and $[\tilde{K} : K] = m$; (2) p > 0and $m/[\tilde{K} : K]$ is a power of p.

In particular, the extension \tilde{K}/K is finite and cyclic.

Lemma 6.12. Assume that (X, C, f) is relatively minimal. By m we denote the multiplicity of the special fiber X_k of f. We define a divisor F on X by $F := X_k/m$. By n we denote the order of the normal bundle of F in the Picard group Pic F. Then:

- (1) The \mathbb{O}_C -module $R^1 f_* \mathbb{O}_X$ is torsion-free if and only if the equality m = n holds.
- (2) If p = 0, then the equality m = n holds. Otherwise, there exists a nonnegative integer e such that the equality $m = np^e$ holds.
- (3) We use the notation introduced in Lemma 6.3. Take m' and n' for (X', C', f') in the same way. By d we denote the degree of u. Assume that F is isomorphic to an elliptic curve and p ∤ d. Then the equalities m = dm' and n = dn' hold.

Proof. Statements (1) and (2) follow from Proposition 1 in [Mitsui 2013]. Let us show Statement (3). Since u' is étale, the equality $u \circ f' = f \circ u'$ gives the equality m = dm'. By $\lambda : F' \to F$ we denote the base change of u' via the inclusion morphism $F \to X$. Since u' is a finite étale surjective morphism of degree d, the base change λ is a finite étale surjective morphism of degree d. Since F is isomorphic to an elliptic curve, the morphism λ may be regarded as a morphism between elliptic curves over k. The morphism λ induces a homomorphism $\lambda^* : \text{Pic } F \to \text{Pic } F'$. Since u' is étale, the divisor F' is equal to the pullback of the divisor F via u', which implies that $N_{F'/X} = \lambda^* N_{F/X}$. Since $p \nmid d$, the relation $p \nmid (n/n')$ holds. Thus, the equality m = dm' and Statement (2) give the equality n = dn'.

Lemma 6.13 [EGA III₁ 1961, 7.7.5(II), 7.8.4, and 7.9.4]. *The following conditions are equivalent:*

- (1) f is cohomologically flat in dimension zero [EGA III₂ 1963, 7.8.1]; *i.e.*, the formation of the direct image $f_* \mathbb{O}_X$ commutes with any base change.
- (2) $R^1 f_* \mathbb{O}_X$ is torsion-free.

Lemma 6.14. We use the notation introduced in Lemma 6.3. Suppose that the reduction of the special fiber of f is isomorphic to an elliptic curve and that $p \nmid \deg v$. Then $R^1 f_* \mathbb{O}_X$ is torsion-free if and only if $R^1 h_* \mathbb{O}_Y$ is torsion-free.

Proof. By [Raynaud 1970, 9.4.2] and Lemma 6.13, we may assume that ξ' is an isomorphism. Then the lemma follows from (1) and (3) of Lemma 6.12.

Proposition 6.15. Suppose that p > 0. Assume that (X, C, f) is relatively minimal. By $_mT$ we denote the type of the special fiber of f. Take the maximal base field \tilde{K} of f (Definition 4.9). By $(X_k)_{red}$ we denote the reduction of the special fiber of f. Then $p \nmid [\tilde{K} : K]$ if and only if one of the following conditions is satisfied:

- (1) $T \neq I_n \ (n \ge 0).$
- (2) $p \nmid m$.
- (3) $(X_k)_{red}$ is isomorphic to a supersingular elliptic curve.
- (4) $(X_k)_{red}$ is isomorphic to an ordinary elliptic curve and $R^1 f_* \mathbb{O}_X$ is torsion-free.

Proof. By (A) and (M) of Proposition 6.11, we have only to consider the case $T = I_0$. By Proposition 6.11(E), the extension \tilde{K}/K corresponds to Ker γ_E and the equality $[\tilde{K}:K] = \# \operatorname{Im} \gamma_E$ holds. If $(X_k)_{\text{red}}$ is isomorphic to a supersingular elliptic curve, then the group Hom $(G_K, E(k))$ is *p*-torsion free (Corollary 6.7). Thus, we may assume that $(X_k)_{\text{red}}$ is isomorphic to an ordinary elliptic curve. By Lemmas 6.3 and 6.14, we may assume that $[\tilde{K}:K]$ is a power of *p*. Then the proposition follows from [Raynaud 1970, 9.4.1(iii)] and Lemma 6.13.

Proposition 6.16. Let L/K be a finite cyclic extension in \overline{K} and (E', C, g') a relatively minimal elliptic fibration with section. By E'_k we denote the special fiber of g'. Then the following two conditions are equivalent:

- (1) There exists a relatively minimal elliptic fibration (X', C, f') satisfying the following conditions:
 - (a) The maximal base field of f' is equal to L (Definition 4.9).
 - (b) The Jacobian fibration of f' is given by g'.
- (2) *The following conditions are satisfied:*
 - (a) If E'_k is not of type I_n $(n \ge 0)$, then L = K.
 - (b) If p > 0 and E'_k is isomorphic to a supersingular elliptic curve, then p ↓ [L : K].

Proof. Propositions 6.11 and 6.15 show that Condition (1) implies Condition (2). Let us show the converse. We may assume that E'_k is of type I_n $(n \ge 0)$. Put d := [L : K] and $G_{L/K} := \text{Gal}(L/K)$. First, we consider the case n > 0. By assumption, there exists an element $\gamma_M \in \text{Hom}(G_{L/K}, \mathbb{Q}/\mathbb{Z}) \subset \text{Hom}(G_K, \mathbb{Q}/\mathbb{Z})$ of order d. Since ϕ_M is surjective, the case n > 0 follows from Proposition 6.11(M). Next, we consider the case n = 0. By assumption, there exists an element $\gamma_E \in \text{Hom}(G_{L/K}, E(k)) \subset \text{Hom}(G_K, E(k))$ of order d. Since ϕ_E is surjective, the case n = 0 follows from Proposition 6.11(E).

Proposition 6.17. Let (Y, C, h) be a relatively minimal elliptic fibration with special fiber of type I_n $(n \ge 0)$. Let $\xi : Y \to X$ be a finite étale surjective *C*-morphism of degree *d*. We regard the restriction $\xi_K : Y_K \to X_K$ of ξ to the generic fibers as a homomorphism between elliptic curves, which is determined by the choice of an element of Y(K). By *G* we denote the subgroup of Y(K) consisting of all *d*-torsion elements. Put $H := \xi_K(G)$. By \overline{G} and \overline{H} we denote the sets of the closures of all elements of *G* and *H* in *Y* and *X*, respectively. Assume that $p \nmid d$. Then:

- (1) $d \mid n, \#G = d^2$, and #H = d.
- (2) All elements of \overline{G} are disjoint.
- (3) All elements of \overline{H} are disjoint.
- (4) There exists an irreducible component of the special fiber of f that intersects with all elements of \overline{H} , and any other irreducible component of the special fiber of f is disjoint from all elements of \overline{H} .

Proof. Since ξ is étale, the relation $d \mid n$ holds and the special fiber of f is of type I_l , where we set l := n/d. We may regard the smooth loci \hat{X} and \hat{Y} of f and h as the Néron models of X_K and Y_K , respectively. By the Néron mapping property, the homomorphism ξ_K induces the unique C-homomorphism $\hat{\xi} : \hat{Y} \to \hat{X}$, which is the restriction of ξ to the smooth loci \hat{X} and \hat{Y} . The restriction $\hat{\xi}_k$ of $\hat{\xi}$ to the special fibers of \hat{X} and \hat{Y} is a finite étale surjective k-homomorphism of degree d. If n = 0, then $\hat{\xi}_k$ is a homomorphism between elliptic curves. If n > 0, then $\hat{\xi}_k$ is the homomorphism

$$\mathbb{G}_{m,k} \times (\mathbb{Z}/n\mathbb{Z}) \longrightarrow \mathbb{G}_{m,k} \times (\mathbb{Z}/l\mathbb{Z}), \quad (z, e \mod n) \longmapsto (z, e \mod l).$$

Thus, the closure of G in Y is finite and étale over C [Bosch et al. 1990, 7.3.2], which concludes the proof. \Box

6C. *Elliptic surfaces with prescribed orbifolds.* In this subsection, we use the following notation. Let *k* be an algebraically closed field of characteristic $p \ge 0$ and *C* a connected proper smooth *k*-curve with function field *K*. An elliptic fibration (X, C, f) is said to be *trivial* if there exists an elliptic curve *F* over *k* such that *X* is *C*-isomorphic to the *C*-scheme $F \times_k C$. Recall the following result on the global-to-local map:

Proposition 6.18 [Cossec and Dolgachev 1989, 5.4.6]. Let E_K be an elliptic curve over K. Take the minimal regular C-model (E, C, g) of E_K . Assume that the elliptic fibration (E, C, g) is nontrivial. For each closed point s on C, we put $K_s := \operatorname{Frac} \mathbb{O}_{C,s}^{\operatorname{sh}}$ and $E_{K_s} := E_K \times_K K_s$. Then the global-to-local map $H^1(K, E_K) \to \bigoplus_{s \in C} H^1(K_s, E_{K_s})$ is surjective.

Theorem 6.19. Let (C, B) be a connected cyclic orbifold k-curve and (E, C, g) a nontrivial relatively minimal elliptic fibration with section. We use the notation B_s/K_s introduced in Definition 3.6. Then the following two conditions are equivalent:

- (1) There exists a relatively minimal elliptic fibration (X, C, f) satisfying the following conditions:
 - (a) The orbifold associated to f is isomorphic to (C, B) (Definition 4.23).
 - (b) The Jacobian fibration of f is given by g.
- (2) The following conditions are satisfied for any closed point s on C:
 - (a) If $g^{-1}(s)$ is not of type I_n $(n \ge 0)$, then $B_s = K_s$.
 - (b) If p > 0 and g⁻¹(s) is isomorphic to a supersingular elliptic curve, then p ↓ [B_s : K_s].

Proof. The theorem follows from Propositions 6.16 and 6.18.

Proposition 6.20. Let (X, C, f) be a relatively minimal elliptic fibration with Jacobian fibration (E, C, g). Then the following conditions are equivalent:

- (1) $\chi(\mathbb{O}_X) \leq 0.$
- (2) $\chi(\mathbb{O}_X) = 0.$
- (3) The reduction of any closed fiber of f is isomorphic to an elliptic curve.
- (4) g is smooth.

Proof. The equivalence of (1) and (2) follows from Proposition 2 in [Mitsui 2014]. The equivalence of (2) and (3) follows from Proposition 2 in [Mitsui 2013]. The equivalence of (3) and (4) follows from Proposition 6.4. \Box

Corollary 6.21. Let (C, B) be a connected cyclic orbifold k-curve. Then there exists a relatively minimal elliptic fibration (X, C, f) with $\chi(\mathbb{O}_X) > 0$ such that the orbifold associated to f is isomorphic to (C, B) (Definition 4.23).

Proof. Take a relatively minimal elliptic fibration (E, C, g) satisfying the following conditions: (1) g is not smooth; (2) g admits a section; (3) for any $s \in \text{Supp } B$, the closed fiber $g^{-1}(s)$ is isomorphic to an ordinary elliptic curve. Since the elliptic fibration (E, C, g) is nontrivial, Theorem 6.19 gives a relatively minimal elliptic fibration (X, C, f) such that the orbifold associated to f is isomorphic to (C, B). Proposition 6.20 shows that the inequality $\chi(\mathbb{O}_X) > 0$ holds.

6D. *Fundamental groups of elliptic surfaces.* In this subsection, we use the following notation. Let k be an algebraically closed field of characteristic $p \ge 0$, C a connected proper smooth k-curve with function field K, and (X, C, f) an elliptic fibration. We denote the intersection number of divisors D_1 and D_2 on X by $D_1 \cdot D_2$.

Lemma 6.22. Assume that two sections D_1 and D_2 of f satisfy the following:

- (1) $\mathbb{O}_X(D_1 D_2)|_{X_K}$ is torsion in $\operatorname{Pic}(X_K)$, where X_K is the generic fiber of f.
- (2) For any closed point x on C, there exists an irreducible component of the fiber $f^{-1}(x)$ that intersects with both of D_1 and D_2 , and any other irreducible component of the fiber $f^{-1}(x)$ is disjoint from both of D_1 and D_2 .

Then the equality $D_1 \cdot D_2 = -\chi(\mathbb{O}_X)$ holds.

Proof. First, we assume that $D_1 = D_2$. Put $D := D_1 = D_2$ and $\mathcal{F} := \mathbb{O}_X(D)/\mathbb{O}_X$. Since the genus of X_K is equal to one and the effective divisor $D|_{X_K}$ on X_K is of degree one, the long exact sequence induced by the functor f_* and the exact sequence of \mathbb{O}_X -modules

$$0 \longrightarrow \mathbb{O}_X \longrightarrow \mathbb{O}_X(D) \longrightarrow \mathcal{F} \longrightarrow 0$$

gives an isomorphism $f_* \mathscr{F} \to R^1 f_* \mathscr{O}_X$. In particular, the equalities

 $D \cdot D = \deg \mathbb{O}_X(D)|_D = \deg f_* \mathcal{F} = \deg R^1 f_* \mathbb{O}_X$

hold. The Riemann–Roch theorem for the line bundle $R^1 f_* \mathbb{O}_X$ on *C* and the Leray spectral sequence for *f* give the equalities

$$\deg R^1 f_* \mathbb{O}_X = \chi(R^1 f_* \mathbb{O}_X) - \chi(\mathbb{O}_C) = -\chi(\mathbb{O}_X).$$

Thus, the equality $D \cdot D = -\chi(\mathbb{O}_X)$ holds.

Next, we consider the general case. By *n* we denote the order of $\mathbb{O}_X(D_1-D_2)|_{X_K}$ in Pic(X_K) (Condition (1)). Then $n(D_1 - D_2)$ is linearly equivalent to a vertical divisor *V*. Condition (2) gives the equality $D_0 \cdot (D_1 - D_2) = 0$ for any vertical prime divisor D_0 . Thus, the equality $V \cdot V = 0$ holds, which gives the equality $(D_1 - D_2) \cdot (D_1 - D_2) = 0$. Therefore, the first case shows the general case. \Box

Theorem 6.23. Choose a smooth closed fiber $i : X_0 \to X$ of f. Take a geometric point \bar{x}_0 on X_0 . Put $\bar{x} := i(\bar{x}_0)$ and $\bar{s} := f(\bar{x})$. By (C, B) we denote the connected proper orbifold k-curve associated to f (Definition 4.23). The morphisms i and f induce canonical homomorphisms $i_* : \pi_1(X_0, \bar{x}_0) \to \pi_1(X, \bar{x})$ and $f_*^{\text{orb}} : \pi_1(X, \bar{x}) \to \pi_1(C, B, \bar{s})$, respectively (Theorem 1.1). Then:

- (1) If $\chi(\mathbb{O}_X) > 0$, then i_* is trivial and f_*^{orb} is an isomorphism.
- (2) If $\chi(\mathbb{O}_X) = 0$, then i_* is injective.

Remark 6.24. In the complex analytic case, Statement (2) does not hold in general for topological fundamental groups although Statement (1) holds and i_* is nontrivial whenever $\chi(\mathbb{O}_X) = 0$ [Friedman and Morgan 1994, 2.2.1 and 2.7.2]. For example, if X is a Hopf surface, then $\chi(\mathbb{O}_X) = 0$, Ker $i_* \cong \mathbb{Z}$, Coker $i_* = 0$, and $\pi_1(X) \cong \text{Im } i_* \cong \mathbb{Z} \oplus \mathbb{Z}/n\mathbb{Z}$ for some positive integer n.

Proof. By (E, C, g) we denote the Jacobian fibration of f. First, let us show Statement (1). We have only to show the following: for any connected étale covering space $\xi : Y \to X$, any connected component of $\xi^{-1}(X_0)$ is X_0 -isomorphic to X_0 . Assume that the above statement does not hold. Choose ξ that does not satisfy the above statement. Take the Stein factorization

$$Y \xrightarrow{h} C' \xrightarrow{u} C$$

of $f \circ \xi : Y \to X \to C$. Take the elliptic fibration (X', C', f') and the étale morphisms $\xi' : Y \to X'$ and $u' : X' \to X$ given by Lemma 6.3. By assumption, the morphism ξ' is not an isomorphism. Replacing ξ by ξ' , we may assume that $(Y, C, f \circ \xi)$ is an elliptic fibration.

Since ξ is étale, any closed fiber is of type ${}_{m}I_{n}$ $(n \ge 0)$. Since $\chi(\mathbb{O}_{X}) > 0$, Proposition 6.20 shows that f admits a closed fiber of type ${}_{m}I_{n}$ (n > 0). In particular, the j-invariant of g is nonconstant (Proposition 6.4). By d we denote the degree of ξ . If p > 0, then $p \nmid d$ since f admits a closed fiber that is isomorphic to a supersingular elliptic curve (Corollary 6.7) and ξ is étale. Choose a connected proper smooth k-curve C' and a finite morphism $u : C' \to C$ satisfying the following condition: the morphism u induces an extension of the function fields K'/K; by J_K we denote the Jacobian of the generic fiber of Y/C; then $Y(K') \neq \emptyset$ and $J_K(K')$ contains $d^2 d$ -torsion elements. Take a desingularization X' of $X \times_C C'$ and the canonical projection $u' : X' \to X$. Since ξ' is étale, the base change $\xi' : Y' \to X'$ of ξ via u' is étale. Thus, we obtain an étale C'-morphism $\xi'' : Y'' \to X''$ between the minimal regular C'-models of Y' and X' after successive blowing-downs of (-1)-curves on Y' and X'. Replacing ξ by ξ'' , we may assume that ξ is a morphism between Jacobian fibrations and Y(K) contains $d^2 d$ -torsion elements.

By *H* we denote the image of the *d*-torsion elements of Y(K) under ξ . By \overline{H} we denote the set of the closures of all elements of *H* in *X*. Proposition 6.17 shows the following: (a) #H = d > 1; (b) all elements of \overline{H} are disjoint; (c) for any closed point *x* on *C*, there exists an irreducible component of the fiber $f^{-1}(x)$ that intersects with all elements of \overline{H} , and any other irreducible component of the fiber $f^{-1}(x)$ that $f^{-1}(x)$ is disjoint from all elements of \overline{H} . This contradicts Lemma 6.22 since $\chi(\mathbb{O}_X) > 0$. Therefore the homomorphism i_* is trivial.

Next, let us show Statement (2). We have only to show that, for any connected étale covering space $\tau : X'_0 \to X_0$, there exists an étale covering $\xi : Y \to X$ such that

any connected component of the preimage $\xi^{-1}(X_0)$ is X_0 -isomorphic to X'_0 . Since $\chi(\mathbb{O}_X) = 0$, Proposition 6.20 shows that g is smooth. Take an integer $n \ge 3$ so that $p \nmid n$. Since the *n*-torsion *C*-subgroup scheme of E/C is finite and étale, we may take a connected étale covering space $u : C' \to C$ satisfying the following condition: let $u' : E' \to E$ denote the base change of u via g; then E'(C') contains n^2 *n*-torsion elements. Since g is smooth, the *j*-invariant of E is contained in k. Thus, the elliptic curve E' over C' induces a constant morphism from C' to the moduli scheme of elliptic curves with level n. Therefore, we obtain a C'-isomorphism $E' \cong X_0 \times_k C'$. Take the base change $\tau' : E'' \to E'$ of τ via the structure morphism $C' \to \text{Spec } k$. Take the *C*-morphism $h : X \to E$ given by Lemma 6.6. Then the base change of $u' \circ \tau'$ via h is the desired morphism ξ .

Lemma 6.25. Let R be a strictly Henselian excellent discrete valuation ring of equicharacteristic. By \hat{R} we denote the completion of R with respect to the maximal ideal. Put K := Frac R and $\hat{K} := \text{Frac } \hat{R}$. Let \hat{L}/\hat{K} be a finite Galois extension. Then there exists a unique extension L/K in \hat{L} such that $\hat{L} = \hat{K}L$. Furthermore, the extension L/K is Galois, of degree $[\hat{L} : \hat{K}]$, and linearly disjoint from \hat{K}/K .

Proof. Since *R* is algebraically closed in \hat{R} by the approximation property [Bosch et al. 1990, 3.6.9], we have only to show the existence of *L*. We denote the characteristic of *R* by *l*. Put $d := [\hat{L} : \hat{K}]$. By assumption, the extension \hat{L}/\hat{K} is solvable. Thus, by induction on *d*, we may assume that $l \nmid d$ or d = l > 0. The case $l \nmid d$ follows from Kummer theory since *R* contains a primitive *d*-th root of unity. The case d = l > 0 follows from Lemma 5.9(2) and Artin–Schreier theory.

Finally, we give a proof of the criterion for simple-connectedness of elliptic surfaces:

Proof of Theorem 1.2. We use the notation introduced in Theorem 6.23. Theorems 1.1 and 6.23 show that $\pi_1(X)$ is trivial if and only if $\pi_1(C, B)$ is trivial and $\chi(\mathbb{O}_X) > 0$. Proposition 6.11 shows that the orbifold (C, B) is cyclic. Thus, Theorem 1.3 shows that $\pi_1(C, B)$ is trivial if and only if $C \cong \mathbb{P}^1_k$, $B^t = B$, $\# \operatorname{Supp} B \leq 2$, and $\gcd(n_s, n_t) = 1$ for $s \neq t$, where we put $n_s := [B_s : K_s]$ for each $s \in \operatorname{Supp} B$. Lemma 6.25 and Propositions 4.6, 6.11, and 6.15 imply that the above conditions on (C, B) are equivalent to Conditions (2)–(6).

Let us show that each of Conditions (1)–(6) is necessary. We remark that $\chi(\mathbb{O}_X) > 0$ if and only if the Jacobian fibration of f is not smooth (Proposition 6.20). The necessity of Conditions (1) and (2) is clear. The necessity of Conditions (3) and (4) follows from Corollary 6.21. The necessity of Conditions (5) and (6) follows from Proposition 6.15 and Theorem 6.19.

Kentaro Mitsui

Appendix A: Triangle groups and projective special linear groups

The result of this section is used in the proof of Proposition 5.4. Let *a*, *b*, and *c* be integers greater than 1. We define the *triangle group* $\Delta(a, b, c)$ by

$$\Delta(a, b, c) := \langle x, y, z \mid x^a = y^b = z^c = xyz = \mathrm{id} \rangle.$$

Let *p* be a prime number and *q* a power of *p*. In this section, we study homomorphisms $\Delta(a, b, c) \rightarrow SL(2, \mathbb{F}_q)$ and $\Delta(a, b, c) \rightarrow PSL(2, \mathbb{F}_q)$ that preserve the orders of *x*, *y*, and *z*. Take an algebraic closure $\overline{\mathbb{F}_q}$ of \mathbb{F}_q . For each positive integer *n* prime to *p*, we take a primitive *n*-th root of unity ζ_n in $\overline{\mathbb{F}_q}$. Put $\mu_n := \zeta_n + \zeta_n^{-1}$. The proofs of Lemmas A.1–A.4 are straightforward.

Lemma A.1. $\mu_n \in \mathbb{F}_q$ if and only if $n \mid (q^2 - 1)$.

Lemma A.2. We have equalities $\# SL(2, \mathbb{F}_q) = q(q^2 - 1)$ and $\# PSL(2, \mathbb{F}_q) = q(q^2 - 1)/\gcd(2, p - 1)$. The projection $SL(2, \mathbb{F}_q) \rightarrow PSL(2, \mathbb{F}_q)$ maps any element of order *n* to an element of order *n*/gcd(2, *p* - 1, *n*).

Lemma A.3. Take $X \in SL(2, \mathbb{F}_q)$. Then the image of X in $SL(2, \overline{\mathbb{F}_q})$ is conjugate to

$$\begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix}$$
, $\begin{pmatrix} -1 & 1 \\ 0 & -1 \end{pmatrix}$, or $\begin{pmatrix} \alpha & 0 \\ 0 & \alpha^{-1} \end{pmatrix}$,

where $\alpha \in \overline{\mathbb{F}_q}^{\times}$. By *n* we denote the order of α in $\overline{\mathbb{F}_q}^{\times}$. If $p \neq 2$, then the order of *X* is equal to *p*, 2*p*, or *n*, respectively. Otherwise, the order of *X* is equal to 2, 2, or *n*, respectively. The order of the image of *X* in PSL(2, \mathbb{F}_q) is equal to *p*, *p*, or *n*/gcd(2, *p* - 1, *n*), respectively.

Lemma A.4. Take $X \in SL(2, \mathbb{F}_q)$. Let *n* be an integer prime to *p*. Assume that n > 2 and tr $X = \mu_n$. Then the order of X is equal to *n*.

Lemma A.5. Assume that a, b, and c are greater than 2 and divide $q^2 - 1$. Then there exist X, Y, and Z in $SL(2, \mathbb{F}_q)$, of orders a, b, and c, respectively, such that XYZ = I, where I is the identity matrix of $SL(2, \mathbb{F}_q)$.

Proof. Lemma A.1 shows that μ_a , μ_b , and μ_c are contained in \mathbb{F}_q . Put

$$X := \begin{pmatrix} 0 & -1 \\ 1 & \mu_a \end{pmatrix} \in \mathrm{SL}(2, \mathbb{F}_q).$$

We have only to construct Y and Z in SL(2, \mathbb{F}_q) so that tr $Y = \mu_b$, tr $Z = \mu_c$, and XYZ = 1 (Lemma A.4). We write

$$Y = \begin{pmatrix} \alpha & \beta \\ \gamma & \delta \end{pmatrix}.$$

Then we have only to choose α , β , γ , and δ in \mathbb{F}_q so that $\alpha\delta - \beta\gamma = 1$, $\alpha + \delta = \mu_b$, and $\beta - \gamma + \mu_a \delta = \mu_c$. Thus, we have only to show that there exists a solution $(\alpha, \beta) \in \mathbb{F}_q \times \mathbb{F}_q$ of the equation $F(\alpha, \beta) = 0$, where we put

$$F(u, v) := u^{2} - \mu_{a}uv + v^{2} - \mu_{b}u + (\mu_{a}\mu_{b} - \mu_{c})v + 1.$$

If elements α_0 , α_1 , β_0 , and β_1 in \mathbb{F}_q satisfy $\alpha = \alpha_0 + \alpha_1$ and $\beta = \beta_0 + \beta_1$, then the equality

$$F(\alpha,\beta) = F(\alpha_0,\beta_0) + \frac{\partial F}{\partial u}(\alpha_0,\beta_0)\alpha_1 + \frac{\partial F}{\partial v}(\alpha_0,\beta_0)\beta_1 + \alpha_1^2 - \mu_a \alpha_1 \beta_1 + \beta_1^2$$

holds. Note that the equalities

$$\begin{cases} (\partial F/\partial u)(u,v) = 2u - \mu_a v - \mu_b, \\ (\partial F/\partial v)(u,v) = -\mu_a u + 2v + \mu_a \mu_b - \mu_c \end{cases}$$

hold. Since a > 2 and gcd(p, a) = 1, the inequality $\mu_a^2 \neq 4$ holds. Thus, we may take $(\alpha_0, \beta_0) \in \mathbb{F}_q \times \mathbb{F}_q$ so that $(\partial F/\partial u)(\alpha_0, \beta_0) = (\partial F/\partial v)(\alpha_0, \beta_0) = 0$. Therefore, we have only to show that there exists a solution $(\alpha_1, \beta_1) \in \mathbb{F}_q \times \mathbb{F}_q$ of the equation $G(\alpha_1, \beta_1) = -F(\alpha_0, \beta_0)$, where we put $G(u, v) := u^2 - \mu_a uv + v^2$. If $p \neq 2$ (resp. p = 2), then the quadratic form G(u, v) is nondegenerate (resp. nondefective), which concludes the proof.

Lemma A.6. Assume that the following conditions are satisfied:

- (1) If $p \neq 2$, then a = p or 2p.
- (2) If p = 2, then a = 2.
- (3) *b* and *c* are greater than 2 and divide $q^2 1$.

Then there exist X, Y, and Z in $SL(2, \mathbb{F}_q)$ of orders a, b, and c, respectively, such that XYZ = I, where I is the identity matrix of $SL(2, \mathbb{F}_q)$.

Proof. Lemma A.1 shows that μ_b and μ_c are contained in \mathbb{F}_q . First, we consider the case a = p. We define X, Y, and Z in SL(2, \mathbb{F}_q) by

$$X := \begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix}, \quad Y := \begin{pmatrix} \zeta_b & 0 \\ \mu_c - \mu_b & \zeta_b^{-1} \end{pmatrix}, \quad \text{and} \quad Z := \begin{pmatrix} \zeta_b^{-1} & -\zeta_b^{-1} \\ \mu_b - \mu_c & \mu_c - \zeta_b^{-1} \end{pmatrix}.$$

Then the order of X is equal to p. Moreover, the equalities tr $Y = \mu_b$, tr $Z = \mu_c$, and XYZ = I hold. Thus, the elements X, Y, and Z satisfy the desired conditions (Lemma A.4). Next, we consider the case $p \neq 2$ and a = 2p. In that case, we have only to replace X, Y, and Z by

$$X := \begin{pmatrix} -1 & 1 \\ 0 & -1 \end{pmatrix}, \quad Y := \begin{pmatrix} \zeta_b & 0 \\ \mu_b + \mu_c & \zeta_b^{-1} \end{pmatrix}, \quad \text{and} \quad Z := \begin{pmatrix} -\zeta_b^{-1} & -\zeta_b^{-1} \\ \mu_b + \mu_c & \zeta_b^{-1} + \mu_c \end{pmatrix}. \quad \Box$$

Proposition A.7. Assume that a, b, and c are pairwise coprime. Take the elements x, y, and z of $\Delta(a, b, c)$ in the definition of $\Delta(a, b, c)$. Put $G := SL(2, \mathbb{F}_q)$ or $PSL(2, \mathbb{F}_q)$. Then there exists a homomorphism $\phi : \Delta(a, b, c) \rightarrow G$ such that the orders of $\phi(x), \phi(y), and \phi(z)$ are equal to a, b, and c, respectively, if and only if the following three conditions are satisfied:

- (1) abc | #G.
- (2) If $G = SL(2, \mathbb{F}_q)$ and one of a, b, and c is equal to 2, then p = 2.
- (3) If an integer u is equal to a, b, or c and is divisible by p, then u satisfies one of the following conditions: (a) u = p: (b) $G = SL(2, \mathbb{F}_q), p \neq 2$, and u = 2p.

Proof. First, let us show the "only if" part. Since *a*, *b*, and *c* are pairwise coprime, the condition on ϕ implies Condition (1). Lemma A.3 and the condition on ϕ imply Condition (3). Assume that $G = SL(2, \mathbb{F}_q)$, a = 2, and $p \neq 2$. Then $\phi(x) = -I$, where *I* is the identity matrix of $SL(2, \mathbb{F}_q)$. Thus, the equality $\phi(y) = -\phi(z)$ holds, which contradicts the assumption that *b* is prime to *c*. Therefore, Condition (2) holds. The "if" part follows from Lemmas A.2, A.5, and A.6.

Lemma A.8. Assume that a, b, and c are pairwise coprime. Let G be a nontrivial finite group. If there exists a surjective homomorphism $\Delta(a, b, c) \rightarrow G$, then G is nonsolvable.

Proof. Assume that *G* is solvable. Then there exists a nontrivial cyclic group *H* and a surjective homomorphism $\phi : \Delta(a, b, c) \to H$. We may write $H = \mathbb{Z}/n\mathbb{Z}$, where *n* is an integer greater than 1. Take the elements *x*, *y*, and *z* of $\Delta(a, b, c)$ in the definition of $\Delta(a, b, c)$. The orders of *x*, *y*, and *z* are equal to *a*, *b*, and *c*, respectively. By \overline{a} , \overline{b} , and \overline{c} we denote the orders of $\phi(x)$, $\phi(y)$, and $\phi(z)$, respectively. Then $\overline{a} \mid a, \overline{b} \mid b, \overline{c} \mid c$ and $\overline{abc} \neq 1$. In particular, the integers $\overline{a}, \overline{b}$, and \overline{c} are pairwise coprime. Since xyz = id, the equality $\phi(x) + \phi(y) + \phi(z) = 0$ holds, which contradicts the facts that $\overline{a}, \overline{b}$, and \overline{c} are pairwise coprime and $\overline{abc} \neq 1$. Therefore, the group *G* is nonsolvable.

Theorem A.9. Assume that a, b, and c are pairwise coprime. If $\{a, b, c\} = \{2, 3, 5\}$, then we suppose that p = 5. Otherwise, we suppose that $2abc | (p^2 - 1)$. Take the elements x, y, and z of $\Delta(a, b, c)$ in the definition of $\Delta(a, b, c)$. Then there exists a surjective homomorphism $\phi : \Delta(a, b, c) \rightarrow PSL(2, \mathbb{F}_p)$ such that the orders of $\phi(x)$, $\phi(y)$, and $\phi(z)$ are equal to a, b, and c, respectively. Furthermore, there exists a prime number p such that $2abc | (p^2 - 1)$.

Remark A.10. If q > 3, then the group $PSL(2, \mathbb{F}_q)$ is noncommutative and simple.

Proof. Let us show the first statement. By Lemma A.2, we may take ϕ in Proposition A.7. We have only to show that ϕ is surjective. Lemma A.8 shows

that the image of ϕ is nonsolvable. Thus, the case $\{a, b, c\} = \{2, 3, 5\}$ follows from the fact that any proper subgroup of PSL $(2, \mathbb{F}_5) \cong A_5$ is solvable. Assume that $\{a, b, c\} \neq \{2, 3, 5\}$. By the classification of the subgroups of PSL $(2, \mathbb{F}_p)$ [Dickson 1958, p. 285, XII, 260], any nonsolvable subgroup of PSL $(2, \mathbb{F}_p)$ is isomorphic to PSL $(2, \mathbb{F}_5)$ or PSL $(2, \mathbb{F}_p)$. Suppose that ϕ is not surjective. Then the image of ϕ is isomorphic to PSL $(2, \mathbb{F}_5)$. Since the order of any nontrivial element in PSL $(2, \mathbb{F}_5) \cong A_5$) is equal to 2, 3 or 5, the equality $\{a, b, c\} = \{2, 3, 5\}$ holds, which contradicts the assumption. Thus, the homomorphism ϕ is surjective. The last statement follows from Dirichlet's theorem on arithmetic progressions.

Appendix B: Comparison between orbifolds and stacks

Let *k* be an algebraically closed field of characteristic $p \ge 0$. For a nonnegative integer *n*, a DM stack \mathcal{G} is said to be *of dimension n* if an atlas of \mathcal{G} is of dimension *n*. A *stack orbifold* \mathcal{G} is a locally Noetherian normal DM stack that admits an open dense substack that is isomorphic to a scheme. A *stack orbifold k*-curve \mathcal{G} is a separated smooth *k*-stack of dimension one that is a stack orbifold. In the following, we see that the notion of an orbifold *k*-curve coincides with the notion of a stack orbifold *k*-curve.

We construct a stack orbifold k-curve from an orbifold k-curve. Let (S, B) be a connected orbifold k-curve. Put $S_0 := S \setminus \text{Supp } B$. Take $s \in S$. Choose an affine open subset U containing s. By $(U, B|_U)$ we denote the orbifold obtained by restricting (S, B) to U. We may take a Galois orbifold trivialization $U' \rightarrow (U, B|_U)$ of $(U, B|_U)$ (Theorem 1.3). We may regard the open subscheme $U \cap S_0$ of U as an open substack of the quotient stack [U'/G]. Pasting [U'/G] for all $s \in S$, we obtain a stack orbifold k-curve with coarse moduli space S.

We construct an orbifold k-curve from a stack orbifold k-curve. Let \mathscr{G} be a connected stack orbifold k-curve. Take an open dense substack \mathscr{G}_0 of \mathscr{G} that is isomorphic to a scheme S_0 . Take the coarse moduli space $\lambda : \mathscr{G} \to S$ of \mathscr{G} [Rydh 2013]. We may regard S_0 as an open subscheme of S. Thus, the equality deg $\lambda = 1$ holds [Vistoli 1989, 1.15]. Therefore, the scheme S is of finite type over k. Since \mathscr{G} is connected and normal, the scheme S is connected and normal. Since k is perfect, the scheme S is a connected smooth k-curve.

By P(S) we denote the set of all closed points on S. Take $s \in P(S)$. By Aut(s) we denote the automorphism group of s. If $s \in S_0$, then Aut(s) is trivial. By S_s and \mathcal{G}_s we denote the schemes obtained by the strict Henselizations of S and \mathcal{G} at s, respectively [Laumon and Moret-Bailly 2000, 6.2.1]. Take the quotient stack $\phi_s : \mathcal{G}_s \to T_s := [\mathcal{G}_s / \operatorname{Aut}(s)]$ and the canonical morphism $\mu_s : T_s \to \mathcal{G}$ [Laumon and Moret-Bailly 2000, 6.2.1]. By B_s , L_s , and K_s , we denote the fields of rational functions of \mathcal{G}_s , T_s , and S_s , respectively [Vistoli 1989, 1.14]. Since the composite

 $\lambda \circ \mu_s : T_s \to S$ induces an isomorphism $K_s \to L_s$, the composite $\lambda \circ \mu_s \circ \phi_s : \mathscr{G}_s \to S$ induces a finite Galois extension B_s/K_s with Galois group Aut(s).

Take a separable closure $\overline{K_s}$ of K_s . We embed B_s in $\overline{K_s}$ over K_s . Since B_s/K_s is Galois, the image does not depend on the choice of the embedding. We define a map B on P(S) by $s \mapsto B_s/K_s$ (Definition 3.6). Then Supp B is locally finite since \mathcal{G}_0 is open dense in \mathcal{G} . Thus, the pair (S, B) is a connected orbifold k-curve.

Theorem B.1. The above two correspondences give an equivalence between the category of orbifold k-curves and orbifold (étale) k-morphisms and the category of stack orbifold k-curves and (étale) k-morphisms that induce qsc morphisms between coarse moduli spaces. In particular, the fundamental group of any orbifold k-curve coincides with the fundamental group of the corresponding stack orbifold k-curve in [Noohi 2004, §4].

Remark B.2. The theorem does not hold for general orbifolds. There exists an orbifold étale covering space of a trivial orbifold (S, B) that is not an étale covering space of the scheme S (Examples 3.24 and 4.16).

Theorem B.1 follows from the local structure theorem on DM stacks [Laumon and Moret-Bailly 2000, 6.2]. The detail of the proof of the theorem is left to the readers.

Acknowledgments

The author thanks the referee for helpful comments. He is grateful to Professor Yuichiro Hoshi, who suggested that we use normalization instead of Stein factorization in studies on homotopy exact sequences induced by nonproper morphisms. This work was supported by the Grant-in-Aid for JSPS (the Japan Society for the Promotion of Science) Fellows (24-1432), the Grant-in-Aid for Young Scientists (B) (25800018) from the JSPS, and the Grant-in-Aid for Scientific Research (S) (24224001) from the JSPS.

References

- [Bosch et al. 1990] S. Bosch, W. Lütkebohmert, and M. Raynaud, *Néron models*, Ergebnisse der Mathematik und ihrer Grenzgebiete (3) **21**, Springer, Berlin, 1990. MR 91i:14034 Zbl 0705.14001
- [Chau 1983] T. C. Chau, "A note concerning Fox's paper on Fenchel's conjecture", *Proc. Amer. Math. Soc.* **88**:4 (1983), 584–586. MR 84m:20038 Zbl 0497.20035
- [Cossec and Dolgachev 1989] F. R. Cossec and I. V. Dolgachev, *Enriques surfaces, I*, Progress in Mathematics **76**, Birkhäuser, Boston, 1989. MR 90h:14052 Zbl 0665.14017
- [Dickson 1958] L. E. Dickson, *Linear groups: With an exposition of the Galois field theory*, Dover, New York, 1958. MR 21 #3488 Zbl 0082.24901
- [EGA III₁ 1961] A. Grothendieck, "Eléments de géométrie algébrique, III: Étude cohomologique des faisceaux cohérents, I", *Inst. Hautes Études Sci. Publ. Math.* **11** (1961), 5–167. MR 0163910 Zbl 0118.36206

- [EGA III₂ 1963] A. Grothendieck, "Eléments de géométrie algébrique, III: Étude cohomologique des faisceaux cohérents, II", *Inst. Hautes Études Sci. Publ. Math.* **17** (1963), 5–91. MR 0163911 Zbl 0122.16102
- [EGA IV₂ 1965] A. Grothendieck, "Eléments de géométrie algébrique, IV: Étude locale des schémas et des morphismes de schémas, II", *Inst. Hautes Études Sci. Publ. Math.* 24 (1965), 5–231. MR 0199181 Zbl 0135.39701
- [EGA IV₃ 1966] A. Grothendieck, "Eléments de géométrie algébrique, IV: Étude locale des schémas et des morphismes de schémas, III", *Inst. Hautes Études Sci. Publ. Math.* 28 (1966), 5–255. MR 0217086 Zbl 0144.19904
- [Fox 1952] R. H. Fox, "On Fenchel's conjecture about *F*-groups", *Mat. Tidsskr. B.* **1952** (1952), 61–65. MR 14,843c Zbl 0049.15404
- [Friedman and Morgan 1994] R. Friedman and J. W. Morgan, Smooth four-manifolds and complex surfaces, Ergebnisse der Mathematik und ihrer Grenzgebiete (3) 27, Springer, Berlin, 1994. MR 95m:57046 Zbl 0817.14017
- [Iitaka 1971] S. Iitaka, "Deformations of compact complex surfaces, III", *J. Math. Soc. Japan* **23** (1971), 692–705. MR 44 #7598 Zbl 0219.32012
- [Katsura and Ueno 1985] T. Katsura and K. Ueno, "On elliptic surfaces in characteristic *p*", *Math. Ann.* **272**:3 (1985), 291–330. MR 87g:14040 Zbl 0553.14019
- [Katsura and Ueno 1986] T. Katsura and K. Ueno, "Multiple singular fibres of type G_a of elliptic surfaces in characteristic *p*", pp. 405–429 in *Algebraic and topological theories* (Kinosaki, 1984), edited by M. Nagata et al., Kinokuniya, Tokyo, 1986. MR 1102270 Zbl 0800.14019
- [Kodaira 1963] K. Kodaira, "On compact analytic surfaces, II", *Ann. of Math.* (2) **77** (1963), 563–626. MR 32 #1730 Zbl 0118.15802
- [Kodaira 1964] K. Kodaira, "On the structure of compact complex analytic surfaces, I", *Amer. J. Math.* **86** (1964), 751–798. MR 32 #4708 Zbl 0137.17501
- [Lang 1986] W. E. Lang, "An analogue of the logarithmic transform in characteristic *p*", pp. 337–340 in *Proceedings of the 1984 Vancouver conference in algebraic geometry*, edited by J. Carrell et al., CMS Conf. Proc. **6**, Amer. Math. Soc., Providence, RI, 1986. MR 846026 Zbl 0601.14033
- [Laumon and Moret-Bailly 2000] G. Laumon and L. Moret-Bailly, *Champs algébriques*, Ergebnisse der Mathematik und ihrer Grenzgebiete (3) **39**, Springer, Berlin, 2000. MR 2001f:14006 Zbl 0945.14005
- [Lieblich and Olsson 2010] M. Lieblich and M. Olsson, "Generators and relations for the étale fundamental group", *Pure Appl. Math. Q.* 6:1 (2010), 209–243. MR 2011b:14055 Zbl 05679716
- [Liu 2002] Q. Liu, *Algebraic geometry and arithmetic curves*, Oxford Graduate Texts in Mathematics **6**, Oxford University Press, 2002. MR 2003g:14001 Zbl 0996.14005
- [Liu et al. 2004] Q. Liu, D. Lorenzini, and M. Raynaud, "Néron models, Lie algebras, and reduction of curves of genus one", *Invent. Math.* 157:3 (2004), 455–518. MR 2005m:14039 Zbl 1060.14037
- [Matsumura 1989] H. Matsumura, *Commutative ring theory*, 2nd ed., Cambridge Studies in Advanced Mathematics **8**, Cambridge University Press, 1989. MR 90i:13001 Zbl 0666.13002
- [Mitsui 2013] K. Mitsui, "Logarithmic transformations of rigid analytic elliptic surfaces", *Math. Ann.* **355**:3 (2013), 1123–1170. MR 3020156 Zbl 1285.14028
- [Mitsui 2014] K. Mitsui, "On a question of Zariski on Zariski surfaces", *Math. Z.* **276**:1–2 (2014), 237–242. MR 3150201 Zbl 1284.14048
- [Moishezon 1977] B. Moishezon, *Complex surfaces and connected sums of complex projective planes*, Lecture Notes in Mathematics **603**, Springer, Berlin-New York, 1977. MR 58 #10931 Zbl 0392.32015

- [Noohi 2004] B. Noohi, "Fundamental groups of algebraic stacks", *J. Inst. Math. Jussieu* **3**:1 (2004), 69–103. MR 2004k:14003 Zbl 1052.14001
- [Raynaud 1970] M. Raynaud, "Spécialisation du foncteur de Picard", Inst. Hautes Études Sci. Publ. Math. 38 (1970), 27–76. MR 44 #227 Zbl 0207.51602
- [Raynaud 1990] M. Raynaud, "p-groupes et réduction semi-stable des courbes", pp. 179–197 in The Grothendieck Festschrift, III, edited by P. Cartier et al., Progress in Mathematics 88, Birkhäuser, Boston, 1990. MR 92m:14025 Zbl 0722.14013
- [Raynaud 1994] M. Raynaud, "Revêtements de la droite affine en caractéristique p > 0 et conjecture d'Abhyankar", *Invent. Math.* **116**:1–3 (1994), 425–462. MR 94m:14034 Zbl 0798.14013
- [Rydh 2013] D. Rydh, "Existence and properties of geometric quotients", *J. Algebraic Geom.* **22**:4 (2013), 629–669. MR 3084720 Zbl 1278.14003
- [Seifert 1933] H. Seifert, "Topologie Dreidimensionaler Gefaserter Räume", *Acta Math.* **60**:1 (1933), 147–238. MR 1555366 Zbl 0006.08304
- [SGA 1 1971] A. Grothendieck, *Revêtements étales et groupe fondamental* (Séminaire de Géométrie Algébrique du Bois Marie 1960–1961), Lecture Notes in Math. 224, Springer, Berlin, 1971. MR 50 #7129 Zbl 0234.14002
- [SGA 4¹/₂ 1977] P. Deligne, *Cohomologie étale* (Séminaire de Géométrie Algébrique du Bois Marie), Lecture Notes in Math. **569**, Springer, Berlin-New York, 1977. MR 57 #3132 Zbl 0349.14008
- [Silverman 2009] J. H. Silverman, *The arithmetic of elliptic curves*, 2nd ed., Graduate Texts in Mathematics **106**, Springer, Dordrecht, 2009. MR 2010i:11005 Zbl 1194.11005
- [Thornton 1967] M. C. Thornton, "Singularly fibered manifolds", *Illinois J. Math.* **11** (1967), 189–201. MR 35 #1031 Zbl 0146.45104
- [Thurston 1980] W. P. Thurston, "The Geometry and Topology of Three-Manifolds", lecture notes, 1980, http://library.msri.org/books/gt3m/.
- [Ue 1986] M. Ue, "On the diffeomorphism types of elliptic surfaces with multiple fibers", *Invent. Math.* **84**:3 (1986), 633–643. MR 87j:57019 Zbl 0595.14028
- [Vistoli 1989] A. Vistoli, "Intersection theory on algebraic stacks and on their moduli spaces", *Invent. Math.* **97**:3 (1989), 613–670. MR 90k:14004 Zbl 0694.14001
- [Xiao 1991] G. Xiao, " π_1 of elliptic and hyperelliptic surfaces", *Internat. J. Math.* 2:5 (1991), 599–615. MR 92j:14049 Zbl 0762.14013

Communicated by Shigefumi Mori Received 2014-10-08 Revised 2015-01-21 Accepted 2015-03-11

mitsui@math.kobe-u.ac.jp Department of Mathematics, Graduate School of Science, Kobe University, Hyogo 657-8501, Japan



Factorially closed subrings of commutative rings

Sagnik Chakraborty, Rajendra Vasant Gurjar and Masayoshi Miyanishi

We prove some new results about factorially closed subrings of commutative rings. We generalize this notion to quasifactorially closed subrings of commutative rings and prove some results about them from algebraic and geometric viewpoints. We show that quasifactorially closed subrings of polynomial and power series rings of dimension at most three are again polynomial (resp. power series) rings in a smaller number of variables. As an application of our results, we give a short proof of a result of Lê Dũng Tráng in connection with the Jacobian problem.

Introduction

We assume throughout the article that the base field k is an algebraically closed field of characteristic 0. Whenever we use topological arguments, k is tacitly assumed to be the field of complex numbers \mathbb{C} . By assuming naturally that k is embedded into \mathbb{C} , we can see that the results proved over \mathbb{C} can be proved over k. For an integral domain S, the field of fractions of S is denoted by Q(S), and the multiplicative group of units by S^* .

The present article grew out of the discussions we had during the workshop *Automorphisms of affine varieties*, held at the Kerala School of Mathematics, India (February 17–22, 2014). In particular, a part of our discussion was inspired by a talk given by Neena Gupta [2014] and a question asked by A. Kanel-Belov.

Let $A \subseteq B$ be integral domains. Then A is said to be *factorially closed*, or *fc*, in B if for any two nonzero elements $b_1, b_2 \in B$, $b_1b_2 \in A$ implies that $b_1, b_2 \in A$. In some papers an fc subring is also called an *inert* subring. Factorially closed subrings appear naturally as the rings of invariants of the action of the additive group G_a , or a connected semisimple group on a polynomial ring.

The notion of fc subring is not well-behaved in the case of local rings due to the existence of too many units. Hence we have introduced a weaker notion: *quasifactorially closed* subrings. For any integral domains $A \subseteq B$, A is said to be quasifactorially closed, or qfc, in B if, for any nonzero $b \in B$, if there exists

MSC2010: primary 13A05; secondary 13B99, 14R05.

Keywords: factorially closed subring.

1138 Sagnik Chakraborty, Rajendra Vasant Gurjar and Masayoshi Miyanishi

some nonzero $b' \in B$ such that $bb' \in A$, then there exists a unit $u \in B$ such that $bu \in A$. It turns out that quasifactorial closedness is more geometric and has several interesting applications. For example, we have proved that a qfc subring of a power series ring in at most three variables is again isomorphic to a power series ring in a smaller number of variables.

The fc property is also related to the property of the existence of nonconstant invertible regular functions on general fibers of the corresponding morphism of schemes.

We now mention the main results proved in this paper (with some hypothesis):

- (1) An inclusion of graded domains $A \subseteq B$ is fc if and only if it is graded fc. Further, A is fc in B if and only if the localization of A at its irrelevant maximal ideal is fc in the corresponding localization of B (Theorems 2 and 3).
- (2) For an inclusion of affine normal domains A ⊆ B the fc locus is always open (Corollary 4.1).
- (3) For an inclusion of affine UFDs $A \subseteq B$ the qfc locus is open if at most finitely many prime elements of A split in B (Theorem 5). (An example in Section 3 shows that the reverse implication is false.)
- (4) If an inclusion of complete local normal domains $A \subseteq B$ over k is qfc then A is algebraically closed in B. Further, any irreducible element of A is irreducible in B (Theorem 6 and its corollaries).
- (5) An fc subring of a polynomial ring in at most three variables is again a polynomial ring (Theorem 1). Similarly, a complete qfc subring of a power series ring in at most three variables is again a power series ring (Theorem 8).
- (6) If an inclusion of affine normal domains $A \subseteq B$ (with a suitable hypothesis) is fc, then a general fiber of the corresponding morphism of affine varieties does not have any nonconstant invertible regular functions (Theorem 11).

Using this we give a new short proof of a result proved by many authors (M. Razar, R. Heitmann, S. Friedland, L. D. Tráng, C. Weber, W. Neumann, P. Norbury) in connection with the Jacobian problem [Neumann and Norbury 1998; Tráng 2008].

In Section 3 we give some examples of ring extensions which shed more light on the fc (and qfc) property.

In Section 4 we have listed some open problems about fc and qfc extensions.

1. Factorially closed subrings

We start with some basic properties of factorial closedness. Some easy proofs have been omitted.

Lemma 1 (local properties of factorial closedness). *The following statements are equivalent for an inclusion of integral domains* $A \subseteq B$:

- (1) The ring A is fc in B.
- (2) For any multiplicatively closed set S in A, $S^{-1}A \subseteq S^{-1}B$ is fc.
- (3) For any prime ideal $\mathfrak{p} \in \operatorname{Spec} A$, $A_{\mathfrak{p}} \subseteq B_{\mathfrak{p}}$ is fc.
- (4) For any maximal ideal $\mathfrak{m} \in \operatorname{Max} A$, $A_{\mathfrak{m}} \subseteq B_{\mathfrak{m}}$ is fc.
- (5) There exist finitely many nonzero elements $a_1, a_2, ..., a_n \in A$, generating the unit ideal, such that A_{a_i} is fc in B_{a_i} for each i = 1, 2, ..., n.

Moreover, if A is normal, the above statements are equivalent to the following one:

(7) For each prime ideal $\mathfrak{p} \in \operatorname{Spec} A$ of height 1, $A_{\mathfrak{p}} \subseteq B_{\mathfrak{p}}$ is fc.

Proof. Omitted.

Lemma 2 (transitive and sandwich properties of factorial closedness). Let $A \subseteq B \subseteq C$ be integral domains.

- (1) If the ring A is fc in B and B is fc in C, then A is fc in C.
- (2) If A is fc in C then it is fc in B. However, in this case B need not be fc in C.

The example $k \subseteq k[t^2] \subseteq k[t]$ shows that *B* need not be fc in *C*.

Lemma 3. Let A be an fc subring of B.

- (1) The ring A is algebraically closed in B.
- (2) If Q(A) is the field of fractions of A, then $Q(A) \cap B = A$. This is the same thing as saying that each principal ideal of A is a contracted ideal.
- (3) If B is integrally closed (or a UFD), then so is A. In fact, in the case of Krull domains, the natural homomorphism of divisor class groups $Cl(A) \rightarrow Cl(B)$ is an injection whenever it is defined.
- (4) Any unit of B is in A.

Proof. The first assertion follows from the slightly more general fact that, for a pair of integral domains $A \subseteq B$, if $B \setminus A$ is closed under multiplication then A is algebraically closed in B.

The other three statements follow from the first one and the next observation. \Box

Remark. For an inclusion of Krull domains $A \subset B$, there is a natural homomorphism $Cl(A) \rightarrow Cl(B)$ if and only if no height 1 prime ideal of *B* contracts to a prime ideal of height > 1 in *A* [Samuel 1964].

1140 Sagnik Chakraborty, Rajendra Vasant Gurjar and Masayoshi Miyanishi

If I is an ideal of A such that the ideal IB is principal, then, since A is fc in B, I itself must be principal. This observation will be implicitly used later.

Lemma 4. Let A be an fc subring of B. Then the Jacobson radical of B, Jac B, is contained in A. Moreover, if Jac $B \neq 0$ then A = B. In particular, if B is semilocal then A = B.

Proof. If $b \in \text{Jac } B$, $1 + b \in B^*$. So, by Lemma 3(4), $1 + b \in A$ implies that $b \in A$. Now, if *b* is a nonzero element in Jac *B*, for any $x \in B$, *xb* is also in Jac *B* and consequently in *A*. So $x \in A$. If *B* is semilocal, and not a field, then Jac $B \neq 0$ and the rest of the assertion follows. If *B* is a field then every nonzero element in *B* is a unit, and since *A* is fc in *B* we again get A = B.

Lemma 5. If $A_1 \subseteq A_2 \subseteq A_3 \subseteq \cdots$ and $B_1 \subseteq B_2 \subseteq B_3 \subseteq \cdots$ are two sequences of integral domains, such that $A_i \subseteq B_i$ is an fc subring for each *i*, then $\bigcup_i A_i \subseteq \bigcup_i B_i$ is also factorially closed.

Lemma 6. If $A \subseteq B \subseteq C$ are integral domains with A an fc subring of B, then, for any subring D of C, $D \cap A \subseteq D \cap B$ is also factorially closed.

Before looking into the ring-theoretic properties of factorial closedness, we would like to describe the structure of a factorially closed subalgebra of the polynomial ring $R = k[x_1, ..., x_n]$. This question can be answered if $n \le 3$, and the answer is simply a polynomial subalgebra. We consider only the case where n = 3. The case n = 2 has a similar answer and is easier.

Theorem 1. Let A be a factorially closed subring of R = k[x, y, z].

- (1) If dim A = 3, then A = R.
- (2) If dim A = 2, then A is a polynomial ring in two variables.
- (3) If dim A = 1, then A = k[f], where f c is irreducible in k[x, y, z] for every $c \in k$.

Proof. (1) In this case, the transcendence degree of A over k is 3. So A, being algebraically closed in k[x, y, z] by Lemma 3(1), must be equal to k[x, y, z].

In the other two cases, since A is a UFD (by Lemma 3(3)) of transcendence degree ≤ 2 , by a result of Zariski [Nagata 1965, p. 52, Theorem 4] A is affine.

So the assertion (3) follows from the fact that, when *A* has dimension 1, *A* is an affine PID with trivial units.

(2) Note that A is a normal affine domain of dimension 2. We assume for simplicity that $k = \mathbb{C}$. Let $Y = \operatorname{Spec} R$ and $X = \operatorname{Spec} A$. The inclusion $A \hookrightarrow R$ defines a dominant morphism $p: Y \to X$. Then every fiber of p is either the empty set or is 1-dimensional. For, if there exists a fiber component D of dimension 2, let it be defined by f = 0 with $f \in R$. Since p(D) is a closed point of X corresponding to a maximal ideal m of A, we have $\mathfrak{m} \subseteq \mathfrak{m} R \subseteq f R$. This implies that any nonzero
element of m is divisible by f, whence $f \in A$. This is a contradiction since A is 2-dimensional. Furthermore, a general fiber of p is irreducible since A is factorially closed in R. By [Miyanishi 1986, Theorem 3], X is isomorphic to either \mathbb{A}^2 or an affine hypersurface $x_1^2 + x_2^3 + x_3^5 = 0$ in \mathbb{A}^3 . But, arguing as in the proof of [Miyanishi 1986, Theorem 4], we can show that the latter case cannot occur.

Let $B := \bigoplus_i B_i$ be a \mathbb{Z} -graded domain and $A := \bigoplus_i A_i$ a graded subring of B, i.e., $A_i \subseteq B_i$ for each i. We say that A is graded factorially closed or gfc, in short, in B if, given any two nonzero homogeneous elements $b_i, b_j \in B, b_i b_j \in A$ implies that $b_i, b_j \in A$. First, the following lemma shows that gfc is a local property:

Lemma 7. Let $A \subseteq B$ be \mathbb{Z} -graded domains. Then the following statements are equivalent:

- (1) The ring A is gfc in B.
- (2) For any multiplicative set S in A, generated by homogeneous elements, $S^{-1}A \subseteq S^{-1}B$ is gfc.
- (3) For any homogeneous prime ideal $\mathfrak{p} \in \operatorname{Spec} A$, $A_{(\mathfrak{p})} \subseteq B_{(\mathfrak{p})}$, where $A_{(\mathfrak{p})}$ and $B_{(\mathfrak{p})}$ denote the localizations of A and B respectively at the multiplicative set consisting of all homogeneous elements of A not contained in \mathfrak{p} , is gfc.

If, moreover, A happens to be positively graded, the above statements are equivalent to the following:

(4) For any homogeneous maximal ideal m ∈ Max A, A_(m) ⊆ B_(m) is graded factorially closed.

Proof. We only show $(3) \implies (1)$. Let $x, y \in B$ be homogeneous elements such that $xy \in A$. So, $x, y \in A_{(p)}$ for every homogeneous prime ideal p. But the set $(A:x) := \{a \in A \mid ax \in A\}$ is a homogeneous ideal in A. So, if $x \notin A$, then (A:x) must be proper ideal and hence contained in a homogeneous prime ideal, leading to a contradiction. For positively graded rings, note that any homogeneous ideal is actually contained in a homogeneous maximal ideal.

Note that properties analogous to the fc property as expressed in Lemmas 1, 2 and 3 also hold for graded factorially closed subrings. The reader is invited to come up with the precise formulations and their proofs.

Next we take our first step in building a bridge between factorial closedness and graded factorial closedness.

Lemma 8. Let $A \subseteq B$ be \mathbb{Z} -graded domains and \mathfrak{p} a homogeneous prime ideal of A. If $A_{\mathfrak{p}} \subseteq B_{\mathfrak{p}}$ is fc then $A_{(\mathfrak{p})} \subseteq B_{(\mathfrak{p})}$ is gfc.

Proof. Let $x, y \in B_{(\mathfrak{p})}$ be homogeneous elements such that $xy \in A_{(\mathfrak{p})}$. Let x = x'/sand y = y'/t with $x', y' \in B$ and $s, t \in \bigcup_i A_i - \mathfrak{p}$. Since $A_{\mathfrak{p}} \subseteq B_{\mathfrak{p}}$ is factorially closed, $x, y \in A_{\mathfrak{p}}$. So there exist nonzero elements in $A, a = \sum_j a_j$ and $\alpha = \sum_j \alpha_j$ with $\alpha \notin \mathfrak{p}$ such that $x'/s = a/\alpha$. Again, α being outside \mathfrak{p} implies that $\alpha_{j^*} \notin \mathfrak{p}$ for some j^* . So $x'\alpha_{j^*} \in A$ and consequently $x \in A_{(\mathfrak{p})}$. Similarly $y \in A_{(\mathfrak{p})}$, and this finishes the proof.

It is natural to ask whether gfc implies fc. Our next few results show that this is indeed true. We first treat the easy case of polynomial ring extensions and then show that the general case, under minor restrictions, reduces to this special case.

Lemma 9. Let A be a factorially closed subring of an integral domain B. Then the polynomial ring A[x] is also factorially closed in B[x].

Proof. Let $f(x), g(x) \in B[x] - \{0\}$ be such that $f(x)g(x) \in A[x]$. It is enough to show that $f(x) \in A[x]$. We consider the following two possible cases:

Case 1: *A is infinite.* Since *f* and *g* can have only finitely many roots, there exist infinitely many $a \in A$ such that f(a), g(a) are nonzero elements of *B* and $f(a)g(a) \in A$, and consequently $f(a) \in A$, since *A* is factorially closed in *B*. In particular, if *f* has degree *n*, there exist n + 1 distinct values in *A*, say $a_1, a_2, \ldots, a_{n+1}$, such that $f(a_i) \in A$ for each $i = 1, 2, \ldots, n + 1$. So, treating the coefficients of *f* as variables, and plugging in the values a_i , we get n + 1 linear equations in n + 1 variables. The simultaneous linear equations have a solution in *B*. If we look at the corresponding Vandermonde matrix, it is obvious that the solution actually lies in Q(A). Since $Q(A) \cap B = A$ by Lemma 3(2), $f(x) \in A[x]$.

Case 2: *A is a field.* Without any loss of generality we may assume that f and g are monic polynomials. Let L be a splitting field of fg over Q(B). The roots of fg, and hence in particular the roots of f, are integral over Q(A). Consequently, the coefficients of f, being symmetric functions of the roots, are integral over Q(A) and hence algebraic over A. But since A is algebraically closed in B by Lemma 3(1), the coefficients are actually in A, and hence $f(x) \in A[x]$.

For the general case, let *A* be a graded factorially closed subring of a \mathbb{Z} -graded domain *B* such that $A_i \neq 0$ and $A_{i+1} \neq 0$ for some integer *i*. We want to show that $A \subseteq B$ is factorially closed. Let *S* be the multiplicative set consisting of all nonzero homogeneous elements of *A*. Note that if $S^{-1}A \subseteq S^{-1}B$ is factorially closed then so is $A \subseteq B$. If $K := (S^{-1}A)_0$ and $\widetilde{B} := (S^{-1}B)_0$, then *K* is a field which is factorially closed in \widetilde{B} . Choose any nonzero elements $a_i \in A_i, a_{i+1} \in A_{i+1}$, and let $t := a_{i+1}/a_i$. Then $t \in (S^{-1}A)_1$ and $S^{-1}A = K[t, t^{-1}]$. To show that $K[t, t^{-1}]$ is factorially closed in $S^{-1}B$, let $b := b_{i_0} + b_{i_1} + \cdots + b_{i_r}$ and $c := c_{j_0} + c_{j_1} + \cdots + c_{j_s}$, with b_{i_0}, b_{i_r} and c_{j_0}, c_{j_s} nonzero, be elements of $S^{-1}B$ such that $bc \in S^{-1}A$. Writing $\overline{b_{i_\alpha}} := b_{i_\alpha} t^{-i_\alpha}$ for $\alpha = 0, 1, \ldots, r$ and $\overline{c_{j_\beta}} := c_{j_\beta} t^{-j_\beta}$ for $\beta = 0, 1, \ldots, s$, we get that

 $b = \overline{b_{i_0}}t^{i_0} + \overline{b_{i_1}}t^{i_1} + \dots + \overline{b_{i_r}}t^{i_r}$, $c = \overline{c_{j_0}}t^{j_0} + \overline{c_{j_1}}t^{j_1} + \dots + \overline{c_{j_s}}t^{j_s} \in \widetilde{B}[t, t^{-1}]$. But since *K* is factorially closed in \widetilde{B} , so is $K[t] \subseteq \widetilde{B}[t]$, and consequently, by Lemmas 1 and 2, $K[t, t^{-1}]$ is factorially closed in $\widetilde{B}[t, t^{-1}]$. So $b, c \in S^{-1}A$, proving that $S^{-1}A \subseteq S^{-1}B$ is factorially closed, and hence $A \subseteq B$ is also factorially closed.

Therefore, we have proved the following result:

Theorem 2. Let $A \subseteq B$ be \mathbb{Z} -graded domains with $A_i \neq 0$ and $A_{i+1} \neq 0$ for some integer *i*. Then A is factorially closed in B if it is graded factorially closed.

Question. Is the hypothesis that A_i , A_{i+1} are nonzero for some *i* necessary?

We do not know the answer to the above question in general. But we sketch below a different proof of Theorem 2 without assuming the condition that A_i and A_{i+1} are nonzero for some *i*. However it works only when *B* is a UFD.

Let $A \subseteq B$ be \mathbb{Z} -graded domains with B a UFD. Now, assuming that A is graded factorially closed in B, we would like to show that A is factorially closed in B. After inverting all nonzero homogeneous elements of A, we may assume that A is of the form $k[t, t^{-1}]$, where t is a homogeneous prime element of positive degree in B. Further, since k[t] is factorially closed in $B_0[t]$ by Theorem 9, it suffices to prove that $B_0[t]$ is factorially closed in $B^+ := \bigoplus_{i \ge 0} B_i$. So, if $f, g \in B^+$ are such that $fg \in B_0[t]$, we want to show that $f, g \in B_0[t]$. Let $f = f_0 + f_1 + \cdots + f_m$ and $g = g_0 + g_1 + \cdots + g_n$, with f_m and g_n nonzero. We can write f and g as

$$f = f'_0 t^{\alpha_0} + f'_1 t^{\alpha_1} + \dots + f'_m t^{\alpha_m}$$
 and $g = g'_0 t^{\beta_0} + g'_1 t^{\beta_1} + \dots + g'_m t^{\beta_m}$,

where $f_i = f'_i t^{\alpha_i}$ and *t* does not divide f'_i unless it is zero, in which case we also take α_i to be zero, and similarly for *g*. If either $f'_i \in B_0$ for each *i* or $g'_j \in B_0$ for each *j*, we are done. Otherwise, we define α_* and β_* to be the minimums of the α_i for $f_i \neq 0$ and the β_j for $g_j \neq 0$, respectively. Let us also define $i^* := \max\{i \mid \alpha_i = \alpha_*\}$ and $j^* := \max\{j \mid \beta_j = \beta_*\}$. Note that $i^* < m$ and $j^* < n$. Looking at the homogeneous component of degree $i^* + j^*$ in fg, we get

$$(fg)_{i^*+j^*} = f'_{i^*}g'_{j^*}t^{\alpha_*+\beta_*} +$$
(elements divisible by $t^{\alpha_*+\beta_*+1}$).

But that means $f'_{i*}g'_{i*}$ must be divisible by t, which is a contradiction.

Finally, we put together the results connecting factorial closedness and graded factorial closedness in the form of the following theorem:

Theorem 3. For positively graded domains $A \subseteq B$, if we assume that $A_1 \neq 0$, then the following statements are equivalent:

- (1) $A \subseteq B$ is fc.
- (2) For any homogeneous prime ideal \mathfrak{p} , $A_{(\mathfrak{p})} \subseteq B_{(\mathfrak{p})}$ is gfc.
- (3) For any homogeneous prime ideal \mathfrak{p} , $A_{\mathfrak{p}} \subseteq B_{\mathfrak{p}}$ is fc.
- (4) For any homogeneous maximal ideal \mathfrak{m} , $A_{(\mathfrak{m})} \subseteq B_{(\mathfrak{m})}$ is gfc.

(5) For any homogeneous maximal ideal \mathfrak{m} , $A_{\mathfrak{m}} \subseteq B_{\mathfrak{m}}$ is fc.

Proof. Follows directly from Lemmas 7 and 8 and Theorem 2.

In particular, if A_0 is a field, A is positively graded, $A_1 \neq (0)$ and m is the irrelevant maximal ideal of A, then $A \subseteq B$ will be factorially closed if $A_m \subseteq B_m$ is factorially closed.

Given a subring *A* of an integral domain *B*, we define the factorially closed locus or fc locus of *A* in *B* to be $\mathfrak{FC}(A:B) := \{\mathfrak{p} \in \text{Spec } A \mid A_{\mathfrak{p}} \subseteq B_{\mathfrak{p}} \text{ is factorially closed}\}.$ We intend to investigate the nature of this fc locus in the Zariski topology. We start with a few definitions: let $A/B := \{b \in B \mid bb' \in A \text{ for some } b' \in B - \{0\}\}$, which is an *A*-module. If \overline{A} denotes the algebraic closure of *A* in *B*, it is easy to see that $A \subseteq Q(A) \cap B \subseteq \overline{A} \subseteq A/B \subseteq B$. Taking *A* to be the ring of integers \mathbb{Z} and *B* to be $\mathbb{Z}[\sqrt{2}, x, y, 1/2y]$, one can see that the inclusions can be proper at each stage. By Lemma 1(2), factorial closedness is preserved under localization. So $\mathfrak{FC}(A:B)$ is closed under generalization. The following lemma gives a necessary and sufficient condition for the nonemptiness of the fc locus.

Lemma 10. With notation as above, the following statements are equivalent:

- (1) There exists a prime ideal $\mathfrak{p} \in \operatorname{Spec} A$ such that $A_{\mathfrak{p}} \subseteq B_{\mathfrak{p}}$ is fc.
- (2) The inclusion $Q(A) \subseteq S^{-1}B$ is fc, where $S := A \{0\}$.
- (3) *There is an equality* $Q(A)^* = (S^{-1}B)^*$.
- (4) There is an equality $A/B = Q(A) \cap B$.

Proof. We will only give the proof that (4) implies (2). The other implications are similar and easy.

Assume that $A/B = Q(A) \cap B$. We will show that Q(A) is fc in $S^{-1}B$.

Let $(b_1/s_1) \cdot (b_2/s_2) \in Q(A)$, where the s_i are nonzero elements of A. Then there is a nonzero element $\alpha \in A$ such that $b_1b_2\alpha \in A$. This implies that $b_i \in A/B$, and hence $b_i \in Q(A)$.

Note that $Q(A)^* = (S^{-1}B)^*$ implies that $B^* \subseteq Q(A)$. But the converse is false as the example $K[xz, yz] \subseteq K[x, y, z]$ with K a field shows.

To give conditions for the openness of the fc locus, we need a few auxiliary lemmas.

Lemma 11. Let $A \subseteq B$ be integral domains. If $\mathfrak{p} \in \text{Spec } A$ is a prime ideal of height 1 which is not in the image of Spec B, then $V(\mathfrak{p}) := {\mathfrak{q} \in \text{Spec } A \mid \mathfrak{p} \subseteq \mathfrak{q}}$ does not meet $\mathfrak{FC}(A : B)$.

Proof. Otherwise, if $q \in V(\mathfrak{p}) \cap \mathfrak{FC}(A : B)$, then $A_{\mathfrak{p}} \subseteq B_{\mathfrak{p}}$ is factorially closed by Lemma 1. But, since \mathfrak{p} is not in the image of Spec *B*, each prime ideal of $B_{\mathfrak{p}}$ contracts to (0) in $A_{\mathfrak{p}}$. Consequently, every nonzero element of $A_{\mathfrak{p}}$ is a unit in $B_{\mathfrak{p}}$ which is a contradiction by Lemma 3. Hence we must have that $V(\mathfrak{p}) \cap \mathfrak{FC}(A : B) = \emptyset$. \Box

Lemma 12. Let $A \subseteq B$ be integral domains with A noetherian and normal. If the image of Spec *B* contains all prime ideals $\mathfrak{p} \in$ Spec *A* of height 1, then either the fc locus $\mathfrak{FC}(A : B)$ is empty or *A* is factorially closed in *B*.

Proof. If $A/B \neq Q(A) \cap B$, we know that the factorially closed locus will be empty. So we are interested in showing that, if $A/B = Q(A) \cap B$, then *A* is factorially closed in *B*. In order to prove factorial closedness, first note that it suffices to prove that any principal ideal of *A* is contracted from some ideal of *B*, or, equivalently, that $xB \cap A = xA$ for any $x \in A$. For, if it is true, let us consider $b_1, b_2 \in B - \{0\}$ such that $b_1b_2 = a \in A$. Since $A/B = Q(A) \cap B$, $b_1 \in Q(A)$. Let $b_1 = \alpha/\beta$, with $\alpha, \beta \in A - \{0\}$. Now $\alpha \in \beta B \cap A = \beta A$, implying that $b_1 \in A$, and consequently *A* is factorially closed in *B*. So all we need to show is that any principal ideal of *A* is contracted for a principal ideal has height 1, and, as a result, using primary decomposition any principal ideal of *A* can be written as a finite intersection of primary ideals of height 1. So it is enough to prove that any height-1 primary ideal of *A* is a contracted ideal. Now, given any prime ideal $\mathfrak{p} \in \text{Spec } A$ of height 1, let us consider the commutative diagram



The local ring A_p is a DVR and pA_p is a contracted ideal. So each pA_p -primary ideal is also contracted. Again, p-primary ideals of A are in a one-to-one correspondence with the pA_p -primary ideals of A_p . So we conclude that each p-primary ideal of A is contracted from some ideal in B, and this completes the proof. \Box

Note. Let $A \subseteq B$ be integral domains with A noetherian. We have proved that if $A_{\mathfrak{p}}$ is a DVR for some $\mathfrak{p} \in \operatorname{Spec} A$ then $\mathfrak{p} \in \mathfrak{FC}(A : B)$ if and only if it is in the image of Spec B.

Now we are in a position to characterize the openness of the fc locus:

Theorem 4. Let $A \subseteq B$ be integral domains with A noetherian and normal. Then $\mathfrak{FC}(A : B)$, if nonempty, is open in Spec A if and only if the image of Spec B misses only finitely many height-1 prime ideals.

Proof. If $\mathfrak{FC}(A : B)$ is open, its complement contains at most finitely many prime ideals of height 1. So, by Lemma 12, the image of Spec *B* misses at most finitely many height-1 prime ideals. Conversely, assume that $\mathfrak{p}_1, \mathfrak{p}_2, \ldots, \mathfrak{p}_n$ are the only prime ideals of height 1 lying outside the image of Spec *B*. In view of Lemma 12, it is enough to show that any $\mathfrak{q} \in$ Spec *A* not contained in $\bigcup_{i=1}^n V(\mathfrak{p}_i)$ is in $\mathfrak{FC}(A : B)$. So, let us choose any $\mathfrak{q} \in$ Spec $A - \bigcup_{i=1}^n V(\mathfrak{p}_i)$. We can find $x \in \bigcap_{i=1}^n \mathfrak{p}_i - \mathfrak{q}$. Considering the inclusion $A_x \subseteq B_x$, all height-1 prime ideals of A_x are in the image of Spec B_x . Since we are only interested in the case when $\mathfrak{FC}(A : B) \neq \emptyset$, we may assume by Lemma 10 that $A/B = Q(A) \cap B$. Consequently $A_x/B_x = Q(A_x) \cap B_x$. Therefore, Lemma 11 applies to show that A_x is factorially closed in B_x . Hence $\mathfrak{FC}(A : B) = \operatorname{Spec} A - (\bigcup_{i=1}^n V(\mathfrak{p}_i))$ is open. \Box

Corollary 4.1. With notation as in Theorem 4, if B is a finitely generated algebra over A then $\mathfrak{FC}(A : B)$ is always open.

Proof. This follows from Theorem 4, since the corresponding dominant morphism of affine schemes Spec $B \rightarrow$ Spec A always contains a nonempty open set in its image, and consequently the image of Spec B can miss at most finitely many height-1 primes of Spec A.

2. Quasifactorially closed subrings

If we attempt to generalize Lemma 9 to the case of formal power series rings $A[[x]] \subseteq B[[x]]$ the attempt fails quite badly. For, suppose that $A \subseteq B$ is factorially closed. If $b \in B - A$, then the element $1 + bx + x^2 + x^3 + x^4 + \cdots$ is a unit in B[[x]] which is not in A[[x]]. So the extension $A[[x]] \subseteq B[[x]]$ is never factorially closed unless A = B. The presence of 'extra units' in the bigger ring turns out to be an obvious obstruction. To rectify this problem, we come up with a weaker notion of quasifactorially closedness.

Recall that, given an inclusion of integral domains $A \subseteq B$, A is said to be quasifactorially closed, or qfc for short, in B if, for any nonzero $b \in B$, if there exists some nonzero $b' \in B$ such that $bb' \in A$, then there exists a unit $u \in B$ such that $bu \in A$.

If *A* and *B* have the same units then the notions of factorial closedness and quasifactorial closedness coincide. Also note that $A \subseteq B$ is quasifactorially closed whenever either *A* or *B* is a field. But quasifactorial closedness, in general, is more of a geometric notion and does not behave well with algebraic operations. For example, although it is closed under localization, we are not yet sure if global information can be retrieved from local data as in Lemma 1. The *sandwich property*, as in Lemma 2, also fails, as any integral domain is always quasifactorially closed in any field containing it. The following example shows that the *transitive property* need not hold true either.

Example. Factorial closedness holds for $K[x] \subseteq K[x, y, z, w]/(xy - zw)$, and A := K[x, y, z, w]/(xy - zw) is qfc in A[1/y]. But K[x] is not qfc in A[1/y], as there is no unit u in A[1/y] such that $uz \in K[x]$.

Note that the above example also shows that in the definition of quasifactorial closedness it may not be possible to get a unit $u \in B$ such that $bu, b'u^{-1} \in A$. In fact, if it were true then one can check that, for integral domains $A \subseteq B \subseteq C$, if $A \subseteq B$ is fc and $B \subseteq C$ is qfc then $A \subseteq C$ would also be qfc, which is clearly not true, as the above example shows.

The following example shows that $A \subseteq B$ being qfc does not imply that $A[x] \subseteq B[x]$ is qfc:

Example. Take any nontrivial algebraic field extension L/K. Then $K \subseteq L$ is qfc but $K[x] \subseteq L[x]$ is not qfc. If $K = \mathbb{R}$ and $L = \mathbb{C}$, take b = ix - 1 and b' = ix + 1. Then $bb' \in \mathbb{R}[x]$, but there is no unit $u \in \mathbb{C}[x]$ such that $bu \in \mathbb{R}[x]$.

Let $A \subseteq B$ be fc. Then any irreducible element of A remains irreducible in B. If A is a UFD but B is not, then prime elements of A need not remain prime in B, as the example $k[x] \subseteq k[x, y, z]/(xy - z^2 - 1)$ shows, where the prime element x of k[x] does not remain a prime in $k[x, y, z]/(xy - z^2 - 1)$. But if B is also a UFD then $A \subseteq B$ is fc if and only if each prime element of A remains a prime in B and $A^* = B^*$. For UFDs $A \subseteq B$, primes of A remaining primes in B is a sufficient condition for qfc. But it is not necessary, as the first example in Section 4 will show. However, it follows from Theorem 6 and its corollaries that the converse is also true in the case of complete local UFDs.

For integral domains $A \subseteq B$, we define the qfc locus of A in B by $\mathfrak{QFC}(A:B) := {\mathfrak{p} \in \text{Spec } A \mid A_{\mathfrak{p}} \subseteq B_{\mathfrak{p}} \text{ is qfc}}$. Just like the fc locus, the qfc locus is also closed under generalization. Note that $\mathfrak{QFC}(A:B)$ is always nonempty since $(0) \in \mathfrak{QFC}(A:B)$. For, let $S = A - {0}$. If $(b_1/s_1).(b_2/c_2) \in Q(A)$ then b_i/s_i are units in $S^{-1}B$. Then $(b_i/s_i).(s_i/b_i) \in Q(A)$, implying that Q(A) is qfc in $S^{-1}B$.

Next, we prove an openness criterion, analogous to Theorem 4, for the qfc locus, albeit for a somewhat restricted class of rings.

Theorem 5. Let $A \subseteq B$ be affine UFDs. Assume that A and B have the same group of units and Q(A) is algebraically closed in Q(B). Then $\mathfrak{QSC}(A : B)$ is a nonempty open set if one, and hence all, of the following equivalent conditions hold:

- (1) Given any prime ideal $\mathfrak{p} \in \operatorname{Spec} A$ of height ≥ 2 , $\mathfrak{p}B$ has height ≥ 2 .
- (2) No prime element of B divides two distinct prime elements of A.
- (3) Any two coprime elements of A continue to be coprime in B.
- (4) There are only finitely many prime elements of A which either split in B or are units in B.

Proof. Since A and B are UFDs, the proof of the equivalence of (1), (2) and (3) in the above statement is easy. This part does not need openness of $\mathfrak{QFC}(A : B)$.

Let V and W denote the irreducible affine varieties corresponding to A and B respectively, and let $f: W \rightarrow V$ be the induced morphism.

First we consider the case when dim A = 1.

By assumption, Q(A) is algebraically closed in Q(B). Then it is well-known (by a suitable application of Bertini's theorem) that only finitely many scheme-theoretic fibers of the morphism $W \rightarrow V$ are either empty or not reduced and irreducible. This shows that (4) is also always true, so that conditions (1)–(4) are equivalent.

Now we will assume that dim $A \ge 2$.

Assume now that the equivalent conditions (1), (2) and (3) hold. We will show that (4) holds.

Again, since Q(A) is algebraically closed in Q(B), there is a proper closed subvariety $S \subset V$ such that the inverse image of any point $p \notin S$ is schemetheoretically reduced and irreducible. By (1), the inverse image of any closed subvariety of V of codim ≥ 2 does not contain any divisor in W. Now we can see that the only possible irreducible divisors $D \subset V$ which split in W are those contained in S.

The image f(W) contains a nonempty Zariski-open subset since f is dominant. Hence f(W) can miss at most finitely many divisors in V. This shows that (4) is true.

Next, we will show that (4) implies (1). Suppose that this is not true. Then there is a closed irreducible subvariety $S \subset V$ of codimension > 1 such that the inverse image of *S* in *W* contains an irreducible divisor Δ , defined by a prime element *q*. Now, if *D* is any irreducible divisor in *V* which contains *S* then the prime element defining *D* will split in *B*. Since dim B > 1, there are infinitely many such prime elements in *A*.

This proves the equivalence of (1)–(4).

Now we will assume that the equivalent conditions (1)–(4) hold. We will show that $\mathfrak{QFC}(A : B)$ is a nonempty open set.

Let p_1, \ldots, p_r be the prime elements in A such that p_i is a non-unit in B and not a prime element in B.

We will show that $\mathfrak{QFC}(A:B) = \operatorname{Spec} A \setminus \bigcup_{i=1}^{r} V(p_i A)$, and hence $\mathfrak{QFC}(A:B)$ is nonempty and open.

First we will show that if a prime element $p \in A$ is not a unit in B and does not remain a prime element in B, then $V(pA) \cap \mathfrak{QFC}(A : B) = \emptyset$.

So, let $p \in A$ be such a prime element and let $q \in V(pA)$. If $A_q \subseteq B_q$ is qfc then so is $A_p \subseteq B_p$, where p := pA. Since p is not a prime element in B, there exists $b_1, b_2 \in B - B^*$ such that $p = b_1b_2$. But $A_p \subseteq B_p$ being qfc implies that either b_1 or b_2 must be a unit in B_p . Without any loss of generality, let us assume that $b_1 \in B_p^*$. So b_1 divides some element $s \in A - p$. But s and p are coprime in A, and hence in B by (3). So b_1 must be a unit in B, which is a contradiction. By assumption, there are only finitely many prime elements $p_i \in A$ such that p_i is a non-unit in B and not a prime element in B. We have already seen that $\bigcup_{i=1}^{n} V(p_i A) \subseteq \mathfrak{QFC}(A : B)^c$. So it suffices to show that any prime ideal of A which does not contain any of the p_i is in $\mathfrak{QFC}(A : B)$. Let $\mathfrak{q} \in \text{Spec } A$ be such a prime ideal. Choose any $a \in \bigcap_{i=1}^{n} p_i A - \mathfrak{q}$. Then $A_a \subseteq B_a$ is qfc since each prime element in A_a continues to be a prime element in B_a . Consequently, $A_{\mathfrak{q}} \subseteq B_{\mathfrak{q}}$ is also qfc, and this completes the proof.

Remark. In Section 3, we will give an example to show that $\mathfrak{QFC}(A : B)$ can be open and nonempty even when there are infinitely many prime elements in *A* which are not units in *B* and are not prime elements in *B*.

The following theorem shows that the qfc property has some nice consequences in the case of complete local domains:

Theorem 6. Let $(A, \mathfrak{m}_A) \subseteq (B, \mathfrak{m}_B)$ be local domains such that $\mathfrak{m}_A = \mathfrak{m}_B \cap A$ and $A/\mathfrak{m}_A = B/\mathfrak{m}_B$. Moreover, assume that A is complete in the \mathfrak{m}_A -adic topology and $\bigcap_{n=1}^{\infty} \mathfrak{m}_B^n = (0)$. If A is qfc in B then A is algebraically closed in B.

Proof. Let $b \in B$ be algebraic over A. We will construct a sequence $(a_n) \in A^{\mathbb{N}}$ such that, for each n, $a_{n+1} = a_n + \alpha_1 \alpha_2 \cdots \alpha_n \alpha_{n+1}$ and $b = a_n + \alpha_1 \alpha_2 \cdots \alpha_n \beta_n$ for some $\alpha_1, \alpha_2, \ldots, \alpha_n, \alpha_{n+1} \in \mathfrak{m}_A$ and $\beta_n \in \mathfrak{m}_B$. Any such sequence will be a Cauchy sequence in the \mathfrak{m}_A -adic topology of A which converges to b in the \mathfrak{m}_B -adic topology of B, implying that $b \in A$.

Since b is algebraic over A, there exist elements $c_0, c_1, \ldots, c_r \in A$, with c_0 and c_r nonzero, such that

$$c_0 + c_1 b + \dots + c_r b^r = 0,$$

implying that $b(c_1 + c_2b + \dots + c_rb^{r-1}) \in A$. Since *A* is qfc in *B*, there exists a unit $u \in B^*$ such that $bu = a \in A$ or, equivalently, $b = au^{-1}$. We can write $u^{-1} = u'_1 + b_1$ for some $u'_1 \in A^*$ and $b_1 \in \mathfrak{m}_B$, so that $b = a(u'_1 + b_1)$. Setting $a_1 := au'_1$, the induction hypothesis is satisfied for n = 1.

Next, suppose that we have already found elements $a_1, a_2, ..., a_n \in A$ satisfying the required conditions. To find a_{n+1} , note that $b = a_n + \alpha_1 \alpha_2 \cdots \alpha_n \beta_n$, implying that $\alpha_1 \alpha_2 \cdots \alpha_n \beta_n$ is also algebraic over A, and consequently there exists a unit $u_{n+1} \in B^*$ such that $\beta_n = \alpha_{n+1}u_{n+1}$ for some $\alpha_{n+1} \in \mathfrak{m}_A$. Writing u_{n+1} as $u_{n+1} = u'_{n+1} + \beta_{n+1}$, where $u'_{n+1} \in A^*$ and $\beta_{n+1} \in \mathfrak{m}_B$, we get

$$b = a_n + \alpha_1 \alpha_2 \cdots \alpha_n \alpha_{n+1} (u'_{n+1} + \beta_{n+1}).$$

It is obvious that $a_{n+1} := a_n + \alpha_1 \alpha_2 \cdots \alpha_n \alpha_{n+1} u'_{n+1}$ satisfies the required properties. This, together with induction, completes the proof.

With notation as in Theorem 6, we have the following easy corollaries:

Corollary 6.1. *There is an equality* $Q(A) \cap B = A$.

Corollary 6.2. If B is normal, then so is A.

Corollary 6.3. If B satisfies the ascending chain condition for principal ideals, then so does A.

Corollaries 6.1, 6.2 and 6.3 are immediate consequence of Theorem 6.

Corollary 6.4. Any irreducible element of A remains irreducible in B.

This can be proved in the same way as the proof of Theorem 6. We leave the details to the reader.

Corollary 6.5. If $a \in A$ is a prime element of B, then it is already a prime element in A.

Corollary 6.6. Let $a, a' \in A$. Then aA = a'A if and only if aB = a'B. In particular, if two elements of A are not associates in A, they cannot become associates in B.

This follows easily from Corollary 6.1. For, if $a' \in aB$, then, writing a' = ab with $b \in B$, by Corollary 6.1 we have $b \in Q(A) \cap B = A$.

Corollary 6.7. If B is a UFD, then so is A.

Proof. Corollary 6.3 shows that any element in A can be written as a product of irreducible elements in A. By Corollary 6.4, any irreducible element in A remains irreducible and hence a prime in B. Now Corollary 6.5 finishes the proof.

Corollary 6.8. If two elements of A have no common factor in A, they cannot have a common factor in B.

Proof. If $b \in \mathfrak{m}_B$ is a common factor of $a, a' \in \mathfrak{m}_A$, then there is a unit $u \in B^*$ such that $bu \in \mathfrak{m}_A$. But then, by Corollary 6.1, bu is a common factor of a and a' in A, leading to a contradiction.

Corollary 6.9. If B is a UFD, then, for any prime ideal $\mathfrak{p} \in \text{Spec } A$ of height ≥ 2 , $\mathfrak{p}B$ has height ≥ 2 .

Now we prove an analogue of Theorem 1 for power series rings. First, we consider the 2-dimensional case.

Theorem 7. Let $k \subsetneq A \subsetneq B := k[[x, y]]$ be a noetherian complete (with respect to its maximal ideal) local qfc subring of B, the power series ring in two variables. Then A is isomorphic to a power series ring in one variable over k.

Proof. By Corollary 6.7, *A* is a UFD. Let $p \in A$ be a prime element. By Corollary 6.4, *p* is a prime element in *B*, and since *A* is qfc in *B* we have $pB \cap A = pA$. This gives an inclusion of integral domains $A/pA \subseteq B/pB$. Now dim B/pB = 1, and hence any two elements in B/pB are analytically dependent. Thus, dim $A/pA \leq 1$. Now dim $A \leq 2$. If dim A = 1, then *A* is clearly isomorphic to a power series ring in one variable over *k*.

Now assume that dim A = 2. We will show that A = k[[x, y]]. For that, choose any two relatively prime elements u, v from the maximal ideal of A. By Corollaries 6.4 and 6.6, u and v are nonassociate prime elements of k[[x, y]]. So, in particular, the extension of the maximal ideal of A to B is (x, y)-primary. Since A, B are complete, we infer that B is integral over A. But then by Theorem 6 A must be equal to k[[x, y]].

Next we consider the case when dim B = 3.

Theorem 8. Let $k \subseteq A \subseteq B := k[[X, Y, Z]]$, where A is a 2-dimensional noetherian complete (with respect to its maximal ideal) local qfc subring of B. Then A is isomorphic to a power series ring in two variables over k.

Proof. The proof is similar to the proof of Theorem 7.

Note that *A* is a UFD by Corollary 6.7.

By Brieskorn's theorem [1968], either A is isomorphic to a power series in two variables over k, or $A \cong k[[u, v, w]]/(u^2 + v^3 + w^5)$. We have to show that A cannot be isomorphic to $k[[u, v, w]]/(u^2 + v^3 + w^5)$. By the argument in the proof of Theorem 7, the extended ideal $(\overline{u}, \overline{v}, \overline{w})B$ has height > 1. We know that A is the ring of invariants of the binary icosahedral group of order 120 acting on a power series ring k[[s, t]]. The morphism Spec $k[[s, t]] \setminus \{(s, t)\} \rightarrow$ Spec $A \setminus \{(\overline{u}, \overline{v}, \overline{w})\}$ is finite unramified. Since Spec $B \setminus V((\overline{u}, \overline{v}, \overline{w}))$ is simply connected, by covering space theory we have a factorization

Spec
$$B \setminus V((\overline{u}, \overline{v}, \overline{w})) \to$$
 Spec $k[[s, t]] \setminus \{(s, t)\} \to$ Spec $A \setminus \{(\overline{u}, \overline{v}, \overline{w})\}$.

By Hartog's theorem, we have $A \subset k[[s, t]] \subseteq B$. But then A is not algebraically closed in B, contradicting Theorem 6. This shows that A is isomorphic to a power series ring in two variables over k.

Question. In Theorem 8, is the assumption dim A = 2 necessary, i.e., can a proper qfc subring of k[[x, y, z]] have dimension > 2?

It is well-known that, if $A \subseteq B$ are affine normal domains over an algebraically closed field of characteristic 0 such that Q(A) is algebraically closed in Q(B), then a general fiber of the morphism Spec $B \rightarrow$ Spec A is irreducible. By Theorem 6, if $A \subseteq B$ are complete normal domains over an algebraically closed field of characteristic 0 such that A is qfc in B, then Q(A) is algebraically closed in Q(B). In view of the above observation we can ask the following question:

Question. Let (V, p), (W, q) be normal complex analytic germs and $f: (W, q) \rightarrow (V, p)$ a complex analytic morphism such that the analytic local ring of V is algebraically closed in that of W. Is a general fiber of f irreducible?

We have the following modest result as an affirmative answer to this question:

Theorem 9. Let (W, q) be a normal complex analytic germ and $f : W \to \mathbb{C}$ a complex analytic morphism of germs. Assume that the ring $\mathbb{C}\{f\} \subset \mathbb{O}_{W,q}$ is qfc. Then a general fiber of f is connected.

Proof. We will use a result of Tráng [1977] on the topology of singular points, which generalizes Milnor's results.

Tráng [1977] proved that there are positive numbers $0 < \delta \ll \epsilon \ll 1$ such that if D is a disc of radius δ in \mathbb{C} then the morphism $B_{\epsilon} \cap W \cap f^{-1}(D - \{0\}) \rightarrow D - \{0\}$ is a topological fiber bundle, where B_{ϵ} is a ball of radius ϵ with center q in \mathbb{C}^n , such that $(W, q) \subseteq (\mathbb{C}^n, 0)$ is a closed embedding of germs. Since $\mathbb{C}\{f\} \subset \mathbb{O}_{W,q}$ is qfc, the fiber $\{f = 0\}$ is irreducible by Corollary 6.4. We have a long exact sequence of homotopy groups

$$\pi_1(F) \longrightarrow \pi_1(B_{\epsilon} \cap W \cap f^{-1}(D - \{0\})) \longrightarrow \pi_1(D - \{0\}) \longrightarrow \pi_0(F)$$
$$\longrightarrow \pi_0(B_{\epsilon} \cap W \cap f^{-1}(D - \{0\})) \longrightarrow \pi_0(D - \{0\}) \longrightarrow (1).$$

Here *F* is a general fiber of $B_{\epsilon} \cap W \cap f^{-1}(D - \{0\}) \to D - \{0\}$. Both $B_{\epsilon} \cap W \cap f^{-1}(D - \{0\})$ and $D - \{0\}$ are connected. Since $\{f = 0\}$ is reduced and irreducible, a small transverse loop in $B_{\epsilon} \cap W \cap f^{-1}(D - \{0\})$ maps onto the generator of the fundamental group of $D - \{0\}$, hence the homomorphism

$$\pi_1(B_{\epsilon} \cap W \cap f^{-1}(D - \{0\})) \to \pi_1(D - \{0\})$$

is surjective. It follows that *F* is connected, and this proves the result.

The next result is an interesting consequence of the property of being factorially closed. To state the result, we need a definition. Let $f: Y \to X$ be a dominant morphism of smooth algebraic varieties such that the general fibers are irreducible and reduced. Then there exist an open immersion $\iota: Y \hookrightarrow W$ and a projective morphism $\overline{f}: W \to X$ such that $f = \overline{f} \circ \iota$, where W is a smooth algebraic variety and $D := W \setminus Y$ is a divisor with simple normal crossings. Let $D = D_1 + \cdots + D_r$ be the irreducible decomposition. We further assume that D intersects transversally the fiber $F_P = \overline{f}^{-1}(P)$ for every closed point $P \in X$. We say that \overline{f} is an SNCcompletion of f. Suppose that for every $P \in X$ and every $1 \le i \le r$ the intersection $D_i \cdot F_P$ is irreducible and reduced. If there exists such an SNC-completion of f, we say that f is *fiberwise integral at infinity*. If there is an open set U of X such that $f: f^{-1}(U) \to U$ has a completion which is fiberwise integral at infinity, then we say that f is generically fiberwise integral at infinity. This condition is equivalent to saying that the generic fiber Y_{η} , with η the generic point of X, can be embedded into a projective smooth variety W_{η} defined over the field $k(\eta)$ in such a way that $D_{\eta} = W_{\eta} \setminus Y_{\eta}$ is a divisor consisting of geometrically integral smooth components with simple normal crossings.

Theorem 10. Let A be an affine domain of dimension 1 over k. Assume that A is fc in a regular affine domain R over k. Let X = Spec A, Y = Spec R and $f : Y \to X$ be the induced morphism. Assume that f has an SNC-completion which is generically fiberwise integral at infinity. Then there is a maximal ideal \mathfrak{m} of A such that the affine domain $R/\mathfrak{m}R$ is regular and has no nontrivial units.

Proof. Since A is fc in R and R is normal, it follows that A is normal, and hence regular as dim A = 1. A general fiber of the morphism f is reduced and irreducible. In particular, by Bertini's theorem, R/mR is a regular affine domain for all but finitely many maximal ideals in A. Removing from X the closed points corresponding to these maximal ideals, we may assume that f is a smooth morphism. Let $\overline{f}: W \to X$ be an SNC-completion which is fiberwise integral at infinity. Here we may have to replace X by a suitable open set. Let $D = W \setminus Y$ be the divisor at infinity and let $D = D_1 + D_2 + \cdots + D_r$ be the irreducible decomposition of D. If the result is not true, then we may assume that R/mR has a nontrivial unit for every maximal ideal m of A. Note that, by definition, each D_i meets each fiber F_P of \overline{f} transversally and the intersection $D_i \cdot F_P$ is integral, i.e., irreducible and reduced.

Let P be a closed point of X. The fiber $f^{-1}(P)$ has a nonconstant unit u_P , and the divisor (u_P) in $F_P := \overline{f}^{-1}(P)$ has the form $\sum_i a(P)_i D_i|_{F_P}$ with $a(P)_i \in \mathbb{Z}$. Note that the subgroup $\sum_i \mathbb{Z}D_i$ of Pic(W), which is generated by the irreducible components of D, is a countable group. Choosing a nonconstant unit u_P for every $P \in X(k)$, we have a mapping $P \mapsto (u_P)$ from X(k) to the group $\sum_i \mathbb{Z}D_i$, where X(k) is the set of closed points of X and (u_P) is identified with $\sum_i a(P)_i D_i$. Since each $D_i \cap F_P$ is irreducible and reduced for each *i*, such an identification is possible. Then we can find a fixed divisor $D_0 = \sum_i a_i D_i$ and an infinite set Λ of X(k) such that $D_0 \cdot F_P = (u_P)$ for each P of A. This means that the line bundle $\mathbb{O}(D_0)$ on W restricts to a trivial line bundle on F_P for each $P \in \Lambda$. By the upper-semicontinuity theorem [Hartshorne 1977, Chapter III, Theorem 12.8], the set of points in X such that the restriction of D_0 to F_P is trivial is a closed subvariety T of X containing the infinite set A. (Use the theorem for \mathscr{L} and \mathscr{L}^{-1} so that $\dim_k H^0(F_P, \mathscr{L}|_{F_P}) \geq 0$ and $\dim_k H^0(F_P, \mathcal{L}^{-1}|_{F_P}) \ge 0.)$ Since dim X = 1, T = X and D_0 restricts to a trivial line bundle on every fiber of \overline{f} . By [Hartshorne 1977, Chapter III, Exercise 12.4], D_0 is linearly equivalent to the pullback by \overline{f} of a divisor of the form $\sum_{j=1}^{s} b_j Q_j$ on X. Thus, the restriction of D_0 to $\overline{f}^{-1}(X \setminus \{Q_1, \ldots, Q_s\})$ is linearly equivalent to zero. Write D_0 as the divisor of a rational function (φ) on $\overline{f}^{-1}(X \setminus \{Q_1, \ldots, Q_s\})$. Then φ gives a nonconstant unit of $f^{-1}(X \setminus \{Q_1, \ldots, Q_s\})$. Since the units on $f^{-1}(X \setminus \{Q_1, \ldots, Q_s\})$ and $X \setminus \{Q_1, \ldots, Q_s\}$ are the same by the assumption of factorial closedness, φ is constant on each fiber of f. However, φ restricts onto the unit u_P up to a nonzero constant for every $P \in \Lambda$. This is a contradiction because u_P is not a constant.

Without the assumption that f has an SNC-completion which is fiberwise integral at infinity, Theorem 10 does not hold.

Example. Let *W* be the Hirzebruch surface $\mathbb{P}^1 \times \mathbb{P}^1$ with vertical and horizontal \mathbb{P}^1 fibrations. Let $p_1 : W \to \mathbb{P}^1$ be the vertical one with a fiber *L*, and let the horizontal one p_2 be given by a linear system |M|. Let D_1 be an irreducible curve such that $D_1 \sim 2M + L$. Then the restriction $p_1|_{D_1} : D_1 \to \mathbb{P}^1$, being a double covering, has two branch points. Let L_1, L_2 be two fibers of p_1 over these branch points. Let $D = D_1 + L_1 + L_2$, and let $Y := W \setminus D$ and $X := \mathbb{P}^1 - \{\text{two branch points}\}$. Let $f = p_1|_Y$. Then every fiber of $f : Y \to X$ is irreducible; hence k(X) is algebraically closed in k(Y) and it is easy to see that *A* is factorially closed in *R*, where X = Spec A and Y = Spec R, because $A = k[t, t^{-1}]$ and every prime element of *A* is t - c with some nonzero constant $c \in k$. Then the fiber over t = cis irreducible. Hence t - c is a prime element in *R*. Furthermore, the units of *R* are the same as the units of *A* because the only linear relation among the components of *D* is the one between L_1 and L_2 . But every closed fiber of *f* has a nontrivial unit because it is isomorphic to \mathbb{A}^1_* . Note that *R* is not factorial since $\text{Pic}(R) = \mathbb{Z}/2\mathbb{Z}$.

Remark. In this example, the fibration $f : Y \to X$ is a twisted \mathbb{A}^1_* -fibration. Let X' be the curve D_1 with two ramifying points for $p_1|_{D_1}$ removed, and let $f' : Y' \to X'$ be the base change of f by $X' \to X$. Then $Y' \cong A^1_* \times \mathbb{A}^1_*$. Write Y = Spec R, X = Spec A and X' = Spec A'. Then A is fc in R, but A' is not fc in $R' := R \otimes_A A'$. In fact, $R'^*/k^* \cong \mathbb{Z} \times \mathbb{Z}$ and $A'^*/k^* \cong \mathbb{Z}$. If A' were fc in R', then we must have $R'^* = A'^*$. Note that A'/A is a finite étale extension. Hence the factorial closedness is not preserved even by an étale base change.

Using Theorem 10 we can now give a very short proof of a result of [Neumann and Norbury 1998].

Theorem 11. Let $f, g \in \mathbb{C}[X, Y]$ be a pair of polynomials in two variables with nonzero constant Jacobian determinant. Suppose that the following conditions are satisfied:

- (a) For all $c \in \mathbb{C}$, the polynomial f c is irreducible and defines a rational curve.
- (b) Let C² ⊂ Y be an open embedding in a smooth quasiprojective surface such that f : C² → C extends to a proper morphism Y → C and Y \ C² is a simple normal crossing divisor such that each irreducible component of Y \ C² is a cross-section of the morphism Y → C.

Then $\{f = 0\} \cong \mathbb{C}$ *, and hence the Jacobian Conjecture is true for the pair* (f, g)*.*

Remark. In [Neumann and Norbury 1998] *f* is called a simple rational polynomial.

Proof. By assumption, f - c is an irreducible polynomial for all constants c. Also, \mathbb{C}^2 has no nonconstant invertible regular functions. Hence the morphism $\mathbb{C}^2 \to \mathbb{C}$ is an fc morphism. Condition (b) implies that $f : \mathbb{C}^2 \to \mathbb{C}$ is generically fiberwise integral at infinity. By Theorem 10 and condition (a), for all but finitely many $c \in \mathbb{C}$ the affine curve $\{f = c\}$ is smooth rational irreducible with no nonconstant invertible regular functions. Hence it is isomorphic to \mathbb{C} . By the Abhyankar–Moh–Suzuki theorem, after a suitable automorphism of $\mathbb{C}[X, Y]$ the polynomial f is mapped onto X. It is well-known that this implies that the Jacobian Conjecture is true for (f, g).

The next result is another interesting example of qfc subrings.

Theorem 12. Let $A = \mathbb{C}\{z_1, z_2, ..., z_n\}/P$ be an analytic local domain which is a *UFD*. Then A is qfc in \hat{A} .

Proof. We use Artin's approximation theorem [1968].

It is known that \hat{A} is also a UFD. To show that A is qfc in \hat{A} , it follows easily from the definition of a qfc subring that it is enough to show that any prime element of A remains a prime element in \hat{A} .

Suppose that $f \in A$ is a prime element. Assume that there are non-units g, h in \hat{A} such that f = gh. Let P be generated by f_1, f_2, \ldots, f_r . Let w_1, w_2, \ldots, w_r , w_{r+1}, w_{r+2} be new indeterminates. Consider the system of equations in the variables $z_1, \ldots, z_n, w_1, w_2, \ldots, w_{r+2}$

$$f_1 - w_1 = 0 = f_2 - w_2 = \dots = f_r - w_r = f - w_{r+1}w_{r+2}$$

This system has solutions $w_1 = f_1, \ldots, w_r = f_r, w_{r+1} = g, w_{r+2} = h$ in $\mathbb{C}[[z_1, \ldots, z_n]]$. By Artin's theorem, we can find solutions g_0, h_0 in $\mathbb{C}\{z_1, \ldots, z_n\}$ such that $f = g_0 h_0$ modulo P and g_0, h_0 approximate g, h to any order. In particular, g_0 and h_0 cannot be units. Thus, every prime element in A remains a prime element in \hat{A} , and consequently A is qfc in \hat{A} .

Corollary 12.1. Any element of $\mathbb{C}[[z_1, \ldots, z_n]]$ which is algebraic over $\mathbb{C}\{z_1, \ldots, z_n\}$ is itself convergent. (Here, z_1, z_2, \ldots, z_n are indeterminates over \mathbb{C} .)

Question. Is Theorem 12 valid without assuming that A is a UFD?

3. Examples

We give some examples which shed more light on fc and qfc extensions.

(1) The following example shows that a qfc extension $A \subseteq B$ of local domains can be quite strange if A is not complete.

Let *k* be any field. Consider $A := k[x, e^x - 1]_{(x, e^x - 1)}$ and B := k[[x]]. Then $A \subseteq B$ is a local inclusion of local UFDs. Note that *A* is a regular local ring of dimension 2, whereas *B* has only one nonzero prime ideal, namely (*x*). From these observations we can easily deduce the following properties:

- (a) Infinitely many primes of A split in B.
- (b) Infinitely many distinct primes of A become associates in B.
- (c) The dimension of A is bigger than the dimension of B.
- (d) A prime element of *B*, namely *x*, divides infinitely many distinct prime elements of *A*.
- (2) The properties of being a qfc extension and a flat extension are independent.

For, a ring of invariants A of a semisimple group acting on a polynomial ring B is fc in B, but the extension is not flat in general.

On the other hand, the extension $k[t^2] \subseteq k[t]$ is flat but not qfc.

(3) For an extension of normal affine domains $A \subseteq B$, the set of points $\mathfrak{m} \in \operatorname{Max} B$ such that $A_{\mathfrak{m}\cap A} \subseteq B_{\mathfrak{m}}$ is qfc is in general not Zariski-open in Max *B*.

An example of this is the inclusion $A := k[x] \subseteq B := k[x, y, z]/(xy - z^2)$. If \mathfrak{m}_0 is the maximal ideal corresponding to the origin (0, 0, 0) in B, then $A_{(x)}$ is qfc in $B_{\mathfrak{m}_0}$, but for maximal ideals corresponding to nearby points $(0, \lambda, 0)$ this is not true.

Remark. One may ask a similar question for fc extensions. But at least in the case of affine domains it is not very interesting, for then by Lemma 4 $A_{m\cap A} = B_m$. So *A* and *B* must be birational where the set of such points is clearly open.

(4) The ring extension $A := k[xy] \subseteq B := k[x, y]$ is such that A is algebraically closed in B and $A^* = B^*$, but the fc locus $\mathfrak{FC}(A : B)$ is empty.

(5) The ring extension $A := k[x] \subset B := k[x, y, z]/(x^2 + y^2 + z^2 - 1)$ has the property that any irreducible element of A remains irreducible in B, the extension is faithfully flat and both rings have same units, but A is not fc in B. Note that B is not factorial.

(6) Let $A := k[x, xy] \subset B := k[x, y]$, where x, y are indeterminates. Then $Q(A) \subset Q(B)$ is maximally algebraic. Any element of the form x + axy is a prime element in A but not a prime element in B, where $a \in k^*$.

If q is any prime ideal in A other than (x, xy), then either x or xy is a unit in A_q . Hence both x, y are units in B_q . It follows that any prime element in A_q is either a prime element in B_q or a unit in B_q . This shows that $\mathfrak{FC}(A : B) = \operatorname{Spec} A \setminus \{(x, xy)\}$ is nonempty and open, but infinitely many prime elements in A are non-units in B and are not prime elements in B.

If p is any height-1 prime ideal in A, then at least one of x, xy does not lie in p. Hence, in B_p , x is always a prime element. From this we see that $A_p \subset B_p$ is qfc.

Clearly A is not fc in B. Since A, B are UFDs and have the same units, A is not qfc in B.

This shows that the local analogue of Lemma 1(6) does not hold for the qfc property.

4. Open problems

- (1) Let A ⊆ B be normal complete local domains over k such that A is qfc in B. Is the power series ring in one variable A [[x]] qfc in B [[x]]?
- (2) Suppose that $A \subseteq B$ are normal affine domains such that for any maximal ideal $\mathfrak{m} \subset A$ the extension $A_{\mathfrak{m}} \subseteq B_{\mathfrak{m}}$ is qfc. Is A qfc in B?
- (3) Let $A \subseteq B$ be an qfc inclusion of normal complete domains over k. Is dim $A \leq \dim B$?
- (4) Is any fc subring of a PID also a PID?

Acknowledgments

Chakraborty is supported by the Council of Scientific and Industrial Research of India under the Shyama Prasad Mukherjee Fellowship.

We are thankful to S. M. Bhatwadekar for correcting the statement of Theorem 5 and suggesting example (6) in Section 3.

References

- [Artin 1968] M. Artin, "On the solutions of analytic equations", *Invent. Math.* **5** (1968), 277–291. MR 38 #344 Zbl 0172.05301
- [Brieskorn 1968] E. Brieskorn, "Rationale Singularitäten komplexer Flächen", *Invent. Math.* **4** (1968), 336–358. MR 36 #5136 Zbl 0219.14003
- [Gupta 2014] N. Gupta, "On Zariski's cancellation problem in positive characteristic", *Adv. Math.* **264** (2014), 296–307. MR 3250286
- [Hartshorne 1977] R. Hartshorne, *Algebraic geometry*, Graduate Texts in Mathematics **52**, Springer, New York-Heidelberg, 1977. MR 57 #3116 Zbl 0367.14001
- [Miyanishi 1986] M. Miyanishi, "Normal affine subalgebras of a polynomial ring", pp. 37–51 in *Algebraic and topological theories* (Kinosaki, 1984), edited by M. Nagata et al., Kinokuniya, Tokyo, 1986. MR 1102251 Zbl 0800.14018
- [Nagata 1965] M. Nagata, *Lectures on the fourteenth problem of Hilbert*, Tata Institute of Fundamental Research, Bombay, 1965. MR 35 #6663 Zbl 0182.54101
- [Neumann and Norbury 1998] W. D. Neumann and P. Norbury, "Nontrivial rational polynomials in two variables have reducible fibres", *Bull. Austral. Math. Soc.* **58**:3 (1998), 501–503. MR 99j:14013 Zbl 0946.14038
- [Samuel 1964] P. Samuel, *Lectures on unique factorization domains*, TIFR **30**, Tata Institute of Fundamental Research, Bombay, 1964. MR 35 #5428 Zbl 0184.06601
- [Tráng 1977] L. D. Tráng, "Some remarks on relative monodromy", pp. 397–403 in *Real and complex singularities* (Oslo, 1976), edited by P. Holm, Sijthoff and Noordhoff, Alphen aan den Rijn, 1977. MR 57 #16296 Zbl 0428.32008
- [Tráng 2008] L. D. Tráng, "Simple rational polynomials and the Jacobian conjecture", *Publ. Res. Inst. Math. Sci.* **44**:2 (2008), 641–659. MR 2009e:14111 Zbl 1152.14050

Communicated by Kei-Ic	hi Watanabe	
Received 2014-12-22	Revised 2015-05-04	Accepted 2015-05-09
sagnik@math.tifr.res.in	School of Ma Tata Institute Homi-Bhabha	thematics, of Fundamental Research, Road, Navy Nagar, Mumbai 400005, India
gurjar@math.tifr.res.in	School of Ma Tata Institute Homi-Bhabha	thematics, of Fundamental Research, Road, Navy Nagar, Mumbai 400005, India
miyanisi@amber.plala.or.	p Research Centr University, 2-1	er for Mathematical Sciences, Kwansei Gakuin , Gakuen, Sanda 669-1337, Japan



1161



Coherent analogues of matrix factorizations and relative singularity categories

Alexander I. Efimov and Leonid Positselski

We define the triangulated category of relative singularities of a closed subscheme in a scheme. When the closed subscheme is a Cartier divisor, we consider matrix factorizations of the related section of a line bundle, and their analogues with locally free sheaves replaced by coherent ones. The appropriate exotic derived category of coherent matrix factorizations is then identified with the triangulated category of relative singularities, while the similar exotic derived category of locally free matrix factorizations is its full subcategory. The latter category is identified with the kernel of the direct image functor corresponding to the closed embedding of the zero locus and acting between the conventional (absolute) triangulated categories of singularities. Similar results are obtained for matrix factorizations of infinite rank; and two different "large" versions of the triangulated category of relative singularities, corresponding to the approaches of Orlov and Krause, are identified in the case of a Cartier divisor. A version of the Thomason-Trobaugh-Neeman localization theorem is proven for coherent matrix factorizations and disproven for locally free matrix factorizations of finite rank. Contravariant (coherent) and covariant (quasicoherent) versions of the Serre-Grothendieck duality theorems for matrix factorizations are established, and pull-backs and push-forwards of matrix factorizations are discussed at length. A number of general results about derived categories of the second kind for curved differential graded modules (CDG-modules) over quasicoherent CDGalgebras are proven on the way. Hochschild (co)homology of matrix factorization categories are discussed in an appendix.

Introduction	
1. Exotic derived categories of quasicoherent CDG-modules	1165
1.1. CDG-rings and CDG-modules	1165
1.2. Quasicoherent CDG-algebras	1165

Intro du ation

MSC2010: primary 14F05; secondary 13D09, 16G99.

Keywords: matrix factorizations, relative singularities of Cartier divisors, triangulated categories of singularities, derived categories of the second kind, coderived categories, direct and inverse images, covariant Serre–Grothendieck duality, localization theory,

1.3.	Derived categories of the second kind	1167
1.4.	Finite flat dimension theorem	
1.5.	Fully faithful embedding	1173
1.6.	Finite homological dimension theorem	1176
1.7.	Gorenstein case	1181
1.8.	Pull-backs and push-forwards	1182
1.9.	Morphisms of finite flat dimension	1186
1.10.	Supports of CDG-modules	1189
2. Tr	iangulated categories of relative singularities	1192
2.1.	Relative singularity category	1192
2.2.	Matrix factorizations	1194
2.3.	Exotic derived categories of matrix factorizations	1195
2.4.	Regular and Gorenstein scheme cases	1196
2.5.	Serre-Grothendieck duality	1198
2.6.	w-flat matrix factorizations	1205
2.7.	Main theorem	1206
2.8.	Infinite matrix factorizations	1214
2.9.	Stable derived category	1218
2.10.	Relative stable derived category	1222
3. Si	ipports, pull-backs, and push-forwards	1225
3.1.	Supports	1225
3.2.	Locality of local freeness	1228
3.3.	Nonlocalization of local freeness	1233
3.4.	Pull-backs and push-forwards in singularity categories	1236
3.5.	Push-forwards of matrix factorizations	1238
3.6.	Push-forwards for morphisms of finite flat dimension	1240
3.7.	Duality and push-forwards	1244
3.8.	Duality and pull-backs	1247
Appen	dix A. Quasicoherent graded modules	1249
A.1.	Flat quasicoherent sheaves	1249
A.2.	Locally projective quasicoherent graded modules	1250
A.3.	Injective quasicoherent graded modules	1250
Appen	Appendix B. Hochschild (co)homology of matrix factorizations	
B.1.	Locally free matrix factorizations of finite rank	1253
B.2.	Coherent matrix factorizations	1264
Acknowledgments		1289
References		1289

Introduction

A matrix factorization of an element w in a commutative ring R is a pair of square matrices (Φ, Ψ) of the same size, with entries from R, such that both the products $\Phi\Psi$ and $\Psi\Phi$ are equal to w times the identity matrix. In the coordinate-free language, a matrix factorization is a pair of finitely generated free R-modules M^0 and M^1 together with R-module homomorphisms $M^0 \to M^1$ and $M^1 \to M^0$ such that both the compositions $M^0 \to M^1 \to M^0$ and $M^1 \to M^0 \to M^1$ are equal to the multiplication with w. Matrix factorizations were introduced by Eisenbud [1980] and used by Buchweitz [1986] for the study of the maximal Cohen–Macaulay modules over hypersurface local rings.

Another name for this notion is "D-branes in the Landau–Ginzburg B model" (as suggested by Kontsevich) [Kapustin and Li 2003]; in this context, the element w is called the *potential*. One generalizes the above definition, replacing free modules with projective modules [Kapustin and Li 2003; Orlov 2004], with locally free sheaves [Orlov 2012], and finally with coherent sheaves [Lin and Pomerleano 2013]. The importance of the latter generalization is emphasized in the present paper.

Being particular cases of curved DG-modules over a curved DG-ring [Kapustin and Li 2003; Positselski 2011b], matrix factorizations form a DG-category. So one can consider the corresponding category of closed degree-zero morphisms up to chain homotopy, which is a triangulated category. Generally speaking, however, the homotopy category is "too big" for most purposes, and one would like to pass from it to an appropriately defined derived category. One can use the homotopy category in lieu of the derived one when dealing with projective modules [Kapustin and Li 2003; Orlov 2004]; for locally free matrix factorizations over a nonaffine scheme, there is an option of working with the quotient category of the homotopy category by the locally contractible objects [Polishchuk and Vaintrob 2011, Definition 3.13]. When dealing with coherent (analogues of) matrix factorizations, having some kind of derived category construction is apparently unavoidable.

The relevant concept of a derived category is that of the derived category of the second kind, as developed in [Positselski 2010; 2011b]. There are several versions of this notion; the appropriate one for quasicoherent sheaves is called the *coderived category* and for coherent sheaves it is the *absolute derived category*. The absolute derived category of locally free matrix factorizations was studied in [Orlov 2012]; for coherent matrix factorizations over a smooth variety, it was considered in [Lin and Pomerleano 2013]. These two absolute derived categories are equivalent for regular schemes, but can be different otherwise (as we show with an explicit counterexample).

The *triangulated category of singularities* of a Noetherian scheme was defined by D. Orlov [2004] as the quotient category of the bounded derived category of coherent sheaves by its full triangulated subcategory of perfect complexes, i.e., the objects locally presentable as finite complexes of locally free sheaves. This triangulated category vanishes if and only if the Noetherian scheme is regular. It was shown in [Orlov 2004, Theorem 3.9], under mild assumptions on an affine regular Noetherian scheme X and a potential (regular function) w on it, that the homotopy category of locally free matrix factorizations of w over X is equivalent to the triangulated category of singularities of the zero locus X_0 of w in X.

Orlov [2012] showed that the affineness assumption on X can be dropped in this result if one replaces the homotopy category of locally free matrix factorizations with their absolute derived category. He also considers the general case of a nonaffine singular scheme X, for which he obtains a fully faithful functor from the absolute derived category of locally free matrix factorizations over X to the triangulated category of singularities of X_0 . The problem of studying the difference between these two triangulated categories was posed in the introduction to [Orlov 2012].

The first aim of the present paper is to provide an alternative proof of these results of Orlov for regular schemes, an alternative generalization of them to singular schemes, and a more precise version of Orlov's original generalization. We replace the triangulated category at the source of Orlov's fully faithful functor by a "larger" category (containing the original one) and the triangulated category at the target by a "smaller" category (a quotient of the original one), thereby transforming this functor into an equivalence of triangulated categories. We also describe the image of Orlov's fully faithful functor as the kernel of a certain other triangulated functor.

More precisely, we show that the absolute derived category of coherent matrix factorizations of w over X is equivalent to what we call the *triangulated category of singularities of* X_0 *relative to* X. The latter category is a certain quotient category of the triangulated category of singularities of X_0 ; it measures, roughly speaking, how much worse are the singularities of X_0 compared to those of X. As to the image of Orlov's fully faithful embedding, it consists precisely of those objects of the conventional (absolute) triangulated category of singularities of X_0 whose direct images vanish in the triangulated category of singularities of X.

The paper consists of three sections and two appendices. In Section 1, we prove three rather general technical assertions about derived categories of the second kind for curved differential graded modules (CDG-modules) over a quasicoherent CDG-algebra with a restriction on the homological dimension. One of them, claiming that certain embeddings of DG-categories of CDG-modules induce equivalences of the derived categories of the second kind, is a generalization of [Polishchuk and Positselski 2012, Theorem 3.2] based on a modification of the same argument, originally introduced for the proof of [Positselski 2010, Theorem 7.2.2].

The idea of the proof of the other assertion, according to which certain natural functors between derived categories of the second kind are fully faithful, is new.

1162

The third technical assertion explains when the coderived category coincides with the absolute derived category of the same class of CDG-modules: e.g., for the locally projective CDG-modules this is true.

A version of (the former two of) these results is used in Section 2 to extend Orlov's cokernel functor from the absolute derived category of locally free matrix factorizations to the absolute derived category of coherent ones. This extension of the cokernel functor admits a simple construction of a functor in the opposite direction, suggested in [Lin and Pomerleano 2013]. We use these constructions to obtain a new proof of Orlov's theorem, and our own generalization of it to the singular case.

When X is regular, Orlov's and our results amount to the same assertion since the absolute derived categories of locally free and coherent matrix factorizations are equivalent by our Theorem 1.4. When X is singular, the natural functor between these two absolute derived categories is fully faithful by our Proposition 1.5, and Orlov's full-and-faithfulness theorem follows from ours by virtue of an appropriate semiorthogonality property.

We also compare a "large" version of the triangulated category of relative singularities with the coderived category of quasicoherent matrix factorizations, strengthening some results of Polishchuk and Vaintrob [2011]. A "large" version of the absolute triangulated category of singularities, defined by Orlov [2004], is identified with H. Krause's stable derived category [2005] in the case of a divisor in a regular scheme. A similar result is proven in the case of a Cartier divisor in a singular scheme, where we extend Krause's theory by defining the *relative stable derived category*. For any closed subscheme of finite flat dimension in a separated Noetherian scheme, the relative stable derived category is compactly generated by its full triangulated subcategory equivalent to the triangulated category of relative singularities.

The homotopy categories of unbounded complexes of projective modules over a ring and injective quasicoherent sheaves over a scheme were studied by Jørgensen [2005] and Krause [2005]; subsequently, Iyengar and Krause [2006] constructed an equivalence between these two categories for rings with dualizing complexes. These results were extended to quasicoherent sheaves over schemes by Neeman [2008] and Murfet [2007], who found a way to define a replacement of the homotopy category of (nonexistent) projective sheaves in terms of the flat ones. The equivalence between these two categories is a covariant version of Serre–Grothendieck duality [Hartshorne 1966]. It is also very similar to the derived comodule-contramodule correspondence theory, developed by the second author [Positselski 2010; 2011b].

Serre–Grothendieck duality for matrix factorizations in the situation of a smooth variety X (and an isolated singularity of X_0) was studied in [Murfet 2013]. In this paper we extend the duality to matrix factorizations over much more general schemes X, constructing an equivalence between two "large" exotic derived categories, namely, the coderived category of flat (or locally free) matrix factorizations

of possibly infinite rank and the coderived category of quasicoherent matrix factorizations. Unless X is Gorenstein, this equivalence is *not* provided by the natural functor induced by the embedding of DG-categories, but rather differs from it in that the tensor product with the dualizing complex has to be taken along the way. A contravariant Serre duality in the form of an auto-antiequivalence of the absolute derived category of coherent matrix factorizations is also obtained.

There was some attention paid to pull-backs and push-forwards of matrix factorizations recently [Polishchuk and Vaintrob 2011; 2014; Dyckerhoff and Murfet 2013]. In Section 3, we approach this topic with our techniques, constructing the push-forwards of locally free matrix factorizations of infinite rank for any morphism of finite flat dimension between schemes of finite Krull dimension, and the pushforwards of locally free matrix factorizations of finite rank for any such morphism for which the induced morphism of the zero loci of w is proper. At the price of having to adjoin the images of idempotent endomorphisms, the preservation of finite rank under push-forwards is proven assuming only the support of the matrix factorization [Polishchuk and Vaintrob 2011] to be proper over the base.

Push-forwards of quasicoherent matrix factorizations are well-defined for any morphism of Noetherian schemes, and push-forwards of coherent matrix factorizations exist under properness assumptions similar to the above. A general study of category-theoretic and set-theoretic supports of quasicoherent and coherent CDG-modules is undertaken in this paper in order to obtain an independent proof of the preservation of coherence under the push-forwards not based on the passage to the triangulated categories of singularities.

The compatibility with pull-backs and push-forwards is an organic part of Serre–Grothendieck duality theory. The contravariant duality agrees with push-forwards of coherent sheaves (or matrix factorizations) with respect to proper morphisms [Hartshorne 1966], while the covariant duality transforms the conventional inverse image of flat matrix factorizations into the extraordinary inverse image of quasicoherent ones [Positselski 2012]. We use the latter result in order to construct the extraordinary inverse image functor of Hartshorne and Deligne, which is denoted by $f^{!}$ in [Hartshorne 1966] and which we denote by f^{+} , in the case of quasicoherent matrix factorizations.

Appendix A contains proofs of some basic facts about flat, locally projective, and injective quasicoherent graded modules which are occasionally used in the main body of the paper. Appendix B can be viewed as a complement to the paper [Polishchuk and Positselski 2012]. While Section B.1 contains some variations of and improvements on the results about Hochschild (co)homology of (C)DG-categories and (locally free) matrix factorizations in [loc. cit.], Section B.2 presents an alternative approach to the Hochschild (co)homology of coherent matrix factorizations based on the techniques developed in the main body of this paper.

1164

1. Exotic derived categories of quasicoherent CDG-modules

1.1. *CDG-rings and CDG-modules.* A *CDG-ring* (curved differential graded ring) B = (B, d, h) is defined as a graded ring $B = \bigoplus_{i \in \mathbb{Z}} B^i$ endowed with an odd derivation $d : B \to B$ of degree 1 and an element $h \in B^2$ such that $d^2(b) = [h, b]$ for all $b \in B$ and d(h) = 0. So one should have $d : B^i \to B^{i+1}$ and $d(ab) = d(a)b + (-1)^{|a|}ad(b)$; the brackets [-,-] denote the supercommutator $[a, b] = ab - (-1)^{|a||b|}ba$. The element *h* is called the *curvature element*.

A morphism of CDG-rings $B \to A$ is a pair (f, a), with a morphism of graded rings $f : B \to A$ and an element $a \in A^1$ such that $f(d_B b) = d_A f(b) + [a, f(b)]$ for all $b \in B$ and $f(h_B) = h_A + d_A a + a^2$. The composition of morphisms of CDG-rings is defined by the obvious rule $(f, a) \circ (g, b) = (f \circ g, a + f(b))$. The element *a* is called the *change-of-connection element*. A discussion of the origins of these definitions can be found in the paper [Positsel'skiĭ 1993], where the above terminology first appeared (see also an earlier paper [Getzler and Jones 1990], where the motivation was entirely different).

A *left CDG-module* $M = (M, d_M)$ over a CDG-ring B is a graded B-module endowed with an odd derivation $d_M : M \to M$ compatible with the derivation don B such that $d_M^2(m) = hm$ for all $m \in M$. Given a morphism of CDG-rings $(f, a) : B \to A$ and a CDG-module (M, d) over A, the CDG-module (M, d') over Bis defined by the rule d'(m) = d(m) + am.

Given graded left *B*-modules *M* and *N*, homogeneous *B*-module morphisms $f: M \to N$ of degree *n* are defined as homogeneous maps supercommuting with the action of *B*; i.e., $f(bm) = (-1)^{n|b|} bf(m)$. When *M* and *N* are CDG-modules, the homogeneous *B*-module morphisms $M \to N$ form a complex of abelian groups with the differential $d(f)(m) = d(f(m)) - (-1)^{|f|} f(d(m))$. The curvature-related terms cancel out in the computation of the square of this differential, so one has $d^2(f) = 0$. Therefore, left CDG-modules over *B* form a DG-category.

Two aspects of the above definitions are worth pointing out. First, the CDG-rings or modules have *no* cohomology modules, as their differentials do not square to zero. Second, given a CDG-ring B, there is *no* natural way to define a CDG-module structure on the free graded B-module B (though B is naturally a CDG-bimodule over itself, in the appropriate sense).

We refer the reader to [Positselski 2011b, Section 3.1] or [Positselski 2010, Sections 0.4.3–0.4.5] for more detailed discussions of the above notions. We will not need to consider any gradings different from \mathbb{Z} -gradings in this paper, though all the general results will be equally applicable in the Γ -graded situation in the sense of [Polishchuk and Positselski 2012, Section 1.1].

1.2. *Quasicoherent CDG-algebras.* Throughout this paper, unless specified otherwise, *X* is a separated Noetherian scheme with enough vector bundles; in other

1166

words, it is assumed that every coherent sheaf on X is the quotient sheaf of a locally free sheaf of finite rank. Note that the class of all schemes satisfying these conditions is closed under the passages to open and closed subschemes [Orlov 2004, Section 1.2] and contains all regular separated Noetherian schemes [Hartshorne 1977, Exercise III.6.8].

Recall the definition of a *quasicoherent CDG-algebra* from [Positselski 2011b, Appendix B]. A quasicoherent CDG-algebra \mathcal{B} over X is a graded quasicoherent \mathcal{O}_X -algebra such that for each affine open subscheme $U \subset X$, the graded ring $\mathcal{B}(U)$ is endowed with a structure of CDG-ring, i.e., a (not necessarily \mathcal{O}_X -linear) odd derivation $d : \mathcal{B}(U) \to \mathcal{B}(U)$ of degree 1 and an element $h \in \mathcal{B}^2(U)$. For each pair of embedded affine open subschemes $U \subset V \subset X$, an element $a_{UV} \in \mathcal{B}^1(U)$ is fixed such that the restriction morphism $\mathcal{B}(V) \to \mathcal{B}(U)$ together with the element a_{UV} form a morphism of CDG-rings. The obvious compatibility condition is imposed for triples of embedded affine open subschemes $U \subset V \subset W \subset X$.

A quasicoherent left CDG-module \mathcal{M} over \mathcal{B} is an \mathcal{O}_X -quasicoherent (or, equivalently, \mathcal{B} -quasicoherent) sheaf of graded left modules over \mathcal{B} together with a family of differentials $d : \mathcal{M}(U) \to \mathcal{M}(U)$ defined for all affine open subschemes $U \subset X$ such that $\mathcal{M}(U)$ is a CDG-module over $\mathcal{B}(U)$ and the appropriate compatibility condition holds with respect to the restriction morphisms of CDG-rings $\mathcal{B}(V) \to \mathcal{B}(U)$. Specifically, for a quasicoherent left CDG-module \mathcal{M} , one should have

$$d(s)|_U = d(s|_U) + a_{UV}s|_U$$
 for any $s \in \mathcal{M}(V)$

Quasicoherent left CDG-modules over a quasicoherent CDG-algebra \mathcal{B} form a DG-category [Positselski 2011b]. The complex of morphisms between CDG-modules \mathcal{N} and \mathcal{M} is the graded abelian group of homogeneous \mathcal{B} -module morphisms $f: \mathcal{N} \to \mathcal{M}$ with the differential d(f) defined locally as the supercommutator of f with the differentials in $\mathcal{N}(U)$ and $\mathcal{M}(U)$. We denote this DG-category by \mathcal{B} -qcoh.

We will call a quasicoherent graded algebra \mathcal{B} over X Noetherian if the graded ring $\mathcal{B}(U)$ is left Noetherian for any affine open subscheme $U \subset X$. Equivalently, \mathcal{B} is Noetherian if the abelian category of quasicoherent graded left \mathcal{B} -modules is a locally Noetherian Grothendieck category. In this case, the full DG-subcategory in \mathcal{B} -qcoh formed by CDG-modules whose underlying graded \mathcal{B} -modules are coherent (i.e., finitely generated over \mathcal{B}) is denoted by \mathcal{B} -coh.

Given a quasicoherent graded left \mathcal{B} -module \mathcal{M} and a quasicoherent graded right \mathcal{B} -module \mathcal{N} , one can define their tensor product $\mathcal{N} \otimes_{\mathcal{B}} \mathcal{M}$, which is a quasicoherent graded \mathcal{O}_X -module. A quasicoherent graded left \mathcal{B} -module \mathcal{M} is called *flat* if the functor $- \otimes_{\mathcal{B}} \mathcal{M}$ is exact on the abelian category of quasicoherent graded right \mathcal{B} -modules. Equivalently, \mathcal{M} is flat if the graded left $\mathcal{B}(U)$ -module $\mathcal{M}(U)$ is flat for any affine open subscheme $U \subset X$. The *flat dimension* of a quasicoherent graded module \mathcal{M} is the minimal length of its flat left resolution.

The full DG-subcategory in \mathcal{B} -qcoh formed by CDG-modules whose underlying graded \mathcal{B} -modules are flat is denoted by \mathcal{B} -qcoh_{fl}, and the full subcategory formed by CDG-modules whose underlying graded \mathcal{B} -modules have finite flat dimension is denoted by \mathcal{B} -qcoh_{ffd}. The similarly defined DG-categories of coherent CDG-modules are denoted by \mathcal{B} -coh_{fl} and \mathcal{B} -coh_{ffd}.

All the above DG-categories of quasicoherent CDG-modules (and the similar ones defined below in this paper) admit shifts and twists, and, in particular, cones. It follows that their homotopy categories $H^0(\mathcal{B}\text{-qcoh})$, $H^0(\mathcal{B}\text{-qcoh}_{fl})$, $H^0(\mathcal{B}\text{-coh})$, etc. are triangulated. Besides, to any finite complex (of objects and closed morphisms) in one of these DG-categories, one can assign its total object, which is an object of (i.e., a CDG-module belonging to) the same DG-category [Positselski 2011b, Section 1.2].

The DG-categories \mathcal{B} -qcoh and \mathcal{B} -qcoh_{fl} also admit infinite direct sums. Hence in these two DG-categories one can totalize even an unbounded complex by taking infinite direct sums along the diagonals.

The DG-category \mathcal{B} -qcoh also admits infinite products (which one can obtain using the coherator construction from [Thomason and Trobaugh 1990, Section B.14]), but these are not well-behaved (neither exact nor local), so we will not use them.

1.3. *Derived categories of the second kind.* The nonexistence of the cohomology groups for curved structures stands in the way of the conventional definition of the derived category of CDG-modules, which therefore does not seem to make sense. The suitable class of constructions of derived categories for CDG-modules is that of the derived categories of the second kind [Positselski 2010; 2011b].

Let \mathcal{B} be a quasicoherent CDG-algebra over X; assume that the quasicoherent graded algebra \mathcal{B} is Noetherian. Then a coherent CDG-module over \mathcal{B} is called *absolutely acyclic* if it belongs to the minimal thick subcategory of the homotopy category of coherent CDG-modules $H^0(\mathcal{B}\text{-coh})$ containing the total CDG-modules of all the short exact sequences of coherent CDG-modules over \mathcal{B} (with closed morphisms between them). The quotient category of $H^0(\mathcal{B}\text{-coh})$ by the thick subcategory of absolutely acyclic CDG-modules is called the *absolute derived category* of coherent CDG-modules over \mathcal{B} and denoted by $D^{abs}(\mathcal{B}\text{-coh})$ [Positselski 2011b].

For any quasicoherent CDG-algebra \mathcal{B} over X, a quasicoherent CDG-module over \mathcal{B} is called *coacyclic* if it belongs to the minimal triangulated subcategory of the homotopy category of quasicoherent CDG-modules $H^0(\mathcal{B}\text{-qcoh})$ containing the total CDG-modules of all the short exact sequences of quasicoherent CDG-modules over \mathcal{B} and closed under infinite direct sums. The quotient category of $H^0(\mathcal{B}\text{-coh})$ by the thick subcategory of coacyclic CDG-modules is called the *coderived category* of quasicoherent CDG-modules over \mathcal{B} and denoted by $D^{co}(\mathcal{B}\text{-qcoh})$ [Positselski 2010; 2011b]. Given an exact subcategory E in the abelian category of quasicoherent graded left \mathcal{B} -modules, one can define the *absolute derived category of left CDG-modules over* \mathcal{B} *with the underlying graded* \mathcal{B} -modules belonging to E as the quotient category of the corresponding homotopy category by its minimal thick subcategory containing the total CDG-modules of all the exact triples of CDG-modules with the underlying graded \mathcal{B} -modules belonging to E. The objects of the latter subcategory are called *absolutely acyclic with respect to* E (or with respect to the DG-category of CDG-modules with the underlying graded modules belonging to E) [Polishchuk and Positselski 2012].

In particular, one defines the absolute derived categories $D^{abs}(\mathcal{B}\text{-coh}_{ffd})$ and $D^{abs}(\mathcal{B}\text{-coh}_{fl})$ as the quotient categories of the homotopy categories $H^0(\mathcal{B}\text{-coh}_{ffd})$ and $H^0(\mathcal{B}\text{-coh}_{fl})$ by the thick subcategories of CDG-modules absolutely acyclic with respect to $\mathcal{B}\text{-coh}_{ffd}$ and $\mathcal{B}\text{-coh}_{fl}$, respectively.

When the exact subcategory E is closed under infinite direct sums, the thick subcategory of CDG-modules *coacyclic with respect to* E is the minimal triangulated subcategory of the homotopy category CDG-modules with the underlying graded modules belonging to E, containing the total CDG-modules of all the exact triples of CDG-modules with the underlying graded modules belonging to E and closed under infinite direct sums. The quotient category by this thick subcategory is called the *coderived category of left CDG-modules over* \mathcal{B} with the underlying graded *modules belonging to* E [Positselski 2010; Polishchuk and Positselski 2012].

Thus one defines the coderived category $D^{co}(\mathcal{B}\operatorname{-qcoh}_{fl})$ as the quotient categories of the homotopy category $H^0(\mathcal{B}\operatorname{-qcoh}_{fl})$ by the thick subcategory of CDG-modules coacyclic with respect to $\mathcal{B}\operatorname{-qcoh}_{fl}$. A little more care is needed for the definition of the coderived category $D^{co}(\mathcal{B}\operatorname{-qcoh}_{ffd})$ since the class of graded modules of finite flat dimension is not in general closed under infinite direct sums. An object $\mathcal{M} \in H^0(\mathcal{B}\operatorname{-qcoh}_{ffd})$ is said to be *coacyclic with respect to* $\mathcal{B}\operatorname{-qcoh}_{ffd}$ if there exists an integer $d \ge 0$ such that \mathcal{M} is coacyclic with respect to the exact category of quasicoherent CDG-modules of flat dimension at most d. The coderived category of quasicoherent CDG-modules of finite flat dimension is, by the definition, the quotient category of $H^0(\mathcal{B}\operatorname{-qcoh}_{ffd})$ by the above-defined thick subcategory of coacyclic CDG-modules [Polishchuk and Positselski 2012, Section 3.2].

Remark 1.3. One may wonder whether coacyclicity (absolute acyclicity) of quasicoherent CDG-modules (of a certain class) is a local notion. One general approach to this kind of problem is to consider the Mayer–Vietoris/Čech exact sequence

$$0 \longrightarrow \mathcal{M} \longrightarrow \bigoplus_{\alpha} j_{U_{\alpha}} * j_{U_{\alpha}}^* \mathcal{M} \longrightarrow \bigoplus_{\alpha < \beta} j_{U_{\alpha} \cap \mathcal{U}_{\beta}} * j_{U_{\alpha} \cap \mathcal{U}_{\beta}}^* \mathcal{M} \longrightarrow \cdots \longrightarrow 0$$

for a finite affine open covering U_{α} of X. Since the inverse and direct images with respect to affine open embeddings are exact and compatible with direct sums, they

1168

preserve coacyclicity (absolute acyclicity). Hence if the restrictions of \mathcal{M} to all U_{α} are coacyclic (absolutely acyclic), then so is \mathcal{M} itself.

Alternatively, one can base this kind of argument on the implications of the Noetherianness assumption, rather than the separatedness assumption. For this purpose, one replaces a quasicoherent CDG-module \mathcal{M} with its injective resolution (see Lemma 1.7(b)) before writing down its Čech resolution. In this approach, the covering need not be affine, as injective coacyclic objects are contractible, and direct images preserve contractibility; but it is important that the restrictions to open subschemes should preserve injectivity of quasicoherent graded \mathcal{B} -modules (see [Hartshorne 1966, Theorem II.7.18] and Theorem A.3; cf. [Thomason and Trobaugh 1990, Appendix B]).

When one is working with coherent CDG-modules, the Čech sequence argument is to be used in conjunction with Proposition 1.5 below. (Cf. Sections 1.10 and 3.2.)

1.4. *Finite flat dimension theorem.* The next theorem is our main technical result on which the proofs in Section 2 are based.

Though we generally prefer the coderived categories of (various classes of) infinitely generated CDG-modules over their absolute derived categories, technical considerations sometimes force us to deal with the latter (see Remark 1.5). Therefore, let $D^{abs}(\mathcal{B}\text{-qcoh}_{fl})$, $D^{abs}(\mathcal{B}\text{-qcoh}_{ffd})$, and $D^{abs}(\mathcal{B}\text{-qcoh})$ denote the absolute derived categories of (flat, of finite flat dimension, or arbitrary) quasicoherent CDG-modules over a quasicoherent CDG-algebra \mathcal{B} .

Theorem 1.4. (a) For any quasicoherent CDG-algebra \mathcal{B} over X, the functor

 $\mathsf{D}^{\mathsf{co}}(\mathcal{B}\operatorname{-qcoh}_{\mathsf{fl}}) \longrightarrow \mathsf{D}^{\mathsf{co}}(\mathcal{B}\operatorname{-qcoh}_{\mathsf{ffd}})$

induced by the embedding of DG-categories \mathcal{B} -qcoh_{fl} $\rightarrow \mathcal{B}$ -qcoh_{ffd} is an equivalence of triangulated categories.

(b) For any quasicoherent CDG-algebra \mathcal{B} over X, the functor

 $\mathsf{D}^{\mathsf{abs}}(\mathcal{B}\operatorname{-\mathsf{qcoh}}_{\mathsf{fl}})\longrightarrow\mathsf{D}^{\mathsf{abs}}(\mathcal{B}\operatorname{-\mathsf{qcoh}}_{\mathsf{ffd}})$

induced by the embedding of DG-categories \mathcal{B} -qcoh_{fl} $\rightarrow \mathcal{B}$ -qcoh_{fd} is an equivalence of triangulated categories.

(c) For any quasicoherent CDG-algebra B over X such that the underlying quasicoherent graded algebra B is Noetherian, the functor

$$\mathsf{D}^{\mathsf{abs}}(\mathcal{B}\operatorname{-coh}_{\mathsf{fl}}) \longrightarrow \mathsf{D}^{\mathsf{abs}}(\mathcal{B}\operatorname{-coh}_{\mathsf{ffd}})$$

induced by the embedding of DG-categories \mathcal{B} -coh_{fl} $\rightarrow \mathcal{B}$ -coh_{ffd} is an equivalence of triangulated categories.

Proof. The proof follows that of [Polishchuk and Positselski 2012, Theorem 3.2] (see also [Positselski 2010, Theorem 7.2.2]) with some modifications. We will prove part (a); the proofs of parts (b) and (c) are similar. (Alternatively, parts (b) and (c) can be deduced from Proposition 1.5(a) and (b) below.)

Given an affine open subscheme $U \subset X$ and a graded module P over the graded ring $\mathcal{B}(U)$, one can construct the freely generated CDG-module $G^+(P)$ over the CDG-ring $\mathcal{B}(U)$ in the way explained in [Positselski 2011b, proof of Theorem 3.6]. The elements of $G^+(P)$ are formal expressions of the form p + dq, where $p, q \in P$. Given a quasicoherent graded module \mathcal{P} over \mathcal{B} , the CDG-modules $G^+(\mathcal{P}(U))$ glue together to form a quasicoherent CDG-module $G^+(\mathcal{P})$ over \mathcal{B} . For any quasicoherent CDG-module \mathcal{M} over \mathcal{B} , there is a bijective correspondence between morphisms of graded \mathcal{B} -modules $\mathcal{P} \to \mathcal{M}$ and closed morphisms of CDG-modules $G^+(\mathcal{P}) \to \mathcal{M}$ over \mathcal{B} . There is a natural short exact sequence of quasicoherent graded \mathcal{B} -modules $\mathcal{P} \to \mathcal{P}[-1]$. The quasicoherent CDG-module $G^+(\mathcal{P})$ is naturally contractible with the contracting homotopy $t_{\mathcal{P}}$ given by the composition

$$G^+(\mathcal{P}) \longrightarrow \mathcal{P}[-1] \longrightarrow G^+(\mathcal{P})[-1].$$

Due to our assumption on X, for any quasicoherent \mathcal{O}_X -module \mathcal{K} over X there exists a surjective morphism $\mathcal{E} \to \mathcal{K}$ onto \mathcal{K} from a direct sum \mathcal{E} of locally free sheaves of finite rank on X. Hence for any quasicoherent graded \mathcal{B} -module \mathcal{M} there is a surjective morphism onto \mathcal{M} from a flat quasicoherent graded \mathcal{B} -module $\mathcal{P} = \bigoplus_n \mathcal{B} \otimes_{\mathcal{O}_X} \mathcal{E}_n[n]$, and for any quasicoherent CDG-module \mathcal{M} over \mathcal{B} there is a surjective closed morphism onto \mathcal{M} from the CDG-module $\mathcal{G}^+(\mathcal{P}) \in \mathcal{B}$ -qcoh_{fl}. (In fact, parts (a) and (b) of this theorem can be proven without the assumption of enough vector bundles on X since there are always enough flat sheaves; see Remark 2.6 and Lemma A.1.)

Now the construction from [loc. cit., proof of Theorem 3.6] provides for any object \mathcal{M} of \mathcal{B} -qcoh_{ffd} a closed morphism onto \mathcal{M} from an object of \mathcal{B} -qcoh_{fl} with the cone absolutely acyclic with respect to \mathcal{B} -qcoh_{ffd}. To obtain this morphism, one picks a finite left resolution of \mathcal{M} consisting of objects from \mathcal{B} -qcoh_{fl} with closed morphisms between them, and takes the total CDG-module of this resolution. By [loc. cit., Lemma 1.6], it follows that the triangulated category $D^{co}(\mathcal{B}$ -qcoh_{ffd}) is equivalent to the quotient category of $H^0(\mathcal{B}$ -qcoh_{fl}) by its intersection in $H^0(\mathcal{B}$ -qcoh_{ffd}. It only remains to show that any object of $H^0(\mathcal{B}$ -qcoh_{fl}) that is coacyclic with respect to \mathcal{B} -qcoh_{ffd}.

Let us call a quasicoherent CDG-module \mathcal{M} over \mathcal{B} *d*-flat if its underlying quasicoherent graded \mathcal{B} -module \mathcal{M} has flat dimension not exceeding *d*. A *d*-flat quasicoherent CDG-module is said to be *d*-coacyclic if it is homotopy equivalent to a CDG-module obtained from the total CDG-modules of exact triples of *d*-flat

1170

CDG-modules using the operations of cone and infinite direct sum. Our goal is to show that any 0-flat *d*-coacyclic CDG-module is 0-coacyclic. For this purpose, we will prove that any (d-1)-flat *d*-coacyclic CDG-module is (d-1)-coacyclic; the desired assertion will then follow by induction.

It suffices to construct for any *d*-coacyclic CDG-module \mathcal{M} a (d-1)-coacyclic CDG-module \mathcal{L} with a (d-1)-coacyclic CDG-submodule \mathcal{K} such that the quotient CDG-module \mathcal{L}/\mathcal{K} is isomorphic to \mathcal{M} . Then if \mathcal{M} is (d-1)-flat, it would follow that both the cone of the morphism $\mathcal{K} \to \mathcal{L}$ and the total CDG-module of the exact triple $\mathcal{K} \to \mathcal{L} \to \mathcal{M}$ are (d-1)-coacyclic, so \mathcal{M} also is. The construction is based on four lemmas similar to those in [Polishchuk and Positselski 2012, Section 3.2].

Lemma A. Let \mathcal{M} be the total CDG-module of an exact triple of d-flat quasicoherent CDG-modules $\mathcal{M}' \to \mathcal{M}'' \to \mathcal{M}'''$ over \mathcal{B} . Then there exists a surjective closed morphism onto \mathcal{M} from a contractible 0-flat CDG-module \mathcal{P} with a (d-1)coacyclic kernel \mathcal{K} .

Proof. Choose 0-flat quasicoherent CDG-modules \mathcal{P}' and \mathcal{P}''' such that there exist surjective closed morphisms $\mathcal{P}' \to \mathcal{M}'$ and $\mathcal{P}''' \to \mathcal{M}''$. Then there exists a surjective morphism from the exact triple of CDG-modules $\mathcal{P}' \to \mathcal{P}' \oplus \mathcal{P}''' \to \mathcal{P}'''$ onto the exact triple $\mathcal{M}' \to \mathcal{M}'' \to \mathcal{M}'''$. The rest of the proof is similar to that in [Polishchuk and Positselski 2012].

Lemma B. (a) Let $\mathcal{K}' \subset \mathcal{L}'$ and $\mathcal{K}'' \subset \mathcal{L}''$ be (d-1)-coacyclic CDG-submodules in (d-1)-coacyclic CDG-modules, and let $\mathcal{L}'/\mathcal{K}' \to \mathcal{L}''/\mathcal{K}''$ be a closed morphism of CDG-modules. Then there exists a (d-1)-coacyclic CDG-module \mathcal{L} with a (d-1)-coacyclic CDG-submodule \mathcal{K} such that

$$\mathcal{L}/\mathcal{K} \simeq \operatorname{cone}(\mathcal{L}'/\mathcal{K}' \to \mathcal{L}''/\mathcal{K}'').$$

(b) In the situation of (a), assume that the morphism L'/K' → L"/K" is injective with a d-flat cokernel M₀. Then there exists a (d-1)-coacyclic CDG-module L₀ with a (d-1)-coacyclic CDG-submodule K₀ such that L₀/K₀ ≃ M₀.

Proof. The proof is similar to that in [Polishchuk and Positselski 2012].

Lemma C. For any contractible d-flat CDG-module \mathcal{M} there exists a surjective closed morphism onto \mathcal{M} from a contractible 0-flat CDG-module \mathcal{L} with a (d-1)-coacyclic kernel \mathcal{K} .

Proof. Let $p : \mathcal{P} \to \mathcal{M}$ be a surjective morphism onto the quasicoherent graded \mathcal{B} -module \mathcal{M} from a flat quasicoherent graded \mathcal{B} -module \mathcal{P} , and $\tilde{p} : G^+(\mathcal{P}) \to \mathcal{M}$ be the induced surjective closed morphism of quasicoherent CDG-modules. Let $t : \mathcal{M} \to \mathcal{M}$ be a contracting homotopy for \mathcal{M} and $t_{\mathcal{P}} : G^+(\mathcal{P}) \to G^+(\mathcal{P})$ be the natural contracting homotopy for $G^+(\mathcal{P})$. Then $\tilde{u} = \tilde{p}t_{\mathcal{P}} - t\tilde{p} : G^+(\mathcal{P}) \to \mathcal{M}$ is a closed morphism of quasicoherent CDG-modules of degree -1. Denote by u

the restriction of \tilde{u} to $\mathcal{P} \subset G^+(\mathcal{P})$. There exists a surjective morphism from a flat quasicoherent graded \mathcal{B} -module \mathcal{Q} onto the fibered product of the morphisms $p: \mathcal{P} \to \mathcal{M}$ and $u: \mathcal{P} \to \mathcal{M}$. Hence we obtain a surjective morphism of quasicoherent graded \mathcal{B} -modules $q: \mathcal{Q} \to \mathcal{P}$ and a morphism of quasicoherent graded \mathcal{B} -modules $v: \mathcal{Q} \to \mathcal{P}$ of degree -1 such that uq = pv.

The morphism q induces a surjective closed morphism of quasicoherent CDGmodules $\tilde{q}: G^+(Q) \to G^+(\mathcal{P})$. The morphism \tilde{q} is homotopic to zero with the natural contracting homotopy $\tilde{q}t_Q = t_{\mathcal{P}}\tilde{q}$. The morphism v induces a closed morphism of CDG-modules $\tilde{v}: G^+(Q) \to G^+(\mathcal{P})$ of degree -1. The morphism $t_{\mathcal{P}}\tilde{q} - \tilde{v}$ is another contracting homotopy for \tilde{q} . The latter homotopy forms a commutative square with the morphisms \tilde{p} , $\tilde{p}\tilde{q}$, and the contracting homotopy tfor the CDG-module \mathcal{M} .

Let \mathcal{N} be the kernel of the morphism $\tilde{p}\tilde{q}: G^+(\mathcal{Q}) \to \mathcal{M}$ and \mathcal{K} be the kernel of the morphism $\tilde{p}: G^+(\mathcal{P}) \to \mathcal{M}$. Then the natural surjective closed morphism $r: \mathcal{N} \to \mathcal{K}$ is homotopic to zero; the restriction of the map $t_{\mathcal{P}}\tilde{q} - \tilde{v}$ provides the contracting homotopy that we need. In addition, the kernel $G^+(\ker q)$ of the morphism r is contractible. So the cone of the morphism r is isomorphic to $\mathcal{K} \oplus \mathcal{N}[1]$, and on the other hand, there is an exact triple $G^+(\ker q)[1] \to \operatorname{cone}(r) \to \operatorname{cone}(\operatorname{id}_{\mathcal{K}})$. Since \mathcal{K} is (d-1)-flat and ker q is flat, this proves that \mathcal{K} is (d-1)-coacyclic. It remains to take $\mathcal{L} = G^+(\mathcal{P})$.

Lemma D. Let $\mathcal{M} \to \mathcal{M}'$ be a homotopy equivalence of d-flat CDG-modules such that \mathcal{M}' is the quotient CDG-module of a (d-1)-coacyclic CDG-module by a (d-1)-coacyclic CDG-submodule. Then \mathcal{M} is also such a quotient.

Proof. The proof is similar to that in [Polishchuk and Positselski 2012]. \Box

It is clear that the property of a CDG-module to be presentable as the cokernel of an injective closed morphism of (d-1)-coacyclic CDG-modules is stable under infinite direct sums. This finishes our construction and the proof of Theorem. \Box

Remark 1.4. The assertion of part (c) of Theorem 1.4 can be equivalently rephrased with flat modules replaced by locally projective ones. Indeed, a finitely presented module over a ring is flat if and only if it is projective.

In the infinitely generated situation of parts (a) and (b), flatness of quasicoherent sheaves is different from their local projectivity (which is a stronger condition), but the assertions remain true after one replaces the former with the latter. The same applies to Proposition 1.5(a) below. Indeed, by Theorem A.2, for any quasicoherent graded algebra \mathcal{B} over an affine scheme U, projectivity of a graded module over the graded ring $\mathcal{B}(U)$ is a local notion. Taking this fact into account, our proof goes through for locally projective quasicoherent graded modules in place of flat ones and the locally projective dimension (defined as the minimal length of a locally projective resolution) in place of the flat dimension.

When $\mathcal{B} = \mathcal{O}_X$, local projectivity of quasicoherent modules is equivalent to local freeness [Bass 1963, Corollary 4.5]. Furthermore, in this case, assuming additionally that X has finite Krull dimension, the classes of quasicoherent sheaves of finite flat dimension and of finite locally projective dimension coincide [Raynaud and Gruson 1971, Corollaire II.3.3.2].

1.5. *Fully faithful embedding.* The next proposition is stronger than Theorem 1.4 in some respects, and is proven by an entirely different technique.

Proposition 1.5. (a) For any quasicoherent CDG-algebra \mathcal{B} over X, the functor $\mathsf{D}^{\mathsf{abs}}(\mathcal{B}\operatorname{-qcoh}_{\mathsf{fl}}) \to \mathsf{D}^{\mathsf{abs}}(\mathcal{B}\operatorname{-qcoh})$ induced by the embedding of DG-categories $\mathcal{B}\operatorname{-qcoh}_{\mathsf{fl}} \to \mathcal{B}\operatorname{-qcoh}$ is fully faithful.

Furthermore, let \mathcal{B} be a quasicoherent CDG-algebra over X such that the underlying quasicoherent graded algebra \mathcal{B} is Noetherian. Then

- (b) the functor D^{abs}(B-coh_{fl}) → D^{abs}(B-coh) induced by the embedding of DG-categories B-coh_{fl} → B-coh is fully faithful;
- (c) the functor D^{abs}(B-coh) → D^{abs}(B-qcoh) induced by the embedding of DG-categories B-coh → B-qcoh is fully faithful;
- (d) the functor D^{abs}(B-coh) → D^{co}(B-qcoh) induced by the embedding of DG-categories B-coh → B-qcoh is fully faithful and its image forms a set of compact generators for D^{co}(B-qcoh).

Proof. The proof of part (d) in the case when X is affine can be found in [Positselski 2011b, Section 3.11] (the part concerning compact generation belongs to D. Arinkin). The proof in the general case is similar, and part (c) can be also proven in the way similar to [loc. cit., Theorem 3.11.1]. Part (b) in the affine case is easy and follows from the semiorthogonality property of CDG-modules with projective underlying graded modules and absolutely acyclic/contraacyclic CDG-modules [loc. cit., Theorem 3.5(b)] since finitely generated flat modules over a Noetherian ring are projective. A detailed proof of part (b) in the general case is given below; and the proof of part (a) (which does not automatically simplify in the affine case) is similar.

We will show that any morphism $\mathcal{E} \to \mathcal{L}$ from a CDG-module $\mathcal{E} \in H^0(\mathcal{B}\operatorname{-coh}_{fl})$ to a CDG-module $\mathcal{L} \in H^0(\mathcal{B}\operatorname{-coh})$ absolutely acyclic with respect to $\mathcal{B}\operatorname{-coh}$ can be annihilated by a morphism $\mathcal{P} \to \mathcal{E}$ from a CDG-module $\mathcal{P} \in H^0(\mathcal{B}\operatorname{-coh}_{fl})$ with a cone of the morphism $\mathcal{P} \to \mathcal{E}$ being absolutely acyclic with respect to $\mathcal{B}\operatorname{-coh}_{fl}$. By the definition, the CDG-module \mathcal{L} is a direct summand of a CDG-module homotopy equivalent to a CDG-module obtained from the totalizations of exact triples of CDG-modules in $\mathcal{B}\operatorname{-coh}$ using the operation of passage to the cone of a closed morphism repeatedly. It suffices to consider the case when \mathcal{L} itself is obtained from totalizations of exact triples using cones. We proceed by induction in the number of operations of passage to the cone in such a construction of \mathcal{L} .

So we assume that there is a distinguished triangle $\mathcal{K} \to \mathcal{L} \to \mathcal{M} \to \mathcal{K}[1]$ in $H^0(\mathcal{B}\text{-coh})$ such that \mathcal{M} is the total CDG-module of an exact triple of CDG-modules in $\mathcal{B}\text{-coh}$, while the CDG-module \mathcal{K} has the desired property with respect to morphisms into it from all CDG-modules $\mathcal{F} \in H^0(\mathcal{B}\text{-coh}_{fl})$. If we knew that the object \mathcal{M} also has the same property, it would follow that the composition $\mathcal{E} \to \mathcal{L} \to \mathcal{M}$ can be annihilated by a morphism $\mathcal{F} \to \mathcal{E}$ with $\mathcal{F} \in H^0(\mathcal{B}\text{-coh}_{fl})$ and a cone absolutely acyclic with respect to $\mathcal{B}\text{-coh}_{fl}$. The composition $\mathcal{F} \to \mathcal{E} \to \mathcal{L}$ then factorizes through \mathcal{K} , and the morphism $\mathcal{F} \to \mathcal{K}$ can be annihilated by a morphism $\mathcal{P} \to \mathcal{F}$ with $\mathcal{P} \in H^0(\mathcal{B}\text{-coh}_{fl})$ and a cone absolutely acyclic with respect to $\mathcal{B}\text{-coh}_{fl}$. The composition $\mathcal{P} \to \mathcal{F}$ by a morphism $\mathcal{P} \to \mathcal{F}$ with $\mathcal{P} \in H^0(\mathcal{B}\text{-coh}_{fl})$ and a cone absolutely acyclic with respect to $\mathcal{B}\text{-coh}_{fl}$.

Thus it remains to construct a morphism $\mathcal{F} \to \mathcal{E}$ with the required properties annihilating a morphism $\mathcal{E} \to \mathcal{M}$, where \mathcal{M} is the total CDG-module of an exact triple of CDG-modules $\mathcal{U} \to \mathcal{V} \to \mathcal{W}$. For any graded module \mathcal{N} over \mathcal{B} , morphisms of graded \mathcal{B} -modules $\mathcal{N} \to \mathcal{M}$ of degree n are represented by triples (f, g, h), where $f : \mathcal{N} \to \mathcal{U}$ is a morphism of degree n + 1, $g : \mathcal{N} \to \mathcal{V}$ is a morphism of degree n, and $h : \mathcal{N} \to \mathcal{W}$ is a morphism of degree n - 1. Denote the closed morphisms in the exact triple $\mathcal{U} \to \mathcal{V} \to \mathcal{W}$ by $j : \mathcal{U} \to \mathcal{V}$ and $k : \mathcal{V} \to \mathcal{W}$.

Lemma E. Let \mathcal{N} be a CDG-module over \mathcal{B} and \mathcal{M} be the total CDG-module of an exact triple of CDG-modules $\mathcal{U} \to \mathcal{V} \to \mathcal{W}$ as above. Then

(a) the differential of a morphism of graded \mathcal{B} -modules $\mathcal{N} \to \mathcal{M}$ of degree n represented by a triple (f, g, h) is given by the rule

$$d(f,g,h) = (-df, -jf + dg, kg - dh);$$

(b) when (f, g, h) is a closed morphism of CDG-modules of degree n and the morphism of graded B-modules h : N → W can be lifted to a morphism of graded B-modules t : N → V of degree n − 1, the morphism (f, g, h) is homotopic to zero.

Proof. We know that the complex of morphisms in the DG-category of CDG-modules $Hom_{\mathcal{B}}(\mathcal{N}, \mathcal{M})$ is the total complex of the bicomplex of abelian groups

$$\operatorname{Hom}_{\mathcal{B}}(\mathcal{N},\mathcal{U})\longrightarrow \operatorname{Hom}_{\mathcal{B}}(\mathcal{N},\mathcal{V})\longrightarrow \operatorname{Hom}_{\mathcal{B}}(\mathcal{N},\mathcal{W}).$$

The formula in (a) is the formula for the differential of a total complex.

Furthermore, the sequence $0 \to \operatorname{Hom}_{\mathcal{B}}(\mathcal{N}, \mathcal{U}) \to \operatorname{Hom}_{\mathcal{B}}(\mathcal{N}, \mathcal{V}) \to \operatorname{Hom}_{\mathcal{B}}(\mathcal{N}, \mathcal{W})$ is exact. Let $\operatorname{Hom}'_{\mathcal{B}}(\mathcal{N}, \mathcal{W})$ denote the cokernel of the morphism of complexes $\operatorname{Hom}_{\mathcal{B}}(\mathcal{N}, \mathcal{U}) \to \operatorname{Hom}_{\mathcal{B}}(\mathcal{N}, \mathcal{V})$; then $\operatorname{Hom}'_{\mathcal{B}}(\mathcal{N}, \mathcal{W})$ is a subcomplex of $\operatorname{Hom}_{\mathcal{B}}(\mathcal{N}, \mathcal{W})$ and the total complex of the bicomplex

$$\operatorname{Hom}_{\mathcal{B}}(\mathcal{N},\mathcal{U})\longrightarrow \operatorname{Hom}_{\mathcal{B}}(\mathcal{N},\mathcal{V})\longrightarrow \operatorname{Hom}_{\mathcal{B}}'(\mathcal{N},\mathcal{W})$$

1174

is an acyclic subcomplex of $\operatorname{Hom}_{\mathcal{B}}(\mathcal{N}, \mathcal{M})$. Hence any cocycle in $\operatorname{Hom}_{\mathcal{B}}(\mathcal{N}, \mathcal{M})$ that belongs to this subcomplex is a coboundary.

To present the same argument using our letter notation for morphisms, assume that kt = h. Then k(dt - g) = dh - kg = 0, so there exists a morphism of graded \mathcal{B} -modules $s : \mathcal{N} \to \mathcal{U}$ of degree n such that dt - g = js. Then jds = -dg = -jf; hence ds = -f and d(s, t, 0) = (f, g, h).

Recall the notation $G^+(Q)$ for the CDG-module freely generated by a graded \mathcal{B} -module \mathcal{Q} (see the beginning of the proof of Theorem 1.4).

Lemma F. Let \mathcal{M} be the total CDG-module of an exact triple of CDG-modules $\mathcal{U} \to \mathcal{V} \to \mathcal{W}$ as above, and let \mathcal{Q} be a graded \mathcal{B} -module. Assume that a morphism of graded \mathcal{B} -modules $p : \mathcal{Q} \to \mathcal{M}$ of degree n with the components (f, g, h) is given such that the component $h : \mathcal{Q} \to \mathcal{W}$ can be lifted to a morphism of graded \mathcal{B} -modules $t : \mathcal{Q} \to \mathcal{V}$ of degree n - 1. Let $\tilde{p} : G^+(\mathcal{Q}) \to \mathcal{M}$ be the induced closed morphism of graded \mathcal{B} -modules of degree n and $(\tilde{f}, \tilde{g}, \tilde{h})$ be its three components. Then the morphism of graded \mathcal{B} -modules $\tilde{h} : G^+(\mathcal{Q}) \to \mathcal{W}$ can be lifted to a morphism of graded \mathcal{B} -modules $\tilde{t} : G^+(\mathcal{Q}) \to \mathcal{W}$ can be lifted to a morphism of graded \mathcal{B} -modules $\tilde{t} : G^+(\mathcal{Q}) \to \mathcal{W}$ can be lifted to a morphism of graded \mathcal{B} -modules $\tilde{t} : G^+(\mathcal{Q}) \to \mathcal{W}$ can be lifted to a morphism of graded \mathcal{B} -modules $\tilde{t} : G^+(\mathcal{Q}) \to \mathcal{V}$ of degree n - 1.

Proof. Notice that any closed morphism of CDG-modules $G^+(Q) \to \mathcal{M}$ is homotopic to zero since the CDG-module $G^+(Q)$ is contractible. The conclusion of the lemma is stronger, and we will need its full strength. The argument consists of a computation in the letter notation for morphisms.

For any CDG-module \mathcal{N} over \mathcal{B} , morphisms of graded \mathcal{B} -modules

 $\tilde{r}: G^+(\mathcal{Q}) \longrightarrow \mathcal{N}$

of degree n-1 are uniquely determined by their restriction to Q and the restriction to Q of their differential $d\tilde{r}$, which can be arbitrary morphisms of graded \mathcal{B} -modules $Q \to \mathcal{N}$ of degrees n-1 and n, respectively. Extend our morphism $t : Q \to \mathcal{V}$ to a morphism of graded \mathcal{B} -modules $\tilde{t} : G^+(Q) \to \mathcal{V}$ of degree n-1 such that $(d\tilde{t})|_Q = g$. Then $k\tilde{t}|_Q = kt = h = \tilde{h}|_Q$ and $(d(k\tilde{t}))|_Q = k(d\tilde{t})|_Q = kg = k\tilde{g}|_Q = (d\tilde{h})|_Q$ by Lemma E(a), and hence $k\tilde{t} = \tilde{h}$.

Now represent a closed morphism $\mathcal{E} \to \mathcal{M}$ by a triple (f, g, h) of morphisms of degrees 1, 0, and -1, respectively. Let \mathcal{Q} be a flat coherent graded \mathcal{B} -module mapping surjectively onto the fibered product of the morphisms $k : \mathcal{V} \to \mathcal{W}$ and $h : \mathcal{E} \to \mathcal{W}$ (see the beginning of the proof of Theorem 1.4 again). Then there is a surjective morphism of graded \mathcal{B} -modules $q : \mathcal{Q} \to \mathcal{E}$ and its composition with the morphism $h : \mathcal{E} \to \mathcal{W}$ can be lifted to a morphism of graded \mathcal{B} -modules $t : \mathcal{Q} \to \mathcal{V}$ of degree -1. Consider the induced morphism of CDG-modules $\tilde{q} : G^+(\mathcal{Q}) \to \mathcal{E}$. By Lemma F, the composition $h\tilde{q} : G^+(\mathcal{Q}) \to \mathcal{W}$ can be lifted to a morphism of graded \mathcal{B} -modules $\tilde{t} : G^+(\mathcal{Q}) \to \mathcal{V}$ of degree -1. 1176

Let \mathcal{R} denote the kernel of the closed morphism \tilde{q} . Then the cone \mathcal{F} of the embedding $\mathcal{R} \to G^+(\mathcal{Q})$ maps naturally onto \mathcal{E} with the cone absolutely acyclic with respect to \mathcal{B} -coh_{fl}. As a graded \mathcal{B} -module, the CDG-module \mathcal{F} is isomorphic to $G^+(\mathcal{Q}) \oplus \mathcal{R}[1]$; the composition $\mathcal{F} \to \mathcal{E} \to \mathcal{M}$ factorizes through the direct summand $G^+(\mathcal{Q})$, where it is defined by the triple $(f\tilde{q}, g\tilde{q}, h\tilde{q})$. Since the morphism $h\tilde{q}$ can be lifted to \mathcal{V} , so can the corresponding component $\mathcal{F} \to \mathcal{W}$ of the morphism $\mathcal{F} \to \mathcal{M}$. Thus the latter morphism is homotopic to zero by Lemma E(b).

In some cases, the use of Lemma F in the above proof of part (b) can be avoided. Assume that X is a projective scheme over a Noetherian ring and the category of coherent graded \mathcal{B} -modules is equivalent to the category of coherent modules over some coherent (graded) \mathcal{O}_X -algebra \mathcal{A} . In this situation, one takes \mathcal{Q} to be the graded \mathcal{B} -module corresponding to the (graded) \mathcal{A} -module induced from a large enough finite direct sum of (shifts of) copies of a sufficiently negative invertible \mathcal{O}_X -module; then there is a surjective morphism of graded \mathcal{B} -modules $\mathcal{Q} \to \mathcal{E}$ and any morphism of graded \mathcal{B} -modules $\mathcal{G}^+(\mathcal{Q}) \to \mathcal{W}$ lifts to \mathcal{V} .

Remark 1.5. We do *not* know how to extend the proof of Proposition 1.5 (a) and (b) to the coderived categories of quasicoherent CDG-modules. Instead, this argument appears to be well-suited for use with the *contraderived* categories (see [Positselski 2011b, Section 3.3] for the definition). In particular, it allows to show that the contraderived category of left CDG-modules over a CDG-ring B with a right coherent underlying graded ring is equivalent to the contraderived category of CDG-modules whose underlying graded B-modules are flat (cf. [loc. cit., paragraph after the proof of Theorem 3.8]).

This is the main reason why we sometimes find it easier to deal with the absolute derived rather than the coderived categories of infinitely generated CDG-modules (cf. Remark 2.8). On the other hand, for the coderived category of quasicoherent CDG-modules we have the compact generation result (part (d) of Proposition 1.5), the results and arguments of Sections 1.7, 1.10, 2.5, 2.9, etc. The conditions under which these two versions of the construction of the derived category of the second kind for a given class of CDG-modules lead to the same triangulated category are discussed below in Section 1.6.

1.6. *Finite homological dimension theorem.* Let \mathcal{B} -qcoh_{lp} denote the DG-category of quasicoherent CDG-modules over \mathcal{B} whose underlying graded \mathcal{B} -modules are locally projective (see Remark 1.4 and Theorem A.2). Denote by $D^{co}(\mathcal{B}$ -qcoh_{lp}) and $D^{abs}(\mathcal{B}$ -qcoh_{lp}) the corresponding coderived and absolute derived categories.

Theorem 1.6. The triangulated categories $D^{co}(\mathcal{B}\operatorname{-qcoh}_{lp})$ and $D^{abs}(\mathcal{B}\operatorname{-qcoh}_{lp})$ coincide; i.e., every CDG-module over \mathcal{B} that is coacyclic with respect to $\mathcal{B}\operatorname{-qcoh}_{lp}$ is also absolutely acyclic with respect to $\mathcal{B}\operatorname{-qcoh}_{lp}$.
Proof. The reason for this assertion to be true is that the exact category of locally projective graded \mathcal{B} -modules has finite homological dimension [Orlov 2004, Lemma 1.12] and exact functors of infinite direct sums. If this exact category also had enough injectives, the simple argument from [Positselski 2011b, Theorem 3.6(a) and Remark 3.6] would suffice to establish the desired $D^{co} = D^{abs}$ isomorphism for it (see also [Positselski 2010, Remark 2.1]). The lengthy argument below is designed to provide a way around the injective objects issue in this kind of proof.

Our aim is to show that for any closed morphism $\mathcal{P} \to \mathcal{L}$ from a CDG-module $\mathcal{P} \in \mathcal{B}\operatorname{-qcoh}_{lp}$ to a CDG-module \mathcal{L} absolutely acyclic with respect to $\mathcal{B}\operatorname{-qcoh}_{lp}$, there exists an exact sequence $0 \to \mathcal{Q}_d \to \mathcal{Q}_{d-1} \to \cdots \to \mathcal{Q}_0 \to \mathcal{P} \to 0$ of CDG-modules and closed morphisms in $\mathcal{B}\operatorname{-qcoh}_{lp}$ such that the induced morphism from the total CDG-module of $\mathcal{Q}_d \to \cdots \to \mathcal{Q}_0$ to \mathcal{L} is homotopic to zero. Here *d* is a fixed integer equal to the homological dimension of the exact category of locally projective graded $\mathcal{B}\operatorname{-modules}$, which does not exceed the number of open subsets in an affine covering of *X* minus one.

Taking $\mathcal{P} = \mathcal{L}$ and the morphism $\mathcal{P} \to \mathcal{L}$ to be the identity, we will then conclude that \mathcal{P} is isomorphic to a direct summand of the total CDG-module of $\mathcal{Q}_d \to \cdots \to \mathcal{Q}_0 \to \mathcal{P}$ in $H^0(\mathcal{B}\operatorname{-qcoh}_{lp})$. Hence an object of $H^0(\mathcal{B}\operatorname{-qcoh}_{lp})$ is absolutely acyclic with respect to $\mathcal{B}\operatorname{-qcoh}_{lp}$ if and only if it is isomorphic to a direct summand of the total CDG-module of a (d+2)-term exact sequence of CDG-modules from $\mathcal{B}\operatorname{-qcoh}_{lp}$ with closed morphisms between them. It will immediately follow that the class of CDG-modules absolutely acyclic with respect to $\mathcal{B}\operatorname{-qcoh}_{lp}$ is closed under infinite direct sums, so it coincides with the class of coacyclic CDG-modules.

We can suppose that there exists a sequence of distinguished triangles

$$\mathcal{K}_{i-1} \longrightarrow \mathcal{K}_i \longrightarrow \mathcal{M}_i \longrightarrow \mathcal{K}_{i-1}[1]$$

in $H^0(\mathcal{B}\operatorname{-qcoh}_{\mathsf{lp}})$ such that $\mathcal{K}_0 = 0$, $\mathcal{K}_n = \mathcal{L}$, and \mathcal{M}_i is the total CDG-module of an exact triple $\mathcal{U}_i \to \mathcal{V}_i \to \mathcal{W}_i$ of CDG-modules from $\mathcal{B}\operatorname{-qcoh}_{\mathsf{lp}}$ for all $1 \le i \le n$. We will start with constructing an exact sequence $0 \to \mathcal{Q}'_n \to \cdots \to \mathcal{Q}'_0 \to \mathcal{P} \to 0$ with the above properties, but of the length *n* rather than *d*. Then we will use the finite homological dimension property of locally projective graded \mathcal{B} -modules in order to obtain the desired resolution \mathcal{Q}_{\bullet} of a fixed length *d* from a resolution \mathcal{Q}'_{\bullet} .

Lemma G. Let \mathcal{M} be the total CDG-module of an exact triple $\mathcal{U} \to \mathcal{V} \to \mathcal{W}$ of CDG-modules from \mathcal{B} -qcoh_{lp} and $\mathcal{K} \to \mathcal{L} \to \mathcal{M} \to \mathcal{K}[1]$ be a distinguished triangle in $H^0(\mathcal{B}$ -qcoh_{lp}). Then for any CDG-module $\mathcal{P} \in \mathcal{B}$ -qcoh_{lp} and a morphism $\mathcal{P} \to \mathcal{L}$ in $H^0(\mathcal{B}$ -qcoh_{lp}) there exists an exact triple $\mathcal{R} \to \mathcal{Q} \to \mathcal{P}$ of CDG-modules from \mathcal{B} -qcoh_{lp} and a morphism $\mathcal{R}[1] \to \mathcal{K}$ in $H^0(\mathcal{B}$ -qcoh_{lp}) such that the composition $\mathcal{F} \to \mathcal{P} \to \mathcal{L}$, where \mathcal{F} is the cone of the closed morphism $\mathcal{R} \to \mathcal{Q}$, is equal to the composition $\mathcal{F} \to \mathcal{R}[1] \to \mathcal{K} \to \mathcal{L}$ in $H^0(\mathcal{B}$ -qcoh_{lp}).

Proof. The argument is based on Lemmas E and F from Section 1.5. We can assume that \mathcal{L} is the cone of a closed morphism $\mathcal{M}[-1] \to \mathcal{K}$ and fix a closed morphism $\mathcal{P} \to \mathcal{L}$ representing the given morphism in the homotopy category. Arguing as in the proof of Proposition 1.5, we can construct a surjective closed morphism $\mathcal{Q}' \to \mathcal{P}$ onto \mathcal{P} from a CDG-module $\mathcal{Q}' \in \mathcal{B}$ -qcoh_{lp} such that the composition $\mathcal{Q}' \to \mathcal{P} \to \mathcal{L} \to \mathcal{M} \to \mathcal{W}[-1]$ lifts to a morphism of graded \mathcal{B} -modules $\mathcal{Q}' \to \mathcal{V}[-1]$. Here it suffices to apply the functor G^+ to the fibered product of the morphisms of graded \mathcal{B} -modules $\mathcal{P} \to \mathcal{W}[-1]$ and $\mathcal{V}[-1] \to \mathcal{W}[-1]$ and use Lemma F.

Then the morphism $Q' \to M$ is homotopic to zero with a natural contracting homotopy (provided by the proof of Lemma E), so the morphism $Q' \to \mathcal{L}$ factorizes, up to a homotopy, as the composition of a naturally defined closed morphism $Q' \to \mathcal{K}$ and the closed morphism $\mathcal{K} \to \mathcal{L}$. Set Q to be the cocone of the closed morphism $Q' \to \mathcal{K}$; then we have a surjective closed morphism $Q \to Q'$ such that the composition $Q \to Q' \to \mathcal{K}$ is homotopic to zero.

Let \mathcal{R} be kernel of the morphism $\mathcal{Q} \to \mathcal{P}$ and \mathcal{F} be the cone of the morphism $\mathcal{R} \to \mathcal{Q}$; then there is a natural closed morphism $\mathcal{F} \to \mathcal{P}$. Using Lemma E and arguing as in the end of the proof of Proposition 1.5 again, we can conclude that the composition $\mathcal{F} \to \mathcal{P} \to \mathcal{L} \to \mathcal{M}$ is homotopic to zero. Indeed, the composition $\mathcal{F} \to \mathcal{M} \to \mathcal{W}[-1]$ lifts to a graded \mathcal{B} -module morphism $\mathcal{F} \to \mathcal{V}[-1]$ since $\mathcal{F} \simeq \mathcal{Q} \oplus \mathcal{R}[-1]$ as a graded \mathcal{B} -module, the morphism $\mathcal{F} \to \mathcal{M}$ factorizes through the projection of \mathcal{F} onto \mathcal{Q} , and the morphism $\mathcal{Q} \to \mathcal{Q}' \to \mathcal{W}[-1]$ lifts to a graded \mathcal{B} -module morphism $\mathcal{Q} \to \mathcal{Q}' \to \mathcal{W}[-1]$ lifts to a graded \mathcal{B} -module morphism $\mathcal{Q} \to \mathcal{Q}' \to \mathcal{W}[-1]$ lifts to a graded \mathcal{B} -module morphism $\mathcal{Q} \to \mathcal{Q}' \to \mathcal{W}[-1]$ lifts to a graded \mathcal{B} -module morphism $\mathcal{Q} \to \mathcal{Q}' \to \mathcal{W}[-1]$ lifts to a graded \mathcal{B} -module morphism $\mathcal{Q} \to \mathcal{Q}' \to \mathcal{V}[-1]$ by our construction.

Notice that the contracting homotopy that we have obtained for the closed morphism $\mathcal{F} \to \mathcal{M}$ forms a commutative diagram with the closed morphisms $\mathcal{Q} \to \mathcal{F}, \ \mathcal{Q} \to \mathcal{Q}'$, and the contracting homotopy that we have previously had for the closed morphism $\mathcal{Q}' \to \mathcal{M}$ (since so do the liftings $\mathcal{F} \to \mathcal{V}[-1]$ and $\mathcal{Q}' \to \mathcal{V}[-1]$). This allows to factorize, up to a homotopy, the closed morphism $\mathcal{F} \to \mathcal{L}$ as the composition of a closed morphism $\mathcal{F} \to \mathcal{K}$ and the closed morphism $\mathcal{K} \to \mathcal{L}$ in such a way that the morphism $\mathcal{F} \to \mathcal{K}$ forms a commutative diagram with the closed morphisms $\mathcal{Q} \to \mathcal{F}, \ \mathcal{Q} \to \mathcal{Q}'$, and the closed morphism $\mathcal{Q}' \to \mathcal{K}$ that we have previously constructed. The composition $\mathcal{Q} \to \mathcal{F} \to \mathcal{K}$, being equal to the composition $\mathcal{Q} \to \mathcal{Q}' \to \mathcal{K}$, is homotopic to zero; hence the morphism $\mathcal{F} \to \mathcal{K}$ factorizes through the closed morphism $\mathcal{F} \to \mathcal{R}[1]$ in $H^0(\mathcal{B}\operatorname{-qcoh}_{lp})$. \Box

Applying Lemma G to the morphism $\mathcal{P} \to \mathcal{L}$ and the distinguished triangle $\mathcal{K}_{n-1} \to \mathcal{L} \to \mathcal{M}_n \to \mathcal{K}_{n-1}$, we obtain an exact triple $\mathcal{R}'_0 \to \mathcal{Q}'_0 \to \mathcal{P}$ and a morphism $\mathcal{R}'_0[1] \to \mathcal{K}_{n-1}$ in $H^0(\mathcal{B}\operatorname{-qcoh}_{lp})$. Applying the same lemma again to the morphism $\mathcal{R}'_0[1] \to \mathcal{K}_{n-1}$ and the distinguished triangle $\mathcal{K}_{n-2} \to \mathcal{K}_{n-1} \to \mathcal{M}_{n-1} \to \mathcal{K}_{n-2}[1]$, we construct an exact triple $\mathcal{R}'_1 \to \mathcal{Q}'_1 \to \mathcal{R}'_0$ and a morphism $\mathcal{R}'_1[2] \to \mathcal{K}_{n-2}$, etc. Finally we obtain an exact triple $\mathcal{R}'_{n-1} \to \mathcal{Q}'_{n-1} \to \mathcal{Q}'_{n-1} \to \mathcal{R}'_{n-2}$ and a morphism $\mathcal{R}'_{n-1}[n] \to \mathcal{K}_0 = 0$.

Let us check that the natural morphism from the total CDG-module of the complex $0 \to \mathcal{R}'_{n-1} \to \mathcal{Q}'_{n-1} \to \cdots \to \mathcal{Q}'_0$ to the CDG-module \mathcal{L} is homotopic to zero. Denote this morphism by f_n . It factorizes naturally through the cone \mathcal{F}_0 of the closed morphism $\mathcal{R}'_0 \to \mathcal{Q}'_0$, and the morphism $\mathcal{F}_0 \to \mathcal{L}$ is homotopic to the composition $\mathcal{F}_0 \to \mathcal{R}'_0[1] \to \mathcal{K}_{n-1} \to \mathcal{L}$. Hence, up to the homotopy, the morphism f_n factorizes through the morphism f_{n-1} from the total CDG-module of the complex $0 \to \mathcal{R}'_{n-1} \to \mathcal{Q}'_{n-1} \to \cdots \to \mathcal{Q}'_1$ to \mathcal{K}_{n-1} induced by the morphism $\mathcal{R}'_0[1] \to \mathcal{K}_{n-1}$. Continuing to argue in this way, we conclude that the morphism f factorizes, up to a homotopy, through the morphism $f_0 : \mathcal{R}'_{n-1}[n] \to \mathcal{K}_0 = 0$.

It remains to "cut" our exact sequence of an unknown length *n* to a fixed size *d*. For this purpose, we will assume that n > d and construct from our exact sequence of length *n* another exact sequence with the same properties, but of the length n - 1. This part of the argument is based on the following lemma.

Lemma H. For any CDG-module $\mathcal{M} \in \mathcal{B}\operatorname{-qcoh}_{\mathsf{lp}}$, locally projective graded $\mathcal{B}\operatorname{-mod}$ ule \mathcal{E} , and a homogeneous surjective morphism of locally projective graded $\mathcal{B}\operatorname{-mod}$ ules $\mathcal{E} \to \mathcal{M}$, there exist a CDG-module $\mathcal{Q} \in \mathcal{B}\operatorname{-qcoh}_{\mathsf{lp}}$, a surjective closed morphism of CDG-modules $\mathcal{Q} \to \mathcal{M}$, and a homogeneous surjective morphism of locally projective graded $\mathcal{B}\operatorname{-modules} \mathcal{Q} \to \mathcal{E}$ such that the triangle $\mathcal{Q} \to \mathcal{E} \to \mathcal{M}$ commutes.

Proof. For any open subscheme $U \subset X$, one can simply define $Q^i(U)$ as the abelian group of all pairs $(e' \in \mathcal{E}^{i+1}(U), e \in \mathcal{E}^i(U))$ such that df(e) = f(e'), where f denotes the morphism of graded \mathcal{B} -modules $\mathcal{E} \to \mathcal{M}$ and d is the differential in \mathcal{M} . The action of \mathcal{B} in Q is defined by the formula $b(e', e) = ((-1)^{|b|}be' + d(b)e, be)$; the differential in Q is given by the obvious rule d(e', e) = (he, e'). The morphism $Q \to \mathcal{E}$ is defined as $(e', e) \mapsto e$; the morphism $Q \to \mathcal{M}$, given by $(e', e) \mapsto f(e)$, obviously commutes with the differentials.

It remains to check that the graded \mathcal{B} -module \mathcal{Q} is locally projective. This can be done by comparing the above construction with the constructions of the freely (co)generated CDG-modules $G^+(\mathcal{E})$ and $G^-(\mathcal{E})$ from [Positselski 2011b, proof of Theorem 3.6] (see the beginning of the proof of Theorem 1.4). One can simply define $G^-(\mathcal{E})$ as being isomorphic to $G^+(\mathcal{E})[1]$. Since \mathcal{M} is a CDG-module, there is a natural closed morphism of CDG-modules $\mathcal{M} \to G^-(\mathcal{M})$. The CDG-module \mathcal{Q} is the fibered product of the surjective closed morphism of CDG-modules $G^-(\mathcal{E}) \to$ $G^-(\mathcal{M})$ and the closed morphism $\mathcal{M} \to G^-(\mathcal{M})$; hence the graded \mathcal{B} -module \mathcal{Q} is locally projective. The morphism $\mathcal{Q} \to \mathcal{E}$ is induced by the natural morphism of graded \mathcal{B} -modules $G^-(\mathcal{E}) \to \mathcal{E}$. It forms a commutative diagram with the morphism $\mathcal{E} \to \mathcal{M}$, since the composition $\mathcal{M} \to G^-(\mathcal{M}) \to \mathcal{M}$ is the identity morphism. \Box

The exact sequence of CDG-modules

$$0 \longrightarrow \mathcal{R}'_{n-1} \longrightarrow \mathcal{Q}'_{n-1} \longrightarrow \cdots \longrightarrow \mathcal{Q}'_0 \longrightarrow \mathcal{P} \longrightarrow 0$$

represents a certain Yoneda Ext class of degree *n* between the locally projective graded \mathcal{B} -modules \mathcal{P} and \mathcal{R}'_{n-1} . Since the homological dimension of the exact category of such \mathcal{B} -modules is equal to *d* and we assume that n > d, this Ext class has to vanish. This means that there exists an exact sequence of locally projective graded \mathcal{B} -modules $0 \to \mathcal{R}'_{n-1} \to \mathcal{E}_{n-1} \to \cdots \to \mathcal{E}_0 \to \mathcal{P} \to 0$ mapping to our original exact sequence, with the maps on the rightmost and leftmost terms being the identity maps, such that the embedding of \mathcal{B} -modules $\mathcal{R}'_{n-1} \to \mathcal{E}_{n-1}$ splits.

As explained in [Positselski 2011a, proof of Lemma 4.4], one can assume the morphisms $\mathcal{E}_i \to \mathcal{Q}'_i$ to be surjective. Applying Lemma H, we obtain a surjective closed morphism of CDG-modules $\mathcal{Q}_0 \to \mathcal{Q}'_0$ and a morphism of graded \mathcal{B} -modules $\mathcal{Q}_0 \to \mathcal{E}_0$ forming a commutative triangle with the morphism $\mathcal{E}_0 \to \mathcal{Q}'_0$. Applying Lemma H to the surjective morphism of fibered products $\mathcal{Q}_0 \times_{\mathcal{E}_0} \mathcal{E}_1 \to \mathcal{Q}_0 \times_{\mathcal{Q}'_0} \mathcal{Q}'_1$, we obtain a surjective closed morphism $\mathcal{Q}_1 \to \mathcal{Q}'_1$ and a closed morphism $\mathcal{Q}_1 \to \mathcal{Q}_0$ forming a commutative square with the closed morphisms $\mathcal{Q}_0 \to \mathcal{Q}'_0$ and $\mathcal{Q}'_1 \to \mathcal{Q}'_0$. Besides, the sequence $\mathcal{Q}_1 \to \mathcal{Q}_0 \to \mathcal{P}$ is exact at \mathcal{Q}_0 . We also obtain a morphism of graded \mathcal{B} -modules $\mathcal{Q}_1 \to \mathcal{E}_1$ forming a commutative triangle with the morphisms to \mathcal{E}_0 .

Proceeding in this way, we construct a sequence $\mathcal{Q}_{n-2} \to \cdots \to \mathcal{Q}_0 \to \mathcal{P} \to 0$, which is exact at all the middle terms, maps onto the sequence $\mathcal{Q}'_{n-2} \to \cdots \to \mathcal{Q}'_0 \to \mathcal{P}$ by closed morphisms, and maps into the sequence $\mathcal{E}_{n-2} \to \cdots \to \mathcal{E}_0 \to \mathcal{P}$ so that the triangle of the maps of sequences commutes. Finally, notice that $\mathcal{E}_{n-1} \simeq \mathcal{E}_{n-2} \times_{\mathcal{Q}'_{n-2}} \mathcal{Q}'_{n-1}$, and set $\mathcal{Q}_{n-1} = \mathcal{Q}_{n-2} \times_{\mathcal{Q}'_{n-2}} \mathcal{Q}'_{n-1}$. Then the exact sequence of CDG-modules $0 \to \mathcal{R}'_{n-1} \to \mathcal{Q}_{n-1} \to \cdots \to \mathcal{Q}_0 \to \mathcal{P} \to 0$ maps onto the exact sequence $0 \to \mathcal{R}'_{n-1} \to \mathcal{Q}'_{n-1} \to \cdots \to \mathcal{Q}'_0 \to \mathcal{P} \to 0$ by closed morphisms, and this map of exact sequences factorizes through the exact sequence of graded \mathcal{B} -modules $0 \to \mathcal{R}'_{n-1} \to \mathcal{E}_{n-1} \to \cdots \to \mathcal{E}_0 \to \mathcal{P} \to 0$. The composition of the morphism $\mathcal{Q}_{n-1} \to \mathcal{E}_{n-1}$ with the splitting $\mathcal{E}_{n-1} \to \mathcal{R}'_{n-1}$ of the embedding $\mathcal{R}'_{n-1} \to \mathcal{E}_{n-1}$ \mathcal{E}_{n-1} provides a graded \mathcal{B} -module splitting $\mathcal{Q}_{n-1} \to \mathcal{R}'_{n-1}$ of the embedding of CDG-modules $\mathcal{R}'_{n-1} \to \mathcal{Q}_{n-1}$.

Denote by \mathcal{R}_{n-2} the image of the morphism of CDG-modules $\mathcal{Q}_{n-1} \to \mathcal{Q}_{n-2}$. The morphism from the total CDG-module of the complex $\mathcal{R}'_{n-1} \to \mathcal{Q}'_{n-1} \to \cdots \to \mathcal{Q}'_0$ to the CDG-module \mathcal{L} is homotopic to zero; hence so is the morphism to \mathcal{L} from the total CDG-module of the complex $\mathcal{R}'_{n-1} \to \mathcal{Q}_{n-1} \to \cdots \to \mathcal{Q}_0$. The latter morphism factorizes naturally through the total CDG-module of the complex $\mathcal{R}_{n-2} \to \mathcal{Q}_{n-2} \to \cdots \to \mathcal{Q}_0$. The cone of this closed morphism between two total CDG-modules is homotopy equivalent to the total CDG-module of the exact triple $\mathcal{R}'_{n-1} \to \mathcal{Q}_{n-1} \to \mathcal{R}_{n-2}$. Since this exact triple splits as an exact triple of graded \mathcal{B} -modules, its total CDG-module of $\mathcal{R}'_{n-1} \to \mathcal{Q}_{n-1} \to \cdots \to \mathcal{Q}_0$ and $\mathcal{R}_{n-2} \to \mathcal{Q}_{n-2} \to \cdots \to \mathcal{Q}_0$ is a homotopy equivalence.

It follows that the natural morphism from the total CDG-module of the resolution $\mathcal{R}_{n-2} \rightarrow \mathcal{Q}_{n-2} \rightarrow \cdots \rightarrow \mathcal{Q}_0$ of the CDG-module \mathcal{P} to the CDG-module \mathcal{L} is homotopic to zero, and we are done.

So far we have only considered flat coherent CDG-modules over quasicoherent CDG-algebras \mathcal{B} whose underlying quasicoherent graded algebras are Noetherian. But the latter restriction is not necessary, as flat and locally finitely presented (or, which is equivalent, locally projective and finitely generated) quasicoherent graded \mathcal{B} -modules always form an exact subcategory of flat (or locally projective) graded \mathcal{B} -modules. The notation \mathcal{B} -coh_{lp} (understood in the obvious sense as the DG-category of CDG-modules over \mathcal{B} with coherent and locally projective underlying graded \mathcal{B} -modules) is synonymous with \mathcal{B} -coh_{fl} (see Remark 1.4).

Corollary 1.6. The functor $D^{abs}(\mathcal{B}\text{-}coh_{lp}) \rightarrow D^{co}(\mathcal{B}\text{-}qcoh_{lp})$ induced by the embedding of DG-categories $\mathcal{B}\text{-}coh_{lp} \rightarrow \mathcal{B}\text{-}qcoh_{lp}$ is fully faithful.

Proof. When \mathcal{B} is Noetherian, one can show that the functor $D^{abs}(\mathcal{B}\text{-}coh_{lp}) \rightarrow D^{abs}(\mathcal{B}\text{-}qcoh_{lp})$ is fully faithful by comparing parts (a)–(c) of Proposition 1.5 (with the flatness condition replaced by the local projectivity). In the general case, one proves this assertion directly, using an argument similar to the proof of Proposition 1.5(a) and (b). Then it remains to use Theorem 1.6.

When every flat quasicoherent graded module over \mathcal{B} has finite locally projective dimension (see Remark 1.4), one has

$$\begin{split} \mathsf{D}^{\mathsf{co}}(\mathcal{B}\text{-}\mathsf{qcoh}_{\mathsf{lp}}) &\simeq \mathsf{D}^{\mathsf{co}}(\mathcal{B}\text{-}\mathsf{qcoh}_{\mathsf{fl}}) \simeq \mathsf{D}^{\mathsf{co}}(\mathcal{B}\text{-}\mathsf{qcoh}_{\mathsf{ffd}}), \\ \mathsf{D}^{\mathsf{abs}}(\mathcal{B}\text{-}\mathsf{qcoh}_{\mathsf{lp}}) &\simeq \mathsf{D}^{\mathsf{abs}}(\mathcal{B}\text{-}\mathsf{qcoh}_{\mathsf{fl}}) \simeq \mathsf{D}^{\mathsf{abs}}(\mathcal{B}\text{-}\mathsf{qcoh}_{\mathsf{fld}}) \end{split}$$

by appropriate versions of Theorem 1.4. Consequently, it follows from Theorem 1.6 that $D^{abs}(\mathcal{B}\text{-qcoh}_{fl}) = D^{co}(\mathcal{B}\text{-qcoh}_{fl})$ and $D^{abs}(\mathcal{B}\text{-qcoh}_{ffd}) = D^{co}(\mathcal{B}\text{-qcoh}_{ffd})$ in this case. Thus the functor $D^{abs}(\mathcal{B}\text{-coh}_{fl}) \rightarrow D^{co}(\mathcal{B}\text{-qcoh}_{fl})$ is fully faithful; when \mathcal{B} is Noetherian, so is the functor $D^{abs}(\mathcal{B}\text{-coh}_{fd}) \rightarrow D^{co}(\mathcal{B}\text{-qcoh}_{fd})$.

1.7. *Gorenstein case.* Here we establish a sufficient condition for the functor $D^{co}(\mathcal{B}\operatorname{-qcoh}_{fl}) \rightarrow D^{co}(\mathcal{B}\operatorname{-qcoh})$ to be an equivalence of triangulated categories.

Let \mathcal{B} -qcoh_{inj} denote the full DG-subcategory in \mathcal{B} -qcoh consisting of the CDGmodules whose underlying quasicoherent graded \mathcal{B} -modules are injective. Furthermore, let \mathcal{B} -qcoh_{fid} be the full DG-subcategory in \mathcal{B} -qcoh consisting of the CDG-modules whose underlying quasicoherent graded \mathcal{B} -modules have finite injective dimension (i.e., admit a finite right resolution by injective quasicoherent graded \mathcal{B} -modules). Let $D^{abs}(\mathcal{B}$ -qcoh_{fid}) and $D^{co}(\mathcal{B}$ -qcoh_{fid}) denote the corresponding derived categories of the second kind. (The difficulty in the definition of the latter category, similar to the difficulty in the definition of $D^{co}(\mathcal{B}$ -qcoh_{fid}) discussed in Section 1.3, does not actually arise, as it is clear from part (a) of the next lemma.)

- **Lemma 1.7.** (a) For any quasicoherent CDG-algebra \mathcal{B} over X, the natural functors $H^0(\mathcal{B}\operatorname{-qcoh}_{inj}) \to D^{\mathsf{abs}}(\mathcal{B}\operatorname{-qcoh}_{fid}) \to D^{\mathsf{co}}(\mathcal{B}\operatorname{-qcoh}_{fid})$ are equivalences of triangulated categories.
- (b) Let B be a quasicoherent CDG-algebra over X whose underlying quasicoherent graded algebra B is Noetherian. Then the functor H⁰(B-qcoh_{inj}) → D^{co}(B-qcoh) induced by the embedding B-qcoh_{inj} → B-qcoh is an equivalence of triangulated categories.

Proof. Part (a) is provided by [Positselski 2011b, Theorem and Remark in Section 3.6]. Part (b) is a particular case of [loc. cit., Theorem and Remark in Section 3.7] since the class of injective quasicoherent graded \mathcal{B} -modules is closed under infinite direct sums in its assumptions. (Cf. [Lin and Pomerleano 2013, Proposition 2.4].)

Proposition 1.7. Let \mathcal{B} be a quasicoherent CDG-algebra over X such that the quasicoherent graded algebra \mathcal{B} is Noetherian and the classes of quasicoherent graded \mathcal{B} -modules of finite flat dimension and of finite injective dimension coincide. Then the functors $D^{abs}(\mathcal{B}\operatorname{-qcoh}_{fl}) \rightarrow D^{co}(\mathcal{B}\operatorname{-qcoh}_{fl}) \rightarrow D^{co}(\mathcal{B}\operatorname{-qcoh})$ induced by the embedding $\mathcal{B}\operatorname{-qcoh}_{fl} \rightarrow \mathcal{B}\operatorname{-qcoh}$ are equivalences of triangulated categories.

Proof. Since \mathcal{B} -qcoh_{ffd} = \mathcal{B} -qcoh_{fid}, the isomorphism of categories $D^{abs}(\mathcal{B}$ -qcoh_{ffd}) = $D^{co}(\mathcal{B}$ -qcoh_{ffd}) follows from part (a) of Lemma 1.7. Applying Theorem 1.4, we obtain the isomorphism of categories $D^{abs}(\mathcal{B}$ -qcoh_{fl}) → $D^{co}(\mathcal{B}$ -qcoh_{fl}). Similarly, it suffices to compare parts (a) and (b) of Lemma 1.7 in order to conclude that the functor $D^{co}(\mathcal{B}$ -qcoh_{fid}) → $D^{co}(\mathcal{B}$ -qcoh_{fid}) → $D^{co}(\mathcal{B}$ -qcoh_{fid}) → D^{co}(\mathcal{B}-qcoh_{fid}) → D^{co}(\mathcal{B} -qcoh_{fid}) → D^{co}(\mathcal{B} -qcoh_{fid}). (Cf. [Positselski 2011b, Section 3.9].)

1.8. *Pull-backs and push-forwards.* Let $f : Y \to X$ be a morphism of separated Noetherian schemes, \mathcal{B}_X be a quasicoherent CDG-algebra over X, and \mathcal{B}_Y be a quasicoherent CDG-algebra over Y. A *morphism of quasicoherent CDG-algebras* $\mathcal{B}_X \to \mathcal{B}_Y$ compatible with the morphism $Y \to X$ is the data of a CDG-ring morphism $\mathcal{B}_X(U) \to \mathcal{B}_Y(V)$ for each pair of affine open subschemes $U \subset X$ and $V \subset Y$ such that $f(V) \subset U$. This data should satisfy the obvious compatibility condition: for any affine open subschemes $U' \subset U$ and $V' \subset V$ such that $f(V') \subset U'$, the square diagram of CDG-ring morphisms between the CDG-rings $\mathcal{B}_X(U)$, $\mathcal{B}_X(U')$, $\mathcal{B}_Y(V)$, and $\mathcal{B}_Y(V')$ must be commutative.

Let $\mathcal{B}_X \to \mathcal{B}_Y$ be a morphism of quasicoherent CDG-algebras compatible with a morphism of schemes $Y \to X$. Then for any quasicoherent left CDG-module \mathcal{M} over \mathcal{B}_X , the quasicoherent graded left module $f^*\mathcal{M} = \mathcal{B}_Y \otimes_{f^{-1}\mathcal{B}_X} f^{-1}\mathcal{M}$ over \mathcal{B}_Y has a natural structure of quasicoherent CDG-module over \mathcal{B}_Y . Similarly, for any quasicoherent left CDG-module \mathcal{N} over \mathcal{B}_Y the quasicoherent graded left module

 $f_*\mathcal{N}$ over \mathcal{B}_X has a natural structure of quasicoherent CDG-module over \mathcal{B}_X . These CDG-module structures are defined in terms of the CDG-ring morphisms $\mathcal{B}_X(U) \to \mathcal{B}_Y(V)$. The above constructions provide the underived direct and inverse image functors, which can be viewed as triangulated functors $f^*: H^0(\mathcal{B}_X\operatorname{-qcoh}) \to H^0(\mathcal{B}_Y\operatorname{-qcoh})$ and $f_*: H^0(\mathcal{B}_Y\operatorname{-qcoh}) \to H^0(\mathcal{B}_X\operatorname{-qcoh})$. The functor f_* is right adjoint to the functor f^* .

The derived inverse image functor $\mathbb{L}f^*$ is in general only defined on CDG-modules satisfying certain finite flat dimension conditions. Restricting the functor f^* to flat CDG-modules, we obtain a triangulated functor

$$H^{0}(\mathcal{B}_{X}\operatorname{-qcoh}_{\mathsf{fl}}) \longrightarrow H^{0}(\mathcal{B}_{Y}\operatorname{-qcoh}_{\mathsf{fl}}),$$

which takes objects coacyclic with respect to \mathcal{B}_X -qcoh_{fl} to objects coacyclic with respect to \mathcal{B}_Y -qcoh_{fl} since the inverse image preserves infinite direct sums and short exact sequences of flat quasicoherent graded modules. Hence there is the induced triangulated functor $D^{co}(\mathcal{B}_X$ -qcoh_{fl}) $\rightarrow D^{co}(\mathcal{B}_Y$ -qcoh_{fl}). Applying Theorem 1.4(a), we construct the derived inverse image functor

$$\mathbb{L}f^*: \mathsf{D}^{\mathsf{co}}(\mathcal{B}_X\operatorname{-qcoh}_{\mathsf{ffd}}) \longrightarrow \mathsf{D}^{\mathsf{co}}(\mathcal{B}_Y\operatorname{-qcoh}_{\mathsf{ffd}}).$$

Assuming that there are enough vector bundles on X and Y, and restricting the functor f^* to flat coherent CDG-modules, we obtain a triangulated functor $H^0(\mathcal{B}_X\operatorname{-coh}_{fl}) \to H^0(\mathcal{B}_Y\operatorname{-coh}_{fl})$, which induces a triangulated functor

$$D^{abs}(\mathcal{B}_X\operatorname{-coh}_{fl}) \longrightarrow D^{abs}(\mathcal{B}_Y\operatorname{-coh}_{fl}).$$

Assuming additionally that the quasicoherent graded algebras B_X and B_Y are Noetherian and applying Theorem 1.4(c), we construct the derived inverse image functor

$$\mathbb{L}f^*: \mathsf{D}^{\mathsf{abs}}(\mathcal{B}_X\operatorname{-coh}_{\mathsf{ffd}}) \longrightarrow \mathsf{D}^{\mathsf{abs}}(\mathcal{B}_Y\operatorname{-coh}_{\mathsf{ffd}}).$$

When f is an affine morphism, the direct image of quasicoherent sheaves is an exact functor (preserving also infinite direct sums), so the functor $f_*: H^0(\mathcal{B}_Y\operatorname{-qcoh}) \to H^0(\mathcal{B}_X\operatorname{-qcoh})$ induces a triangulated functor $D^{co}(\mathcal{B}_Y\operatorname{-qcoh}) \to D^{co}(\mathcal{B}_X\operatorname{-qcoh})$. To construct the derived direct image functor between the coderived categories in the general case, we need to use injective resolutions.

From now on we assume that \mathcal{B}_X and \mathcal{B}_Y are Noetherian; so Lemma 1.7(b) is applicable to \mathcal{B}_Y . Restricting the functor f_* to the full subcategory $H^0(\mathcal{B}_Y\operatorname{-qcoh}_{\operatorname{inj}}) \subset H^0(\mathcal{B}_Y\operatorname{-qcoh})$ and composing it with the localization functor $H^0(\mathcal{B}_X\operatorname{-qcoh}) \to D^{\operatorname{co}}(\mathcal{B}_X\operatorname{-qcoh})$, we obtain the derived direct image functor

$$\mathbb{R} f_* : \mathsf{D}^{\mathsf{co}}(\mathcal{B}_Y \operatorname{-qcoh}) \longrightarrow \mathsf{D}^{\mathsf{co}}(\mathcal{B}_X \operatorname{-qcoh}).$$

Proposition 1.8. Assume that there are enough vector bundles on X and Y. Then the functors $\mathbb{L}f^* : D^{abs}(\mathcal{B}_X\operatorname{-coh}_{\mathrm{ffd}}) \to D^{abs}(\mathcal{B}_Y\operatorname{-coh}_{\mathrm{ffd}})$ and $\mathbb{R}f_* : D^{co}(\mathcal{B}_Y\operatorname{-qcoh}) \to$ $D^{co}(\mathcal{B}_X\operatorname{-qcoh})$ are partially adjoint to each other in the following sense: for any objects $\mathcal{M} \in D^{abs}(\mathcal{B}_X\operatorname{-coh}_{\mathrm{ffd}})$ and $\mathcal{N} \in D^{co}(\mathcal{B}_Y\operatorname{-qcoh})$, there is a natural isomorphism of abelian groups

 $\operatorname{Hom}_{\mathsf{D}^{\mathrm{co}}(\mathcal{B}_X\operatorname{-qcoh})}(\iota_X\mathcal{M}, \mathbb{R}f_*\mathcal{N}) \simeq \operatorname{Hom}_{\mathsf{D}^{\mathrm{co}}(\mathcal{B}_Y\operatorname{-qcoh})}(\iota_Y\mathbb{L}f^*\mathcal{M}, \mathcal{N}),$

where

1184

$$\iota_X : \mathsf{D}^{\mathsf{abs}}(\mathcal{B}_X\operatorname{-coh}_{\mathsf{ffd}}) \longrightarrow \mathsf{D}^{\mathsf{co}}(\mathcal{B}_X\operatorname{-qcoh}),$$
$$\iota_Y : \mathsf{D}^{\mathsf{abs}}(\mathcal{B}_X\operatorname{-coh}_{\mathsf{ffd}}) \longrightarrow \mathsf{D}^{\mathsf{co}}(\mathcal{B}_Y\operatorname{-qcoh})$$

are the natural fully faithful triangulated functors.

Proof. The functors ι_X and ι_Y are fully faithful by Theorem 1.4(c) and Proposition 1.5(b) and (d). Using Theorem 1.4(c), let us assume that $\mathcal{M} \in \mathsf{D}^{\mathsf{abs}}(\mathcal{B}_X\operatorname{-coh}_{\mathsf{fl}})$. We can also assume that $\mathcal{N} \in H^0(\mathcal{B}_Y\operatorname{-qcoh}_{\mathsf{inj}})$.

Then the left-hand side is the (filtered) inductive limit of

$$\operatorname{Hom}_{H^0(\mathcal{B}_X\operatorname{-qcoh})}(\mathcal{M}'', f_*\mathcal{N})$$

over all morphisms $\mathcal{M}'' \to \mathcal{M}$ in $H^0(\mathcal{B}_X$ -qcoh) with a cone coacyclic with respect to \mathcal{B}_X -qcoh. According to the proofs of Proposition 1.5(b) and [Positselski 2011b, Theorem 3.11.1], any morphism from \mathcal{M} to an object coacyclic with respect to \mathcal{B}_X -qcoh factorizes through an object absolutely acyclic with respect to \mathcal{B}_X -coh_{fl}. Thus the above inductive limit coincides with the similar limit taken over all morphisms $\mathcal{M}' \to \mathcal{M}$ in $H^0(\mathcal{B}_X$ -coh_{fl}) with a cone absolutely acyclic with respect to \mathcal{B}_X -coh_{fl}.

By [loc. cit., Theorem 3.5(a), Remark 3.5, and Lemma 1.3], the right-hand side is isomorphic to $\operatorname{Hom}_{H^0(\mathcal{B}_Y\operatorname{-qcoh})}(f^*\mathcal{M},\mathcal{N})$ and to $\operatorname{Hom}_{H^0(\mathcal{B}_Y\operatorname{-qcoh})}(f^*\mathcal{M}',\mathcal{N})$ since the objects of $H^0(\mathcal{B}_Y\operatorname{-qcoh}_{\operatorname{inj}})$ are right orthogonal to any coacyclic objects in $H^0(\mathcal{B}_Y\operatorname{-qcoh})$. So the assertion follows from the adjointness of the functors f^* and f_* on the level of the homotopy categories of quasicoherent CDG-modules. \Box

Remark 1.8. It is not immediately obvious from the above construction that the derived functor $\mathbb{R}f_*$ is compatible with the compositions; i.e., for $g: Z \to Y$ and $f: Y \to X$, one has $\mathbb{R}(fg)_* \simeq \mathbb{R}f_* \circ \mathbb{R}g_*$. The problem is that the direct image functor f_* does not preserve injectivity of quasicoherent graded modules in general. When the derived direct image functors are adjoint to appropriately defined derived inverse images (see Section 1.9 below for some results of this kind), the problem reduces to checking that the derived inverse images are compatible with the compositions, which may be easier to see from our definitions.

One general approach to this problem is to replace injective quasicoherent graded \mathcal{B} -modules with quasicoherent graded \mathcal{B} -modules that are flabby as sheaves of

graded abelian groups in our construction of the derived direct images. The class of flabby sheaves of abelian groups is closed under infinite direct sums since the underlying topological space of the scheme is Noetherian; it is also always closed under extensions and cokernels of injective morphisms. Whenever the quasicoherent graded algebra \mathcal{B} is Noetherian, all injective quasicoherent graded \mathcal{B} -modules are flabby by Theorem A.3. Therefore, the coderived category of flabby quasicoherent CDG-modules over \mathcal{B} is equivalent to the homotopy category $H^0(\mathcal{B}-qcoh_{inj})$ by a version of Lemma 1.7(b); hence it is also equivalent to the coderived category of all quasicoherent CDG-modules $D^{co}(\mathcal{B}-qcoh)$ (cf. the proof of Proposition 1.7).

The direct images preserve exact triples of flabby sheaves, so derived direct images can be defined using flabby resolutions. The direct images also take flabby sheaves to flabby sheaves; hence the desired compatibility of their derived functors with the compositions of scheme morphisms follows.

Moreover, assuming additionally that the scheme has finite Krull dimension, the absolute derived category of flabby quasicoherent CDG-modules is equivalent to $D^{abs}(\mathcal{B}\text{-qcoh})$ by a dual version of Theorem 1.4(b), as the "flabby dimension" of any quasicoherent graded \mathcal{B} -module is finite. This allows us to define the derived direct images on the absolute derived categories of quasicoherent CDG-modules (another approach to this question is to use the construction from the proof of Proposition 1.9 below). Notice that all our constructions of derived inverse images are also applicable to the categories $D^{abs}(\mathcal{B}\text{-qcoh})$.

Finally, let us point out that for any morphism of quasicoherent CDG-algebras $\mathcal{B}_X \to \mathcal{B}_Y$ with Noetherian underlying quasicoherent graded algebras \mathcal{B}_X and \mathcal{B}_Y compatible with a morphism of separated Noetherian schemes $f : Y \to X$, the functor $\mathbb{R} f_*$ has a right adjoint functor

$$f^{!}: \mathsf{D}^{\mathsf{co}}(\mathcal{B}_{X}\operatorname{-qcoh}) \longrightarrow \mathsf{D}^{\mathsf{co}}(\mathcal{B}_{Y}\operatorname{-qcoh}).$$

Indeed, the triangulated category $D^{co}(\mathcal{B}_Y\text{-qcoh})$ is compactly generated by Proposition 1.5(d), and the functor $\mathbb{R}f_*$ preserves infinite direct sums since the class of injective quasicoherent graded \mathcal{B}_Y -modules is closed under infinite direct sums, due to Noetherianness of \mathcal{B}_Y . So it remains to apply [Neeman 1996, Theorem 4.1].

There is a special situation when one can construct the above functor $f^!$ explicitly. Assume that $f: Y \to X$ is an affine morphism. Let us say that the quasicoherent graded algebra \mathcal{B}_Y is *finite* over \mathcal{B}_X if for any affine open subscheme $U \subset X$, the graded $\mathcal{B}_X(U)$ -module $\mathcal{B}_Y(f^{-1}(U))$ is finitely generated, or in other words, if the quasicoherent graded \mathcal{B}_X -module $f_*\mathcal{B}_Y$ is coherent.

Let $\mathcal{B}_X \to \mathcal{B}_Y$ be a morphism of Noetherian quasicoherent CDG-algebras compatible with an affine morphism of separated Noetherian schemes $f: Y \to X$ such that the quasicoherent graded algebra \mathcal{B}_Y is finite over \mathcal{B}_X . Given a quasicoherent graded left module \mathcal{M} over \mathcal{B}_X , we set $(f^!\mathcal{M})(f^{-1}(U))$ to be the graded left module of homogeneous morphisms (of various degrees) $\operatorname{Hom}_{\mathcal{B}_X(U)}(\mathcal{B}_Y(f^{-1}(U)), \mathcal{M}(U))$ over the graded ring $\mathcal{B}_Y(f^{-1}(U))$ for any affine open subscheme $U \subset X$. Due to the finiteness condition on \mathcal{B}_Y over \mathcal{B}_X , for any affine open subscheme $V \subset U$, there are natural isomorphisms $(f^!\mathcal{M})(f^{-1}(V)) \simeq \mathcal{O}_X(V) \otimes_{\mathcal{O}_X(U)} (f^!\mathcal{M})(f^{-1}(U)) \simeq$ $\mathcal{O}_Y(f^{-1}(V)) \otimes_{\mathcal{O}_Y(f^{-1}(U))} (f^!\mathcal{M})(f^{-1}(U))$, which allow us to extend the assignment $f^{-1}(U) \mapsto (f^!\mathcal{M})(f^{-1}(U))$ to a quasicoherent graded module $f^!(\mathcal{M})$ over the quasicoherent graded algebra \mathcal{B}_Y .

Given a quasicoherent CDG-module \mathcal{M} over \mathcal{B}_X , the conventional rule

$$d(g)(m) = d(g(m)) - (-1)^{|g|}g(d(m))$$

(with the usual change-of-connection modifications) defines the structure of a quasicoherent CDG-module over \mathcal{B}_Y on the quasicoherent graded module $f^!(\mathcal{M})$. This construction provides a triangulated functor $f^!: H^0(\mathcal{B}_X\operatorname{-qcoh}) \to H^0(\mathcal{B}_Y\operatorname{-qcoh})$ right adjoint to the triangulated functor $f_*: H^0(\mathcal{B}_Y\operatorname{-qcoh}) \to H^0(\mathcal{B}_X\operatorname{-qcoh})$. Restricting the functor $f^!: H^0(\mathcal{B}_X\operatorname{-qcoh}) \to H^0(\mathcal{B}_Y\operatorname{-qcoh})$ to the full subcategory of injective quasicoherent CDG-modules in $H^0(\mathcal{B}_X\operatorname{-qcoh})$ and taking into account Lemma 1.7(b), we obtain the right derived functor

$$\mathbb{R}f^!: \mathsf{D}^{\mathsf{co}}(\mathcal{B}_X\operatorname{-qcoh}) \longrightarrow \mathsf{D}^{\mathsf{co}}(\mathcal{B}_Y\operatorname{-qcoh}),$$

which is right adjoint to the (underived, as the morphism f is affine) direct image functor $f_* : D^{co}(\mathcal{B}_Y\text{-qcoh}) \to D^{co}(\mathcal{B}_X\text{-qcoh})$. In other words, the functor $\mathbb{R}f^!$ coincides with the above adjoint functor $f^! : D^{co}(\mathcal{B}_X\text{-qcoh}) \to D^{co}(\mathcal{B}_Y\text{-qcoh})$ in our special case.

1.9. *Morphisms of finite flat dimension.* Let $f : Y \to X$ be a morphism of separated Noetherian schemes, and let $\mathcal{B}_X \to \mathcal{B}_Y$ be a compatible morphism of quasicoherent CDG-algebras. We will say that the quasicoherent graded algebra \mathcal{B}_Y has finite flat dimension over \mathcal{B}_X if (the left derived functor of) the functor of inverse image f^* acting between the abelian categories of quasicoherent graded modules over \mathcal{B}_X and \mathcal{B}_Y has finite homological dimension. Equivalently, for any affine open subschemes $U \subset X$ and $V \subset Y$ such that $f(V) \subset U$, the graded right $\mathcal{B}_X(U)$ -module $\mathcal{B}_Y(V)$ should have finite flat dimension.

A quasicoherent graded \mathcal{B}_X -module is said to be *adjusted to* f^* if its derived inverse image under f, as an object of the derived category of the abelian category of quasicoherent graded \mathcal{B}_Y -modules, coincides with the underived inverse image. Denote the DG-category of quasicoherent CDG-modules over \mathcal{B}_X whose underlying graded \mathcal{B}_X -modules are adjusted to f^* by \mathcal{B}_X -qcoh_{*f*-adj}. When \mathcal{B}_X is Noetherian, let \mathcal{B}_X -coh_{*f*-adj} denote the similarly defined DG-category of coherent CDG-modules. We will use our usual notation for the absolute derived and coderived categories of these DG-categories of CDG-modules.

Lemma 1.9. Assume that the quasicoherent graded algebra \mathcal{B}_Y has finite flat dimension over \mathcal{B}_X . Then

- (a) the functor $D^{co}(\mathcal{B}_X\operatorname{-qcoh}_{f\operatorname{-adj}}) \to D^{co}(\mathcal{B}_X\operatorname{-qcoh})$ induced by the embedding of DG-categories $\mathcal{B}_X\operatorname{-qcoh}_{f\operatorname{-adj}} \to \mathcal{B}_X\operatorname{-qcoh}$ is an equivalence of triangulated categories;
- (b) the functor D^{abs}(B_X-qcoh_{f-adj}) → D^{abs}(B_X-qcoh) induced by the embedding of DG-categories B_X-qcoh_{f-adj} → B_X-qcoh is an equivalence of triangulated categories;
- (c) if there are enough vector bundles on X and \mathcal{B}_X is Noetherian, the functor $\mathsf{D}^{\mathsf{abs}}(\mathcal{B}_X\operatorname{-coh}_{f\operatorname{-adj}}) \to \mathsf{D}^{\mathsf{abs}}(\mathcal{B}_X\operatorname{-coh})$ induced by the embedding of DG-categories $\mathcal{B}_X\operatorname{-coh}_{f\operatorname{-adj}} \to \mathcal{B}_X\operatorname{-coh}$ is an equivalence of triangulated categories.

Proof. This is a version of Theorem 1.4, provable in the same way (cf. Corollary 2.6 below). The assertions hold, because any quasicoherent graded \mathcal{B}_X -module has a finite left resolution consisting of quasicoherent CDG-modules adjusted to f^* , and it is similar for coherent CDG-modules.

The functor of inverse image $f^*: H^0(\mathcal{B}_X\operatorname{-qcoh}) \to H^0(\mathcal{B}_Y\operatorname{-qcoh})$ takes CDGmodules coacyclic with respect to $\mathcal{B}_X\operatorname{-qcoh}_{f\operatorname{-adj}}$ to CDG-modules coacyclic with respect to $\mathcal{B}_Y\operatorname{-qcoh}$, and hence induces a triangulated functor $D^{\operatorname{co}}(\mathcal{B}_X\operatorname{-qcoh}_{f\operatorname{-adj}}) \to D^{\operatorname{co}}(\mathcal{B}_Y\operatorname{-qcoh})$. Taking Lemma 1.9 into account, we construct the derived inverse image functor

$$\mathbb{L}f^*: \mathsf{D}^{\mathsf{co}}(\mathcal{B}_X\operatorname{-qcoh}) \longrightarrow \mathsf{D}^{\mathsf{co}}(\mathcal{B}_Y\operatorname{-qcoh}).$$

One shows that this functor is left adjoint to the functor $\mathbb{R} f_*$ constructed in Section 1.8 in the way analogous to (but simpler than) the proof of Proposition 1.8.

When there are enough vector bundles on X, and \mathcal{B}_X and \mathcal{B}_Y are Noetherian, we construct the derived inverse image functor

 $\mathbb{L}f^*: \mathsf{D}^{\mathsf{abs}}(\mathcal{B}_X\operatorname{-\mathsf{coh}}) \longrightarrow \mathsf{D}^{\mathsf{abs}}(\mathcal{B}_Y\operatorname{-\mathsf{coh}})$

in a similar way.

Let $\mathcal{B}_X^{\text{op}}$ and $\mathcal{B}_Y^{\text{op}}$ denote the quasicoherent graded algebras with the opposite multiplication to \mathcal{B}_X and \mathcal{B}_Y .

Proposition 1.9. When $\mathcal{B}_Y^{\text{op}}$ has finite flat dimension over $\mathcal{B}_X^{\text{op}}$, the derived inverse image functor $\mathbb{L}f^*$: $\mathsf{D}^{\text{co}}(\mathcal{B}_X\operatorname{-qcoh}_{\mathrm{ffd}}) \to \mathsf{D}^{\text{co}}(\mathcal{B}_Y\operatorname{-qcoh}_{\mathrm{ffd}})$ constructed in Section 1.8 has a right adjoint functor

$$\mathbb{R}f_*: \mathsf{D}^{\mathsf{co}}(\mathcal{B}_Y\operatorname{-qcoh}_{\mathsf{ffd}}) \longrightarrow \mathsf{D}^{\mathsf{co}}(\mathcal{B}_X\operatorname{-qcoh}_{\mathsf{ffd}}).$$

Proof. Let $\{U_{\alpha}\}$ be a finite affine covering of Y. To any object $\mathcal{N} \in \mathcal{B}_{Y}$ -qcoh_{ffd}, assign the total CDG-module $\mathbb{R}_{\{U_{\alpha}\}}f_*\mathcal{N}$ of the finite Čech complex

$$\bigoplus_{\alpha} f|_{U_{\alpha}} * (\mathcal{N}|_{U_{\alpha}}) \longrightarrow \bigoplus_{\alpha < \beta} f|_{U_{\alpha} \cap \mathcal{U}_{\beta}} * (\mathcal{N}|_{U_{\alpha} \cap \mathcal{U}_{\beta}}) \longrightarrow \cdots$$

of CDG-modules over \mathcal{B}_X .

The terms of this complex belong to \mathcal{B}_X -qcoh_{ffd} since the morphism $f|_V : V \to X$ is affine for any intersection V of a nonempty subset of affine open subschemes $U_{\alpha} \subset Y$ and the quasicoherent graded algebra $\mathcal{B}_Y^{\text{op}}$ has finite flat dimension over $\mathcal{B}_X^{\text{op}}$. Hence one has $\mathbb{R}_{\{U_{\alpha}\}} f_* \mathcal{N} \in \mathcal{B}_X$ -qcoh_{ffd}; it is clear that $\mathbb{R}_{\{U_{\alpha}\}} f_*$ is a DG-functor \mathcal{B}_Y -qcoh_{ffd} $\to \mathcal{B}_X$ -qcoh_{ffd} taking coacyclic objects to coacyclic objects. So we have the induced functor $\mathbb{R} f_*$ between the coderived categories.

It remains to obtain the adjunction isomorphism

 $\operatorname{Hom}_{\mathsf{D}^{co}(\mathcal{B}_{X}\operatorname{-\mathsf{qcoh}_{ffd}})}(\mathcal{M}, \mathbb{R}f_*\mathcal{N}) \simeq \operatorname{Hom}_{\mathsf{D}^{co}(\mathcal{B}_{Y}\operatorname{-\mathsf{qcoh}_{ffd}})}(\mathbb{L}f^*\mathcal{M}, \mathcal{N})$

for $\mathcal{M} \in \mathsf{D}^{co}(\mathcal{B}_X\operatorname{-qcoh}_{\mathrm{ffd}})$. Denote by \mathcal{N}_+ the total CDG-module of the finite complex

$$C^{\bullet}_{\{U_{\alpha}\}}\mathcal{N} = \left(\bigoplus_{\alpha} j_{U_{\alpha}} * j^{*}_{U_{\alpha}}\mathcal{N} \longrightarrow \bigoplus_{\alpha < \beta} j_{U_{\alpha} \cap \mathcal{U}_{\beta}} * j^{*}_{U_{\alpha} \cap \mathcal{U}_{\beta}}\mathcal{N} \longrightarrow \cdots\right)$$

of CDG-modules over \mathcal{B}_Y (where $j_V : V \to Y$ denotes the embedding of an affine open subscheme). Then we have $\mathbb{R}_{\{U_\alpha\}} f_* \mathcal{N} \simeq f_* \mathcal{N}_+$. There is a natural closed morphism $\mathcal{N} \to \mathcal{N}_+$ of CDG-modules over \mathcal{B}_Y with the cone coacyclic (and even absolutely acyclic) with respect to \mathcal{B}_Y -qcoh_{ffd}.

For any CDG-module $Q \in B_Y$ -qcoh_{ffd} such that $f_*Q \in B_X$ -qcoh_{ffd}, there is a natural map

$$\psi : \operatorname{Hom}_{\mathsf{D}^{\mathrm{co}}(\mathcal{B}_{X}\operatorname{-\mathsf{qcoh}}_{\operatorname{ffd}})}(\mathcal{M}, f_{*}\mathcal{Q}) \longrightarrow \operatorname{Hom}_{\mathsf{D}^{\mathrm{co}}(\mathcal{B}_{Y}\operatorname{-\mathsf{qcoh}}_{\operatorname{ffd}})}(\mathbb{L}f^{*}\mathcal{M}, \mathcal{Q}).$$

Indeed, by the proof of Theorem 1.4(a), any morphism $\mathcal{M} \to f_*\mathcal{Q}$ in $\mathsf{D}^{\mathsf{co}}(\mathcal{B}_X\operatorname{-qcoh}_{\mathsf{ffd}})$ can be represented as a fraction formed by a morphism $\mathcal{M}' \to \mathcal{M}$ in $H^0(\mathcal{B}_X\operatorname{-qcoh}_{\mathsf{ffd}})$, with $\mathcal{M}' \in \mathcal{B}_X\operatorname{-qcoh}_{\mathsf{fl}}$ and a cone coacyclic with respect to $\mathcal{B}_X\operatorname{-qcoh}_{\mathsf{ffd}}$, and a morphism $\mathcal{M}' \to f_*\mathcal{Q}$ in $H^0(\mathcal{B}_X\operatorname{-qcoh}_{\mathsf{ffd}})$. To such a fraction, the map ψ assigns the related morphism $\mathbb{L} f^*\mathcal{M} = f^*\mathcal{M}' \to \mathcal{Q}$.

For a fixed \mathcal{M} , the map ψ is a morphism of cohomological functors of the argument $\mathcal{Q} \in H^0(\mathcal{B}_Y\operatorname{-qcoh}_{\mathrm{ffd}})$ with $f_*\mathcal{Q} \in \mathcal{B}_X\operatorname{-qcoh}_{\mathrm{ffd}}$. Thus in order to show that it is an isomorphism for $\mathcal{Q} = \mathcal{N}_+$, it suffices to check that it is an isomorphism for $\mathcal{Q} = j_{V*}\mathcal{P}$ for every affine $V \subset Y$ and $\mathcal{P} \in \mathcal{B}_Y|_V\operatorname{-qcoh}_{\mathrm{ffd}}$. This follows from the adjunction isomorphism

$$\operatorname{Hom}_{\mathsf{D}^{\mathrm{co}}(\mathcal{B}_{X}\operatorname{-\mathsf{qcoh}}_{\mathrm{ffd}})}(\mathcal{M}, f|_{V} * \mathcal{P}) \simeq \operatorname{Hom}_{\mathsf{D}^{\mathrm{co}}(\mathcal{B}_{Y}|_{V}\operatorname{-\mathsf{qcoh}}_{\mathrm{ffd}})}(\mathbb{L}f|_{V}^{*}\mathcal{M}, \mathcal{P})$$

and the similar isomorphism for the embedding j_V , which hold because the functors $f|_{V*}$ and j_{V*} are exact, the morphisms $f|_V$ and j_V being affine.

Remark 1.9. One can also use the above Čech complex approach in order to construct a version of the derived functor $\mathbb{R} f_* : D^{co}(\mathcal{B}_Y\text{-qcoh}) \to D^{co}(\mathcal{B}_X\text{-qcoh})$. One can check that this construction agrees with the injective resolution construction from Section 1.8, using the fact that the restrictions of injective quasicoherent graded \mathcal{B}_Y -modules to open subschemes are injective (Theorem A.3). Alternatively, in the assumption of finite flat dimension of \mathcal{B}_Y over \mathcal{B}_X , one checks that both constructions provide functors right adjoint to $\mathbb{L} f^*$, hence they are isomorphic.

This allows us to conclude that the derived functors $\mathbb{R} f_*$ acting on arbitrary quasicoherent CDG-modules and quasicoherent CDG-modules of finite flat dimension form a commutative diagram with the natural functors from the coderived categories of the latter to the coderived categories of the former.

1.10. Supports of CDG-modules. Let X be a Noetherian scheme. The *set-theoretic* support of a quasicoherent sheaf \mathcal{M} on X is the minimal closed subset $T \subset X$ such that the restriction of \mathcal{M} to the open subscheme $X \setminus T$ vanishes. Given a Noetherian quasicoherent graded algebra \mathcal{B} over X and a quasicoherent graded \mathcal{B} -module \mathcal{M} , the set-theoretic support $T = \text{Supp } \mathcal{M}$ of \mathcal{M} is defined similarly. It only depends on the underlying quasicoherent \mathcal{O}_X -module of \mathcal{M} .

Let \mathcal{B} be a quasicoherent CDG-algebra over X whose underlying quasicoherent graded algebra \mathcal{B} is Noetherian. Fix a closed subset $T \subset X$. Denote by \mathcal{B} -qcoh_T the full DG-subcategory in \mathcal{B} -qcoh consisting of all the quasicoherent CDG-modules whose underlying quasicoherent graded \mathcal{B} -modules have their set-theoretic supports contained in T. The DG-category \mathcal{B} -coh_T of coherent CDG-modules with settheoretic support in T is defined similarly.

Let $D^{co}(\mathcal{B}\operatorname{-qcoh}_T)$ and $D^{abs}(\mathcal{B}\operatorname{-coh}_T)$ denote the coderived and the absolute derived category of these DG-categories of CDG-modules. Finally, let $\mathcal{B}\operatorname{-qcoh}_{T,\operatorname{inj}}$ denote the DG-category of quasicoherent CDG-modules over \mathcal{B} whose underlying quasicoherent graded modules are injective objects of the abelian category of quasicoherent graded \mathcal{B} -modules with set-theoretic support contained in T.

- **Proposition 1.10.** (a) The functor $H^0(\mathcal{B}\operatorname{-qcoh}_{T,\operatorname{inj}}) \to \mathsf{D^{co}}(\mathcal{B}\operatorname{-qcoh}_T)$ induced by the embedding of DG-categories $\mathcal{B}\operatorname{-qcoh}_{T,\operatorname{inj}} \to \mathcal{B}\operatorname{-qcoh}_T$ is an equivalence of triangulated categories.
- (b) The functor D^{abs}(B-coh_T) → D^{co}(B-qcoh_T) induced by the embedding of DG-categories B-coh_T → B-qcoh_T is fully faithful and its image is a set of compact generators of the target category.
- (c) The functor $D^{co}(\mathcal{B}\operatorname{-qcoh}_T) \to D^{co}(\mathcal{B}\operatorname{-qcoh})$ induced by the embedding of $DG\operatorname{-categories} \mathcal{B}\operatorname{-qcoh}_T \to \mathcal{B}\operatorname{-qcoh}$ is fully faithful.

(d) The functor $D^{abs}(\mathcal{B}\text{-}coh_T) \to D^{abs}(\mathcal{B}\text{-}coh)$ induced by the embedding of DGcategories $\mathcal{B}\text{-}coh_T \to \mathcal{B}\text{-}coh$ is fully faithful.

Proof. Part (a) is essentially a particular case of [Positselski 2011b, Theorem and Remark in Section 3.7]. It is only important here that there are enough injective objects in the abelian category of quasicoherent graded \mathcal{B} -modules supported settheoretically in T and the class of such injective objects is closed under infinite direct sums. This is so because the abelian category in question is a locally Noetherian Grothendieck category (since X and \mathcal{B} are Noetherian). Part (b) can be proven in the same way as the results of [loc. cit., Section 3.11]. Part (d) follows from parts (b) and (c) and Proposition 1.5(d).

Finally, part (c) follows from part (a), Lemma 1.7(b), and the fact that any injective object \mathcal{J} in the category of quasicoherent graded \mathcal{B} -modules supported set-theoretically in T is also an injective object in the category of arbitrary quasicoherent graded \mathcal{B} -modules. The latter is essentially a reformulation of the Artin–Rees lemma.

Indeed, it suffices to check that for any coherent graded \mathcal{B} -module \mathcal{M} and its coherent graded \mathcal{B} -submodule \mathcal{N} , any morphism of quasicoherent graded \mathcal{B} -modules $\phi : \mathcal{N} \to \mathcal{J}$ can be extended to \mathcal{M} . Let Z be a closed subscheme structure on the closed subset $T \subset X$. Then there is an integer $n \ge 0$ such that the morphism ϕ annihilates $\mathcal{I}_Z^n \mathcal{N}$ (where \mathcal{I}_Z is the sheaf of ideals of the closed subscheme Z). By Lemma A.3, there exists $m \ge 0$ such that $\mathcal{I}_Z^m \mathcal{M} \cap \mathcal{N} \subset \mathcal{I}_Z^n \mathcal{N}$. Then there exists a morphism $\mathcal{M}/\mathcal{I}_Z^m \mathcal{M} \to \mathcal{J}$ of quasicoherent graded \mathcal{B} -modules supported settheoretically in T which extends the given morphism into \mathcal{J} from the quasicoherent graded \mathcal{B} -submodule $\mathcal{N}/(\mathcal{I}_Z^m \mathcal{M} \cap \mathcal{N}) \subset \mathcal{M}/\mathcal{I}_Z^m \mathcal{M}$.

Let $U \subset X$ denote the open subscheme $X \setminus T$.

Theorem 1.10. (a) The functor of restriction to the open subscheme $D^{co}(\mathcal{B}\operatorname{-qcoh}) \rightarrow D^{co}(\mathcal{B}|_U\operatorname{-qcoh})$ is the Verdier localization functor by the thick subcategory

$$\mathsf{D}^{\mathsf{co}}(\mathcal{B}\operatorname{-qcoh}_T) \subset \mathsf{D}^{\mathsf{co}}(\mathcal{B}\operatorname{-qcoh}).$$

In particular, the kernel of the restriction functor coincides with the subcategory $D^{co}(\mathcal{B}\operatorname{-qcoh}_T)$.

(b) The functor of restriction to the open subscheme $D^{abs}(\mathcal{B}\text{-}coh) \rightarrow D^{abs}(\mathcal{B}|_U\text{-}coh)$ is the Verdier localization functor by the triangulated subcategory

$$D^{abs}(\mathcal{B}\text{-}coh_T) \subset D^{abs}(\mathcal{B}\text{-}coh).$$

In particular, the kernel of the restriction functor coincides with the thick envelope of (i.e., the minimal thick subcategory containing) $D^{abs}(\mathcal{B}\text{-}coh_T)$ in $D^{abs}(\mathcal{B}\text{-}coh)$.

Proof. Let $j: U \to X$ denote the natural open embedding. To prove part (a), consider the functor $\mathbb{R}j_*: D^{co}(\mathcal{B}|_U \operatorname{-qcoh}) \to D^{co}(\mathcal{B}\operatorname{-qcoh})$ as constructed in Section 1.8. Since the quasicoherent graded algebra $\mathcal{B}|_U$ is flat over \mathcal{B} , the functor $\mathbb{R}j_*$ is right adjoint to the restriction functor $j^*: D^{co}(\mathcal{B}\operatorname{-qcoh}) \to D^{co}(\mathcal{B}|_U \operatorname{-qcoh})$. Obviously, the composition $j^*\mathbb{R}j_*$ is the identity functor. It follows that the functor j^* is a Verdier localization functor by its kernel, which is the full subcategory consisting of all the cones of the adjunction morphisms $\mathcal{M} \to \mathbb{R}j_*j^*\mathcal{M}$, where $\mathcal{M} \in D^{co}(\mathcal{B}\operatorname{-qcoh})$.

Represent the object \mathcal{M} by a CDG-module with an injective underlying quasicoherent graded \mathcal{B} -module. By Theorem A.3, the quasicoherent graded $\mathcal{B}|_U$ -module $j^*\mathcal{M}$ is then also injective, so we have $\mathbb{R}j_*j^*\mathcal{M} = j_*j^*\mathcal{M}$. Obviously, both the kernel and the cokernel of the closed morphism of CDG-modules $\mathcal{M} \to j_*j^*\mathcal{M}$ belong to \mathcal{B} -qcoh_T, and it follows, in view of Proposition 1.10(c), that the cone also belongs to $\mathcal{D}^{co}(\mathcal{B}$ -qcoh_T).

To prove part (b), notice first that any coherent CDG-module over $\mathcal{B}|_U$ can be extended to a coherent CDG-module over \mathcal{B} (because a coherent sheaf \mathcal{K} on U can be extended to a coherent subsheaf of $j_*\mathcal{K}$), so the restriction functor is essentially surjective. Taking this observation into account, part (b) follows from part (a), Proposition 1.10(b), Proposition 1.5(d), and the standard results about localization of compactly generated triangulated categories [Neeman 1992, Lemma 2.5 to Theorem 2.1].

Define the *category-theoretic support* supp \mathcal{M} of a quasicoherent CDG-module \mathcal{M} over \mathcal{B} as the minimal closed subset $T \subset X$ such that the restriction $\mathcal{M}|_U$ of \mathcal{M} to the open subscheme $U = X \setminus T$ is a coacyclic CDG-module over $\mathcal{B}|_U$. In other words, $X \setminus \text{supp } \mathcal{M}$ is the union of all open subschemes $V \subset X$ such that $\mathcal{M}|_V$ is a coacyclic CDG-module over $\mathcal{B}|_V$ (see Remark 1.3). Obviously, one has supp $\mathcal{M} \subset \text{Supp } \mathcal{M}$.

The category-theoretic support of a coherent CDG-module \mathcal{M} over \mathcal{B} can be equivalently defined as the minimal closed subset $T \subset X$ such that the restriction $\mathcal{M}|_U$ of \mathcal{M} to the open subscheme $U = X \setminus T$ is absolutely acyclic. Indeed, any CDG-module from $\mathcal{B}|_U$ -coh that is coacyclic with respect to $\mathcal{B}|_U$ -qcoh is also absolutely acyclic with respect to $\mathcal{B}|_U$ -coh by Proposition (d).

- **Corollary 1.10.** (a) For any quasicoherent CDG-module \mathcal{M} over \mathcal{B} with categorytheoretic support supp \mathcal{M} contained in T, there exists a quasicoherent CDGmodule \mathcal{M}' over \mathcal{B} such that \mathcal{M} is isomorphic to \mathcal{M}' in $D^{co}(\mathcal{B}-qcoh)$ and set-theoretic support Supp \mathcal{M}' is contained in T.
- (b) For any coherent CDG-module M over B with category-theoretic support supp M contained in T, there exists a coherent CDG-module M' over B such that M is isomorphic to a direct summand of M' in D^{abs}(B-coh) and set-theoretic support Supp M' is contained in T.

Proof. Follows immediately from Theorem 1.10.

Remark 1.10. One can prove that the restriction functor in Theorem 1.10(a) is a Verdier localization functor without assuming the quasicoherent graded algebra \mathcal{B} to be Noetherian. Indeed, one can construct a right adjoint functor \mathbb{R}_{j*} to the restriction functor j^* in the way similar to that of Proposition 1.9; then it is easy to see that $j^*\mathbb{R}_{j*}$ is the identity functor.

When \mathcal{B} is Noetherian, Theorem 1.10 can be generalized as follows. Let S and T be closed subsets in X; set $U = X \setminus T$. Then the restriction functor $\mathsf{D}^{\mathsf{co}}(\mathcal{B}\operatorname{-qcoh}_S) \to \mathsf{D}^{\mathsf{co}}(\mathcal{B}|_U\operatorname{-qcoh}_{U\cap S})$ is the Verdier localization functor by the thick subcategory $\mathsf{D}^{\mathsf{co}}(\mathcal{B}\operatorname{-qcoh}_{T\cap S})$, and the restriction functor $\mathsf{D}^{\mathsf{abs}}(\mathcal{B}\operatorname{-coh}_S) \to \mathsf{D}^{\mathsf{abs}}(\mathcal{B}|_U\operatorname{-coh}_{U\cap S})$ is the Verdier localization functor $\mathsf{D}^{\mathsf{abs}}(\mathcal{B}\operatorname{-coh}_S) \to \mathsf{D}^{\mathsf{abs}}(\mathcal{B}|_U\operatorname{-coh}_{U\cap S})$ is the Verdier localization functor by the triangulated subcategory $\mathsf{D}^{\mathsf{abs}}(\mathcal{B}\operatorname{-coh}_{T\cap S})$. The proof is similar to the above.

It is not difficult to deduce from the latter assertions, using the result of [Neeman 1996, Theorem 2.1(5)], that the property of an object of $D^{co}(\mathcal{B}\text{-qcoh})$ to belong to the thick envelope of $D^{abs}(\mathcal{B}\text{-coh})$ is local in X. Using the Čech exact sequence as in Remark 1.3, one can easily see that the property of an object of $D^{abs}(\mathcal{B}\text{-qcoh})$ to belong to $D^{abs}(\mathcal{B}\text{-qcoh})$ is also local.

We do not know whether the property of an object of $D^{abs}(\mathcal{B}\text{-coh})$ or $D^{abs}(\mathcal{B}\text{-qcoh}_{fl})$ to belong to $D^{abs}(\mathcal{B}\text{-coh}_{fl})$ is local in general. In the particular case of matrix factorizations, such results will be proven in Section 3.2 using the connection with singularity categories (cf. Remark 3.6).

Notice that the theory of localization for compactly generated triangulated categories, on which the proof of Theorem 1.10(b) is based, was originally applied in algebraic geometry for the purposes of the Thomason–Trobaugh localization theory of perfect complexes. In this section we apply it to coherent CDG-modules. In fact, we will see in Section 3.3 that the localization theory *fails* for locally free matrix factorizations of finite rank.

2. Triangulated categories of relative singularities

2.1. *Relative singularity category.* Recall that X denotes a separated Noetherian scheme with enough vector bundles. The *triangulated category of singularities* $D^{b}_{Sing}(X)$ of the scheme X is defined [Orlov 2004, Section 1.2] as the quotient category of the bounded derived category D^b(X-coh) of coherent sheaves on X by its thick subcategory *Perf*(X) of perfect complexes on X.

The perfect complexes, in our assumptions, can be simply defined as bounded complexes of locally free sheaves of finite rank, so $Perf(X) = D^{b}(X \cdot coh_{if})$ is the bounded derived category of the exact category X-coh_{if} of locally free sheaves of finite rank on X. Equivalently, the perfect complexes are the compact objects of the

unbounded derived category of quasicoherent sheaves D(X-qcoh) on the scheme X [Neeman 1996, Examples 1.10–1.11 and Corollary 2.3].

Let $Z \subset X$ be a closed subscheme such that \mathcal{O}_Z has finite flat dimension as an \mathcal{O}_X -module. In this case the derived inverse image functor $\mathbb{L}i^*$ for the closed embedding $i : Z \to X$ acts on the bounded derived categories of coherent sheaves, $D^b(X\text{-coh}) \to D^b(Z\text{-coh})$. We call the quotient category of $D^b(Z\text{-coh})$ by the thick subcategory generated by the objects in the image of this functor the *triangulated category of singularities of Z relative to X* and denote it by $D^b_{Sing}(Z/X)$.

Note that the triangulated category of relative singularities $D_{Sing}^{b}(Z/X)$ is a quotient category of the conventional (absolute) triangulated category of singularities $D_{Sing}^{b}(Z)$ of the scheme Z. Indeed, the thick subcategory $Perf(Z) \subset D^{b}(Z-coh)$ is generated by any ample family of vector bundles on Z since any such family is a set of compact generators of the unbounded derived category of quasicoherent sheaves D(Z-qcoh) on Z [Neeman 1996]; in particular, it is generated by the restrictions to Z of vector bundles from X (see also Lemma 2.8).

The functor $\mathbb{L}i^*$: $D^b(X-coh) \rightarrow D^b(Z-coh)$ induces a triangulated functor

$$i^{\circ}: \mathsf{D}^{\mathsf{b}}_{Sing}(X) \longrightarrow \mathsf{D}^{\mathsf{b}}_{Sing}(Z).$$

Furthermore, since the sheaf $i_*\mathcal{O}_Z$ belongs to Perf(X), the functor $i_*:D^b(Z\operatorname{-coh}) \to D^b(X\operatorname{-coh})$ takes Perf(Z) to Perf(X) (cf. [Orlov 2004, paragraphs before Proposition 1.14]). Hence the functor i_* induces a triangulated functor $i_o:D^b_{Sing}(Z) \to D^b_{Sing}(X)$ right adjoint to i° . The triangulated category $D^b_{Sing}(Z/X)$ is the quotient category of $D^b_{Sing}(Z)$ by the thick subcategory generated by the image of the functor i° .

When X is regular, any coherent sheaf on X has a finite resolution by locally free sheaves of finite rank. So $D_{Sing}^{b}(X) = 0$, and hence the triangulated categories $D_{Sing}^{b}(Z)$ and $D_{Sing}^{b}(Z/X)$ coincide. The converse is also true: the structure sheaf of the reduced scheme structure on the closure of any singular point of X is not a perfect complex on X, so $D_{Sing}^{b}(X) \neq 0$ when X is not regular.

Remark 2.1. Roughly speaking, the triangulated category of relative singularities $D_{Sing}^{b}(Z/X)$ measures how much worse the singularities of Z are compared to the singularities of X in a neighborhood of Z.

The basic formal properties of $D_{Sing}^{b}(Z/X)$ are similar to those of $D_{Sing}^{b}(Z)$. When the \mathcal{O}_{X} -module \mathcal{O}_{Z} has finite flat dimension, the derived category $D^{b}(X-coh)$ is generated by coherent sheaves adjusted to i^{*} . Let $\mathbb{E}_{Z/X}$ denote the minimal full subcategory of the abelian category of coherent sheaves on Z containing the restrictions of such coherent sheaves from X and closed under extensions and the kernels of epimorphisms of sheaves. Then $\mathbb{E}_{Z/X}$ is naturally an exact category and its bounded derived category $D^{b}(\mathbb{E}_{Z/X})$ is equivalent to the thick subcategory of $D^{b}(Z\operatorname{-coh})$ generated by the derived restrictions of coherent sheaves from X, so $D^{b}_{Sing}(Z/X) = D^{b}(Z\operatorname{-coh})/D^{b}(E_{Z/X})$. One can define the E-homological dimension of a coherent sheaf (or bounded complex) on Z as the minimal length of a left resolution consisting of objects from $E_{Z/X}$. This dimension does not depend on the choice of a resolution (in the same sense that the conventional flat dimension doesn't). The thick subcategory $D^{b}(E_{Z/X})$ consists of those objects of $D^{b}(Z\operatorname{-coh})$ that have finite E-homological dimensions.

Unlike in the case of perfect complexes, we do not know whether the property to belong to $E_{Z/X}$ or $D^{b}(E_{Z/X})$ is local, though. In the case when Z is a Cartier divisor, locality can be established using Theorem 2.7 below and Remark 1.3.

2.2. *Matrix factorizations.* Following [Polishchuk and Vaintrob 2011], we will consider matrix factorizations of a global section of a line bundle. So let \mathcal{L} be a line bundle (invertible sheaf) on X and $w \in \mathcal{L}(X)$ be a fixed section, called the *potential*.

Let $\mathcal{B} = (X, \mathcal{L}, w)$ denote the following \mathbb{Z} -graded quasicoherent CDG-algebra over X. The component \mathcal{B}^n is isomorphic to $\mathcal{L}^{\otimes n/2}$ for $n \in 2\mathbb{Z}$ and vanishes for $n \in 2\mathbb{Z} + 1$, the multiplication in \mathcal{B} being given by the natural isomorphisms $\mathcal{L}^{\otimes n/2} \otimes_{\mathcal{O}_X} \mathcal{L}^{\otimes m/2} \to \mathcal{L}^{\otimes (n+m)/2}$. For any affine open subscheme $U \subset X$, the differential on $\mathcal{B}(U)$ is zero, and the curvature element is $w|_U \in \mathcal{B}^2(U) = \mathcal{L}(U)$. The elements a_{UV} defining the restriction morphisms of CDG-rings $\mathcal{B}(V) \to \mathcal{B}(U)$ all vanish.

The category of quasicoherent \mathbb{Z} -graded \mathcal{B} -modules is equivalent to the category of quasicoherent $\mathbb{Z}/2$ -graded \mathcal{O}_X -modules, the equivalence assigning to a graded \mathcal{B} -module \mathcal{M} the pair of \mathcal{O}_X -modules which we denote symbolically by $\mathcal{U}^0 = \mathcal{M}^0$ and $\mathcal{U}^1 \otimes \mathcal{L}^{\otimes 1/2} = \mathcal{M}^1$. Conversely, $\mathcal{M}^n \simeq \mathcal{U}^{n \mod 2} \otimes_{\mathcal{O}_X} \mathcal{L}^{\otimes n/2}$ for all $n \in \mathbb{Z}$ (the meaning of the notation in the right-hand side being the obvious one). This equivalence of abelian categories preserves all the properties of coherence, flatness, flat dimension, local projectivity/local freeness, etc. that we were interested in in Section 1.

Following [Lin and Pomerleano 2013], we will consider CDG-modules over $\mathcal{B} = (X, \mathcal{L}, w)$ whose underlying graded \mathcal{B} -modules correspond to coherent or quasicoherent \mathcal{O}_X -modules, rather than just locally free sheaves (as in the conventional matrix factorizations). A quasicoherent CDG-module over (X, \mathcal{L}, w) is the same thing as a pair of quasicoherent \mathcal{O}_X -modules \mathcal{U}^0 and $\mathcal{U}^1 \otimes \mathcal{L}^{\otimes 1/2}$ endowed with \mathcal{O}_X -linear morphisms $\mathcal{U}^0 \to \mathcal{U}^1 \otimes \mathcal{L}^{\otimes 1/2}$ and $\mathcal{U}^1 \otimes \mathcal{L}^{\otimes 1/2} \to \mathcal{U}^0 \otimes_{\mathcal{O}_X} \mathcal{L}$ such that both compositions

$$\mathcal{U}^{0} \to \mathcal{U}^{1} \otimes \mathcal{L}^{\otimes 1/2} \to \mathcal{U}^{0} \otimes_{\mathcal{O}_{X}} \mathcal{L} \quad \text{and} \quad \mathcal{U}^{1} \otimes \mathcal{L}^{\otimes 1/2} \to \mathcal{U}^{0} \otimes_{\mathcal{O}_{X}} \mathcal{L} \to \mathcal{U}^{1} \otimes_{\mathcal{O}_{X}} \mathcal{L}^{\otimes 3/2}$$

are equal to the multiplications with w.

2.3. *Exotic derived categories of matrix factorizations.* The following corollary is a restatement of the results of Section 1 in the application to the quasicoherent CDG-algebra $\mathcal{B} = (X, \mathcal{L}, w)$. We will use the notation (X, \mathcal{L}, w) -coh_{lf} (instead of the previously introduced \mathcal{B} -coh_{fl}) for the DG-category of locally free matrix factorizations of finite rank, and the notation (X, \mathcal{L}, w) -qcoh_{lf} (instead of the previously introduced \mathcal{B} -qcoh_{lp}) for the DG-category of locally free matrix factorizations of possibly infinite rank (see Remark 1.4). The rest of our notation system for various classes of quasicoherent CDG-modules over $\mathcal{B} = (X, \mathcal{L}, w)$ remains in use.

In addition, we also denote by (X, \mathcal{L}, w) -qcoh_{lfd} the DG-category of quasicoherent CDG-modules of finite locally free/locally projective dimension over (X, \mathcal{L}, w) (see Remark 1.4 again). Let $D^{co}((X, \mathcal{L}, w)$ -qcoh_{lfd}) and $D^{abs}((X, \mathcal{L}, w)$ -qcoh_{lfd}) be the corresponding derived categories of the second kind.

Corollary 2.3. (a) The functor $D^{co}((X, \mathcal{L}, w) \operatorname{-qcoh}_{fl}) \to D^{co}((X, \mathcal{L}, w) \operatorname{-qcoh}_{ffd})$ induced by the embedding of DG-categories $(X, \mathcal{L}, w) \operatorname{-qcoh}_{fl} \to (X, \mathcal{L}, w) \operatorname{-qcoh}_{ffd}$ is an equivalence of triangulated categories.

(b) The functor $D^{abs}((X, \mathcal{L}, w) \operatorname{-qcoh}_{fl}) \to D^{abs}((X, \mathcal{L}, w) \operatorname{-qcoh}_{ffd})$ induced by the embedding of DG-categories $(X, \mathcal{L}, w) \operatorname{-qcoh}_{fl} \to (X, \mathcal{L}, w) \operatorname{-qcoh}_{ffd}$ is an equivalence of triangulated categories.

(c) The functors

$$\begin{split} \mathsf{D}^{\mathsf{co}}((X,\mathcal{L},w)\operatorname{-qcoh}_{\mathsf{lf}}) &\longrightarrow \ \mathsf{D}^{\mathsf{co}}((X,\mathcal{L},w)\operatorname{-qcoh}_{\mathsf{lfd}}), \\ \mathsf{D}^{\mathsf{abs}}((X,\mathcal{L},w)\operatorname{-qcoh}_{\mathsf{lf}}) &\longrightarrow \ \mathsf{D}^{\mathsf{abs}}((X,\mathcal{L},w)\operatorname{-qcoh}_{\mathsf{lfd}}) \end{split}$$

induced by the embedding of DG-categories (X, \mathcal{L}, w) -qcoh_{lf} $\rightarrow (X, \mathcal{L}, w)$ -qcoh_{lf} are equivalences of triangulated categories.

(d) The triangulated categories $D^{co}((X, \mathcal{L}, w) \operatorname{-qcoh}_{lf})$ and $D^{abs}((X, \mathcal{L}, w) \operatorname{-qcoh}_{lf})$ coincide, as do the categories $D^{co}((X, \mathcal{L}, w) \operatorname{-qcoh}_{lfd})$ and $D^{abs}((X, \mathcal{L}, w) \operatorname{-qcoh}_{lfd})$. The natural functors between these four categories form a commutative square of equivalences of triangulated categories.

(e) When the scheme X has finite Krull dimension, the functors

$$D^{co}((X, \mathcal{L}, w) \operatorname{-qcoh}_{\mathsf{lf}}) \longrightarrow D^{co}((X, \mathcal{L}, w) \operatorname{-qcoh}_{\mathsf{fl}}),$$
$$D^{abs}((X, \mathcal{L}, w) \operatorname{-qcoh}_{\mathsf{lf}}) \longrightarrow D^{abs}((X, \mathcal{L}, w) \operatorname{-qcoh}_{\mathsf{fl}})$$

induced by the embedding of DG-categories (X, \mathcal{L}, w) -qcoh_{lf} $\rightarrow (X, \mathcal{L}, w)$ -qcoh_{fl} are equivalences of triangulated categories. The natural functors between these four categories form a commutative square of equivalences.

(f) When the scheme X has finite Krull dimension, the triangulated category $D^{co}((X, \mathcal{L}, w)-qcoh_{fl})$ coincides with $D^{abs}((X, \mathcal{L}, w)-qcoh_{fl})$ and the triangulated

category $D^{co}((X, \mathcal{L}, w)-qcoh_{ffd})$ coincides with $D^{abs}((X, \mathcal{L}, w)-qcoh_{ffd})$. The natural functors between these four categories form a commutative square of equivalences.

(g) The functor $D^{abs}((X, \mathcal{L}, w) \operatorname{coh}_{\mathsf{lf}}) \to D^{abs}((X, \mathcal{L}, w) \operatorname{coh}_{\mathsf{ffd}})$ induced by the embedding of DG-categories $(X, \mathcal{L}, w) \operatorname{coh}_{\mathsf{lf}} \to (X, \mathcal{L}, w) \operatorname{coh}_{\mathsf{ffd}}$ is an equivalence of triangulated categories.

(h) The triangulated functors

 $\mathsf{D}^{\mathsf{abs}}((X,\mathcal{L},w)\operatorname{-qcoh}_{\mathsf{lf}}) \to \mathsf{D}^{\mathsf{abs}}((X,\mathcal{L},w)\operatorname{-qcoh}_{\mathsf{fl}}) \to \mathsf{D}^{\mathsf{abs}}((X,\mathcal{L},w)\operatorname{-qcoh})$

induced by the embeddings of DG-categories (X, \mathcal{L}, w) -qcoh_{lf} \rightarrow (X, \mathcal{L}, w) -qcoh_{fl} \rightarrow (X, \mathcal{L}, w) -qcoh are fully faithful.

(i) The triangulated functor D^{abs}((X, L, w)-coh_{lf}) → D^{abs}((X, L, w)-coh) induced by the embedding of DG-categories (X, L, w)-coh_{lf} → (X, L, w)-coh is fully faithful.
(j) The triangulated functor D^{abs}((X, L, w)-coh_{lf}) → D^{co}((X, L, w)-qcoh_{lf}) induced by the embedding of DG-categories (X, L, w)-coh_{lf} → (X, L, w)-qcoh_{lf} is fully faithful.

(k) The triangulated functor $D^{abs}((X, \mathcal{L}, w) \operatorname{-coh}) \to D^{abs}((X, \mathcal{L}, w) \operatorname{-qcoh})$ induced by the embedding of DG-categories $(X, \mathcal{L}, w) \operatorname{-coh} \to (X, \mathcal{L}, w) \operatorname{-qcoh}$ is fully faithful.

(1) The triangulated functor $D^{abs}((X, \mathcal{L}, w)\text{-coh}) \to D^{co}((X, \mathcal{L}, w)\text{-qcoh})$ induced by the embedding of DG-categories $(X, \mathcal{L}, w)\text{-coh} \to (X, \mathcal{L}, w)\text{-qcoh}$ is fully faithful and its image forms a set of compact generators for $D^{co}((X, \mathcal{L}, w)\text{-qcoh})$.

Proof. Parts (a), (b) and (g) are particular cases of Theorem 1.4, and the proof of part (c) is similar (see Remark 1.4). Part (g) also essentially follows from Proposition 1.5(b) (and part (b) can be proven similarly). Parts (h), (i), (k) and (l) are particular cases of Proposition 1.5 (except for "locally free half" of part (h), which is similar to the "flat half"). Part (d) is Theorem 1.6 together with part (c). Part (j) is Corollary 1.6. Part (e) follows from parts (a)–(c) and Remark 1.4 (cf. the discussion in the end of Section 1.6). Part (f) follows from parts (a), (b), (d) and (e); alternatively, it can be proven directly in the way similar to part (d), using the fact that the exact category of flat quasicoherent sheaves on *X* has finite homological dimension when the Krull dimension of *X* is finite.

2.4. *Regular and Gorenstein scheme cases.* When the scheme *X* is regular or Gorenstein, the assertions of Corollary 2.3 simplify as follows.

Corollary 2.4. (a) When the scheme X is Gorenstein of finite Krull dimension, the functors

 $\mathsf{D}^{\mathsf{abs}}((X,\mathcal{L},w)\operatorname{-qcoh}_{\mathsf{fl}})\longrightarrow \mathsf{D}^{\mathsf{co}}((X,\mathcal{L},w)\operatorname{-qcoh}_{\mathsf{fl}})\longrightarrow \mathsf{D}^{\mathsf{co}}((X,\mathcal{L},w)\operatorname{-qcoh})$

induced by the embedding of DG-categories (X, \mathcal{L}, w) -qcoh_{fl} $\rightarrow (X, \mathcal{L}, w)$ -qcoh are equivalences of triangulated categories.

- (b) When the scheme X is regular of finite Krull dimension, the natural functors between the categories D^{abs}((X, L, w)-qcoh_{fl}), D^{co}((X, L, w)-qcoh_{fl}), D^{abs}((X, L, w)-qcoh), and D^{co}((X, L, w)-qcoh) form a commutative square of equivalences of triangulated categories.
- (c) When the scheme X is regular, the natural functor $D^{abs}((X, \mathcal{L}, w)\text{-coh}_{lf}) \rightarrow D^{abs}((X, \mathcal{L}, w)\text{-coh})$ is an equivalence of triangulated categories.

Proof. Part (a) is a particular case of Proposition 1.7. Part (c) follows from Corollary 2.3(g) since any coherent sheaf on a regular scheme has finite flat dimension. In the assumptions of part (b), the functor $D^{abs}((X, \mathcal{L}, w)$ -qcoh) $\rightarrow D^{co}((X, \mathcal{L}, w)$ -qcoh) is an isomorphism of triangulated categories by [Positselski 2011b, Theorem 3.6(a) and Remark 3.6] since the abelian category of quasicoherent sheaves on a regular scheme of finite Krull dimension has finite homological dimension and enough injectives (cf. Theorem 1.6). The remaining assertions of part (b) follow from Corollary 2.3(a) and (b), or alternatively from part (a).

Assuming that X has finite Krull dimension, the assertions of Corollaries 2.3 and 2.4 may be summarized by the following commutative diagram of triangulated functors. Here, as above, \mathcal{B} denotes the quasicoherent CDG-algebra (X, \mathcal{L}, w) :



The four categories in the left lower area are coderived categories coinciding with absolute derived categories (of the same classes of quasicoherent CDG-modules). The five double lines between these four categories are equivalences, as is the upper left horizontal line. All the arrows going down are fully faithful functors. The image of the rightmost vertical arrow is a set of compact generators in the target category. The only arrow going up is a Verdier localization functor.

Nothing is claimed about the long horizontal arrow in the right lower area of the diagram in general; but when X is Gorenstein, this functor is an equivalence of categories. When X is regular, all the arrows going right are equivalences of categories (so the whole diagram reduces to one triangulated category with infinite direct sums, containing a full triangulated subcategory of compact generators).

Recall also that, by Lemma 1.7, for any X we have a commutative diagram of triangulated functors

$$H^{0}(\mathcal{B}\operatorname{-qcoh}_{\operatorname{inj}}) = \mathbb{D}^{\operatorname{co}=\operatorname{abs}}(\mathcal{B}\operatorname{-qcoh}_{\operatorname{fid}}) \xrightarrow{\qquad} \mathbb{D}^{\operatorname{co}}(\mathcal{B}\operatorname{-qcoh})$$

with equivalences of categories in the upper line. The fully faithful embedding $D^{abs}(\mathcal{B}\text{-qcoh}_{fid}) \rightarrow D^{abs}(\mathcal{B}\text{-qcoh})$, which in the Gorenstein case (of finite Krull dimension) coincides with the embedding $D^{abs}(\mathcal{B}\text{-qcoh}_{ffd}) \rightarrow D^{abs}(\mathcal{B}\text{-qcoh})$, is always right adjoint to the localization functor $D^{abs}(\mathcal{B}\text{-qcoh}) \rightarrow D^{co}(\mathcal{B}\text{-qcoh})$.

Remark 2.4. When X is an affine Noetherian scheme of finite Krull dimension, the embeddings of DG-categories (X, \mathcal{L}, w) -qcoh_{lp} $\rightarrow (X, \mathcal{L}, w)$ -qcoh_{fl} $\rightarrow (X, \mathcal{L}, w)$ -qcoh induce equivalences $H^0(\mathcal{B}$ -qcoh_{lp}) $\simeq D^{abs}(\mathcal{B}$ -qcoh_{fl}) $\simeq D^{ctr}(\mathcal{B}$ -qcoh) between the homotopy category of (locally) projective matrix factorizations of infinite rank, the absolute derived category of flat matrix factorizations, and the contraderived category of arbitrary quasicoherent matrix factorizations (see [Positselski 2011b, Section 3.8]; cf. Remark 1.5).

2.5. Serre–Grothendieck duality. The aim of this section is to show that the somewhat mysterious long horizontal arrow in the above large diagram is actually a functor between two equivalent triangulated categories, for a rather wide class of schemes X. The functor $D^{co}((X, \mathcal{L}, w)-qcoh_{fl}) \rightarrow D^{co}((X, \mathcal{L}, w)-qcoh)$ in the above diagram, which is induced by the embedding of DG-categories (X, \mathcal{L}, w) -qcoh_{fl} $\rightarrow (X, \mathcal{L}, w)$ -qcoh, is *not* the equivalence that we have in mind, however (unless the scheme is Gorenstein). Instead, the equivalence between the categories $D^{co}((X, \mathcal{L}, w)$ -qcoh_{fl}) and $D^{co}((X, \mathcal{L}, w)$ -qcoh) is constructed using a dualizing complex on X [Hartshorne 1966, Section V.2].

Before recalling the definition of a dualizing complex, let us discuss the notion of the *quasicoherent internal Hom*. Given quasicoherent sheaves \mathcal{M} and \mathcal{N} over X, the quasicoherent sheaf $\mathcal{H}om_{X-qc}(\mathcal{M}, \mathcal{N})$ is defined by the isomorphism $\operatorname{Hom}_{\mathcal{O}_X}(-\otimes_{\mathcal{O}_X} \mathcal{M}, \mathcal{N}) \simeq \operatorname{Hom}_{\mathcal{O}_X}(-, \mathcal{H}om_{X-qc}(\mathcal{M}, \mathcal{N}))$ of functors from the category of quasicoherent sheaves to the category of abelian groups. Equivalently, the quasicoherent sheaf $\mathcal{H}om_{X-qc}(\mathcal{M}, \mathcal{N})$ can be obtained by applying the coherator functor [Thomason and Trobaugh 1990, Sections B.12–B.14] to the sheaf of \mathcal{O}_X -modules $\mathcal{H}om_{\mathcal{O}_X}(\mathcal{M}, \mathcal{N})$. Whenever \mathcal{M} is a coherent sheaf, the sheaf $\mathcal{H}om_{\mathcal{O}_X}(\mathcal{M}, \mathcal{N})$ of \mathcal{O}_X -module internal Hom is quasicoherent, and $\mathcal{H}om_{X-qc}(\mathcal{M}, \mathcal{N}) \simeq \mathcal{H}om_{\mathcal{O}_X}(\mathcal{M}, \mathcal{N})$.

Notice that the construction of the sheaf $\mathcal{H}om_{X-qc}(\mathcal{M}, \mathcal{N})$ is *not* local in general; i.e., it does not commute with the restrictions of quasicoherent sheaves to open subschemes; when the sheaf \mathcal{M} is coherent, it does.

- **Lemma 2.5.** (a) For any injective quasicoherent sheaf \mathcal{J} over a separated Noetherian scheme X, the functor $\mathcal{M} \mapsto \mathcal{H}om_{X-qc}(\mathcal{M}, \mathcal{J})$ is exact.
- (b) For any flat quasicoherent sheaf F and injective quasicoherent sheaf J over X, the quasicoherent sheaves F ⊗_{OX} J and Hom_{X-qc}(F, J) are injective.
- (c) For any injective quasicoherent sheaves \$\mathcal{J}'\$ and \$\mathcal{J}\$ over \$X\$, the quasicoherent sheaf \$\mathcal{H}\$ on \$X\$-qc(\$\mathcal{J}', \$\mathcal{J}\$) is flat.

Proof. The second assertion of part (b) is obvious from the universal property defining $\mathcal{H}om_{X-qc}$. To prove the first one, notice that injectivity of quasicoherent sheaves over a Noetherian scheme is a local property ([Hartshorne 1966, Lemma II.7.16 and Theorem II.7.18] or Theorem A.3), a flat quasicoherent sheaf over an affine scheme is a filtered inductive limit of locally free sheaves of finite rank [Bourbaki 1980, Numéros 1.5–6], and injectivity of modules over a Noetherian ring is preserved by filtered inductive limits.

The proof of parts (a) and (c) follows the argument in [Murfet 2007, Lemma 8.7]. Choose a finite affine covering U_{α} of the scheme X and consider the morphism $\mathcal{J} \to \bigoplus_{\alpha} j_{U_{\alpha}*} j_{U_{\alpha}}^* \mathcal{J}$. Being an embedding of injective quasicoherent sheaves, it splits, so \mathcal{J} is a direct summand of the direct sum of $j_{U_{\alpha}*} j_{U_{\alpha}}^* \mathcal{J}$. Hence it suffices to prove both assertions in the case when $\mathcal{J} = j_{V*}\mathcal{J}''$, where \mathcal{J}'' is an injective quasicoherent sheaf on an affine open subscheme $V \subset X$.

Now we have $\mathcal{H}om_{X-qc}(\mathcal{M}, j_{V*}\mathcal{J}'') \simeq j_{V*}\mathcal{H}om_{V-qc}(j_V^*\mathcal{M}, \mathcal{J}'')$. Since $V \to X$ is a flat affine morphism, the functor j_{V*} is exact and preserves the flatness of quasicoherent sheaves. This proves part (a) and reduces (c) to the case of an affine scheme X=V. It remains to apply [Cartan and Eilenberg 1956, Proposition VI.5.3]. \Box

For our purposes, a *dualizing complex* \mathcal{D}_X^{\bullet} on X is a finite complex of injective quasicoherent sheaves such that the cohomology sheaves of \mathcal{D}_X^{\bullet} are coherent and for any coherent sheaf \mathcal{M} over X, the natural morphism of finite complexes of quasicoherent sheaves $\mathcal{M} \to \mathcal{H}om_{X-qc}(\mathcal{H}om_{X-qc}(\mathcal{M}, \mathcal{D}_X^{\bullet}), \mathcal{D}_X^{\bullet})$ is a quasi-isomorphism. Note that it follows from the former two conditions on \mathcal{D}_X^{\bullet} that the complex $\mathcal{H}om_{X-qc}(\mathcal{M}, \mathcal{D}_X^{\bullet})$, bas coherent cohomology sheaves. This makes the conditions imposed on \mathcal{D}_X^{\bullet} actually local in X, so the restriction $\mathcal{D}_U^{\bullet} = \mathcal{D}_X^{\bullet}|_U$ of the complex of sheaves \mathcal{D}_X^{\bullet} to an open subscheme $U \subset X$ is a dualizing complex on U.

Given a quasicoherent CDG-algebra \mathcal{B} over X, a quasicoherent left CDG-module \mathcal{M} over \mathcal{B} , and a complex of quasicoherent sheaves \mathcal{F}^{\bullet} on X, consider the complexes of quasicoherent left CDG-modules $\mathcal{F}^{\bullet} \otimes_{\mathcal{O}_X} \mathcal{M}$ and $\mathcal{H}om_{X-qc}(\mathcal{F}^{\bullet}, \mathcal{M})$ over \mathcal{B} . Taking their totalizations (formed, if necessary, by taking infinite direct sums along the diagonals), construct two triangulated functors $H^0(\mathcal{B}\text{-qcoh}) \to H^0(\mathcal{B}\text{-qcoh})$ depending on a complex \mathcal{F}^{\bullet} . Given a right CDG-module \mathcal{N} over \mathcal{B} (see [Positselski 2011b, Sections 3.1 and B.1]), similarly construct a complex of quasicoherent left

CDG-modules $\mathcal{H}om_{X-qc}(\mathcal{N}, \mathcal{F}^{\bullet})$ over \mathcal{B} , obtaining a triangulated functor from the homotopy category of right CDG-modules $H^0(qcoh-\mathcal{B})$ to $H^0(\mathcal{B}-qcoh)$.

In the particular case of matrix factorizations, we conclude that the covariant functors $\mathcal{F}^{\bullet} \otimes_{\mathcal{O}_X} -$ and $\mathcal{H}om_{X\text{-qc}}(\mathcal{F}^{\bullet}, -)$ take quasicoherent matrix factorizations of a potential $w \in \mathcal{L}(X)$ to (complexes of) quasicoherent matrix factorizations of w, while the contravariant functor $\mathcal{H}om_{X\text{-qc}}(-, \mathcal{F}^{\bullet})$ transforms quasicoherent matrix factorizations of the opposite potential $-w \in \mathcal{L}(X)$ into (complexes of) quasicoherent matrix factorizations of w. Of course, the quasicoherent CDG-algebras (X, \mathcal{L}, w) and $(X, \mathcal{L}, -w)$ over a scheme X are naturally isomorphic, but we prefer to keep the distinction between the two.

The next proposition provides the matrix factorization version of the conventional (contravariant) Serre–Grothendieck duality for bounded complexes of coherent sheaves. We assume that X is a separated Noetherian scheme with a dualizing complex \mathcal{D}_X^{\bullet} . Recall that any such scheme has finite Krull dimension [Hartshorne 1966, Corollary V.7.2]. We denote by D^{op} the opposite category to a category D.

Proposition 2.5. The triangulated functor

$$\mathcal{H}om_{X-qc}(-, \mathcal{D}_X^{\bullet}) : H^0((X, \mathcal{L}, -w)-qcoh)^{op} \longrightarrow H^0((X, \mathcal{L}, w)-qcoh)$$

induces a well-defined triangulated functor between the absolute derived categories $D^{abs}((X, \mathcal{L}, -w)\operatorname{-qcoh})^{op}$ and $D^{abs}((X, \mathcal{L}, w)\operatorname{-qcoh})$ taking the full triangulated subcategory $D^{abs}((X, \mathcal{L}, -w)\operatorname{-coh})^{op} \subset D^{abs}((X, \mathcal{L}, -w)\operatorname{-qcoh})^{op}$ into the full subcategory $D^{abs}((X, \mathcal{L}, w)\operatorname{-coh}) \subset D^{abs}((X, \mathcal{L}, w)\operatorname{-qcoh})$. The composition of the duality functors $D^{abs}((X, \mathcal{L}, w)\operatorname{-coh}) \to D^{abs}((X, \mathcal{L}, -w)\operatorname{-coh})^{op} \to D^{abs}((X, \mathcal{L}, w)\operatorname{-coh})$ is the identity functor.

Proof. The functor $\mathcal{H}om_{X-qc}(-, \mathcal{D}_X^{\bullet})$ preserves absolute acyclicity, because \mathcal{D}_X^{\bullet} is a complex of injective quasicoherent sheaves, so Lemma 2.5(a) applies. Given a coherent matrix factorization \mathcal{M} , the finite complex of matrix factorizations $\mathcal{H}om_{X-qc}(-, \mathcal{D}_X^{\bullet})$ has coherent cohomology matrix factorizations, so one can use its canonical truncations in order to prove by induction that its totalization belongs to the triangulated subcategory $D^{abs}((X, \mathcal{L}, w)$ -coh).

Finally, for any quasicoherent matrix factorization \mathcal{M} , consider the bicomplex of matrix factorizations $\mathcal{H}om_{X-qc}(\mathcal{H}om_{X-qc}(\mathcal{M}, \mathcal{D}_X^{\bullet}), \mathcal{D}_X^{\bullet})$ and take its totalization in the two directions where it is a complex, obtaining a complex of matrix factorizations. Then there is a natural morphism of finite complexes of matrix factorizations $\mathcal{M} \to \mathcal{H}om_{X-qc}(\mathcal{H}om_{X-qc}(\mathcal{M}, \mathcal{D}_X^{\bullet}), \mathcal{D}_X^{\bullet}))$, which is a quasi-isomorphism of complexes of matrix factorizations when \mathcal{M} is coherent. The induced closed morphism of the total matrix factorizations is an isomorphism in $D^{abs}((X, \mathcal{L}, w)$ -qcoh) since the totalization of a finite acyclic complex of matrix factorizations is absolutely acyclic. It

remains to use the fact that the functor $D^{abs}((X, \mathcal{L}, w)\text{-coh}) \rightarrow D^{abs}((X, \mathcal{L}, w)\text{-qcoh})$ is fully faithful (see Corollary 2.3(k)) again.

The next result is our covariant Serre–Grothendieck duality theorem for matrix factorizations. It is the matrix factorization analogue of the similar results for complexes of projective and injective modules [Iyengar and Krause 2006, Theorem 4.2] and sheaves [Murfet 2007, Theorem 8.4]. It also strongly resembles the *derived comodule-contramodule correspondence* theory (see [Positselski 2010, Corollaries 5.4 and 6.3; 2011b, Theorem 5.2]; cf. Remark 2.4 above). Notice that our proof is more akin to the arguments in [Positselski 2010; 2011b] than those of [Iyengar and Krause 2006; Murfet 2007] in that we give a direct proof of the covariant duality independent of both the contravariant duality and any descriptions of the compact objects in the categories to be compared.

Theorem 2.5. The functors

$$\mathcal{D}_{X}^{\bullet} \otimes_{\mathcal{O}_{X}} - : H^{0}((X, \mathcal{L}, w) \operatorname{-qcoh}_{\mathsf{fl}}) \longrightarrow H^{0}((X, \mathcal{L}, w) \operatorname{-qcoh}_{\mathsf{inj}}),$$

$$\mathcal{H}om_{X-\mathsf{qc}}(\mathcal{D}_{X}^{\bullet}, -) : H^{0}((X, \mathcal{L}, w) \operatorname{-qcoh}_{\mathsf{inj}}) \longrightarrow H^{0}((X, \mathcal{L}, w) \operatorname{-qcoh}_{\mathsf{fl}})$$

induce mutually inverse equivalences between the coderived categories

$$D^{co}((X, \mathcal{L}, w) \operatorname{-qcoh}_{fl})$$
 and $D^{co}((X, \mathcal{L}, w) \operatorname{-qcoh})$

Proof. Recall that $H^0((X, \mathcal{L}, w)\operatorname{-qcoh}_{inj}) \simeq \mathsf{D^{co}}((X, \mathcal{L}, w)\operatorname{-qcoh})$ by Lemma 1.7(b) and $\mathsf{D^{abs}}((X, \mathcal{L}, w)\operatorname{-qcoh}_{fl}) = \mathsf{D^{co}}((X, \mathcal{L}, w)\operatorname{-qcoh}_{fl})$ by Corollary 2.3(f) (though we will reprove the latter fact rather than use it in the following argument; see also Remark 2.6 below and Lemma A.1). The functor

$$\mathcal{D}_X^{\bullet} \otimes_{\mathcal{O}_X} - : H^0((X, \mathcal{L}, w) \operatorname{-qcoh}_{\mathsf{fl}}) \longrightarrow H^0((X, \mathcal{L}, w) \operatorname{-qcoh}_{\mathsf{ini}})$$

obviously takes matrix factorizations coacyclic with respect to (X, \mathcal{L}, w) -qcoh_{fl} to matrix factorizations coacyclic with respect to (X, \mathcal{L}, w) -qcoh_{inj}, which are all contractible. It remains to check that the induced functors are mutually inverse.

Let \mathcal{E} be a matrix factorization from (X, \mathcal{L}, w) -qcoh_{fl}. As in the previous proof, consider the bicomplex of matrix factorizations $\mathcal{H}om_{X-qc}(\mathcal{D}_X^{\bullet}, \mathcal{D}_X^{\bullet} \otimes_{\mathcal{O}_X} \mathcal{E})$ and take its total complex of matrix factorizations. Then there is a natural morphism $\mathcal{E} \to \mathcal{H}om_{X-qc}(\mathcal{D}_X^{\bullet}, \mathcal{D}_X^{\bullet} \otimes_{\mathcal{O}_X} \mathcal{E})$ of finite complexes of matrix factorizations from (X, \mathcal{L}, w) -qcoh_{fl}. To prove that the induced morphism of the total matrix factorizations is an isomorphism in $\mathsf{D}^{co}((X, \mathcal{L}, w)$ -qcoh_{fl}), we once again use the fact that the totalization of a finite acyclic complex of matrix factorizations is absolutely acyclic. So it suffices to check that for any flat quasicoherent sheaf \mathcal{F} over X, the natural morphism $\mathcal{F} \to \mathcal{H}om_{X-qc}(\mathcal{D}_X^{\bullet}, \mathcal{D}_X^{\bullet} \otimes_{\mathcal{O}_X} \mathcal{F})$ is a quasi-isomorphism of complexes of flat quasicoherent sheaves. This will be done below. Similarly, let \mathcal{M} be a matrix factorization from (X, \mathcal{L}, w) -qcoh_{inj}. Consider the morphism of finite complexes of injective matrix factorizations given by $\mathcal{D}_X^{\bullet} \otimes_{\mathcal{O}_X} \mathcal{H}om_{X-qc}(\mathcal{D}_X^{\bullet}, \mathcal{M}) \to \mathcal{M}$. To prove that the cone of the induced morphism of the total matrix factorizations is contractible, it suffices to check that for any injective quasicoherent sheaf \mathcal{J} over X, the natural morphism of complexes of injective sheaves $\mathcal{D}_X^{\bullet} \otimes_{\mathcal{O}_X} \mathcal{H}om_{X-qc}(\mathcal{D}_X^{\bullet}, \mathcal{J}) \to \mathcal{J}$ is a quasi-isomorphism.

Let ${}^{\prime}\mathcal{D}_{X}^{\bullet}$ denote a finite complex of coherent sheaves over X endowed with a quasi-isomorphism ${}^{\prime}\mathcal{D}_{X}^{\bullet} \to \mathcal{D}_{X}^{\bullet}$. Then the morphism $\mathcal{H}om_{X-qc}(\mathcal{D}_{X}^{\bullet}, \mathcal{D}_{X}^{\bullet} \otimes_{\mathcal{O}_{X}} \mathcal{F}) \to \mathcal{H}om_{X-qc}({}^{\prime}\mathcal{D}_{X}^{\bullet}, \mathcal{D}_{X}^{\bullet} \otimes_{\mathcal{O}_{X}} \mathcal{F})$ is a quasi-isomorphism for any flat quasicoherent sheaf \mathcal{F} . The construction of the composition

$$\mathcal{F} \longrightarrow \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{D}_X^{\bullet}, \mathcal{D}_X^{\bullet} \otimes_{\mathcal{O}_X} \mathcal{F}) \longrightarrow \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{D}_X^{\bullet}, \mathcal{D}_X^{\bullet} \otimes_{\mathcal{O}_X} \mathcal{F})$$

is local in X, so it suffices to check that the composition is a quasi-isomorphism when X is affine. Then, using the passage to the filtered inductive limit, we may assume that \mathcal{F} is locally free of finite rank, and further that $\mathcal{F} = \mathcal{O}_X$. It remains to recall that the morphism $\mathcal{O}_X \to \mathcal{H}om_{X-qc}(\mathcal{D}_X^{\bullet}, \mathcal{D}_X^{\bullet})$ is a quasi-isomorphism by the definition of \mathcal{D}_X^{\bullet} .

Let ${}^{\prime\prime}\mathcal{D}_X^{\bullet}$ be a bounded-above complex of flat quasicoherent sheaves mapping quasi-isomorphically to ${}^{\prime}\mathcal{D}_X^{\bullet}$. Then for any injective quasicoherent sheaf \mathcal{J} over X there are quasi-isomorphisms

$${}^{"}\mathcal{D}^{\bullet}_{X} \otimes_{\mathcal{O}_{X}} \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{D}^{\bullet}_{X},\mathcal{J}) \longrightarrow \mathcal{D}^{\bullet}_{X} \otimes_{\mathcal{O}_{X}} \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{D}^{\bullet}_{X},\mathcal{J}),$$

$${}^{"}\mathcal{D}^{\bullet}_{X} \otimes_{\mathcal{O}_{X}} \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{D}^{\bullet}_{X},\mathcal{J}) \longrightarrow {}^{"}\mathcal{D}^{\bullet}_{X} \otimes_{\mathcal{O}_{X}} \mathcal{H}om_{X-\mathsf{qc}}({}^{'}\mathcal{D}^{\bullet}_{X},\mathcal{J})$$

forming a commutative diagram with the evaluation morphisms into \mathcal{J} . Hence it remains to check that the morphism ${}^{\prime\prime}\mathcal{D}_X^{\bullet} \otimes_{\mathcal{O}_X} \mathcal{H}om_{X-qc}({}^{\prime}\mathcal{D}_X^{\bullet}, \mathcal{J}) \to \mathcal{J}$ is a quasi-isomorphism, which is a local question. Assume further that ${}^{\prime\prime}\mathcal{D}_X^{\bullet}$ is a bounded-above complex of locally free sheaves of finite rank. Then there is a natural isomorphism of complexes of sheaves

$${}^{''}\mathcal{D}_X^{\bullet} \otimes_{\mathcal{O}_X} \mathcal{H}om_{X-\mathsf{qc}}({}^{'}\mathcal{D}_X^{\bullet},\mathcal{J}) \simeq \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{H}om_{X-\mathsf{qc}}({}^{''}\mathcal{D}_X^{\bullet},{}^{'}\mathcal{D}_X^{\bullet}),\mathcal{J}).$$

The related morphism

$$\mathcal{H}om_{X-\mathsf{qc}}(\mathcal{H}om_{X-\mathsf{qc}}(''\mathcal{D}_X^{\bullet},'\mathcal{D}_X^{\bullet}),\mathcal{J})\longrightarrow \mathcal{J}$$

is induced by the natural morphism of complexes $\mathcal{O}_X \to \mathcal{H}om_{X-qc}(\mathcal{D}_X^{\bullet}, \mathcal{D}_X^{\bullet})$. The latter is again a quasi-isomorphism essentially by the definition of \mathcal{D}_X^{\bullet} .

From this point on we resume assuming that X has enough vector bundles. Notice that the equivalence functor

$$\mathcal{D}_X^{\bullet} \otimes_{\mathcal{O}_X} - : \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w) \operatorname{-qcoh}_{\mathsf{lf}}) \longrightarrow \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w) \operatorname{-qcoh})$$

that we constructed takes the full triangulated subcategory $D^{abs}((X, \mathcal{L}, w)\text{-coh}_{lf}) \subset D^{co}((X, \mathcal{L}, w)\text{-qcoh}_{lf})$ into the full triangulated subcategory $D^{abs}((X, \mathcal{L}, w)\text{-coh}) \subset D^{co}((X, \mathcal{L}, w)\text{-qcoh})$. This is so because the dualizing complex \mathcal{D}_X^{\bullet} has bounded coherent cohomology sheaves.

Now we will use Proposition 2.5 and Theorem 2.5 in order to construct compact generators of the triangulated category $D^{co}((X, \mathcal{L}, w)-qcoh_{lf})$ (cf. [Jørgensen 2005; Neeman 2008]).

Consider the abelian category $Z^0((X, \mathcal{L}, -w)$ -coh) of coherent matrix factorizations of -w and closed morphisms of degree 0 between them, and its exact subcategory of locally free matrix factorizations of finite rank $Z^0((X, \mathcal{L}, -w)$ -coh_{lf}). The natural functor between the bounded-above derived categories of our abelian category and its exact subcategory

$$D^{-}(Z^{0}((X, \mathcal{L}, -w)\operatorname{-coh}_{\mathsf{lf}})) \longrightarrow D^{-}(Z^{0}((X, \mathcal{L}, -w)\operatorname{-coh}))$$

is an equivalence of triangulated categories. The vector bundle duality functor $\mathcal{H}om_{X-qc}(-, \mathcal{O}_X) : Z^0((X, \mathcal{L}, -w) \operatorname{-coh}_{\mathsf{lf}})^{\mathsf{op}} \to Z^0((X, \mathcal{L}, w) \operatorname{-coh}_{\mathsf{lf}})$ induces a triangulated functor $\mathsf{D}^-(Z^0((X, \mathcal{L}, -w) \operatorname{-coh}_{\mathsf{lf}}))^{\mathsf{op}} \to \mathsf{D}^+(Z^0((X, \mathcal{L}, w) \operatorname{-coh}_{\mathsf{lf}}))$ taking bounded-above complexes to bounded-below ones.

Let $D^+(Z^0((X, \mathcal{L}, w)-\operatorname{qcoh}_{\mathsf{lf}}))$ denote the bounded-below derived category of the exact category of locally free matrix factorizations of possibly infinite rank. Since the bounded-below acyclic complexes over any exact category with infinite direct sums are coacyclic [Positselski 2010, Lemma 2.1], there is a well-defined, triangulated direct sum totalization functor $D^+(Z^0((X, \mathcal{L}, w)\operatorname{-qcoh}_{\mathsf{lf}})) \to D^{co}((X, \mathcal{L}, w)\operatorname{-qcoh}_{\mathsf{lf}})$. Consider the composition

$$Z^{0}((X, \mathcal{L}, -w)\operatorname{-coh})^{\operatorname{op}} \longrightarrow D^{-}(Z^{0}((X, \mathcal{L}, -w)\operatorname{-coh}))^{\operatorname{op}}$$

$$\simeq D^{-}(Z^{0}((X, \mathcal{L}, -w)\operatorname{-coh}_{\mathsf{lf}}))^{\operatorname{op}} \longrightarrow D^{+}(Z^{0}((X, \mathcal{L}, w)\operatorname{-coh}_{\mathsf{lf}}))$$

$$\longrightarrow D^{+}(Z^{0}((X, \mathcal{L}, w)\operatorname{-qcoh}_{\mathsf{lf}})) \longrightarrow D^{\operatorname{co}}((X, \mathcal{L}, w)\operatorname{-qcoh}_{\mathsf{lf}}),$$

where two of the functors are the duality and the totalization discussed above, while the other two are the natural embedding and the functor induced by such.

One easily checks that this composition takes cones of closed morphisms in $Z^0((X, \mathcal{L}, -w)$ -coh) to cocones in $D^{co}((X, \mathcal{L}, w)$ -qcoh_{lf}); hence it induces a triangulated functor $H^0((X, \mathcal{L}, -w)$ -coh)^{op} \rightarrow D^{co}((X, \mathcal{L}, w)-qcoh_{lf}). Similarly, the above composition takes the totalizations of short exact sequences in $(X, \mathcal{L}, -w)$ -coh to objects corresponding to the totalizations of short exact sequences in (X, \mathcal{L}, w) -qcoh_{lf}; one checks this by considering a left locally free resolution of a short exact sequence of coherent matrix factorizations. Thus we obtain a triangulated functor

$$\Omega: \mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, -w)\operatorname{-coh})^{\mathsf{op}} \longrightarrow \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w)\operatorname{-qcoh}_{\mathsf{lf}}).$$

Corollary 2.5. The functor Ω is fully faithful, and its image forms a set of compact generators in $D^{co}((X, \mathcal{L}, w)\operatorname{-qcoh}_{\mathsf{lf}})$. The following diagram of triangulated functors is commutative:



Here υ , κ , and γ denote the fully faithful functors induced by the natural embeddings of DG-categories of CDG-modules. The two upper vertical lines are the natural contravariant dualities (antiequivalences) on the (absolute derived) categories of locally free matrix factorizations of finite rank and coherent matrix factorizations. The lower horizontal line is the equivalence of categories from Theorem 2.5, and the middle horizontal arrow is the fully faithful functor discussed after the proof of Theorem 2.5.

The above diagram is to be compared with the following subdiagram of the large diagram in the end of Section 2.4:



Here λ denotes the triangulated functor induced by the embedding of DG-categories of CDG-modules (X, \mathcal{L}, w) -qcoh_{lf} $\rightarrow (X, \mathcal{L}, w)$ -qcoh.

Notice that it is clear from these two diagrams that the functor λ is an equivalence of triangulated categories whenever the functor v is. Indeed, if v is an equivalence of categories, then the image of κ is a set of compact generators in the target category, and λ is an infinite direct sum-preserving triangulated functor identifying triangulated subcategories of compact generators, and hence λ is an equivalence. In this case, the functor $\mathcal{D}^{\bullet}_X \otimes_{\mathcal{O}_X}$ – becomes an autoequivalence of the triangulated category $\mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w)$ -qcoh) and restricts to an autoequivalence of its full subcategory of compact generators $\mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w)$ -coh).

Proof of Corollary 2.5. The assertions in the first sentence follow from the second one, as we know γ to be fully faithful and its image to be a set of compact generators

by Corollary 2.3(1). The commutativity of both squares and the upper left triangle is clear. To check commutativity of the lower right triangle, consider a coherent matrix factorization \mathcal{M} of the potential -w; let \mathcal{E}_{\bullet} be its left resolution in the abelian category $Z^0((X, \mathcal{L}, -w)$ -coh) whose terms \mathcal{E}_n belong to $Z^0((X, \mathcal{L}, -w)$ -coh_{lf}). Then the finite complex of matrix factorizations $\mathcal{H}om_{X-qc}(\mathcal{M}, \mathcal{D}_X^{\bullet})$ maps quasiisomorphically to the bounded-below complex of injective matrix factorizations $\mathcal{H}om_{X-qc}(\mathcal{E}_{\bullet}, \mathcal{D}_X^{\bullet}) \simeq \mathcal{D}_X^{\bullet} \otimes_{\mathcal{O}_X} \mathcal{H}om_{X-qc}(\mathcal{E}_{\bullet}, \mathcal{O}_X)$, so the cone of the corresponding morphism of the total matrix factorizations is coacyclic.

2.6. *w-flat matrix factorizations.* From now on we will assume that for any affine open subscheme $U \subset X$ the element $w|_U$ is not a zero divisor in the $\mathcal{O}(U)$ -module $\mathcal{L}(U)$; in other words, the morphism of sheaves $w : \mathcal{O}_X \to \mathcal{L}$ is injective.

The following results will be used in the proof of the main theorem and its analogues below. Let us call a quasicoherent \mathcal{O}_X -module \mathcal{E} w-flat if the map $w : \mathcal{E} \to \mathcal{E} \otimes_{\mathcal{O}_X} \mathcal{L}$ is injective. Notice that any submodule of a w-flat module is w-flat, so the "w-flat dimension" of a quasicoherent sheaf over X never exceeds 1.

Denote by (X, \mathcal{L}, w) -coh_{*w*-fl} the DG-category of coherent CDG-modules over (X, \mathcal{L}, w) with *w*-flat underlying graded \mathcal{O}_X -modules and by (X, \mathcal{L}, w) -qcoh_{*w*-fl} the similar DG-category of quasicoherent CDG-modules. Let $D^{abs}((X, \mathcal{L}, w)$ -coh_{*w*-fl}), $D^{abs}((X, \mathcal{L}, w)$ -qcoh_{*w*-fl}, and $D^{co}((X, \mathcal{L}, w)$ -qcoh_{*w*-fl}) denote the corresponding derived categories of the second kind.

Furthermore, denote by (X, \mathcal{L}, w) -coh_{*w*-fl∩ffd} the DG-category of coherent CDGmodules over (X, \mathcal{L}, w) whose underlying graded \mathcal{O}_X -modules are both *w*-flat and of finite flat dimension, and by (X, \mathcal{L}, w) -qcoh_{*w*-fl∩lfd} the DG-category of *w*-flat quasicoherent CDG-modules of finite locally free dimension. Let the corresponding exotic derived categories be denoted by $D^{abs}((X, \mathcal{L}, w)$ -coh_{*w*-fl∩lfd}), $D^{abs}((X, \mathcal{L}, w)$ -qcoh_{*w*-fl∩lfd}), and $D^{co}((X, \mathcal{L}, w)$ -qcoh_{*w*-fl∩lfd}).

Corollary 2.6. (a) The functor $D^{co}((X, \mathcal{L}, w) \operatorname{-qcoh}_{w-fl}) \to D^{co}((X, \mathcal{L}, w) \operatorname{-qcoh})$ induced by the embedding of DG-categories $(X, \mathcal{L}, w) \operatorname{-qcoh}_{w-fl} \to (X, \mathcal{L}, w) \operatorname{-qcoh}$ is an equivalence of triangulated categories.

(b) The functor $D^{abs}((X, \mathcal{L}, w) \operatorname{-qcoh}_{w-fl}) \to D^{abs}((X, \mathcal{L}, w) \operatorname{-qcoh})$ induced by the embedding of DG-categories $(X, \mathcal{L}, w) \operatorname{-qcoh}_{w-fl} \to (X, \mathcal{L}, w) \operatorname{-qcoh}$ is an equivalence of triangulated categories.

(c) The functor $D^{abs}((X, \mathcal{L}, w) \operatorname{coh}_{w-fl}) \to D^{abs}((X, \mathcal{L}, w) \operatorname{coh})$ induced by the embedding of DG-categories $(X, \mathcal{L}, w) \operatorname{coh}_{w-fl} \to (X, \mathcal{L}, w) \operatorname{coh}$ is an equivalence of triangulated categories.

(d) The functor $D^{co}((X, \mathcal{L}, w) \operatorname{-qcoh}_{w-\operatorname{fl}\cap\operatorname{lfd}}) \to D^{co}((X, \mathcal{L}, w) \operatorname{-qcoh}_{\operatorname{lfd}})$ induced by the embedding of DG-categories $(X, \mathcal{L}, w) \operatorname{-qcoh}_{w-\operatorname{fl}\cap\operatorname{lfd}} \to (X, \mathcal{L}, w) \operatorname{-qcoh}_{\operatorname{lfd}}$ is an equivalence of triangulated categories.

(e) The functor $D^{abs}((X, \mathcal{L}, w) \operatorname{-qcoh}_{w-fi\cap lfd}) \to D^{abs}((X, \mathcal{L}, w) \operatorname{-qcoh}_{lfd})$ induced by the embedding of DG-categories $(X, \mathcal{L}, w) \operatorname{-qcoh}_{w-fi\cap lfd} \to (X, \mathcal{L}, w) \operatorname{-qcoh}_{lfd}$ is an equivalence of triangulated categories.

(f) The functor $D^{abs}((X, \mathcal{L}, w) \operatorname{coh}_{w-fi\cap ffd}) \to D^{abs}((X, \mathcal{L}, w) \operatorname{coh}_{ffd})$ induced by the embedding of DG-categories $(X, \mathcal{L}, w) \operatorname{coh}_{w-fi\cap ffd} \to (X, \mathcal{L}, w) \operatorname{coh}_{ffd}$ is an equivalence of triangulated categories.

Proof. The proofs are analogous to those of Corollary 2.3(a)–(c) and (g) (except that no induction in d is needed, as it suffices to consider the case d = 1). Parts (d), (e), (f) are analogous to parts (a), (b), (c), respectively. Parts (b), (c), (e), and (f) can be also proven in the way similar to Corollary 2.3(h) and (i).

Remark 2.6. The assertions of parts (a) and (b) hold under somewhat weaker assumptions than above: namely, one does not need to assume the existence of enough vector bundles on X. And one can make parts (d) and (e) hold without vector bundles by replacing the finite locally free dimension condition in their formulation with the finite flat dimension condition. The reason is that there are enough flat sheaves on any reasonable scheme (see Lemma A.1).

In fact, even part (c) does not depend on the existence of vector bundles since a surjective morphism onto a given coherent sheaf \mathcal{M} from a *w*-flat coherent sheaf can be easily constructed, e.g., by starting from a surjective morphism onto \mathcal{M} from a flat quasicoherent sheaf \mathcal{F} and picking a large enough coherent subsheaf in \mathcal{F} . Accordingly, one does not need vector bundles to prove the equivalence of categories in the lower horizontal line in Theorem 2.7 below and the other two equivalences in Theorem 2.8. Replacing locally free sheaves with flat ones in the relevant definitions and assuming the Krull dimension to be finite, one can have the whole of Proposition 2.8 hold without vector bundles as well.

Another alternative is to use *very flat* quasicoherent sheaves, which there are always enough of and which always form a category of finite homological dimension on a quasicompact semiseparated scheme [Positselski 2012, Section 4.1]. Similarly, the existence of vector bundles is not needed for the validity of Theorem 1.4(a) and (b), Proposition 1.5(a), (c), and (d), all the assertions of Sections 1.7 and 1.10, Corollary 2.3(a), (b), (f), (k), and (l), Corollary 2.4(a) and (b), Proposition 2.5, Theorem 2.5, and some other results.

2.7. *Main theorem.* Let $X_0 \subset X$ be the closed subscheme defined locally by the equation w = 0, and $i : X_0 \to X$ be the natural closed embedding. The next theorem is the main result of this paper.

Theorem 2.7. There is a natural equivalence of triangulated categories

$$D^{abs}((X, \mathcal{L}, w)\text{-coh}) \simeq D^{b}_{Sing}(X_0/X).$$

Together with the functor $\Sigma : D^{abs}((X, \mathcal{L}, w) \operatorname{-coh}_{lf}) \to D^{b}_{Sing}(X_{0})$ constructed in [Orlov 2012], this equivalence forms the following diagram of triangulated functors:

where the upper horizontal arrow Σ is fully faithful, the left vertical arrow is fully faithful, the right vertical arrow is a Verdier localization functor, and the lower horizontal line $\mathbb{L}\Xi = \Upsilon^{-1}$ is an equivalence of categories. The square is commutative; the three diagonal arrows i_{\bullet} , i° , i_{\circ} (the middle one pointing down and the two other ones pointing up) are adjoint.

Furthermore, the image of the functor Σ is precisely the full subcategory of objects annihilated by the functor i_{\circ} , or equivalently, by the functor i_{\bullet} . In other words, the image of Σ is equal both to the left and to the right orthogonal complements to the thick subcategory generated by the image of the functor i° ; that is, an object $\mathcal{F} \in \mathsf{D}^{\mathsf{b}}_{Sing}(X_0)$ is isomorphic to $\Sigma(\mathcal{M})$ for some $\mathcal{M} \in \mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w)\operatorname{-coh}_{\mathsf{lf}})$ if and only if for every $\mathcal{E} \in \mathsf{D}^{\mathsf{b}}_{Sing}(X)$, one has

$$\operatorname{Hom}_{\mathsf{D}^{\mathsf{b}}_{\operatorname{Sing}}(X_0)}(i^{\circ}\mathcal{E},\mathcal{F})=0,$$

or equivalently, for every $\mathcal{E} \in \mathsf{D}^{\mathsf{b}}_{Sing}(X)$, one has $\operatorname{Hom}_{\mathsf{D}^{\mathsf{b}}_{Sino}}(X_0)(\mathcal{F}, i^{\circ}\mathcal{E}) = 0$.

The thick subcategory generated by the image of the functor i° is the kernel of the right vertical arrow. So the upper horizontal arrow and the right vertical arrow are included into "exact sequences" of triangulated categories (as marked by the zeros at the ends; there is no exactness at the uppermost rightmost end).

When X is a regular scheme, the functor

$$D^{abs}((X, \mathcal{L}, w)\text{-coh}_{lf}) \longrightarrow D^{abs}((X, \mathcal{L}, w)\text{-coh})$$

is an equivalence of categories by Corollary 2.4(c), and so is the functor $D_{Sing}^{b}(X_0) \rightarrow D_{Sing}^{b}(X_0/X)$ (as explained in Section 2.1). Hence it follows that the functor Σ is

an equivalence of categories, too. Thus we recover the result of Orlov [2012, Theorem 3.5] claiming the equivalence of triangulated categories $D^{abs}((X, \mathcal{L}, w)\text{-coh}_{lf}) \simeq D^{b}_{Sing}(X_0)$ for a regular X.

A counterexample in Section 3.3 will show that when X is not regular, the functor $D^{abs}((X, \mathcal{L}, w)\text{-coh}_{lf}) \rightarrow D^{abs}((X, \mathcal{L}, w)\text{-coh})$ does not have to be an equivalence, and indeed, the thick subcategory generated by $D^{abs}((X, \mathcal{L}, w)\text{-coh}_{lf})$ can be a proper strictly full subcategory in $D^{abs}((X, \mathcal{L}, w)\text{-coh})$.

Proof of the lower horizontal equivalence. To obtain the equivalence of triangulated categories in the lower horizontal line, we will construct triangulated functors in both directions and then check that they are mutually inverse. Given a bounded complex of coherent sheaves \mathcal{F}^{\bullet} over X_0 , consider the CDG-module $\Upsilon(\mathcal{F}^{\bullet})$ over (X, \mathcal{L}, w) with the underlying coherent graded module given by the rule

$$\Upsilon^n(\mathcal{F}^{\bullet}) = \bigoplus_{m \in \mathbb{Z}} i_* \mathcal{F}^{n-2m} \otimes_{\mathcal{O}_X} \mathcal{L}^{\otimes m}$$

and the differential induced by the differential on \mathcal{F}^{\bullet} . Since $d^2 = 0$ on \mathcal{F}^{\bullet} and w acts by zero in $i_*\mathcal{F}^j$, this is a CDG-module. It is clear that Υ is a well-defined triangulated functor $D^{\mathsf{b}}(X_0\operatorname{-coh}) \to D^{\mathsf{abs}}((X, \mathcal{L}, w)\operatorname{-coh})$ since the derived category of bounded complexes over an abelian category coincides with their absolute derived category.

Let us check that Υ annihilates the image of the functor $\mathbb{L}i^*$. It suffices to consider a *w*-flat coherent sheaf \mathcal{E} on X and check that $\Upsilon(\operatorname{coker} w) = 0$, where $w: \mathcal{E} \otimes_{\mathcal{O}_X} \mathcal{L}^{\otimes -1} \to \mathcal{E}$. Indeed, $\Upsilon(\operatorname{coker} w)$ is the cokernel of the injective morphism of contractible coherent CDG-modules $\mathcal{N} \to \mathcal{M}$, where $\mathcal{N}^{2n+1} = \mathcal{M}^{2n+1} = \mathcal{E} \otimes_{\mathcal{O}_X} \mathcal{L}^{\otimes n}$ and $\mathcal{N}^{2n} = \mathcal{E} \otimes_{\mathcal{O}_X} \mathcal{L}^{\otimes n-1}$, while $\mathcal{M}^{2n} = \mathcal{E} \otimes_{\mathcal{O}_X} \mathcal{L}^{\otimes n}$ for $n \in \mathbb{Z}$.

This provides the desired triangulated functor

$$\Upsilon: \mathsf{D}^{\mathsf{b}}_{Sing}(X_0/X) \longrightarrow \mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w)\operatorname{-coh}).$$

The functor in the opposite direction is a version of Orlov's cokernel functor, but in our situation it has to be constructed as a derived functor since the functor of the cokernel of an arbitrary morphism is not exact. Recall the equivalence of triangulated categories $D^{abs}((X, \mathcal{L}, w)\text{-coh}_{w-fl}) \rightarrow D^{abs}((X, \mathcal{L}, w)\text{-coh})$ from Corollary 2.6(c).

Define the functor $\Xi : Z^0((X, \mathcal{L}, w) \operatorname{coh}_{w-\mathrm{fl}}) \to \mathsf{D}^{\mathsf{b}}_{Sing}(X_0/X)$ from the category of *w*-flat coherent CDG-modules over (X, \mathcal{L}, w) and closed morphisms of degree 0 between them to the triangulated category of relative singularities by the rule

$$\Xi(\mathcal{M}) = \operatorname{coker}(d : \mathcal{M}^{-1} \to \mathcal{M}^{0}) = \operatorname{coker}(i^{*}d : i^{*}\mathcal{M}^{-1} \to i^{*}\mathcal{M}^{0}),$$

where the former cokernel, which is by definition a coherent sheaf on X annihilated by w, is considered as a coherent sheaf on X_0 . One can immediately see that the functor Ξ transforms morphisms homotopic to zero into morphisms factorizable

through the restrictions to X_0 of w-flat coherent sheaves on X. Hence the functor Ξ factorizes through the homotopy category $H^0((X, \mathcal{L}, w)\operatorname{-coh}_{w-fl})$.

It is explained in [Polishchuk and Vaintrob 2011, Lemma 3.12] that the functor Ξ is triangulated (see also Lemma 3.6 below) and in [Orlov 2012, Proposition 3.2] that the functor Ξ factorizes through $D^{abs}((X, \mathcal{L}, w) \operatorname{-coh}_{w-fl})$. The latter assertion can be also deduced by considering the complex (1.3) from [Polishchuk and Vaintrob 2011]. Indeed, the complex $i^*\mathcal{M}$ corresponding to the total CDG-module \mathcal{M} of an exact triple in $\mathcal{B}\operatorname{-coh}_{w-fl}$ is the total complex of an exact triple of complexes in the exact category $\mathsf{E}_{X_0/X}$ from Remark 2.1; hence the complex $i^*\mathcal{M}$ is exact subcategory in the abelian category of coherent sheaves over X_0 . So we obtain the triangulated functor

$$\Xi: \mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w) \operatorname{-coh}_{w-\mathrm{fl}}) \longrightarrow \mathsf{D}^{\mathsf{b}}_{Sing}(X_0/X),$$

and consequently, the left derived functor

$$\mathbb{L}\Xi: \mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w)\operatorname{-coh}) \longrightarrow \mathsf{D}^{\mathsf{b}}_{Sing}(X_0/X).$$

Let us check that the two functors Υ and $\mathbb{L}\Xi$ are mutually inverse. For any *w*-flat coherent CDG-module \mathcal{M} over (X, \mathcal{L}, w) , there is a natural surjective closed morphism of CDG-modules $\phi : \mathcal{M} \to \Upsilon \Xi (\mathcal{M})$ with a contractible kernel. Clearly, $\phi : \mathrm{Id} \to \Upsilon \mathbb{L}\Xi$ is an (iso)morphism of functors.

Conversely, any object of $D_{Sing}^{b}(X_0/X)$ can be represented by a coherent sheaf on X_0 , and any morphism in $D_{Sing}^{b}(X_0/X)$ is isomorphic to a morphism coming from the abelian category of such coherent sheaves. Indeed, the bounded-above derived category $D^{-}(X_0$ -coh) of coherent sheaves over X_0 is equivalent to the boundedabove derived category $D^{-}(X_0$ -coh_{lf}) of locally free sheaves; using a truncation far enough to the left, one can represent any object or morphism in $D_{Sing}^{b}(X_0/X)$ by a long enough shift of a coherent sheaf or a morphism of coherent sheaves. Now for any coherent sheaf \mathcal{F} on X_0 , there is a natural distinguished triangle

$$\mathcal{F} \otimes_{\mathcal{O}_{X_0}} i^* \mathcal{L}^{\otimes -1}[1] \longrightarrow \mathbb{L} i^* i_* \mathcal{F} \longrightarrow \mathcal{F} \longrightarrow \mathcal{F} \otimes_{\mathcal{O}_{X_0}} i^* \mathcal{L}^{\otimes -1}[2]$$

in D^b(X_0 -coh), which provides a natural isomorphism $\mathcal{F} \simeq \mathcal{F} \otimes_{\mathcal{O}_{X_0}} i^* \mathcal{L}^{\otimes -1}[2]$ in D^b_{Sing}(X_0/X).

Let \mathcal{F} be a coherent sheaf on X_0 ; pick a vector bundle \mathcal{E} on X together with a surjective morphism $\mathcal{E} \to i_*\mathcal{F}$ with the kernel \mathcal{E}' . Then the CDG-module \mathcal{M} over (X, \mathcal{L}, w) with the components $\mathcal{M}^{2n} = \mathcal{E} \otimes_{\mathcal{O}_X} \mathcal{L}^{\otimes n}$ and $\mathcal{M}^{2n-1} = \mathcal{E}' \otimes_{\mathcal{O}_X} \mathcal{L}^{\otimes n}$ maps surjectively onto $\Upsilon(\mathcal{F})$ with a contractible kernel, and $\mathbb{L} \cong \Upsilon(\mathcal{F}) = \Xi(\mathcal{M}) = \mathcal{F}$ (cf. [Lin and Pomerleano 2013, Lemma 2.18]). Denote the isomorphism we have constructed by $\psi : \mathbb{L} \cong \Upsilon(\mathcal{F}) \to \mathcal{F}$. The composition $\Upsilon \psi \circ \phi \Upsilon : \Upsilon(\mathcal{F}) \to$ $\Upsilon(\mathcal{F}) \to \Upsilon(\mathcal{F})$ is clearly the identity morphism. It is obvious that ψ commutes with any morphism of coherent sheaves \mathcal{F} on X_0 , but checking that it commutes with all morphisms, or all isomorphisms, in $\mathsf{D}^{\mathsf{b}}_{Sing}(X_0/X)$ is a little delicate (cf. Remark 2.7 below).

Notice that $\Upsilon \psi$ is an (iso)morphism of functors since $\phi \Upsilon$ is, and consequently $\mathbb{L}\Xi \Upsilon \psi$ is an (iso)morphism of functors. Thus it remains to check that the functor $\mathbb{L}\Xi \Upsilon$ is faithful, i.e., does not annihilate any morphisms. Indeed, any morphism in $D^{b}_{Sing}(X_0/X)$ is isomorphic to a morphism coming from the abelian category of coherent sheaves on X_0 , and the functor $\mathbb{L}\Xi \Upsilon$ transforms such morphisms into isomorphic ones. The construction of the equivalence of categories in the lower horizontal line is finished. One still has to check that the isomorphisms ϕ commute with the isomorphisms $\Upsilon \Xi (\mathcal{M}[1]) \simeq \Upsilon \Xi (\mathcal{M})[1]$, but this is straightforward.

Alternatively, one can use *w*-flat coherent sheaves on *X* or objects of the exact category $E_{X_0/X}$ of coherent sheaves on X_0 (as applicable) instead of the locally free sheaves everywhere in the above argument.

Proof of "exactness" in the upper line. We start with a discussion of the three adjoint functors in the right upper corner. The functor i_{\circ} right adjoint to the functor $i^{\circ} : D^{b}_{Sing}(X) \to D^{b}_{Sing}(X_{0})$ was constructed in Section 2.1.

To construct the left adjoint functor to i° , notice that the right derived functor of the subsheaf with scheme-theoretic support in the closed subscheme

$$\mathbb{R}i^{!}: \mathsf{D}^{\mathsf{b}}(X\operatorname{-coh}) \longrightarrow \mathsf{D}^{\mathsf{b}}(X_{0}\operatorname{-coh})$$

only differs from the functor $\mathbb{L}i^*$ by a shift and a twist; $\mathbb{R}i^!\mathcal{E}^{\bullet} \simeq \mathbb{L}i^*\mathcal{E}^{\bullet} \otimes_{\mathcal{O}_{X_0}} \mathcal{L}|_{X_0}[-1]$. One can check this first for *w*-flat coherent sheaves \mathcal{E} , when both objects to be identified are shifts of sheaves, so it suffices to compare their direct images under *i*, which are both computed by the same two-term complex $\mathcal{E} \to \mathcal{E} \otimes_{\mathcal{O}_X} \mathcal{L}$; then replace a complex \mathcal{E}^{\bullet} with a finite complex of *w*-flat coherent sheaves (for a general result of this kind, see [Neeman 1996, Theorem 5.4]).

Hence the functor $\mathbb{R}i^{!}$ takes Perf(X) to $Perf(X_{0})$ and induces a triangulated functor $i^{\bullet}: \mathbb{D}^{b}_{Sing}(X) \to \mathbb{D}^{b}_{Sing}(X_{0})$ right adjoint to i_{\circ} . It follows that the functor $i_{\bullet}(\mathcal{F}) = i_{\circ}(\mathcal{F}) \otimes_{\mathcal{O}_{X}} \mathcal{L}[-1]$ is left adjoint to the functor i° .

To prove the vanishing of the composition of functors in the upper line and the orthogonality assertions, notice that

$$\operatorname{Hom}_{\operatorname{D}^{\mathrm{b}}_{\operatorname{Sing}}(X_0)}(i^{\circ}\mathcal{E}, \Sigma\mathcal{M}) \simeq \operatorname{Hom}_{\operatorname{D}^{\mathrm{b}}_{\operatorname{Sing}}(X)}(\mathcal{E}, i_{\circ}\Sigma\mathcal{M})$$

and $i_*\Sigma(\mathcal{M}) = \operatorname{coker}(\mathcal{M}^{-1} \to \mathcal{M}^0) \in Perf(X)$ for any $\mathcal{M} \in D^{\operatorname{abs}}((X, \mathcal{L}, w)\operatorname{-coh}_{\mathsf{lf}})$ since the morphism $\mathcal{M}^{-1} \to \mathcal{M}^0$ of locally free sheaves on X is injective. Similarly,

$$\operatorname{Hom}_{\operatorname{D}^{\mathrm{b}}_{\operatorname{Sing}}(X_0)}(\Sigma\mathcal{M}, i^{\circ}\mathcal{E}) \simeq \operatorname{Hom}_{\operatorname{D}^{\mathrm{b}}_{\operatorname{Sing}}(X)}(i_{\bullet}\Sigma\mathcal{M}, \mathcal{E})$$

and $i_{\bullet}\Sigma(\mathcal{M}) = i_{\circ}\Sigma(\mathcal{M}) \otimes_{\mathcal{O}_X} \mathcal{L}[-1] = 0$ in $\mathsf{D}^{\mathsf{b}}_{Sing}(X)$.

Obviously, our derived cokernel functor $\mathbb{L}\Xi$ makes a commutative diagram with the cokernel functor Σ from [Orlov 2012]. The left vertical arrow is fully faithful by Corollary 2.3(i). The assertion that the upper horizontal arrow is fully faithful is due to Orlov [2012, Theorem 3.4]. We have just obtained a new proof of it with our methods. Indeed, it follows from orthogonality that the functor $D_{Sing}^{b}(X_0) \rightarrow$ $D_{Sing}^{b}(X_0/X)$ induces isomorphisms on the groups of morphisms between any two objects, one of which comes from $D^{abs}((X, \mathcal{L}, w)\operatorname{-coh}_{lf})$. Conversely, Orlov's theorem together with the orthogonality argument and the equivalence of categories in the lower horizontal line imply that the left vertical arrow is fully faithful.

Now assume that $i_{\circ}\mathcal{F} = 0$ for some $\mathcal{F} \in D^{b}_{Sing}(X_{0})$. Clearly, there exists $m \geq 0$ and a coherent sheaf \mathcal{K} on X_{0} such that $\mathcal{F} \simeq \mathcal{K}[m]$ in $D^{b}_{Sing}(X_{0})$. Then $i_{*}\mathcal{K}$ is a perfect complex, i.e., a coherent sheaf of finite flat dimension on X. Let us view it as an object of (X, \mathcal{L}, w) -coh_{ffd}; i.e., consider the CDG-module \mathcal{N} over (X, \mathcal{L}, w) with the components $\mathcal{N}^{2n} = i_{*}\mathcal{K} \otimes_{\mathcal{O}_{X}} \mathcal{L}^{\otimes n}$ and $\mathcal{N}^{2n+1} = 0$.

The construction of the cokernel functor Σ can be straightforwardly extended to w-flat coherent matrix factorizations of finite flat dimension, providing a triangulated functor

$$\widetilde{\Sigma}$$
: $\mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w) \operatorname{-coh}_{w-\mathsf{fl}\cap\mathsf{ffd}}) \longrightarrow \mathsf{D}^{\mathsf{b}}_{Sing}(X_0).$

The functor $\tilde{\Sigma}$ is well-defined since one has $i^*\mathcal{M} \in Perf(X_0)$ for any *w*-flat coherent sheaf \mathcal{M} of finite flat dimension on *X*. Using the equivalence of triangulated categories $\mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w)\operatorname{-coh}_{w\operatorname{-fl}\cap \mathrm{fld}}) \simeq \mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w)\operatorname{-coh}_{\mathrm{fld}})$ from Corollary 2.6(f), one constructs the derived functor

$$\mathbb{L}\widetilde{\Sigma}: \mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w) \operatorname{-coh}_{\mathrm{ffd}}) \longrightarrow \mathsf{D}^{\mathsf{b}}_{Sing}(X_0)$$

in the same way as it was done above for the derived functor $\mathbb{L}\Xi$. Since the functor $\mathbb{D}^{abs}((X, \mathcal{L}, w)\operatorname{-coh}_{\mathsf{lf}}) \to \mathbb{D}^{abs}((X, \mathcal{L}, w)\operatorname{-coh}_{\mathsf{ffd}})$ is an equivalence of categories by Corollary 2.3(g), the (essential) images of the functors Σ and $\mathbb{L}\widetilde{\Sigma}$ coincide.

Let us check that $\mathbb{L}\widetilde{\Sigma}(\mathcal{N}) \simeq \mathcal{K}$ as an object of $\mathsf{D}^{\mathsf{b}}_{Sing}(X_0)$. We argue as above, picking a vector bundle \mathcal{E} on X together with a surjective morphism $\mathcal{E} \to i_*\mathcal{K}$ with the kernel \mathcal{E}' . Then the CDG-module \mathcal{M} over (X, \mathcal{L}, w) with the components $\mathcal{M}^{2n} = \mathcal{E} \otimes_{\mathcal{O}_X} \mathcal{L}^{\otimes n}$ and $\mathcal{M}^{2n-1} = \mathcal{E}' \otimes_{\mathcal{O}_X} \mathcal{L}^{\otimes n}$ maps surjectively onto \mathcal{N} with a contractible kernel. Hence the object $\mathcal{M} \in (X, \mathcal{L}, w)$ -coh_{w-fl \cap ffd} is isomorphic to \mathcal{N} in $\mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w)$ -coh_{ffd}), and we have $\mathbb{L}\widetilde{\Sigma}(\mathcal{N}) = \widetilde{\Sigma}(\mathcal{M}) = \mathcal{K}$. Therefore, the object $\mathcal{K} \in \mathsf{D}^{\mathsf{b}}_{Sing}(X_0)$ belongs to the (essential) image of the functor Σ , and it follows that so does the object $\mathcal{F} \simeq \mathcal{K}[m]$.

One can strengthen the above argument so as to obtain a construction of the (partial) inverse functor Δ to the functor Σ similar to the above construction of the functor Υ inverse to the functor $\mathbb{L}\Xi$. Consider the full subcategory $F_{X_0/X} \subset X_0$ -coh in the abelian category of coherent sheaves on X_0 consisting of all the sheaves \mathcal{F}

such that the sheaf $i_*\mathcal{F}$ has finite flat dimension (i.e., is a perfect complex) on X. The category $F_{X_0/X}$ contains all the locally free sheaves on X_0 and is closed under the kernels of surjections, the cokernels of embeddings, and the extensions.

Hence $F_{X_0/X}$ is an exact subcategory in X_0 -coh. The natural functor

$$\mathsf{D}^{\mathsf{b}}(\mathsf{F}_{X_0/X}) \longrightarrow \mathsf{D}^{\mathsf{b}}(X_0\operatorname{-coh})$$

is fully faithful; its image coincides with the kernel of the composition of the direct image and Verdier localization functors $D^{b}(X_{0}\text{-coh}) \rightarrow D^{b}(X\text{-coh}) \rightarrow D^{b}_{Sing}(X)$. Accordingly, the quotient category $D^{b}(\mathsf{F}_{X_{0}/X})/D^{b}(X_{0}\text{-coh}_{lf})$ is identified with the kernel of the direct image functor $i_{\circ}: D^{b}_{Sing}(X_{0}) \rightarrow D^{b}_{Sing}(X)$.

Now the functor

$$\Delta: \mathsf{D}^{\mathsf{b}}(\mathsf{F}_{X_0/X})/\mathsf{D}^{\mathsf{b}}(X_0\operatorname{-coh}_{\mathsf{lf}}) \longrightarrow \mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w)\operatorname{-coh}_{\mathsf{ffd}})$$

is constructed in the way similar to the construction of the functor Υ , by taking the direct image from X_0 to X and applying the periodicity summation. That is,

$$\Delta^n(\mathcal{F}^{\bullet}) = \bigoplus_{m \in \mathbb{Z}} i_* \mathcal{F}^{n-2m} \otimes_{\mathcal{O}_X} \mathcal{L}^{\otimes m}$$

for any $\mathcal{F}^{\bullet} \in \mathsf{D}^{\mathsf{b}}(\mathsf{F}_{X_0/X})$. One checks that the functor Δ is inverse to the functor $\mathbb{L}\widetilde{\Sigma}$, the latter being viewed as a functor taking values in the triangulated subcategory $\mathsf{D}^{\mathsf{b}}(\mathsf{F}_{X_0/X})/\mathsf{D}^{\mathsf{b}}(X_0\operatorname{-coh}_{\mathsf{lf}}) \subset \mathsf{D}^{\mathsf{b}}_{Sing}(X_0)$, in the same way as it was done above for the functors Υ and $\mathbb{L}\Xi$. This provides yet another proof of the fact that the functor Σ is fully faithful, together with another proof of our description of its image. It is also obvious from the constructions that the functor Δ makes a commutative diagram with the functor Υ .

Remark 2.7. The somewhat tricky technical argument in the first part of the above proof can be clarified and generalized using the approach developed by the first author in [Efimov 2013, Appendix A].

Let C be an abelian category, $L : C \to C$ be its covariant autoequivalence, and $w : \mathrm{Id} \to L$ be a natural transformation commuting with L (that is for any object $B \in C$, one has $w_{L(B)} = L(w_B)$). Let MF(C, L, w) denote the abelian category of "matrix factorizations of w in C", i.e., pairs of objects U^0 , $L^{1/2}(U^1) \in C$ endowed with pairs of morphisms $U^0 \to L^{1/2}(U^1)$, $L^{1/2}(U^1) \to L(U^0)$ such that the compositions

$$U^0 \longrightarrow L^{1/2}(U^1) \longrightarrow L(U^0)$$
 and $L^{1/2}(U^1) \longrightarrow L(U^0) \longrightarrow L^{3/2}(U^1)$

are equal to w_{U^0} and $w_{L^{1/2}(U^1)}$, respectively. Given a matrix factorization $M = (U^0, U^1)$, one sets

$$M^n = L^{n/2}(U^{n \mod 2}).$$
Passing to the quotient category by the ideal of morphisms homotopic to zero, one obtains the homotopy category of matrix factorizations of w in C, and their absolute derived category, denoted by $D^{abs}(C, L, w)$, is produced by the Verdier localization procedure similar to the one discussed in Section 1.3. (Cf. [Positselski 2011a, Remark 4.3].)

Let $C_0 \subset C$ denote the full subcategory formed by all the objects $A \in C$ for which $w_A = 0$; so C_0 is an abelian subcategory in C closed under subobjects and quotient objects. An object $B \in C$ is said to have no *w*-torsion if the morphism w_B is injective; and one says that the potential (natural transformation) *w* does not divide zero in C if every object of C is the quotient object of an object without *w*-torsion. Let $i_* : C_0 \to C$ denote the exact identity embedding functor and $i^* : C \to C_0$ be the functor left adjoint to i_* , so $i^*(B) = \operatorname{coker}(w_{L^{-1}(B)} : L^{-1}(B) \to B)$. Assuming that *w* does not divide zero (as we do in the sequel), one can construct the left derived functor $\mathbb{L}i^* : D^{\mathrm{b}}(C) \to D^{\mathrm{b}}(C_0)$ with $\mathbb{L}_s i^*(B) = 0$ for all $s \neq 0, 1$ and any object $B \in C$. The functor $\mathbb{L}i^*$ is left adjoint to the triangulated functor $i_* : D^{\mathrm{b}}(C_0) \to D^{\mathrm{b}}(C)$ induced by the identity embedding $i_* : C_0 \to C$.

Similarly, let $v^n : C_0 \to MF(C, L, w)$ denote the exact functor assigning to an object $A \in C_0$ the matrix factorization M with $M^n = A$ and $M^{n+1} = 0$, and let $\xi^n : MF(C, L, w) \to C_0$ be the functor left adjoint to v^n , assigning the object coker $(M^{n-1} \to M^n) \in C_0$ to a matrix factorization M. Considering the bounded derived category $D^bMF(C, L, w)$ of the abelian category MF(C, L, w), one can construct the left derived functor

$$\mathbb{L}\xi^{n}: \mathsf{D}^{\mathsf{b}}MF(\mathsf{C}, L, w) \longrightarrow \mathsf{D}^{\mathsf{b}}(\mathsf{C}_{0});$$

once again, the functor $\mathbb{L}\xi^n$ is left adjoint to $\upsilon^n : D^{\mathrm{b}}(\mathsf{C}_0) \to D^{\mathrm{b}}MF(\mathsf{C}, L, w)$ and one has $\mathbb{L}_s\xi^n(M) = 0$ for all $s \neq 0, 1$ and any matrix factorization $M \in MF(\mathsf{C}, L, w)$.

Then the composition of the functor $\upsilon^n : D^b(C_0) \to D^b MF(C, L, w)$ with the totalization functor $D^b MF(C, L, w) \to D^{abs}(C, L, w)$ induces an equivalence of triangulated categories

$$\Upsilon^{n}: \mathsf{D}^{\mathsf{b}}(\mathsf{C}_{0})/\langle \mathbb{L}i^{*}\mathsf{D}^{\mathsf{b}}(\mathsf{C})\rangle \longrightarrow \mathsf{D}^{\mathsf{abs}}(\mathsf{C}, L, w)$$

between the quotient category of the derived category $D^{b}(C_{0})$ by the thick subcategory generated by the image of the functor $\mathbb{L}i^{*}$ and the absolute derived category of matrix factorizations. The composition of the functor $\mathbb{L}\xi^{n} : D^{b}MF(C, L, w) \rightarrow$ $D^{b}(C_{0})$ with the Verdier localization functor $D^{b}(C_{0}) \rightarrow D^{b}(C_{0})/\langle \mathbb{L}i^{*}D^{b}(C)\rangle$ factorizes through the totalization functor $D^{b}MF(C, L, w) \rightarrow D^{abs}(C, L, w)$, providing the triangulated functor

$$\mathbb{L}\Xi^{n}: \mathsf{D}^{\mathsf{abs}}(\mathsf{C}, L, w) \longrightarrow \mathsf{D}^{\mathsf{b}}(\mathsf{C}_{0})/\langle \mathbb{L}i^{*}\mathsf{D}^{\mathsf{b}}(\mathsf{C})\rangle$$

inverse to Υ^n .

Indeed, let $F^n: MF(C, L, w) \to C$ denote the forgetful functor taking a matrix factorization M to the object $M^n \in C$, and let $G^{n-}: C \to MF(C, L, w)$ denote the functor left adjoint to F^{n-1} (and right adjoint to F^n); so the functor G^{n-} takes an object $B \in C$ to a contractible matrix factorization M with $M^{n-1} = M^n = B$ (cf. the constructions of the functors G^+ and G^- in the proofs in Sections 1.4 and 1.6). It is claimed that the induced triangulated functors $G^{n-}: D^b(C) \to D^bMF(C, L, w)$ and $v^n: D^b(C_0) \to D^bMF(C, L, w)$ are fully faithful and their images form a semiorthogonal decomposition of the derived category $D^bMF(C, L, w)$.

To check the first assertion, it suffices to notice that the triangulated functor G^{n-} is left adjoint to the functor $F^{n-1} : D^b MF(C, L, w) \to D^b(C)$, and their composition $F^{n-1} \circ G^{n-}$ is the identity endofunctor on $D^b(C)$. Similarly, the composition of triangulated functors $\mathbb{L}\xi^n \circ \upsilon^n$ is the identity endofunctor on $D^b(C_0)$, so υ^n is fully faithful as a functor between the derived categories. Furthermore, one has $F^{n-1} \circ \upsilon^n = 0 = \mathbb{L}\xi^n \circ G^{n-}$, implying the semiorthogonality. Finally, for any matrix factorization M whose terms are objects without w-torsion, there is a short exact sequence

$$0 \longrightarrow G^{n-}F(M) \longrightarrow M \longrightarrow \upsilon^n \xi^n M \longrightarrow 0$$

in MF(C, L, w) and $\mathbb{L}\xi^n M = \xi^n M$, proving the decomposition claim.

Now we notice that for any object $B \in C$ having no *w*-torsion, there is a short exact sequence

$$0 \longrightarrow G^{(n+2)-}(B) \longrightarrow G^{(n+1)-}(B) \longrightarrow \upsilon^n j^* B \longrightarrow 0$$

in MF(C, L, w). According to (the proof of) [Efimov 2013, Proposition A.3(1) and (2)], the totalization functor $D^bMF(C, L, w) \rightarrow D^{abs}(C, L, w)$ is the Verdier localization functor by the thick subcategory generated by the objects of the form $G^{n-}(B) = G^{(n+2)-}L(B)$ and $G^{(n+1)-}(B) \in MF(C, L, w) \subset D^bMF(C, L, w)$. The assertions about the existence of triangulated functors Υ^n and $\mathbb{L}\Xi^n$ and their being mutually inverse equivalences of categories follow from these observations.

Returning to a separated Noetherian scheme X with enough vector bundles and the Cartier divisor $X_0 \subset X$ of a global section w of a line bundle \mathcal{L} on X, the above approach based on [loc. cit., Proposition A.3] provides an elegant construction of Orlov's triangulated cokernel functor $\Sigma : D^{abs}((X, \mathcal{L}, w) \operatorname{-coh}_{lf}) \to D^{b}_{Sing}(X_0)$ in addition to a proof of our equivalence of categories $D^{abs}((X, \mathcal{L}, w) \operatorname{-coh}) \simeq D^{b}_{Sing}(X_0/X)$.

2.8. *Infinite matrix factorizations.* Following [Orlov 2004, paragraphs after Remark 1.9], one can define a "large" version of the triangulated category of singularities $D'_{Sing}(X)$ of a scheme X as the quotient category of the bounded derived category of quasicoherent sheaves $D^{b}(X$ -qcoh) by the thick subcategory $D^{b}(X$ -qcoh_{lf}) of bounded complexes of locally free sheaves (of infinite rank). When X has

finite Krull dimension, the latter subcategory coincides with the thick subcategory $D^{b}(X-qcoh_{fl})$ of bounded complexes of flat sheaves (see Remark 1.4).

Similarly, let $Z \subset X$ be a closed subscheme such that \mathcal{O}_Z has finite flat dimension as an \mathcal{O}_X -module. Let us define a "large" triangulated category of relative singularities $\mathsf{D}'_{Sing}(Z/X)$ as the quotient category of $\mathsf{D}^{\mathsf{b}}(Z\operatorname{-qcoh})$ by the minimal thick subcategory containing the image of the functor $\mathbb{L}i^* : \mathsf{D}^{\mathsf{b}}(X\operatorname{-qcoh}) \to \mathsf{D}^{\mathsf{b}}(Z\operatorname{-qcoh})$ and closed under those infinite direct sums that exist in $\mathsf{D}^{\mathsf{b}}(Z\operatorname{-qcoh})$. The quotient category of $\mathsf{D}^{\mathsf{b}}(Z\operatorname{-qcoh})$ by the minimal thick subcategory containing $\mathbb{L}i^*\mathsf{D}^{\mathsf{b}}(X\operatorname{-qcoh})$ (without the direct sum closure) will be also of interest to us; let us denote it by $\mathsf{D}''_{Sine}(Z/X)$.

Lemma 2.8. The triangulated categories $D'_{Sing}(Z/X)$ and $D''_{Sing}(Z/X)$ are quotient categories of $D'_{Sing}(Z)$. When the scheme X is regular of finite Krull dimension, these three triangulated categories coincide.

Proof. To prove the first assertion, let us show that any locally free sheaf on Z, considered as an object of $D^b(Z\operatorname{-qcoh})$, is a direct summand of a bounded complex whose terms are direct sums of locally free sheaves of finite rank restricted from X. Indeed, pick a finite left resolution of a given locally free sheaf on Z with the middle terms as above, long enough compared to the number of open subsets in an affine covering of Z. Then the corresponding Ext class between the cohomology sheaves at the rightmost and leftmost terms has to vanish in view of the Mayer–Vietoris sequence for Ext groups between quasicoherent sheaves [Orlov 2004, Lemma 1.12]. Hence the rightmost term is a direct summand of the complex formed by the middle terms.

The second assertion holds for the categories $D''_{Sing}(Z/X)$ and $D'_{Sing}(Z)$ since any quasicoherent sheaf on a regular scheme of finite Krull dimension has a finite left resolution consisting of locally free sheaves. To identify these two categories with $D'_{Sing}(Z/X)$, one needs to know that the subcategory of bounded complexes of locally free sheaves on Z is closed under those infinite direct sums that exist in D^b(Z-qcoh). The latter is true for any Noetherian scheme Z of finite Krull dimension with enough vector bundles since the finitistic projective dimension of a commutative ring of finite Krull dimension is finite [Raynaud and Gruson 1971, Théorème II.3.2.6].

Now let \mathcal{L} be a line bundle on X, $w \in \mathcal{L}(X)$ be a global section corresponding to an injective morphism of sheaves $\mathcal{O}_X \to \mathcal{L}$, and $X_0 \subset X$ be the locus of w = 0.

Proposition 2.8. There is a natural equivalence of triangulated categories

$$D^{abs}((X, \mathcal{L}, w)$$
-qcoh) $\simeq D''_{Sing}(X_0/X)$

Together with the infinite-rank version $\Sigma' : D^{abs}((X, \mathcal{L}, w) \operatorname{-qcoh}_{\mathsf{lf}}) \to D'_{Sing}(X_0)$ of Orlov's cokernel functor Σ from [Orlov 2012], this equivalence forms the following diagram of triangulated functors:



where the upper horizontal arrow Σ' is fully faithful, the left vertical arrow is fully faithful, the right vertical arrow is the Verdier localization functor by the thick subcategory generated by the image of the diagonal down arrow i°, and the lower horizontal line is an equivalence of categories. The square is commutative; the three diagonal arrows i_{\bullet} , i° , i_{\bullet} are adjoint.

Furthermore, the image of the functor Σ' is precisely the full subcategory of objects annihilated by the functor i_{\circ} , or equivalently, by the functor i_{\bullet} . In other words, the image of Σ' is equal both to the left and to the right orthogonal complements to (the thick subcategory generated by) the image of the functor i° .

Proof. The proof is completely similar to that of Theorem 2.7. It uses Corollaries 2.6(b), 2.3(h), 2.6(e), and 2.3(c). The first assertion can be also obtained as a particular case of the result of Remark 2.7.

Alternatively, one can prove that the functor Σ' is fully faithful in the same way as it was done for the functor Σ in [Orlov 2012, Theorem 3.4], and deduce the assertion that the left vertical arrow is fully faithful from the orthogonality.

Note that one can check in a straightforward way that the functor Σ' annihilates the objects coacyclic with respect to (X, \mathcal{L}, w) -qcoh_{lf}. This provides another proof of Corollary 2.3(d), working in the assumption that w is a local nonzero-divisor. \Box

The functors Σ and Σ' together with the direct image functors i_{\circ} form the commutative diagram of an embedding of "exact sequences" of triangulated functors:

The leftmost vertical arrow is fully faithful by Corollary 2.3(j). The other two vertical arrows are fully faithful by Orlov's theorem [2004, Proposition 1.13] claiming

that the functor $D_{Sing}^{b}(X) \to D'_{Sing}(X)$ is fully faithful for any separated Noetherian scheme X with enough vector bundles. The leftmost nontrivial terms in both lines are the kernels of the rightmost arrows by Theorem 2.7 and Proposition 2.8.

Theorem 2.8. There is a natural equivalence of triangulated categories

 $\mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w)\operatorname{-qcoh}) \simeq \mathsf{D}'_{Sing}(X_0/X)$

forming a commutative diagram of triangulated functors:



with the equivalences of categories from Theorem 2.7 and Proposition 2.8. The upper vertical arrows are fully faithful, the lower ones are Verdier localization functors, and the vertical compositions are fully faithful. The categories in the lower line admit arbitrary direct sums, and the images of the vertical compositions are sets of compact generators in the target categories.

Proof. The construction of the desired equivalence of categories is very similar to the construction of the equivalence of categories in Theorem 2.7 and Proposition 2.8. Using Corollary 2.6(a), one defines the infinite-rank version of the functor $\mathbb{L}\Xi$, then shows that the obvious infinite-rank version of the functor Υ is inverse to it. Notice that the functor $\Xi : Z^0((X, \mathcal{L}, w)\operatorname{-qcoh}_{w-fl}) \to D^b(X_0\operatorname{-qcoh})$ preserves infinite direct sums and the functor $\Upsilon : D^b(X_0\operatorname{-qcoh}) \to D^{\operatorname{co}}((X, \mathcal{L}, w)\operatorname{-qcoh})$ preserves those infinite direct sums that exist in $D^b(X_0\operatorname{-qcoh})$, so the functors

$$\begin{split} \Xi : \mathsf{D}^{\mathsf{co}}((X,\mathcal{L},w)\operatorname{-qcoh}_{w\operatorname{-fl}}) &\longrightarrow \mathsf{D}'_{Sing}(X_0/X), \\ \Upsilon : \mathsf{D}'_{Sing}(X_0/X) &\longrightarrow \mathsf{D}^{\mathsf{co}}((X,\mathcal{L},w)\operatorname{-qcoh}) \end{split}$$

are well-defined.

The upper left vertical arrow is fully faithful by Corollary 2.3(k); it follows that the upper right vertical arrow is fully faithful, too. The assertions about the vertical compositions are proved similarly. The category $D'_{Sing}(X_0/X)$ admits arbitrary direct sums, since the category $D^{co}((X, \mathcal{L}, w)$ -qcoh) does. By Corollary 2.3(l), the left vertical composition is fully faithful and its image is a set of compact generators in the target, so the right vertical composition has the same properties. The following square diagram of triangulated functors is commutative:

$$\begin{array}{c} \mathsf{D}^{\mathsf{co}}((X,\mathcal{L},w)\operatorname{-qcoh}_{\mathsf{lf}}) \xrightarrow{\Sigma'} \mathsf{D}'_{Sing}(X_{0}) \\ \\ \downarrow \qquad \qquad \downarrow \\ \mathsf{D}^{\mathsf{co}}((X,\mathcal{L},w)\operatorname{-qcoh}) = \mathsf{D}'_{Sing}(X_{0}/X) \end{array}$$

The upper horizontal arrow Σ' is fully faithful; the right vertical arrow is a Verdier localization functor. The lower line is an equivalence of triangulated categories. Nothing is claimed about the left vertical arrow in general.

When the scheme X is Gorenstein of finite Krull dimension, the left vertical arrow is an equivalence of categories by Corollary 2.4(a). When X is also regular, the right vertical arrow is an equivalence of categories by Lemma 2.8. So Σ' is an equivalence of categories $D^{abs}((X, \mathcal{L}, w)\operatorname{-qcoh}_{lf}) \simeq D'_{Sing}(X_0)$ and we have obtained a strengthened version of [Polishchuk and Vaintrob 2011, Theorem 4.2] (in the scheme case).

Remark 2.8. It is well-known that the Verdier localization functor of a triangulated category with infinite direct sums by a thick subcategory closed under infinite direct sums preserves infinite direct sums [Neeman 2001, Lemma 3.2.10]. This result is not applicable to the localization functors $D^{b}(X-qcoh) \rightarrow D'_{Sing}(X)$ and $D^{b}(Z-qcoh) \rightarrow D'_{Sing}(Z/X)$, as the category $D^{b}(X-qcoh)$ does not admit arbitrary infinite direct sums.

Using the equivalence of categories from Theorem 2.8 and the observation that the functor Υ preserves infinite direct sums, one can show that the localization functor $D^{b}(X_{0}-qcoh) \rightarrow D'_{Sing}(X_{0}/X)$ takes those infinite direct sums that exist in $D^{b}(X_{0}-qcoh)$ into direct sums in the triangulated category of relative singularities $D'_{Sing}(X_{0}/X)$ of the zero locus of w in X. However, there is *no* obvious reason why the localization functor $D^{b}(X_{0}-qcoh) \rightarrow D'_{Sing}(X_{0})$ should take those infinite direct sums that exist in $D^{b}(X_{0}-qcoh)$ into direct sums in the absolute triangulated category of singularities $D'_{Sing}(X_{0})$.

That is the problem one encounters attempting to prove that the kernel of the localization functor $D'_{Sing}(X_0) \rightarrow D'_{Sing}(X_0/X)$ is semiorthogonal to the image of the functor Σ' .

2.9. Stable derived category. Following Krause [2005], we define the stable derived category of a Noetherian scheme X as the homotopy category of acyclic unbounded complexes of injective quasicoherent sheaves on X. As explained below, this is another (and in some respects better) "large" version of the triangulated category of singularities of X; for this reason, we denote it by $D_{Sing}^{st}(X)$.

In view of Lemma 1.7(b) (see also [Positselski 2010, Remark 5.4]), one can equivalently define $D_{Sing}^{st}(X)$ as the quotient category of the homotopy category of acyclic complexes of quasicoherent sheaves over X by the thick subcategory of coacyclic complexes, or as the full subcategory of acyclic complexes in the coderived category $D^{co}(X$ -qcoh) of (complexes of) quasicoherent sheaves over X. It is the latter definition that will be used in the sequel.

Clearly, the category $D_{Sing}^{st}(X)$ has arbitrary infinite direct sums. Krause [2005, Corollary 5.4] constructs a fully faithful functor $D_{Sing}^{b}(X) \rightarrow D_{Sing}^{st}(X)$ and proves that its image is a set of compact generators of the target category.

Theorem 2.9. For any separated Noetherian scheme Z with enough vector bundles, there is a natural triangulated functor $D'_{Sing}(Z) \rightarrow D^{st}_{Sing}(Z)$ forming a commutative diagram with the natural functors from $D^{b}_{Sing}(Z)$ into both these categories. The composition

$$\mathsf{D}^{\mathsf{b}}(Z\operatorname{-\mathsf{qcoh}})\longrightarrow \mathsf{D}'_{Sing}(Z)\longrightarrow \mathsf{D}^{\mathsf{st}}_{Sing}(Z)$$

preserves those infinite direct sums that exist in $D^{b}(Z\operatorname{-qcoh})$. When $Z = X_{0}$ is a divisor in a regular separated Noetherian scheme of finite Krull dimension, the functor $D'_{Sing}(X_{0}) \rightarrow D^{st}_{Sing}(X_{0})$ is an equivalence of triangulated categories.

Proof. The construction of the functor $D_{Sing}^{b}(Z) \rightarrow D_{Sing}^{st}(Z)$ in [Krause 2005] is given in terms of the Verdier localization functor $Q : D^{co}(Z-qcoh) \rightarrow D(Z-qcoh)$ by the triangulated subcategory $D_{Sing}^{st}(Z) \subset D^{co}(Z-qcoh)$ and its adjoint functors on both sides, which exist according to [loc. cit., Corollary 4.3]. The proof of our theorem is based on explicit constructions of the restrictions of these adjoint functors to some subcategories of bounded complexes in D(Z-qcoh).

It is well known that the Verdier localization functor $H^0(Z\operatorname{-qcoh}) \to D(Z\operatorname{-qcoh})$ from the homotopy category of (complexes of) quasicoherent sheaves on Z to their derived category has a right adjoint functor $D(Z\operatorname{-qcoh}) \to H^0(Z\operatorname{-qcoh})$. The objects in the image of this functor are called *homotopy injective complexes* of quasicoherent sheaves on Z. The composition $D(Z\operatorname{-qcoh}) \to H^0(Z\operatorname{-qcoh}) \to D^{co}(Z\operatorname{-qcoh})$ provides the functor $Q_\rho : D(Z\operatorname{-qcoh}) \to D^{co}(Z\operatorname{-qcoh})$ right adjoint to Q. In particular, any bounded-below complex in $D(Z\operatorname{-qcoh})$ has a bounded-below injective resolution and any bounded-below complex of injectives is homotopy injective. Furthermore, any bounded-below acyclic complex is coacyclic [Positselski 2010, Lemma 2.1]. It follows that any bounded-below complex from $D^+(Z\operatorname{-qcoh})$, considered as an object of $D^{co}(Z\operatorname{-qcoh})$, represents its own image under the functor Q_ρ .

On the other hand, any bounded-above complex from D(Z-qcoh) has a locally free left resolution defined uniquely up to a quasi-isomorphism of complexes in the exact category of locally free sheaves; i.e., there is an equivalence of bounded above derived categories $D^{-}(Z-qcoh_{lf}) \simeq D^{-}(Z-qcoh)$. Since the exact category $Z-qcoh_{lf}$ has finite homological dimension, any acyclic complex in it is coacyclic (and even absolutely acyclic [loc. cit., Remark 2.1]), so there are natural functors $D^{-}(Z\operatorname{-qcoh}_{lf}) \rightarrow D(Z\operatorname{-qcoh}_{lf}) \simeq D^{co}(Z\operatorname{-qcoh}_{lf}) \rightarrow D^{co}(Z\operatorname{-qcoh}).$

Lemma 2.9. The composition of the embedding $D^-(Z\operatorname{-qcoh}) \to D(Z\operatorname{-qcoh})$ with the functor $Q_{\lambda} : D(Z\operatorname{-qcoh}) \to D^{\operatorname{co}}(Z\operatorname{-qcoh})$ left adjoint to Q is isomorphic to the functor $D^-(Z\operatorname{-qcoh}) \to D^{\operatorname{co}}(Z\operatorname{-qcoh})$ constructed above.

Proof. We have to show that $\operatorname{Hom}_{\mathsf{D}^{co}(Z\operatorname{-qcoh})}(\mathcal{L}^{\bullet}, \mathcal{E}^{\bullet}) = 0$ for any bounded-above complex of locally free sheaves \mathcal{L}^{\bullet} and any acyclic complex \mathcal{E}^{\bullet} of quasicoherent sheaves on Z. Let us check that any morphism $\mathcal{L}^{\bullet} \to \mathcal{E}^{\bullet}$ in $H^0(Z\operatorname{-qcoh})$ factorizes through a coacyclic complex of quasicoherent sheaves. Clearly, we can assume that the complex \mathcal{E}^{\bullet} is bounded above, too. Let \mathcal{K}^{\bullet} be the cocone of a closed morphism of complexes $\mathcal{L}^{\bullet} \to \mathcal{E}^{\bullet}$; then \mathcal{K}^{\bullet} is bounded above and the composition $\mathcal{K}^{\bullet} \to \mathcal{L}^{\bullet} \to \mathcal{E}^{\bullet}$ is homotopic to zero. Pick a bounded-above complex of locally free sheaves \mathcal{F}^{\bullet} together with a quasi-isomorphism $\mathcal{F}^{\bullet} \to \mathcal{K}^{\bullet}$. Then the cone of the composition $\mathcal{F}^{\bullet} \to \mathcal{K}^{\bullet} \to \mathcal{L}^{\bullet}$, being a bounded-above acyclic complex of locally free sheaves, is coacyclic. Since the composition $\mathcal{F}^{\bullet} \to \mathcal{L}^{\bullet} \to \mathcal{E}^{\bullet}$ is homotopic to zero, up to homotopy, through this cone.

Now we can describe the action of the functor $I_{\lambda} : D^{co}(Z \operatorname{-qcoh}) \to D^{st}_{Sing}(Z \operatorname{-qcoh})$ left adjoint to the embedding $D^{st}_{Sing}(Z \operatorname{-qcoh}) \to D^{co}(Z \operatorname{-qcoh})$ on bounded-above complexes in $D^{co}(Z \operatorname{-qcoh})$. If \mathcal{K}^{\bullet} is a bounded-above complex of quasicoherent sheaves and \mathcal{F}^{\bullet} is its locally free left resolution, then the cone of the closed morphism $\mathcal{F}^{\bullet} \to \mathcal{K}^{\bullet}$ represents the object $I_{\lambda}(\mathcal{K}^{\bullet}) \in D^{st}_{Sing}(Z \operatorname{-qcoh})$. In view of Lemma 2.9, this cone is functorial and does not depend on the choice of \mathcal{F}^{\bullet} for the usual semiorthogonality reasons.

The embedding of compact generators $D_{Sing}^{b}(Z) \rightarrow D_{Sing}^{st}(Z)$ is constructed in [Krause 2005] as the functor induced by the restriction of the composition $I_{\lambda} \circ Q_{\rho} : D(Z \operatorname{-qcoh}) \rightarrow D_{Sing}^{st}(Z)$ to the full subcategory $D^{b}(Z\operatorname{-coh}) \subset D(Z\operatorname{-qcoh})$. Let us explain why this is so. By Proposition 1.5(d) (cf. [loc. cit., Proposition 2.3 and Remark 3.8]), the natural functor $D^{b}(Z\operatorname{-coh}) \rightarrow D^{co}(Z\operatorname{-qcoh})$ is fully faithful and its image is a set of compact generators in the target. This is the image of $D^{b}(Z\operatorname{-coh}) \subset D(Z\operatorname{-qcoh})$ under the functor Q_{ρ} , as constructed above. It is clear from the above construction of the functor Q_{λ} that it preserves compactness (and in fact coincides with the functor Q_{λ} and I_{λ} , being left adjoints, preserve infinite direct sums, and I_{λ} is a Verdier localization functor by the image of Q_{λ} , it follows that the image of any set of compact generators of $D^{co}(Z\operatorname{-qcoh})$ under I_{λ} is a set of compact generators of $D_{Sine}^{st}(Z)$ [Neeman 1996, Theorem 2.1(4)].

In order to define the desired functor

$$\mathsf{D}'_{Sing}(Z) \longrightarrow \mathsf{D}^{\mathsf{st}}_{Sing}(Z),$$

restrict the same composition $I_{\lambda} \circ Q_{\rho}$ to the full subcategory $D^{b}(Z\operatorname{-qcoh}) \subset D(Z\operatorname{-qcoh})$. According to the above, this restriction assigns to any bounded complex of quasicoherent sheaves \mathcal{K}^{\bullet} on Z the cone of a morphism $\mathcal{F}^{\bullet} \to \mathcal{K}^{\bullet}$ into it from its locally free left resolution \mathcal{F}^{\bullet} . Clearly, the functor

$$D^{b}(Z\operatorname{-qcoh}) \longrightarrow D^{st}_{Sing}(Z)$$

that we have constructed preserves those infinite direct sums that exist in $D^{b}(Z\operatorname{-qcoh})$ and annihilates the triangulated subcategory $D^{b}(Z\operatorname{-qcoh}_{lf}) \subset D^{b}(Z\operatorname{-qcoh})$. So we have the induced functor $D'_{Sing}(Z) \to D^{st}_{Sing}(Z)$, and the first two assertions of the theorem are proven.

To prove the last assertion, we use the results of Section 2.8. Assume that $Z = X_0$ is the zero locus of a section $w \in \mathcal{L}(X)$ of a line bundle on X; as usual, $w : \mathcal{O}_X \to \mathcal{L}$ has to be an injective morphism of sheaves. Then by Theorem 2.8 and Lemma 2.8, the category $D'_{Sing}(Z)$ admits infinite direct sums and the image of the fully faithful functor $D^{b}_{Sing}(X_0) \to D'_{Sing}(X_0)$ is a set of compact generators in the target. Furthermore, it follows from the proof of Theorem 2.8 that any object of $D'_{Sing}(X_0)$ can be represented by a quasicoherent sheaf on X_0 and the direct sum of an infinite family of such objects is represented by the direct sums of such sheaves (see Remark 2.8). Thus the functor $D'_{Sing}(Z) \to D^{st}_{Sing}(Z)$, being an infinite direct sum-preserving triangulated functor identifying triangulated subcategories of compact generators, is an equivalence of triangulated categories.

We keep the assumptions of Theorem 2.9 and the notation of the last paragraph of its proof; i.e., X is a regular separated Noetherian scheme of finite Krull dimension with enough vector bundles and $X_0 \subset X$ is the divisor of zeros of a locally nonzerodividing section $w \in \mathcal{L}(X)$. The closed embedding $X_0 \to X$ is denoted by *i*.

Corollary 2.9. The functor

$$\Lambda : \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w) \operatorname{-qcoh}_{\mathsf{lf}}) \simeq \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w) \operatorname{-qcoh}) \longrightarrow \mathsf{D}^{\mathsf{st}}_{Sing}(X_0)$$

assigning to a locally free (or just w-flat) quasicoherent matrix factorization \mathcal{M} the acyclic complex of locally free (or quasicoherent) sheaves $i^*\mathcal{M}$ on X_0 is an equivalence of triangulated categories.

Proof. Given a *w*-flat matrix factorization \mathcal{M} , the complex of sheaves $i^*\mathcal{M}$ on X_0 is acyclic by [Polishchuk and Vaintrob 2011, Lemma 1.5]. Clearly, the assignment $\mathcal{M} \longmapsto i^*\mathcal{M}$ defines a triangulated functor $\mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w) \operatorname{-qcoh}_{w-\mathrm{fl}}) \to \mathsf{D}^{\mathsf{st}}_{Sing}(X_0)$.

To prove that this functor is an equivalence of categories, it suffices to identify it, up to a shift, with the composition of the equivalences

$$\mathsf{D}^{\mathsf{co}}((X,\mathcal{L},w)\operatorname{-qcoh}_{\mathsf{lf}})\longrightarrow \mathsf{D}'_{Sing}(X_0)\longrightarrow \mathsf{D}^{\mathsf{st}}_{Sing}(X_0).$$

Here one simply notices that for any $\mathcal{M} \in D^{co}((X, \mathcal{L}, w)\operatorname{-qcoh}_{\mathsf{lf}})$ the complex $i^*\mathcal{M}$ is isomorphic in $D^{\mathsf{st}}_{Sing}(X_0)$ to its canonical truncation $\tau_{\leq 1}i^*\mathcal{M}$, and the latter complex is the cocone of the morphism into $\Sigma(\mathcal{M})$ from one of its left locally free resolutions. So the functor Λ is identified with $\Sigma[-1]$.

2.10. *Relative stable derived category.* The goal of this section is to generalize the results of the previous one to the case of a singular Noetherian scheme X. The relative version of stable derived category, defined for a closed embedding of finite flat dimension $i : Z \to X$, is equivalent to the categories $D'_{Sing}(X_0/X)$ and $D^{co}((X, \mathcal{L}, w)$ -qcoh) in the case of the Cartier divisor $Z = X_0$ corresponding to a locally nonzero-dividing section w of a line bundle \mathcal{L} on X.

Let X be a separated Noetherian scheme of finite Krull dimension and $i : Z \to X$ be a closed embedding of schemes such that $i_*\mathcal{O}_Z$ has finite flat dimension as an \mathcal{O}_X -module. According to Section 1.9, there is a left derived inverse image functor $\mathbb{L}i^* : D^{co}(X\operatorname{-qcoh}) \to D^{co}(Z\operatorname{-qcoh})$. This functor forms a commutative diagram with the similar functor $\mathbb{L}i^* : D(X\operatorname{-qcoh}) \to D(Z\operatorname{-qcoh})$, and consequently, takes acyclic complexes in $D^{co}(X\operatorname{-qcoh})$ to acyclic complexes in $D^{co}(Z\operatorname{-qcoh})$.

Proposition 2.10. *The following four triangulated categories are naturally equiva-lent:*

(a) the full subcategory in $D^{co}(Z\operatorname{-qcoh})$ consisting of all the objects annihilated by the direct image functor $i_* : D^{co}(Z\operatorname{-qcoh}) \to D^{co}(X\operatorname{-qcoh});$

(b) the quotient category of the homotopy category of complexes over Z-qcoh whose direct images are coacyclic complexes over X-qcoh by the thick subcategory of coacyclic complexes over Z-qcoh;

(c) the quotient category of $D^{co}(Z\operatorname{-qcoh})$ by its minimal triangulated subcategory, containing the objects in $\mathbb{L}i^*D^{co}(X\operatorname{-qcoh})$ and closed under infinite direct sums;

(d) the quotient category of the full subcategory of acyclic complexes in $D^{co}(Z \operatorname{-qcoh})$ by its minimal triangulated subcategory, containing the left derived inverse images of acyclic complexes in $D^{co}(X\operatorname{-qcoh})$ and closed under infinite direct sums.

Proof. The equivalence of (a) and (b) is obvious. To show that the natural functor from the category (d) to the category (c) is an equivalence, notice that the minimal triangulated subcategory containing flat quasicoherent sheaves and closed under infinite direct sums together with the triangulated subcategory of acyclic complexes form a semiorthogonal decomposition of $D^{co}(X-qcoh)$, and similarly for Z [Positselski 2012, Corollary A.4.7]. Since flat quasicoherent sheaves on Z belong to the thick subcategory in $D^{b}(Z-qcoh) \subset D^{co}(Z-qcoh)$ generated by the inverse images of flat quasicoherent sheaves from X (see the proof of Lemma 2.8), the assertion follows.

Finally, the functor $\mathbb{L}i^*$ preserves infinite direct sums and compactness of objects since its right adjoint functor i_* preserves infinite direct sums. Hence the minimal

triangulated subcategory in $D^{co}(Z-qcoh)$ containing $Li^*D^{co}(X-qcoh)$ and closed under infinite direct sums is compactly generated by some objects which are compact in $D^{co}(Z-qcoh)$. By Brown representability, the quotient category in (c) is equivalent to the right orthogonal complement to this triangulated subcategory, which is the kernel category in (a).

We call any of the equivalent triangulated categories in Proposition 2.10 the *relative stable derived category of* Z *over* X and denote it by $D_{Sing}^{st}(Z/X)$ (cf. [Becker 2014, Section 2]). In particular, defining the relative stable derived category by the construction (c), we have natural triangulated functors

$$D^{b}(Z\operatorname{-qcoh}) \longrightarrow D^{co}(Z\operatorname{-qcoh}) \longrightarrow D^{st}_{Sing}(Z/X).$$

Clearly, the composition $D^{b}(Z\operatorname{-qcoh}) \to D^{st}_{Sing}(Z/X)$ factorizes through the relative singularity category $D'_{Sing}(Z/X)$, providing a natural functor $D'_{Sing}(Z/X) \to D^{st}_{Sing}(Z/X)$.

Lemma 2.10. The composition of triangulated functors

$$\mathsf{D}^{\mathsf{b}}_{Sing}(Z/X) \longrightarrow \mathsf{D}'_{Sing}(Z/X) \longrightarrow \mathsf{D}^{\mathsf{st}}_{Sing}(Z/X)$$

is fully faithful and its image forms a set of compact generators for the triangulated category $D_{Sing}^{st}(Z/X)$.

Proof. By Proposition 1.5(d), the full triangulated subcategory $D^{abs}(Z\text{-coh})$ compactly generates the triangulated category $D^{co}(Z\text{-qcoh})$, and similarly this holds for X. In view of the construction (c) and the argument in the proof of Proposition 2.10, the assertion follows from [Neeman 1992, Theorem 2.1].

Now let \mathcal{L} be a line bundle on X, let $w \in \mathcal{L}(X)$ be a locally nonzero-dividing section of \mathcal{L} , and let $i : X_0 \to X$ be closed embedding of the zero locus of w. Defining the category $\mathsf{D}^{\mathsf{st}}_{Sing}(X_0/X)$ by the construction (d), let $\mathbb{L}\Lambda : \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w)\operatorname{-qcoh}) \to \mathsf{D}^{\mathsf{st}}_{Sing}(X_0/X)$ be the triangulated functor assigning to a w-flat quasicoherent matrix factorization \mathcal{M} the acyclic complex $i^*\mathcal{M}$ over $X_0\operatorname{-qcoh}$.

Since any bounded-below acyclic complex over X_0 -qcoh is coacyclic, and any bounded-above complex belongs to the minimal triangulated subcategory in $D^{co}(X_0$ -qcoh) generated by its terms and closed under infinite direct sums, the following diagram of triangulated functors is commutative (cf. Corollary 2.9):



Theorem 2.10. For any locally nonzero-dividing section w of a line bundle \mathcal{L} on a separated Noetherian scheme X of finite Krull dimension, all the three functors on the above diagram are equivalences of triangulated categories.

Proof. The functor $\mathbb{L}\Xi$ is an equivalence by Theorem 2.8. To show that the functor $\mathbb{L}\Lambda$ is an equivalence, let us check that it identifies compact generators. By Proposition 1.5(d), the category $\mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w)\operatorname{-qcoh})$ is compactly generated by its full triangulated subcategory $\mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w)\operatorname{-coh})$, while according to Lemma 2.10 the category $\mathsf{D}^{\mathsf{st}}_{Sing}(X_0/X)$ is compactly generated by its full triangulated subcategory $\mathsf{D}^{\mathsf{st}}_{Sing}(X_0/X)$. The restriction of the functor $\mathbb{L}\Lambda$ being an equivalence between these two subcategories (in view of commutativity of the diagram and) by Theorem 2.7, it follows that the functor $\mathbb{L}\Lambda$ itself is an equivalence, too.

Remark 2.10. Another proof of Theorem 2.10 can be obtained using the approach based on [Efimov 2013, Appendix A]. In the notation and assumptions of Remark 2.7, suppose that C is an abelian category with exact functors of arbitrary infinite direct sums. Then so is the abelian category MF(C, L, w); the full abelian subcategory $C_0 \subset C$ is closed under infinite direct sums; and the triangulated functors $i_*, \mathbb{L}i^*, v^n, \mathbb{L}\xi^n, F^n, G^{n-}$ act between the coderived categories $D^{co}(C), D^{co}(C_0)$, and $D^{co}MF(C, L, w)$.

As in Remark 2.7, one proves that the functors $G^{n-}: D^{co}(C) \to D^{co}MF(C, L, w)$ and $v^n: D^{co}(C_0) \to D^{co}MF(C, L, w)$ are fully faithful and their images form a semiorthogonal decomposition of the coderived category $D^{co}MF(C, L, w)$. By (the proof of) [loc. cit., Proposition A.3(3) and (4)], the totalization functor

$$D^{co}MF(C, L, w) \longrightarrow D^{co}(C, L, w)$$

acting between the coderived category of the abelian category MF(C, L, w) and the coderived category of matrix factorizations $D^{co}(C, L, w)$ (defined as in Section 1.3) is the Verdier localization by the minimal triangulated subcategory containing the objects $G^{n-}(B)$ and $G^{(n+1)-}(B)$ for all $B \in C$ and closed under infinite direct sums.

It follows that the composition of the functor $v^n : D^{co}(C_0) \to D^{co}MF(C, L, w)$ with the totalization functor $D^{co}MF(C, L, w) \to D^{co}(C, L, w)$ induces an equivalence of triangulated categories

$$\mathsf{D}^{\mathsf{co}}(\mathsf{C}_0)/\langle \mathbb{L}i^*\mathsf{D}^{\mathsf{co}}(\mathsf{C})\rangle_{\oplus} \longrightarrow \mathsf{D}^{\mathsf{co}}(\mathsf{C},L,w)$$

between the quotient category of the coderived category $D^{co}(C_0)$ by its minimal triangulated subcategory containing the image of the functor $\mathbb{L}i^*: D^{co}(C) \to D^{co}(C_0)$ and closed under infinite direct sums, and the coderived category of matrix factorizations. The composition of the functor $\mathbb{L}\xi^n : D^{co}(C_0) \to D^{co}(C_0)$ with the Verdier localization functor $D^{co}(C_0) \to D^{co}(C_0)/\langle \mathbb{L}i^*D^{co}(C) \rangle_{\oplus}$ factorizes

through the totalization functor, providing the inverse equivalence $D^{co}(C, L, w) \rightarrow D^{co}(C_0)/\langle \mathbb{L}i^*D^{co}(C)\rangle_{\oplus}$.

Returning to quasicoherent matrix factorizations of a global section $w \in \mathcal{L}(X)$ of a line bundle \mathcal{L} on a separated Noetherian scheme X with the zero locus $X_0 \subset X$, we obtain direct constructions of two mutually inverse triangulated equivalences between the coderived category $D^{co}((X, \mathcal{L}, w)$ -qcoh) and the relative stable derived category $D^{st}_{Sinv}(X_0/X)$ as defined in part (c) of Proposition 2.10.

3. Supports, pull-backs, and push-forwards

3.1. *Supports.* This section paves the ground for the results about preservation of finite rank or coherence by the push-forwards of matrix factorizations with proper supports, which will be proven in Sections 3.5–3.6.

Let X be a separated Noetherian scheme and $T \subset X$ be a Zariski closed subset. Denote by X-coh_T the abelian category of coherent sheaves on X with set-theoretic support in T; and we will use similar notation for quasicoherent sheaves.

It is a well-known fact (essentially, a reformulation of the Artin–Rees lemma) that the embedding of abelian categories X-qcoh_T \rightarrow X-qcoh takes injectives to injectives. It follows that the functor $D^{b}(X$ -coh_T) \rightarrow $D^{b}(X$ -coh) is fully faithful. Clearly, its image is a thick subcategory and the corresponding quotient category can be naturally identified with $D^{b}(U$ -coh), where $U = X \setminus T$ (cf. Section 1.10).

Assume additionally that X has enough vector bundles. Let $Perf_T(X) \subset Perf(X)$ denote the full subcategory of perfect complexes with the cohomology sheaves settheoretically supported in T. By the above result, $Perf_T(X)$ can be considered as a thick subcategory in $D^b(X-\operatorname{coh}_T)$. According to [Orlov 2011, Lemma 2.6], the functor $D^b(X-\operatorname{coh}_T)/Perf_T(X) \to D^b_{Sing}(X)$ induced by the embedding $D^b(X-\operatorname{coh}_T) \to$ $D^b(X-\operatorname{coh})$ is fully faithful. We denote the source (or the image) category of this functor by $D^b_{Sing}(X, T)$.

By [Chen 2010, Theorem 1.3], the restriction functor $D_{Sing}^{b}(X) \rightarrow D_{Sing}^{b}(U)$ is the Verdier localization functor by the triangulated subcategory $D_{Sing}^{b}(X, T)$. In particular, the kernel of the restriction functor coincides with the thick envelope of (i.e., the minimal thick subcategory containing) $D_{Sing}^{b}(X, T)$ in $D_{Sing}^{b}(X)$.

Now we are going to establish the similar results for the triangulated categories of relative singularities. Let $i : Z \to X$ be a closed subscheme such that $i_*\mathcal{O}_Z \in Perf(X)$, and let $Perf(Z/X) = D^b(\mathbb{E}_{Z/X})$ (see Remark 2.1) denote the thick subcategory in $D^b(Z$ -coh) generated by $\mathbb{L}i^*D^b(X$ -coh). Let $T \subset Z$ be a Zariski closed subset; put $U = X \setminus T$ and $V = Z \setminus T$. We denote by $Perf_T(Z/X)$ the full subcategory of all objects of Perf(Z/X) with the cohomology sheaves set-theoretically supported in T. Consider it as a thick subcategory in $D^b(Z$ -coh $_T$), and denote by $D^b_{Sing}(Z/X, T)$ the quotient category $D^b(Z$ -coh $_T)/Perf_T(Z/X)$.

- **Lemma 3.1.** (a) The functor $D^{b}_{Sing}(Z/X, T) \rightarrow D^{b}_{Sing}(Z/X)$ induced by the embedding $D^{b}(Z\operatorname{-coh}_{T}) \rightarrow D^{b}(Z\operatorname{-coh})$ is fully faithful.
- (b) The restriction functor D^b_{Sing}(Z/X) → D^b_{Sing}(V/U) is the Verdier localization functor by the triangulated subcategory D^b_{Sing}(Z/X, T). In particular, the kernel of the restriction functor coincides with the thick envelope of D^b_{Sing}(Z/X, T) in D^b_{Sing}(Z/X).

Proof. The proof of (a) is similar to that of [Orlov 2011, Lemma 2.6]. One only needs to notice that the tensor product of an object of Perf(Z/X) with an object of Perf(Z) belongs to Perf(Z/X). This follows from the fact that Perf(Z) as a thick subcategory in $D^{b}(Z$ -coh) is generated by the restrictions of vector bundles from X (see Section 2.1). Part (b) is true since the thick subcategory $Perf(V/U) \subset D^{b}(V$ -coh) is generated by the restriction functor $Perf(Z/X) \rightarrow Perf(V/U)$, which is true because any coherent sheaf on U can be extended to a coherent sheaf on X. \Box

Let \mathcal{L} be a line bundle over X and $w \in \mathcal{L}(X)$ be a section; set $X_0 = \{w = 0\} \subset X$. The definitions of the set-theoretic and category-theoretic supports Supp \mathcal{M} and supp \mathcal{M} of a coherent matrix factorization $\mathcal{M} \in (X, \mathcal{L}, w)$ -coh were given (in a greater generality of coherent CDG-modules) in Section 1.10.

Given a locally free matrix factorization of finite rank $\mathcal{M} \in (X, \mathcal{L}, w)$ -coh_{lf}, define the *(category-theoretic) support* supp $\mathcal{M} \subset X$ as the minimal closed subset $T \subset X$ such that the restriction $\mathcal{M}|_U$ of \mathcal{M} to the open subscheme $U = X \setminus T$ is absolutely acyclic with respect to $(U, \mathcal{L}|_U, w|_U)$ -coh_{lf}. By Corollary 2.3(i), the definitions of category-theoretic supports of coherent matrix factorizations and of locally free matrix factorizations of finite rank agree when they are both applicable.

Equivalently, for a locally free matrix factorization \mathcal{M} of finite rank over X, the open subscheme $X \setminus \text{supp } \mathcal{M}$ is the union of all affine open subschemes $U \subset X$ such that the matrix factorization $\mathcal{M}|_U$ is contractible (see Remark 1.3). For any coherent matrix factorization \mathcal{M} , one has supp $\mathcal{M} \subset X_0$ since any matrix factorization of an invertible potential is contractible (cf. [Polishchuk and Vaintrob 2011, Section 5]).

Let $T \subset X$ be a closed subset. Denote by $D_T^{abs}((X, \mathcal{L}, w)\operatorname{-coh}_{\mathsf{lf}})$ (respectively $D_T^{abs}((X, \mathcal{L}, w)\operatorname{-coh})$) the quotient category of the homotopy category of locally free matrix factorizations of finite rank (resp. coherent matrix factorizations) supported category-theoretically inside T by the thick subcategory of matrix factorizations absolutely acyclic with respect to $(X, \mathcal{L}, w)\operatorname{-coh}_{\mathsf{lf}}$ (resp. $(X, \mathcal{L}, w)\operatorname{-coh})$. Clearly, the functors $D_T^{abs}((X, \mathcal{L}, w)\operatorname{-coh}_{\mathsf{lf}}) \to D^{abs}((X, \mathcal{L}, w)\operatorname{-coh}_{\mathsf{lf}})$ and $D_T^{abs}((X, \mathcal{L}, w)\operatorname{-coh}) \to D^{abs}((X, \mathcal{L}, w)\operatorname{-coh})$ are fully faithful [loc. cit.].

By the definition, the thick subcategories

$$\begin{split} \mathsf{D}_{T}^{\mathsf{abs}}((X,\mathcal{L},w)\text{-}\mathsf{coh}_{\mathsf{lf}}) \subset \mathsf{D}^{\mathsf{abs}}((X,\mathcal{L},w)\text{-}\mathsf{coh}_{\mathsf{lf}}),\\ \mathsf{D}_{T}^{\mathsf{abs}}((X,\mathcal{L},w)\text{-}\mathsf{coh}) \subset \mathsf{D}^{\mathsf{abs}}((X,\mathcal{L},w)\text{-}\mathsf{coh}) \end{split}$$

only depend on the intersection $X_0 \cap T$ (rather than the whole of T). Equivalently, they can be defined as the full subcategories of objects annihilated by the restriction functors $D^{abs}((X, \mathcal{L}, w)\text{-coh}_{lf}) \to D^{abs}((U, \mathcal{L}|_U, w|_U)\text{-coh}_{lf})$ and $D^{abs}((X, \mathcal{L}, w)\text{-coh}) \to D^{abs}((U, \mathcal{L}|_U, w|_U)\text{-coh})$, where $U = X \setminus T$.

As in Section 1.10, we denote by $D^{abs}((X, \mathcal{L}, w) \operatorname{-coh}_T)$ the absolute derived category of coherent matrix factorizations with set-theoretic support in T. The functor $D^{abs}((X, \mathcal{L}, w) \operatorname{-coh}_T) \to D^{abs}((X, \mathcal{L}, w) \operatorname{-coh})$ is fully faithful by Proposition 1.10(d). By Corollary 1.10(b), the full subcategory

$$\mathsf{D}_T^{\mathsf{abs}}((X,\mathcal{L},w)\operatorname{-coh}) \subset \mathsf{D}^{\mathsf{abs}}((X,\mathcal{L},w)\operatorname{-coh})$$

is the thick envelope of the full subcategory $D^{abs}((X, \mathcal{L}, w)\text{-coh}_T)$.

Now assume that $w : \mathcal{O}_X \to \mathcal{L}$ is an injective morphism of sheaves.

Proposition 3.1. (a) The equivalence of categories

$$\mathsf{D}^{\mathsf{abs}}((X,\mathcal{L},w)\operatorname{-coh})\simeq\mathsf{D}^{\mathsf{b}}_{Sing}(X_0/X)$$

identifies the triangulated subcategory $D^{abs}((X, \mathcal{L}, w)\text{-coh}_T)$ with the triangulated subcategory $D^{b}_{Sing}(X_0/X, X_0 \cap T)$. In particular, the former triangulated subcategory only depends on the intersection $X_0 \cap T$.

(b) The full preimage of the thick envelope of the triangulated subcategory

$$\mathsf{D}^{\mathsf{b}}_{Sing}(X_0, X_0 \cap T) \subset \mathsf{D}^{\mathsf{b}}_{Sing}(X_0)$$

under the fully faithful functor Σ : $D^{abs}((X, \mathcal{L}, w)$ -coh_{lf}) $\rightarrow D^{b}_{Sing}(X_0)$ coincides with the triangulated subcategory $D^{abs}_T((X, \mathcal{L}, w)$ -coh_{lf}).

Proof. Part (b) follows from the fact that the thick envelope of $D^{b}_{Sing}(X_0, X_0 \cap T)$ is the kernel of the restriction functor $D^{b}_{Sing}(X_0) \to D^{b}_{Sing}(X_0 \setminus T)$, the similar fact for $D^{abs}_{T}((X, \mathcal{L}, w)\text{-coh}_{lf})$, and the compatibility of the functors Σ with the restrictions to open subschemes, together with their full-and-faithfulness.

To prove part (a), notice first that the functor Υ obviously takes $D_{Sing}^{b}(X_0/X, X_0 \cap T)$ into $D^{abs}((X, \mathcal{L}, w) \operatorname{-coh}_T)$. Let us check that the functor $\mathbb{L}\Xi$ takes $D^{abs}((X, \mathcal{L}, w) \operatorname{-coh}_T)$ into $D_{Sing}^{b}(X_0/X, X_0 \cap T)$. Let \mathcal{M} be a coherent matrix factorization supported set-theoretically in T. Present \mathcal{M} as the cokernel of an injective morphism of w-flat coherent matrix factorizations $\mathcal{K} \to \mathcal{N}$. Since the functor $\mathbb{L}\Xi$ is triangulated, the object $\mathbb{L}\Xi(\mathcal{M}) \in D_{Sing}^{b}(X_0/X)$ is isomorphic to the cone of the morphism $\Xi(\mathcal{K}) \to \Xi(\mathcal{N})$ (cf. Lemma 3.6). The morphism $\Xi(\mathcal{K}) \to \Xi(\mathcal{N})$ of coherent sheaves on X_0 is an isomorphism outside T, so its kernel and cokernel are supported in $X_0 \cap T$. Thus the cone is quasi-isomorphic to a two-term complex of coherent sheaves on X_0 with the terms supported set-theoretically in $X_0 \cap T$. \Box

3.2. Locality of local freeness. The aim of this section is to show that the property of an object of $D^{abs}((X, \mathcal{L}, w)\operatorname{-qcoh}_{fl})$ or $D^{abs}((X, \mathcal{L}, w)\operatorname{-coh})$ to be a direct summand of an object from $D^{abs}((X, \mathcal{L}, w)\operatorname{-coh}_{lf})$ is local in a separated Noetherian scheme X with a dualizing complex and enough vector bundles, assuming that the potential $w \in \mathcal{L}(X)$ is not locally zero-dividing.

Let Z be a Noetherian scheme of finite Krull dimension with enough vector bundles. Recall that the natural functor $D^{b}_{Sing}(Z) \rightarrow D'_{Sing}(Z)$ is fully faithful [Orlov 2004, Proposition 1.13] (cf. Section 2.8).

Proposition 3.2. Let $Z = U \cup V$ be a covering by two open subschemes. Then any object of $\mathsf{D}'_{Sing}(Z)$ whose restrictions to U and V belong to the full subcategories $\mathsf{D}^{\mathsf{b}}_{Sing}(U) \subset \mathsf{D}'_{Sing}(U)$ and $\mathsf{D}^{\mathsf{b}}_{Sing}(V) \subset \mathsf{D}'_{Sing}(V)$, respectively, is a direct summand of an object belonging to the full subcategory $\mathsf{D}^{\mathsf{b}}_{Sing}(Z) \subset \mathsf{D}'_{Sing}(Z)$.

Proof. Consider the bounded derived category of quasicoherent sheaves $D^{b}(Z\operatorname{-qcoh})$ on Z and two full triangulated subcategories $D^{b}(Z\operatorname{-coh})$ and $D^{b}(Z\operatorname{-qcoh}_{fl})$ in it. Clearly, the intersection $D^{b}(Z\operatorname{-coh}) \cap D^{b}(Z\operatorname{-qcoh}_{fl})$ coincides with the full subcategory of perfect complexes $Perf(Z) = D^{b}(Z\operatorname{-coh}_{lf}) \subset D^{b}(Z\operatorname{-qcoh})$.

Lemma 3.2. Any morphism from an object of the full subcategory $D^{b}(Z\operatorname{-qcoh}_{fl})$ into an object of the full subcategory $D^{b}(Z\operatorname{-coh}) \subset D^{b}(Z\operatorname{-qcoh})$ factorizes through an object belonging to $D^{b}(Z\operatorname{-coh}_{lf})$.

Proof. See the proof of [Orlov 2004, Proposition 1.13].

It follows from Lemma 3.2 (by the way of the octahedron axiom) that any object \mathcal{K}^{\bullet} of the full triangulated subcategory $D^{b}(Z\operatorname{-qcoh}_{fl-c}$ generated by $D^{b}(Z\operatorname{-qcoh}_{fl})$ and $D^{b}(Z\operatorname{-coh})$ in $D^{b}(Z\operatorname{-qcoh})$ can be included in a distinguished triangle

$$\mathcal{F}^{\bullet} \longrightarrow \mathcal{K}^{\bullet} \longrightarrow \mathcal{M}^{\bullet} \longrightarrow \mathcal{F}^{\bullet}[1],$$

with $\mathcal{F}^{\bullet} \in D^{b}(Z\operatorname{-qcoh}_{fl})$ and $\mathcal{M}^{\bullet} \in D^{b}(Z\operatorname{-coh})$. Besides, the natural functor $D^{b}(Z\operatorname{-qcoh}_{fl})/D^{b}(Z\operatorname{-coh}_{lf}) \to D^{b}(Z\operatorname{-qcoh})/D^{b}(Z\operatorname{-coh})$ is fully faithful.

To prove Proposition 3.2, one has to show that any object $\mathcal{K}^{\bullet} \in \mathsf{D}^{\mathsf{b}}(Z\operatorname{-qcoh})$ whose restrictions to U and V belong to the subcategories $\mathsf{D}^{\mathsf{b}}(U\operatorname{-qcoh})_{\mathsf{fl-c}}$ and $\mathsf{D}^{\mathsf{b}}(V\operatorname{-qcoh}_{\mathsf{fl-c}})$, respectively, is a direct summand of an object from $\mathsf{D}^{\mathsf{b}}(Z\operatorname{-qcoh})_{\mathsf{fl-c}} \subset$ $\mathsf{D}^{\mathsf{b}}(Z\operatorname{-qcoh})$. According to the above, there exist two objects $\mathcal{F}_{U}^{\bullet} \in \mathsf{D}^{\mathsf{b}}(U\operatorname{-qcoh}_{\mathsf{fl}})$ and $\mathcal{F}_{V}^{\bullet} \in \mathsf{D}^{\mathsf{b}}(V\operatorname{-qcoh}_{\mathsf{fl}})$ and two morphisms $\mathcal{F}_{U}^{\bullet} \to \mathcal{K}^{\bullet}|_{U}$ and $\mathcal{F}_{V}^{\bullet} \to \mathcal{K}^{\bullet}|_{V}$ whose cones belong to $\mathsf{D}^{\mathsf{b}}(U\operatorname{-coh})$ and $\mathsf{D}^{\mathsf{b}}(V\operatorname{-coh})$, respectively.

Set $W = U \cap V \subset Z$; then the restrictions of \mathcal{F}_U^{\bullet} and \mathcal{F}_V^{\bullet} to W are isomorphic in $D^{b}(W\operatorname{-qcoh})/D^{b}(W\operatorname{-coh})$, and consequently, in $D^{b}(W\operatorname{-qcoh}_{fl})/D^{b}(W\operatorname{-coh}_{lf})$, too. Notice that the category $Perf(W) = D^{b}(W\operatorname{-coh}_{lf})$ is idempotent complete, and therefore, a thick subcategory in $D^{b}(W\operatorname{-qcoh}_{fl})$. It follows that there exists a finite complex of flat quasicoherent sheaves \mathcal{F}_W^{\bullet} on W together with two morphisms

 $\mathcal{F}_U^{\bullet}|_W \to \mathcal{F}_W^{\bullet}$ and $\mathcal{F}_V^{\bullet}|_W \to \mathcal{F}_W^{\bullet}$ whose cones are perfect complexes. Denote the cocones of these morphisms by \mathcal{G}_W^{\bullet} and \mathcal{H}_W^{\bullet} .

For any object A of a triangulated category D, let us denote by 'A the object $A \oplus A[1]$. For any triangulated subcategory $C \subset D$, whenever an object $A \in D$ is a direct summand of an object from C, the object 'A belongs to C, as $A \oplus B \in C$ implies $A \oplus A[1] \in C$ in view of the distinguished triangle $A \oplus B \rightarrow A \oplus B \rightarrow A \oplus A[1] \rightarrow A[1] \oplus B[1]$ [Thomason 1997, Theorem 2.1].

By the Thomason–Trobaugh theorem [1990, Section 5], the objects ${}^{'}\mathcal{G}_{W}^{\bullet}$ and ${}^{'}\mathcal{H}_{W}^{\bullet}$ can be extended to perfect complexes on U and V, respectively. Moreover, these extensions $\mathcal{G}_{U}^{\bullet} \in \mathsf{D}^{\mathsf{b}}(U\text{-coh}_{\mathsf{lf}})$ and $\mathcal{H}_{V}^{\bullet} \in \mathsf{D}^{\mathsf{b}}(V\text{-coh}_{\mathsf{lf}})$ can be chosen in such a way that the morphisms ${}^{'}\mathcal{G}_{W}^{\bullet} \to {}^{'}\mathcal{F}_{U}^{\bullet}|_{W}$ and ${}^{'}\mathcal{H}_{W}^{\bullet} \to {}^{'}\mathcal{F}_{V}^{\bullet}|_{W}$ would be extendable to morphisms $\mathcal{G}_{U}^{\bullet} \to {}^{'}\mathcal{F}_{U}^{\bullet}$ and $\mathcal{H}_{V}^{\bullet} \to {}^{'}\mathcal{F}_{V}^{\bullet}$ [Neeman 1996, Theorem 2.1(4) and (5)].

Furthermore, the objects ${}^{\prime}\mathcal{G}_{U}^{\bullet}$ and ${}^{\prime}\mathcal{H}_{V}^{\bullet}$ can be extended to perfect complexes \mathcal{G}^{\bullet} and \mathcal{H}^{\bullet} on the whole scheme Z so that the compositions of morphisms

$${}^{\prime}\mathcal{G}_{U}^{\bullet} \longrightarrow {}^{\prime\prime}\mathcal{F}_{U}^{\bullet} \longrightarrow {}^{\prime\prime}\mathcal{K}^{\bullet}|_{U} \text{ and } {}^{\prime\prime}\mathcal{H}_{V}^{\bullet} \longrightarrow {}^{\prime\prime}\mathcal{F}_{V}^{\bullet} \longrightarrow {}^{\prime\prime}\mathcal{K}^{\bullet}|_{V}$$

would be extendable to morphisms $\mathcal{G}^{\bullet} \to {}^{"}\mathcal{K}^{\bullet}$ and $\mathcal{H}^{\bullet} \to {}^{"}\mathcal{K}^{\bullet}$. Denote by $\mathcal{K}_{(1)}^{\bullet}$ a cone of the morphism $\mathcal{G}^{\bullet} \oplus \mathcal{H}^{\bullet} \to {}^{"}\mathcal{K}^{\bullet}$, by $\mathcal{F}_{U,(1)}^{\bullet}$ a cone of the morphism $\mathcal{G}_{U}^{\bullet} \to {}^{"}\mathcal{F}_{U}^{\bullet}$, and by $\mathcal{F}_{V,(1)}^{\bullet}$ a cone of the morphism $\mathcal{H}_{V}^{\bullet} \to {}^{"}\mathcal{F}_{V}^{\bullet}$. We have come back to the original situation with an object $\mathcal{K}_{(1)}^{\bullet} \in \mathsf{D}^{\mathsf{b}}(Z\operatorname{-qcoh})$, two objects $\mathcal{F}_{U,(1)}^{\bullet} \in \mathsf{D}^{\mathsf{b}}(U\operatorname{-qcoh}_{\mathsf{fl}})$ and $\mathcal{F}_{V,(1)}^{\bullet} \in \mathsf{D}^{\mathsf{b}}(V\operatorname{-qcoh}_{\mathsf{fl}})$, and two morphisms

$$\mathcal{F}^{\bullet}_{U,(1)} \longrightarrow \mathcal{K}^{\bullet}_{(1)}|_{U} \text{ and } \mathcal{F}^{\bullet}_{V,(1)} \longrightarrow \mathcal{K}^{\bullet}_{(1)}|_{V}$$

whose cones belong to $D^{b}(U\text{-coh})$ and $D^{b}(V\text{-coh})$, respectively. In addition, the objects $\mathcal{F}_{U,(1)}^{\bullet}|_{W}$ and $\mathcal{F}_{V,(1)}^{\bullet}|_{W}$ are now isomorphic in $D^{b}(W\text{-qcoh}_{fl})$.

The construction does not guarantee commutativity of the diagram formed by the isomorphism

$$\mathcal{F}^{\bullet}_{U,(1)}|_{W} = \mathcal{F}^{\bullet}_{W,(1)} \simeq \mathcal{F}^{\bullet}_{V,(1)}|_{W}$$

and the restrictions of the morphisms $\mathcal{F}^{\bullet}_{U,(1)} \to \mathcal{K}^{\bullet}_{(1)}$ and $\mathcal{F}^{\bullet}_{V,(1)} \to \mathcal{K}^{\bullet}_{(1)}$ to W. However, the original choice of the morphisms

$$\mathcal{F}_U^{\bullet}|_W \longrightarrow \mathcal{F}_W^{\bullet} \quad \text{and} \quad \mathcal{F}_V^{\bullet}|_W \longrightarrow \mathcal{F}_W^{\bullet}$$

makes this diagram commute in the quotient category $D^{b}(W\operatorname{-qcoh})/D^{b}(W\operatorname{-coh})$. Hence the difference of two morphisms $\mathcal{F}_{W,(1)}^{\bullet} \rightrightarrows \mathcal{K}_{(1)}^{\bullet}|_{W}$ factorizes through a bounded complex of coherent sheaves on W, and consequently (according to Lemma 3.2) also through a perfect complex on W. Denote the latter by $\mathcal{E}^{\bullet} \in D^{b}(W\operatorname{-coh}_{\mathsf{lf}})$. Now let $j: U \to Z$, $k: V \to Z$, and $h: W \to Z$ denote the natural open embeddings. Consider the square diagram formed by the morphisms

$$\mathbb{R}j_*\mathcal{F}^{\bullet}_{U,(1)} \oplus \mathbb{R}k_*\mathcal{F}^{\bullet}_{V,(1)} \longrightarrow \mathbb{R}h_*\mathcal{F}^{\bullet}_{U,(1)}|_W,$$
$$\mathbb{R}j_*\mathcal{K}^{\bullet}_{(1)}|_U \oplus \mathbb{R}k_*\mathcal{K}^{\bullet}_{(1)}|_V \longrightarrow \mathbb{R}h_*\mathcal{K}^{\bullet}_{(1)}|_W.$$

According to the above, this diagram is not necessarily commutative; but it can be made commutative by adding the new direct summand $\mathbb{R}h_*\mathcal{E}^{\bullet}$ to the term $\mathbb{R}j_*\mathcal{K}^{\bullet}_{(1)}|_U \oplus \mathbb{R}k_*\mathcal{K}^{\bullet}_{(1)}|_V$ with the morphism $\mathbb{R}h_*\mathcal{E}^{\bullet} \to \mathbb{R}h_*\mathcal{K}^{\bullet}_{(1)}|_W$ induced by the morphism $\mathcal{E}^{\bullet} \to \mathcal{K}^{\bullet}_{(1)}|_W$ and the morphism $\mathbb{R}j_*\mathcal{F}^{\bullet}_{U,(1)} \oplus \mathbb{R}k_*\mathcal{F}^{\bullet}_{V,(1)}$ equal to zero on the first direct summand and induced by the morphism $\mathcal{F}^{\bullet}_{V,(1)}|_W \simeq \mathcal{F}^{\bullet}_{W,(1)} \to \mathcal{E}^{\bullet}$ on the second one.

Let \mathcal{F}^{\bullet} denote a cocone of the morphism

$$\mathbb{R}j_*\mathcal{F}^{\bullet}_{U,(1)} \oplus \mathbb{R}k_*\mathcal{F}^{\bullet}_{V,(1)} \longrightarrow \mathbb{R}h_*\mathcal{F}^{\bullet}_{U,(1)}|_W$$

and \mathcal{L}^{\bullet} denote a cocone of the morphism

$$\mathbb{R}j_*\mathcal{K}^{\bullet}_{(1)}|_U \oplus \mathbb{R}k_*\mathcal{K}^{\bullet}_{(1)}|_V \oplus \mathbb{R}h_*\mathcal{E}^{\bullet} \longrightarrow \mathbb{R}h_*\mathcal{K}^{\bullet}_{(1)}|_W.$$

Then the commutative square can be extended to a morphism of distinguished triangles, so we obtain a morphism $\mathcal{F}^{\bullet} \to \mathcal{L}^{\bullet}$. Since $\mathcal{K}^{\bullet}_{(1)}$ is a cocone of the morphism

$$\mathbb{R}j_*\mathcal{K}^{\bullet}_{(1)}|_U \oplus \mathbb{R}k_*\mathcal{K}^{\bullet}_{(1)}|_V \longrightarrow \mathbb{R}h_*\mathcal{K}^{\bullet}_{(1)}|_W,$$

there is also a distinguished triangle $\mathcal{K}^{\bullet}_{(1)} \to \mathcal{L}^{\bullet} \to \mathbb{R}h_*\mathcal{E} \to \mathcal{K}^{\bullet}_{(1)}[1].$

Notice that the complexes \mathcal{F}^{\bullet} and $\mathbb{R}h_*\mathcal{E}^{\bullet}$ belong to $D^{b}(Z\operatorname{-qcoh}_{fl})$ (since the class of bounded complexes of flat quasicoherent sheaves is preserved by the derived direct images with respect to flat morphisms of Noetherian schemes; cf. Proposition 1.9). Furthermore, the complex $\mathbb{R}h_*\mathcal{E}^{\bullet}$ is perfect over W. Restricting to W our morphism of distinguished triangles, and recalling that cones of the morphisms

$$\mathcal{F}^{\bullet}_{U,(1)} \longrightarrow \mathcal{K}^{\bullet}_{(1)}|_{U} \text{ and } \mathcal{F}^{\bullet}_{V,(1)} \longrightarrow \mathcal{K}^{\bullet}_{(1)}|_{V}$$

are coherent complexes over U and V, one easily concludes that a cone of the morphism $\mathcal{F}^{\bullet} \to \mathcal{L}^{\bullet}$ is a coherent complex over W.

Denote this cone temporarily by $\mathcal{K}^{\bullet}_{(2)}$. Clearly, in order to show that the original complex \mathcal{K}^{\bullet} is a direct summand of an object from $D^{b}(Z\operatorname{-qcoh})_{\mathrm{fl-c}}$ in $D^{b}(Z\operatorname{-qcoh})$ (which is our goal), it suffices to check that the complex $\mathcal{K}^{\bullet}_{(2)}$ is as well. It also follows from the constructions that the restrictions of the complex $\mathcal{K}^{\bullet}_{(2)}$ to U and V belong to $D^{b}(U\operatorname{-qcoh})_{\mathrm{fl-c}}$ and $D^{b}(V\operatorname{-qcoh}_{\mathrm{fl-c}})$, respectively. Dropping the lower index and redenoting $\mathcal{K}^{\bullet}_{(2)}$ simply by \mathcal{K}^{\bullet} , we are coming back to the situation in the beginning of the proof with the new knowledge that \mathcal{K}^{\bullet} may be assumed to be a coherent complex over W.

The next segment of our proof is based on the localization theory for coderived categories of quasicoherent sheaves on Noetherian schemes (similar to the Thomason– Trobaugh–Neeman theorem for the conventional derived categories, the difference being that arbitrary bounded complexes of coherent sheaves play the role of perfect complexes). What we need is a particular case of the theory developed in Section 1.10 (corresponding to the choice of the quasicoherent CDG-algebra \mathcal{O}_Z over Z).

Specifically, it follows from Proposition 1.5(d) and Theorem 1.10 together with [Neeman 1996, Theorem 2.1(5)] that any morphism from an object of $D^{b}(W\text{-coh})$ into a restriction to W of an object \mathcal{K}^{\bullet} from $D^{b}(Z\text{-qcoh})$ (or even from $D^{co}(Z\text{-qcoh})$) can be extended to a morphism to \mathcal{K}^{\bullet} from an object of $D^{b}(Z\text{-coh})$. Applying this assertion to the identity morphism $\mathcal{K}^{\bullet}|_{W} \to \mathcal{K}^{\bullet}|_{W}$ in the above situation, we obtain a morphism $\mathcal{M}^{\bullet} \to \mathcal{K}^{\bullet}$ into \mathcal{K}^{\bullet} from a coherent complex \mathcal{M}^{\bullet} over Z that is a quasi-isomorphism over W. Passing to a cone of this morphism, we may assume \mathcal{K}^{\bullet} to be acyclic over W.

By Corollary 1.10, such a complex \mathcal{K}^{\bullet} is quasi-isomorphic to a (bounded) complex of quasicoherent sheaves on Z whose terms are concentrated set-theoretically in the complement $Z \setminus W$. The latter is a disjoint union of two nonintersecting closed subsets in Z, namely, the complements $S = Z \setminus U$ and $T = Z \setminus V$. Now the complex \mathcal{K}^{\bullet} decomposes into a direct sum of two complexes with set-theoretic supports inside S and T, respectively.

One can consider the two direct summands separately. We have to show that any bounded complex of quasicoherent sheaves \mathcal{K}^{\bullet} on Z, which is supported set-theoretically in T and whose restriction to U belongs to $D^{b}(U\operatorname{-qcoh})_{\mathrm{fl-c}}$, itself belongs to $D^{b}(Z\operatorname{-qcoh})_{\mathrm{fl-c}}$. Arguing as in the beginning of this proof, we have an object $\mathcal{G}^{\bullet} \in D^{b}(U\operatorname{-qcoh}_{\mathrm{fl}})$ together with a morphism $\mathcal{G}^{\bullet} \to \mathcal{K}^{\bullet}|_{U}$ whose cone belongs to $D^{b}(U\operatorname{-coh})$. The restriction $\mathcal{G}^{\bullet}|_{W}$ then belongs to both $D^{b}(W\operatorname{-qcoh}_{\mathrm{fl}})$ and $D^{b}(W\operatorname{-coh})$, and is, therefore, a perfect complex on W.

Again by the Thomason–Trobaugh theorem, the object ${}^{\mathcal{G}} {}^{\bullet}|_{W}$ can be extended to a perfect complex \mathcal{H}^{\bullet} on V. A cocone of the morphism

$$\mathbb{R}j_*'\mathcal{G}^{\bullet} \oplus \mathbb{R}k_*\mathcal{H}^{\bullet} \longrightarrow \mathbb{R}h_*'\mathcal{G}^{\bullet}|_W$$

provides an object $\mathcal{F}^{\bullet} \in D^{\mathsf{b}}(Z\operatorname{-qcoh}_{\mathsf{fl}})$ isomorphic to $'\mathcal{G}^{\bullet}$ over U and to \mathcal{H}^{\bullet} over V. Now the morphism $'\mathcal{G}^{\bullet} \to '\mathcal{K}^{\bullet}|_U$ over U extends uniquely to a morphism $\mathcal{F}^{\bullet} \to '\mathcal{K}^{\bullet}$ over Z since the set-theoretic support of $'\mathcal{K}^{\bullet}$ is contained in a closed subset lying inside U. A cone of the morphism $\mathcal{F}^{\bullet} \to '\mathcal{K}^{\bullet}$ is a coherent complex on Z since it is so in restrictions to U and V. Thus, the proposition is proven. \Box

Now let X be a separated Noetherian scheme of finite Krull dimension with enough vector bundles, \mathcal{L} be a line bundle on X, and $w \in \mathcal{L}(X)$ be a locally nonzero-dividing potential. Let $X_0 \subset X$ be the zero locus of w.

Corollary 3.2. Let $X = U \cap V$ be a covering by two open subschemes. Then any object of $D^{co}((X, \mathcal{L}, w)\operatorname{-qcoh}_{fl})$ whose restrictions to U and V belong to the full triangulated subcategories

$$\begin{split} \mathsf{D}^{\mathsf{abs}}((U,\mathcal{L}|_U,w|_U)\text{-}\mathsf{coh}_{\mathsf{lf}}) \subset \mathsf{D}^{\mathsf{co}}((U,\mathcal{L}|_U,w|_U)\text{-}\mathsf{qcoh}_{\mathsf{fl}}),\\ \mathsf{D}^{\mathsf{abs}}((V,\mathcal{L}|_V,w|_V)\text{-}\mathsf{coh}_{\mathsf{lf}}) \subset \mathsf{D}^{\mathsf{co}}((V,\mathcal{L}|_V,w|_V)\text{-}\mathsf{qcoh}_{\mathsf{fl}}), \end{split}$$

respectively, is a direct summand of an object from the full triangulated subcategory $D^{abs}((X, \mathcal{L}, w)\text{-coh}_{lf}) \subset D^{co}((X, \mathcal{L}, w)\text{-qcoh}_{fl}).$

Proof. By Proposition 2.8, the category $D^{co}((X, \mathcal{L}, w) \operatorname{-qcoh}_{fl})$ is a full triangulated subcategory of the triangulated category $D'_{Sing}(X_0)$. The (essential) intersection of the full subcategories $D^{co}((X, \mathcal{L}, w) \operatorname{-qcoh}_{fl})$ and $D^{b}_{Sing}(X_0)$ in $D'_{Sing}(X_0)$ is the triangulated category $D^{abs}((X, \mathcal{L}, w) \operatorname{-coh}_{lf})$.

Indeed, an object of $\mathcal{F} \in D^{b}_{Sing}(X_{0})$ belongs to $D^{abs}((X, \mathcal{L}, w)\operatorname{-coh}_{lf})$ if and only if the object $i_{\circ}\mathcal{F}$ vanishes in $D^{b}_{Sing}(X)$ (Theorem 2.7); an object $\mathcal{F} \in D'_{Sing}(X_{0})$ belongs to $D^{co}((X, \mathcal{L}, w)\operatorname{-qcoh}_{fl})$ if and only if the object $i_{\circ}\mathcal{F}$ vanishes in $D'_{Sing}(X)$ (Proposition 2.8); and the functor $D^{b}_{Sing}(X) \to D'_{Sing}(X)$ is fully faithful. Moreover, the (essential) intersection of $D^{co}((X, \mathcal{L}, w)\operatorname{-qcoh}_{fl})$ with the thick

Moreover, the (essential) intersection of $D^{co}((X, \mathcal{L}, w)-qcoh_{fl})$ with the thick envelope of $D^{b}_{Sing}(X_{0})$ in $D'_{Sing}(X_{0})$ is the thick envelope of $D^{abs}((X, \mathcal{L}, w)-coh_{lf})$ in $D'_{Sing}(X_{0})$. Indeed, let \mathcal{M} be an object of the intersection; then $\mathcal{M} \oplus \mathcal{M}[1]$ belongs to both $D^{co}((X, \mathcal{L}, w)-qcoh_{fl})$ and $D^{b}_{Sing}(X_{0})$, hence also to $D^{abs}((X, \mathcal{L}, w)-coh_{lf})$, and consequently \mathcal{M} belongs to the thick envelope of $D^{abs}((X, \mathcal{L}, w)-coh_{lf})$.

Now let \mathcal{K} be our object of $\mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w)\operatorname{-qcoh}_{\mathsf{fl}})$; it can be also viewed as an object of $\mathsf{D}'_{Sing}(X_0)$. If its restrictions to U and V belong to $\mathsf{D}^{\mathsf{abs}}((U, \mathcal{L}|_U, w|_U)\operatorname{-coh}_{\mathsf{lf}})$ and $\mathsf{D}^{\mathsf{abs}}((V, \mathcal{L}|_V, w|_V)\operatorname{-coh}_{\mathsf{lf}})$, they also belong to $\mathsf{D}^{\mathsf{b}}_{Sing}(U_0) \subset \mathsf{D}'_{Sing}(U_0)$ and $\mathsf{D}^{\mathsf{b}}_{Sing}(V_0) \subset \mathsf{D}'_{Sing}(V_0)$ (where we set $U_0 = U \cap X_0$ and $V_0 = V \cap X_0$).

Applying Proposition 3.2, we can conclude that \mathcal{K} belongs to the thick envelope of $D^{b}_{Sine}(X_0)$ in $D'_{Sine}(X_0)$. The assertion of Corollary 3.2 follows from the above. \Box

Assume additionally that the scheme X admits a dualizing complex $\mathcal{D}_{\mathbf{Y}}^{\bullet}$.

Theorem 3.2. Let $X = U \cap V$ be a covering by two open subschemes. Then any object of $D^{abs}((X, \mathcal{L}, w)\text{-coh})$ whose restrictions to U and V belong to the thick envelopes of the triangulated subcategories

$$D^{abs}((U, \mathcal{L}|_U, w|_U) \operatorname{-coh}_{\mathsf{lf}}) \subset D^{abs}((U, \mathcal{L}|_U, w|_U) \operatorname{-coh}),$$

$$D^{abs}((V, \mathcal{L}|_V, w|_V) \operatorname{-coh}_{\mathsf{lf}}) \subset D^{abs}((V, \mathcal{L}_V, w|_V) \operatorname{-coh})$$

itself belongs to the thick envelope of the triangulated subcategory

$$D^{abs}((X, \mathcal{L}, w)\text{-coh}_{lf}) \subset D^{abs}((X, \mathcal{L}, w)\text{-coh}).$$

Proof. The argument is based on the Serre–Grothendieck duality theory for matrix factorizations as developed in Section 2.5, which allows us to reduce the question to

the result of Corollary 3.2. Specifically, let \mathcal{M} be our coherent matrix factorization over X. Replacing, if necessary, \mathcal{M} with $\mathcal{M} \oplus \mathcal{M}[1]$, we may assume the restrictions of \mathcal{M} to U and V to be isomorphic to locally free matrix factorizations of finite rank.

Let us apply the construction of functor

$$\Omega: \mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w)\operatorname{-coh})^{\mathsf{op}} \longrightarrow \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, -w)\operatorname{-qcoh}_{\mathsf{fl}})$$

from Section 2.5 to the matrix factorization \mathcal{M} . That is, we pick a left resolution of \mathcal{M} by locally free matrix factorizations of finite rank, dualize by applying $\operatorname{Hom}_{X\operatorname{-qc}}(-, \mathcal{O}_X)$, and totalize using infinite direct sums. By Corollary 2.5, the functor Ω is fully faithful; it also identifies $\operatorname{D}^{\operatorname{abs}}((X, \mathcal{L}, w)\operatorname{-coh}_{|\mathrm{f}})^{\operatorname{op}}$ with $\operatorname{D}^{\operatorname{abs}}((X, \mathcal{L}, -w)\operatorname{-coh}_{|\mathrm{f}})$. Hence it suffices to check that the matrix factorization $\Omega(\mathcal{M})$ belongs to the thick envelope of $\operatorname{D}^{\operatorname{abs}}((X, \mathcal{L}, -w)\operatorname{-coh}_{|\mathrm{f}})$ in $\operatorname{D}^{\operatorname{co}}((X, \mathcal{L}, -w)\operatorname{-qcoh}_{|\mathrm{f}})$. But we know as much from Corollary 3.2.

3.3. *Nonlocalization of local freeness.* The lack of a workable notion of the conventional derived category (as opposed to the coderived category) for quasicoherent matrix factorizations stands in the way of a direct extension of the Thomason–Trobaugh–Neeman localization theorem for perfect complexes [Thomason and Trobaugh 1990; Neeman 1992; 1996] to locally free matrix factorizations of finite rank. We have seen in Section 1.10 how the localization theory can be developed for coherent matrix factorizations. In this section we demonstrate a counterexample showing that the localization theory, in its conventional form, actually does *not* hold for locally free matrix factorizations.

In other words, the restriction

$$\mathsf{D}^{\mathsf{abs}}((X,\mathcal{L},w)\operatorname{-coh}_{\mathsf{lf}})\longrightarrow \mathsf{D}^{\mathsf{abs}}((U,\mathcal{L}|_U,w|_U)\operatorname{-coh}_{\mathsf{lf}})$$

for an open subscheme $U \subset X$ is not always a Verdier quotient functor, even up to the direct summands. Moreover, the triangulated category $D^{abs}((X, \mathcal{L}, w)\text{-coh}_{lf})$ may fail to be generated by a single object, unlike in the case of the categories of perfect complexes on quasicompact quasiseparated schemes.

All the potentials in our example will be simply regular functions, i.e., sections of the trivial line bundle \mathcal{O}_X or \mathcal{O}_U , etc.; so we drop the line bundle \mathcal{L} from our notation in the rest of the section and write simply $\mathsf{D}^{\mathsf{abs}}((X, w)\operatorname{-coh}_{\mathsf{lf}})$ or $\mathsf{D}^{\mathsf{abs}}((X, w)\operatorname{-coh})$, etc. For simplicity, we will work over the basic field of complex numbers \mathbb{C} .

Consider the 3-dimensional affine quadratic cone

$$X = \{xy = zw\} \subset \mathbb{A}^4 = \operatorname{Spec} \mathbb{C}[x, y, z, w].$$

Further, let us take the open subset

$$U = \{ z \neq 0 \} \subset X$$

Clearly, we have an isomorphism of pairs (algebraic variety, regular function on it)

$$(U,w) \xrightarrow{\sim} (\mathbb{A}_{t_1,t_2}^2 \times \mathbb{G}_m, t_1 t_2), \quad (x, y, z, w) \longmapsto \left(\left(x, \frac{y}{z}\right), z\right),$$
(1)

where we denote $\mathbb{A}^2_{t_1,t_2} = \operatorname{Spec} \mathbb{C}[t_1,t_2]$ and, as usual, $\mathbb{G}_m = \mathbb{A}^1 \setminus \{0\}$.

Lemma 3.3. (a) We have a natural equivalence of triangulated categories

 $\mathsf{D}^{\mathsf{abs}}((U, w)\operatorname{-coh}) \simeq \mathsf{D}^{\mathsf{abs}}((\mathbb{G}_m, 0)\operatorname{-coh}).$

(b) The restriction functor

$$D^{abs}((X, w)\text{-coh}) \longrightarrow D^{abs}((U, w)\text{-coh})$$

is an equivalence.

Here the category of matrix factorizations of the zero potential $D^{abs}((Y, 0)\text{-coh})$ is, of course, simply the derived category of 2-periodic complexes of coherent sheaves on a smooth variety Y.

Proof. Part (a): By (1), we have the equivalence

$$\mathsf{D}^{\mathsf{abs}}((U, w)\operatorname{-coh}) \simeq \mathsf{D}^{\mathsf{abs}}((\mathbb{A}^2_{t_1, t_2} \times \mathbb{G}_m, t_1 t_2)\operatorname{-coh}).$$

By Knörrer periodicity (cf. [Orlov 2006, Theorem 3.1]), we have the equivalence

$$\mathsf{D}^{\mathsf{abs}}((\mathbb{A}^2_{t_1,t_2} \times \mathbb{G}_m, t_1t_2)\operatorname{-coh}) \simeq \mathsf{D}^{\mathsf{abs}}((\mathbb{G}_m, 0)\operatorname{-coh}).$$

Part (b): Let us put $D = X \setminus U$. By Theorem 1.10(b) (see also Section 3.1), we have the short exact sequence of triangulated categories

$$0 \longrightarrow \mathsf{D}_D^{\mathsf{abs}}((X, w)\operatorname{-coh}) \longrightarrow \mathsf{D}^{\mathsf{abs}}((X, w)\operatorname{-coh}) \longrightarrow \mathsf{D}^{\mathsf{abs}}((U, w)\operatorname{-coh}) \longrightarrow 0$$

Thus, we need to show that the category $D_D^{abs}((X, w))$ -coh) is zero. It suffices to check that the category $D^{abs}((D, w)$ -coh) is zero.

Let us put $S = \{xy = 0\} \subset \mathbb{A}^2$. Then we have the isomorphism

$$(D,w) \xrightarrow{\sim} (S \times \mathbb{A}^1_t, t), \quad (x, y, 0, w) \longmapsto ((x, y), w).$$

Since $D^{abs}((\mathbb{A}^1_t, t)\text{-coh}) = 0$, it follows that

$$\mathsf{D}^{\mathsf{abs}}((D, w)\operatorname{-coh}) \simeq \mathsf{D}^{\mathsf{abs}}((S \times \mathbb{A}^1_t, t)\operatorname{-coh}) = 0.$$

Since U is smooth, we have the equivalence

$$\mathsf{D}^{\mathsf{abs}}((U, w)\operatorname{-coh}_{\mathsf{lf}}) \simeq \mathsf{D}^{\mathsf{abs}}((U, w)\operatorname{-coh}).$$

Now we turn to the category $D^{abs}((X, w)\text{-coh}_{lf})$. As usual, we put

$$X_0 = \{w = 0\} \subset X.$$

According to Theorem 2.7, the triangulated category $D^{abs}((X, w)\operatorname{-coh}_{lf})$ is equivalent to the kernel of the direct image functor $i_{\circ}: D^{b}_{Sing}(X_{0}) \to D^{b}_{Sing}(X)$ acting between the triangulated categories of singularities of the schemes X_{0} and X. This can be rephrased by saying that $D^{abs}((X, w)\operatorname{-coh}_{lf})$ is equivalent to the quotient category of the category of bounded complexes of coherent sheaves on X_{0} whose direct images are perfect complexes on X by the category of perfect complexes on X_{0} . Denoting the triangulated category of coherent complexes on X_{0} whose direct images are perfect on X by $Perf(X_{0}, X) \subset D^{b}(X_{0}\operatorname{-coh})$, we have the equivalence of triangulated categories

$$\mathsf{D}^{\mathsf{abs}}((X, w) \operatorname{-coh}_{\mathsf{lf}}) \simeq \operatorname{Perf}(X_0, X) / \operatorname{Perf}(X_0).$$
⁽²⁾

Note that we have the natural isomorphism

 $X_0 \simeq S \times \mathbb{A}^1$, $(x, y, z, 0) \longmapsto ((x, y), z)$.

It follows immediately that

$$\mathsf{D}^{\mathsf{b}}_{Sing}(X_0) \simeq \mathsf{D}^{\mathsf{abs}}((\mathbb{A}^1, 0)\operatorname{-\mathsf{coh}}). \tag{3}$$

Proposition 3.3. (a) We have the natural equivalence of triangulated categories

 $Perf(X_0, X)/Perf(X_0) \simeq D^{abs}((\mathbb{G}_m, 0)\text{-coh})_{0\text{-dim}},$

where $D^{abs}((\mathbb{G}_m, 0)\text{-coh})_{0\text{-dim}} \subset D^{abs}((\mathbb{G}_m, 0)\text{-coh})$ is the subcategory of complexes with zero-dimensional support.

Moreover, we have a commutative diagram of fully faithful triangulated functors:

$$\begin{array}{c|c} Perf(X_0, X)/Perf(X_0) & \longrightarrow & \mathsf{D}^{\mathsf{b}}_{Sing}(X_0) \\ & & & \\ & & & \\ & & & \\ \mathsf{D}^{\mathsf{abs}}((\mathbb{G}_m, 0)\operatorname{-coh})_{0\operatorname{-dim}} & \xrightarrow{j_*} & \mathsf{D}^{\mathsf{abs}}((\mathbb{A}^1, 0)\operatorname{-coh}), \end{array}$$

where $j : \mathbb{G}_m \to \mathbb{A}^1$ is the open embedding.

(b) We have a commutative diagram of fully faithful triangulated functors and equivalences:

$$\begin{array}{c|c} \mathsf{D}^{\mathsf{abs}}((X,w)\operatorname{-coh}_{\mathsf{lf}}) & \longrightarrow & \mathsf{D}^{\mathsf{abs}}((U,w)\operatorname{-coh}_{\mathsf{lf}}) \\ & & & \\ & & \\ & & \\ \mathsf{D}^{\mathsf{abs}}((\mathbb{G}_m,0)\operatorname{-coh})_{0\operatorname{-dim}} & & \longrightarrow & \mathsf{D}^{\mathsf{abs}}((\mathbb{G}_m,0)\operatorname{-coh}), \end{array}$$

where ι is the tautological embedding.

Proof. Part (a): Indeed, from the equivalence (3) we have the natural fully faithful triangulated functor

$$Perf(X_0, X)/Perf(X_0) \longrightarrow D^{abs}((\mathbb{A}^1, 0)\text{-coh}).$$

Let us denote by $T \subset D^{abs}((\mathbb{A}^1, 0)\text{-coh})$ the essential image of this functor. For each $z_0 \in \mathbb{C} \setminus \{0\}$ we have a line $l_{z_0} := \{y = 0, z = z_0\} \subset X_0$. Since $l_{z_0} \subset U$ and U is smooth, the coherent sheaf $\mathcal{O}_{l_{z_0}}$ is contained in $Perf(X_0, X)$. Further, its image in $D^{b}_{Sing}(X_0)$ corresponds to the skyscraper $\mathcal{O}_{z_0} \in D^{abs}((\mathbb{A}^1, 0)\text{-coh})$ under the equivalence (3). It follows that the triangulated category T contains $j_*(\mathbb{D}^{abs}((\mathbb{G}_m, 0)\text{-coh})_{0\text{-dim}})$ as a full subcategory.

Suppose that T is strictly bigger than $j_*(\mathbb{D}^{abs}((\mathbb{G}_m, 0)\operatorname{-coh})_{0\operatorname{-dim}})$. Then it contains an object $\mathcal{F}_0 = \mathcal{O}_0 \oplus \mathcal{O}_0[1] \in \mathbb{D}^{abs}((\mathbb{A}^1, 0)\operatorname{-coh})$, where \mathcal{O}_0 is the structure sheaf of the origin. Denote by $O \in X_0$ the origin (0, 0, 0, 0). Then the image of the coherent sheaf $\mathcal{O}_O \in X_0\operatorname{-coh}$ in $\mathbb{D}^b_{Sing}(X_0)$ corresponds to \mathcal{F}_0 under the equivalence (3). But the object $\mathcal{O}_O \in \mathbb{D}^b(X_0\operatorname{-coh})$ is not relatively perfect under the inclusion $X_0 \to X$ (i.e., it does not belong to $Perf(X_0, X)$) since O is the singular point of X. We get a contradiction.

Thus, we have an equivalence $T \simeq j_*(\mathbb{D}^{abs}((\mathbb{G}_m, 0)\text{-coh})_{0\text{-dim}})$. This proves (a). Part (b) follows immediately from part (a) and the equivalence (2).

In particular, we see that the functor $D^{abs}((X, w)\operatorname{-coh}_{lf}) \to D^{abs}((U, w)\operatorname{-coh}_{lf})$ is not essentially surjective, even up to the direct summands. Moreover, the triangulated category $D^{abs}((X, w)\operatorname{-coh}_{lf})$ does not even have a countable set of generators.

3.4. *Pull-backs and push-forwards in singularity categories.* Let $f : Y \to X$ be a morphism of separated Noetherian schemes with enough vector bundles. The morphism f is said to have *finite flat dimension* if the derived inverse image functor $\mathbb{L}f^* : D^-(X\operatorname{-qcoh}) \to D^-(Y\operatorname{-qcoh})$ takes $D^b(X\operatorname{-qcoh})$ to $D^b(Y\operatorname{-qcoh})$.

In this case, the functor $\mathbb{L} f^*$ induces the inverse image functors on the triangulated categories of singularities

$$f^{\circ}: \mathsf{D}'_{Sing}(X) \longrightarrow \mathsf{D}'_{Sing}(Y)$$
$$f^{\circ}: \mathsf{D}^{\mathsf{b}}_{Sing}(X) \longrightarrow \mathsf{D}^{\mathsf{b}}_{Sing}(Y).$$

Under the same assumption of finite flat dimension, the derived direct image functor $\mathbb{R}f_*: D^{b}(Y\operatorname{-qcoh}) \to D^{b}(X\operatorname{-qcoh})$ takes $D^{b}(Y\operatorname{-qcoh}_{fl})$ to $D^{b}(X\operatorname{-qcoh}_{fl})$, as one can see by computing $\mathbb{R}f_*$ in terms of an affine covering of Y in the spirit of the proof of Proposition 1.9. When the scheme X has finite Krull dimension, one has $D^{b}(X\operatorname{-qcoh}_{fl}) = D^{b}(X\operatorname{-qcoh}_{lf})$, so the functor $\mathbb{R}f_*$ induces the direct image functor

$$f_{\circ}: \mathsf{D}'_{Sing}(Y) \longrightarrow \mathsf{D}'_{Sing}(X),$$

which is right adjoint to f° .

Whenever the morphism f is proper of finite type and has finite flat dimension, the functor $\mathbb{R} f_*$ takes $D^b(Y\text{-coh})$ to $D^b(X\text{-coh})$ [Grothendieck 1961, Théorème 3.2.1] and induces the direct image functor

$$f_{\circ}: \mathsf{D}^{\mathsf{b}}_{Sing}(Y) \longrightarrow \mathsf{D}^{\mathsf{b}}_{Sing}(X),$$

which is right adjoint to f° [Orlov 2004, paragraphs before Proposition 1.14]. More generally, for a morphism f of finite flat dimension and any closed subset $T \subset Y$ such that (a closed subscheme structure on) T is proper of finite type over X, the functor $\mathbb{R} f_*$ takes $D^{b}(Y \operatorname{-coh}_T)$ to $D^{b}(X \operatorname{-coh})$ and induces the direct image functor

$$f_{\circ}: \mathsf{D}^{\mathsf{b}}_{Sing}(Y, T) \longrightarrow \mathsf{D}^{\mathsf{b}}_{Sing}(X).$$

Indeed, the intersection of $D^{b}(X\operatorname{-qcoh}_{fl})$ and $D^{b}(X\operatorname{-coh})$ in $D^{b}(X\operatorname{-qcoh})$ is equal to $D^{b}(X\operatorname{-coh}_{lf})$, as any complex of finite flat dimension with bounded coherent cohomology is easily seen to be perfect.

Let $Z \subset X$ and $W \subset Y$ be closed subschemes such that \mathcal{O}_Z is a perfect \mathcal{O}_X -module, \mathcal{O}_W is a perfect \mathcal{O}_Y -module, and $f(W) \subset Z$. Assume that both morphisms $f: Y \to X$ and $f|_W: W \to Z$ have finite flat dimensions. Then the derived inverse image functor $\mathbb{L}f|_W^* : D^{\mathrm{b}}(Z\operatorname{-qcoh}) \to D^{\mathrm{b}}(W\operatorname{-qcoh})$ induces the inverse image functors on the triangulated categories of relative singularities

$$f^{\circ}: \mathsf{D}'_{Sing}(Z/X) \longrightarrow \mathsf{D}'_{Sing}(W/Y)$$
$$f^{\circ}: \mathsf{D}^{\mathsf{b}}_{Sing}(Z/X) \longrightarrow \mathsf{D}^{\mathsf{b}}_{Sing}(W/Y).$$

Now let $Z \subset X$ be a closed subscheme; set $W = Z \times_X Y$. Denote the closed embeddings $Z \to X$ and $W \to Y$ by *i* and *i'*, respectively; also let *f'* denote the morphism $f|_W : W \to Z$. Assume that *W* coincides with the derived product of *Z* and *Y* over *X*; i.e., $\mathbb{L} f^* i_* \mathcal{O}_Z = i'_* \mathcal{O}_W$. Assume further that $i_* \mathcal{O}_Z$ is a perfect \mathcal{O}_X -module; then also $i'_* \mathcal{O}_W$ is a perfect \mathcal{O}_Y -module.

For any $\mathcal{M} \in D^{b}(Y\operatorname{-qcoh})$, there is a natural morphism

$$\phi_{\mathcal{M}}: \mathbb{L}i^* \mathbb{R}f_* \mathcal{M} \longrightarrow \mathbb{R}f'_* \mathbb{L}i'^* \mathcal{M}$$

in D^b(Z-qcoh). Using the projection formula for tensor products with perfect complexes, one easily checks that the morphism $i_*\phi_M$ is an isomorphism. Hence, so is the morphism ϕ_M since the functor i_* does not annihilate any objects of the derived category. Hence we obtain the induced functor of direct image

$$f_{\circ}: \mathsf{D}'_{Sing}(W/Y) \longrightarrow \mathsf{D}'_{Sing}(Z/X)$$

When the morphism f is proper of finite type, there is also the induced functor

$$f_{\circ}: \mathsf{D}^{\mathsf{b}}_{Sing}(W/Y) \longrightarrow \mathsf{D}^{\mathsf{b}}_{Sing}(Z/X)$$

Assume additionally that the morphism f has finite flat dimension; then so does the morphism f'. In this case, the functor $f_{\circ}: D'_{Sing}(W/Y) \to D'_{Sing}(Z/X)$ is right adjoint to the functor $f^{\circ}: D'_{Sing}(Z/X) \to D'_{Sing}(W/Y)$. When the morphism f is proper of finite type, the functor $f_{\circ}: D^{b}_{Sing}(W/Y) \to D^{b}_{Sing}(Z/X)$ is right adjoint to the functor $f^{\circ}: D^{b}_{Sing}(Z/X) \to D^{b}_{Sing}(W/Y)$.

Remark 3.4. In the case when Z is a Cartier divisor in X, we will construct the functor $f_o: D^b_{Sing}(W/Y) \to D^b_{Sing}(Z/X)$ under somewhat weaker assumptions below in Section 3.5. Namely, it will suffice that the morphism $f': W \to Z$ be proper of finite type, while the morphism $f: Y \to Z$ need not be. A generalization to the case of proper support will also be obtained.

3.5. *Push-forwards of matrix factorizations.* Let $f : Y \to X$ be a morphism of separated Noetherian schemes with enough vector bundles, \mathcal{L} be a line bundle on X, and $w \in \mathcal{L}(X)$ be a section.

Set $\mathcal{B}_X = (X, \mathcal{L}, w)$ and $\mathcal{B}_Y = (Y, f^*\mathcal{L}, f^*w)$; then there is a natural morphism of CDG-algebras $\mathcal{B}_X \to \mathcal{B}_Y$ compatible with the morphism of schemes $f : Y \to X$. Therefore, according to Section 1.8, there are the derived inverse image functors

$$\begin{split} \mathbb{L}f^* : \mathsf{D}^{\mathsf{co}}((X,\mathcal{L},w)\operatorname{-qcoh}_{\mathsf{ffd}}) &\longrightarrow \mathsf{D}^{\mathsf{co}}((Y,f^*\mathcal{L},f^*w)\operatorname{-qcoh}_{\mathsf{ffd}}), \\ \mathbb{L}f^* : \mathsf{D}^{\mathsf{abs}}((X,\mathcal{L},w)\operatorname{-coh}_{\mathsf{ffd}}) &\longrightarrow \mathsf{D}^{\mathsf{abs}}((Y,f^*\mathcal{L},f^*w)\operatorname{-coh}_{\mathsf{ffd}}) \end{split}$$

and the derived direct image functor

$$\mathbb{R}f_*: \mathsf{D}^{\mathsf{co}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-qcoh}) \longrightarrow \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w)\operatorname{-qcoh}).$$

The latter two functors are "partially adjoint" to each other.

Given a triangulated category D, we denote by \overline{D} its idempotent completion. By [Balmer and Schlichting 2001, Section 1], the category \overline{D} has the natural structure of a triangulated category.

Lemma 3.5. For any closed subset $T \subset Y$ such that (for a closed subscheme structure on T) the morphism $f|_T : T \to X$ is proper of finite type, the functor $\mathbb{R}f_*$ takes the full subcategory $\mathsf{D}^{\mathsf{abs}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-coh}_T) \subset \mathsf{D}^{\mathsf{co}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-qcoh})$ into the full subcategory $\mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w)\operatorname{-coh}) \subset \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w)\operatorname{-qcoh})$, thus defining a triangulated functor of direct image

$$\mathbb{R}f_*: \mathsf{D}^{\mathsf{abs}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-coh}_T) \longrightarrow \mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w)\operatorname{-coh}).$$

Consequently, there is the triangulated functor

$$\overline{\mathbb{R}f_*} \colon \overline{\mathsf{D}_T^{\mathsf{abs}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-coh})} \longrightarrow \overline{\mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w)\operatorname{-coh})}.$$

Proof. We will use the construction of the functor

 $\mathbb{R}f_*: \mathsf{D}^{\mathsf{co}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-qcoh}) \longrightarrow \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w)\operatorname{-qcoh})$

similar to the one in the proof of Proposition 1.9 (see Remark 1.9). According to this construction, given a matrix factorization $\mathcal{M} \in (Y, f^*\mathcal{L}, f^*w)$ -qcoh, the object $\mathbb{R} f_*\mathcal{M} \in D^{co}((X, \mathcal{L}, w)$ -qcoh) is represented by the total matrix factorization $\mathbb{R}_{\{U_{\alpha}\}} f_*\mathcal{M}$ of the finite Čech complex $f_*C^{\bullet}_{\{U_{\alpha}\}}\mathcal{M}$ of matrix factorizations on X. The derived functor of direct image of complexes of quasicoherent sheaves $\mathbb{R} f_* : D^{b}(Y$ -qcoh) $\rightarrow D^{b}(X$ -qcoh) can be constructed in the same way.

By [Grothendieck 1961, Théorème 3.2.1], the latter functor takes $D^{b}(Y \operatorname{-coh}_{T})$ into $D^{b}(X\operatorname{-coh})$. Hence the cohomology matrix factorizations of the finite complex of matrix factorizations $f_*C^{\bullet}_{\{U_{\alpha}\}}\mathcal{M}$ belong to $(X, \mathcal{L}, w)\operatorname{-coh}$ when the matrix factorization \mathcal{M} belongs to $(Y, f^*\mathcal{L}, f^*w)\operatorname{-coh}_{T}$. It follows that the object $\mathbb{R}f_*\mathcal{M}$ belongs to $D^{\operatorname{abs}}((X, \mathcal{L}, w)\operatorname{-coh}) \subset D^{\operatorname{co}}((X, \mathcal{L}, w)\operatorname{-qcoh})$ in this case.

To prove the last assertion, it remains to apply Corollary 1.10(b).

Now assume that both morphisms of sheaves $w : \mathcal{O}_X \to \mathcal{L}$ and $f^*w : \mathcal{O}_Y \to f^*\mathcal{L}$ are injective. Let $X_0 \subset X$ and $Y_0 \subset Y$ denote the closed subschemes defined locally by the equations w = 0 and $f^*w = 0$, respectively. In this setting, we will compare the constructions of direct image functors for matrix factorizations and for the triangulated categories of relative singularities, and prove the assertions of Lemma 3.5 in a different way. Recall that in Section 3.4 we constructed the functor of direct image $f_o : D'_{Sing}(Y_0/Y) \to D'_{Sing}(X_0/X)$.

Proposition 3.5. (a) Whenever the morphism $f_0 = f|_{Y_0} : Y_0 \to X_0$ is proper of finite type, the functor $\mathbb{R}f_*$ takes the full subcategory

$$\mathsf{D}^{\mathsf{abs}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-coh}) \subset \mathsf{D}^{\mathsf{co}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-qcoh})$$

into the full subcategory

 $\mathsf{D}^{\mathsf{abs}}((X,\mathcal{L},w)\operatorname{\mathsf{-coh}})\subset\mathsf{D}^{\mathsf{co}}((X,\mathcal{L},w)\operatorname{\mathsf{-qcoh}}),$

thus defining a triangulated functor

$$\mathbb{R}f_*: \mathsf{D}^{\mathsf{abs}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-coh}) \longrightarrow \mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w)\operatorname{-coh}).$$

(b) For any closed subset $T \subset Y_0$ such that (for a closed subscheme structure on T) the morphism $f_0|_T : T \to X_0$ is proper of finite type, the functor f_\circ takes the full subcategory $\mathsf{D}^{\mathsf{b}}_{Sing}(Y_0/Y, T) \subset \mathsf{D}'_{Sing}(Y_0/Y)$ into the full subcategory $\mathsf{D}^{\mathsf{b}}_{Sing}(X_0/X) \subset \mathsf{D}'_{Sing}(X_0/X)$, thus defining a triangulated functor

$$f_{\circ}: \mathsf{D}^{\mathsf{b}}_{Sing}(Y_0/Y, T) \longrightarrow \mathsf{D}^{\mathsf{b}}_{Sing}(X_0/X).$$

(c) The equivalences of categories $D^{abs}((Y, f^*\mathcal{L}, f^*w) \operatorname{-coh}_T) \simeq D^b_{Sing}(X_0/X, T)$ from Proposition 3.1(a) and $D^{abs}((X, \mathcal{L}, w) \operatorname{-coh}) \simeq D^b_{Sing}(X_0/X)$ from Theorem 2.7 transform the direct image functor

$$\mathbb{R}f_*: \mathsf{D}^{\mathsf{abs}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-coh}_T) \longrightarrow \mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w)\operatorname{-coh})$$

from Lemma 3.5 into the direct image functor f_{\circ} from part (b).

Proof. Part (a) follows from Lemma 3.5 and Proposition 3.1(a), or alternatively, from part (b) and the proof of part (c) below. In part (b), the fact of key importance is that the functor $D^{b}_{Sing}(X_0/X) \rightarrow D'_{Sing}(X_0/X)$ is fully faithful (by Theorem 2.8). The functor f_{\circ} takes $D^{b}_{Sing}(Y_0/Y, T)$ into $D^{b}_{Sing}(X_0/X)$ because the functor $\mathbb{R} f_{0*} : D^{b}(Y_0\text{-qcoh}) \rightarrow D^{b}(X_0\text{-qcoh})$ takes $D^{b}(Y_0\text{-coh}_T)$ into $D^{b}(X_0\text{-coh})$ [Grothendieck 1961]. To prove part (c), we will check that the equivalences of categories from Theorem 2.8 transform the functor $\mathbb{R} f_* : D^{co}((Y, f^*\mathcal{L}, f^*w)\text{-qcoh}) \rightarrow$ $D^{co}((X, \mathcal{L}, w)\text{-qcoh})$ into the functor $f_{\circ} : D'_{Sing}(Y_0/Y) \rightarrow D'_{Sing}(X_0/X)$. (Together with part (b) and Proposition 3.1(a), this will also provide another proof of Lemma 3.5.)

For this purpose, extend the functor

$$\Upsilon_Y : \mathsf{D}^{\mathsf{b}}(Y_0\operatorname{-\mathsf{qcoh}}) \longrightarrow \mathsf{D}^{\mathsf{co}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-\mathsf{qcoh}})$$

to the functor $\tilde{\Upsilon}_Y : D^+(Y_0\text{-qcoh}) \to D^{\text{co}}((Y, f^*\mathcal{L}, f^*w)\text{-qcoh})$ in the obvious way (taking infinite direct sums of quasicoherent sheaves in the construction of the matrix factorization $\tilde{\Upsilon}_Y(\mathcal{F}^{\bullet})$). The functor $\tilde{\Upsilon}_Y$ is well-defined since any boundedbelow acyclic complex of quasicoherent sheaves is coacyclic [Positselski 2010, Lemma 2.1]. Furthermore, the functor $\tilde{\Upsilon}_Y$ can be presented as the composition of the "periodicity summation" functor $D^+(Y_0\text{-qcoh}) \to D^{\text{co}}((Y_0, i'^* f^*\mathcal{L}, 0)\text{-qcoh})$ taking values in the coderived category of quasicoherent matrix factorizations of the zero potential on Y_0 , and the functor of direct image $i'_* : D^{\text{co}}((Y_0, i'^* f^*\mathcal{L}, 0)\text{-qcoh}) \to$ $D^{\text{co}}((Y, f^*\mathcal{L}, f^*w)\text{-qcoh})$ with respect to the closed embedding i'.

The functors

$$\mathbb{R}f_{0*}: \mathsf{D}^+(Y_0\operatorname{-qcoh}) \longrightarrow \mathsf{D}^+(X_0\operatorname{-qcoh}),$$
$$\mathbb{R}f_*: \mathsf{D}^{\operatorname{co}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-qcoh}) \longrightarrow \mathsf{D}^{\operatorname{co}}((X, \mathcal{L}, w)\operatorname{-qcoh})$$

form a commutative diagram with the functors $\tilde{\Upsilon}_X$ and $\tilde{\Upsilon}_Y$. Indeed, the "periodicity summations" of bounded-below complexes of quasicoherent sheaves on Y_0 and X_0 , taking injective resolutions to injective resolutions, obviously commute with the derived direct images with respect to f', as the direct image preserves infinite direct sums. Furthermore, the derived direct images of quasicoherent matrix factorizations are compatible with the compositions of morphisms of schemes (see Remark 1.8), and hence also commute with each other. It follows that the functors $\mathbb{R} f_*$ and f_0 agree as they should. (Alternatively, one can prove this in the way similar to the proof of Proposition 3.6 below.)

3.6. *Push-forwards for morphisms of finite flat dimension.* Let $f : Y \to X$ be a morphism of finite flat dimension between separated Noetherian schemes with enough vector bundles, \mathcal{L} be a line bundle on X, and $w \in \mathcal{L}(X)$ be a section. As

in Section 3.5, we have a natural morphism of CDG-algebras $\mathcal{B}_X = (X, \mathcal{L}, w) \rightarrow \mathcal{B}_Y = (Y, f^*\mathcal{L}, f^*w)$ compatible with the morphism of schemes $Y \rightarrow X$.

The quasicoherent graded algebra \mathcal{B}_Y has finite flat dimension over \mathcal{B}_X . Therefore, according to Section 1.9, there are derived inverse image functors

$$\mathbb{L}f^*: \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w)\operatorname{-qcoh}) \longrightarrow \mathsf{D}^{\mathsf{co}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-qcoh})$$
$$\mathbb{L}f^*: \mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w)\operatorname{-coh}) \longrightarrow \mathsf{D}^{\mathsf{abs}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-coh}),$$

the former of which is left adjoint to the functor

 $\mathbb{R}f_*: \mathsf{D}^{\mathsf{co}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-qcoh}) \longrightarrow \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w)\operatorname{-qcoh})$

from Section 3.5.

Furthermore, according to Proposition 1.9, there is a derived direct image functor

$$\mathbb{R}f_*: \mathsf{D}^{\mathsf{co}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-qcoh}_{\mathsf{ffd}}) \simeq \mathsf{D}^{\mathsf{co}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-qcoh}_{\mathsf{fl}}) \longrightarrow \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w)\operatorname{-qcoh}_{\mathsf{ffd}}) \simeq \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w)\operatorname{-qcoh}_{\mathsf{fl}}),$$

which is right adjoint to the functor

$$\mathbb{L}f^*: \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w)\operatorname{-qcoh}_{\mathsf{ffd}}) \longrightarrow \mathsf{D}^{\mathsf{co}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-qcoh}_{\mathsf{ffd}})$$

from Section 3.5.

Now assume that X and Y have finite Krull dimensions. Recall that the natural triangulated functors $D^{abs}((X, \mathcal{L}, w)\operatorname{-coh}_{\mathsf{lf}}) \to D^{\mathsf{co}}((X, \mathcal{L}, w)\operatorname{-qcoh}_{\mathsf{fl}})$ and $D^{abs}((Y, f^*\mathcal{L}, f^*w)\operatorname{-coh}_{\mathsf{lf}}) \to D^{\mathsf{co}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-qcoh}_{\mathsf{fl}})$ are fully faithful by Corollary 2.3(e) and (j).

As in the second half of Section 3.5, assume that both morphisms of sheaves $w : \mathcal{O}_X \to \mathcal{L}$ and $f^*w : \mathcal{O}_Y \to f^*\mathcal{L}$ are injective, and denote by $f_0 : Y_0 \to X_0$ the induced morphism between the zero loci schemes of f^*w and w. Since the morphism f has finite flat dimension, so does the morphism f_0 .

Proposition 3.6. (a) Whenever the morphism f_0 is proper of finite type, the functor

 $\mathbb{R}f_*: \mathsf{D}^{\mathsf{co}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-qcoh}_{\mathsf{fl}}) \longrightarrow \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w)\operatorname{-qcoh}_{\mathsf{fl}})$

takes the full subcategory $D^{abs}((Y, f^*\mathcal{L}, f^*w) \operatorname{-coh}_{lf}) \subset D^{co}((Y, f^*\mathcal{L}, f^*w) \operatorname{-qcoh}_{fl})$ into the full subcategory $D^{abs}((X, \mathcal{L}, w) \operatorname{-coh}_{lf}) \subset D^{co}((X, \mathcal{L}, w) \operatorname{-qcoh}_{fl})$. Besides, the functor

$$f_{0\circ}: \mathsf{D}^{\mathsf{b}}_{Sing}(Y_0) \longrightarrow \mathsf{D}^{\mathsf{b}}_{Sing}(X_0)$$

takes the full subcategory $D^{abs}((Y, f^*\mathcal{L}, f^*w)\text{-coh}_{lf}) \subset D^{b}_{Sing}(Y_0)$ into the full subcategory $D^{abs}((X, \mathcal{L}, w)\text{-coh}_{lf}) \subset D^{b}_{Sing}(X_0)$. Both restrictions define the same triangulated functor

$$\mathbb{R}f_*: \mathsf{D}^{\mathsf{abs}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-coh}_{\mathsf{lf}}) \longrightarrow \mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w)\operatorname{-coh}_{\mathsf{lf}}).$$

(b) For any closed subset $T \subset Y_0$ such that (for a closed subscheme structure on T) the morphism $f_0|_T : T \to X_0$ is proper of finite type, the functor

$$\mathbb{R}f_*: \mathsf{D}^{\mathsf{co}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-qcoh}_{\mathsf{fl}}) \longrightarrow \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w)\operatorname{-qcoh}_{\mathsf{fl}})$$

takes the full subcategory $D_T^{abs}((Y, f^*\mathcal{L}, f^*w) \operatorname{-coh}_{\mathsf{lf}}) \subset D^{\operatorname{co}}((Y, f^*\mathcal{L}, f^*w) \operatorname{-qcoh}_{\mathsf{fl}})$ into the thick envelope of the full subcategory

 $\mathsf{D}^{\mathsf{abs}}((X,\mathcal{L},w)\operatorname{-coh}_{\mathsf{lf}})\subset\mathsf{D}^{\mathsf{co}}((X,\mathcal{L},w)\operatorname{-qcoh}_{\mathsf{fl}}).$

Besides, the triangulated functor

$$\overline{f_{0\circ}}:\overline{\mathsf{D}^{\mathsf{b}}_{Sing}(Y_0,T)}\longrightarrow\overline{\mathsf{D}^{\mathsf{b}}_{Sing}(X_0)}$$

takes the full subcategory $D_T^{abs}((Y, f^*\mathcal{L}, f^*w)\operatorname{-coh}_{lf}) \subset \overline{D_{Sing}^b(Y_0, T)}$ into the thick envelope of the full subcategory

$$\mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w)\operatorname{-coh}_{\mathsf{lf}}) \subset \mathsf{D}^{\mathsf{b}}_{Sing}(X_0).$$

Both restrictions define the same triangulated functor

 $\overline{\mathbb{R}f_*}:\overline{\mathsf{D}_T^{\mathsf{abs}}((Y,f^*\mathcal{L},f^*w)\operatorname{-coh}_{\mathsf{lf}})}\longrightarrow\overline{\mathsf{D}^{\mathsf{abs}}((X,\mathcal{L},w)\operatorname{-coh}_{\mathsf{lf}})}.$

Proof. Both categories $D^{co}((X, \mathcal{L}, w)\operatorname{-qcoh}_{fl})$ and $D^{b}_{Sing}(X_{0})$ are full triangulated subcategories of the triangulated category $D'_{Sing}(X_{0})$ (see Proposition 2.8 and [Orlov 2004, Proposition 1.13]). According to the proof of Corollary 3.2, the intersection of $D^{co}((X, \mathcal{L}, w)\operatorname{-qcoh}_{fl})$ with (the thick envelope of) $D^{b}_{Sing}(X_{0})$ in $D'_{Sing}(X_{0})$ (is the thick envelope of) the subcategory $D^{abs}((X, \mathcal{L}, w)\operatorname{-coh}_{lf}) \subset D'_{Sing}(X_{0})$.

Thus it suffices to show that the direct image functor

$$\mathbb{R}f_*: \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w)\operatorname{-qcoh}_{\mathsf{fl}}) \longrightarrow \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w)\operatorname{-qcoh}_{\mathsf{fl}})$$

agrees with the direct image functor $f_{0\circ}: \mathsf{D}'_{Sing}(Y_0) \to \mathsf{D}'_{Sing}(X_0)$. The latter assertion does not depend on any properness assumptions.

Recall that the derived functor $\mathbb{R} f_*$ was constructed in the proof of Proposition 1.9 in terms of the Čech complex whose terms are direct sums of the CDG-modules $f|_{V*}\mathcal{M}|_V$, where $\mathcal{M} \in \mathsf{D^{co}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-qcoh}_{\mathrm{ffd}})$ and $V \subset Y$. The derived direct image $\mathbb{R} f_{0*} : \mathsf{D^b}(Y_0\operatorname{-qcoh}) \to \mathsf{D^b}(X_0\operatorname{-qcoh})$ can be constructed in the similar way; moreover, one can use for this purpose the restriction to Y_0 of an affine open covering U_{α} of the scheme Y.

We will make use of the flat dimension analogue of Corollary 2.6(d). Let $\tilde{\Sigma}'_X$ and $\tilde{\Sigma}'_Y$ denote the obvious extensions of the functors Σ' from (X, \mathcal{L}, w) -qcoh_{lf} to the category of w-flat matrix factorizations of finite flat dimension (X, \mathcal{L}, w) -qcoh_{w-fl \cap ffd} and from $(Y, f^*\mathcal{L}, f^*w)$ -qcoh_{lf} to $(Y, f^*\mathcal{L}, f^*w)$ -qcoh_f (see the proofs of Proposition 2.8 and Theorem 2.7). Notice that the direct image functors $f|_{V*}$

take f^*w -flat sheaves to w-flat sheaves and $(V, f^*\mathcal{L}|_V, f^*w|_V)$ -qcoh $_{f^*w$ -fl \cap ffd to (X, \mathcal{L}, w) -qcoh $_{w$ -fl \cap ffd.

Let \mathcal{N} be a matrix factorization from $(Y, f^*\mathcal{L}, f^*w)$ -qcoh_{f^*w -fl \cap ffd}. Since the open subschemes V are presumed to be affine, there are natural isomorphisms

$$\widetilde{\Sigma}'_X(f|_{V*}\mathcal{N}|_V) \simeq f_0|_{V \cap Y_0} * \widetilde{\Sigma}'_Y(\mathcal{N})|_{V \cap Y_0}$$

of quasicoherent sheaves on X_0 . Now it remains to use the next lemma.

Lemma 3.6. Let $\mathcal{M}^{-n} \to \cdots \to \mathcal{M}^N$ be a finite complex of matrix factorizations from (X, \mathcal{L}, w) -qcoh_{w-fl \cap ffd} and \mathcal{M} be its totalization. Then the complex $\tilde{\Sigma}'(\mathcal{M}^{-n}) \to \cdots \to \tilde{\Sigma}'(\mathcal{M}^N)$ and the quasicoherent sheaf $\tilde{\Sigma}'(\mathcal{M})$ on X_0 represent naturally isomorphic objects in the triangulated category of singularities $\mathsf{D}'_{Sing}(X_0)$. The same applies to a finite complex of matrix factorizations from (X, \mathcal{L}, w) -qcoh_{w-fl}, the functor Ξ , and the triangulated category of relative singularities $\mathsf{D}'_{Sing}(X_0/X)$.

Proof. For each $-n \le p \le N$, the restriction of the matrix factorization \mathcal{M}^p to the closed subscheme $X_0 \subset X$ is an unbounded complex of quasicoherent sheaves $i^*\mathcal{M}^{p,\bullet}$. By [Polishchuk and Vaintrob 2011, Lemma 1.5], this complex is acyclic.

The complex $\tilde{\Sigma}'(\mathcal{M}^{-n}) \to \cdots \to \tilde{\Sigma}'(\mathcal{M}^N)$ of quasicoherent sheaves on X_0 is quasi-isomorphic to the total complex of the bicomplex $\mathcal{K}^{\bullet,\bullet}$ with the terms $\mathcal{K}^{p,0} = i^* \mathcal{M}^{p,0}, \quad \mathcal{K}^{p,-1} = i^* \mathcal{M}^{p,-1}, \quad \mathcal{K}^{p,-2} = \ker(i^* \mathcal{M}^{p,-1} \to i^* \mathcal{M}^{p,0}), \text{ and } \mathcal{K}^{p,q} = 0$ for $q \neq 0, -1, -2$. Similarly, the quasicoherent sheaf $\tilde{\Sigma}'(\mathcal{M})$ on X_0 is quasi-isomorphic to the total complex of the bicomplex $\mathcal{E}^{\bullet,\bullet}$ with the terms $\mathcal{E}^{p,p} = i^* \mathcal{M}^{p,p}, \quad \mathcal{E}^{p,p-1} = i^* \mathcal{M}^{p,p-1}, \quad \mathcal{E}^{p,p-2} = \ker(i^* \mathcal{M}^{p,p-1} \to i^* \mathcal{M}^{p,p}),$ and $\mathcal{E}^{p,q} = 0$ for $q - p \neq 0, -1, -2$.

We can assume that $N, n \ge 0$. Consider the bicomplex $\mathcal{F}^{\bullet,\bullet}$ with the terms $\mathcal{F}^{p,q} = i^* \mathcal{M}^{p,q}$ for $-n-1 \le q \le N$, $\mathcal{F}^{p,-n-2} = \ker(i^* \mathcal{M}^{p,-n-1} \to i^* \mathcal{M}^{p,-n})$, and $\mathcal{F}^{p,q} = 0$ for q < -n-2 or q > N. Then there are natural surjective morphisms of bicomplexes $\mathcal{F}^{\bullet,\bullet} \to \mathcal{K}^{\bullet,\bullet}$ and $\mathcal{F}^{\bullet,\bullet} \to \mathcal{E}^{\bullet,\bullet}$. The kernels of both morphisms are the direct sums of a finite bicomplex of quasicoherent sheaves of finite flat dimension on X_0 and a finite bicomplex of quasicoherent sheaves on X_0 with acyclic columns. Thus both morphisms become isomorphisms in $D'_{Sing}(X_0)$. \Box

Remark 3.6. One would like to have a theory of set-theoretic supports for locally free matrix factorizations of finite rank that would allow us to prove Proposition 3.6 in the way similar to the proof of Lemma 3.5. However, we do not know how to do this. In particular, we do *not* know whether every locally free matrix factorization of finite rank with category-theoretic support in T is isomorphic in the absolute derived category to a direct summand of an object represented by a coherent matrix factorization of finite flat dimension with set-theoretic support in T (cf. Corollary 1.10 and Section 3.3).

Another alternative approach to proving Proposition 3.6 would be to show that the intersection of the full subcategories $D^{abs}((X, \mathcal{L}, w)\text{-coh})$ and $D^{abs}((X, \mathcal{L}, w)\text{-qcoh}_{lf})$ in the absolute derived category $D^{abs}((X, \mathcal{L}, w)\text{-qcoh})$ coincides with the full subcategory $D^{abs}((X, \mathcal{L}, w)\text{-qcoh}_{lf})$. We do *not* know whether this is true.

3.7. *Duality and push-forwards.* In the following two sections we discuss the compatibility properties of the derived direct and inverse image functors for matrix factorizations with the Serre–Grothendieck duality functors from Section 2.5.

Let X be a separated Noetherian scheme with a dualizing complex \mathcal{D}_X^{\bullet} , and let $f: Y \to X$ be a separated morphism of finite type. As usually, we set $\mathcal{D}_Y^{\bullet} = f^+ \mathcal{D}_X^{\bullet}$, where f^+ is the functor denoted by $f^!$ in [Hartshorne 1966] (right adjoint to $\mathbb{R} f_*$ for proper morphisms f and left adjoint to $\mathbb{R} f_*$ for open embeddings f; see [Neeman 1996, Example 4.2] and [Hartshorne 1966, Remark before Proposition V.8.5 and Deligne's Appendix]). This formula defines the dualizing complex \mathcal{D}_Y^{\bullet} up to a natural quasi-isomorphism only, and we presume this derived category object (as well as \mathcal{D}_X^{\bullet}) to be represented by a finite complex of injective quasicoherent sheaves.

Proposition 3.7. Let $T \subset Y_0$ be a closed subset such that (for some closed subscheme structure on T) the morphism $f|_T : T \to X_0$ is proper. Then the derived direct image functor

$$\overline{\mathbb{R}f_*}:\overline{\mathsf{D}_T^{\mathsf{abs}}((Y,f^*\mathcal{L},f^*w)\text{-}\mathsf{coh})}\longrightarrow\overline{\mathsf{D}^{\mathsf{abs}}((X,\mathcal{L},w)\text{-}\mathsf{coh})}$$

and the similar functor for the potential -w form a commutative diagram with the Serre duality functors

$$\mathcal{H}om_{X\operatorname{-qc}}(-,\mathcal{D}_X^{\bullet}): \mathsf{D}^{\mathsf{abs}}((X,\mathcal{L},-w)\operatorname{-coh})^{\mathsf{op}} \longrightarrow \mathsf{D}^{\mathsf{abs}}((X,\mathcal{L},w)\operatorname{-coh}),$$
$$\mathcal{H}om_{Y\operatorname{-qc}}(-,\mathcal{D}_Y^{\bullet}): \mathsf{D}_T^{\mathsf{abs}}((Y,f^*\mathcal{L},-f^*w)\operatorname{-coh})^{\mathsf{op}} \longrightarrow \mathsf{D}_T^{\mathsf{abs}}((Y,f^*\mathcal{L},f^*w)\operatorname{-coh}).$$

Two proofs of Proposition 3.7 are given below. One of them is based on the theory of set-theoretic supports of coherent CDG-modules developed in Section 1.10 and the arguments similar to the proof of Lemma 3.5. It does not depend on the assumption about w and f^*w being local nonzero-divisors and does not mention the zero loci. The other proof is based on the passage to the triangulated categories of relative singularities and uses Proposition 3.5(c).

First proof. First of all, the duality functor

 $\mathcal{H}om_{Y\text{-qc}}(-, \mathcal{D}_Y^{\bullet}) : \mathsf{D}^{\mathsf{abs}}((Y, f^*\mathcal{L}, -f^*w)\text{-qcoh})^{\mathsf{op}} \longrightarrow \mathsf{D}^{\mathsf{abs}}((Y, f^*\mathcal{L}, f^*w)\text{-qcoh})$ obviously takes the full subcategory $\mathsf{D}^{\mathsf{abs}}((Y, f^*\mathcal{L}, -f^*w)\text{-coh}_T)^{\mathsf{op}}$ into

$$\mathsf{D}^{\mathsf{abs}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-coh}_T)$$

and vice versa. Furthermore, for any quasicoherent sheaf \mathcal{K} on Y denote by $\Gamma_T \mathcal{K} \subset \mathcal{K}$ the maximal quasicoherent subsheaf with set-theoretic support in T.

Then for any matrix factorization $\mathcal{M} \in \mathsf{D}^{\mathsf{abs}}((Y, f^*\mathcal{L}, -f^*w)\operatorname{-coh}_T)$, the natural morphism $\mathcal{H}om_{Y\operatorname{-qc}}(\mathcal{M}, \Gamma_T \mathcal{D}_Y^{\bullet}) \to \mathcal{H}om_{Y\operatorname{-qc}}(\mathcal{M}, \mathcal{D}_Y^{\bullet})$ is an isomorphism in $\mathsf{D}^{\mathsf{abs}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-coh}_T)$.

As in the proof of Lemma 3.5, we will use the construction of the functor

$$\mathbb{R}f_*: \mathsf{D}^{\mathsf{abs}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-qcoh}) \longrightarrow \mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w)\operatorname{-qcoh})$$

similar to the one from the proof of Proposition 1.9 (see Remarks 1.8 and 1.9). Let $\{U_{\alpha}\}$ and $\{V_{\beta}\}$ be two affine open coverings of the scheme Y. For any matrix factorization $\mathcal{N} \in (Y, f^*\mathcal{L}, -f^*w)$ -qcoh, there is a natural morphism of bicomplexes of matrix factorizations

$$f_*C^{\bullet}_{\{U_{\alpha}\}} \mathcal{H}om_{Y-\mathsf{qc}}(\mathcal{N}, \ \Gamma_T\mathcal{D}^{\bullet}_Y) \longrightarrow \mathcal{H}om_{X-\mathsf{qc}}(f_*\mathcal{N}, \ f_*C^{\bullet}_{\{U_{\alpha}\}}\Gamma_T\mathcal{D}^{\bullet}_Y).$$

Passing to the total complexes and taking the composition with the adjunction morphism $f_*C_{\{U_\alpha\}} \circ \Gamma_T \mathcal{D}_Y^{\bullet} = \mathbb{R} f_*(\Gamma_T \mathcal{D}_Y^{\bullet}) \to \mathcal{D}_X^{\bullet}$, we obtain a natural morphism of complexes of matrix factorizations

$$f_*C^{\bullet}_{\{U_{\alpha}\}} \mathcal{H}om_{Y-\mathsf{qc}}(\mathcal{N}, \, \Gamma_T\mathcal{D}^{\bullet}_Y) \longrightarrow \mathcal{H}om_{X-\mathsf{qc}}(f_*\mathcal{N}, \, \mathcal{D}^{\bullet}_X)$$

(cf. [Neeman 1996, beginning of Section 6]).

Substituting $\mathcal{N} = C^{\bullet}_{\{V_{\beta}\}}\mathcal{M}$ for some $\mathcal{M} \in (Y, f^*\mathcal{L}, -f^*w)$ -qcoh, we get a natural morphism of bicomplexes of matrix factorizations

$$f_*C^{\bullet}_{\{U_{\alpha}\}} \mathcal{H}om_{Y-\mathsf{qc}}(C^{\bullet}_{\{V_{\beta}\}}\mathcal{M}, \Gamma_T\mathcal{D}^{\bullet}_Y) \longrightarrow \mathcal{H}om_{X-\mathsf{qc}}(f_*C^{\bullet}_{\{V_{\beta}\}}\mathcal{M}, \mathcal{D}^{\bullet}_X).$$

When \mathcal{M} is a coherent matrix factorization supported set-theoretically in T, the induced morphism of the total complexes is a quasi-isomorphism of complexes of matrix factorizations by the conventional Serre–Grothendieck duality theorem for bounded derived categories of coherent sheaves and proper morphisms of schemes (see [Hartshorne 1966, Theorem VII.3.3] or [Neeman 1996, Section 6]). Hence the induced morphism of the total matrix factorizations is an isomorphism in $D^{abs}((X, \mathcal{L}, w)$ -qcoh), and consequently also in $D^{abs}((X, \mathcal{L}, w)$ -coh).

Second proof. Assume that w and f^*w are locally nonzero-dividing sections of the respective line bundles. Let $i: X_0 \to X$ be the zero locus of w and $i': Y_0 \to Y$ be the zero locus of f^*w . As above, we set $\mathcal{D}_{X_0}^{\bullet} = \mathbb{R}i^!\mathcal{D}_X^{\bullet}$ and $\mathcal{D}_{Y_0}^{\bullet} = \mathbb{R}i'!\mathcal{D}_Y^{\bullet}$ [Hartshorne 1966, Proposition V.2.4], and presume all these dualizing complexes to be finite complexes of injective quasicoherent sheaves.

The duality functor

$$\mathcal{H}om_{Y-\mathsf{qc}}(-,\mathcal{D}_Y^{\bullet}):\mathsf{D}^{\mathsf{abs}}((Y,f^*\mathcal{L},-f^*w)\operatorname{-coh})^{\mathsf{op}}\longrightarrow\mathsf{D}^{\mathsf{abs}}((Y,f^*\mathcal{L},f^*w)\operatorname{-coh})$$

is compatible with the restrictions to the open subscheme $Y \setminus T$ and thus identifies the full subcategories $D_T^{abs}((Y, f^*\mathcal{L}, -f^*w)\text{-coh})^{op}$ and $D_T^{abs}((Y, f^*\mathcal{L}, f^*w)\text{-coh})$. To prove the proposition, we will define the Serre duality functors on the triangulated categories of relative singularities $D_{Sing}^{b}(Y_0/Y)$ and $D_{Sing}^{b}(X/X_0)$, then check that the equivalences of triangulated categories $\mathbb{L}\Xi = \Upsilon^{-1}$ commute with the dualities, and finally reduce to the conventional Serre–Grothendieck duality theorem for bounded complexes of coherent sheaves.

The duality functor $\mathcal{H}om_{X_0\text{-qc}}(-, \mathcal{D}^{\bullet}_{X_0})$: $D^{\mathsf{b}}(X_0\text{-coh})^{\mathsf{op}} \to D^{\mathsf{b}}(X_0\text{-coh})$ takes objects of the form $\mathbb{L}i^*\mathcal{K}^{\bullet}$, where $\mathcal{K}^{\bullet} \in D^{\mathsf{b}}(X\text{-coh})$, to similar objects. Indeed, one has

$$\mathcal{H}om_{X_0-\mathsf{qc}}(\mathbb{L}i^*\mathcal{K}^{\bullet},\mathcal{D}_{X_0}^{\bullet})\simeq \mathbb{R}i^!\mathcal{H}om_{X-\mathsf{qc}}(\mathcal{K}^{\bullet},\mathcal{D}_{X}^{\bullet})$$

[loc. cit., Proposition V.8.5] and $\mathbb{R}i^! \simeq \mathcal{L}|_{X_0}[-1] \otimes_{\mathcal{O}_{X_0}} \mathbb{L}i^*$ (see the proof of Theorem 2.7). Therefore, we have the induced duality functor

$$\mathcal{H}om_{X_0\text{-qc}}(-,\mathcal{D}_{X_0}^{\bullet}):\mathsf{D}^{\mathsf{b}}_{Sing}(X_0/X)^{\mathsf{op}}\longrightarrow\mathsf{D}^{\mathsf{b}}_{Sing}(X_0/X).$$

Similarly, the duality functor

1246

$$\mathcal{H}om_{Y_0\text{-}\mathsf{qc}}(-,\mathcal{D}^{\bullet}_{Y_0}):\mathsf{D}^{\mathsf{b}}(Y_0\text{-}\mathsf{coh})^{\mathsf{op}}\longrightarrow\mathsf{D}^{\mathsf{b}}(Y_0\text{-}\mathsf{coh})$$

takes the full subcategory $D^{b}(Y_{0}-\operatorname{coh}_{T})^{\operatorname{op}}$ into $D^{b}(Y_{0}-\operatorname{coh}_{T})$ and $\operatorname{Perf}_{T}(Y_{0}/Y)^{\operatorname{op}}$ into $\operatorname{Perf}_{T}(Y_{0}/Y)$. Hence the induced duality functor

$$\mathcal{H}om_{Y_0\text{-qc}}(-, \mathcal{D}^{\bullet}_{Y_0}) : \mathsf{D}^{\mathsf{b}}_{Sing}(Y_0/Y, T)^{\mathsf{op}} \longrightarrow \mathsf{D}^{\mathsf{b}}_{Sing}(Y_0/Y, T).$$

Checking that the equivalence of categories $D^{abs}((X, \mathcal{L}, w)\text{-coh}) \simeq D^{b}_{Sing}(X_0/X)$ commutes with the dualities is easily done using the functor Υ . It suffices to notice the functorial quasi-isomorphism $\mathcal{H}om_{X\text{-qc}}(i_*\mathcal{F}^{\bullet}, \mathcal{D}^{\bullet}_X) \simeq i_*\mathcal{H}om_{X_0\text{-qc}}(\mathcal{F}^{\bullet}, \mathcal{D}^{\bullet}_{X_0})$ for any complex $\mathcal{F}^{\bullet} \in D^{b}(X_0\text{-coh})$ [loc. cit., Theorem III.6.7]. The same applies to the equivalence of categories

$$\mathsf{D}_T^{\mathsf{abs}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-coh}) \simeq \mathsf{D}_{Sing}^{\mathsf{b}}(Y_0/Y, T).$$

Furthermore, by Proposition 3.5(c), the equivalences of categories $\mathbb{L}\Xi = \Upsilon^{-1}$ transform the derived direct image functor

$$\overline{\mathbb{R}f_*}: \overline{\mathsf{D}_T^{\mathsf{abs}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-coh})} \longrightarrow \overline{\mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w)\operatorname{-coh})}$$

into (the idempotent closure of) the direct image functor $f_{\circ} : D^{b}_{Sing}(Y_{0}/Y, T) \rightarrow D^{b}_{Sing}(X_{0}/X)$.

Finally, the direct image functor $f_{\circ} : D^{b}_{Sing}(Y_{0}/Y, T) \to D^{b}_{Sing}(X_{0}/X)$ commutes with the Serre duality functors since so do the derived direct image functors $\mathbb{R}f|_{T*}: D^{b}(\tilde{T}\text{-coh}) \to D^{b}(X_{0}\text{-coh})$ for all the closed subscheme structures $\tilde{T} \subset Y_{0}$ on the closed subset T and the similar functors related to the closed embeddings $\tilde{T}' \to \tilde{T}''$ of various such subscheme structures into each other. This is the conventional Serre–Grothendieck duality theorem for proper morphisms of schemes. \Box

3.8. Duality and pull-backs. Let X be a separated Noetherian scheme with a dualizing complex \mathcal{D}_X^{\bullet} and $f: Y \to X$ be a separated morphism of finite type; set $\mathcal{D}_Y^{\bullet} = f^+ \mathcal{D}_X^{\bullet}$. Let \mathcal{L} be a line bundle on X and $w \in \mathcal{L}(X)$ be a section.

Let us first suppose that the morphism f is smooth of relative dimension n. Then the functor $f^+ : D^+(X\operatorname{-qcoh}) \to D^+(Y\operatorname{-qcoh})$ is naturally isomorphic to $\omega_{Y/X}[n] \otimes_{\mathcal{O}_Y} f^*$, where $\omega_{Y/X}$ is the line bundle of relative top forms.

In particular, $\mathcal{D}_Y^{\bullet} \simeq \omega_{Y/X}[n] \otimes_{\mathcal{O}_Y} f^* \mathcal{D}_X^{\bullet}$ (where $f^* \mathcal{D}_X^{\bullet}$ is also presumed to have been replaced by a complex of injectives). Then it is clear that the equivalences of categories

$$\begin{split} \mathcal{D}^{\bullet}_X \otimes_{\mathcal{O}_X} - &: \mathsf{D^{co}}((X,\mathcal{L},w)\operatorname{-qcoh}_{\mathrm{fl}}) \longrightarrow \mathsf{D^{co}}((X,\mathcal{L},w)\operatorname{-qcoh}), \\ f^*\mathcal{D}^{\bullet}_X \otimes_{\mathcal{O}_Y} - &: \mathsf{D^{co}}((Y,f^*\mathcal{L},f^*w)\operatorname{-qcoh}_{\mathrm{fl}}) \longrightarrow \mathsf{D^{co}}((Y,f^*\mathcal{L},f^*w)\operatorname{-qcoh}) \end{split}$$

from Section 2.5 transform the inverse image functor for flat matrix factorizations $f^* : D^{co}((X, \mathcal{L}, w)\operatorname{-qcoh}_{fl}) \to D^{co}((Y, f^*\mathcal{L}, f^*w)\operatorname{-qcoh}_{fl})$ into the (underived, as the morphism f is flat) inverse image functor for quasicoherent matrix factorizations $f^* : D^{co}((X, \mathcal{L}, w)\operatorname{-qcoh}) \to D^{co}((Y, f^*\mathcal{L}, f^*w)\operatorname{-qcoh}).$

Furthermore, for any quasicoherent matrix factorization \mathcal{M} on X there is a natural morphism of finite complexes of matrix factorizations $f^* \mathcal{H}om_{X-qc}(\mathcal{M}, \mathcal{D}_X^{\bullet}) \rightarrow \mathcal{H}om_{Y-qc}(f^*\mathcal{M}, f^*\mathcal{D}_X^{\bullet})$ on Y. When \mathcal{M} is a coherent matrix factorization, this is a quasi-isomorphism of complexes of matrix factorizations (since the similar assertion holds for coherent sheaves [Hartshorne 1966, Proposition II.5.8]), so the related morphism of total matrix factorizations has an absolutely acyclic cone. Thus the antiequivalences of categories

$$\mathcal{H}om_{X-\mathsf{qc}}(-,\mathcal{D}_X^{\bullet}):\mathsf{D}^{\mathsf{abs}}((X,\mathcal{L},-w)\operatorname{-coh})^{\mathsf{op}}\longrightarrow\mathsf{D}^{\mathsf{abs}}((X,\mathcal{L},w)\operatorname{-coh}),$$
$$\mathcal{H}om_{Y-\mathsf{qc}}(-,f^*\mathcal{D}_X^{\bullet}):\mathsf{D}^{\mathsf{abs}}((Y,f^*\mathcal{L},-f^*w)\operatorname{-coh})^{\mathsf{op}}\longrightarrow\mathsf{D}^{\mathsf{abs}}((Y,f^*\mathcal{L},f^*w)\operatorname{-coh})$$

form a commutative diagram with the inverse image functors f^* for coherent matrix factorizations.

Now suppose that f is a proper morphism of finite type. The following theorem describes the compatibility property of the covariant Serre–Grothendieck duality with the inverse images of matrix factorizations (cf. [Positselski 2012, Theorem 5.15.3], where a similar result is proven for complexes of quasicoherent sheaves).

Theorem 3.8. The equivalences of categories

$$\mathcal{D}_{X}^{\bullet} \otimes_{\mathcal{O}_{X}} - : \mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w) \operatorname{-qcoh}_{\mathsf{fl}}) \longrightarrow \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w) \operatorname{-qcoh}),$$
$$\mathcal{D}_{Y}^{\bullet} \otimes_{\mathcal{O}_{Y}} - : \mathsf{D}^{\mathsf{abs}}((Y, f^{*}\mathcal{L}, f^{*}w) \operatorname{-qcoh}_{\mathsf{fl}}) \longrightarrow \mathsf{D}^{\mathsf{co}}((Y, f^{*}\mathcal{L}, f^{*}w) \operatorname{-qcoh})$$

transform the inverse image functor

 $f^*: \mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w)\operatorname{-qcoh}_{\mathsf{fl}}) \longrightarrow \mathsf{D}^{\mathsf{abs}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-qcoh}_{\mathsf{fl}})$

into the functor $f^!$: $D^{co}((X, \mathcal{L}, w)$ -qcoh) $\rightarrow D^{co}((Y, f^*\mathcal{L}, f^*w)$ -qcoh) right adjoint to the direct image functor

$$\mathbb{R}f_*: \mathsf{D}^{\mathsf{co}}((Y, f^*\mathcal{L}, f^*w)\operatorname{-qcoh}) \longrightarrow \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w)\operatorname{-qcoh})$$

(see the end of Section 1.8).

Proof. For any quasicoherent matrix factorization \mathcal{N} on Y and any flat quasicoherent matrix factorization \mathcal{E} on X, we have to construct an isomorphism

$$\operatorname{Hom}_{\mathsf{D}^{\mathrm{co}}((X,\mathcal{L},w)\operatorname{-qcoh})}(\mathbb{R}f_*\mathcal{N}, \mathcal{D}_X^{\bullet}\otimes_{\mathcal{O}_X}\mathcal{E})$$

\$\approx Hom_{\mathbb{D}^{\mathrm{co}}((Y,f^*\mathcal{L},f^*w)\operatorname{-qcoh})}(\mathcal{N}, \mathcal{D}_Y^{\bullet}\otimes_{\mathcal{O}_Y}f^*\mathcal{E}).\$\$\$

The composition

$$\operatorname{Hom}_{Y}(\mathcal{N}, \mathcal{D}_{Y}^{\bullet} \otimes_{\mathcal{O}_{Y}} f^{*}\mathcal{E}) \longrightarrow \operatorname{Hom}_{X}(\mathbb{R}f_{*}\mathcal{N}, \mathbb{R}f_{*}(\mathcal{D}_{Y}^{\bullet} \otimes_{\mathcal{O}_{Y}} f^{*}\mathcal{E}))$$
$$\simeq \operatorname{Hom}_{X}(\mathbb{R}f_{*}\mathcal{N}, f_{*}\mathcal{D}_{Y}^{\bullet} \otimes_{\mathcal{O}_{X}} \mathcal{E}) \longrightarrow \operatorname{Hom}_{X}(\mathbb{R}f_{*}\mathcal{N}, \mathcal{D}_{X}^{\bullet} \otimes_{\mathcal{O}_{X}} \mathcal{E})$$

provides a morphism from the right-hand to the left-hand side. Here all the Hom functors are taken in the coderived categories of quasicoherent matrix factorizations on Y and X; the middle isomorphism holds since $\mathcal{D}_Y^{\bullet} \otimes_{\mathcal{O}_Y} f^* \mathcal{E}$ is an injective matrix factorization on Y (so the derived direct image can be computed for it by applying the underived direct image functor f_* termwise) and by the projection formula; the last morphism is induced by the adjunction $f_*\mathcal{D}_Y^{\bullet} \to \mathcal{D}_Y^{\bullet}$.

Furthermore, on both sides of the desired isomorphism we have injective matrix factorizations in the second arguments of the Hom functors; hence the Hom can be computed in the homotopy category of matrix factorizations instead of the coderived category in both cases. Finally, one can assume \mathcal{N} to be an injective matrix factorization, too, and compute $\mathbb{R} f_* \mathcal{N} = f_* \mathcal{N}$ termwise (alternatively, one could use the Čech construction). Similarly, the tensor products in the second arguments are totalizations of termwise tensor products.

Now one can fix the components involved for both matrix factorizations \mathcal{N} and \mathcal{E} , obtaining a morphism of finite complexes of abelian groups of the same kind as above, but related to (one-term) complexes of quasicoherent sheaves rather than matrix factorizations. The latter is an isomorphism by [Positselski 2012, Theorem 5.15.3]. It remains to notice that the totalization of an acyclic finite complex of (unbounded) complexes of abelian groups is acyclic.

The next corollary is a matrix factorization version of the main result of Deligne's appendix to [Hartshorne 1966] (see also [Positselski 2012, Section 5.16]).

Corollary 3.8. For any morphism of finite type between separated Noetherian schemes with dualizing complexes $f : Y \to X$, a line bundle \mathcal{L} on X, and a section
$w \in \mathcal{L}(X)$, one can define a triangulated functor

$$f^+: \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w)\operatorname{-qcoh}) \longrightarrow \mathsf{D}^{\mathsf{co}}((Y, f^*\mathcal{L}, f^*w))\operatorname{-qcoh})$$

in such a way that

- (i) for an open embedding f, one has $f^+ = f^*$, and more generally, for a smooth morphism f of relative dimension n one has $f^+ = \omega_{Y/X}[n] \otimes_{\mathcal{O}_Y} f^*$;
- (ii) for a proper morphism f, the functor $f^+ = f^!$ is right adjoint to $\mathbb{R}f_*$;
- (iii) the construction is compatible with the compositions of the morphisms f.

Proof. It suffices to define f^+ : $D^{co}((X, \mathcal{L}, w)$ -qcoh) $\rightarrow D^{co}((Y, f^*\mathcal{L}, f^*w)$ -qcoh) as the functor corresponding to the inverse image of flat quasicoherent matrix factorizations f^* : $D^{abs}((X, \mathcal{L}, w)$ -qcoh_{fl}) $\rightarrow D^{abs}((Y, f^*\mathcal{L}, f^*w)$ -qcoh_{fl}) under the identifications of categories

$$\mathcal{D}_{X}^{\bullet} \otimes_{\mathcal{O}_{X}} -: \mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w) \operatorname{-qcoh}_{\mathsf{fl}}) \longrightarrow \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w) \operatorname{-qcoh}),$$
$$\mathcal{D}_{Y}^{\bullet} \otimes_{\mathcal{O}_{Y}} -: \mathsf{D}^{\mathsf{abs}}((Y, f^{*}\mathcal{L}, f^{*}w) \operatorname{-qcoh}_{\mathsf{fl}}) \longrightarrow \mathsf{D}^{\mathsf{co}}((Y, f^{*}\mathcal{L}, f^{*}w) \operatorname{-qcoh}),$$

where \mathcal{D}_X^{\bullet} is any dualizing complex on X and $\mathcal{D}_Y^{\bullet} = f^+ \mathcal{D}_X^{\bullet}$.

Appendix A. Quasicoherent graded modules

A.1. *Flat quasicoherent sheaves.* I am grateful to A. Neeman for suggesting that a result of the following kind can be proven without much difficulty.

Lemma A.1. On any quasicompact semiseparated scheme, any quasicoherent sheaf is the quotient sheaf of a flat quasicoherent sheaf.

Proof. Let *X* be our scheme. Assume that a quasicoherent sheaf \mathcal{M} over *X* is flat over an open subscheme $V \subset X$; given an affine open subscheme $U \subset X$, we will construct a surjective morphism $\mathcal{N} \to \mathcal{M}$ onto \mathcal{M} from a quasicoherent sheaf \mathcal{N} over *X* that is flat over $U \cup V$. Let *j* denote the embedding $U \to X$. There exists a surjective morphism onto $j^*\mathcal{M}$ from a flat quasicoherent sheaf \mathcal{F} over *U*; let \mathcal{K} denote the kernel of this morphism of sheaves.

Since the morphism $j: U \to X$ is affine and flat, the functor j_* is exact and preserves flatness. Consider the pull-back of the exact triple $j_*\mathcal{K} \to j_*\mathcal{F} \to j_*j^*\mathcal{M}$ with respect to the morphism $\mathcal{M} \to j_*j^*\mathcal{M}$; denote the middle term of the resulting exact triple by \mathcal{N} . One has $\mathcal{N}|_U = \mathcal{F}|_U$, so \mathcal{N} is flat over U. Furthermore, the sheaf $j^*\mathcal{M}$ is flat over $V \cap U$; hence, so is the sheaf \mathcal{K} . The embedding $U \cap V \to V$ is an affine flat morphism, so the sheaf $j_*\mathcal{K}$ is flat over V. From the exact triple $j_*\mathcal{K} \to \mathcal{N} \to \mathcal{M}$, we conclude that \mathcal{N} is flat over V.

It follows immediately that any quasicoherent graded module over a quasicoherent graded algebra \mathcal{B} over X is a quotient module of a flat quasicoherent graded module.

A.2. *Locally projective quasicoherent graded modules.* The following result is essentially due to Raynaud and Gruson [1971] (for a discussion, see [Drinfeld 2006, Section 2]); here we just briefly explain how to deduce the formulation that interests us from their assertions.

Theorem A.2. Let X be an affine scheme and $\{U_{\alpha}\}$ be its finite affine covering. Let B be a quasicoherent graded algebra over X and P be a quasicoherent graded module over B. Then the graded $\mathcal{B}(X)$ -module $\mathcal{P}(X)$ is projective if and only if the graded $\mathcal{B}(U_{\alpha})$ -module $\mathcal{P}(U_{\alpha})$ is projective for every α .

Proof. First of all, a graded module P over a graded ring B is projective if and only if it is projective as an ungraded module. Indeed, if P is graded projective, then it is a homogeneous direct summand of a free graded module; hence P is also ungraded projective. Conversely, pick a homogeneous (of degree 0) surjective homomorphism $F \rightarrow P$ onto a given graded module P from a free graded module F. If P is ungraded projective, this homomorphism has a (perhaps nonhomogeneous) section s, and the homogeneous component of s of degree 0 provides a homogeneous section. Hence it suffices to consider ungraded modules over an ungraded quasicoherent algebra \mathcal{B} .

It is clear that if $\mathcal{P}(X)$ is a projective $\mathcal{B}(X)$ -module, then $\mathcal{P}(V)$ is a projective $\mathcal{B}(V)$ -module for any affine open subscheme $V \subset X$. Conversely, assume that the $\mathcal{B}(U_{\alpha})$ -module $\mathcal{P}(U_{\alpha})$ is projective for every α . Then by the result of [Kaplansky 1958], the $\mathcal{B}(U_{\alpha})$ -modules $\mathcal{P}(U_{\alpha})$ are direct sums of countably generated modules, and it follows easily that so is the $\mathcal{B}(X)$ -module $\mathcal{P}(X)$ (essentially, since a connected graph with an at most countable set of edges at each vertex has a countable number of vertices). Hence we can assume the $\mathcal{B}(X)$ -module $\mathcal{P}(X)$ to be countably generated.

Besides, the $\mathcal{B}(U_{\alpha})$ -modules $\mathcal{P}(U_{\alpha})$ are flat; hence so is the $\mathcal{B}(X)$ -module $\mathcal{P}(X)$. By [Raynaud and Gruson 1971, Corollaire II.2.2.2], it remains to show that the $\mathcal{B}(X)$ -module $\mathcal{P}(X)$ satisfies the Mittag-Leffler condition; this can be easily deduced from the similar property of the $\mathcal{B}(U_{\alpha})$ -modules $\mathcal{P}(U_{\alpha})$ using the formulation of this condition given in Proposition II.2.1.4(ii) or Propositions II.2.1.4(ii) and II.2.1.1(i) of [Raynaud and Gruson 1971] (cf. Sections II.2.5 and II.3.1 of the same paper). \Box

A.3. *Injective quasicoherent graded modules.* The following result is a noncommutative generalization of a theorem of Hartshorne [1966, Theorem II.7.18] about injective quasicoherent sheaves on Noetherian schemes. Our proof method, based on the Artin–Rees lemma, is different from the one in [loc. cit.].

Theorem A.3. Let \mathcal{B} be a Noetherian quasicoherent graded algebra over a Noetherian scheme X. Then any injective object in the category of quasicoherent graded left modules over \mathcal{B} is also an injective object of the category of arbitrary sheaves of graded \mathcal{B} -modules over X.

Consequently, the restriction $\mathcal{J}|_U$ of an injective quasicoherent graded module \mathcal{J} over \mathcal{B} to an open subscheme $U \subset X$ is an injective quasicoherent graded module over $\mathcal{B}|_U$. Conversely, if U_{α} is an open covering of X and the quasicoherent graded $\mathcal{B}|_{U_{\alpha}}$ -modules $\mathcal{J}|_{U_{\alpha}}$ are injective, then a quasicoherent graded \mathcal{B} -module \mathcal{J} is injective. Besides, the underlying sheaf of graded abelian groups of any injective quasicoherent graded \mathcal{B} -module \mathcal{J} is flabby.

Proof. First of all, notice that the abelian category \mathcal{B} -qcoh of quasicoherent graded modules over \mathcal{B} is a locally Noetherian Grothendieck category with coherent graded modules forming the subcategory of Noetherian generators [Hartshorne 1977, Exercise II.5.15]; so, in particular, \mathcal{B} -qcoh has enough injectives and the assertions of Theorem A.3 are not vacuous. The category of sheaves of graded \mathcal{B} -modules \mathcal{B} -mod has similar properties, with the extensions by zero of the restrictions of \mathcal{B} to (small) open subschemes of X forming a set of Noetherian generators [Hartshorne 1966, Theorem II.7.8].

Secondly, let us check that the main result in the first paragraph implies the assertions in the second one. Indeed, injective sheaves of graded \mathcal{B} -modules have all the properties we are interested in. They remain injective after being restricted to an open subscheme since the extension by zero from an open subscheme is an exact functor. They are flabby since given two open subschemes $U \subset V \subset X$ and j_U , j_V being their identity embeddings $U, V \to X$, the morphism of sheaves of graded \mathcal{B} -modules $j_{U!}\mathcal{B}|_U \to j_{V!}\mathcal{B}|_V$ is injective. And their property is local [loc. cit., Lemma II.7.16] because sheaves of graded \mathcal{B} -modules supported inside one of the subschemes U_{α} form a set of generators of the category \mathcal{B} -mod.

Now let \mathcal{J} be an injective quasicoherent graded module over \mathcal{B} . To prove the main assertion, we have to show that for any open subscheme $U \subset X$ and a subsheaf of graded \mathcal{B} -modules $\mathcal{G} \subset j_{U!}\mathcal{B}|_U$, any homogeneous morphism of sheaves of graded \mathcal{B} -modules $\mathcal{G} \to \mathcal{J}$ can be extended to a similar morphism $j_{U!}\mathcal{B}|_U \to \mathcal{J}$. Indeed, \mathcal{G} is a subsheaf of graded \mathcal{B} -modules in the coherent graded \mathcal{B} -module \mathcal{B} ; hence according to the following proposition, there exists a quasicoherent graded \mathcal{B} -module $\mathcal{G} \subset \mathcal{F} \subset \mathcal{B}$ such that the morphism $\mathcal{G} \to \mathcal{J}$ can be extended to a homogeneous morphism of quasicoherent graded \mathcal{B} -modules $\mathcal{F} \to \mathcal{J}$.

Since \mathcal{J} is injective in \mathcal{B} -qcoh, the latter morphism can in turn be extended to a similar morphism $\mathcal{B} \to \mathcal{J}$. Restricting to $j_{U!}\mathcal{B}|_U$, we obtain the desired morphism of sheaves of graded \mathcal{B} -modules $j_{U!}\mathcal{B}|_U \to \mathcal{J}$.

Proposition A.3. In the assumptions of Theorem A.3, let \mathcal{E} be a coherent graded left \mathcal{B} -module, $\mathcal{G} \subset \mathcal{E}$ be a subsheaf of graded \mathcal{B} -modules, \mathcal{M} be a quasicoherent graded \mathcal{B} -module, and $\phi : \mathcal{G} \to \mathcal{M}$ be a morphism of sheaves of graded \mathcal{B} -modules. Then there exists a coherent graded \mathcal{B} -module $\mathcal{G} \subset \mathcal{F} \subset \mathcal{E}$ such that the morphism ϕ can be extended to \mathcal{F} . 1252

Proof. Before proving Proposition A.3, let us reformulate its conclusion as follows. In the same setting, there exists a quasicoherent graded \mathcal{B} -module \mathcal{K} together with an injective morphism $\mathcal{M} \to \mathcal{K}$ and a morphism $\mathcal{E} \to \mathcal{K}$ forming a commutative diagram with the embedding $\mathcal{G} \to \mathcal{E}$ and the morphism $\phi : \mathcal{G} \to \mathcal{M}$. Indeed, if a coherent \mathcal{B} -module \mathcal{F} exists, one can take \mathcal{K} to be the fibered coproduct of \mathcal{E} and \mathcal{M} over \mathcal{F} ; conversely, if a quasicoherent \mathcal{B} -module \mathcal{K} exists, one can take \mathcal{F} to be the full preimage of $\mathcal{M} \subset \mathcal{K}$ under the morphism $\mathcal{E} \to \mathcal{K}$. Notice also that one can always replace \mathcal{M} with its sufficiently big coherent graded \mathcal{B} -submodule.

Now let us state the version of Artin-Rees lemma that we will use.

Lemma A.3. In the assumptions of Theorem A.3, let \mathcal{M} be a coherent graded \mathcal{B} -module, $\mathcal{N} \subset \mathcal{M}$ a coherent graded \mathcal{B} -submodule, and $Z \subset X$ a closed subscheme with the sheaf of ideals $\mathcal{I}_Z \subset \mathcal{O}_X$. Then for any $n \ge 0$, there exists $m \ge 0$ such that the intersection $\mathcal{I}_Z^m \mathcal{M} \cap \mathcal{N}$ is contained in $\mathcal{I}_Z^n \mathcal{N}$.

Proof. Clearly, the question is local, so it suffices to consider the case of an affine scheme X. Then (the graded version of) the Artin–Rees lemma for ideals generated by central elements in noncommutative Noetherian rings [Goodearl and Warfield 1989, Theorem 13.3] applies.

Being a Noetherian object, the sheaf of graded \mathcal{B} -modules \mathcal{G} is generated by a finite number of homogeneous sections $s_n \in \mathcal{G}(U_n)$, where $U_n \subset X$ are some open subschemes. If all of these subschemes coincide with X, the sheaf \mathcal{G} , being a subsheaf of a coherent sheaf generated by global sections, is itself coherent, so there is nothing to prove. In the general case, we will argue by induction on the number of open subschemes U_n that are not equal to X.

Let $U = U_1 \subsetneq X$ be one such open subscheme, and $T = X \setminus U$ be its closed complement. We can assume that \mathcal{M} is a coherent graded \mathcal{B} -module. Let \mathcal{N} denote its maximal coherent graded \mathcal{B} -submodule supported set-theoretically in T. Applying Lemma A.3 to $\mathcal{N} \subset \mathcal{M}$, we conclude that there is a closed subscheme structure $i : Z \to X$ on T such that the morphism $\mathcal{N} \to i_*i^*\mathcal{M}$ is injective. Consequently, so is the morphism $\mathcal{M} \to i_*i^*\mathcal{M} \oplus j_*j^*\mathcal{M}$, where j denotes the open embedding $U \to X$.

Let us show that there is a thicker closed subscheme structure $i': Z' \to X$ on T such that the kernel of the morphism of sheaves $i'_*i'^*\mathcal{G} \to i'_*i'^*\mathcal{E}$ is contained in the kernel of the morphism of sheaves $i'_*i'^*\mathcal{G} \to i_*i^*\mathcal{G}$. Indeed, there exists a finite collection of subsheaves of graded \mathcal{B} -modules in \mathcal{G} , each of them an extension by zero of a coherent graded $\mathcal{B}|_V$ -module from some open subscheme $V \subset X$ such that the stalk of \mathcal{G} at each point of X coincides with the stalk of one of these subsheaves. So the assertion reduces to the case when \mathcal{G} is a coherent graded \mathcal{B} -submodule in \mathcal{E} when it is an equivalent reformulation of Lemma A.3.

Let $\mathcal{H} \subset i'^*\mathcal{E}$ denote the image of the morphism of sheaves of graded $i'^*\mathcal{B}$ -modules $i'^*\mathcal{G} \to i'^*\mathcal{E}$ over the scheme Z'. Let $\iota : Z \to Z'$ be the natural closed embedding. Then, according to the above, the morphism of sheaves of graded $i'^*\mathcal{B}$ -modules $i'^*\mathcal{G} \to \iota_*i^*\mathcal{G}$ induces a morphism $\mathcal{H} \to \iota_*i^*\mathcal{G}$.

The sheaf of graded $i'^*\mathcal{B}$ -modules \mathcal{H} is generated by the images of the restrictions of the sections s_n , $n \ge 2$, to the closed subschemes $Z' \cap U_n \subset U_n$. Hence the induction assumption is applicable to \mathcal{H} , and we can conclude that there exists a quasicoherent graded $i'^*\mathcal{B}$ -module \mathcal{K} on the scheme Z' together with an injective morphism $\iota_*i^*\mathcal{M} \to \mathcal{K}$ and a morphism $i'^*\mathcal{E} \to \mathcal{K}$ forming a commutative diagram with the embedding $\mathcal{H} \to i'^*\mathcal{E}$ and the composition $\mathcal{H} \to \iota_*i^*\mathcal{G} \to \iota_*i^*\mathcal{M}$.

Similarly, the sheaf of graded $\mathcal{B}|_U$ -modules $j^*\mathcal{G}$ is generated by the restrictions of the sections s_n to the open subschemes $U_1 \cap U_n \subset U_n$, among which the (restriction of) the section s_1 is a global section over $U = U_1$. Hence the induction assumption is applicable to $j^*\mathcal{G}$, and there exists a quasicoherent graded $\mathcal{B}|_U$ -module \mathcal{L} together with an injective morphism $j^*\mathcal{M} \to \mathcal{L}$ and a morphism $j^*\mathcal{E} \to \mathcal{L}$ forming a commutative diagram with the embedding $j^*\mathcal{G} \to j^*\mathcal{E}$ and the morphism $j^*\mathcal{G} \to j^*\mathcal{M}$.

Now the injective morphism $\mathcal{M} \to i'_*\mathcal{K} \oplus j_*\mathcal{L}$ (whose first component is the composition $\mathcal{M} \to i_*i^*\mathcal{M} \simeq i'_*\iota_*i^*\mathcal{M} \to i'_*\mathcal{K}$) and the morphism $\mathcal{E} \to i'_*\mathcal{K} \oplus j_*\mathcal{L}$ provide the desired commutative diagram of morphisms of sheaves of graded \mathcal{B} -modules over X.

Appendix B. Hochschild (co)homology of matrix factorizations

This appendix complements the paper [Polishchuk and Positselski 2012] in two ways. Section B.1 contains some modifications and improvements of the main results of [loc. cit.] generally, and as applied to locally free matrix factorizations of finite rank in particular. The main thrust consists of replacing the finite homological dimension conditions in [loc. cit.] with the Noetherianness conditions to the (limited) extent possible.

Section B.2, on the other hand, presents an elementary approach to the computation of Hochschild (co)homology of coherent matrix factorizations, entirely unrelated to that in [loc. cit.] and not based on any notion of Hochschild (co)homology of the second kind, but rather on the Serre–Grothendieck duality theory.

B.1. *Locally free matrix factorizations of finite rank.* In Sections B.1.1–B.1.4, we start with a bit of categorical nonsense, following the lines of [Polishchuk and Posit-selski 2012, Sections 3.3–3.5], but with the additional coherence/Noetherianness conditions imposed from the very beginning. We use the notation from [loc. cit.] rather than that of the main body of this paper. Then in Section B.1.5, we turn to locally free matrix factorizations of finite rank over certain possibly singular, affine algebraic varieties. Finally, Section B.1.6 presents an improvement over

the discussion of matrix factorizations over smooth affine varieties in [loc. cit., Section 4.8]. An example of an application of our techniques to nonaffine varieties can be found in the preprint [Efimov 2012].

B.1.1. Coherent and Noetherian CDG-categories. Let $(\Gamma, \sigma, \mathbf{1})$ be a grading group data [Polishchuk and Positselski 2012, Section 1.1] and $B^{\#}$ be a small Γ -graded preadditive category [Positselski 2011a, Section A.1]. Both left and right Γ -graded $B^{\#}$ -modules form abelian categories.

A Γ -graded $B^{\#}$ -module is said to be *finitely generated* (respectively, *finitely presented*) if it is a quotient module of a finitely generated free Γ -graded $B^{\#}$ -module [Polishchuk and Positselski 2012, Section 1.5] (respectively, the cokernel of a morphism of finitely generated free Γ -graded $B^{\#}$ -modules).

A Γ -graded preadditive category $B^{\#}$ is called *left Noetherian* if any submodule of a finitely generated Γ -graded left $B^{\#}$ -module is finitely generated, or equivalently, if the abelian category of Γ -graded left $B^{\#}$ -modules is locally Noetherian. A Γ -graded preadditive category $B^{\#}$ is called *left coherent* if any submodule of a finitely presented Γ -graded left $B^{\#}$ -module is finitely presented.

Let *B* be a small (Γ -graded) CDG-category [loc. cit., Section 1.2] and $B^{\#}$ be its underlying Γ -graded preadditive category. Following [loc. cit.], we denote the DG-categories of left and right CDG-modules over *B* by *B*-mod^{cdg} and mod^{cdg}-*B*. The DG-subcategories of left CDG-modules whose underlying Γ -graded $B^{\#}$ -modules are flat or injective are denoted by B-mod^{cdg} and B-mod^{cdg}_{inj} \subset *B*-mod^{cdg}. Similarly, the DG-subcategories of left and right CDG-modules over *B* whose underlying Γ -graded $B^{\#}$ -modules are projective and finitely generated are denoted by *B*-mod^{cdg} and mod^{cdg}_{fgp}-*B*.

Assuming that the Γ -graded category $B^{\#}$ is left Noetherian, the DG-subcategory of left CDG-modules whose underlying Γ -graded $B^{\#}$ -modules are finitely generated is denoted by B-mod^{cdg}_{fg} $\subset B$ -mod^{cdg}. Assuming that the Γ -graded category $B^{\#}$ is right coherent, the DG-subcategory of right CDG-modules whose underlying Γ -graded $B^{\#}$ -modules are finitely presented is denoted by mod^{cdg}_{fp}-B.

The coderived and contraderived categories of left CDG-modules over *B* are denoted by $D^{co}(B \text{-mod}^{cdg})$ and $D^{ctr}(B \text{-mod}^{cdg})$, respectively [loc. cit., Section 3.2]. Assuming that the Γ -graded category $B^{\#}$ is right coherent, the class of flat Γ -graded left *B*-modules [loc. cit., Section 2.2] is closed under infinite products, so the contraderived category $D^{ctr}(B \text{-mod}_{fl}^{cdg})$ is well-defined. The homotopy category of the DG-category $B \text{-mod}_{inj}^{cdg}$ is denoted, as usually, by $H^0(B \text{-mod}_{inj}^{cdg})$. In the respective assumptions of left Noetherianness or right coherence of the

In the respective assumptions of left Noetherianness or right coherence of the Γ -graded category $B^{\#}$, the absolute derived categories of CDG-modules with finitely generated or finitely presented underlying Γ -graded $B^{\#}$ -modules are denoted by $D^{abs}(B-mod_{fg}^{cdg})$ and $D^{abs}(mod_{fg}^{cdg}-B)$, respectively.

B.1.2. Derived functors of the second kind. Let k be a commutative ring and B be a small k-linear CDG-category. Assume that the Γ -graded category $B^{\#}$ is left Noetherian. Let L and M be left CDG-modules over B; suppose that the Γ -graded left $B^{\#}$ -module $L^{\#}$ underlying the CDG-module L over B is finitely generated.

As in [Polishchuk and Positselski 2012, §§ 2.1–2], we denote by $Z^0(B \operatorname{-mod}^{\operatorname{cdg}})$ and $Z^0(\operatorname{mod}^{\operatorname{cdg}}-B)$ the abelian categories of left and right CDG-modules over *B*. Let $Z^0(B \operatorname{-mod}_{\operatorname{fg}}^{\operatorname{cdg}}) \subset Z^0(B \operatorname{-mod}^{\operatorname{cdg}})$ and $H^0(B \operatorname{-mod}_{\operatorname{fg}}^{\operatorname{cdg}}) \subset H^0(B \operatorname{-mod}^{\operatorname{cdg}})$ denote the abelian and homotopy categories of left CDG-modules over *B* with finitely generated underlying Γ -graded $B^{\#}$ -modules, and $Z^0(\operatorname{mod}_{\operatorname{fp}}^{\operatorname{cdg}}-B) \subset Z^0(\operatorname{mod}^{\operatorname{cdg}}-B)$ and $H^0(\operatorname{mod}_{\operatorname{fp}}^{\operatorname{cdg}}-B) \subset H^0(\operatorname{mod}^{\operatorname{cdg}}-B)$ be the similar categories of right CDG-modules with finitely presented underlying Γ -graded modules.

Let J^{\bullet} be a right resolution of M in $Z^{0}(B \operatorname{-mod}^{\operatorname{cdg}})$ such that the Γ -graded left $B^{\#}$ -modules $J^{i\#}$ are injective, and let J be the total CDG-module of the complex of CDG-modules J^{\bullet} constructed by taking infinite direct sums along the diagonals. Then the complex $\operatorname{Tot}^{\oplus} \operatorname{Hom}^{B}(L, J^{\bullet})$ computing $\operatorname{Ext}_{B}^{II}(L, M)$ [loc. cit., Section 2.2] is isomorphic to the complex $\operatorname{Hom}^{B}(L, J)$ [loc. cit., formula (6)], which computes the *k*-modules of morphisms from *L* into M[*] in the coderived category $D^{\operatorname{co}}(B \operatorname{-mod}^{\operatorname{cdg}})$ [Positselski 2011b, Theorems 3.5(a) and 3.7]. Thus,

$$H^* \operatorname{Ext}_{B}^{II}(L, M) \simeq \operatorname{Hom}_{\operatorname{D^{co}}(B\operatorname{-mod}^{\operatorname{cdg}})}(L, M[*]).$$

Just as in [Polishchuk and Positselski 2012, Section 3.3], one can lift this isomorphism from the level of cohomology modules to that of the derived category D(k-mod) in the following way. Consider the functor

$$\operatorname{Hom}^{B}: H^{0}(B\operatorname{-mod}^{\operatorname{cdg}})^{\operatorname{op}} \times H^{0}(B\operatorname{-mod}^{\operatorname{cdg}}) \longrightarrow D(k\operatorname{-mod}),$$

and restrict it to the full subcategory $H^0(B \text{-mod}_{inj}^{cdg})$ in the second argument. This restriction factorizes through the coderived category $D^{co}(B \text{-mod}^{cdg})$ in the first argument. Taking into account [Positselski 2011b, Theorem 3.7], we obtain a right derived functor

$$D^{co}(B-mod^{cdg})^{op} \times D^{co}(B-mod^{cdg}) \longrightarrow D(k-mod)$$

Restricting to the full subcategory $D^{abs}(B - \text{mod}_{fg}^{cdg})^{op} \subset D^{co}(B - \text{mod}^{cdg})^{op}$ [loc. cit., Theorem 3.11.1] in the first argument, we have the derived functor

$$D^{abs}(B\operatorname{-mod}_{fg}^{cdg})^{op} \times D^{co}(B\operatorname{-mod}^{cdg}) \longrightarrow D(k\operatorname{-mod}).$$
(4)

The composition of this functor with the localization functors

$$Z^{0}(B\operatorname{-mod}_{\mathrm{fg}}^{\mathrm{cdg}}) \longrightarrow \mathrm{D}^{\mathrm{abs}}(B\operatorname{-mod}_{\mathrm{fg}}^{\mathrm{cdg}}) \quad \text{and} \quad Z^{0}(B\operatorname{-mod}^{\mathrm{cdg}}) \longrightarrow \mathrm{D}^{\mathrm{co}}(B\operatorname{-mod}^{\mathrm{cdg}})$$

agrees with the derived functor $\operatorname{Ext}_{B}^{II}$ where the former is defined.

Now assume that the Γ -graded category $B^{\#}$ is right coherent. Consider the functor [Polishchuk and Positselski 2012, formula (5)]

$$\otimes_B : H^0(\mathrm{mod}^{\mathrm{cdg}} \cdot B) \times H^0(B \cdot \mathrm{mod}^{\mathrm{cdg}}) \longrightarrow \mathrm{D}(k \cdot \mathrm{mod})$$

and restrict it to the Cartesian product of full subcategories

$$H^{0}(\operatorname{mod}_{\operatorname{fp}}^{\operatorname{cdg}}-B) \times H^{0}(B\operatorname{-mod}_{\operatorname{fl}}^{\operatorname{cdg}}) \subset H^{0}(\operatorname{mod}^{\operatorname{cdg}}-B) \times H^{0}(B\operatorname{-mod}^{\operatorname{cdg}})$$

Since the tensor product with a finitely presented Γ -graded right $B^{\#}$ -module commutes with infinite products of Γ -graded left $B^{\#}$ -modules, this restriction factorizes through the contraderived category $D^{ctr}(B - \text{mod}_{fl}^{cdg})$ in the second argument. Clearly, it also factorizes through the absolute derived category $D^{abs}(\text{mod}_{fp}^{cdg}-B)$ in the first argument.

By Remark 1.5 of the main body of this paper (see also [Positselski 2012, Proposition A.3.1(b)]), the natural functor $D^{ctr}(B - \text{mod}_{fl}^{cdg}) \rightarrow D^{ctr}(B - \text{mod}^{cdg})$ is an equivalence of triangulated categories. Hence we obtain the left derived functor

$$D^{abs}(mod_{fp}^{cdg}-B) \times D^{ctr}(B-mod^{cdg}) \longrightarrow D(k-mod).$$
(5)

Up to composing with the localization functors $Z^0(\text{mod}_{fp}^{cdg}-B) \rightarrow D^{abs}(\text{mod}_{fp}^{cdg}-B)$ and $Z^0(B-\text{mod}^{cdg}) \rightarrow D^{ctr}(B-\text{mod}^{cdg})$, this functor agrees with the derived functor Tor^{*B*,*II*} from [Polishchuk and Positselski 2012, Section 2.2] where the former is defined.

Indeed, let *N* be an object of $Z^0 \pmod{}_{fp}^{cdg} \cdot B$. Let P_{\bullet} be a left resolution of an object $M \in Z^0(B - \text{mod}^{cdg})$ by left CDG-modules over *B* with flat underlying Γ -graded $B^{\#}$ -modules, and let *P* be the total CDG-module of the complex P_{\bullet} constructed by taking infinite products along the diagonals. Then the complex $\text{Tot}^{\sqcap}(N \otimes_B P_{\bullet})$ computing $\text{Tor}^{B,II}(N, M)$ is isomorphic to the complex $N \otimes_B P$ computing the derived functor (5) on the objects *N* and *M*.

B.1.3. Comparison of the two theories. Let C be a small k-linear (Γ -graded) DG-category. The above constructions applicable to CDG-categories and CDG-modules over them can be also applied to DG-categories and DG-modules as a particular case. Following [Polishchuk and Positselski 2012], we denote the DG-categories of left and right DG-modules over C by C-mod^{dg} and mod^{dg}-C, and generally use the upper index "dg" instead of "cdg" in the notation related to DG-modules.

As in [loc. cit., Sections 2.1, 3.1 and 3.4], we denote by $H^0(C \operatorname{-mod}^{dg})_{inj}$ and $H^0(C \operatorname{-mod}^{dg})_{fl}$ the homotopy categories of h-injective and h-flat left DG-modules over *C*. The notation $H^0(C \operatorname{-mod}_{inj}^{dg})_{inj}$ and $H^0(C \operatorname{-mod}_{fl}^{dg})_{fl}$ stands for the full triangulated subcategories in $H^0(C \operatorname{-mod}^{dg})$ formed by h-injective DG-modules over *C* whose underlying Γ -graded $C^{\#}$ -modules are injective, or h-flat DG-modules whose underlying Γ -graded $C^{\#}$ -modules are flat, respectively. Finally, we let

1256

 $H^{0}(C\operatorname{-mod}_{\operatorname{fgp}}^{\operatorname{dg}})_{\operatorname{prj}} \subset H^{0}(C\operatorname{-mod}^{\operatorname{dg}})$ and $H^{0}(\operatorname{mod}_{\operatorname{fgp}}^{\operatorname{dg}}-C)_{\operatorname{fl}} \subset H^{0}(\operatorname{mod}^{\operatorname{dg}}-C)$ denote the full triangulated subcategories of h-projective left and h-flat right DG-modules whose underlying Γ -graded $C^{\#}$ -modules are projective and finitely generated.

Assume that the Γ -graded category $C^{\#}$ is left Noetherian. Let L be an object of $Z^0(C \operatorname{-mod}_{\mathrm{fg}}^{\mathrm{dg}})$. Given a left DG-module M over C, pick its injective resolution J^{\bullet} in the exact category $Z^0(C \operatorname{-mod}^{\mathrm{dg}})$ [loc. cit., Section 2.1]. Let $\operatorname{Tot}^{\oplus}(J^{\bullet}) \rightarrow$ $\operatorname{Tot}^{\sqcap}(J^{\bullet})$ be the natural closed morphism between the total DG-modules of the complex J^{\bullet} constructed by taking infinite direct sums and infinite products along the diagonals. Then the induced morphism of complexes of k-modules

$$\operatorname{Hom}^{C}(L, \operatorname{Tot}^{\oplus}(J^{\bullet})) \longrightarrow \operatorname{Hom}^{C}(L, \operatorname{Tot}^{\sqcap}(J^{\bullet}))$$

represents the comparison morphism $\operatorname{Ext}_{C}^{H}(L, M) \to \operatorname{Ext}_{C}(L, M)$ [loc. cit., formula (10)] in D(*k*-mod) between the two kinds of Ext objects for the DG-modules *L* and *M*.

Similarly, assume that the Γ -graded category $C^{\#}$ is right coherent. Let N be an object of $Z^0 \pmod{_{\text{fp}}}{C}$. Given a left DG-module M over C, pick its projective resolution P_{\bullet} in the exact category $Z^0(C \operatorname{-mod}^{\mathrm{dg}})$. Let $\operatorname{Tot}^{\oplus}(P_{\bullet}) \to \operatorname{Tot}^{\sqcap}(P_{\bullet})$ be the natural closed morphism between the total DG-modules of the complex P_{\bullet} constructed by taking infinite direct sums and infinite products along the diagonals. Then the induced morphism of complexes of k-modules

$$N \otimes_{\mathbb{C}} \operatorname{Tot}^{\oplus}(P_{\bullet}) \longrightarrow N \otimes_{\mathbb{C}} \operatorname{Tot}^{\sqcap}(P_{\bullet})$$

represents the comparison morphism $\text{Tor}^{C}(N, M) \to \text{Tor}^{C, II}(N, M)$ [loc. cit., formula (9)] in D(*k*-mod) between the two kinds of Tor objects for the DG-modules N and M.

Proposition A. Assume that the Γ -graded category $C^{\#}$ is left Noetherian. Let L be a left DG-module over C whose underlying Γ -graded left $C^{\#}$ -module is finitely generated, and let M be a left DG-module over C. Then the natural morphism $\operatorname{Ext}_{C}^{II}(L, M) \to \operatorname{Ext}_{C}(L, M)$ is an isomorphism provided that either

- (i) the object $M \in D^{co}(C \operatorname{-mod}^{dg})$ belongs to the image of the fully faithful functor $H^0(C \operatorname{-mod}_{\operatorname{inj}}^{\operatorname{dg}})_{\operatorname{inj}} \to D^{co}(C \operatorname{-mod}^{\operatorname{dg}}); or$
- (ii) the object $L \in D^{abs}(C \operatorname{-mod}^{dg})$ belongs to the image of the fully faithful functor $H^0(C \operatorname{-mod}_{fgp}^{dg})_{prj} \to D^{abs}(C \operatorname{-mod}_{fg}^{dg}).$

Proof. Let J^{\bullet} be an injective resolution of the DG-module M in the exact category $Z^{0}(C \operatorname{-mod}^{\mathrm{dg}})$. Then the natural morphism $M \to \operatorname{Tot}^{\oplus}(J^{\bullet})$ is always an isomorphism in $D^{\mathrm{co}}(C \operatorname{-mod}^{\mathrm{dg}})$ [Positselski 2011b, proof of Theorem 3.7], while the morphism $M \to \operatorname{Tot}^{\Box}(J^{\bullet})$ is an isomorphism in the conventional derived category $D(C \operatorname{-mod}^{\mathrm{dg}})$ [loc. cit., proofs of Theorems 1.4-5]. Furthermore, one has $\operatorname{Tot}^{\oplus}(J^{\bullet}) \in H^{0}(C \operatorname{-mod}^{\mathrm{dg}}_{\mathrm{inj}})$ and $\operatorname{Tot}^{\Box}(J^{\bullet}) \in H^{0}(C \operatorname{-mod}^{\mathrm{dg}}_{\mathrm{inj}})_{\mathrm{inj}}$.

Part (i): The functor is fully faithful by [loc. cit., Theorem 3.5(a) and Lemma 1.3]. According to formula (4) from Section B.1.2 and [Polishchuk and Positselski 2012, Section 3.1], both kinds of Ext involved are well-defined as functors of the argument $M \in D^{co}(C \operatorname{-mod}^{dg})$. Hence one can assume $M \in H^0(C \operatorname{-mod}^{dg})_{inj}$. Then both morphisms $M \to \operatorname{Tot}^{\oplus}(J^{\bullet})$ and $M \to \operatorname{Tot}^{\square}(J^{\bullet})$ are homotopy equivalences by semiorthogonality; hence so is the morphism $\operatorname{Tot}^{\oplus}(J^{\bullet}) \to \operatorname{Tot}^{\square}(J^{\bullet})$ and the assertion follows.

Part (ii): In view of the first paragraph of this proof, a cone *K* of the morphism $\operatorname{Tot}^{\oplus}(J^{\bullet}) \to \operatorname{Tot}^{\sqcap}(J^{\bullet})$ in $H^{0}(C\operatorname{-mod}^{\operatorname{dg}})$ is an acyclic DG-module over *C* whose underlying Γ -graded $C^{\#}$ -module is injective. Hence the complex of morphisms $\operatorname{Hom}^{C}(-, K)$ is a well-defined functor $\operatorname{D}^{\operatorname{abs}}(C\operatorname{-mod}_{\operatorname{fg}}^{\operatorname{dg}})^{\operatorname{op}} \to \operatorname{D}(k\operatorname{-mod})$ annihilating $H^{0}(C\operatorname{-mod}_{\operatorname{fgp}}^{\operatorname{dg}})_{\operatorname{prj}}$.

Proposition B. Assume that the Γ -graded category $C^{\#}$ is right coherent. Let N be a right DG-module over C whose underlying Γ -graded right $C^{\#}$ -module is finitely presented, and let M be a left DG-module over C. Then the natural morphism $\operatorname{Tor}^{C}(N, M) \to \operatorname{Tor}^{C, II}(N, M)$ is an isomorphism provided that either

- (i) there exists a closed morphism $P \rightarrow M$ into M from a DG-module $P \in H^0(C \operatorname{-mod}_{\mathrm{fl}}^{\mathrm{dg}})_{\mathrm{fl}}$ with a cone contraacyclic with respect to $C \operatorname{-mod}_{\mathrm{fl}}^{\mathrm{dg}}$ or completely acyclic with respect to $C \operatorname{-mod}_{\mathrm{fl}}^{\mathrm{dg}}$ (see [Polishchuk and Positselski 2012, Sections 3.2 and 4.7]); or
- (ii) the object $N \in D^{abs}(\text{mod}^{dg}-C)$ belongs to the image of the fully faithful functor $H^{0}(\text{mod}_{fgp}^{dg}-C)_{fl} \to D^{abs}(\text{mod}_{fp}^{dg}-C).$

Proof. Let P_{\bullet} be a projective resolution of the DG-module M in the exact category $Z^0(C \operatorname{-mod}^{\mathrm{dg}})$. Then the natural morphism $\operatorname{Tot}^{\Box}(P_{\bullet}) \to M$ is always an isomorphism in $\operatorname{D}^{\operatorname{ctr}}(C \operatorname{-mod}^{\operatorname{dg}})$ [Positselski 2011b, proof of Theorem 3.8], while the morphism $\operatorname{Tot}^{\oplus}(P_{\bullet}) \to M$ is an isomorphism in $\operatorname{D}(C \operatorname{-mod}^{\operatorname{dg}})$ [loc. cit., proof of Theorem 1.4]. Furthermore, one has $\operatorname{Tot}^{\Box}(P_{\bullet}) \in H^0(C \operatorname{-mod}^{\operatorname{dg}}_{\mathrm{fl}})$ and $\operatorname{Tot}^{\oplus}(P_{\bullet}) \in H^0(C \operatorname{-mod}^{\operatorname{dg}}_{\mathrm{fl}})_{\mathrm{fl}}$.

Part (i): Acyclic DG-modules in the second argument are annihilated by the functor Tor^{*C*} by [Polishchuk and Positselski 2012, Section 3.1], while contraacyclic DG-modules in the second argument are annihilated by the functor Tor^{*C*, *II*} (*N*, –) according to the formula (5). The latter also applies to DG-modules completely acyclic with respect to *C*-mod^{dg}_{fl} since the functor of a tensor product with a finitely presented DG-module preserves infinite direct sums and products. So one can replace *M* with *P* and assume that $M \in H^0(C - \text{mod}_{fl}^{dg})_{fl}$.

Then a cone of the morphism $\operatorname{Tot}^{\sqcap}(P_{\bullet}) \to M$ is contraacyclic with respect to C-mod^{dg} with a flat underlying Γ -graded $C^{\#}$ -module, and hence also contraacyclic

1258

with respect to $C \operatorname{-mod}_{\mathrm{fl}}^{\mathrm{dg}}$. On the other hand, a cone of the morphism $\operatorname{Tot}^{\oplus}(P_{\bullet}) \to M$ is acyclic and h-flat. It follows that the functor $N \otimes_C -$ transforms both of these morphisms, and therefore also the morphism $\operatorname{Tot}^{\oplus}(P_{\bullet}) \to \operatorname{Tot}^{\Box}(P_{\bullet})$, into quasi-isomorphisms of complexes of *k*-modules.

Part (ii): A cone *K* of the morphism $\operatorname{Tot}^{\oplus}(P_{\bullet}) \to \operatorname{Tot}^{\sqcap}(P_{\bullet})$ in $H^{0}(C\operatorname{-mod}^{\mathrm{dg}})$ is an acyclic DG-module over *C* whose underlying Γ -graded $C^{\#}$ -module is flat. Hence the tensor product $-\otimes_{C} K$ is a well-defined functor $\operatorname{D}^{\mathrm{abs}}(\operatorname{mod}_{\mathrm{fp}}^{\mathrm{dg}}-C) \to \operatorname{D}(k\operatorname{-mod})$ annihilating $H^{0}(\operatorname{mod}_{\mathrm{fp}}^{\mathrm{dg}}-C)_{\mathrm{fl}}$.

In particular, assuming that the category $C^{\#}$ is left Noetherian, the natural morphism $\operatorname{Ext}_{C}^{II}(L, M) \to \operatorname{Ext}_{C}(L, M)$ is an isomorphism for all $L \in C\operatorname{-mod}_{\mathrm{fg}}^{\mathrm{dg}}$ and $M \in C\operatorname{-mod}^{\mathrm{dg}}$ provided that the Verdier localization functor $D^{\mathrm{co}}(C\operatorname{-mod}^{\mathrm{dg}}) \to$ $D(C\operatorname{-mod}^{\mathrm{dg}})$ is an equivalence of triangulated categories. Assuming that the category $C^{\#}$ is right coherent, the natural morphism $\operatorname{Tor}^{C}(N, M) \to \operatorname{Tor}^{C, II}(N, M)$ is an isomorphism for all $N \in \operatorname{mod}_{\mathrm{fp}}^{\mathrm{dg}}$ -C and $M \in C\operatorname{-mod}^{\mathrm{dg}}$ provided that the Verdier localization functor $D^{\mathrm{ctr}}(C\operatorname{-mod}^{\mathrm{dg}}) \to D(C\operatorname{-mod}^{\mathrm{dg}})$ is an equivalence of categories, or alternatively, that any acyclic DG-module from $C\operatorname{-mod}_{\mathrm{fl}}^{\mathrm{dg}}$ is completely acyclic with respect to $C\operatorname{-mod}_{\mathrm{fl}}^{\mathrm{dg}}$.

B.1.4. Comparison for the DG-category of CDG-modules. Let B be a k-linear CDG-category and $C = \text{mod}_{fgp}^{cdg}$ -B be the DG-category of right CDG-modules over B whose underlying Γ -graded B[#]-modules are projective and finitely generated. The DG-categories of (left or right) CDG-modules over B and DG-modules over C are naturally equivalent [Polishchuk and Positselski 2012, Sections 1.5 and 2.6] (as are the categories of Γ -graded modules over B[#] and C[#]).

Following [loc. cit., Section 3.5], we denote by M_C the DG-module over C corresponding to a CDG-module M over B.

Let k^{\vee} be an injective cogenerator of the abelian category of *k*-modules. Introduce the notation B-mod^{cdg}_{prj} $\subset B$ -mod^{cdg} for the DG-category of left CDG-modules over *B* with projective underlying Γ -graded $B^{\#}$ -modules. The results below in this section are to be compared with those from [loc. cit., Sections 3.5 and 4.7].

Proposition A'. Assume that the Γ -graded category $B^{\#}$ is left Noetherian. Let L be a left CDG-module over B whose underlying Γ -graded left $B^{\#}$ -module $L^{\#}$ is finitely generated, and let M be a left CDG-module over B. Then the natural morphism $\operatorname{Ext}_{C}^{II}(L_{C}, M_{C}) \to \operatorname{Ext}_{C}(L_{C}, M_{C})$ is an isomorphism provided that either

- (i) the object M belongs to the minimal triangulated subcategory of D^{co}(B-mod^{cdg}) containing the objects Hom_k(F, k[∨]) for all F ∈ H⁰(mod^{cdg}_{fgp}-B) and closed under infinite products; or
- (ii) the object L belongs to the minimal thick subcategory of $D^{abs}(B-mod_{fg}^{cdg})$ containing the image of $H^0(B-mod_{fgp}^{cdg})$.

Proof. Part (i): The equivalence of categories

$$H^{0}(C\operatorname{-mod}_{\operatorname{inj}}^{\operatorname{dg}})_{\operatorname{inj}} \simeq D(C\operatorname{-mod}^{\operatorname{dg}})$$

makes the embedding functor $H^0(C \operatorname{-mod}_{\operatorname{inj}}^{\operatorname{dg}})_{\operatorname{inj}} \to D^{\operatorname{co}}(C \operatorname{-mod}^{\operatorname{dg}})$ right adjoint to the localization functor $D^{\operatorname{co}}(C \operatorname{-mod}^{\operatorname{dg}}) \to D(C \operatorname{-mod}^{\operatorname{dg}})$. It follows that the functor $H^0(C \operatorname{-mod}_{\operatorname{inj}}^{\operatorname{dg}})_{\operatorname{inj}} \to D^{\operatorname{co}}(C \operatorname{-mod}^{\operatorname{dg}})$ preserves infinite products (also, all infinite products exist in the coderived category since it is compactly generated [Positselski 2011b, Theorem 3.11.2]). Since the category $H^0(C \operatorname{-mod}_{\operatorname{inj}}^{\operatorname{dg}})_{\operatorname{inj}}$ is the minimal triangulated subcategory of $H^0(C \operatorname{-mod}^{\operatorname{dg}})$ containing the objects $\operatorname{Hom}_k(F_C, k^{\vee})$ and closed under infinite products [loc. cit., Theorem 1.5], the assertion follows from Proposition A(i).

Part (ii): The equivalence of absolute derived categories

$$D^{abs}(B-mod_{fg}^{cdg}) \simeq D^{abs}(C-mod_{fg}^{dg})$$

takes objects of the full subcategory $H^0(B \operatorname{-mod}_{fgp}^{cdg}) \subset D^{abs}(B \operatorname{-mod}_{fg}^{cdg})$ to representable (and, consequently, perfect and h-projective) DG-modules in

$$H^0(C\operatorname{-mod}_{\operatorname{fgp}}^{\operatorname{dg}}) \subset \mathrm{D}^{\operatorname{abs}}(C\operatorname{-mod}_{\operatorname{fg}}^{\operatorname{dg}})$$

so it remains to apply Proposition A(ii).

Proposition B'. Assume that the Γ -graded category $B^{\#}$ is right coherent. Let N be a right CDG-module over B whose underlying Γ -graded right $B^{\#}$ -module $N^{\#}$ is finitely presented, and let M be a left CDG-module over B. Then the natural morphism $\operatorname{Tor}^{C}(N_{C}, M_{C}) \rightarrow \operatorname{Tor}^{C, II}(N_{C}, M_{C})$ is an isomorphism provided that either

(i) the object M belongs to the minimal triangulated subcategory of

$$H^{0}(B\operatorname{-mod}_{\operatorname{prj}}^{\operatorname{cdg}}) \subset \mathrm{D}^{\operatorname{ctr}}(B\operatorname{-mod}^{\operatorname{cdg}})$$

containing the image of $H^0(B\operatorname{-mod}_{\operatorname{fgp}}^{\operatorname{cdg}})$ and closed under infinite direct sums; or

(ii) the object N belongs to the minimal thick subcategory of $D^{abs}(mod_{fp}^{cdg}-B)$ containing the image of $H^0(mod_{fgp}^{cdg}-B)$.

Proof. Similar to that of Proposition A' and based on Proposition B.

Now assume that the commutative ring k has finite weak homological dimension and all the Γ -graded k-modules of morphisms in the category $B^{\#}$ are flat. Clearly, the DG-categories of left and right CDG-modules over the CDG-category $B \otimes_k B^{\text{op}}$ are naturally equivalent, as are the DG-categories of left and right DG-modules over the DG-category $C \otimes_k C^{\text{op}}$. The DG-category of CDG-modules over $B \otimes_k B^{\text{op}}$ is also naturally equivalent to the DG-category of DG-modules over $C \otimes_k C^{\text{op}}$

1260

[Polishchuk and Positselski 2012, Section 2.6]. As above, we denote by M_C the DG-module over $C \otimes_k C^{\text{op}}$ corresponding to a CDG-module M over $B \otimes_k B^{\text{op}}$.

To any left CDG-module *G* and right CDG-module *F* over *B*, one can assign the left CDG-module $G \otimes_k F$ and the right CDG-module $F \otimes_k G$ (corresponding to each other under the above equivalence) over the CDG-category $B \otimes_k B^{\text{op}}$. There are also the natural *diagonal* CDG-module *B* over $B \otimes_k B^{\text{op}}$ and DG-module *C* over $C \otimes_k C^{\text{op}}$ [loc. cit., Section 2.4]; these also correspond to each other with respect to the above equivalence of DG-categories.

For any DG-module M_C over $C \otimes_k C^{\text{op}}$, we are interested in the comparison morphisms between the two kinds of Hochschild cohomology $HH^{II,*}(C, M_C) \rightarrow$ $HH^*(C, M_C)$ and Hochschild homology $HH_*(C, M_C) \rightarrow HH_*^{II}(C, M_C)$ [loc. cit., formula (23)].

Proposition C. Assume that the Γ -graded category $B^{\#} \otimes_k B^{\# op}$ is Noetherian and the diagonal Γ -graded module $B^{\#}$ over it is finitely generated. Let M be a CDG-module over $B \otimes_k B^{op}$. Then the natural morphism $HH^{II,*}(C, M_C) \rightarrow$ $HH^*(C, M_C)$ is an isomorphism provided that either

(i) the object M belongs to the minimal triangulated subcategory of

$$D^{co}(B \otimes_k B^{op}-mod^{cdg})$$

containing the CDG-modules $Hom_k(F \otimes_k G, k^{\vee})$ for all $F \in H^0(\text{mod}_{fgp}^{cdg}-B)$ and $G \in H^0(B\text{-mod}_{fgp}^{cdg})$ and closed under infinite products; or

(ii) the diagonal CDG-module B over $B \otimes_k B^{\text{op}}$ belongs to the minimal thick subcategory of

 $D^{abs}(B \otimes_k B^{op}-mod_{fg}^{cdg})$

containing the CDG-modules $G \otimes_k F$ for all $F \in H^0(\text{mod}_{fgp}^{cdg} - B)$ and $G \in H^0(B - \text{mod}_{fgp}^{cdg})$.

Proposition D. Assume that the Γ -graded category $B^{\#} \otimes_k B^{\# op}$ is coherent and the diagonal Γ -graded module $B^{\#}$ over it is finitely presented. Let M be a CDG-module over $B \otimes_k B^{op}$. Then the natural morphism $HH_*(C, M_C) \to HH_*^{II}(C, M_C)$ is an isomorphism provided that either

(i) the object M belongs to the minimal triangulated subcategory of

$$H^{0}(B\operatorname{-mod}_{\operatorname{prj}}^{\operatorname{cdg}}) \subset \mathrm{D}^{\operatorname{ctr}}(B \otimes_{k} B^{\operatorname{op}}\operatorname{-mod}^{\operatorname{cdg}})$$

containing the CDG-modules $G \otimes_k F$ for all $F \in H^0(\text{mod}_{fgp}^{cdg} \cdot B)$ and $G \in H^0(B \cdot \text{mod}_{fgp}^{cdg})$ and closed under infinite direct sums; or

(ii) the diagonal CDG-module B over $B \otimes_k B^{\text{op}}$ belongs to the minimal thick subcategory of $D^{\text{abs}}(B \otimes_k B^{\text{op}}-\text{mod}_{fg}^{\text{cdg}})$ containing the CDG-modules $G \otimes_k F$ for all $F \in H^0(\text{mod}_{fgp}^{\text{cdg}}-B)$ and $G \in H^0(B-\text{mod}_{fgp}^{\text{cdg}})$.

Proofs of Propositions C and D. Similar to the proofs of Propositions A' and B'. \Box

In particular, assume that the Γ -graded category $B^{\#} \otimes_k B^{\# op}$ is Noetherian and the diagonal Γ -graded module $B^{\#}$ over it is finitely generated. Suppose that the diagonal CDG-module B over $B \otimes_k B^{op}$ belongs to the minimal thick subcategory of $D^{abs}(B \otimes_k B^{op}-mod_{fg}^{cdg})$ containing the CDG-modules $G \otimes_k F$ for all $F \in$ $H^0(mod_{fgp}^{cdg}-B)$ and $G \in H^0(B-mod_{fgp}^{cdg})$. Then, according to [Polishchuk and Positselski 2012, formulas (44-45) in Section 2.6] and parts (ii) of Propositions C and D, there are natural isomorphisms

$$HH^{*}(C, M_{C}) \simeq HH^{II,*}(C, M_{C}) \simeq HH^{II,*}(B, M),$$
 (6)

$$HH_*(C, M_C) \simeq HH_*^{II}(C, M_C) \simeq HH_*^{II}(B, M)$$
(7)

for any CDG-module M over $B \otimes_k B^{\text{op}}$. Specializing to the case of the diagonal CDG-module M = B and DG-module $M_C = C$, we obtain

$$HH^*(C) \simeq HH^{II,*}(C) \simeq HH^{II,*}(B),$$

$$HH_*(C) \simeq HH^{II}_*(C) \simeq HH^{II}_*(B).$$
(8)

B.1.5. Locally free matrix factorizations. Let k be a regular commutative Noetherian ring of finite Krull dimension and X be an affine scheme of finite type over Spec k. Let $w \in \mathcal{O}(X)$ be a global regular function on X. Consider the $\mathbb{Z}/2$ -graded CDG-algebra B over k with $B^0 = \mathcal{O}(X)$, $B^1 = 0$, d = 0, and $h = -w \in B^0$. We will find it convenient to denote the CDG-algebra B simply by (X, h) = (X, -w) (cf. Section 2.2 of the main body of this paper).

Then $C = \text{mod}_{fgp}^{cdg}$ -*B* is the $\mathbb{Z}/2$ -graded DG-category of locally free matrix factorizations of finite rank of the potential *w* on *X*. Furthermore, one has $B \otimes_k B^{op} = (X \times_k X, w_2 - w_1)$, where $w_i = p_i^* w \in \mathcal{O}(X \times_k X)$, i = 1, 2, and $p_i : X \times_k X \to X$ denote the coordinate projections. Let $\Delta : X \to X \times_k X$ be the diagonal embedding and $\Delta_* \mathcal{O}_X$ be the corresponding coherent sheaf on $X \times_k X$.

Consider the coherent matrix factorization of the potential $w_2 - w_1$ on $X \times X$ whose even-degree component is the sheaf $\Delta_* \mathcal{O}_X$, while the odd-degree component vanishes. We will denote this "diagonal" matrix factorization simply by $\Delta_* \mathcal{O}_X \in H^0((X \times_k X, w_2 - w_1) - \text{mod}_{fg}^{cdg})$. Applying the machinery of the previous sections leads to the following result (cf. [Polishchuk and Positselski 2012, Sections 4.8–4.10]).

1262

Corollary B.1.5. Suppose that the diagonal matrix factorization $\Delta_*\mathcal{O}_X$ belongs to the minimal thick subcategory of $D^{abs}((X \times_k X, w_2 - w_1) \operatorname{-mod}_{fg}^{cdg})$ containing the external tensor products of locally free matrix factorizations of finite rank $p_1^*G \otimes_k p_2^*F$ for all $G \in H^0((X, -w) \operatorname{-mod}_{fgp}^{cdg})$ and $F \in H^0((X, w) \operatorname{-mod}_{fgp}^{cdg})$. Then the natural isomorphisms (8) hold for the CDG-algebra B = (X, w) and the DG-category of locally free matrix factorizations $C = \operatorname{mod}_{fgp}^{cdg} - B$.

Notice that the condition under which the conclusion of Corollary B.1.5 has been proven is a rather strong one, particularly when X is not assumed to be a regular scheme. Then it is not even clear when or why the diagonal matrix factorization $\Delta_* \mathcal{O}_X$ should belong to the thick envelope of the full triangulated subcategory of locally free matrix factorizations

$$H^{0}((X \times_{k} X, w_{2} - w_{1}) \operatorname{-mod}_{\operatorname{fgp}}^{\operatorname{cdg}}) \subset \mathsf{D}^{\operatorname{abs}}((X \times_{k} X, w_{2} - w_{1}) \operatorname{-mod}_{\operatorname{fg}}^{\operatorname{cdg}})$$

on $X \times_k X$, let alone to the thick subcategory generated by external tensor products of locally free matrix factorizations from the two copies of X.

B.1.6. Smooth stratifications. A scheme X of finite type over a field k is said to admit a smooth stratification [Efimov 2013] if it can be presented as a disjoint union of its locally closed subsets $X = \bigsqcup_{\alpha} S_{\alpha}$ so that each S_{α} , when endowed with the structure of a reduced locally closed subscheme in X, becomes a smooth scheme over k. In particular, every scheme of finite type over a perfect field k admits a smooth stratification, as any regular scheme of finite type over a perfect field is smooth over it [Grothendieck 1967, Corollaires 17.15.2 and 17.15.13]. Notice that a scheme of finite type over a field admits a smooth stratification if and only if its maximal reduced closed subscheme does.

The definition of a *regular stratification* of a Noetherian scheme is similar, except that the strata S_{α} are only required to be regular schemes in their reduced locally closed subscheme structures. Any scheme of finite type over a field admits a regular stratification [Grothendieck 1965, Scholie 7.8.3(iii)–(iv) and Proposition 7.8.6(i)].

Let X be a smooth affine scheme over a field k and $w \in \mathcal{O}(X)$ be a regular function on X. Set $X_0 = \{w = 0\} \subset X$ to be the zero locus of w. The following result is a slight generalization of [Polishchuk and Positselski 2012, Corollary 4.8.A] based on the above definitions.

Corollary B.1.6. Assume that there exists a closed subscheme $Z \subset X$ such that $w : X \setminus Z \to \mathbb{A}^1_k$ is a smooth morphism, $w|_Z = 0$, and the scheme Z admits a smooth stratification over k. Then the conditions of Corollary B.1.5 are satisfied, so its conclusions apply.

Proof. According to the argument in [Polishchuk and Positselski 2012, Section 4.8], it suffices to show that the bounded derived category of coherent sheaves on $Z \times Z$

is generated by external tensor products of coherent sheaves on the two Cartesian factors. This is a particular case of the following lemma. \Box

Lemma B.1.6. Let Z' and Z" be schemes of finite type over a field k. Assume that the scheme Z' admits a smooth stratification. Then the bounded derived category of coherent sheaves $D^b((Z' \times Z'')$ -coh) on the Cartesian product $Z' \times_k Z''$ coincides with its minimal thick subcategory containing the external tensor products $\mathcal{K}' \otimes_k \mathcal{K}''$ of coherent sheaves on \mathcal{K}' on Z' and \mathcal{K}'' on Z''.

Proof. One proceeds by induction on the total number of strata in a smooth stratification of Z' and a regular stratification of Z''. Clearly, one can replace Z' and Z'' with their maximal reduced closed subschemes. Now if S_{α_0} is an open stratum in Z' and T_{β_0} is an open stratum in Z'', then S_{α_0} is smooth as an open subscheme in Z' and T_{β_0} is regular as an open subscheme in Z'', while the induction assumption applies to $(Z' \setminus S_{\alpha_0}) \times_k Z''$ and $Z' \times_k (Z'' \setminus T_{\beta_0})$. The scheme $S_{\alpha_0} \times_k T_{\beta_0}$ is regular since it is smooth over a regular scheme. The rest of the argument is based on [Orlov 2011, Proposition 2.7] and follows the lines of [Lin and Pomerleano 2013, proof of Theorem 3.7].

B.2. *Coherent matrix factorizations.* In this section, we return to the notation system typical for the main body of this paper. The notion of a critical value of a regular function on a singular variety is defined in Section B.2.1. In Section B.2.2 we show that the external tensor product of coherent matrix factorizations is a fully faithful functor between the absolute derived categories and provide a sufficient condition for the pretriangulated extension of its DG-category version to be a quasiequivalence. The Hochschild cohomology of the DG-category corresponding to the absolute derived category of coherent matrix factorizations of a potential having no critical values but zero is computed in Section B.2.4.

The notion of cotensor product of complexes of quasicoherent sheaves and quasicoherent matrix factorizations is discussed in Sections B.2.5–B.2.6 and used in order to compute the Hochschild homology of the (same) DG-category of coherent matrix factorizations in Section B.2.7. The direct sum formula for the Hochschild (co)homology of the DG-categories of coherent matrix factorizations of a potential with several critical values is established in Section B.2.8.

In some sense, the results of this section (as compared to those of Section B.1) suggest that the DG-category corresponding to the absolute derived category of coherent matrix factorizations on a singular variety may be better behaved than the similar category of locally free matrix factorizations of finite rank. Other (and in some way related) arguments in support of the same conclusion are provided by the results of the papers [Lunts 2010; Efimov 2013] showing that the DG-category corresponding to the absolute derived category of coherent matrix factorizations is

smooth (and even homotopically finitely presented), under suitable conditions on the field k. (Cf. the counterexample in Section 3.3.)

B.2.1. *Noncritical functions.* Let k be a field and X be a scheme of finite type over Spec k. Let $f \in \mathcal{O}(X)$ be a global regular function on X.

Let *Y* be a scheme of finite type over Spec *k* and $g \in \mathcal{O}(Y)$ be a global regular function. Let $p_1: X \times_k Y \to X$ and $p_2: X \times_k Y \to Y$ be the natural projections. Consider the regular function $f_1 + g_2 = p_1^* f + p_2^* g$ on $X \times_k Y$.

Suppose that $f: X \to \mathbb{A}_k^1$ is a flat morphism from X to the affine line (when k is algebraically closed, this means that the function f - c is a local nonzero-divisor on X for every $c \in k$). Then the morphism $f_1 + g_2 : X \times_k Y \to \mathbb{A}_k^1$ is also flat as it is the composition of two flat morphisms

$$X \times_k Y \longrightarrow \mathbb{A}^1_k \times_k Y \longrightarrow \mathbb{A}^1_k$$

(the former morphism being flat since the morphism $f: X \to \mathbb{A}^1_k$ is and the latter one because the polynomial x + g does not divide zero in B[x] for any commutative ring *B* and element $g \in B$). In particular, it follows that the function $f_1 + g_2$ is a local nonzero-divisor on the Cartesian product $X \times_k Y$.

A function $f \in \mathcal{O}(X)$ is said to be *noncritical* (or to *have no critical values*) if for any regular function $g \in \mathcal{O}(Y)$ on a scheme Y of finite type over Spec k the absolute derived category of coherent matrix factorizations $D^{abs}((X \times_k Y, \mathcal{O}, f_1 + g_2)\text{-coh})$ vanishes (i.e., is equivalent to the zero category). According to Remark 1.3 and Theorem 1.10(b), this condition is local in both X and Y.

Therefore, given a scheme X of finite type over Spec k and a regular function $f \in \mathcal{O}(X)$, there is a unique maximal open subscheme $X'_f \subset X$ where the function f is noncritical. We will see below that the open subscheme X'_f is always dense in X if the morphism $f : X \to \mathbb{A}^1_k$ is flat and the field k has zero characteristic.

Similarly, there is a unique maximal open subscheme $\mathbb{A}_{k,f}^1 \subset \mathbb{A}_k^1$ such that the restriction of f to its full preimage in X is noncritical. The scheme $\mathbb{A}_{k,f}^1$ is always nonempty if the field k has zero characteristic. The points in the complement $\mathbb{A}_k^1 \setminus \mathbb{A}_{k,f}^1$ are called the *critical values* of f. In particular, one says that f has no critical values but zero if the restriction of f to $f^{-1}(\mathbb{A}_k^1 \setminus \{0\}) \subset X$ is noncritical.

Notice that when the schemes X and Y are separated and the morphism of schemes $f : X \to \mathbb{A}^1_k$ is flat, the category $D^{abs}((X \times_k Y, \mathcal{O}, f_1 + g_2)\text{-coh})$ is equivalent to the triangulated category $D^{b}_{Sing}(\{f_1 + g_2 = 0\}/X \times_k Y)$ of relative singularities of the zero locus of the function $f_1 + g_2$ on $X \times_k Y$ (see Theorem 2.7).

Remark B.2.1. It would be interesting to have a geometric characterization of noncriticality of functions on singular schemes. For example, how does our definition of noncriticality relate to the condition that the differential of f at every closed point $x \in X$ be a nonzero element of the Zariski cotangent space T_x^*X ? We do *not* know this; cf. the smooth stratification approach below. 1266

Lemma B.2.1. Let $X = \bigsqcup_{\alpha} S_{\alpha}$ be a scheme of finite type over Spec k presented as a disjoint union of its locally closed subsets, endowed with their reduced locally closed subscheme structures. Let \mathcal{L} be a line bundle on X and $w \in$ $\mathcal{L}(X)$ be its global section. In this setting, if the absolute derived categories $\mathsf{D}^{\mathsf{abs}}((S_{\alpha}, \mathcal{L}|_{S_{\alpha}}, w|_{S_{\alpha}})$ -coh) vanish for all α , then so does the absolute derived category $\mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w)$ -coh).

Proof. Proceeding by induction on the number of strata in the stratification S_{α} , it suffices to consider the case when there are only two of them, namely, a closed subset $S \subset X$ and its open complement $X \setminus S$. One can also replace X with its maximal reduced closed subscheme. Then the desired assertion follows from Theorem 1.10(b) since the triangulated category $D^{abs}((X, \mathcal{L}, w)\text{-coh}_S)$ is generated by the image of the natural functor $D^{abs}((S, \mathcal{L}|_S, w|_S)\text{-coh}) \to D^{abs}((X, \mathcal{L}, w)\text{-coh}_S)$.

Proposition B.2.1. Let X be a scheme of finite type over Spec k and $f \in O(X)$ be a regular function on X. Let $X = \bigsqcup_{\alpha} S_{\alpha}$ be a smooth stratification of the scheme X over k (see Section B.1.6) such that the morphisms of schemes $f|_{S_{\alpha}} : S_{\alpha} \to \mathbb{A}_{k}^{1}$ are smooth for all α . Then the function f is noncritical on X.

Proof. Let *Y* be a scheme of finite type over Spec *k* and $g \in \mathcal{O}(Y)$ be a regular function. We have to show that the triangulated category $D^{abs}((X \times_k Y, \mathcal{O}, f_1 + g_2)\text{-coh})$ vanishes. Choosing a stratification of *Y* by regular locally closed subschemes and applying Lemma B.2.1, one can assume that *X* is smooth over *k* and *Y* is regular.

Then the scheme $X \times_k Y$ is also regular, the derivative of the function $f_1 + g_2 \in \mathcal{O}(X \times_k Y)$, viewed as an element of the Zariski cotangent space, does not vanish at any points where the function itself does (and, in a sense, at any other closed points, too), and it follows that the zero locus of $f_1 + g_2$ in $X \times_k Y$ is also a regular scheme. It remains to use Theorem 2.7 (or [Orlov 2012, Theorem 3.5] and Corollary 2.4(c)).

It follows from Proposition B.2.1 that for any scheme of finite type X with a smooth stratification $X = \bigsqcup_{\alpha} S_{\alpha}$ over Spec k and any regular function $f \in \mathcal{O}(X)$, the set of critical values of the function f on X is contained in the union of the sets of critical values of the functions $f|_{S_{\alpha}}$. In particular, if the characteristic of k is zero, then all of these sets are finite.

B.2.2. *External tensor products.* Let X' and X'' be separated schemes of finite type over a field k, and let $w' \in \mathcal{O}(X')$ and $w'' \in \mathcal{O}(X'')$ be regular functions. Let $X' \times_k X''$ be the Cartesian product, p_1 and p_2 be its natural projections onto the factors X' and X'', and $w'_1 + w''_2 = p_1^* w' + p_2^* w''$ be the related regular function

on $X' \times_k X''$. Then there is the external tensor product functor

$$\otimes_{k} : \mathsf{D}^{\mathsf{co}}((X', \mathcal{O}, w')\operatorname{-qcoh}) \times \mathsf{D}^{\mathsf{co}}((X'', \mathcal{O}, w'')\operatorname{-qcoh}) \longrightarrow \mathsf{D}^{\mathsf{co}}((X' \times_{k} X'', \mathcal{O}, w'_{1} + w''_{2})\operatorname{-qcoh}),$$
(9)

which restricts to the similar functor

$$\otimes_{k} : \mathsf{D}^{\mathsf{abs}}((X', \mathcal{O}, w')\operatorname{-coh}) \times \mathsf{D}^{\mathsf{abs}}((X'', \mathcal{O}, w'')\operatorname{-coh}) \longrightarrow \mathsf{D}^{\mathsf{abs}}((X' \times_{k} X'', \mathcal{O}, w'_{1} + w''_{2})\operatorname{-coh})$$
(10)

on coherent matrix factorizations.

Proposition B.2.2. Let \mathcal{K}' and \mathcal{M}' be coherent matrix factorizations of the potential w' on the scheme X', and let \mathcal{K}'' and \mathcal{M}'' be coherent matrix factorizations of the potential w'' on the scheme X''. Then the natural map of $\mathbb{Z}/2$ -graded k-vector spaces of morphisms

$$\operatorname{Hom}_{\mathsf{D}^{\mathsf{abs}}((X',\mathcal{O},w')\operatorname{-coh})}(\mathcal{K}',\mathcal{M}'[*]) \otimes_{k} \operatorname{Hom}_{\mathsf{D}^{\mathsf{abs}}((X'',\mathcal{O},w'')\operatorname{-coh})}(\mathcal{K}'',\mathcal{M}''[*]) \longrightarrow \operatorname{Hom}_{\mathsf{D}^{\mathsf{abs}}((X'\times_{k}X'',\mathcal{O},w'_{1}+w''_{2})\operatorname{-coh})}(\mathcal{K}'\otimes_{k}\mathcal{K}'',\mathcal{M}'\otimes_{k}\mathcal{M}''[*])$$
(11)

induced by the additive functor of two arguments (10) is an isomorphism.

Proof. By Proposition 1.5(d), it suffices to show that the natural map

$$\operatorname{Hom}_{\mathsf{D}^{\mathrm{co}}((X',\mathcal{O},w')\operatorname{-qcoh})}(\mathcal{K}',\mathcal{M}'[*]) \otimes_{k} \operatorname{Hom}_{\mathsf{D}^{\mathrm{co}}((X'',\mathcal{O},w'')\operatorname{-qcoh})}(\mathcal{K}'',\mathcal{M}''[*]) \longrightarrow \operatorname{Hom}_{\mathsf{D}^{\mathrm{co}}((X'\times_{k}X'',\mathcal{O},w'_{1}+w''_{2})\operatorname{-qcoh})}(\mathcal{K}'\otimes_{k}\mathcal{K}'',\mathcal{M}'\otimes_{k}\mathcal{M}''[*])$$
(12)

induced by the functor (9) is an isomorphism for any coherent matrix factorizations $\mathcal{K}', \mathcal{K}''$ and quasicoherent matrix factorizations $\mathcal{M}', \mathcal{M}''$ of the potentials w' and w''. One easily checks that the desired assertion holds for the Hom spaces in the homotopy categories of matrix factorizations (since it holds for morphisms between the external tensor products of coherent and quasicoherent sheaves).

Furthermore, one can assume the quasicoherent matrix factorizations \mathcal{M}' and \mathcal{M}'' to be injective. Then the Hom spaces in the left-hand side of the map (12) coincide with the similar Hom spaces computed in the homotopy categories of matrix factorizations. Let \mathcal{I}^{\bullet} be a right resolution of $\mathcal{M}' \otimes_k \mathcal{M}''$ in the abelian category of quasicoherent matrix factorizations (and closed morphisms between them) consisting of injective matrix factorizations, and let \mathcal{J} be the total matrix factorization of the complex \mathcal{I}^{\bullet} constructed by taking infinite direct sums along the diagonals. Then the *k*-vector spaces of morphisms from $\mathcal{K}' \otimes_k \mathcal{K}''$ into \mathcal{J} in the homotopy category of matrix factorizations are isomorphic to the right-hand side of (12) [Positselski 2011b, Theorem 3.7].

Alexander I. Efimov and Leonid Positselski

1268

It remains to show that the spaces of morphisms from $\mathcal{K}' \otimes_k \mathcal{K}''$ to $\mathcal{M}' \otimes_k \mathcal{M}''$ in the homotopy category of matrix factorizations are isomorphic to the similar spaces of morphisms from $\mathcal{K}' \otimes_k \mathcal{K}''$ to \mathcal{J} . Indeed, taking the termwise Hom from $\mathcal{K}' \otimes_k \mathcal{K}''$ preserves exactness of the sequence $0 \to \mathcal{M}' \otimes_k \mathcal{M}'' \to \mathcal{I}^{\bullet}$ since the higher Ext spaces from the components of $\mathcal{K}' \otimes_k \mathcal{K}''$ into those of $\mathcal{M}' \otimes_k \mathcal{M}''$ in the abelian category of quasicoherent sheaves on $X' \otimes_k X''$ vanish. The latter assertion can be checked for affine schemes X', X'' using projective resolutions and then globally for the cohomology of quasicoherent sheaves using, e.g., the Čech approach.

Theorem B.2.2. Assume that the morphisms of schemes

$$w': X' \to \mathbb{A}^1_k$$
 and $w'': X'' \to \mathbb{A}^1_k$

are flat. Suppose that there exist closed subschemes $Z' \subset X'$ and $Z'' \subset X''$ such that $w'|_{Z'} = 0 = w''_{Z''}$, the functions w' and w'' are noncritical on $X' \setminus Z'$ and $X'' \setminus Z''$, and the scheme Z' admits a smooth stratification over k. Then the absolute derived category $D^{abs}((X' \times_k X'', \mathcal{O}, w'_1 + w''_2)$ -coh) coincides with its minimal thick subcategory containing the image of the functor (10).

Proof. By the definition of noncriticality, one has

$$\begin{split} \mathsf{D}^{\mathsf{abs}}((((X' \setminus Z') \times_k X''), \mathcal{O}, w_1' + w_2'')\operatorname{-coh}) &= 0 \\ &= \mathsf{D}^{\mathsf{abs}}(((X' \times_k (X'' \setminus Z'')), \mathcal{O}, w_1' + w_2'')\operatorname{-coh}). \end{split}$$

Therefore, any coherent matrix factorization of the potential $w'_1 + w''_2$ on $X' \times_k X''$ has its category-theoretic support inside $Z' \times_k Z''$, and is consequently isomorphic in $D^{abs}((X' \times_k X'', \mathcal{O}, w'_1 + w''_2)$ -coh) to a direct summand of an object represented by a coherent matrix factorization supported set-theoretically inside $Z' \times_k Z''$ (see Corollary 1.10(b)). It follows that the triangulated category

$$D^{abs}((X' \times_k X'', O, w'_1 + w''_2) - \operatorname{coh})$$

is generated by the direct images of coherent matrix factorizations of the zero potential from the closed embedding $Z' \times_k Z'' \to X' \times_k X''$.

Furthermore, let X'_0 , X''_0 , and Y_0 denote the zero loci of the functions w', w'', and $w'_1 + w''_2$ on X', X'', and $X' \times_k X''$, respectively. Denote the natural closed embeddings by $i' : X'_0 \to X'$, $i'' : X''_0 \to X''$, $\iota : X'_0 \times X''_0 \to Y_0$, and $h : Y_0 \to$ $X' \times X''$. The external tensor product functor (cf. [Polishchuk and Positselski 2012, Lemma 4.8.B])

$$\otimes_k : \mathsf{D}^{\mathsf{b}}_{Sing}(X'_0/X') \times \mathsf{D}^{\mathsf{b}}_{Sing}(X''_0/X'') \longrightarrow \mathsf{D}^{\mathsf{b}}_{Sing}(Y_0/(X' \times_k X''))$$
(13)

is well-defined since for any bounded complexes of coherent sheaves \mathcal{F}^{\bullet} on X' and \mathcal{K}^{\bullet} on X''_0 one has $\iota_*(\mathbb{L}i'^*\mathcal{F}^{\bullet}\otimes_k \mathcal{K}^{\bullet}) \simeq \mathbb{L}h^*((\mathrm{id}_{X'} \times i'')_*(\mathcal{F}^{\bullet}\otimes_k \mathcal{K}^{\bullet}))$. Indeed, the

square diagram of closed embeddings

$$\begin{array}{cccc} X'_{\mathbf{0}} \times_{k} X''_{\mathbf{0}} & \longrightarrow & X' \times_{k} X''_{\mathbf{0}} \\ & \downarrow & & \downarrow \\ & Y_{\mathbf{0}} & \longrightarrow & X' \times X'' \end{array}$$

is Cartesian and the higher derived tensor products related to the construction of this relative Cartesian product of schemes all vanish.

The functor $\Upsilon : D_{Sing}^{b}(Y_0/(X' \times_k X'')) \to D^{abs}((X' \times_k X'', \mathcal{O}, w'_1 + w''_2)\text{-coh})$ (see Section 2.7) and the similar functors for the potentials w' and w'' on X' and X'' transform the external product functor (10) into the external tensor product functor (13). By the assumption, one has $Z' \subset X'_0$ and $Z'' \subset X''_0$. It remains to apply Lemma B.1.6 in order to finish the proof of the theorem. \Box

B.2.3. Internal Hom of matrix factorizations. Let X be a separated Noetherian scheme. Let \mathcal{L} be a line bundle on X and $w', w'' \in \mathcal{L}(X)$ be its global sections. Then given a matrix factorization $\mathcal{U}^0 \to \mathcal{U}^1 \otimes \mathcal{L}^{\otimes 1/2} \to \mathcal{U}^0 \otimes_{\mathcal{O}_X} \mathcal{L}$ of the potential w' and a matrix factorization $\mathcal{V}^0 \to \mathcal{V}^1 \otimes \mathcal{L}^{\otimes 1/2} \to \mathcal{V}^0 \otimes_{\mathcal{O}_X} \mathcal{L}$ of the potential w'' on the scheme X (in the symbolic notation of Section 2.2), one can construct the matrix factorization

$$\begin{aligned} \mathcal{U}^{0} \otimes_{\mathcal{O}_{X}} \mathcal{V}^{0} \oplus \mathcal{U}^{1} \otimes_{\mathcal{O}_{X}} \mathcal{V}^{1} \\ & \longrightarrow \mathcal{U}^{1} \otimes \mathcal{L}^{\otimes 1/2} \otimes_{\mathcal{O}_{X}} \mathcal{V}^{0} \oplus \mathcal{U}^{0} \otimes_{\mathcal{O}_{X}} \mathcal{V}^{1} \otimes \mathcal{L}^{\otimes 1/2} \\ & \longrightarrow \mathcal{U}^{0} \otimes_{\mathcal{O}_{X}} \mathcal{V}^{0} \otimes_{\mathcal{O}_{X}} \mathcal{L} \oplus \mathcal{U}^{1} \otimes_{\mathcal{O}_{X}} \mathcal{V}^{1} \otimes_{\mathcal{O}_{X}} \mathcal{L} \end{aligned}$$

of the potential w' + w'' on *X*. Here the tensor product $\mathcal{U}^1 \otimes_{\mathcal{O}_X} \mathcal{V}^1$ is defined as the sheaf $(\mathcal{U}^1 \otimes \mathcal{L}^{\otimes 1/2}) \otimes_{\mathcal{O}_X} (\mathcal{V}^1 \otimes \mathcal{L}^{\otimes 1/2}) \otimes_{\mathcal{O}_X} \mathcal{L}^{\otimes -1}$ on *X*, while the differential on the tensor product of matrix factorizations is given by the conventional rule $d(u \otimes v) = d(u) \otimes v + (-1)^{|u|} u \otimes d(v).$

We denote the matrix factorization so obtained by $\mathcal{U} \otimes_{\mathcal{O}_X} \mathcal{V}$ and call it the *tensor* product of two matrix factorizations \mathcal{U} and \mathcal{V} of two sections w' and w'' of the same line bundle \mathcal{L} on a scheme X. Restricting to the cases when one or both matrix factorizations are flat, and passing to the coderived categories, one obtains the induced tensor product functors

$$\otimes_{\mathcal{O}_X} : \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w')\operatorname{-qcoh}_{\mathsf{fl}}) \times \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w'')\operatorname{-qcoh}_{\mathsf{fl}}) \longrightarrow \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w' + w'')\operatorname{-qcoh}_{\mathsf{fl}})$$
(14)

and

$$\otimes_{\mathcal{O}_X} : \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w')\operatorname{-qcoh}_{\mathsf{fl}}) \times \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w'')\operatorname{-qcoh}) \longrightarrow \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w' + w'')\operatorname{-qcoh}).$$
(15)

The functors (14) and (15) are well-defined since the tensor product with a flat (quasicoherent) matrix factorization takes a short exact sequence of flat matrix factorizations, the tensor product with a flat matrix factorization takes a short exact sequence of quasicoherent matrix factorizations to a short exact sequence of quasicoherent matrix factorizations, and the tensor product with a quasicoherent matrix factorization takes a short exact sequence of quasicoherent matrix factorizations. Also, the tensor product functor preserves infinite direct sums.

Given a quasicoherent matrix factorization $\mathcal{U}^0 \to \mathcal{U}^1 \otimes \mathcal{L}^{\otimes 1/2} \to \mathcal{U}^0 \otimes_{\mathcal{O}_X} \mathcal{L}$ of a potential $w' \in \mathcal{L}(X)$ and a quasicoherent matrix factorization $\mathcal{V}^0 \to \mathcal{V}^1 \otimes \mathcal{L}^{\otimes 1/2} \to \mathcal{V}^1 \otimes_{\mathcal{O}_X} \mathcal{L}$ of a potential $w'' \in \mathcal{L}(X)$ on the scheme X, one can construct the quasicoherent matrix factorization

$$\mathcal{H}om_{X-\mathsf{qc}}(\mathcal{U}^{0},\mathcal{V}^{0}) \oplus \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{U}^{1},\mathcal{V}^{1}) \longrightarrow \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{U}^{0},\mathcal{V}^{1}\otimes\mathcal{L}^{\otimes 1/2}) \oplus \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{U}^{1},\mathcal{V}^{0})\otimes\mathcal{L}^{\otimes 1/2} \longrightarrow \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{U}^{0},\mathcal{V}^{0})\otimes_{\mathcal{O}_{X}}\mathcal{L} \oplus \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{U}^{1},\mathcal{V}^{1})\otimes_{\mathcal{O}_{X}}\mathcal{L}$$

of the potential w'' - w' on X. Here the sheaf $\mathcal{H}om_{X-qc}(\mathcal{U}^1, \mathcal{V}^0) \otimes \mathcal{L}^{\otimes 1/2}$ is defined as the tensor product $\mathcal{H}om_{X-qc}(\mathcal{U}^1 \otimes \mathcal{L}^{\otimes 1/2}, \mathcal{V}^0) \otimes_{\mathcal{O}_X} \mathcal{L}$, while the differential on the internal Hom is given by the conventional rule $d(g)(u) = d(g(u)) - (-1)^{|g|}g(d(u))$.

We denote the matrix factorization so obtained by $\mathcal{H}om_{X-qc}(\mathcal{U}, \mathcal{V})$ and call it the matrix factorization of *quasicoherent internal Hom* between the quasicoherent matrix factorizations \mathcal{U} and \mathcal{V} of two sections w' and w'' of the same line bundle \mathcal{L} on a scheme X. Restricting to the case when the matrix factorization in the second argument is injective, one obtains the induced internal Hom functor

$$\mathcal{H}om_{X-\mathsf{qc}} : \mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w') \operatorname{-qcoh})^{\mathsf{op}} \times H^0((X, \mathcal{L}, w'') \operatorname{-qcoh}_{\mathsf{inj}}) \longrightarrow \mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w'' - w') \operatorname{-qcoh}), \quad (16)$$

which can be also viewed as the right derived internal Hom functor

$$\mathbb{R} \mathcal{H}om_{X-\mathsf{qc}} : \mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w') \operatorname{-qcoh})^{\mathsf{op}} \times \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w'') \operatorname{-qcoh}) \longrightarrow \mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w'' - w') \operatorname{-qcoh}).$$
(17)

Remark B.2.3. Alternatively, one could restrict the quasicoherent internal Hom functor to pairs of quasicoherent matrix factorizations which are *both* injective, obtaining the triangulated functor

$$\mathcal{H}om_{X-\mathsf{qc}} : H^{0}((X, \mathcal{L}, w') \operatorname{-qcoh}_{\mathsf{inj}})^{\mathsf{op}} \times H^{0}((X, \mathcal{L}, w'') \operatorname{-qcoh}_{\mathsf{inj}}) \longrightarrow H^{0}((X, \mathcal{L}, w'' - w') \operatorname{-qcoh}_{\mathsf{fl}}), \quad (18)$$

which can be also viewed as a derived internal Hom functor

$$\mathbb{LR} \mathcal{H}om_{X-qc} : \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w') \operatorname{-qcoh})^{\mathsf{op}} \times \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w'') \operatorname{-qcoh}) \longrightarrow \mathsf{D}^{\mathsf{abs}}((X, \mathcal{L}, w'' - w') \operatorname{-qcoh}_{\mathsf{fl}})$$
(19)

that is a left derived functor in its first argument and a right derived functor in the second one. Notice that the derived functor so obtained does *not* agree with the right derived functor defined above; i.e., the composition of the functor (19) with the natural fully faithful functor $D^{abs}((X, \mathcal{L}, w''-w')\operatorname{-qcoh}_{fl}) \rightarrow D^{abs}((X, \mathcal{L}, w''-w')\operatorname{-qcoh})$ and the Verdier localization functor $D^{abs}((X, \mathcal{L}, w')\operatorname{-qcoh}) \rightarrow D^{co}((X, \mathcal{L}, w')\operatorname{-qcoh})$ is *not* isomorphic to the functor (17).

In particular, when w' = w'', the functors (16) and (17) take values in the absolute derived category of quasicoherent matrix factorizations of the zero potential $0 \in \mathcal{L}(X)$. The objects of this category are simply complexes of quasicoherent sheaves \mathcal{M}^{\bullet} on X endowed with a 2-periodicity isomorphism $\mathcal{M}^{\bullet}[2] \simeq \mathcal{M}^{\bullet} \otimes_{\mathcal{O}_X} \mathcal{L}$. So there is a natural forgetful functor

$$\mathsf{D}^{\mathsf{co}}((X,\mathcal{L},0)\operatorname{-qcoh}) \longrightarrow \mathsf{D}^{\mathsf{co}}(X\operatorname{-qcoh}) \tag{20}$$

and the similar functors acting on the homotopy, absolute derived, etc. categories of flat, coherent, locally free, etc. matrix factorizations.

Furthermore, there is the derived global sections functor

$$\mathbb{R}\Gamma(X,-): \mathsf{D}^{\mathsf{co}}(X\operatorname{-\mathsf{qcoh}}) \longrightarrow \mathsf{D}(\mathbb{Z}\operatorname{-\mathsf{mod}})$$
(21)

taking values in the derived category of abelian groups and defined using either the injective resolutions or the Čech construction (see Sections 1.8–1.9). In fact, the functor (21) factorizes through the conventional derived category D(X-qcoh).

Composing the forgetful functor with the functor of underived global sections of complexes of quasicoherent sheaves, one obtains a triangulated functor

$$\Gamma(X,-): H^0((X,\mathcal{L},0)\operatorname{-qcoh}) \longrightarrow \mathsf{D}(\mathbb{Z}\operatorname{-mod}).$$
(22)

Alternatively, the functor (22) can be defined as the functor $\operatorname{Hom}_{(X,\mathcal{L},0)\operatorname{-qcoh}}(\mathcal{O}_X, -)$, where the structure sheaf \mathcal{O}_X is viewed as a matrix factorization $(\mathcal{U}^0, \mathcal{U}^1)$ of the potential $0 \in \mathcal{L}(X)$ with the components $\mathcal{U}^0 = \mathcal{O}_X$ and $\mathcal{U}^1 \otimes \mathcal{L}^{\otimes 1/2} = 0$.

Similarly, composing the functors (20) and (21), one obtains a triangulated functor

$$\mathbb{R}\Gamma(X,-): \mathsf{D}^{\mathsf{co}}((X,\mathcal{L},0)\operatorname{-qcoh}) \longrightarrow \mathsf{D}(\mathbb{Z}\operatorname{-mod}),$$
(23)

which can be alternatively described as the functor $\operatorname{Hom}_{\mathsf{D}^{co}((X,\mathcal{L},0)\operatorname{-qcoh})}(\mathcal{O}_X,-[*])$. In the case when $\mathcal{L} = \mathcal{O}_X$, the functors (22) and (23) can be viewed as taking values in the derived category of $\mathbb{Z}/2\operatorname{-graded}$ (2-periodic) complexes of abelian groups. 1272 Alexander I. Efimov and Leonid Positselski

For any quasicoherent matrix factorizations \mathcal{K} and \mathcal{M} of a potential $w \in \mathcal{L}(X)$ on the scheme X there is a natural isomorphism of complexes of abelian groups

$$\operatorname{Hom}_{(X,\mathcal{L},w)\operatorname{-qcoh}}(\mathcal{K},\mathcal{M}) \simeq \Gamma(X,\operatorname{Hom}_{X\operatorname{-qc}}(\mathcal{K},\mathcal{M})), \tag{24}$$

and more generally, for any quasicoherent matrix factorizations \mathcal{K} and \mathcal{E} of potentials w' and $w'' \in \mathcal{L}(X)$ and a quasicoherent matrix factorization \mathcal{M} of the potential w' + w'' on the scheme X there is a natural isomorphism of complexes

$$\operatorname{Hom}_{(X,\mathcal{L},w'+w'')\operatorname{-qcoh}}(\mathcal{K}\otimes_{\mathcal{O}_X}\mathcal{E},\mathcal{M})\simeq\operatorname{Hom}_{(X,\mathcal{L},w'')\operatorname{-qcoh}}(\mathcal{E},\mathcal{H}om_{X\operatorname{-qc}}(\mathcal{K},\mathcal{M})).$$
(25)

Lemma B.2.3. Let \mathcal{K} be a quasicoherent matrix factorization and \mathcal{M} be an injective quasicoherent matrix factorization of a potential $w \in \mathcal{L}(X)$. Let

$$\mathcal{H}om_{X-\mathsf{qc}}(\mathcal{K},\mathcal{M})\longrightarrow \mathcal{J}$$

be a closed morphism with a coacyclic cone between quasicoherent matrix factorizations of the potential $0 \in \mathcal{L}(X)$ from the matrix factorization of quasicoherent internal Hom into an injective matrix factorization \mathcal{J} . Then the induced morphism

$$\Gamma(X, \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{K}, \mathcal{M})) \longrightarrow \Gamma(X, \mathcal{J})$$

is a quasi-isomorphism of complexes of abelian groups.

Proof. Let $0 \to \mathcal{H}om_{X-qc}(\mathcal{K}, \mathcal{M}) \to \mathcal{I}^{\bullet}$ be a right resolution of the matrix factorization $\mathcal{H}om_{X-qc}(\mathcal{K}, \mathcal{M})$ by injective matrix factorization \mathcal{I}^i . Then one can take \mathcal{J} to be the total matrix factorization of the complex \mathcal{I}^{\bullet} constructed by passing to the infinite direct sums along the diagonals. Notice that the functor of global sections of quasicoherent sheaves on X commutes with the infinite direct sums. It remains to show that the functor $\Gamma(X, -) = \operatorname{Hom}_{(X, \mathcal{L}, w'')-\operatorname{qcoh}}(\mathcal{O}_X, -)$ preserves the exactness of the sequence $0 \to \mathcal{H}om_{X-\operatorname{qc}}(\mathcal{K}, \mathcal{M}) \to \mathcal{I}^{\bullet}$ (cf. the proof of Proposition B.2.2).

In fact, we claim that the Ext groups from flat quasicoherent sheaves to the components of $\mathcal{H}om_{X-qc}(\mathcal{K}, \mathcal{M})$ vanish in the abelian category X-qcoh. This assertion follows from the results of [Positselski 2012, Lemma 2.5.3(c) and Corollary 4.1.9(b)] (the argument is based essentially on the above Lemma A.1).

We recall the constructions of the (underived and derived) direct and inverse image functors for matrix factorizations from Sections 1.8–1.9 and 3.5–3.6. In addition to the conventional adjunction of the (underived) direct and inverse image functors f_* and f^* (as mentioned in Section 1.8), there is also the "internal Hom adjunction", formulated as follows.

Let $f: Z \to Y$ be a morphism of separated Noetherian schemes, \mathcal{L} be a line bundle on Y, and $w', w'' \in \mathcal{L}(Y)$ be two global sections. Let $\mathcal{K} \in H^0((Y, \mathcal{L}, w')\operatorname{-qcoh})$

and $\mathcal{M} \in H^0((Z, f^*\mathcal{L}, f^*w'')$ -qcoh) be quasicoherent matrix factorizations on *Y* and *Z*. Then there is a natural isomorphism

$$f_* \mathcal{H}om_{Z-qc}(f^*\mathcal{K}, \mathcal{M}) \simeq \mathcal{H}om_{Y-qc}(\mathcal{K}, f_*\mathcal{M})$$
(26)

of quasicoherent matrix factorizations of the potential w'' - w' on Y.

Now let X be a separated scheme of finite type over a field k, and let $w', w'' \in \mathcal{O}(X)$ be two global regular functions on X. Denote by p_1 and p_2 the natural projections $X \times_k X \rightrightarrows X$, and consider the regular function $w'_1 + w''_2 = p_1^* w' + p_2^* w''$ on $X \times_k X$. Let $\Delta : X \to X \times_k X$ denote the diagonal map.

Let \mathcal{N} and \mathcal{K} be quasicoherent matrix factorizations of the potentials w' and w''on X. Then there is a natural isomorphism $\mathcal{N} \otimes_{\mathcal{O}_X} \mathcal{K} \simeq \Delta^*(\mathcal{N} \otimes_k \mathcal{K})$ of matrix factorizations of the potential w' + w'' on X. Therefore, given a quasicoherent matrix factorization \mathcal{M} of the potential $w' + w'' \in \mathcal{O}(X)$, one has a natural isomorphism of $\mathbb{Z}/2$ -graded complexes of abelian groups

 $\operatorname{Hom}_{(X,\mathcal{O},w'')\operatorname{-qcoh}}(\mathcal{K},\mathcal{H}om_{X\operatorname{-qc}}(\mathcal{N},\mathcal{M}))$

$$\simeq \operatorname{Hom}_{(X \times_k X, \mathcal{O}, w_1' + w_2') \operatorname{-qcoh}}(\mathcal{N} \otimes_k \mathcal{K}, \Delta_* \mathcal{M}).$$
(27)

Proposition B.2.3. (a) Assume that the matrix factorization \mathcal{N} is coherent and the matrix factorization \mathcal{M} is injective. Let $\Delta_*\mathcal{M} \to \mathcal{J}$ be a closed morphism with a coacyclic cone between quasicoherent matrix factorizations of the potential $w'_1 + w''_2$ on $X \times_k X$ from the direct image $\Delta_*\mathcal{M}$ into an injective matrix factorization \mathcal{J} . Then there is a natural closed morphism with a coacyclic cone $\mathcal{H}om_{X-qc}(\mathcal{N}, \mathcal{M}) \to p_{2*}\mathcal{H}om_{X\times_k X-qc}(p_1^*\mathcal{N}, \mathcal{J})$ of quasicoherent matrix factorizations of the potential w'' on X, and the matrix factorization $p_{2*}\mathcal{H}om_{X\times_k X-qc}(p_1^*\mathcal{N}, \mathcal{J})$ is injective.

(b) There is a natural isomorphism of $\mathbb{Z}/2$ -graded complexes of abelian groups

 $\operatorname{Hom}_{(X,\mathcal{O},w'')\operatorname{-qcoh}}(\mathcal{K}, p_{2*}\operatorname{\mathcal{H}om}_{X\times_k X\operatorname{-qc}}(p_1^*\mathcal{N},\mathcal{J})) \\ \simeq \operatorname{Hom}_{(X\times_k X,\mathcal{O},w_1'+w_2'')\operatorname{-qcoh}}(\mathcal{N}\otimes_k \mathcal{K},\mathcal{J}).$

Proof. Part (a): The desired closed morphism is provided by the composition

$$\mathcal{H}om_{X-\mathsf{qc}}(\mathcal{N},\mathcal{M}) \simeq p_{2*}\Delta_* \mathcal{H}om_{X-\mathsf{qc}}(\Delta^* p_1^*\mathcal{N},\mathcal{M})$$
$$\simeq p_{2*}\mathcal{H}om_{X\times_k X-\mathsf{qc}}(p_1^*\mathcal{N},\Delta_*\mathcal{M}) \longrightarrow p_{2*}\mathcal{H}om_{X\times_k X-\mathsf{qc}}(p_1^*\mathcal{N},\mathcal{J}).$$

To prove that this morphism has a coacyclic cone, pick a right resolution \mathcal{I}^{\bullet} of the matrix factorization $\Delta_* \mathcal{M}$ on $X \times_k X$ by injective matrix factorizations, and take \mathcal{J} to be the totalization of the complex of matrix factorizations \mathcal{I}^{\bullet} constructed by passing to the infinite direct sums along the diagonals.

Then the complex of matrix factorizations $0 \to \mathcal{H}om_{X \times_k X \operatorname{-qc}}(p_1^* \mathcal{N}, \Delta_* \mathcal{M}) \to \mathcal{H}om_{X \times_k X \operatorname{-qc}}(p_1^* \mathcal{N}, \mathcal{I}^{\bullet})$ is acyclic since for any affine open subscheme $U \subset X$ the

higher Ext spaces between the components of the restrictions of p_1^*N and Δ_*M to $U \times_k U$ vanish. The latter assertion follows from the adjunction of derived functors $\mathbb{L}\Delta^*$ and $\Delta_* = \mathbb{R}\Delta_*$ or $p_1^* = \mathbb{L}p_1^*$ and $\mathbb{R}p_{1*}$ together with the agreement of the derived direct/inverse images of (complexes of) quasicoherent sheaves with the compositions of morphisms of separated Noetherian schemes.

It remains to show that our complex will stay acyclic after applying the direct image functor p_{2*} . According to the argument in the proof of Lemma B.2.3, the components of the matrix factorizations $\mathcal{H}om_{X\times_k X\text{-qc}}(p_1^*\mathcal{N}, \mathcal{I}^i)$ are acyclic for the direct image; so are the components of the matrix factorization $\mathcal{H}om_{X\times_k X\text{-qc}}(p_1^*\mathcal{N}, \Delta_*\mathcal{M})$, in view of the above local argument and since for any affine open subscheme $U \subset X$, the higher Ext spaces between the components of the restrictions of $p_1^*\mathcal{N}$ and $\Delta_*\mathcal{M}$ to $X \times_k U$ vanish. The latter assertion is checked in the same way as above.

Finally, the claim that the matrix factorization in question is injective follows from the computation in part (b), which shows that the left-hand side is an exact functor of the argument \mathcal{K} , because the right-hand side is.

Part (b) is straightforward:

1274

$$\begin{split} \operatorname{Hom}_{(X,\mathcal{O},w'')\operatorname{-qcoh}}(\mathcal{K}, \ p_{2*} \operatorname{\mathcal{H}om}_{X \times_k X\operatorname{-qc}}(p_1^*\mathcal{N},\mathcal{J})) \\ &\simeq \operatorname{Hom}_{(X \times_k X,\mathcal{O},w_2'')\operatorname{-qcoh}}(p_2^*\mathcal{K}, \operatorname{\mathcal{H}om}_{X \times_k X\operatorname{-qc}}(p_1^*\mathcal{N},\mathcal{J})) \\ &\simeq \operatorname{Hom}_{(X \times_k X,\mathcal{O},w_1'+w_2'')\operatorname{-qcoh}}(p_1^*\mathcal{N} \otimes_{\mathcal{O}_{X \times_k X}} p_2^*\mathcal{K},\mathcal{J}) \\ &\simeq \operatorname{Hom}_{(X \times_k X,\mathcal{O},w_1'+w_2'')\operatorname{-qcoh}}(\mathcal{N} \otimes_k \mathcal{K},\mathcal{J}). \end{split}$$

B.2.4. *Hochschild cohomology.* Our goal in the rest of this appendix is to compute the Hochschild (co)homology of the DG-category $DG^{abs}((X, \mathcal{O}, w)\text{-coh})$ corresponding to the triangulated category $D^{abs}((X, \mathcal{O}, w)\text{-coh})$. The word "corresponding" here means, first of all, that there is a natural equivalence of (triangulated) categories $H^0DG^{abs}((X, \mathcal{O}, w)\text{-coh}) \simeq D^{abs}((X, \mathcal{O}, w)\text{-coh})$ (see [Positselski 2011b, Section 1.2]).

As the absolute derived category is constructed from the homotopy category of matrix factorizations using the Verdier localization procedure, so the DG-category $DG^{abs}((X, \mathcal{O}, w)\text{-coh})$ is obtained by applying a DG-version of localization to the DG-category of coherent matrix factorizations $(X, \mathcal{O}, w)\text{-coh}$ of the potential w on the scheme X (see Section 1.2). Several such localization procedures are known, leading to naturally quasiequivalent DG-categories. As the Hochschild (co)homology of DG-categories are preserved by quasiequivalences [Polishchuk and Positselski 2012, Sections 2.1 and 2.4], it is not very important which localization procedure to choose. To be specific, let us say that we prefer Drinfeld's localization [Drinfeld 2004]. Similarly one localizes the DG-category (X, \mathcal{O}, w) -qcoh and

obtains a DG-category $DG^{co}((X, \mathcal{O}, w)$ -qcoh) "corresponding" to the coderived category $D^{co}((X, \mathcal{O}, w)$ -qcoh).

Our method will naturally allow us to compute the Hochschild cohomology $HH^*(\mathsf{DG}^{\mathsf{abs}}((X, \mathcal{O}, w)\text{-}\mathsf{coh}))$ together with its structure of an associative (in fact, supercommutative, but we will neither prove nor use this fact) $\mathbb{Z}/2$ -graded algebra over k. Similarly, the Hochschild homology $HH_*(\mathsf{DG}^{\mathsf{abs}}((X, \mathcal{O}, w)\text{-}\mathsf{coh}))$ will be computed together with its structure of a $\mathbb{Z}/2$ -graded module over the $\mathbb{Z}/2$ -graded associative algebra $HH^*(\mathsf{DG}^{\mathsf{abs}}((X, \mathcal{O}, w)\text{-}\mathsf{coh}))$.

Let X be a separated scheme of finite type over a field k and $w \in \mathcal{O}(X)$ be a global regular function. Assume that the morphism of schemes $w : X \to \mathbb{A}_k^1$ is flat. Consider the Cartesian square $X \times_k X$ and endow it with the potential (global function) $w_2 - w_1 = p_2^*(w) - p_1^*(w)$.

Any $\mathbb{Z}/2$ -graded complex of quasicoherent sheaves \mathcal{K}^{\bullet} on X can be viewed as a matrix factorization of the potential $0 \in \mathcal{O}(X)$. Furthermore, one can take its direct image $\Delta_*\mathcal{K}^{\bullet}$ with respect to the diagonal embedding $\Delta : X \to X \times_k X$ and consider it as a quasicoherent matrix factorization of the potential $w_2 - w_1$ on $X \times_k X$. Given a bounded \mathbb{Z} -graded complex of quasicoherent sheaves \mathcal{K}^{\bullet} , one can associate a $\mathbb{Z}/2$ -complex with it (by taking direct sums of all terms with the same parity) and then apply the above constructions.

Theorem B.2.4. Assume that there exists a closed subscheme $Z \subset X$ such that $w|_Z = 0$, the function w is noncritical on $X \setminus Z$, and the scheme Z admits a smooth stratification over k. In particular, if the field k is perfect, it suffices to require that the function w on X have no critical values but zero (and take $Z = \{w = 0\}$). Then there is a natural isomorphism between the Hochschild cohomology algebra $HH^*(\mathsf{DG}^{\mathsf{abs}}((X, \mathcal{O}, w)\operatorname{-coh}))$ and the Ext algebra

$$\operatorname{Hom}_{\mathsf{D}^{\mathrm{co}}((X \times_k X, \mathcal{O}, w_2 - w_1) \operatorname{-qcoh})}(\Delta_* \mathcal{D}^{\bullet}_X, \Delta_* \mathcal{D}^{\bullet}_X[*]),$$

where \mathcal{D}_X^{\bullet} denotes a dualizing complex on X.

Proof. By the definition, the Hochschild cohomology algebra of a $\mathbb{Z}/2$ -graded DG-category DG is the $\mathbb{Z}/2$ -graded algebra $\operatorname{Hom}_{D(DG\otimes_k DG^{op}-mod)}(DG, DG[*])$, where the Hom is taken in the conventional derived category $D(DG \otimes_k DG^{op}-mod)$ of DG-bimodules over DG (or DG-modules over $DG \otimes_k DG^{op}$) between two copies of the diagonal DG-bimodule DG over DG [Polishchuk and Positselski 2012, Sections 2.4 and 3.1].

Specializing to the case of the DG-category $DG_w = DG^{abs}((X, \mathcal{O}, w)\text{-coh})$, we notice, first of all, that the contravariant Serre–Grothendieck duality functor $\mathcal{M} \mapsto \mathcal{H}om_{\mathcal{O}_X}(\mathcal{M}, \mathcal{D}_X^{\bullet})$ (see Section 2.5) provides a quasiequivalence between the DG-categories $DG^{abs}((X, \mathcal{O}, w)\text{-coh})^{op}$ and $DG^{abs}((X, \mathcal{O}, -w)\text{-coh})$. Furthermore, the external tensor product is a DG-functor

$$\mathsf{DG}^{\mathsf{abs}}((X,\mathcal{O},-w)\operatorname{-coh}) \otimes_k \mathsf{DG}^{\mathsf{abs}}((X,\mathcal{O},w)\operatorname{-coh}) \longrightarrow \mathsf{DG}^{\mathsf{abs}}((X \times_k X,\mathcal{O},w_2 - w_1)\operatorname{-coh}), \quad (28)$$

which, according to Proposition B.2.2 and Theorem B.2.2, induces an equivalence between the derived categories of (left or right) DG-modules over the two DG-categories in the left-hand and right-hand sides. Composing the Serre–Grothendieck duality with the external tensor product, we obtain (perhaps, after replacing our DG-categories with naturally quasiequivalent ones) a DG-functor

$$\mathsf{DG}^{\mathsf{abs}}((X,\mathcal{O},w)\operatorname{-coh})^{\mathsf{op}} \otimes_k \mathsf{DG}^{\mathsf{abs}}((X,\mathcal{O},w)\operatorname{-coh}) \longrightarrow \mathsf{DG}^{\mathsf{abs}}((X \times_k X,\mathcal{O},w_2 - w_1)\operatorname{-coh})$$
(29)

having the same property with respect to the derived categories of DG-modules over the left-hand and right-hand sides as the DG-functor (28).

We are interested specifically in the diagonal right DG-module over $DG_w^{op} \otimes_k DG_w$, that is, the contravariant functor from $DG_w^{op} \otimes_k DG_w$ to the DG-category of $\mathbb{Z}/2$ graded complexes of *k*-vector spaces taking an object ($\mathcal{M}^{op}, \mathcal{K}$) to the complex Hom_{DG}(\mathcal{K}, \mathcal{M}). It is claimed that the diagonal DG-module is naturally quasiisomorphic to the DG-module obtained by composing the DG-functor (29) with the right DG-module over the right-hand side represented by the object

$$\Delta_* \mathcal{D}_X^{\bullet} \in \mathsf{DG}^{\mathsf{abs}}((X \times_k X, \mathcal{O}, w_2 - w_1) \operatorname{-coh}) \subset \mathsf{DG}^{\mathsf{co}}((X \times_k X, \mathcal{O}, w_2 - w_1) \operatorname{-qcoh}).$$

Indeed, for any quasicoherent matrix factorizations \mathcal{K} and \mathcal{N} of the potentials w and -w on X there is a natural isomorphism of $\mathbb{Z}/2$ -graded complexes of abelian groups (see (27))

$$\operatorname{Hom}_{(X,\mathcal{O},w)\operatorname{-qcoh}}(\mathcal{K}, \mathcal{H}om_{\mathcal{O}_X}(\mathcal{N}, \mathcal{D}_X^{\bullet})) \simeq \operatorname{Hom}_{(X \times_k X, \mathcal{O}, w_2 - w_1)\operatorname{-qcoh}}(\mathcal{N} \otimes_k \mathcal{K}, \Delta_* \mathcal{D}_X^{\bullet}).$$

Proposition B.2.3 shows how one can pass from this isomorphism to a quasi-isomorphism of the similar complexes of morphisms in the DG-categories $DG^{co}((X, \mathcal{O}, w) -qcoh)$ and $DG^{co}((X \times_k X, \mathcal{O}, w_2 - w_1)-qcoh)$.

Now morphisms between representable DG-modules in the derived category of DG-modules over a DG-category DG are computed by the complex of morphisms in DG between the representing objects, so our proof is finished. $\hfill\square$

Remark B.2.4. Given a separated scheme X of finite type over a field k, let \mathcal{D}_X^{\bullet} be a dualizing complex on X and $\mathcal{D}_{X \times_k X}^{\bullet}$ be a dualizing complex on $X \times_k X$ such

1276

that $\mathcal{D}_X^{\bullet} \simeq \mathbb{R}\Delta^!(\mathcal{D}_{X \times_k X}^{\bullet})$. Then the antiequivalence of absolute derived categories $\mathcal{H}om_{X \times_k X-qc}(-, \mathcal{D}_{X \times_k X}^{\bullet}) : \mathbb{D}^{\mathsf{abs}}((X \times_k X, \mathcal{O}, w_1 - w_2) \operatorname{-coh})^{\mathsf{op}} \simeq \mathbb{D}^{\mathsf{abs}}((X \times_k X, \mathcal{O}, w_2 - w_1) \operatorname{-coh})$

from Proposition 2.5 transforms the object $\Delta_* \mathcal{O}_X \in D^{abs}((X \times_k X, \mathcal{O}, w_1 - w_2)$ -coh) into the object $\Delta_* \mathcal{D}_X^{\bullet} \in D^{abs}((X \times_k X, \mathcal{O}, w_2 - w_1)$ -coh) (see Proposition 3.7). Therefore, in the assumptions of Theorem B.2.4, the Hochschild cohomology algebra $HH^*(D^{abs}((X, \mathcal{O}, w)$ -coh)) of the DG-category $DG^{abs}((X, \mathcal{O}, w)$ -coh) can be also identified with the Ext algebra

$$\operatorname{Hom}_{\mathsf{D}^{\mathsf{abs}}((X \times_k X, \mathcal{O}, w_1 - w_2) \operatorname{-coh})}(\Delta_* \mathcal{O}_X, \Delta_* \mathcal{O}_X[*])^{\operatorname{op}}$$

(cf. Remark B.2.7 below).

B.2.5. Cotensor product of complexes of quasicoherent sheaves. Let X be a separated Noetherian scheme. Then there is a tensor product functor on the coderived category of (\mathbb{Z} -graded complexes of) flat quasicoherent sheaves on X (cf. [Murfet 2007, Chapter 6])

$$\otimes_{\mathcal{O}_X} : \mathsf{D}^{\mathsf{co}}(X\operatorname{-qcoh}_{\mathsf{fl}}) \times \mathsf{D}^{\mathsf{co}}(X\operatorname{-qcoh}_{\mathsf{fl}}) \longrightarrow \mathsf{D}^{\mathsf{co}}(X\operatorname{-qcoh}_{\mathsf{fl}}), \tag{30}$$

and a similar functor of the tensor product on the coderived categories of flat and arbitrary quasicoherent sheaves (see [Positselski 2012, Section 4.12])

$$\otimes_{\mathcal{O}_X} : \mathsf{D}^{\mathsf{co}}(X\operatorname{-qcoh}_{\mathsf{fl}}) \times \mathsf{D}^{\mathsf{co}}(X\operatorname{-qcoh}) \longrightarrow \mathsf{D}^{\mathsf{co}}(X\operatorname{-qcoh}).$$
(31)

Now let \mathcal{D}_X^{\bullet} be a dualizing complex on X (viewed, as usual, as a finite complex of injective quasicoherent sheaves). Then the equivalence of triangulated categories $D^{co}(X\operatorname{-qcoh}_{fl}) \simeq D^{co}(X\operatorname{-qcoh})$ constructed using the dualizing complex \mathcal{D}_X^{\bullet} (see [Murfet 2007, Chapter 8]) transforms the tensor product functor (30) into the tensor product functor (31). One can use the same equivalence of categories to define a tensor triangulated category structure with the unit object \mathcal{D}_X^{\bullet} on the coderived category $D^{co}(X\operatorname{-qcoh})$. We call this operation the *cotensor product* of complexes of quasicoherent sheaves on X and denote it by

$$\Box_{\mathcal{D}_{Y}^{\bullet}}: \mathsf{D}^{\mathsf{co}}(X\operatorname{-qcoh}) \times \mathsf{D}^{\mathsf{co}}(X\operatorname{-qcoh}) \longrightarrow \mathsf{D}^{\mathsf{co}}(X\operatorname{-qcoh}).$$
(32)

Explicitly, $\mathcal{N}^{\bullet} \Box_{\mathcal{D}_{X}^{\bullet}} \mathcal{M}^{\bullet} = \mathcal{D}_{X}^{\bullet} \otimes_{\mathcal{O}_{X}} \mathcal{H}om_{X-qc}(\mathcal{D}_{X}^{\bullet}, \mathcal{N}^{\bullet}) \otimes_{\mathcal{O}_{X}} \mathcal{H}om_{X-qc}(\mathcal{D}_{X}^{\bullet}, \mathcal{M}^{\bullet})$ for any complexes of injective quasicoherent sheaves \mathcal{N}^{\bullet} and \mathcal{M}^{\bullet} on X (cf. Lemma 1.7(b)) and also $\mathcal{N}^{\bullet} \Box_{\mathcal{D}_{X}^{\bullet}} \mathcal{M}^{\bullet} = \mathcal{H}om_{X-qc}(\mathcal{D}_{X}^{\bullet}, \mathcal{N}^{\bullet}) \otimes_{\mathcal{O}_{X}} \mathcal{M}^{\bullet}$ for any complex of injective quasicoherent sheaves \mathcal{N}^{\bullet} and any complex of quasicoherent sheaves \mathcal{M}^{\bullet} on X.

Recall that the full triangulated subcategory of bounded-below complexes in $D^{co}(X-qcoh)$ is equivalent to $D^+(X-qcoh)$ (see [Positselski 2010, Lemma 2.1 and

Remark 4.1] or [Positselski 2012, Lemma A.1.2]). Denote by $D^+_{coh}(X-qcoh)$ the full triangulated subcategory in $D^+(X-qcoh)$ consisting of complexes with coherent cohomology sheaves; then the category $D^+_{coh}(X-qcoh)$ can be also viewed as a full triangulated subcategory in $D^{co}(X-qcoh)$.

For any complexes of quasicoherent sheaves \mathcal{N}^{\bullet} and \mathcal{M}^{\bullet} on X there is a natural morphism of complexes of quasicoherent sheaves

$$\mathcal{D}_{X}^{\bullet} \otimes_{\mathcal{O}_{X}} \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{D}_{X}^{\bullet}, \mathcal{N}^{\bullet}) \otimes_{\mathcal{O}_{X}} \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{D}_{X}^{\bullet}, \mathcal{M}^{\bullet}) \longrightarrow \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{H}om_{X-\mathsf{qc}}(\mathcal{N}^{\bullet}, \mathcal{D}_{X}^{\bullet}) \otimes_{\mathcal{O}_{X}} \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{M}^{\bullet}, \mathcal{D}_{X}^{\bullet}), \mathcal{D}_{X}^{\bullet})$$
(33)

on X defined in terms of the composition morphisms

$$\mathcal{H}om_{X-\mathsf{qc}}(\mathcal{D}_X^{\bullet},\mathcal{K}^{\bullet})\otimes_{\mathcal{O}_X}\mathcal{H}om_{X-\mathsf{qc}}(\mathcal{K}^{\bullet},\mathcal{D}_X^{\bullet})\longrightarrow\mathcal{H}om_{X-\mathsf{qc}}(\mathcal{D}_X^{\bullet},\mathcal{D}_X^{\bullet})$$

for complexes of quasicoherent sheaves \mathcal{K}^{\bullet} on X and the natural quasi-isomorphism

$$\mathcal{D}_X^{\bullet} \otimes_{\mathcal{O}_X} \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{D}_X^{\bullet}, \mathcal{D}_X^{\bullet}) \otimes_{\mathcal{O}_X} \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{D}_X^{\bullet}, \mathcal{D}_X^{\bullet}) \longrightarrow \mathcal{D}_X^{\bullet}.$$

Theorem B.2.5. For any bounded-below complexes of injective quasicoherent sheaves \mathcal{N}^{\bullet} and \mathcal{M}^{\bullet} with coherent cohomology sheaves on a separated Noetherian scheme X with a dualizing complex \mathcal{D}_X^{\bullet} , the natural morphism (33) is a homotopy equivalence of bounded-below complexes of injective quasicoherent sheaves on X with coherent cohomology sheaves.

Proof. By Lemma 2.5(b) and (c), both sides of (33) are bounded-below complexes of injective quasicoherent sheaves. Since the functor

$$\mathcal{H}om_{X-\mathsf{qc}}(-, \mathcal{D}_X^{\bullet}) : \mathsf{D}(X-\mathsf{qcoh}) \longrightarrow \mathsf{D}(X-\mathsf{qcoh})$$

takes $D^+_{coh}(X-qcoh) \subset D^+(X-qcoh)$ into $D^-(X-coh) \subset D^-(X-qcoh)$ and vice versa, while the derived tensor product functor

 $\otimes_X^{\mathbb{L}} : \mathsf{D}^-(X\operatorname{\!-qcoh}) \times \mathsf{D}^-(X\operatorname{\!-qcoh}) \longrightarrow \mathsf{D}^-(X\operatorname{\!-qcoh})$

takes $D^{-}(X-coh) \times D^{-}(X-coh)$ into $D^{-}(X-coh)$, the right-hand side has coherent cohomology sheaves.

It remains to prove the homotopy equivalence claim. Since the homotopy category of bounded-below complexes of injectives is equivalent to $D^+(X-qcoh)$, one only has to check that the map is a quasi-isomorphism. Let us first show that it suffices to do so for complexes of sheaves on affine open subschemes $U \subset X$.

Indeed, for any quasicoherent sheaves \mathcal{E} and \mathcal{K} on X there is a natural morphism of quasicoherent sheaves $\mathcal{H}om_{X-qc}(\mathcal{E},\mathcal{K})|_U \to \mathcal{H}om_{U-qc}(\mathcal{E}|_U,\mathcal{K}|_U)$ on U. The morphism of complexes of quasicoherent sheaves $\mathcal{H}om_{X-qc}(\mathcal{E}^{\bullet},\mathcal{K}^{\bullet})|_U \to$

 $\mathcal{H}om_{U-qc}(\mathcal{E}^{\bullet}|_U, \mathcal{K}^{\bullet}|_U)$ is a quasi-isomorphism whenever the complex \mathcal{E}^{\bullet} has coherent cohomology, \mathcal{K}^{\bullet} is a complex of injective quasicoherent sheaves, and one of the complexes \mathcal{E}^{\bullet} and \mathcal{K}^{\bullet} is finite.

Finally, the tensor product in the right-hand side preserves quasi-isomorphisms of bounded-above complexes of flat quasicoherent sheaves, while the one in the left-hand side is well-defined on the coderived category of (complexes of) flat quasicoherent sheaves. It remains to notice that the functor $\operatorname{Hom}_{X-\operatorname{qc}}(\mathcal{D}_X^{\bullet}, -)$ in the equivalence of categories in Theorem 2.5 agrees with the restrictions to open subschemes since so does its inverse functor $\mathcal{D}_X^{\bullet} \otimes_{\mathcal{O}_X} -$.

Now that we are on an affine scheme U, pick bounded-above complexes of vector bundles \mathcal{N}^{\bullet} and \mathcal{M}^{\bullet} isomorphic to $\mathcal{H}om_{U-qc}(\mathcal{N}^{\bullet}, \mathcal{D}_{U}^{\bullet})$ and $\mathcal{H}om_{U-qc}(\mathcal{M}^{\bullet}, \mathcal{D}_{U}^{\bullet})$, respectively, in D⁻(U-coh). Given the isomorphisms

$$\mathcal{H}om_{U-\mathsf{qc}}(\mathcal{D}_{U}^{\bullet}, \mathcal{H}om_{U-\mathsf{qc}}(\mathcal{N}^{\bullet}, \mathcal{D}_{U}^{\bullet})) \simeq \mathcal{H}om_{U-\mathsf{qc}}(\mathcal{N}^{\bullet}, \mathcal{H}om_{U-\mathsf{qc}}(\mathcal{D}_{U}^{\bullet}, \mathcal{D}_{U}^{\bullet}))$$
$$\simeq \mathcal{H}om_{U-\mathsf{qc}}(\mathcal{N}^{\bullet}, \mathcal{O}_{U}),$$

and similar isomorphisms for $'\mathcal{M}^{\bullet}$ in $\mathsf{D}^{\mathsf{co}}(U\operatorname{-qcoh}_{\mathsf{fl}})$, the assertion reduces to the obvious isomorphism of complexes

$$\mathcal{D}_{X}^{\bullet} \otimes_{\mathcal{O}_{X}} \mathcal{H}om_{U-\mathsf{qc}}(\mathcal{N}^{\bullet}, \mathcal{O}_{U}) \otimes_{\mathcal{O}_{U}} \mathcal{H}om_{U-\mathsf{qc}}(\mathcal{M}^{\bullet}, \mathcal{O}_{U}) \\ \simeq \operatorname{Hom}_{U-\mathsf{qc}}(\mathcal{N}^{\bullet} \otimes_{\mathcal{O}_{X}} \mathcal{M}^{\bullet}, \mathcal{D}_{U}^{\bullet}). \quad \Box$$

For any complexes of quasicoherent sheaves \mathcal{K}^{\bullet} and \mathcal{M}^{\bullet} on X we denote by $\mathcal{H}om_{X-qc}^{\oplus}(\mathcal{K}^{\bullet}, \mathcal{M}^{\bullet})$ the complex of quasicoherent sheaves on X obtained by totalizing the bicomplex of quasicoherent internal Hom sheaves $\mathcal{H}om_{X-qc}(\mathcal{K}^i, \mathcal{M}^j)$ by taking infinite direct sums along the diagonals. Assuming that \mathcal{M}^{\bullet} is a complex of injective quasicoherent sheaves, the complex $\mathcal{H}om_{X-qc}^{\oplus}(\mathcal{K}^{\bullet}, \mathcal{M}^{\bullet})$ is absolutely acyclic with respect to X-qcoh whenever the complex \mathcal{K}^{\bullet} is (see Lemma 2.5(a)).

In the same assumption, the complex $\mathcal{H}om_{X-qc}^{\oplus}(\mathcal{K}^{\bullet}, \mathcal{M}^{\bullet})$ is also coacyclic with respect to X-qcoh whenever the complex of quasicoherent sheaves \mathcal{K}^{\bullet} is acyclic and bounded from above [Positselski 2010, Lemma 2.1]. Therefore, representing the second argument of $\mathcal{H}om_{X-qc}^{\oplus}$ by complexes of injectives, one can construct the right derived functors

$$\mathbb{R} \mathcal{H}om_{X-\mathsf{qc}}^{\oplus} : \mathsf{D}^{\mathsf{abs}}(X-\mathsf{qcoh})^{\mathsf{op}} \times \mathsf{D}^{\mathsf{co}}(X-\mathsf{qcoh}) \longrightarrow \mathsf{D}^{\mathsf{abs}}(X-\mathsf{qcoh})$$
(34)

and

$$\mathbb{R} \mathcal{H}om_{X-\mathsf{qc}}^{\oplus} : \mathsf{D}^{-}(X-\mathsf{qcoh})^{\mathsf{op}} \times \mathsf{D}^{\mathsf{co}}(X-\mathsf{qcoh}) \longrightarrow \mathsf{D}^{\mathsf{co}}(X-\mathsf{qcoh})$$
(35)

of the functor $\mathcal{H}om_{X-qc}^{\oplus}$.

For any complexes of quasicoherent sheaves \mathcal{N}^{\bullet} and \mathcal{M}^{\bullet} on X there is a natural morphism of complexes of quasicoherent sheaves

$$\mathcal{N}^{\bullet} \otimes_{\mathcal{O}_{X}} \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{D}_{X}^{\bullet}, \mathcal{M}^{\bullet}) \longrightarrow \mathcal{H}om_{X-\mathsf{qc}}^{\oplus}(\mathcal{H}om_{X-\mathsf{qc}}(\mathcal{N}^{\bullet}, \mathcal{D}_{X}^{\bullet}), \mathcal{M}^{\bullet})$$
(36)

on X defined in terms of the composition/evaluation morphism

$$\mathcal{N}^{\bullet} \otimes_{\mathcal{O}_X} \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{N}^{\bullet}, \mathcal{D}_X^{\bullet}) \otimes_{\mathcal{O}_X} \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{D}_X^{\bullet}, \mathcal{M}^{\bullet}) \longrightarrow \mathcal{M}^{\bullet}.$$

Proposition B.2.5. For any bounded-below complex of injective quasicoherent sheaves \mathcal{N}^{\bullet} with coherent cohomology sheaves and any complex of injective quasicoherent sheaves \mathcal{M}^{\bullet} on X, the natural morphism (36) is a homotopy equivalence of complexes of injective quasicoherent sheaves on X.

Proof. It suffices to check that the morphism (36) is an isomorphism in $D^{co}(X-qcoh)$. Since both sides of the desired isomorphism are well-defined as functors of the argument $\mathcal{N}^{\bullet} \in D^+(X-qcoh)$ taking values in $D^{co}(X-qcoh)$, one can freely replace \mathcal{N}^{\bullet} with any quasi-isomorphic bounded-below complex of quasicoherent sheaves. The same applies to the bounded-above complex $\mathcal{H}om_{X-qc}(\mathcal{N}^{\bullet}, \mathcal{D}_X^{\bullet})$ in the right-hand side of (36).

Since all the functors involved are local in X up to isomorphism in the relevant triangulated categories, it suffices to consider complexes of sheaves over affine open subschemes $U \subset X$ (see Remark 1.3). Representing the object $\mathcal{H}om_{U-qc}(\mathcal{N}^{\bullet}, \mathcal{D}_{U}^{\bullet}) \in D^{-}(U\text{-coh}) \subset D^{-}(U\text{-qcoh})$ by a bounded-above complex of vector bundles ' \mathcal{N}^{\bullet} , it remains to notice the isomorphism of complexes

$$\mathcal{H}om_{U-\mathsf{qc}}(\mathcal{N}^{\bullet}, \mathcal{D}_{U}^{\bullet}) \otimes_{\mathcal{O}_{U}} \mathcal{F}^{\bullet} \simeq \mathcal{H}om_{U-\mathsf{qc}}^{\oplus}(\mathcal{N}^{\bullet}, \mathcal{D}_{U}^{\bullet} \otimes_{\mathcal{O}_{U}} \mathcal{F}^{\bullet})$$

for any complex of quasicoherent sheaves \mathcal{F}^{\bullet} on U and point out that the functor $\mathcal{H}om_{U-nc}^{\oplus}(\mathcal{N}^{\bullet}, -)$ takes a homotopy equivalence

$$\mathcal{D}^{\bullet}_{U} \otimes_{\mathcal{O}_{U}} \mathcal{H}om_{U-\mathsf{qc}}(\mathcal{D}^{\bullet}_{U}, \mathcal{M}^{\bullet}) \longrightarrow \mathcal{M}^{\bullet}$$

to a homotopy equivalence.

In the particular cases when either \mathcal{N}^{\bullet} is a finite complex of quasicoherent sheaves with coherent cohomology sheaves, or \mathcal{N}^{\bullet} is a bounded-below complex of quasicoherent sheaves with coherent cohomology sheaves and \mathcal{M}^{\bullet} is a boundedbelow complex of quasicoherent sheaves, the direct sum totalization of the bicomplex $\mathcal{H}om_{X-qc}$ in the right-hand side of the isomorphism (36) in the coderived category $D^{co}(X-qcoh)$ is no different from the conventional direct product totalization.

Finally, let X be a separated scheme of finite type over a field k and $\pi: X \to \operatorname{Spec} k$ be its structure morphism. Then $\mathcal{D}_X^{\bullet} \simeq \pi^+ \mathcal{O}_{\operatorname{Spec} k}$ (see Section 3.7) is a natural choice of the dualizing complex on X. Let $\pi \times_k \pi : X \times_k X \to \operatorname{Spec} k$ be the structure morphism of the Cartesian square of X over k. Then the dualizing complex

1280

 $\mathcal{D}_{X \times_k X}^{\bullet} = (\pi \times_k \pi)^+ \mathcal{O}_{\text{Spec }k}$ on $X \times_k X$ is quasi-isomorphic to the external tensor product $\mathcal{D}_X^{\bullet} \otimes_k \mathcal{D}_X^{\bullet}$, and one has $\mathcal{D}_X^{\bullet} \simeq \Delta^+(\mathcal{D}_{X \times_k X}^{\bullet}) \simeq \mathbb{R}\Delta^!(\mathcal{D}_X^{\bullet} \otimes_k \mathcal{D}_X^{\bullet})$, where $\Delta: X \to \tilde{X} \times_k X$ denotes the diagonal map.

The equivalence of categories $D^{co}(X \times_k X \operatorname{-qcoh}_{fl}) \simeq D^{co}(X \times_k X \operatorname{-qcoh})$ constructed using the dualizing complex $\mathcal{D}^{\bullet}_{X \times_k X}$ and the similar equivalence $\mathsf{D}^{\mathsf{co}}(X\operatorname{-qcoh}_{\mathsf{fl}}) \simeq$ $D^{co}(X-qcoh)$ constructed using the dualizing complex \mathcal{D}_X^{\bullet} transform the external tensor product functor

$$\otimes_k : \mathsf{D}^{\mathsf{co}}(X\operatorname{\!-qcoh}_{\mathsf{fl}}) \times \mathsf{D}^{\mathsf{co}}(X\operatorname{\!-qcoh}_{\mathsf{fl}}) \longrightarrow \mathsf{D}^{\mathsf{co}}(X \times_k X\operatorname{\!-qcoh}_{\mathsf{fl}})$$

into the external tensor product functor

$$\otimes_k : \mathsf{D}^{\mathsf{co}}(X\operatorname{-qcoh}) \times \mathsf{D}^{\mathsf{co}}(X\operatorname{-qcoh}) \longrightarrow \mathsf{D}^{\mathsf{co}}(X \times_k X\operatorname{-qcoh})$$

since so do the functors $\mathcal{D}_X^{\bullet} \otimes_{\mathcal{O}_X} -$ and $\mathcal{D}_{X \times_k X}^{\bullet} \otimes_{\mathcal{O}_X \times_k X} -$. Let \mathcal{N}^{\bullet} and \mathcal{M}^{\bullet} be two complexes of injective quasicoherent sheaves on X, and let \mathcal{J}^{\bullet} be a complex of injective quasicoherent sheaves on $X \times_k X$ isomorphic to $\mathcal{N}^{\bullet} \otimes_k \mathcal{M}^{\bullet}$ in $\mathsf{D}^{\mathsf{co}}(X \times_k X \operatorname{-qcoh})$. Then in the coderived categories of quasicoherent sheaves one has

$$\mathcal{N}^{\bullet} \Box_{\mathcal{D}_{X}^{\bullet}} \mathcal{M}^{\bullet} = \mathcal{D}_{X}^{\bullet} \otimes_{\mathcal{O}_{X}} \Delta^{*}(\mathcal{H}om_{\mathcal{O}_{X}}(\mathcal{D}_{X}^{\bullet}, \mathcal{N}^{\bullet}) \otimes_{k} \mathcal{H}om_{\mathcal{O}_{X}}(\mathcal{D}_{X}^{\bullet}, \mathcal{M}^{\bullet}))$$
$$\simeq \mathcal{D}_{X}^{\bullet} \otimes_{\mathcal{O}_{X}} \Delta^{*} \mathcal{H}om_{\mathcal{O}_{X}}(\mathcal{D}_{X \times_{k} X}^{\bullet}, \mathcal{J}^{\bullet}) \simeq \mathbb{R}\Delta^{!}(\mathcal{N}^{\bullet} \otimes_{k} \mathcal{M}^{\bullet})$$

by [Positselski 2012, Theorem 5.15.3] applied to the proper morphism (actually, closed embedding) Δ . We have obtained the formula

$$\mathcal{N}^{\bullet} \square_{\pi^{+}\mathcal{O}_{\text{Spec}\,k}} \mathcal{M}^{\bullet} \simeq \mathbb{R}\Delta^{!}(\mathcal{N}^{\bullet} \otimes_{k} \mathcal{M}^{\bullet})$$
(37)

for the cotensor product of complexes of quasicoherent sheaves on the scheme X(see the end of Section 1.8 for the notation $\mathbb{R} f$ as applied to objects of the coderived category of quasicoherent sheaves).

B.2.6. Cotensor product of matrix factorizations. The equivalences of triangulated categories

$$\mathsf{D}^{\mathsf{co}}((X,\mathcal{L},w'')\operatorname{-qcoh}_{\mathsf{fl}}) \simeq \mathsf{D}^{\mathsf{co}}((X,\mathcal{L},w'')\operatorname{-qcoh})$$
$$\mathsf{D}^{\mathsf{co}}((X,\mathcal{L},w'+w'')\operatorname{-qcoh}_{\mathsf{fl}}) \simeq \mathsf{D}^{\mathsf{co}}((X,\mathcal{L},w'+w'')\operatorname{-qcoh})$$

constructed using a dualizing complex $\mathcal{D}_{X}^{\bullet}$ (see Section 2.5) transform the tensor product functor (14) into the tensor product functor (15). So one can use the same equivalences of categories together with the similar equivalence

$$\mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w')\operatorname{-qcoh}_{\mathsf{fl}}) \simeq \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w')\operatorname{-qcoh})$$

(constructed using the same dualizing complex \mathcal{D}_X^{\bullet}) in order to define a triangulated functor of two arguments

$$\Box_{\mathcal{D}_{X}^{\bullet}}: \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w')\operatorname{-qcoh}) \times \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w'')\operatorname{-qcoh}) \longrightarrow \mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w' + w'')\operatorname{-qcoh}), \quad (38)$$

which we call the *cotensor product* of matrix factorizations.

As in the case of complexes of quasicoherent sheaves, one explicitly has

$$\mathcal{N} \square_{\mathcal{D}_X^{\bullet}} \mathcal{M} = \mathcal{D}_X^{\bullet} \otimes_{\mathcal{O}_X} \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{D}_X^{\bullet}, \mathcal{N}) \otimes_{\mathcal{O}_X} \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{D}_X^{\bullet}, \mathcal{M})$$

for any injective quasicoherent matrix factorizations \mathcal{N} and \mathcal{M} on X, and also

$$\mathcal{N} \square_{\mathcal{D}_{V}^{\bullet}} \mathcal{M} = \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{D}_{X}^{\bullet}, \mathcal{N}) \otimes_{\mathcal{O}_{X}} \mathcal{M}$$

for any injective quasicoherent matrix factorization \mathcal{N} and any quasicoherent matrix factorization \mathcal{M} on X. As in Section B.2.3, \mathcal{N} and \mathcal{M} must be matrix factorizations of two sections w' and w'' of the same line bundle \mathcal{L} on a scheme X; then the cotensor product $\mathcal{N} \square_{\mathcal{D}_X^{\bullet}} \mathcal{M}$ is a matrix factorization of the section w' + w'' of the line bundle \mathcal{L} .

Remark B.2.6. While a matrix factorization version of Proposition B.2.5 is presented below, Remark B.2.3 explains the reason why a matrix factorization version of Theorem B.2.5 cannot be formulated in the way similar to the version for complexes of quasicoherent sheaves above. Still, let \mathcal{N} and \mathcal{M} be coherent matrix factorizations of sections w' and w'' of the same line bundle \mathcal{L} on a separated Noetherian scheme X with enough vector bundles. Let \mathcal{P} and \mathcal{Q} be coherent matrix factorizations of the potentials -w' and $-w'' \in \mathcal{L}(X)$ isomorphic to $\mathcal{H}om_{X-qc}(\mathcal{N}, \mathcal{D}_X^{\bullet})$ and $\mathcal{H}om_{X-qc}(\mathcal{M}, \mathcal{D}_X^{\bullet})$ in the respected coderived categories.

Let \mathcal{E}_{\bullet} and \mathcal{F}_{\bullet} be left resolutions of the matrix factorizations \mathcal{P} and \mathcal{Q} by locally free matrix factorizations of finite rank (of the respected potentials). Then the totalizations of the bounded-below complexes of matrix factorizations $\mathcal{H}om_{X-qc}(\mathcal{E}_{\bullet}, \mathcal{D}_X^{\bullet})$, $\mathcal{H}om_{X-qc}(\mathcal{F}_{\bullet}, \mathcal{D}_X^{\bullet})$, and $\mathcal{H}om_{X-qc}(\mathcal{E}_{\bullet} \otimes_{\mathcal{O}_X} \mathcal{F}_{\bullet}, \mathcal{D}_X^{\bullet})$ represent objects naturally isomorphic to \mathcal{N} , \mathcal{M} and $\mathcal{N} \Box_{\mathcal{D}_X^{\bullet}} \mathcal{M}$ in the coderived categories of matrix factorizations of the potentials w', w'', and w' + w'' (cf. Corollary 2.5).

For any quasicoherent matrix factorizations \mathcal{M} and \mathcal{N} of sections w' and w'' of the same line bundle \mathcal{L} on the scheme X, there is a natural morphism of quasicoherent matrix factorizations of the section w' + w'' of the line bundle \mathcal{L} on X

$$\mathcal{N} \otimes_{\mathcal{O}_X} \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{D}_X^{\bullet}, \mathcal{M}) \longrightarrow \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{H}om_{X-\mathsf{qc}}(\mathcal{N}, \mathcal{D}_X^{\bullet}), \mathcal{M})$$
(39)

constructed in the same way as it was done for complexes of quasicoherent sheaves in Section B.2.5.

1282

Proposition B.2.6. For any coherent matrix factorization \mathcal{N} and injective quasicoherent matrix factorization \mathcal{M} of sections w' and w'' of the same line bundle \mathcal{L} on a separated Noetherian scheme X, the natural morphism (39) is an isomorphism in the coderived category of quasicoherent matrix factorizations of the potential $w' + w'' \in \mathcal{L}(X)$.

Proof. The argument follows the lines of the proof of Proposition B.2.5. The lefthand side of the desired isomorphism is well-defined as a functor of the argument $\mathcal{N} \in \mathsf{D^{co}}((X, \mathcal{L}, w'')\text{-qcoh})$ taking values in $\mathsf{D^{co}}((X, \mathcal{L}, w' + w'')\text{-qcoh})$, while the right-hand side is well-defined as a functor of the argument $\mathcal{N} \in \mathsf{D^{abs}}((X, \mathcal{L}, w'')\text{-coh})$ taking values, say, in the same coderived category. Besides, the right-hand side, viewed as an object of the coderived category, only depends on the matrix factorization $\mathcal{H}om_{X-gc}(\mathcal{N}, \mathcal{D}_{Y}^{\bullet})$ viewed as an object of the absolute derived category.

Furthermore, the contravariant Serre–Grothendieck duality $\mathcal{H}om_{X-qc}(-, \mathcal{D}_X^{\bullet})$ is well-defined as a functor $D^{abs}((X, \mathcal{L}, w'')$ -qcoh) $\rightarrow D^{abs}((X, \mathcal{L}, -w'')$ -qcoh) and takes $D^{abs}((X, \mathcal{L}, w'')$ -coh) $\subset D^{abs}((X, \mathcal{L}, w'')$ -qcoh) into $D^{abs}((X, \mathcal{L}, -w'')$ -coh) \subset $D^{abs}((X, \mathcal{L}, -w'')$ -coh), inducing an equivalence between these two subcategories (see Proposition 2.5). In particular, one can conclude that all the functors involved are local in X, and it suffices to prove the desired assertion for matrix factorizations over affine open subschemes $U \subset X$.

Now let \mathcal{K} be a coherent matrix factorization of the potential -w'' isomorphic to $\mathcal{H}om_{U-qc}(\mathcal{N}, \mathcal{D}_U^{\bullet})$ in $\mathsf{D}^{\mathsf{abs}}((U, \mathcal{L}, -w'')\operatorname{-qcoh})$, and let \mathcal{E}_{\bullet} be its left resolution by locally free matrix factorizations of the same potential $-w'' \in \mathcal{L}(U)$. Then the matrix factorization $\mathcal{H}om_{X-qc}(\mathcal{K}, \mathcal{M})$ is isomorphic in $\mathsf{D}^{\mathsf{co}}((X, \mathcal{L}, w' + w'')\operatorname{-qcoh})$ to the totalization of the complex of matrix factorizations $\mathcal{H}om_{X-qc}(\mathcal{E}_{\bullet}, \mathcal{M})$ constructed by taking infinite direct sums along the diagonals; and the matrix factorization $\mathcal{N} \simeq \mathcal{H}om_{X-qc}(\mathcal{K}, \mathcal{D}_X^{\bullet})$ can be described similarly (cf. the proof of Corollary 2.5).

It remains to notice that the functor of tensoring with $\mathcal{H}om_{X-qc}(\mathcal{E}_{\bullet}, \mathcal{O}_X)$ and totalizing by taking infinite direct sums along the diagonals takes the homotopy equivalence

$$\mathcal{D}_X^{\bullet} \otimes_{\mathcal{O}_X} \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{D}_X^{\bullet}, \mathcal{M}) \longrightarrow \mathcal{M}$$

to a homotopy equivalence of matrix factorizations.

As in Section B.2.5, we finish by discussing the case of a separated scheme X of finite type over a field k. From now on we also assume that $\mathcal{L} = \mathcal{O}_X$. So let $w', w'' \in \mathcal{O}(X)$ be two global regular functions on X; as in Section B.2.2, we consider the regular function $w'_1 + w''_2 = p_1^* w' + p_2^* w''$ on $X \times_k X$. We use the dualizing complexes $\mathcal{D}_X^{\bullet} = \pi^+ \mathcal{O}_{\text{Spec } k}$ and $\mathcal{D}_{X \times_k X}^{\bullet} = (\pi \times_k \pi)^+ \mathcal{O}_{\text{Spec } k}$.

The equivalence of categories

$$\mathsf{D}^{\mathsf{co}}((X \times_k X, \mathcal{O}, w_1' + w_2'')\operatorname{\mathsf{-qcoh}}_{\mathsf{fl}}) \simeq \mathsf{D}^{\mathsf{co}}((X \times_k X, \mathcal{O}, w_1' + w_2'')\operatorname{\mathsf{-qcoh}})$$

constructed using the dualizing complex $\mathcal{D}_{X \times_k X}^{\bullet}$ and the similar equivalences of coderived categories of matrix factorizations of the potentials w' and w'' on X constructed using the dualizing complex \mathcal{D}_X^{\bullet} transform the external tensor product functor (cf. (9))

$$\begin{split} \otimes_k : \mathsf{D}^{\mathrm{co}}((X,\mathcal{O},w')\operatorname{-qcoh}_{\mathrm{fl}}) \times \mathsf{D}^{\mathrm{co}}((X,\mathcal{O},w'')\operatorname{-qcoh}_{\mathrm{fl}}) \\ & \longrightarrow \mathsf{D}^{\mathrm{co}}((X \times_k X, \mathcal{O}, w_1' + w_2'')\operatorname{-qcoh}_{\mathrm{fl}}) \end{split}$$

into the external tensor product functor

$$\begin{split} \otimes_k : \mathsf{D}^{\mathsf{co}}((X,\mathcal{O},w')\operatorname{-qcoh}) \times \mathsf{D}^{\mathsf{co}}((X,\mathcal{O},w'')\operatorname{-qcoh}) \\ & \longrightarrow \mathsf{D}^{\mathsf{co}}((X\times_k X,\,\mathcal{O},\,w_1'+w_2'')\operatorname{-qcoh}) \end{split}$$

since so do the functors $\mathcal{D}_X^{\bullet} \otimes_{\mathcal{O}_X} -$ and $\mathcal{D}_{X \times_k X}^{\bullet} \otimes_{\mathcal{O}_{X \times_k X}} -$.

Let \mathcal{N} and \mathcal{M} be injective quasicoherent matrix factorizations of the potentials w' and w'' on X, and let \mathcal{J} be an injective quasicoherent matrix factorization of the potential $w'_1 + w''_2$ on $X \times_k X$ isomorphic to $\mathcal{N} \otimes_k \mathcal{M}$ in

$$\mathsf{D^{co}}((X \times_k X, \mathcal{O}, w_1' + w_2'')$$
-qcoh).

Then in the coderived categories of quasicoherent matrix factorizations one has

$$\mathcal{N} \square_{\mathcal{D}_{X}^{\bullet}} \mathcal{M} = \mathcal{D}_{X}^{\bullet} \otimes_{\mathcal{O}_{X}} \Delta^{*}(\mathcal{H}om_{\mathcal{O}_{X}}(\mathcal{D}_{X}^{\bullet}, \mathcal{N}) \otimes_{k} \mathcal{H}om_{\mathcal{O}_{X}}(\mathcal{D}_{X}^{\bullet}, \mathcal{M}))$$
$$\simeq \mathcal{D}_{X}^{\bullet} \otimes_{\mathcal{O}_{X}} \Delta^{*} \mathcal{H}om_{\mathcal{O}_{X}}(\mathcal{D}_{X \times_{k} X}^{\bullet}, \mathcal{J}) \simeq \mathbb{R}\Delta^{!}(\mathcal{N} \otimes_{k} \mathcal{M})$$

by the result of Theorem 3.8 applied to the proper morphism Δ . We have obtained the formula

$$\mathcal{N} \square_{\pi^+ \mathcal{O}_{\text{Spec}\,k}} \mathcal{M} \simeq \mathbb{R}\Delta^! (\mathcal{N} \otimes_k \mathcal{M}) \tag{40}$$

for the cotensor product of quasicoherent matrix factorizations on the scheme X.

B.2.7. *Hochschild homology.* Let *X* be a separated scheme of finite type over a field *k* and $\pi : X \to \text{Spec } k$ be its structure morphism. Let $w \in \mathcal{O}(X)$ be a global regular function; as in Section B.2.4, we assume that the morphism of schemes $w : X \to \mathbb{A}^1_k$ is flat. Consider the scheme $X \times_k X$ and endow it with the potential $w_2 - w_1 = p_2^*(w) - p_1^*(w)$. Let $\Delta : X \to X \times_k X$ denote the diagonal morphism.

Theorem B.2.7. In the assumptions of Theorem B.2.4, there is a natural isomorphism between the Hochschild homology module $HH_*(DG^{abs}((X, \mathcal{O}, w)\text{-coh}))$ over the algebra $HH^*(DG^{abs}((X, \mathcal{O}, w)\text{-coh}))$ and the Ext module

$$\operatorname{Hom}_{\mathsf{D}^{\mathrm{co}}((X \times_k X, \mathcal{O}, w_2 - w_1) \operatorname{-qcoh})}(\Delta_* \mathcal{O}_X, \Delta_* \mathcal{D}_X^{\bullet}[*])$$
over the algebra

$$\operatorname{Hom}_{\mathsf{D}^{\mathrm{co}}((X \times_k X, \mathcal{O}, w_2 - w_1) \operatorname{-qcoh})}(\Delta_* \mathcal{D}_X^{\bullet}, \Delta_* \mathcal{D}_X^{\bullet}[*]).$$

Here \mathcal{D}_X^{\bullet} *denotes the dualizing complex* $\pi^+ \mathcal{O}_{\text{Spec } k}$ *on X*.

Proof. By the definition, the Hochschild homology of a $\mathbb{Z}/2$ -graded DG-category DG is the $\mathbb{Z}/2$ -graded vector space $\operatorname{Tor}_*^{\mathsf{DG}\otimes_k \mathsf{DG}^{\mathsf{op}}}(\mathsf{DG}, \mathsf{DG})$ for the diagonal right and left DG-modules DG over the DG-category DG $\otimes_k \mathsf{DG}^{\mathsf{op}}$ [Polishchuk and Positselski 2012, Sections 2.4 and 3.1]. This is the conventional derived tensor product ("of the first kind") of a left and a right DG-module over a small DG-category. The Hochschild cohomology algebra $\operatorname{Hom}_{\mathsf{D}(\mathsf{DG}\otimes_k \mathsf{DG}^{\mathsf{op}})}(\mathsf{DG}, \mathsf{DG}[*])$ acts on the Hochschild homology space via its action on, say, the first argument of the Tor.

As in the proof of Theorem B.2.4, we set $\mathsf{DG}_w = \mathsf{DG}^{\mathsf{abs}}((X, \mathcal{O}, w)\operatorname{-coh})$; accordingly, $\mathsf{DG}_{-w} = \mathsf{DG}^{\mathsf{abs}}((X, \mathcal{O}, -w)\operatorname{-coh})$ and

$$\mathsf{DG}_{w_2-w_1} = \mathsf{DG}^{\mathsf{abs}}((X \times_k X, \mathcal{O}, w_2-w_1)\operatorname{-coh}).$$

The DG-functor $DG_w^{op} \otimes_k DG_w \to DG_{w_2-w_1}$ (29) induces a fully faithful functor between the homotopy categories $H^0(DG_w)^{op} \otimes_k H^0(DG_w) \to H^0(DG_{w_2-w_1})$ such that every object in the target category can be obtained from objects in the image using the operations of a cone and the passage to a direct summand.

Let DG(mod-DG^{op}_w \otimes_k DG_w) denote the DG-category version of the (conventional) derived category of right DG-modules over the DG-category DG^{op}_w \otimes_k DG (i.e., contravariant DG-functors from DG^{op}_w \otimes_k DG into the DG-category DG(k-vect) of $\mathbb{Z}/2$ -graded complexes of k-vector spaces). Let DG(mod-DG^{op}_w \otimes_k DG_w)⁰ \subset DG(mod-DG^{op}_w \otimes_k DG_w) denote the full DG-subcategory of DG-modules corresponding to compact objects of the derived category D(mod-DG^{op}_w \otimes_k DG_w) of right DG-modules.

The derived tensor product with the left DG-module DG_w over $DG_w^{op} \otimes_k DG_w$ can be viewed as a covariant DG-functor $DG(\text{mod}-DG_w^{op} \otimes_k DG_w) \rightarrow DG(k$ -vect). We are interested in the restriction of this DG-functor to the DG-subcategory $DG(\text{mod}-DG_w^{op} \otimes_k DG_w)^0$; let us denote it by

$$F : \mathsf{DG}(\mathsf{mod}\mathsf{-}\mathsf{DG}_w^{\mathsf{op}}\otimes_k \mathsf{DG}_w)^0 \longrightarrow \mathsf{D}(k\operatorname{-vect})$$

There is a natural DG-functor $DG_w^{op} \otimes_k DG_w \to DG(\text{mod}-DG_w^{op} \otimes_k DG_w)^0$ assigning to any object of $DG_w^{op} \otimes_k DG_w$ the contravariant DG-functor represented by it. Similarly one constructs a DG-functor $DG_{w_2-w_1} \to DG(\text{mod}-DG_w^{op} \otimes_k DG_w)^0$ whose composition with the DG-functor $DG_w^{op} \otimes_k DG_w \to DG_{w_2-w_1}$ is naturally quasi-isomorphic to the DG-functor $DG_w^{op} \otimes_k DG_w \to DG(\text{mod}-DG_w^{op} \otimes_k DG_w)^0$.

It is claimed that the composition of the DG-functor

$$\mathsf{DG}_{w_2-w_1} \longrightarrow \mathsf{DG}(\mathsf{mod}\mathsf{-}\mathsf{DG}_w^{\mathsf{op}} \otimes_k \mathsf{DG}_w)^0$$

with the DG-functor $F : DG(mod-DG_w^{op} \otimes_k DG_w)^0 \to D(k\text{-vect})$ is naturally quasiisomorphic to the DG-functor $Hom_{DG_{w_2-w_1}}(\Delta_*\mathcal{O}_X, -)$. Since the derived categories of left DG-modules over $DG_{w_1-w_2}$ and $DG_w^{op} \otimes_k DG_w$ are equivalent, it suffices to construct a quasi-isomorphism between the compositions of the two DG-functors in question with the DG-functor $DG_w^{op} \otimes_k DG_w \to DG_{w_2-w_1}$.

Indeed, let $(\mathcal{K}^{op}, \mathcal{M})$ be an object of $\mathsf{DG}_w^{op} \otimes_k \mathsf{DG}_w$. Then the functor of the (derived or underived) tensor product with the diagonal left DG -module DG_w takes the right DG -module over $\mathsf{DG}_w^{op} \otimes_k \mathsf{DG}_w$ represented by $(\mathcal{K}^{op}, \mathcal{M})$ to the complex of k-vector spaces $\mathsf{Hom}_{\mathsf{DG}_w}(\mathcal{K}, \mathcal{M})$. Substituting $\mathcal{K} = \mathcal{H}om_{X-\mathsf{qc}}(\mathcal{N}, \mathcal{D}_X^{\bullet})$ with $\mathcal{N} \in \mathsf{DG}_{-w}$ and assuming \mathcal{M} to be represented by an injective matrix factorization isomorphic to the given coherent one in $\mathsf{DG}^{co}((X, \mathcal{O}, w)\operatorname{-qcoh})$, we have to compute the complex of k-vector spaces $\mathsf{Hom}_{(X,\mathcal{O},w)\operatorname{-qcoh}}(\mathcal{H}om_{X-\mathsf{qc}}(\mathcal{N}, \mathcal{D}_X^{\bullet}), \mathcal{M})$.

Now the formula (24) together with Lemma B.2.3 allow us to interpret this complex as $\mathbb{R}\Gamma(X, \mathcal{H}om_{X-qc}(\mathcal{H}om_{X-qc}(\mathcal{N}, \mathcal{D}_X^{\bullet}), \mathcal{M}))$. According to Proposition B.2.6 together with the formula (40), this is the same as $\mathbb{R}\Gamma(X, \mathbb{R}\Delta^!(\mathcal{N} \otimes_k \mathcal{M}))$, or, in other notation, $\operatorname{Hom}_{\mathsf{DG}^{co}}((X, \mathcal{O}, 0)\operatorname{-qcoh})(\mathcal{O}_X, \mathbb{R}\Delta^!(\mathcal{N} \otimes_k \mathcal{M}))$. Finally, the adjunction of Δ_* and $\mathbb{R}\Delta^!$ allows us to rewrite the complex in question as

$$\operatorname{Hom}_{\mathsf{DG}^{\operatorname{co}}((X \times_k X, \mathcal{O}, w_2 - w_1) \operatorname{-qcoh})}(\Delta_* \mathcal{O}_X, \mathcal{N} \otimes_k \mathcal{M}).$$

The desired quasi-isomorphism of DG-functors is obtained.

It remains to recall that, according to the proof of Theorem B.2.4, the diagonal right DG-module DG_w over $DG_w^{op} \otimes_k DG_w$ is represented by the object $\Delta_* \mathcal{D}_X^{\bullet} \in DG_{w_2-w_1}$, in order to finish our proof here.

Remark B.2.7. The Hochschild homology module $HH_*((DG^{abs}(X, \mathcal{O}, w)\text{-coh}))$ over the Hochschild cohomology algebra $HH^*((DG^{abs}(X, \mathcal{O}, w)\text{-coh}))$ can be also computed as the Ext module

$$\operatorname{Hom}_{\mathsf{D}^{\mathrm{co}}((X \times_k X, \mathcal{O}, w_1 - w_2) \operatorname{-qcoh})}(\Delta_* \mathcal{O}_X, \Delta_* \mathcal{D}_X^{\bullet}[*])$$

over the Ext algebra

$$\operatorname{Hom}_{\mathsf{D}^{\mathsf{abs}}((X \times_k X, \mathcal{O}, w_1 - w_2) \operatorname{-coh})}(\Delta_* \mathcal{O}_X, \Delta_* \mathcal{O}_X[*])^{\operatorname{op}},$$

according to Remark B.2.4. Moreover, the contravariant Serre duality for matrix factorizations over $X \times_k X$ can be used in order to obtain an alternative proof of our Hochschild homology computation. Indeed, for any coherent matrix factorizations

1286

Coherent analogues of matrix factorizations and relative singularity categories 1287

 \mathcal{N} and \mathcal{M} of the potentials -w and w on X there are natural quasi-isomorphisms

$$\begin{aligned} &\operatorname{Hom}_{\mathsf{DG}^{\mathsf{abs}}((X,\mathcal{O},w)\operatorname{-coh})}(\mathcal{H}om_{X\operatorname{-qc}}(\mathcal{N},\mathcal{D}_{X}^{\bullet}),\mathcal{M}) \\ &\simeq \operatorname{Hom}_{\mathsf{DG}^{\mathsf{abs}}((X,\mathcal{O},w)\operatorname{-coh})}(\mathcal{H}om_{X\operatorname{-qc}}(\mathcal{N},\mathcal{D}_{X}^{\bullet}),\mathcal{H}om_{X\operatorname{-qc}}(\mathcal{H}om_{X\operatorname{-qc}}(\mathcal{M},\mathcal{D}_{X}^{\bullet}),\mathcal{D}_{X}^{\bullet})) \\ &\simeq \operatorname{Hom}_{\mathsf{DG}^{\mathsf{abs}}((X\times_{k}X,\mathcal{O},w_{1}-w_{2})\operatorname{-coh})}(\mathcal{H}om_{X\operatorname{-qc}}(\mathcal{N},\mathcal{D}_{X}^{\bullet})\otimes_{k}\mathcal{H}om_{X\operatorname{-qc}}(\mathcal{M},\mathcal{D}_{X}^{\bullet}),\Delta_{*}\mathcal{D}_{X}^{\bullet}) \\ &\simeq \operatorname{Hom}_{\mathsf{DG}^{\mathsf{abs}}((X\times_{k}X,\mathcal{O},w_{1}-w_{2})\operatorname{-coh})}(\mathcal{H}om_{X\times_{k}X\operatorname{-qc}}(\mathcal{N}\otimes_{k}\mathcal{M},\mathcal{D}_{X\times_{k}X}^{\bullet}),\Delta_{*}\mathcal{D}_{X}^{\bullet}) \\ &\simeq \operatorname{Hom}_{\mathsf{DG}^{\mathsf{abs}}((X\times_{k}X,\mathcal{O},w_{2}-w_{1})\operatorname{-coh})}(\Delta_{*}\mathcal{O}_{X},\mathcal{N}\otimes_{k}\mathcal{M}) \end{aligned}$$

by Proposition 2.5 and the proof of Theorem B.2.4. In other words, while the right diagonal DG-module DG_w over $DG_w^{op} \otimes_k DG_w$ is represented by the object

$$\Delta_* \mathcal{D}^{\bullet}_X \in \mathsf{DG}^{\mathsf{abs}}((X \times_k X, \mathcal{O}, w_2 - w_1) \operatorname{-coh})$$

as a contravariant DG-functor on $\mathsf{DG}_w^{\mathsf{op}} \otimes_k \mathsf{DG}_w \subset \mathsf{DG}_{w_2-w_1}$, the left diagonal DG-bimodule DG_w over $\mathsf{DG}_w^{\mathsf{op}} \otimes_k \mathsf{DG}_w$ is represented by the object

$$\Delta_* \mathcal{O}^{\bullet}_X \in \mathsf{DG}^{\mathsf{abs}}((X \times_k X, \mathcal{O}, w_2 - w_1) \operatorname{-coh})$$

as a covariant DG-functor on

$$\mathsf{DG}_w^{\mathsf{op}} \otimes_k \mathsf{DG}_w \subset \mathsf{DG}_{w_2-w_1} = \mathsf{DG}^{\mathsf{abs}}((X \times_k X, \mathcal{O}, w_2-w_1)\text{-coh}).$$

B.2.8. Direct sum over the critical values. Let X be a separated scheme of finite type over a field k and $\pi : X \to \text{Spec } k$ be its structure morphism. As in Sections B.2.5–B.2.7 (see also Section 3.7), we choose the dualizing complex $\mathcal{D}_X^{\bullet} \simeq \pi^+ \mathcal{O}_{\text{Spec } k}$ on X. Let $w \in \mathcal{O}(X)$ be a global regular function on X such that the morphism of schemes $w : X \to \mathbb{A}_k^1$ is flat (cf. [Orlov 2004; 2012]).

Let $c_1, \ldots, c_n \in k$ be a finite number of different elements of the ground field. Assume that there exist closed subschemes $Z_i \subset X$ such that the function w is noncritical on $X \setminus (Z_1 \cup \cdots \cup Z_n)$, the restriction of w to Z_i is equal to the constant c_i , and the schemes Z_i admit smooth stratifications over k.

In particular, if the field k is perfect, it suffices to require that the function w has only a finite number of critical values $c_1, \ldots, c_n \in \mathbb{A}^1_k$ (i.e., the open subscheme $\mathbb{A}^1_{k,f} \subset \mathbb{A}^1_k$ is nonempty; see Section B.2.1), and all of these values belong to the field k (rather than its algebraic closure). When the field k has zero characteristic, the former condition holds automatically. Then one simply takes Z_i to be the zero locus of the function $w_i - c_i$ on X.

Consider the Cartesian square $X \times_k X$ with the global function $w_2 - w_1 = p_2^*(w) - p_1^*(w)$ on it. Let $\Delta : X \to X \times_k X$ denote the diagonal morphism. The following result is to be compared with [Polishchuk and Positselski 2012, Corollary 4.10].

Corollary B.2.8. There are natural isomorphisms of $\mathbb{Z}/2$ -graded k-algebras

$$\bigoplus_{i=1}^{n} HH^{*}(\mathsf{DG}^{\mathsf{abs}}((X, \mathcal{O}, w - c_{i})\operatorname{-coh})) \simeq \operatorname{Hom}_{\mathsf{D}^{\mathsf{co}}((X \times_{k} X, \mathcal{O}, w_{2} - w_{1})\operatorname{-qcoh})}(\Delta_{*}\mathcal{D}_{X}^{\bullet}, \Delta_{*}\mathcal{D}_{X}^{\bullet}[*]) \simeq \operatorname{Hom}_{\mathsf{D}^{\mathsf{abs}}((X \times_{k} X, \mathcal{O}, w_{1} - w_{2})\operatorname{-coh})}(\Delta_{*}\mathcal{O}_{X}, \Delta_{*}\mathcal{O}_{X}[*])^{\operatorname{op}}.$$
(41)

There are also natural isomorphisms of $\mathbb{Z}/2$ -graded k-modules

$$\bigoplus_{i=1}^{n} HH_{*}(\mathsf{DG}^{\mathsf{abs}}((X, \mathcal{O}, w-c_{i})\operatorname{-coh})) \\
\simeq \operatorname{Hom}_{\mathsf{D}^{\mathsf{co}}((X\times_{k}X, \mathcal{O}, w_{2}-w_{1})\operatorname{-qcoh})}(\Delta_{*}\mathcal{O}_{X}, \Delta_{*}\mathcal{D}_{X}^{\bullet}[*]) \\
\simeq \operatorname{Hom}_{\mathsf{D}^{\mathsf{co}}((X\times_{k}X, \mathcal{O}, w_{1}-w_{2})\operatorname{-qcoh})}(\Delta_{*}\mathcal{O}_{X}, \Delta_{*}\mathcal{D}_{X}^{\bullet}[*]) \quad (42)$$

over the $\mathbb{Z}/2$ -graded k-algebra (41).

Proof. For each i = 1, ..., n, let Y_i denote the open subscheme $X \setminus \bigcup_{j \neq i} Z_i \subset X$. Let $w_i \in \mathcal{O}(Y_i)$ denote the restriction of the regular function $w - c_i$ to Y_i . The argument is based on the results of Sections B.2.4 and B.2.7 applied to the schemes Y_i (or their open subschemes) endowed with the potentials w_i .

The restriction of morphisms (in the coderived categories) of quasicoherent matrix factorizations to the open subschemes $Y_i \subset X$ defines a $\mathbb{Z}/2$ -graded *k*-algebra morphism from the (middle or) right-hand side to the left-hand side of (41), and a $\mathbb{Z}/2$ -graded *k*-module morphism from the (middle or) right-hand side to the left-hand side to the left-hand side of (42). It remains to show that these morphisms are isomorphisms.

For this purpose, one can start with replacing $\Delta_* \mathcal{D}^{\bullet}_X$ or $\Delta_* \mathcal{O}_X$ in the second argument of the Hom spaces in the middle or right-hand sides of (41) and (42) with an injective matrix factorization \mathcal{J} on $X \times_k X$ representing the same object in the coderived category. Then one notices that the restriction from $X \times_k X$ to its open subscheme $V = \bigcup_{i=1}^n Y_i \times_k Y_i$ does not change the Hom spaces in the right-hand sides, as the image of Δ is contained in V.

Finally, one writes down the Čech resolution of the matrix factorization $\mathcal{J}|_V$ corresponding to the covering of the scheme V by its open subschemes $Y_i \times_k Y_i$. This is a finite acyclic complex of injective matrix factorizations, so applying the functor $\operatorname{Hom}_{(V, \mathcal{O}, (w_2-w_1)|_V)\operatorname{-qcoh}}(\mathcal{K}, -)$ from any quasicoherent matrix factorization \mathcal{K} preserves its acyclicity. Since the Hom spaces on any intersection of at least two different open subschemes in the covering are zero by Theorems B.2.4– B.2.7 (as w is noncritical on $Y_i \cap Y_j$ for any $i \neq j$), the desired isomorphisms follow. \Box

Remark B.2.8. The Hochschild cohomology algebra and the Hochschild homology module of the DG-category version $DG^{b}(X-coh)$ of the bounded derived category $D^{b}(X-coh)$ of (complexes of) coherent sheaves on a separated scheme X of finite type over a field k can be computed in the way similar to (but simpler than) the

Coherent analogues of matrix factorizations and relative singularity categories 1289

above. The answers are the same as in Theorems B.2.4 and B.2.7:

$$HH^{*}(\mathsf{DG}^{\mathsf{b}}(X\operatorname{-coh})) \simeq \operatorname{Hom}_{\mathsf{D}(X \times_{k} X\operatorname{-qcoh})}(\Delta_{*}\mathcal{D}_{X}^{\bullet}, \ \Delta_{*}\mathcal{D}_{X}^{\bullet}[*])$$
$$\simeq \operatorname{Hom}_{\mathsf{D}^{\mathsf{b}}(X \times_{k} X\operatorname{-coh})}(\Delta_{*}\mathcal{O}_{X}, \Delta_{*}\mathcal{O}_{X}[*])^{\mathsf{op}} \quad (43)$$

and

$$HH_*(\mathsf{DG}^{\mathsf{b}}(X\operatorname{-coh})) \simeq \operatorname{Hom}_{\mathsf{D}(X \times_k X\operatorname{-qcoh})}(\Delta_*\mathcal{O}_X, \ \Delta_*\mathcal{D}_X^{\bullet}[*]), \tag{44}$$

the only difference being that $DG^{b}(X\operatorname{-coh})$ is a \mathbb{Z} -graded DG-category and the righthand sides describe the Hochschild (co)homology as a \mathbb{Z} -graded algebra and module. The only assumption is that the scheme X should admit a smooth stratification over k (i.e., it suffices that the field k be perfect).

Acknowledgments

Positselski is grateful to Daniel Pomerleano and Kevin Lin for sending him a preliminary version of their manuscript [Lin and Pomerleano 2013], which inspired the present work. We also wish to thank Roman Bezrukavnikov, Alexander Kuznetsov, and Amnon Neeman for helpful conversations, Alexander Polishchuk and Henning Krause for stimulating questions, and Dmitri Orlov for his insightful remarks on an early version of this manuscript. Finally, Positselski would like to thank an anonymous referee of an early version for detailed suggestions on the improvement of the exposition. It was initially my attempt to implement those suggestions that eventually resulted in the growth of the size of this paper by a factor of almost seven. Efimov was partially supported by RFBR grant 2998.2014.1 and RFBR research project 15-51-50045. Positselski was supported in part by a Simons Foundation grant and RFBR grants in Moscow and by a fellowship from the Lady Davis Foundation at the Technion. The paper was prepared within the framework of a subsidy granted to the HSE by the Government of the Russian Federation for the implementation of the Global Competitiveness Program.

References

- [Balmer and Schlichting 2001] P. Balmer and M. Schlichting, "Idempotent completion of triangulated categories", *J. Algebra* 236:2 (2001), 819–834. MR 2002a:18013 Zbl 0977.18009
- [Bass 1963] H. Bass, "Big projective modules are free", *Illinois J. Math.* **7** (1963), 24–31. MR 26 #1341 Zbl 0115.26003
- [Becker 2014] H. Becker, "Models for singularity categories", *Adv. Math.* **254** (2014), 187–232. MR 3161097 Zbl 06284998
- [Bourbaki 1980] N. Bourbaki, *Algèbre homologique: Chapitre 10*, Masson, Paris, 1980. MR 82j:18022 Zbl 0455.18010

[Buchweitz 1986] R.-O. Buchweitz, "Maximal Cohen–Macaulay modules and Tate-cohomology over Gorenstein rings", preprint, 1986, available at http://hdl.handle.net/1807/16682.

- [Cartan and Eilenberg 1956] H. Cartan and S. Eilenberg, *Homological algebra*, Princeton University Press, 1956. MR 17,1040e Zbl 0075.24305
- [Chen 2010] X.-W. Chen, "Unifying two results of Orlov on singularity categories", *Abh. Math. Semin. Univ. Hambg.* **80**:2 (2010), 207–212. MR 2011j:14036 Zbl 1214.18013
- [Drinfeld 2004] V. Drinfeld, "DG quotients of DG categories", *J. Algebra* **272**:2 (2004), 643–691. MR 2006e:18018 Zbl 1064.18009
- [Drinfeld 2006] V. Drinfeld, "Infinite-dimensional vector bundles in algebraic geometry: an introduction", pp. 263–304 in *The unity of mathematics*, edited by P. Etingof et al., Progr. Math. **244**, Birkhäuser, Boston, 2006. MR 2007d:14038 Zbl 1108.14012
- [Dyckerhoff and Murfet 2013] T. Dyckerhoff and D. Murfet, "Pushing forward matrix factorizations", *Duke Math. J.* **162**:7 (2013), 1249–1311. MR 3079249 Zbl 1273.14014
- [Efimov 2012] A. I. Efimov, "Cyclic homology of categories of matrix factorizations", preprint, 2012. arXiv 1212.2859
- [Efimov 2013] A. I. Efimov, "Homotopy finiteness of some DG categories from algebraic geometry", preprint, 2013. arXiv 1308.0135
- [Eisenbud 1980] D. Eisenbud, "Homological algebra on a complete intersection, with an application to group representations", *Trans. Amer. Math. Soc.* **260**:1 (1980), 35–64. MR 82d:13013 Zbl 0444.13006
- [Getzler and Jones 1990] E. Getzler and J. D. S. Jones, " A_{∞} -algebras and the cyclic bar complex", *Illinois J. Math.* **34**:2 (1990), 256–283. MR 91e:19001 Zbl 0701.55009
- [Goodearl and Warfield 1989] K. R. Goodearl and R. B. Warfield, Jr., *An introduction to noncommutative Noetherian rings*, London Mathematical Society Student Texts **16**, Cambridge University Press, 1989. MR 91c:16001 Zbl 0679.16001
- [Grothendieck 1961] A. Grothendieck, "Éléments de géométrie algébrique, III: Étude cohomologique des faisceaux cohérents I", *Inst. Hautes Études Sci. Publ. Math.* 11 (1961), 167. MR 29 #1209
- [Grothendieck 1965] A. Grothendieck, "Éléments de géométrie algébrique, IV; Étude locale des schémas et des morphismes de schémas II", *Inst. Hautes Études Sci. Publ. Math.* 24 (1965), 231. MR 33 #7330 Zbl 0135.39701
- [Grothendieck 1967] A. Grothendieck, "Éléments de géométrie algébrique, IV: Étude locale des schémas et des morphismes de schémas IV", *Inst. Hautes Études Sci. Publ. Math.* 32 (1967), 361. MR 39 #220 Zbl 0153.22301
- [Hartshorne 1966] R. Hartshorne, *Residues and duality*, Lecture Notes in Mathematics **20**, Springer, Berlin, 1966. MR 36 #5145 Zbl 0212.26101
- [Hartshorne 1977] R. Hartshorne, *Algebraic geometry*, Graduate Texts in Mathematics **52**, Springer, New York, 1977. MR 57 #3116 Zbl 0531.14001
- [Iyengar and Krause 2006] S. Iyengar and H. Krause, "Acyclicity versus total acyclicity for complexes over Noetherian rings", *Doc. Math.* **11** (2006), 207–240. MR 2007h:16013 Zbl 1119.13014
- [Jørgensen 2005] P. Jørgensen, "The homotopy category of complexes of projective modules", *Adv. Math.* **193**:1 (2005), 223–232. MR 2005m:18013 Zbl 1068.18012
- [Kaplansky 1958] I. Kaplansky, "Projective modules", Ann. of Math (2) **68** (1958), 372–377. MR 20 #6453 Zbl 0083.25802
- [Kapustin and Li 2003] A. Kapustin and Y. Li, "D-branes in Landau–Ginzburg models and algebraic geometry", *J. High Energy Phys.* 12 (2003), 005, 44 pp. MR 2005b:81179b
- [Krause 2005] H. Krause, "The stable derived category of a Noetherian scheme", *Compos. Math.* **141**:5 (2005), 1128–1162. MR 2006e:18019 Zbl 1090.18006

Coherent analogues of matrix factorizations and relative singularity categories 1291

- [Lin and Pomerleano 2013] K. H. Lin and D. Pomerleano, "Global matrix factorizations", *Math. Res. Lett.* **20**:1 (2013), 91–106. MR 3126725 Zbl 1285.14019
- [Lunts 2010] V. A. Lunts, "Categorical resolution of singularities", *J. Algebra* **323**:10 (2010), 2977–3003. MR 2011d:18019 Zbl 1202.18006
- [Murfet 2007] D. Murfet, *The mock homotopy category of projectives and Grothendieck duality*, Ph.D. thesis, Australian National University, 2007, available at http://www.therisingsea.org/thesis.pdf.
- [Murfet 2013] D. Murfet, "Residues and duality for singularity categories of isolated Gorenstein singularities", *Compos. Math.* **149**:12 (2013), 2071–2100. MR 3143706 Zbl 06250163
- [Neeman 1992] A. Neeman, "The connection between the *K*-theory localization theorem of Thomason, Trobaugh and Yao and the smashing subcategories of Bousfield and Ravenel", *Ann. Sci. École Norm. Sup.* (4) **25**:5 (1992), 547–566. MR 93k:18015 Zbl 0868.19001
- [Neeman 1996] A. Neeman, "The Grothendieck duality theorem via Bousfield's techniques and Brown representability", *J. Amer. Math. Soc.* **9**:1 (1996), 205–236. MR 96c:18006 Zbl 0864.14008
- [Neeman 2001] A. Neeman, *Triangulated categories*, Annals of Mathematics Studies **148**, Princeton University Press, 2001. MR 2001k:18010 Zbl 0974.18008
- [Neeman 2008] A. Neeman, "The homotopy category of flat modules, and Grothendieck duality", *Invent. Math.* **174**:2 (2008), 255–308. MR 2009h:16008 Zbl 1184.18008
- [Orlov 2004] D. O. Orlov, "Triangulated categories of singularities and D-branes in Landau–Ginzburg models", *Tr. Mat. Inst. Steklova* **246** (2004), 240–262. In Russian; translated in *Proc. Steklov Math. Inst.* **246**,(2004), 227–248. MR 2006i:81173 Zbl 1101.81093
- [Orlov 2006] D. O. Orlov, "Triangulated categories of singularities, and equivalences between Landau–Ginzburg models", *Mat. Sb.* **197**:12 (2006), 1827–1840. MR 2009g:14013 Zbl 1161.14301
- [Orlov 2011] D. Orlov, "Formal completions and idempotent completions of triangulated categories of singularities", *Adv. Math.* **226**:1 (2011), 206–217. MR 2012c:14035 Zbl 1216.18012
- [Orlov 2012] D. Orlov, "Matrix factorizations for nonaffine LG-models", *Math. Ann.* **353**:1 (2012), 95–108. MR 2910782 Zbl 1243.81178
- [Polishchuk and Positselski 2012] A. Polishchuk and L. Positselski, "Hochschild (co)homology of the second kind I", *Trans. Amer. Math. Soc.* **364**:10 (2012), 5311–5368. MR 2931331 Zbl 1285.16005
- [Polishchuk and Vaintrob 2011] A. Polishchuk and A. Vaintrob, "Matrix factorizations and singularity categories for stacks", *Ann. Inst. Fourier (Grenoble)* **61**:7 (2011), 2609–2642. MR 3112502 Zbl 1278.13014
- [Polishchuk and Vaintrob 2014] A. Polishchuk and A. Vaintrob, "Matrix factorizations and cohomological field theories", *Journ. für die reine und angewandte Math.* (online publication April 2014).
- [Positselski 2010] L. Positselski, *Homological algebra of semimodules and semicontramodules*, Instytut Matematyczny Polskiej Akademii Nauk. Monografie Matematyczne (New Series) **70**, Birkhäuser/Springer Basel AG, 2010. MR 2012c:16030 Zbl 1202.18001
- [Positselski 2011a] L. Positselski, "Mixed Artin–Tate motives with finite coefficients", *Mosc. Math. J.* **11**:2 (2011), 317–402, 407–408. MR 2859239 Zbl 1273.12004
- [Positselski 2011b] L. Positselski, "Two kinds of derived categories, Koszul duality, and comodulecontramodule correspondence", pp. vi+133 Mem. Amer. Math. Soc. 996, Amer. Math. Soc., Providence, RI, 2011. MR 2012f:16023
- [Positselski 2012] L. Positselski, "Contraherent cosheaves", preprint, 2012. arXiv 1209.2995
- [Positsel'skiĭ 1993] L. E. Positsel'skiĭ, "Nonhomogeneous quadratic duality and curvature", *Funk-tsional. Anal. i Prilozhen.* 27:3 (1993), 57–66. In Russian; translated in *Funct. Anal. Appl.* 27:3 (1993), 197–283. MR 95h:16041 Zbl 0826.16041

- [Raynaud and Gruson 1971] M. Raynaud and L. Gruson, "Critères de platitude et de projectivité: Techniques de "platification" d'un module", *Invent. Math.* **13** (1971), 1–89. MR 46 #7219 Zbl 0227.14010
- [Thomason 1997] R. W. Thomason, "The classification of triangulated subcategories", *Compositio Math.* **105**:1 (1997), 1–27. MR 98b:18017 Zbl 0873.18003
- [Thomason and Trobaugh 1990] R. W. Thomason and T. Trobaugh, "Higher algebraic *K*-theory of schemes and of derived categories", pp. 247–435 in *The Grothendieck Festschrift, Vol. III*, edited by P. Cartier et al., Progr. Math. **88**, Birkhäuser, Boston, 1990. MR 92f:19001 Zbl 0731.14001

Communicated by Michel Van den Bergh Received 2014-12-26 Accepted 2015-03-27

1292

efimov@mccme.ru	Department of Algebraic Geometry, Steklov Mathematical Institute of the Russian Academy of Sciences, Gubkina str., 8, Moscow, 119991, Russia, and Laboratory of Algebraic Geometry, Higher School of Economics, 7 Vavilova str., Moscow, 117312, Russia
posic@mccme.ru	Mathematics Department, Technion – Israel Institute of Technology, 32000 Haifa, Israel, and Sector of Algebra and Number Theory, Institute for Information Transmission Problems, Moscow, 127994, Russia, and Laboratory of Algebraic Geometry, National Research University Higher School of Economics, Moscow, 117312, Russia

Guidelines for Authors

Authors may submit manuscripts in PDF format on-line at the Submission page at the ANT website.

Originality. Submission of a manuscript acknowledges that the manuscript is original and and is not, in whole or in part, published or under consideration for publication elsewhere. It is understood also that the manuscript will not be submitted elsewhere while under consideration for publication in this journal.

Language. Articles in *ANT* are usually in English, but articles written in other languages are welcome.

Length There is no a priori limit on the length of an *ANT* article, but *ANT* considers long articles only if the significance-to-length ratio is appropriate. Very long manuscripts might be more suitable elsewhere as a memoir instead of a journal article.

Required items. A brief abstract of about 150 words or less must be included. It should be self-contained and not make any reference to the bibliography. If the article is not in English, two versions of the abstract must be included, one in the language of the article and one in English. Also required are keywords and subject classifications for the article, and, for each author, postal address, affiliation (if appropriate), and email address.

Format. Authors are encouraged to use LATEX but submissions in other varieties of TEX, and exceptionally in other formats, are acceptable. Initial uploads should be in PDF format; after the refereeing process we will ask you to submit all source material.

References. Bibliographical references should be complete, including article titles and page ranges. All references in the bibliography should be cited in the text. The use of BibT_EX is preferred but not required. Tags will be converted to the house format, however, for submission you may use the format of your choice. Links will be provided to all literature with known web locations and authors are encouraged to provide their own links in addition to those supplied in the editorial process.

Figures. Figures must be of publication quality. After acceptance, you will need to submit the original source files in vector graphics format for all diagrams in your manuscript: vector EPS or vector PDF files are the most useful.

Most drawing and graphing packages (Mathematica, Adobe Illustrator, Corel Draw, MATLAB, etc.) allow the user to save files in one of these formats. Make sure that what you are saving is vector graphics and not a bitmap. If you need help, please write to graphics@msp.org with details about how your graphics were generated.

White space. Forced line breaks or page breaks should not be inserted in the document. There is no point in your trying to optimize line and page breaks in the original manuscript. The manuscript will be reformatted to use the journal's preferred fonts and layout.

Proofs. Page proofs will be made available to authors (or to the designated corresponding author) at a Web site in PDF format. Failure to acknowledge the receipt of proofs or to return corrections within the requested deadline may cause publication to be postponed.

Algebra & Number Theory

Volume 9 No. 5 2015

$p\mbox{-}adic$ Hodge-theoretic properties of étale cohomology with mod p coefficients, and the cohomology of Shimura varieties MATTHEW EMERTON and TOBY GEE	1035
Homotopy exact sequences and orbifolds KENTARO MITSUI	1089
Factorially closed subrings of commutative rings SAGNIK CHAKRABORTY, RAJENDRA VASANT GURJAR and MASAYOSHI MIYANISHI	1137
Coherent analogues of matrix factorizations and relative singularity categories ALEXANDER I. EFIMOV and LEONID POSITSELSKI	1159

